Symmetry Analysis and Exact Solutions of Equations of Nonlinear Mathematical Physics

Managing Editor:

M. HAZEWINKEL Centre for Mathematics and Computer Science, Amsterdam, The Netherlands

Symmetry Analysis and Exact Solutions of Equations of Nonlinear Mathematical Physics

W. I. Fushchich, W. M. Shtelen and N. I. Serov Institute of Mathematics,

Institute of Mathematics Kiev, Ukraine



SPRINGER-SCIENCE+BUSINESS MEDIA, B.V.

ISBN 978-90-481-4244-6 ISBN 978-94-017-3198-0 (eBook) DOI 10.1007/978-94-017-3198-0

Printed on acid-free paper

This is an updated, revised and extended translation from the Russian original work Symmetry Analysis and Exact Solutions of Equations of Nonlinear Mathematical Physics, Nauka, Kiev, © 1989

All Rights Reserved © 1993 Springer Science+Business Media Dordrecht Originally published by Kluwer Academic Publishers in 1993 No part of the material protected by this copyright notice may be reproduced or utilized in any form or by any means, electronic or mechanical, including photocopying, recording or by any information storage and retrieval system, without written permission from the copyright owner.

Dedicated to the 150th anniversary of the birthday of the genius Norwegian mathematician Marius Sophus Lie (1842–1899).

We must be grateful to God that he created the world in such a way that everything simple is true and everything complicated is untrue.

> Gregory Skovoroda (an 18th-century Ukrainian philosopher)

Table of Contents

Preface	xiii
Preface to the English Edition	xv
Introduction	xvii

Chapter 1. Poincare Invariant Nonlinear Scalar Equations

1.1.	Nonlinear <i>n</i> -dimensional wave equations invariant under the groups	
	$\tilde{P}(1, n-1), \tilde{P}(1, n) \dots $	1
1.2.	The point and tangent symmetry of the relativistic Hamilton equation	5
1.3.	The polywave equations invariant under the conformal group	10
1.4.	Ansätze and reduction of PDEs. The extended Poincare group $\widetilde{P}(1,2)$	
	and its invariants	16
1.5.	Reduction and exact solutions of the nonlinear d'Alembert equation	21
1.6.	Reduction and solutions of the Liouville equation	28
1.7.	Reduction and solutions of d'Alembert equation with nonlinearities	
	$\sin u$, $\sin u$	32
1.8.	Solutions of eikonal equations	35
1.9.	Symmetry and exact solutions of the Euler-Lagrange-Born-Infeld	
	equation	41
1.10.	Symmetry and exact solutions of the Monge-Ampere equation	47
1.11.	Symmetry of the scalar wave equation with interaction	51

Chapter 2. Poincare-Invariant Systems of Nonlinear PDEs

2.1.	Reduction and exact solutions of the nonlinear massive Dirac equa-	
	tion	55
2.2.	Reduction and exact solutions of the nonlinear massless Dirac equa-	
	tion	70
2.3.	Conformal symmetry and formula of generating solutions for fields	

Table of Contents

	of arbitrary spin. C(1,3)-ungenerative ansätze	78
2.4.	Conformally invariant nonlinear equations for spinor fields and their	
	solutions	91
2.5.	Reduction and exact solutions of coupled nonlinear PDEs for spinor	
	and scalar fields	98
2.6.	Exact solutions of systems of nonlinear equations of quantum elec-	
	trodynamics	102
2.7.	On linearization and general solution of two-dimensional Dirac-Hei-	
	senberg-Thirring and quantum electrodynamics equations	112
2.8.	Symmetry analysis of nonlinear equations of classical electrodynamics	117
2.9.	Solutions of nonlinear equations for vector fields	126
2.10.	Some exact solutions of SU(2) Yang-Mills field theory	129
2.11.	On connection between solutions of Dirac and Maxwell equations,	
	dual Poincare invariance and superalgebras of invariance of Dirac	
	equation	141

Chapter 3. Euclid and Galilei Groups and Nonlinear PDEs for Scalar Fields

3.1. The second-order PDEs invariant under the extended Euclid group $\widetilde{E}(1, n-1)$	147
3.2. Reduction and exact solutions of nonlinear PDEs of the type	
$\Box u + F(u, u)u_0 = 0 \dots \dots \dots \dots \dots \dots \dots \dots \dots $	150
3.3. PDEs admitting Galilei algebras	163
3.4. The Galilean relativistic principle and nonlinear PDEs	169
3.5. Reduction and exact solutions of the nonlinear Schrödinger equation .	176
3.6. Symmetry and some exact solutions of the Hamilton-Jacobi equation .	182
3.7. Symmetry and some exact solutions of the Boussinesq equation	188
3.8. Symmetry properties of Fokker-Planck equations	193

Chapter 4. System of PDEs Invariant Under The Galilei Group

4.1.	The Schrödinger group $Sch(1,3)$: nonequivalent ansätze and final transformations for fields of arbitrary spin \ldots	199
4.2.	Linear and nonlinear systems of PDEs invariant under the Schrö-	
	dinger group $Sch(1,3)$	204
4.3.	Systems of second-order PDEs invariant under the Galilei group	215
4.4.	Solutions of the nonlinear Levi-Leblond equation	225
4.5.	Symmetry analysis of gas dynamics equations	229
4.6.	The Galilean invariant generalization of the Lame equations. The super-	
	algebra of symmetry of the Lame equations	250
4.7.	Reduction and exact solutions of nonlinear Galilei-invariant spinor	
	equations	254
4.8.	Reduction and exact solutions of the Navier-Stokes equation	260

Chapter 5. Some Special Questions

5.1. On nonlocal linearization of nonlinear equations	277
5.2. Symmetry, integrals of motion, and some partial solutions of the	
three-body problem	286
5.3. Non-Lie symmetry and nonlocal transformations	297
5.4. Lie-Backlund symmetry of the Dirac equation	310
5.5. Symmetry of integrodifferential equations	319
5.6. On exact and approximate solutions of the multidimensional Van der	000
Pol equation	323
5.7. Conditional symmetry of PDEs	320
5.8. Nonlocal symmetry of quasirelativistic wave equation	349
5.9. Are maxwell's equations invariant under Gamean transformations:	357
5.11. On solutions of the Schrödinger equation invariant under the Lor-	001
entz algebra	358
5.12. On approximate symmetry and approximate solution of the nonlin-	
ear wave equation with a small parameter	360
Appendix 1. Jacobi elliptic functions	364
Appendix 2	
$\widetilde{\mathbf{D}}(1,2)$ represent the dimensional subalgebras of the extended	
P(1,3)-nonequivalent one-dimensional subalgebras of the extended	260
Poincare algebra $AP(1,3)$	300
Appendix 3.	
Some applications of Campbell-Baker-Hausdorff operator calculus	373
Appendix 4.	
Differential invariants (DI) of Poincare algebras $AP(1,n)$, $AP(1,n)$ and	
conformal algebra $AC(1,n)$	387
Appendix 5.	
Differential invariants (DI) of Galilei algebras $AG(1,n)$, $A\widetilde{G}(1,n)$ and	
Schrödinger algebra $\operatorname{ASch}(1,n) \ldots \ldots \ldots \ldots \ldots \ldots$	389
Appendix 6.	
Compatibility and solutions of the overdetermined d'Alembert-Hamilton-	
system	393
Appendix 7. Q-conditional symmetry of the heat equation	396
Appendix 8. On nonlocal symmetries of nonlinear heat equation	403
References	411
Additional References	422
Index	121
пцех	491

Solving concrete problems an investigator enriches himself, and in this way he discovers new methods and extends his mental outlook. He who searches new methods without any specific problem in mind just wastes time for nothing.

- D. Hilbert

Preface

As far back as D. Bernoulli, Euler, Laplace, d'Alembert, Fourier, Lame, Riemann, Liouville, and many other scientists, the symmetry properties of differential equations (DEs) were used, although implicitly, for finding their exact solutions. But the mathematical foundations of the theory of the symmetry of DEs were developed only at the end of the last century by the eminent Norwegian mathematician Sophus Lie. He was the first to successfully apply this theory—the theory of continuous (Lie) groups—to specific equations.

In recent decades Lie's ideas and methods have become widespread. It is impossible to overestimate the importance of Lie's contribution to modern science and mathematics. At present there are many articles and several monographs devoted to the application and development of Lie's methods. To give a complete bibliography on the subject is impossible. Suffice it to say that more than half of the articles of such popular science journal as "Journal of Mathematical Physics" deal with group (algebraic) investigations which we shall call symmetry analysis. The applications of Lie's theory include such diverse fields as algebraic topology, differential geometry, invariant theory, bifurcation theory, special functions, numerical analysis, control theory, classical mechanics, quantum mechanics, relativity, continuum mechanics, and so on.

This book is devoted to the development and diverse applications of Lie's theory. The main object of investigation is multidimensional systems of nonlinear partial differential equations (PDEs) of hyperbolic and parabolic types which admit the Poincare group, the Schrödinger group, and some other important groups. A key question considered is the construction of exact solutions of nonlinear systems of PDEs by means of symmetry reduction. For this purpose we use the theory of Lie groups and Lie algebras, the theory of representations, subalgebraic structure of Lie algebras, special ansatze, and so on. The table of contents indicates the equations studied. The especially important role played by spinor (spin s = 1/2) field equations is emphasized because their solutions can be used for constructing solutions of other field equations insofar as fields with any spin may be constructed from spin s = 1/2 fields. A brief account of the main ideas of the book is presented in the Introduction.

The book is largely based on the authors' works $[55-109, 176-189, 13-16, 7^*-14^*, 23^*, 24^*]$ carried out in the Institute of Mathematics, Academy of Sciences of the Ukraine. References to other sources is not intended to imply completeness. As a rule, only those works used directly are cited.

The authors wish to express their gratitude to Academician Yu.A. Mitropolsky, and to Academician of Academy of Sciences of the Ukraine O.S. Parasyuk, for basic support and stimulation over the course of many years; to our coworkers in the Department of Applied Studies, I.A. Egorchenko, R.Z. Zhdanov, A.G. Nikitin, I.V. Revenko, V.I. Lagno, and I.M. Tsifra for assistance with the manuscript.

Ukraine, Kiev, October 1988

The Authors

Preface to the English Edition

For the English edition of our book Symmetry Analysis and Exact Solutions of Nonlinear Equations of Mathematical Physics, published in Russian in Kiev (Ukraine) in 1989 we have specially prepared some new paragraphs which naturally supplement and extend the basis text. The new paragraphs deal with solutions of the Navier-Stokes equations, the Fokker-Planck equations, conditional and approximate symmetry, nonlocal Galilei invariance of Maxwell equations, dual Poincare invariance and connection between Dirac and Maxwell equations, solutions of the Schrödinger equation invariant under the Lorentz algebra and some other topics. This, we hope, makes the text more interesting and useful.

Nowadays there arise new effective and constructive methods of mathemnatical description and analysis of nonlinear processes. The collection of these methods can be considered as new scientific direction—nonlinear mathematical physics (NMP). The main distinguishing feature of NMP is nonfulfillment of classical superposition principle. This means that the majority of methods of linear mathematical physics are useless in NMP. Of course, sources of NMP can be discerned in works of classics from Bernoulli, Euler, and especially from Sophus Lie and Poincare, yet only at the end of the 20th century NMP has become a separate, wide and diversified scientific branch with its specific problems and methods.

We are glad that on professor Michiel Hazewinkel's initiative our book is published by Kluwer. We would like to take this opportunity to thank him for his interest in our work. This book should be considered as a natural continuation and further development and application of the ideas and methods of our first book Symmetries of Maxwell's Equations (W.I.Fushchich and A.G.Nikitin, D.Reidel, 1987) to nonlinear equations of mathematical and theoretical physics. In this series of books on symmetry and its applications in mathematical physics, we have three other books written in Russian, which have been or are yet to be published: Symmetry of Quantum Mechanics Equations (W.I.Fushchich and A.G.Nikitin, Moscow: Nauka, 1990), Subgroup Analysis of the Galilei and Poincare Groups and Reduction of Nonlinear Equations (W.I.Fushchich, L.F.Barannik and A.F.Barannik, Kiev: Naukova Dumka, 1991), Nonlinear Spinor Equations: Symmetry and Exact Solutions (W.I.Fushchich and R.Z.Zhdanzov, Kiev: Naukova Dumka, 1992).

Introduction

Since the time of Newton the search for exact solutions of differential equations describing genuine physical phenomena has been the most important issue in the mathematical description of nature. During the past 300 and more years a great number of effective and elegant methods for solving DEs have been developed: the method of special substitutions, the method of separation of variables, the Poisson method, the method of Fourier series expansion, the saddle point method, the method of the inverse scattering transform, and so on. If one looks at these methods from the point of view of group theory, then one sees that they are essentially based on symmetry, and effectively solve those problems which actually possess explicit or implicit symmetry.

Sophus Lie advanced many fundamental ideas and worked out basic methods for studying group properties of DEs. He also obtained many concrete results of great importance in applied mathmatics. In particular he was the first to establish the maximal group of point (local) transformations admitted by the one-dimensional heat equation, and discovered the so-called projective representation of the Galilei group. These results were rediscovered only recently. The well-known Noether theorem on conserved laws is based on Lie's theory of continuous groups. Nowadays, in connection with the modern development of mathematical and theoretical physics, numerous results of Lie are being recognized and rediscovered, and we are witnessing the triumph of Lie theory throughout all of the mathematically based sciences.

A crucial point in the recognition of Lie theory is the fact, established for the first time by Poincare in 1905, that the Lorentz transformations, which leave Maxwell's equations invariant, form a Lie group. In 1909 Bateman [25] and Cunningham [43] established that Maxwell's equations are invariant with respect to the conformal group which includes the Lorentz group as a subgroup. Bateman utilized the symmetry of the linear wave equation to construct its solutions. Later on such solutions were called functionally invariant (V.I. Smirnov and S.L. Sobolev, 1932). Some important ideas on searching for exact solutions of PDEs were suggested by H. Birkhoff [29]. A great number of exact solutions of two-dimensional nonlinear PDEs are given in books of Forsyth [54] and Ames [7]. In Kiev, Lie's methods were developed by V.P. Ermakov (1890–1900), G.V. Pfeifer (1920–1935), and M.K. Kurensky (1930).

In the post-war era of the USSR the first papers on the symmetry of PDEs were published by L.V. Ovsyannikov (1958) (see [161]) and by V.G. Kostenko (1959) [135]. The modern treatment of Lie's theory is given in monographs [27, 123, 124, 159, 161, 162].

It should be noted that Poincare was the first to suggest using the group theory approach for the construction and analysis of a physical theory. Today, symmetry principles are guiding principles in theoretical and mathematical physics. They are often used as selection rules, allowing one to choose from a set of mathematically permissible models (equations) those which possess the desirable properties. Sometimes such symmetry selection leads to a unique equation. For example, among the set of linear systems of PDEs for the two vector functions $\vec{E}(x)$ and $\vec{H}(x)$ there is only one system that is invariant under the Poincare group P(1,3) (P(1,3) is the ten-parameter group that includes the Lorentz transformations and spacetime translations, which at Wigner's suggestion was named in honor of Poincare), namely Maxwell's equations [82]. Analogous properties have been found for many of the basic linear and nonlinear PDEs of theoretical and mathematical physics [66]. It should be noted that some nonlinear PDEs have much wider symmetry than any linear ones. For example, equations such as the Monge-Ampere equation, and the Hamilton-Jacobi and relativistic Hamilton (eikonal) equations, admit such wide groups of transformations which do not admit (and cannot admit in principle) any linear PDE.

From a mathematical point of view it is important to know the maximal (in a certain sense) group of invariance of a given PDE. Of special value is information that provides knowledge of nonlinear invariance transformations that allow the construction of highly nontrivial formulae for generating solutions which yield new families of solutions starting from a known and often trivial solution. Evidently the first result that should be considered important in this context is the discovery of the conformal invariance of Maxwell's equations made by Bateman [25] and Cunningham [43] in 1909.

In [57] it was noted that the Lie approach is highly restricted. The fact is that it falls far short of providing the possibility of finding all symmetries which a system of DEs possesses in as much as Lie symmetry generators, which are always differential operators of first order, are far from being the only symmetry generators. Using the non-Lie approach recently developed [55, 57–61], new invariance algebras (IAs) of Maxwell's equations, the Dirac equation, and many other relativistic as well as nonrelativistic PDEs have been obtained. The basis elements of such IAs are differential operators of any order and even integrodifferential ones. A largely complete review of non-Lie symmetries obtained within the framework of the non-Lie approach is given in reference [82]. We often use the term "non-Lie symmetry," introduced in [61, 63, 64], in order to emphasize that this symmetry cannot be obtained within the framework of classical Lie's method.

Investigation of symmetry and the construction of exact solutions of nonlinear PDEs is the major but not the only task of the present book. In the 5th chapter we consider some related questions.

For studying local (point) symmetry of PDEs we use Lie's algorithm [161,159]. But in many cases, especially for multicomponent systems of PDEs, the direct application of the standard Lie algorithm is conjugated with rather cumbersome calculations. To simplify these cases we shall do the following. Consider an arbitrary nonlinear system of PDEs written in the following symbolic form:

$$L\left(x,\psi(x)\right) = 0,\tag{1}$$

where ψ is a multicomponent function with components $\psi = \{\psi^1, \ldots, \psi^m\}$, and $x \in \mathbb{R}^n$. In the Lie approach the infinitesimal operators (IFOs) of the invariance algebra (IA) of Equation (1) are sought in the form

$$X = \xi^{\mu}(x,\psi)\frac{\partial}{\partial x^{\mu}} + \eta^{k}(x,\psi)\frac{\partial}{\partial \psi^{k}}, \quad \begin{array}{l} \mu = \overline{0,n-1} \\ k = \overline{1,m} \end{array}$$
(2)

where functions ξ^{μ} and η^{k} are determined from the invariance condition

$$\left. \begin{array}{c} XL\\s \end{array} \right|_{L=0} = 0, \tag{3}$$

 X_s is the sth prolongation of the operator (2) which is defined according to the Lie formulae [159, 161]:

 $\nu, \nu_1, \ldots, \nu_s = \overline{0, n-1}$; s is the order of the PDE in question; D_{ν} is the total derivative operator,

$$D_{\nu} = \frac{\partial}{\partial x^{\nu}} + \psi_{\nu}^{k} \frac{\partial}{\partial \psi^{k}} + \psi_{\nu\nu_{1}}^{k} \frac{\partial}{\partial \psi_{\nu_{1}}^{k}} + \dots; \qquad (5)$$

$$\psi_{\nu}^{k} = \frac{\partial \psi^{k}}{\partial x^{\nu}}, \qquad \psi_{\nu\nu_{1}}^{k} = \frac{\partial^{2} \psi^{k}}{\partial x^{\nu} \partial x^{\nu_{1}}}, \quad \text{and so on.}$$

Invariance condition (3) results in a linear system of PDEs, the so-called defining equations, the general solution of which determines the maximal in the Lie sense IA.

If we deal with a linear system of PDEs

$$L(x,\partial)\psi(x) = 0, \tag{6}$$

where L is a linear operator, we can essentially simplify the above algorithm. Since the space of solutions of any linear equation is also linear, then the symmetry operators as well as the invariance transformations may also be only linear in ψ . So we can write the general form of the Lie IFOs in this case as

$$Q = \xi^{\mu}(x)\partial_{\mu} + \eta(x), \tag{7}$$

where $\partial_{\mu} \equiv \frac{\partial}{\partial x^{\mu}}$, and $\eta(x)$ is an $m \times m$ matrix. Note that operator (7) operates in the linear space $\{\psi(x)\}$ of solutions of Equation (6) while its counterpart of the form (2)

$$X = \xi^{\mu}(x)\partial_{\mu} - (\eta(x)\psi)^{k}\frac{\partial}{\partial\psi^{k}}$$
(8)

operates on the manifold $\{(x, \psi)\}$. There is a further point to be made here. According to the Lie algorithm, dependent and independent variables enter IFOs (2) equally and this circumstance deprives us of the possibility of effectively using the operator (matrix) calculus within the framework of the standard Lie algorithm.

It should also be noted that the most general form of IFOs generating linear transformations is

$$X = \xi^{\mu}(x)\partial_{\mu} - \left(\eta(x)\psi + \beta(x)\right)^{k} \frac{\partial}{\partial\psi^{k}}$$
(8')

where $\beta(x)$ is an *m*-component function. In the case of linear PDEs (6) $\beta(x)$ is just an arbitrary solution of the equation. Evidently all the essential

information on point symmetry of the system (6) is contained in operators of the form (7) or (8).

By means of operator (7) the invariance condition of the system (6) can be written as

$$LQ\psi(x) \bigg|_{L\psi(x)=0} = 0.$$
(9)

It clearly means that operator Q transforms solutions of the system to other solutions, that is, its action on the set of solutions does not go out of this set. The condition (9) can be rewritten equivalently as

$$[L,Q]\psi \equiv (LQ - QL)\psi = 0,$$
(10)

or

$$[L,Q] = \lambda(x)L,\tag{11}$$

where $\lambda(x)$ is a certain $m \times m$ matrix. (Generalization of conditions (9)-(11); see Paragraphs 5.3, 5.7.) The general solution of the Equation (11) leads to the maximal (in sense of Lie) IA of the Equation (6). As one can see, using the algorithm based on Equation (11) is much simpler than using that of Lie based on Equations (2), (3).

Now consider the system of nonlinear PDEs

$$L(x,\partial)\psi(x) + F(x,\psi) = 0, \qquad (12)$$

where $F(x, \psi)$ is a smooth *m*-component function depending on x and ψ . If relation (11) still holds for the IFOs (7) and the linear operator L of Equation (12), and, in addition, the following equality is fulfilled (see paragraph 5.5)

$$\xi^{\mu}(x)\frac{\partial F}{\partial x^{\mu}} - (\eta\psi)^{k}\frac{\partial F}{\partial\psi^{k}} + (\lambda(x) + \eta(x))F = 0, \qquad (13)$$

then our nonlinear system (12) is invariant under the IFO (7). The general solutions of Equations (11) and (13) determine the maximal IA of the system (12) in the class of operators (7). It will be noted that, generally speaking, the maximal, in the Lie sense, IA of the nonlinear PDEs (12) is apparently determined by IFOs of the form (8') (for an example, see system (2.8.35)).

The greater part of the present book is devoted to the construction of exact solutions of nonlinear systems of PDEs. The method we are using is explained in Paragraphs 1.4 and 2.1. It is based on the representation of a solution in a special form called ansatz, which is defined as a rule by means of symmetry operators. The generalization of this method is given in Paragraph 5.7, where the concept of conditional invariance is introduced.

So, if an equation (linear or nonlinear) admits an IFO of the type (7), then its solutions can be sought in the form [63]

$$\psi(x) = A(x)\phi(x),\tag{14}$$

where the $m \times m$ matrix A(x) and new variables $\omega = \omega(x)$ are determined from the equations [95]

$$QA(x) \equiv (\xi^{\mu}(x)\partial_{\mu} + \eta(x))A(x) = 0$$

$$\xi^{\mu}(x)\partial_{\mu}\omega(x) = 0.$$
(15)

The ansatz (14) reduces the initial system of PDEs to a system of PDEs with a smaller number of independent variables for the function $\phi(\omega)$. The idea of using the symmetry of equations in searching for their solutions goes back to Lie, and in the past 30 years was considerably advanced by H. Birkhoff [29], Ovsyannikov [161, 162], Ibragimov [123, 124] and others [7, 10, 11, 27, 159]. Numerous important results have been obtained by Winternitz with collaborators [114, 25^{*}, 38^{*}, 39^{*}, 85^{*}].

Symmetry transformations can be used to construct new solutions from known solutions. In particular, if the transformations

$$x_{\mu} \to x'_{\mu} = f_{\mu}(x,\theta),$$

$$\psi(x) \to \psi'(x') = R(x,\theta)\psi(x)$$
(16)

 $(R(x,\theta)$ is a certain nonsingular matrix, θ are group parameters) leave a given PDE invariant, then the function

$$\psi_{II}(x) = R^{-1}(x,\theta)\psi_I(x') \tag{17}$$

will be a solution of the equation as soon as $\psi_I(x)$ is a solution of the equation. Expressions like (17) shall be called formulae of generating solutions [95]. It is appropriate to note that in connection with the operation of generating solutions the concept of G-ungenerative solutions naturally arises [65]. So, a family of solutions is called G-ungenerative if it is transformed into itself when it is subjected to the above-described group multiplication by means of all transformations of the group G. In this case the action of the operation of generating solutions is reduced to transformations of group parameters. (For more details see Paragraphs 2.3 and 4.1. Nonlocal transformations are considered in Paragraph 5.3.)

In discussing exact solutions of DEs one cannot fail to note two more methods which have been intensively developed recently, namely the method of separation of variables [11, 153] and the method of the inverse scattering transform [1, 154, 210]. The basic idea of monograph [153] is that the systems of coordinates which admit separation of variables for linear second-order PDEs can

xxii

be described by means of sets of second-order differential operators $Q^{(2)}$. Solutions with separable variables are just eigenfunctions of operators $Q^{(2)}$ and the constant of separation k^2 is the corresponding eigenvalue, that is

$$Q^{(2)}\psi_k = k^2\psi_k.$$
 (18)

Comparing this relation with the conditions in (15), one easily sees that it is very similar to that of (15). Indeed, (18) can be rewritten as

$$\widetilde{Q}\psi_k = 0, \quad \widetilde{Q} = Q^{(2)} - k^2 I \tag{19}$$

and therefore the solutions in (14) and solutions with separable variables ψ_k are invariant functions of the symmetry operators (7) and (19), respectively. Now it is clear why the method of separation of variables in its standard formulation is inapplicable to nonlinear PDEs. The point is that, firstly, nonlinear PDEs do not admit, as a rule, the unit operator I, and secondly, they may not be invariant under the second-order differential operators even though the latter belong to the enveloping algebra of the first-order symmetry operators.

It seems to us that the most natural way of describing separation of variables can be made in the framework of conditional invariance which is considered in Paragraph 5.7. For example, the classical variants of separation of variables, multiplicative and additive

$$\psi(x,t) = \phi(x)\chi(t), \quad \psi(x,t) = \phi(x) + \chi(t)$$

can be represented in the form of additional differential equations

$$\psi\psi_{xt}=\psi_x\psi_t,\quad\psi_{xt}=0,$$

which may be added to the original equation. Similar ideas were discussed in [160].

The term and concept conditional symmetry (or conditional invariance) was suggested by one of us $[67, 82, 7^*, 59^*]$ and represents by itself a natural generalization of classical Lie concept on invariance of differential equations. In general terms, the idea of this generalization consists in joining some additional equation(s)

$$L_1(x,\psi(x)) = 0$$
 (20)

to the equation in question (1). Equation (20) is constructed so as to enlarge or just to change the symmetry of the basic Equation (1). For example, the maximal invariance algebra, in the Lie sense, of Maxwell equations without additional conditions div $\vec{E} = \text{div } \vec{H} = 0$ is the 10-dimensional Lie algebra which does not contain generators of Lorentz and conformal transformations [82, §42]. And only the full system of Maxwell equations, which is overdetermined, possesses the 17-dimensional invariance algebra which includes the Poincare and conformal algebras. Another example of additional condition (20) is the equation $[7^*, 59^*]$

$$[a^{\mu\nu}(x, u, \underbrace{u}_{1})J_{\mu\nu} + b^{\mu}(x, u, \underbrace{u}_{1})P_{\mu} + c^{\mu}(x, u, \underbrace{u}_{1})K_{\mu} + r(x, u, \underbrace{u}_{1})D]u = F(x, u, \underbrace{u}_{1}),$$
(21)

where $a^{\mu\nu}$, b^{μ} , c^{μ} , r, F are smooth functions of $x \in \mathbb{R}(1,n)$, $u, u = \{\partial u/\partial x^{\mu}\}$; $J_{\mu\nu}, P_{\mu}, K_{\mu}, D$ are basis elements of the conformal algebra (see §2.3). The particular case when elements of Equation (21) are arbitrary constants, solutions of Equation (1), satisfying (21), may coincide with those obtained by Lie's method. Symmetry of Equation (1) under the additional condition (21) we called conditional symmetry [82, 7*, 59*]. Numerous results on the study of conditional symmetry of nonlinear equations of mathematical physics are contained in [44*, 45*, 56*-60*, 82*, 83*, 90*-110*] where, in particular, non-Lie ansatze are constructed which reduce PDEs to ODEs.

The simplest example of an additional condition for nonlinear wave equation

$$\Box u = F(u) \tag{22}$$

is the equation

$$\frac{\partial u}{\partial x^{\mu}}\frac{\partial u}{\partial x_{\mu}} = \begin{cases} \pm 1, \\ 0. \end{cases}$$
(23)

Equation (23) is a particular case of Equation (21). Indeed, letting $a_{\mu\nu} = 0$, $c_{\mu} = r = 0$, $F = \{0, \pm 1\}$, $b_{\mu} = \partial u / \partial x^{\mu}$ in (21) we get (23). Equation (23) has an important property: it possesses a wider symmetry than Equation (22), which is why it can be used effectively to define new (non-Lie) ansatze for Equation (22). The general solution of system (22), (23) is obtained in [60^{*}, 130^{*}] (See Appendix 6).

Let us briefly note the method of the inverse scattering transform. In spite of numerous important results obtained within the framework of this method, it remains as yet applicable mainly to one-dimensional nonlinear PDEs of special form. The generalization to multidimensional cases faces difficulties which, unfortunately, have not yet been overcome.

In conclusion let us make one final remark. In many cases we give symmetry IFOs multiplied by *i*. This is done to make the operators Hermitian. But, since the independent variables considered are always real, all calculations should be done with real coordinates ξ^{μ} and η . This is especially important for nonlinear PDEs.

xxiv

Poincare-Invariant Nonlinear Scalar Equations

In the present chapter we describe the first- and second-order *n*-dimensional nonlinear PDEs which are invariant under the groups $\tilde{P}(1, n-1)$, $\tilde{P}(1, n)$. We investigate local and tangent symmetry of the relativistic Hamilton equation, of the nonlinear d'Alembert equation, of the Euler-Lagrange-Born-Infeld equation, the Monge-Ampere equation, and some other PDEs. For this purpose the Lie method has been used with the exception of Sec. 1.3, where the symmetry of the polywave equation is investigated by the operator method expounded in Sec. 5.5.

Sec. 1.4 is devoted to the description of the method of the construction of exact solutions for nonlinear PDEs, which is applied further to the abovementioned equations and in the following chapters to systems of nonlinear PDEs.

1.1. Nonlinear n-dimensional wave equations invariant under the groups $\widetilde{P}(1, n-1), \widetilde{P}(1, n)$

1. Let us solve the following problem: to describe all nonlinear equations of the form

$$L(x, u(x)) \equiv \Box u + F(x, u) = 0, \tag{1.1.1}$$

where F(x, u) is some smooth function of x and u, invariant under the extended Poincare group which is generated by the operators

$$P_{0} = \partial_{0} \equiv \frac{\partial}{\partial x_{0}}, \quad P_{a} = -\partial_{a} \equiv -\frac{\partial}{\partial x_{a}}, \quad a = \overline{1, n-1};$$

$$J_{\mu\nu} = x_{\mu}P_{\nu} - x_{\nu}P_{\mu}; \quad \mu, \nu = \overline{0, n-1};$$
 (1.1.2)

$$D = x^{\nu} P_{\nu} + \eta(u) \frac{\partial}{\partial u}, \qquad (1.1.3)$$

 $(\eta(u)$ is an arbitrary differentiable function) which satisfy the commutation relations

$$[P_{\mu}, P_{\nu}] = 0, \quad [P_{\sigma}, J_{\mu\nu}] = g_{\sigma\mu}P_{\nu} - g_{\sigma\nu}P_{\mu}, [J_{\mu\nu}, J_{P\sigma}] = g_{\nu\rho}J_{\mu\sigma} + g_{\mu\sigma}J_{\nu\rho} - g_{\mu\rho}J_{\nu\sigma} + g_{\nu\sigma}J_{\mu\rho},$$
(1.1.4)
 $[P_{\mu}, D] = P_{\mu}, \quad [J_{\mu\nu}, D] = 0.$

Theorem 1.1.1. [88] Equation (1.1.1) is invariant under the group $\widetilde{P}(1, n-1)$ iff

$$F(x,u) = \lambda u^k, \quad k \neq 1 \tag{1.1.5}$$

or

$$F(x,u) = \lambda_1 e^{\lambda_2 u}, \quad \lambda_2 \neq 0 \tag{1.1.6}$$

 $(\lambda, \lambda_1, \lambda_2, k \text{ are arbitrary constants}).$

Proof. The necessary and sufficient condition of the invariance of Equation (1.1.1) under the group G according to S.Lie [161] is the condition (3). To write it explicitly it is necessary to construct, via formulae (4) and (5), the second prolongation of infinitesimal operators (IFOs) (2). For the scalar equation the twice-prolonged operator has the form

$$\begin{split} & \sum_{2}^{X} = \xi^{\mu}(x,u)\partial_{\mu} + \eta(x,u)\partial_{u} + \tau_{\mu}\partial_{u_{\mu}} + \tau_{\mu\nu}\partial_{u_{\mu\nu}}, \\ & \tau_{\mu} = \eta_{\mu} + u_{\mu}\eta_{u} - u_{\nu}\xi^{\nu}_{\mu} - u_{\mu}u_{\nu}\xi^{\nu}_{u}, \\ & \tau_{\mu\nu} = \eta_{\mu\nu} + u_{\mu}\eta_{\nu u} + u_{\nu}\eta_{\mu u} + u_{\mu}u_{\nu}\eta_{u u} + u_{\mu\nu}\eta_{u} - \\ & - u_{\mu\sigma}\xi^{\sigma}_{\nu} - u_{\nu\sigma}\xi^{\sigma}_{\mu} - u_{\mu\sigma}u_{\nu}\xi^{\sigma}_{u} - u_{\nu\sigma}u_{\mu}\xi^{\sigma}_{u} - u_{\mu\nu}u_{\sigma}\xi^{\sigma}_{u} - \\ & - u_{\sigma}\xi^{\sigma}_{\mu\nu} - u_{\mu}u_{\sigma}\xi^{\sigma}_{\nu u} - u_{\nu}u_{\sigma}\xi^{\sigma}_{\mu} - u_{\mu}u_{\nu}u_{\sigma}\xi^{\sigma}_{uu}, \end{split}$$
(1.1.7)

where $u_{\mu} \equiv \frac{\partial u}{\partial x_{\mu}}, \ u_{\mu\nu} \equiv \frac{\partial^2 u}{\partial x_{\mu} \partial x_{\nu}}, \ \eta_{\nu} \equiv \frac{\partial \eta}{\partial x_{\nu}}, \ \xi_{\nu}^{\sigma} \equiv \frac{\partial \xi^{\sigma}}{\partial x_{\nu}}$, and so on.

If we take as IFOs the linear combination of the operators (1.1.2), (1.1.3), i.e.,

$$X = (C_{\mu\nu}x_{\nu} + dx_{\mu} + a_{\mu})\frac{\partial}{\partial x_{\mu}} + d \cdot \eta(u)\frac{\partial}{\partial u}$$

where $C_{\mu\nu} = -C_{\nu\mu}$, d, a_{μ} are arbitrary constants; then from the invariance condition (3) we shall get the defining equations for the functions $\eta(u)$ and F(x, u)

$$\eta_{uu} = 0, \quad \partial_{\mu}F(x,u) = 0, \quad (\eta_u - 2)F = \eta \frac{\partial F}{\partial u},$$

Section 1

the general solution of which has the form $\eta(u) = au + b$, where a and b are arbitrary constants,

$$F(x,u) = \begin{cases} \lambda(au+b)^{(a-2)/a}, & a \neq 0, 2, \\ \lambda_1 \exp\{-\frac{2}{b}u\}, & b \neq 0. \end{cases}$$
(1.1.8)

Thus, in the class of Equations (1.1.1) there are only two essentially different equations that are invariant under the group $\tilde{P}(1, n-1)$ which correspond to the nonlinearities (1.1.5), (1.1.6)

$$\Box u + \lambda u^k = 0, \quad k \neq 1, \tag{1.1.9}$$

and

$$\Box u + \lambda_1 e^u = 0. \tag{1.1.10}$$

Remark 1.1.1. By using S.Lie's method it is easy to show that the group $\widetilde{P}(1, n-1)$, n > 2 is the maximal invariance group of Equations (1.1.9), (1.1.10). The basis IFOs of the corresponding Lie algebra have the form (1.1.2), (1.1.3), $\eta(u)$ having the form $(\frac{-2}{k-1})u$ for Equation (1.1.9) and $\eta(u) = -2$ for Equation (1.1.10).

The multiparameter families of exact solutions of Equations (1.1.9), (1.1.10) are constructed in §§1.5, 1.6.

2. Studying group properties of the relativistic Hamilton equation

$$u_{\mu}u^{\mu} \equiv \frac{\partial u}{\partial x^{\mu}} \frac{\partial u}{\partial x_{\mu}} = 1, \qquad \mu = \overline{0,3}, \qquad (1.1.11)$$

we discovered [92] that its invariance group is much wider than $\tilde{P}(1,3)$ and in particular contains the $\tilde{P}(1,4)$ group (for more details on the symmetry of (1.1.11) see §1.2) the action of which is defined in 5-dimensional Poincare-Minkowsky space R(1,4) with coordinates $x = (x_0, x_1, x_2, x_3, x_4 \equiv u)$. In this connection it is natural to consider the problem of describing *n*-dimensional first- and second-order PDEs

$$L_1(x, u, u) = 0, (1.1.12)$$

$$L_2(x, u, u, u) = 0, (1.1.13)$$

which are invariant under the group $\tilde{P}(1,n)$. The solution of this problem gives

Theorem 1.1.2. [86] Equation (1.1.12) is invariant under the Lie algebra $A\tilde{P}(1,n)$ whose basis elements are given by the operators (1.1.2), (1.1.14)

$$P_n = -\partial_u, \quad J_{n\mu} = x_n P_\mu - x_\mu P_n; \quad \mu = \overline{0, n-1},$$

$$D = x^\alpha P_\alpha, \quad \alpha = \overline{0, n}, \quad (x_n \equiv u),$$
(1.1.14)

iff it has the form (1.1.11).

Proof. Let us use Lie's algorithm. The invariance condition in this case takes the form

$$X_{1}L_{1} \Big|_{L_{1}=0} = \left(\xi^{\mu}(x,u)\partial_{\mu} + \eta(x,u)\partial_{u} + \tau_{\nu}\partial_{u_{\nu}}\right)L_{1} \Big|_{L_{1}=0} = 0. \quad (1.1.15)$$

where $\xi^{\alpha} = c^{\alpha\beta}x_{\beta} + d^{\alpha}$, $(\xi^{n} = \eta)$, $c^{\alpha\beta} = -c^{\beta\alpha}$, and τ_{ν} are constructed via formulas (1.1.7). From (1.1.15) we get the defining equations

$$\begin{aligned} \frac{\partial L_1}{\partial x_{\mu}} &= \frac{\partial L_1}{\partial u} = 0, \quad u_{\mu} \frac{\partial L_1}{\partial u_{\nu}} - u_{\nu} \frac{\partial L_1}{\partial u_{\mu}} = 0, \\ & \left(\frac{\partial L_1}{\partial u^{\mu}} - u_{\mu} u_{\nu} \frac{\partial L_1}{\partial u_{\nu}} \right) \Big|_{L_1 = 0} = 0, \end{aligned}$$

whose general solution is the function $L_1 = u^{\nu}u_{\nu} - 1$. Thus in the class of PDEs defined by (1.1.12) there exists only one equation, namely the *n*-dimensional relativistic Hamilton equation, which is invariant under the group P(1, n). The theorem is proved.

Theorem 1.1.3. The two-dimensional Equation (1.1.13) is invariant under the algebra $A\tilde{P}(1,2)$ if it has the form

$$\lambda_1 [(1 - u_\nu u^\nu) \Box u + u^\mu u^\nu u_{\mu\nu}] + \lambda_2 [(1 - u_\nu u^\nu) \det (u_{\mu\nu})]^{1/2} = 0, \quad (1.1.16)$$

where λ_1 , λ_2 are arbitrary constants, $\mu, \nu = 0, 1$.

Proof. Necessity. From the invariance condition

$$\left. \begin{array}{c} XL_2 \\ 2 \end{array} \right|_{L_2=0} = 0,$$

where X_2 is defined in (1.1.7), $\xi^A = c^{AB}x_B + d^A$, $c^{AB} = -c^{BA}$, $A, B = \overline{0, n}$ we get the following system of equations:

$$\frac{\partial L_2}{\partial x_{\mu}} = \frac{\partial L_2}{\partial u} = 0,$$

Section 2

$$\begin{split} \left[\left(u_{\sigma}u_{\nu} - g_{\sigma\nu} \right) & \frac{\partial L_2}{\partial u_{\nu}} + \left(u_{\nu}u_{\mu\sigma} + u_{\mu}u_{\nu\sigma} \right) & \frac{\partial L_2}{\partial x_{\mu\nu}} \right] \Big|_{L_2=0} = 0, \\ \left[u_{\mu} & \frac{\partial L_2}{\partial u_{\nu}} - u_{\nu} & \frac{\partial L_2}{\partial u_{\mu}} + \left(u_{\rho\mu}g^{\sigma\nu} + u_{\sigma\mu}g^{\rho\nu} - u_{\rho\nu}g^{\sigma\mu} - u_{\sigma\nu}g^{\rho\mu} \right) & \frac{\partial L_2}{\partial u_{\rho\sigma}} \right] \Big|_{L_2=0} = 0. \\ \left(u_{\rho\sigma} & \frac{\partial L_2}{\partial u_{\rho\sigma}} \right) \Big|_{L_2=0} = 0. \end{split}$$

We managed to find the general solution of this system when n = 2; it has the form (1.1.16).

Sufficiency. One can make sure by verification that Equation (1.1.16) is invariant under the algebra $A\tilde{P}(1,2)$.

From (1.1.16) we can single out three different equations that are invariant under the group $\tilde{P}(1,2)$: the Euler-Lagrange-Born-Infeld (ELBI) equation

$$(1 - u_{\nu}u^{\nu})\Box u + u^{\mu}u^{\nu}u_{\mu\nu} = 0, \qquad (1.1.17)$$

the Monge-Ampere (MA) equation

$$\det(u_{\mu\nu}) = 0. \tag{1.1.18}$$

and the eikonal equation (1.1.11).

When $n \geq 2$ the following statement holds true.

Theorem 1.1.4. [86] The equation

$$\lambda_1 [(1 - u_\nu u^\nu) \Box u + u^\mu u^\nu u_{\mu\nu}] + \lambda_2 [(1 - u_\nu u^\nu) \det (u_{\mu\nu})]^{\frac{3}{n+4}} = 0, \quad (1.1.19)$$

where $\mu, \nu = \overline{0, n-1}$ is invariant under the group $\widetilde{P}(1, n)$.

The proof is earried out by direct verification. When n = 2, Equation (1.1.19) coincides with (1.1.16). Using Lie's method we can show that $\tilde{P}(1,n)$ is the maximal symmetry group of Equation (1.1.19) [86].

1.2. The point and tangent symmetry of the relativistic Hamilton equation

1. The relativistic analogue of the classical Hamilton equation is the equation

$$u_{\nu}u^{\nu} \equiv \frac{\partial u}{\partial x^{\nu}}\frac{\partial u}{\partial x_{\nu}} = 1, \qquad (1.2.1)$$

In the preceeding paragraph we had shown that Equation (1.2.1) is invariant under the group $\tilde{P}(1,4) \supset \tilde{P}(1,3)$. It so happens that the point symmetry of Equation (1.2.1) is much wider than $\tilde{P}(1,4)$.

Theorem 1.2.1. [92] The maximal local invariance group of Equation (1.2.1) is 21-parameter conformal group C(1,4). The basis elements of the corresponding Lie algebra AC(1,4) have the form

$$\partial_{\alpha} = \frac{\partial}{\partial x_{\alpha}}, \quad J_{\alpha\beta} = x_{\alpha}\partial_{\beta} - x_{\beta}\partial_{\alpha}, \quad D = x^{\alpha}\partial_{\alpha},$$

$$K_{\alpha} = 2x_{\alpha}D - s^{2}\partial_{\alpha}, \quad s^{2} \equiv x^{\alpha}x_{\alpha} = x_{0}^{2} - x_{1}^{2} - x_{2}^{2} - x_{3}^{2} - u^{2}, \quad (1.2.2)$$

$$x^{(4)} = -x_{4} \equiv u, \quad x^{\alpha} = g^{\alpha\beta}x_{\beta}, \quad g^{\alpha\beta} = g_{\alpha\beta} = (1, -1, -1, -1, -1)\delta_{\alpha\beta}$$

(here and below the indices μ, ν range over 0,1,2,3 and the indices $\alpha, \beta, \rho, \sigma$ range over 0,1,2,3,4) and satisfy the commutation relations

$$\begin{aligned} [\partial_{\alpha}, \partial_{\beta}] &= 0, \quad [\partial_{\alpha}, J_{\beta\rho}] = g_{\alpha\beta}\partial_{\rho} - g_{\alpha\rho}\partial_{\beta}, \\ [J_{\alpha\beta}, J_{\rho\sigma}] &= g_{\alpha\sigma}J_{\beta\rho} + g_{\beta\rho}J_{\alpha\sigma} - g_{\alpha\rho}J_{\rho\sigma} - g_{\beta\sigma}J_{\alpha\rho}, \\ [\partial_{\alpha}, D] &= \partial_{\alpha}, \quad [J_{\alpha\beta}, D] = 0, \\ [K_{\alpha}, K_{\beta}] &= 0, \quad [K_{\alpha}, J_{\beta\rho}] = g_{\alpha\beta}K_{\rho} - g_{\alpha\rho}K_{\beta}, \\ [\partial_{\alpha}, K_{\beta}] &= 2(g_{\alpha\beta}D - J_{\alpha\beta}), \quad [D, K_{\alpha}] = K_{\alpha}. \end{aligned}$$

$$(1.2.3)$$

Proof. Let us use Lie's algorithm. From the invariance condition (3) which in the present case takes the form (see formulas (4), (1.1.7))

$$\left[\xi^{\mu}\partial_{\mu}+\eta\partial_{u}+\tau^{\mu}\partial_{u_{\mu}}\right]\left(u^{\nu}u_{\nu}-1\right)\bigg|_{u^{\nu}u_{\nu}=1}=0,$$

we get the system of defining equations

$$\frac{\partial \eta}{\partial x_{\nu}} = \frac{\partial \xi^{\nu}}{\partial u}, \quad \frac{\partial \xi^{0}}{\partial x^{0}} = \frac{\partial \xi^{1}}{\partial x^{1}} = \frac{\partial \xi^{2}}{\partial x^{2}} = \frac{\partial \xi^{3}}{\partial x^{3}} = \frac{\partial \eta}{\partial u} \stackrel{\text{def}}{=} f(x, u),$$
$$\frac{\partial \xi^{\mu}}{\partial x_{\nu}} + \frac{\partial \xi^{\nu}}{\partial x_{\mu}} = 0, \quad \mu \neq \nu$$

or, in the more compact form,

$$\xi_{\alpha,\beta} + \xi_{\beta,\alpha} = g_{\alpha\beta}f, \quad \left(\xi_{\alpha,\beta} \equiv \frac{\partial\xi^{\rho}}{\partial x^{\beta}}g_{\alpha\rho}\right).$$
 (1.2.4)

Equations (1.2.4) are known as Killing equations.

Having convoluted (1.2.4) with the metric tensor $g^{\alpha\beta}$ we get

$$\partial_{\alpha}\xi^{\alpha} = 2f$$

Section 2

Acting with the operator ∂^{β} on (1.2.4) we find

$$\partial_{\beta}\partial^{\beta}\xi_{\alpha} + \partial_{\alpha}\partial^{\beta}\xi_{\beta} = \partial_{\beta}\partial^{\beta}\xi_{\alpha} + 2\partial_{\alpha}f, \quad (\xi_{\alpha} = g_{\alpha\rho}\xi^{\rho}),$$

i.e.,

$$\partial_{\beta}\partial^{\beta}\xi^{\alpha} = -\partial^{\alpha}f. \tag{1.2.5}$$

Taking the divergence of (1.2.5) we get

$$\partial_{\beta}\partial^{\beta}f = 0. \tag{1.2.6}$$

After applying the operator $\partial_{\sigma}\partial^{\sigma}$ to (1.2.4) and using the equalities (1.2.5), (1.2.6) we find that

$$\partial_{\alpha}\partial_{\beta}f = 0.$$

Thus f can be only the linear function of x

$$f(x) = 2d + 4c_{\alpha}x^{\alpha}$$

Finally, we find that the general solution of Equation (1.2.4) has the form

$$\xi^{\alpha} = 2x^{\alpha}c_{\beta}x^{\beta} - c^{\alpha}x^{\beta}x_{\beta} + b^{\alpha\beta}x_{\beta} + dx^{\alpha} + a^{\alpha}, \qquad (1.2.7)$$

where c_{β} , $b_{\alpha\beta} = -b_{\beta\alpha}$, d, a_{α} are arbitrary constants.

From (1.2.7) we get the basis operators (1.2.3). By direct substitution we confirm that the operators (1.2.3) satisfy the commutational relations of the conformal algebra AC(1,4). The theorem is proved.

2. Let us study the invariance of Equation (1.2.1) under contact transformations, which are the natural generalization of the point transformations and the most closely related to them. We shall adduce some facts on contact transformations (for the more detailed information see the books [124, 161, 166, 168]).

Definition 1.2.1. The transformations of the form

$$\begin{aligned} x_{\nu} \to x_{\nu}' &= f_{\nu}(x, u, u, \theta), \quad \theta = \text{const}, \quad u_{1} = \left\{ \frac{\partial u}{\partial x_{\mu}} \right\}, \\ u \to u'(x') &= g(x, u, u, \theta), \\ u_{\nu} \to u_{\nu}' &= h(x, u, \theta, \theta) \end{aligned}$$
(1.2.8)

are called *contact* if they leave invariant the expressions

$$du - u_{\nu}dx_{\nu}, \tag{1.2.9}$$

or, in other words, the following relations hold true

$$u'_{
u} = rac{\partial u'(x')}{\partial x'_{
u}}$$

Relations (1.2.9) lead to quite strong restrictions on the coordinates of the generator of the contact transformations

$$X = \xi^{\mu}(x, u, \underline{u})\partial_{\mu} + \eta(x, u, \underline{u})\partial_{u} + \tau_{\nu}(x, u, \underline{u})\partial_{u_{\nu}}, \qquad (1.2.10)$$

namely

$$\xi^{\mu} = -\frac{\partial W}{\partial u_{\mu}}, \quad \eta = W - u_{\mu} \frac{\partial W}{\partial u_{\mu}},$$

$$\tau_{\mu} = D_{\mu} \eta - u_{\nu} D_{\mu} \xi^{\nu} = \frac{\partial W}{\partial x^{\mu}} + u_{\mu} \frac{\partial W}{\partial u},$$
 (1.2.11)

where W = W(x, u, u) is the characteristic function, and D_{μ} is the operator of full differentiation (5). Note that the more general statement is true: the group of the contact transformations coincides with the prolonged group of the point transformation if the dimension of the space of dependent variables u(x) is more than one, and only for scalar PDEs do these transformations have nontrivial coordinates of the corresponding IFOs having the form (1.2.11). The proof of this fact, known since the time of Lie, may be found in [124].

To find the explicit form of the finite transformations generated by the operators (1.2.10), (1.2.11), it is necessary to solve the following system of Lie's equations

$$\frac{\partial x'_{\nu}}{\partial \theta} = \xi^{\nu}(x', u', u') = -\frac{\partial W'}{\partial u'_{\nu}}, \quad x'_{\nu} \Big|_{\theta=0} = x_{\nu},$$

$$\frac{\partial u'}{\partial \theta} = \eta(x', u', u') = W' - u'_{\mu} \frac{\partial W'}{\partial u'_{\mu}}, \quad u' \Big|_{\theta=0} = u, \quad (1.2.12)$$

$$\frac{\partial u'_{\nu}}{\partial \theta} = \tau_{\nu}(x', u', u') = \frac{\partial W'}{\partial x'_{\nu}} + u'_{\nu} \frac{\partial W'}{\partial u'}, \quad u'_{\nu} \Big|_{\theta=0} = u_{\nu}.$$

The particular case of the contact transformations (1.2.7) are canonical or homogeneous contact transformations when g(x, u, u) = u, and f_{ν}, h_{ν} do not depend on u (this is equivalent to the case when the characteristic function Wdoes not depend on u). For the homogeneous contact transformations, Lie's Equations (1.2.12) have the Hamiltonian structure

$$\frac{\partial x'_{\nu}}{\partial \theta} = -\frac{\partial W}{\partial u'_{\nu}}, \quad \frac{\partial u'_{\nu}}{\partial \theta} = \frac{\partial W}{\partial x'_{\nu}}, \quad x'_{\nu} \Big|_{\theta=0} = x_{\nu}, \quad u'_{\nu} \Big|_{\theta=0} = u_{\nu}. \quad (1.2.13)$$

Section 2

Let us turn again to Equation (1.2.1).

Theorem 1.2.2. [187] The maximal invariance algebra of Equation (1.2.1) in the class of operators (1.2.10), (1.2.11) is infinite-dimensional and is given by the characteristic function

$$W = W(u_{\nu}, x_{\mu} - uu_{\mu}). \tag{1.2.14}$$

Proof. The necessary and sufficient condition of the invariance of Equation (1.2.1) under the contact transformations is the fulfillment of the conditions

$$\widetilde{X}(u_{\nu}u^{\nu}-1)\Big|_{u_{\nu}u^{\nu}=1}=0,$$

with the operator \widetilde{X} from (1.0.10), (1.2.11), or, at greater length

$$\left[u^{\nu}\left(\frac{\partial W}{\partial x^{\nu}}+u_{\nu}\frac{\partial W}{\partial u}\right)\right]\Big|_{u_{\nu}u^{\nu}=1}=u_{\nu}\frac{\partial W}{\partial x^{\nu}}+\frac{\partial W}{\partial u}=0, \quad u_{\nu}u^{\nu}=1. \quad (1.2.15)$$

The general solution of Equations (1.2.15) is defined in (1.2.14). The theorem is proved.

3. Let us consider the relativistic Hamilton equation with the nonlinear right-hand part

$$W^{\nu}W_{\nu} = f(W). \tag{1.2.16}$$

If $f(W) \neq 0$ then, having carried out the transformation

$$W \to u = \int_0^W \frac{dz}{\sqrt{|f(z)|}}$$

we get as a result Equation (1.2.1), or the equation

$$u_{\nu}u^{\nu} = -1. \tag{1.2.17}$$

It is not difficult to understand that the maximal group of the point transformations of Equation (1.2.17) is the conformal group C(2,3) whose generators have the form (1.2.3), but in this case $g_{\alpha\beta}$ has the signature (+ + - -).

If in (1.2.17) f(W) = 0 then we get the eikonal equation

$$u_{\nu}u^{\nu} = 0. \tag{1.2.18}$$

Theorem 1.2.3. The maximal (in Lie's sense) invariance group of Equation (1.2.18) is the infinite-dimensional group generated by the operators $C^{\infty}(1,3) \otimes G^{\infty}$,

$$X = (2x^{\mu}cx - c^{\mu}x^{2} + dx^{\mu} + b^{\mu\nu}x_{\nu} + a^{\mu})\partial_{\mu} + \eta\partial_{u}, \qquad (1.2.19)$$

where $c_{\mu}, b_{\mu\nu} = -b_{\nu\mu}, d, a_{\mu}, \eta$ are arbitrary differentiable functions of u.

We omit the proof because of its similarity to the proof of Theorem 1.2.1.

Theorem 1.2.4. The maximal invariance algebra of Equation (1.2.18) in the class of operators (1.2.10), (1.2.11) is infinite-dimensional and is defined by the characteristic function

$$W = W(u, u_{\nu}, x_0 u_a - x_a u_0), \quad (a = 1, 2, 3)$$

The proof is the same as in the case of Theorem 1.2.2.

1.3. The polywave equations invariant under the conformal group

Consider the equation

$$\Box^{l} u + F(x, u) = 0, \qquad (1.3.1)$$

where $\Box^{l} = \Box^{l-1}\Box$, $l = 1, 2, 3, ...; \Box = \partial_{\mu}\partial^{\mu}$, F(x, u) is a smooth function. We would like to describe all functions $F(x, u) \neq 0$, which ensure the conformal symmetry of Equation (1.3.1). When l > 1 we call Equation (1.3.1), according to [63], the nonlinear polywave equation.

It is well known that the maximal local invariance group of the wave equation $\Box u = 0$ is the conformal group C(1, n - 1) generated by the operators (1.1.2) and (1.3.2)

$$D = x^{\nu} P_{\nu} + k,$$

$$K_{\mu} = 2x_{\mu} D - x^{2} P_{\mu},$$
(1.3.2)

where k (called the conformal degree (see §2.3)) is equal to n/2 - 1.

Let us answer the question of whether the polywave equation

$$\Box^{l} u = 0 \tag{1.3.3}$$

has conformal symmetry.

Theorem 1.3.1. [176] The polywave Equation (1.3.3) is invariant under the group of conformal transformations C(1, n - 1) generated by the operators

Section 3

(1.1.2), (1.3.2), the conformal degree k being equal to k = n/2 - l (i.e., $D = D^{(l)} = x^{\nu} P_{\nu} + \frac{n}{2} - l$).

Proof. Let us write down the invariance condition of Equation (1.3.3) in the operator form [61,64] (see also §5.5):

$$[\Box^l, Q] = \lambda(x) \Box^l, \qquad (1.3.4)$$

where Q is one of the operators (1.1.2), (1.3.2); $\lambda(x)$ is some function. When l = 1 the conditions (1.3.4) have the form

$$[\Box, P_{\mu}] = [\Box, J_{\mu\nu}] = 0,$$
$$[\Box, D] = 2\Box, \quad [\Box, K_{\mu}] = 4x_{\mu}\Box$$

Now we shall use the method of mathematical induction. Let us suppose that when l = s - 1 the following equalities take place

$$[\Box^{s-1}, P_{\mu}] = [\Box^{s-1}, J_{\mu\nu}] = 0$$
$$[\Box^{s-1}, D^{(s-1)}] = 2(s-1)\Box^{s-1}, \qquad (1.3.5)$$
$$[\Box^{s-1}, K_{\mu}^{(s-1)}] = 4(s-1)x_{\mu}\Box^{s-1}$$

If we show that relations (1.3.5) hold true also when l = s then we shall have proven the theorem. Thus, we have

$$\begin{bmatrix} \Box^{s}, P_{\mu} \end{bmatrix} = \begin{bmatrix} \Box^{s}, J_{\mu\nu} \end{bmatrix} = 0$$
$$\begin{bmatrix} \Box^{s}, D^{(s)} \end{bmatrix} = \begin{bmatrix} \Box \cdot \Box^{s-1}, D^{(s-1)} - 1 \end{bmatrix} = \Box \begin{bmatrix} \Box^{s-1}, D^{(s-1)} \end{bmatrix} + \begin{bmatrix} \Box, D^{(s-1)} \end{bmatrix} \Box^{s-1} =$$
$$= 2(s-1)\Box^{s} + 2\Box^{s} = 2s\Box^{s};$$
$$\begin{bmatrix} \Box^{s}, K_{\mu}^{(s)} \end{bmatrix} = \begin{bmatrix} \Box \cdot \Box^{s-1}, K_{\mu}^{(s-1)} - 2x_{\mu} \end{bmatrix} = \Box \left(\begin{bmatrix} \Box^{s-1}, K_{\mu}^{(s-1)} \end{bmatrix} - 2[\Box^{s-1}, x_{\mu}] \right) + \left(\begin{bmatrix} \Box, K_{\mu}^{(s-1)} \end{bmatrix} - 2[\Box, x_{\mu}] \right) \Box^{s-1} = \Box \left(4(s-1)x_{\mu}\Box^{s-1} + 4(s-1)P_{\mu}\Box^{s-1} \right) + \left(4x_{\mu}\Box + 4(s-2)P_{\mu} + 4P_{\mu} \right) \Box^{s-1} =$$
$$= 4sx_{\mu}\Box^{s}.$$

Here we have made use of the equality

$$[\Box^{s}, x_{\mu}] = 2sP_{\mu}\Box^{s-1} \tag{1.3.6}$$

whose correctness may be easily verified.

Thus the theorem is proved (let us note that a more elegant proof may be derived using the formulas (A.3.5), (A.3.6)).

Consider the following question: what functions F(x, u) from (1.3.1) preserve the conformal symmetry of Equation (1.3.3) established in Theorem 1.3.1? When l = 1 the answer is well-known [123]:

$$F(x, u) = cu^{(n+2)/(n-2)}, \quad n \neq 2$$

(where c is an arbitrary constant).

When $l \ge 1$ the following statement holds true [176].

Theorem 1.3.2. Equation (1.3.1) is invariant under the conformal group C(1, n-1) iff

$$F(x,u) = cu^{(n+2l)/(n-2l)}, \quad n \neq 2l$$
(1.3.7)

Proof. Under the transformations generated by the operator (translations)

$$x_{\mu} \rightarrow x'_{\mu} = x_{\mu} + a_{\mu}, \quad u(x) \rightarrow u'(x') = u(x)$$

Equation (1.3.1) takes the form

$$\Box^l u = F(x+a, u).$$

Whence, due to the demand of invariance, we get

$$F(x+a,u) = F(x,u)$$

The solution of this functional relation is

$$F(x,u)=F(u)$$

Furthermore, we shall use Theorem 5.5.1, namely, the conditions (5.5.3) 1°, 2° which in our case have the form (1.3.4) and

$$[\eta(x) + \lambda(x)] F - \eta(x) u F \frac{\partial F}{\partial u} = 0.$$
 (1.3.8)

As it follows from Theorem 1.3.1: $\eta(x) = \lambda(x) = 0$ for the operators $P_{\mu}, J_{\mu\nu}$; $\eta(x) = \frac{n}{2} - l, \ \lambda(x) = 2l$ for the operator $D^{(l)}$ (1.3.3), and at last $\eta(x) = cx(n-2l), \lambda(x) = 4lcx$ for the operator $K = c^{\mu}K_{\mu}$.

Substituting these expressions into (1.3.8) we get

$$(n+2l)F - (n-2l)u\frac{\partial F}{\partial u} = 0$$
(1.3.9)

The general solution of this equation has the form (1.3.7). The theorem is proved.

Section 3

The more general result may be obtained if we suppose that formula (A.3.6) is true for an arbitrary differentiable operator-valued function $F(\hat{x})$. Then the following statement holds true.

Remark 1.3.1. When n = 2l Equation (1.3.1) is invariant under AC(1, n - 1) with basis elements (1.1.2) and

$$D = x^{\nu}P_{\nu} + 2l\partial_u, \qquad K_{\mu} = 2x_{\mu}D - x^2P_{\mu},$$

if it has the form

$$\Box^l u = \lambda e^u.$$

Theorem 1.3.3. A pseudodifferential nonlinear wave equation

$$\Box^r u = F(u) \tag{1.3.10}$$

(where r is an arbitrary number) is invariant under the conformal group C(1, n-1) iff

$$F(u) = cu^{(n+2r)/(n-2r)}, \qquad n \neq 2r.$$
 (1.3.11)

The basis IFOs of the corresponding Lie algebra have the form (1.1.2), (1.3.2), the conformal degree k being equal to

$$k = \frac{n}{2} - r$$
 (1.3.12)

Proof. Let us make use of the invariance conditions in the form (5.5.3) 1° , 2° which coincide with (1.3.4) and (1.3.8) when l = r. We have:

$$\eta(x) = \begin{cases} 0 & \text{for } P_{\mu}, J_{\mu\nu}; \\ k & \text{for } D; \\ 2kcx & \text{for } K = K_{\mu}c^{\mu} \end{cases}$$
(1.3.13)

It is evident that $[\Box^r, P_{\mu}] = 0$. Furthermore, via formula (A.3.6) we find

$$[\Box^r, J_{\mu\nu}] = [\Box, J_{\mu\nu}] r \Box^{r-1} + \cdots = 0,$$

$$[\Box^{r}, D] = [\Box, D]r\Box^{r-1} + \frac{1}{2!} [\Box, [\Box, D]]r(r-1)\Box^{r-2} + \dots = 2r\Box^{r},$$
$$[\Box^{r}, K] = [\Box, K]r\Box^{r-1} + \frac{1}{2!} [\Box, [\Box, K]]r(r-1)\Box^{r-2} + \dots =$$
$$= 4r \left(cx\Box^{r} + \left(r + k - \frac{n}{2} \right) c\partial\Box^{r-1} \right),$$

whence it can be seen that condition (1.3.4) is fulfilled if k has the form (1.3.12) with

$$\lambda(x) = \begin{cases} 0 & \text{for } P_{\mu}, J_{\mu\nu}; \\ 2r & \text{for } D; \\ 4rcx & \text{for } K = c^{\mu}K_{\mu}. \end{cases}$$
(1.3.14)

Having substituted (1.3.13) and (1.3.14) into (1.3.8), we shall obtain the equation for the function F(u) which coincides with (1.3.9) when l = r. The theorem is proved.

Corollary. The equation

$$\Box^r u + c u^q = 0 \tag{1.3.15}$$

is invariant under the conformal algebra C(1, n - 1) (1.3.2) iff the numbers r and q are connected with the following equality:

$$r+k-\frac{n}{2}=0$$

2r+k=kq,

whence, supposing q to be given, we find

$$r = rac{n}{2}rac{q-1}{q+1}, \quad k = rac{n}{q+1}$$

In particular, when n = 4 and q = 2 we get the nonlinear conformally invariant equation

$$\Box^{2/3}u = cu^2 \tag{1.3.16}$$

Note. The operator \Box^r , if r is a fractional number, is to be considered a pseudodifferential operator defined with the help of the integral Fourier transformation

$$\Box^{r} u = -\int \int (p_{\nu} p^{\nu})^{r} e^{i(x-y)p} u(y) dy dp, \qquad (1.3.17)$$

where $dy = dy_0 dy_1 dy_2 dy_3$, $dp = \frac{1}{(2\pi)^4} dp_0 dp_1 dp_2 dp_3$.

To conclude this paragraph let us prove the following statement:

Section 3

Theorem 1.3.4. If the equation

$$\Box u = F(x, u, u) \tag{1.3.18}$$

is invariant under the conformal group C(1, n - 1) then it is equivalent to the equation

$$\Box W + \lambda W^{\frac{n+2}{n-2}} = 0 \tag{1.3.19}$$

Proof. The invariance of Equation (1.3.18) under the algebra AC(1, n - 1) means that there exists the set of operators $P_{\mu}, J_{\mu\nu}, D, K_{\mu}$ satisfying the commutational relations of the algebra AC(1, n - 1) (see §2.3). The formulas (1.1.2) and (1.3.2) define the linear representation of the conformal algebra acting in the space of scalar functions u(x). Evidently the generalization of this representation is possible at the expense of altering the operator D. The most general form of the operator D which, when combined with $P_{\mu}, J_{\mu\nu}$ (1.1.2) and $K_{\mu} = 2x_{\mu}D - x^2P_{\mu}$, satisfies the commutational relations of the algebra AC(1, n - 1), has the form (1.1.3).

Note that if $\eta(u) \neq 0$ then it is possible to transform the function u

$$u \to w = \exp\left\{\frac{2-n}{2}\int \frac{du}{\eta(u)}\right\}, \quad \eta(u) \neq 0 \tag{1.3.20}$$

so that the scale transformations generator D (1.1.3) will change into D (1.3.2), $k = \frac{n}{2} - 1$. That is why we can further use the representation of the algebra AC(1, n - 1) in the form (1.1.2), (1.3.2) without loss of generality.

Evidently the function F from (1.3.18), due to relations $[\Box, P_{\mu}] = [\Box, J_{\mu\nu}] = 0$, has to be constructed of the absolute invariants of the Poincare group, and there are only two such invariants: u and $u_{\nu}u^{\nu}$.

By the action of the scale transformations

$$x_{\mu} \to x'_{\mu} = e^{\theta} x_{\mu}, \quad u(x) \to u'(x') = e^{-\theta(n/2-1)} u(x),$$
 (1.3.21)

generated by the operator D (1.3.2) we have

$$\Box u - F(u, u_{\nu}u^{\nu}) = 0 \to e^{-\theta(\frac{n}{2}+1)} \Box u - F\left(e^{-\theta(\frac{n}{2}-1)}u, e^{-n\theta}u_{\nu}u^{\nu}\right) = 0.$$

From the demand of invariance we get the functional relation for F

$$F(u, u_{\nu}u^{\nu}) = e^{\theta(\frac{n}{2}+1)}F\left(e^{-\theta(\frac{n}{2}-1)}u, e^{-n\theta}u_{\nu}u^{\nu}\right).$$

Differentiating F with respect to θ and then putting $\theta = 0$ we get the differential equation

$$(n+2)F - (n-2)u\frac{\partial F}{\partial u} - 2nv\frac{\partial F}{\partial v} = 0, \quad (v \equiv u_{\mu}u^{\mu}),$$

Chapter 1. Poincare-Invariant Scalar Equations

whose general solution has the form

$$F = u^{\frac{n+2}{n-2}} f\left(u(u_{\mu}u^{\mu})^{\frac{2-n}{2n}}\right), \qquad (1.3.22)$$

where f is an arbitrary differentiable function of the mentioned argument.

Using the explicit form of the conformal transformations (see §2.3) we can show that Equation (1.3.18) with the function f (1.3.22) remains invariant with respect to these transformations only when f = const.

In the case where the coordinate $\eta(u)$ of the operator D (1.1.3) is equal to zero, i.e., $D = x_{\nu}\rho^{\nu}$, it generates the transformations

$$x_{\mu} \to x'\mu = e^{g}x_{\mu}, \quad u(x) \to u'(x') = u(x).$$
 (1.3.23)

Having repeated the same reasoning used while obtaining the expressions (1.3.22), we adduce the final result.

The equation

$$\Box u + u_{\nu} u^{\nu} f(u) = 0 \tag{1.3.24}$$

is invariant under the algebra $A\widetilde{P}(1, n-1) = \{AP(1, n-1)(1.1.2), D(1.3.23)\}.$

We can verify that Equation (1.3.24) is not invariant under conformal transformations generated by the operators $K_{\mu} = 2x_{\mu}xP - x^2P_{\mu}$ (see §2.3) with any f(u). By the way, let us note that with the transformation u = g(w), where $\ddot{g} + \dot{g}^2 f(g) = 0$, Equation (1.3.24) is reduced to the form (1.3.19) with $\lambda = 0$.

1.4. Ansätze and reduction of PDEs. The extended Poincare group $\widetilde{P}(1,2)$ and its invariants

The method of constructing exact solutions of PDEs is expounded. This method is widely used throughout the book.

1. Let us consider as an example the nonlinear wave equation

$$\Box u + \lambda u^k = 0, \tag{1.4.1}$$

where $\Box = \partial_{\mu}\partial^{\mu}$, $\mu = \overline{0, n-1}$, u = u(x). Solutions of Equation (1.4.1), according to [63], are sought in the form

$$u(x) = f(x)\varphi(\omega), \qquad (1.4.2)$$

where $\varphi(w)$ are unknown functions to be found. The explicit form of the new variables $\omega = \omega(x)$ and function f(x) is defined from the conditions described below. Expressions such as (1.4.2) shall be called ansatze. *

^{*} The word "ansatz" in German means "substitution," although to stress that expressions (1.4.2) are not purely substitutions *ad hoc* but are also a method of calculating functions f(x), $\omega(x)$ in explicit form they shall be referred to by the word "ansatz."
Substituting (1.4.2) into (1.4.1) gives

$$(\Box f)\varphi + f\left(\frac{\partial\omega_i}{\partial x_{\mu}}\frac{\partial\omega_k}{\partial x^{\mu}}\varphi_{\omega_i\omega_k} + \Box\omega_i\varphi_{\omega_i}\right) + 2\frac{\partial f}{\partial x_{\mu}}\frac{\partial\omega_i}{\partial x^{\mu}}\varphi_{\omega_i} + \lambda f^k\varphi^k = 0.$$
(1.4.3)

If we take

$$\frac{\partial \omega_i}{\partial x_{\mu}} \frac{\partial \omega_k}{\partial x^{\mu}} = f^{k-1} a_{ik}(\omega),$$

$$f \Box \omega_i + 2 \frac{\partial f}{\partial x_{\mu}} \frac{\partial \omega_i}{\partial x^{\mu}} = f^k b_i(\omega),$$

$$\Box f = f^k c(\omega)$$
(1.4.4)

where a_{ik}, b_i, c_i are some functions depending on ω , only then instead of (1.4.3) we shall have

$$a_{ik}\varphi_{\omega_i\omega_k} + b_i\varphi_{\omega_i} + c\varphi + \lambda\varphi^k = 0. \tag{1.4.5}$$

Conditions (1.4.4), called the *splitting conditions* [63], are the system of equations for defining the explicit form of the new variables $\omega = \omega(x)$ and the functions $f(x), a_{ik}, b_i, c$. The new number of variables $\omega = \{\omega_i\}$ may be $1, 2, \ldots, n-1$. In these cases Equations (1.4.5) will be ODEs or PDEs with the number of independent variables being equal to $1, 2, 3, \ldots, n-1$. The solutions of the reduced Equation (1.4.5), especially in the case of ODEs, can often be found by direct integration.

The simplest case, when the number of independent variables ω in ansatz (1.4.2) is one and f = 1, the splitting conditions (1.4.4) take the form

$$\Box \omega = b(\omega)$$

$$\omega_{\mu}\omega^{\mu} \equiv \frac{\partial \omega}{\partial x^{\mu}} \frac{\partial \omega}{\partial x_{\mu}} = a(\omega)$$
(1.4.4')

By means of a change of variables similar to that of (1.2.16)-(1.2.18), the system (1.4.4') can be transformed to the canonical form

$$\Box \widetilde{\omega} = F(\widetilde{\omega})$$

$$\widetilde{\omega}_{\mu} \widetilde{\omega}^{\mu} \equiv \lambda, \qquad \lambda = 0, 1, -1.$$
(1.4.4")

So, to describe all ansatze $u(x) = \varphi(\omega)$ which reduce Equation (1.4.1) to ODEs one has to find all solutions of system (1.4.4"). This latter problem is fully solved in [60*,130*] (see Appendix 6).

To determine the explicit form of the ansatz (1.4.2) one can effectively use the symmetry of Equation (1.4.1). As had been shown in §1.1, the generators of the invariance algebra of Equation (1.4.1) have the form

$$X = \xi^{\mu}(x)\partial_{\mu} + \eta(x), \qquad (1.4.6)$$

where $\xi^{\mu}(x), \eta(x)$ are some scalar functions. Using the operator (1.4.6) we can determine the explicit expressions for the variables $\omega(x)$ and functions f(x) as solutions of the following equations:

$$\xi^{\mu}(x)\partial_{\mu}\omega(x) = 0, \qquad (1.4.7)$$

$$Xf(x) \equiv \left(\xi^{\mu}(x)\partial_{\mu} + \eta(x)\right)f(x), \qquad (1.4.8)$$

or of the equivalent system of Lagrange-Euler equations

$$\frac{dx_0}{\xi^0} = \frac{dx_1}{\xi^1} = \dots = \frac{dx_{n-1}}{\xi^{n-1}} = \frac{du}{-\eta u} \stackrel{\text{def}}{=} d\tau$$
(1.4.9)

Substituting the ansatz into (1.4.1) we get for $\varphi(\omega)$ an equation not containing f(x). The number of variables of w = w(x) which are the first integrals of (1.4.9) is one less than the number of variables of x.

Evidently the formulated algorithm may be applied to the PDEs for the functions $\varphi(\omega_1, \ldots, \omega_{n-1})$ repeatedly. Surely it is necessary for these equations to possess nontrivial symmetries (for more details see §2.1).

Let us point out that the ansatz (1.4.2) can also work effectively in the case when an equation does not possess a proper Lie symmetry [67]. For example, the Equation [108]

$$\Box u + \left(\frac{\lambda_0}{x_0}\frac{\partial u}{\partial x_0}\right)^2 - \left(\frac{\lambda_1}{x_1}\frac{\partial u}{\partial x_1}\right)^2 - \left(\frac{\lambda_2}{x_2}\frac{\partial u}{\partial x_2}\right)^2 - \left(\frac{\lambda_3}{x_3}\frac{\partial u}{\partial x_3}\right)^2 = 0$$

where $\lambda_0, \lambda_1, \lambda_2, \lambda_3$ are arbitrary parameters, $x_{\mu} \neq 0$ is not Lorentz-invariant, but the Lorentz-invariant ansatz

$$u(x) = \varphi(\omega), \quad \omega = x_{\mu}x^{\mu}$$

reduces it to the ODE

$$\omega rac{d^2 arphi}{d \omega^2} + 2 rac{d arphi}{d \omega} - \lambda^2 \left(rac{d arphi}{d \omega}
ight)^2 = 0$$

(for more details about the reduction and solution of equations with broken symmetry see $\S5.7$).

Let us also note that in some cases (see, for example, Sec. 1.6) the following ansatze proved to be effective

$$u(x) = f(x)\varphi(\omega) + g(x) \tag{1.4.10}$$

Transformations leaving an equation invariant may be used to generate new solutions from known solutions. Thus, if a PDE is invariant with respect to transformations

$$x \to x' = f(x,\theta), \quad u(x) = u'(x') = r(x,\theta)u(x)$$
 (1.4.11)

(θ is a parameter, $f(x, \theta)$, $r(x, \theta)$ are some smooth functions) and if $u_I(x)$ is a solution of this equation then the function

$$u_{II}(x) = r^{-1}(x,\theta)u_{I}(x') \tag{1.4.12}$$

will be a solution too.

In the case when the symmetry transformations are nonlinear with respect to u, i.e.,

$$x' = f(x, u, \theta), \ u'(x') = g(x, u, \theta)$$
(1.4.13)

then to obtain the generating formulas it is necessary to solve the functional equation

$$g(x, u_{II}(x), \theta) = u_I(f(x, u_{II}(x), \theta))$$
(1.4.14)

with respect to $u_{II}(x)$.

In connection with group generating of solutions the notion of G-ungenerative solutions [65] (see also $\S2.3$) arises.

Definition. A set of solutions M is called G-ungenerative if the group G generating does not take out of this set with any transformations from the group G.

2. Let us determine the invariant variables $\omega = \omega(x)$ for the operator that is the linear combination of the generators of the group $\widetilde{P}(1,2)$ (see for example (1.1.2),(1.1.3)).

$$\xi^{\mu}(x)\partial_{\mu} = (\varpi x^{\mu} + c^{\mu\nu}x_{\nu} + d^{\mu})\partial_{\mu}, \qquad (1.4.15)$$

where $\mu, \nu = 0, 1, 2$; $\mathfrak{B}, c_{\mu\nu} = -c_{\nu\mu}, d^{\mu}$ are arbitrary constants.

Let us rewrite (1.4.9) equivalently as a system of ODEs

$$\frac{dx_{\mu}}{d\tau} = \xi^{\mu}(x) = \Re x_{\mu} + c_{\mu\nu}x^{\nu} + d_{\mu}, \qquad (1.4.16)$$

or in matrix form

$$\dot{X} = AX + D, \tag{1.4.17}$$

where

$$X = \begin{pmatrix} x_0 \\ x_1 \\ x_2 \end{pmatrix}, \quad A = \begin{pmatrix} \mathfrak{X} & c_{01} & c_{02} \\ c_{01} & \mathfrak{X} & c_{12} \\ c_{02} & -c_{12} & \mathfrak{X} \end{pmatrix}, \quad D = \begin{pmatrix} d_0 \\ d_1 \\ d_2 \end{pmatrix}.$$

The number of independent solutions of system (1.4.17) is determined by the number of different roots of the characteristic equation

$$\det\left(A - \lambda I\right) = 0 \tag{1.4.18}$$

(where I is the unit matrix.)

In the present case, Equation (1.4.18) has the form

$$(x - \lambda) \left[(x - \lambda)^2 + c_{\mu\nu} c^{\mu\nu} \right] = 0.$$
 (1.4.19)

Whence, having obtained λ , we construct the corresponding solutions of system (1.4.9) and then, having excluded τ from it via (1.4.9), we find the sought-after invariants.

Without going into the details of these rather cumbersome calculations we shall present the final result. Depending on the value of the parameters $x, c_{\mu\nu}, d_{\mu}$, we get the (nonequivalent) invariants of the group $\tilde{P}(1,2)$ [88].

Ν	Invariant variables ω	Conditions on parameters
1.	$(\alpha y)(\beta y)^a, y^2(\beta y)^{-2}$	$\alpha^2 = \alpha\beta = 0, \beta^2 = b < 0.$
2.	$(eta y)(lpha y)^{-1}+\ln{(lpha y)},y^2(lpha y)^{-2}$	$\alpha^2 = \alpha\beta = 0, \beta^2 = b < 0.$
3.	$\ln{(\alpha y)} + \arctan{\frac{\delta y}{\beta y}}, y^2(\alpha y)^{-2}$	$\alpha^2 = -\beta^2 = -\delta^2 = 1,$
		$\alpha\beta = \alpha\delta = \beta\delta = 0.$
4.	$lpha y + \ln (eta y), (\delta y) (eta y)^{-2}$	$\alpha^2 = \delta^2 = \alpha\beta = \beta\delta = 0, \beta^2 = 1,$
		$\alpha\delta = b \neq 0.$
5.	$(\beta y)(\alpha y)^{-1}, (\delta y)(\alpha y)^{-1}$	$\alpha^2 = -\beta^2 = -\delta^2 = 1,$
		$\alpha\beta = \alpha\delta = \beta\delta = 0.$
6.	$y^2 + (\beta y)^2$, $\beta y + \ln (\alpha y)$	$\alpha^2 = \alpha\beta = 0, \beta^2 = 1.$
7.	$(\beta y)^2 - y^2, \beta y + \arctan rac{\delta y}{\alpha y}$	$\alpha^2 = -\beta^2 = -\delta^2 = 1,$
		$\alpha\beta = \alpha\delta = \beta\delta = 0.$
8.	$\frac{1}{2}(\alpha y)^2 + a(\beta y), \frac{1}{3}(\alpha y)^3 +$	$\alpha^2 = \alpha\beta = \beta\delta = 0,$
	$+a(lpha y)(eta y)+a^2(\delta y)$	$\alpha\delta = -\beta^2 = \delta^2 = b \neq 0.$
9.	$\alpha x, x^2$	$\alpha^2 = 1.$
10.	lpha x, eta x.	

Table 1.4.1 The invariant variables of the group $\widetilde{P}(1,2)$.

Here $a, b, a_{\mu}, \alpha_{\mu}, \delta_{\mu}$ are arbitrary constants, satisfying the aforementioned conditions: $y_{\mu} = x_{\mu} + a_{\mu}$.

The invariants 1–10 from Table 1.4.1 are easily generalized for a greater number of variables. For this purpose it is necessary to sum on μ, ν from 0 to an arbitrary n > 2. This is done so that we do not exhaust all possible unequivalent invariants.

The alternate method of constructing unequivalent invariants of the group $\widetilde{P}(1,2)$, as well as any group G, is the following. Instead of solving Equation (1.4.7) or the equivalent system of ODEs (1.4.9) with the functions ξ^{μ} which are the linear combination of the generator's coordinates of the Lie algebra, all at least one-dimensional, non-isomorphic subalgebras of the AG should be found. Then the invariants of these subalgebras' generators are calculated. Certainly the very problem of describing non-isomorphic subalgebras of a given Lie algebra is difficult enough but for the fact that there is an effective algorithm [163] with which the complete description of non-isomorphic subalgebras of the most important Lie algebras is given: P(1,3) [163], P(1,4) [51], $\widetilde{P}(1,n)$ [14–16], P(2,n) [13], G(1,n) [12,68], E(n) [69] (for a comprehensive review of this subject see [139*]).

The explicit form of the non-isomorphic, one-dimensional subalgebras of the algebra AP(1,3) and the corresponding invariants obtained in [163,114] are presented in Table 2.1.1.

In Appendix 2 the one-dimensional non-isomorphic subalgebras of $A\widetilde{P}(1,3)$ are constructed. These results are used in §2.1 and §2.2 when constructing ansatzes for spinor or vector fields.

1.5. Reduction and exact solutions of the nonlinear d'Alembert equation

In the present paragraph we construct the families of exact solutions of the nonlinear wave equation [88]

$$\Box u + \lambda u^k = 0, \tag{1.5.1}$$

where $u = u(x_0, x_1, x_2)$, $\Box = \partial_0^2 - \partial_1^2 - \partial_2^2$, λ , $k \neq 1$ are arbitrary constants.

As shown in §1.1, Equation (1.5.1) is invariant under the algebra $A\tilde{P}(1,2)$. Using this result we take for the operator X (1.4.6) the generators' linear combination of the AI $A\tilde{P}(1,2)$

$$X = (a x_{\mu} + c_{\mu\nu} x^{\mu} + d_{\mu}) \partial_{\mu} + \frac{2a}{k-1}$$
(1.5.2)

and construct by the method elaborated in the preceding paragraph the ansatze (1.4.2). The invariant variables ω_1, ω_2 are presented in Table 1.4.1. The function f(x) may be determined both from (1.4.8) and (1.4.9) using the operator (1.5.2), and from the splitting conditions, (1.4.4). Omitting easy but

rather lengthy calculations we present the explicit form of $\widetilde{P}(1, 2)$ -unequivalent ansatze for the scalar field u(x) with symmetry operators (1.5.2) [88]

$$1^{\circ} - 5^{\circ} \quad u(x) = (\beta y)^{\frac{4}{1-k}} \phi(\omega_1, \omega_2),$$

$$6^{\circ} - 10^{\circ} \quad u(x) = \phi(\omega_1, \omega_2),$$
(1.5.3)

Formulas (1.5.3) are to be considered together with Table 1.4.1 where the corresponding values of invariant variables are represented, and the relationships between arbitrary constants $\alpha_{\nu}, \beta_{\nu}$ are indicated.

Let us substitute the ansatze $1^{\circ}-10^{\circ}$ (1.5.3) into Equation (1.5.1). After some cumbersome calculations we get

$$\begin{split} & 1^{\circ} \cdot a^{2} \omega_{1}^{2} \varphi_{11} - 4 \omega_{1} (a \omega_{2} - a - 1) \varphi_{12} + 4 \omega_{2} (\omega_{2} - 1) \varphi_{22} + \\ & + a [a - 1 + 4(1 - k)] \omega_{1} \varphi_{1} + 2(k - 1)^{-1} [(3k + 1) \omega_{2} - 2(k + 1)] \varphi_{2} + \\ & + 2(k + 1)(k - 1)^{-2} \varphi + \lambda \varphi^{k} = 0; \\ & 2^{\circ} \cdot \varphi_{11} + 4 \varphi_{12} - 4 \omega_{2} \varphi_{22} + 2(3 + k)(1 - k)^{-1} \varphi_{2} + \lambda \varphi^{k} = 0; \\ & 3^{\circ} \cdot [1 - (\omega_{2} - 1)^{-1}] \varphi_{11} - 4(\omega_{2} - 1)(\varphi_{12} - \omega_{2} \varphi_{22}) + (3 + k)(1 - k)^{-1} \varphi_{1} - \\ & -4(1 - k)[(3 - k) \omega_{2} - (1 + k)] \varphi_{2} + 2(1 + k)(1 - k)^{-2} \varphi + \lambda \varphi^{k} = 0; \\ & 4^{\circ} \cdot \omega_{1}^{2} \varphi_{11} - \omega_{1} \omega_{2} (b \omega_{2} - 2) \varphi_{12} + \omega_{2}^{2} \varphi_{22} + 4(1 + k)^{-1} (\omega_{1} \varphi_{1} + \omega_{2} \varphi_{2}) + \\ & + 2(k + 1)(1 - k)^{-2} \varphi + \lambda \varphi^{k} = 0; \\ & 5^{\circ} \cdot (\omega_{1}^{2} + 1) \varphi_{11} + 2 \omega_{1} \omega_{1} \varphi_{22} + (\omega_{2}^{2} + 1) \varphi_{22} + 2(k + 1)(k - 1)^{-1} \times \\ & \times (\omega_{1} \varphi_{1} + \omega_{2} \varphi_{2}) + 2(k + 1)(k - 1)^{-2} \varphi + \lambda^{k} = 0; \\ & 6^{\circ} \cdot 4 \omega_{1} \varphi_{11} + 4 a \varphi_{12} - \varphi_{22} + 4 \varphi_{1} + \lambda \varphi^{k} = 0; \\ & 7^{\circ} \cdot -4 \omega_{1} \varphi_{11} + (1 + a^{2} \omega_{1}^{-1}) \varphi_{22} + 4 \varphi_{1} + \lambda \varphi^{k} = 0; \\ & 8^{\circ} \cdot -\varphi_{11} + 2(\omega_{1} + a^{2}) \varphi_{22} + (\lambda/ba^{2}) \varphi^{k} = 0; \\ & 9^{\circ} \cdot \varphi_{11} + 4 \omega_{1} \varphi_{12} + 4 \omega_{2} \varphi_{22} + \varphi_{2} + \lambda \varphi^{k} = 0; \\ & 10^{\circ} \cdot \alpha^{2} \varphi_{11} + 2 \alpha \beta \varphi_{12} + \beta^{2} \varphi_{22} + \lambda \varphi^{k} = 0, \\ \end{aligned}$$

Here $\varphi_k = \frac{\partial \varphi}{\partial \omega_k}$, $\varphi_{kl} = \frac{\partial^2 \varphi}{\partial \omega_k \partial \omega_l}$, N 1°-10° correspond the numbers of ansatze in (1.5.3) and of invariant variables in Table 1.4.1.

Let us pass to the more detailed analysis of the reduced Equations $1^{\circ}-10^{\circ}$ (1.5.4).

If in 1° (1.5.4) we put $\varphi_2 = 0$ then the substitution

$$\varphi(\omega_1) = \omega_1 \frac{k+1}{k-1} V(\omega_1) \tag{1.5.5}$$

reduces 1° (1.5.4) to the Emden-Fauler equation

$$\omega_1^2 \ddot{V} + 2\omega_1 \dot{V} + \lambda \omega_1^{k+1} V^k = 0, \qquad (1.5.6)$$

where dots denotes differentiation with respect to ω_1 . The partial solution of (1.5.6) is sought in the form

$$V(\omega_1) = c\omega_1^s$$
, with c, s constant (1.5.7)

Substituting this expression into (1.5.6) we get

$$s = -\frac{k+1}{k-1}, \qquad c^{k-1} = -2\frac{k+1}{\lambda(k-1)^2}$$

As a result we find the following solution for Equation (1.5.1):

$$u(x) = (\beta x)^{\frac{2}{1-k}}, \quad \beta^2 = -\lambda \frac{(1-k)^2}{2(1+k)}$$
 (1.5.8)

Putting in 2° (1.5.4) $\varphi_2 = 0$ we get the ODE

$$\varphi_{11} + \lambda \varphi^k = 0, \tag{1.5.9}$$

whose general solution is expressed through the integral

$$\int_{0}^{\varphi} \frac{d\tau}{\sqrt{c_1 + \tau^{k+1}}} = \pm \frac{2}{1-k} (\omega_1 + c_2), \qquad (1.5.10)$$

where c_1 , c_2 are constants of integration. Specifically, when $c_1 = 0$ we have

$$\varphi(\omega_1) = (\omega_1 + c_2)^{\frac{2}{1-k}}$$
(1.5.11)

(When k = 3, the solutions of Equation (1.5.9) are expressed with elliptic functions—see Appendix 1).

Returning to 2° (1.5.3) we find the following solution for Equation (1.5.1):

$$u(x) = [\beta x + \alpha x (c_2 + \ln \alpha x)]^{\frac{2}{1-k}},$$

$$\alpha^2 = \alpha \beta = 0, \quad \beta^2 = -\frac{\lambda}{2} \frac{(1-k)^2}{1+k}.$$
(1.5.12)

Putting in 2° (1.5.4) $\varphi_1 = 0$ we obtain

$$-4\omega_2\varphi_{22} + 2(3+k)(1-k)^{-1}\varphi_2 + \lambda\varphi^k = 0, \qquad (1.5.13)$$

whose partial solution is the function

$$\varphi(\omega_2) = \left[\frac{\lambda}{2} \frac{(k-1)^2}{k-3} \omega_2\right]^{\frac{1}{1-k}}, \quad k \neq 3.$$

This leads to the solution of Equation (1.5.1)

$$u(x) = \left[\frac{\lambda(k-1)^2}{2(k-3)}x_{\nu}x^{\nu}\right]^{\frac{1}{1-k}}, \quad k \neq 3.$$
 (1.5.14)

By a substitution analogous to (1.5.5), Equation 4° (1.5.4), when $\varphi_1 = 0$ or $\varphi_2 = 0$, is reduced to the Emden-Fauler Equation (1.5.6).

When $\varphi_1 = 0$ or $\varphi_2 = 0$, Equation 5° (1.5.4) takes the form

$$(\omega^2 + 1)\ddot{\varphi} + a\omega\dot{\varphi} + b\varphi + \lambda\varphi^k = 0 \qquad (1.5.15)$$

where $\omega = \omega_1$ or ω_2 ; and the overdots denote differentiation with respect to ω . When k = -3 it is possible to integrate Equation (1.5.15) and the result has the form

$$\varphi = \pm \left[2c_1 + \frac{c_2^2 \left(\omega + \sqrt{\omega^2 + 1}\right)^{\pm 2} + c_1^2 - \lambda}{2c_2 \left(\omega + \sqrt{\omega^2 + 1}\right)^{\pm 1}} \right]^{1/2}$$
(1.5.16)

4° (1.5.3) and (1.5.16), when $\omega = \omega_1$ or $\omega = \omega_2$, results in the following solution for Equation (1.5.1):

$$u(x) = \begin{cases} \pm [(\alpha x)(c_1 + \mu \operatorname{ch} Q(x)]^{1/2}, & c_1^2 - 4\lambda = \mu^2 > 0, \\ \pm [(\alpha x)(c_1 + \mu \operatorname{sh} Q(x)]^{1/2}, & c_1^2 - 4\lambda = -\mu^2 > 0, \\ \pm [(\alpha x)Q(x)]^{1/2}, & c_1 - 4\lambda = 0, \end{cases}$$
(1.5.17)

where $Q(x) = c_2 [(\beta x)(\alpha x)^{-1} + (\beta x)^2 (\alpha x)^2 + 1^{1/2}], \alpha^2 = -\beta^2 = 1, \quad \alpha\beta = 0.$ Putting in 6° (1.5.4) $\varphi_2 = 0$, we obtain the ODE

$$\omega_1 \varphi_{11} + \varphi_1 + \frac{\lambda}{4} \varphi^k = 0, \qquad (1.5.18)$$

whose partial solution is

$$\varphi(\omega_1) = \left[-\frac{\lambda}{4} (1-k)^2 \omega \right]^{\frac{1}{1-k}}$$
(1.5.19)

Formulas (1.5.19) and 6° (1.5.3) determine the following solution for Equation (1.5.1):

$$u(x) = \left\{ -\frac{\lambda}{4} (1-k)^2 \left[(\beta y)^2 + y^2 \right] \right\}^{\frac{1}{1-k}}.$$
 (1.5.20)

Putting in 8° (1.5.4) $\varphi_2 = 0$, we get (1.5.9), and using (1.5.11) and 8° (1.5.3) we obtain one more family of solutions for Equation (1.5.1):

$$u(x) = \left[\beta y + \frac{1}{2a}(\alpha y)^2 + c_2\right]^{\frac{2}{1-k}}, \quad \alpha^2 = \alpha\beta = 0, \quad \beta^2 = -1 \quad (1.5.21)$$

24

Equation 10° (1.5.4) is the two-dimensional wave equation, if $\alpha^2 = -\beta^2 = 1$, $\alpha\beta = 0$.

In Equation 9° (1.5.4) let us make the substitution of variables

$$\tau_0 = \sqrt{\omega_1^2 - \omega_2}, \qquad \tau_1 = \omega_1,$$
 (1.5.22)

which transform 9° into the nonlinear Darboux equation

$$\varphi_{\tau_0\tau_0} - \varphi_{\tau_1\tau_1} + \tau_0^{-1}\varphi_{\tau_0} - \lambda\varphi^k = 0$$
 (1.5.23)

To find solutions for Equation (1.5.23) we once again apply the algorithm of §1.4. But first let us investigate its symmetry.

Theorem 1.5.1. The maximal local invariance group of the nonlinear Darboux Equation (1.5.23) is a 2-parameter Lie group generated by operators of the form

$$X_1 = \frac{\partial}{\partial \tau_1}, \qquad X_2 = \tau_0 \frac{\partial}{\partial \tau_0} + \tau_1 \frac{\partial}{\partial \tau_1} - \frac{2}{1-k}$$
(1.5.24)

Theorem 1.5.2. The maximal local invariance group of the *n*-dimensional nonlinear Darboux equation

$$\Box u + \lambda_1 x_0^{-1} u_0 + \lambda_2 u^k = 0 \tag{1.5.25}$$

when $k = \frac{n+2+\lambda_1}{n-2+\lambda_1}$, $\lambda_1 \neq 2-n$ is the conformal group C(n-1) whose generators have the form

$$P_{a} = -\partial_{a}, \quad J_{ab} = x_{a}P_{b} - x_{b}P_{a}; \quad a, b = \overline{1, n - 1},$$

$$D = x_{0}\partial_{0} + x_{a}\partial_{a} - \frac{1}{2}(n - 2 + \lambda_{1}),$$

$$K_{a} = 2x_{a}D - (x_{0}^{2} - x_{b}x^{b})P_{a},$$

(1.5.26)

when $k \neq \frac{n+2+\lambda_1}{n-2+\lambda_1}$ it is the extended Euclidean group $\widetilde{E}(n-1)$ generated by the operators

$$P_a = -\partial_a, \qquad J_{ab} = x_a P_b - x_b P_a,$$

$$D = x_0 \partial_0 + x_a \partial_a - \frac{2}{1-k}.$$
(1.5.27)

The proof of Theorems 1.5.1, 1.5.2 may be obtained by Lie's method.

We seek the solutions of Equation (1.5.23) in the form

$$\varphi(\tau_0, \tau_1) = \tau^{\frac{2}{1-k}} \Phi(V), \quad V = \frac{\tau_1 + a}{\tau_0}, \quad a = \text{const.}$$
 (1.5.28)

Substituting (1.5.28) into (1.5.23) results in the ODE

$$(V^2 - 1)\Phi_{VV} + \frac{k+3}{k-1}V\Phi_V + \frac{4}{(k-1)^2}\Phi + \lambda\Phi^k = 0.$$
(1.5.29)

Knowing the solutions of this equation, we construct by the formulae (1.5.28), (1.5.22), 9° (1.5.3), solutions for equation (1.5.1). Note that when k = 5, (1.5.28) coincides with the nonlinear Legendre equation

$$\frac{d}{dV}\left[(1-V^2)\frac{d\Phi}{dV}\right] - \frac{1}{4}\Phi - \lambda\Phi^5 = 0.$$
 (1.5.30)

To conclude this paragraph we obtain solutions for Equation (1.5.1) containing arbitrary functions. We shall seek these solutions in the form

$$u(x) = [f(x) + g(x)]^{\alpha}$$
(1.5.31)

where $\alpha = (1-k)^{-1}$ or $2(1-k)^{-1}$, $V = \{V^{\nu}\}$, and $W = \{W^{\nu}\}$ are some functions of x.

Let us substitute (1.5.31) into the *n*-dimensional Equation (1.5.1). As a result we obtain

$$(\alpha - 1)F_{\mu}F^{\mu} + (f + g)(f_{\nu\lambda}V_{\mu}^{\lambda\mu} + g_{\nu\lambda}W_{\mu}^{\nu}W^{\lambda\mu} + f_{\nu}\Box V^{\nu} + g_{\nu}\Box W^{\nu}) + \frac{\lambda}{\alpha}(f + g)^{\alpha(k-1)+2} = 0, \qquad (1.5.32)$$

where

$$F_{\mu} = f_{\nu}V_{\mu}^{\nu} + g_{\nu}W_{\mu}^{\nu}, \quad f_{\nu} = \frac{\partial g}{\partial W^{\nu}}, \quad f_{\mu\nu} = \frac{\partial^2 f}{\partial V^{\mu} \partial V^{\nu}}, \quad g_{\mu\nu} = \frac{\partial^2 g}{\partial V^{\mu} \partial V^{\nu}}.$$

Let us adduce some partial solutions of Equation (1.5.32):

a).
$$n \ge 3$$
, $\alpha = (1-k)^{-1}$, $f(V) = (V^0 + c)^2$, $g(W) = W^0 W^1$,
 $V^0 = \alpha x$, $W^0 = \beta x$, $W^1 = \delta x$, $c = \text{const}$, (1.5.33)
 $\alpha \beta = \alpha \delta = \beta^2 = \delta^2 = 0$, $2\alpha^2 = \beta \delta = \lambda \frac{k-1}{k-2}$, $k \ne 3$.

b).
$$n \ge 3$$
, $\alpha = 2(1-k)^{-1}$, $f(V) = V^1 \varphi(V^0)$, $g(W) = W^1 \psi(W^0)$,
 $V^0 = W^0 = \alpha x$, $V^1 = \beta x$, $W^1 = \delta x$, (1.5.34)
 $\alpha^2 = \alpha \beta = \alpha \delta = \beta \delta = 0$, $\beta^2 = -\delta^2 = -1$,

where φ and ψ are arbitrary differentiable functions satisfying the condition $\varphi^2 + \psi^2 = \lambda (k-1)^2/2(k+1)$.

c).
$$n \ge 3$$
, $\alpha = 2(1-k)^{-1}$, $f(V) = f(V^0)$ is an arbitrary smooth function;

$$g(W) = W^0, V^0 = \alpha x, W^0 = \beta x,$$

 $\alpha^2 = \alpha \beta = 0, \quad \beta^2 = -\frac{\lambda}{2} \frac{(1-k)^2}{1+k}$
(1.5.35)

Returning to (1.5.31) we obtain from (1.5.33) through (1.5.35) corresponding solutions for the *n*-dimensional Equation (1.5.1) $(n \ge 3)$

$$u(x) = \left[(\alpha x + c)^2 + (\beta x)(\delta x) \right]^{\frac{1}{1-k}}, \quad c = \text{const}, \ k \neq 3,$$

$$\alpha \beta = \alpha \delta = \beta^2 = \delta^2 = 0, \quad 2\alpha^2 = \beta \delta = \lambda \frac{k-1}{k-3};$$
 (1.5.36)

$$u(x) = \left[\beta x \varphi(\alpha x) + \delta x \psi(\alpha x)\right]^{\frac{2}{1-k}},$$

$$\alpha^2 = \alpha \beta = \alpha \delta = \beta \delta = 0, \quad \beta^2 = -\delta^2 = -1;$$
(1.5.37)

$$u(x) = [f(\alpha x) + \beta x]^{\frac{2}{2-k}},$$

$$\alpha^{2} = \alpha \beta = 0, \quad \beta^{2} = -\frac{\lambda}{2} \frac{(k-1)^{2}}{k+1}.$$
(1.5.38)

The solutions obtained, containing the arbitrary functions, may be useful while solving various problems of mathematical physics, such as the Cauchy problem or some boundary value problems.

In conclusion let us note that in the case of three spacial variables nonlinear wave equation

$$\Box u + \lambda u^3 = 0 \tag{1.5.39}$$

is invariant under the conformal group $C(1,3) \supset \widetilde{P}(1,3)$. The detailed analysis of conformal symmetry will be given in the next chapter (§2.3), but here it is appropriate to consider this question in connection with Equation (1.5.39).

The conformal transformations leaving Equation (1.5.39) invariant have the form

$$x_{\mu} \to x'_{\mu} = \frac{x_{\mu} - c_{\mu}x^2}{\sigma(x,c)}, \quad u(x) \to u'(x') = \sigma(x,c)u(x),$$
 (1.5.40)

where $\sigma(x,c) = 1 - 2cx + c^2 x^2$, c_{μ} are arbitrary constants. According to (1.4.11), (1.4.12) one can easily derive from (1.5.40) the following formula of generating solutions

$$u_{II}(x) = \frac{u_{I}(x')}{\sigma(x,c)}.$$
 (1.5.41)

The conformally invariant ansatz for Equation (1.5.39) has the form

$$u(x) = (x^2)^{-1} \varphi(\omega), \quad \omega = \frac{\beta x}{x^2}, \quad (\beta x \equiv \beta_{\nu} x^{\nu}) \tag{1.5.42}$$

where β_{ν} are arbitrary constants. The substitution of (1.5.42) into (1.5.39) gives the following ODE

$$\beta^2 \frac{d^2 \varphi}{d\omega^2} + \lambda \varphi^3 = 0 \qquad (\beta^2 \equiv \beta^\nu \beta_\nu). \tag{1.5.43}$$

Solutions of this equation are expressed in terms of elliptic functions (see Appendix 1) when $\beta^2 = -1$, $\lambda > 0$, there is a simple partial solution of Equation (1.5.43)

$$\varphi = \sqrt{\frac{2}{\lambda}} \frac{1}{\omega}.$$
 (1.5.44)

So, the expression (1.5.44) together with (1.5.42) result in the following solution of Equation (1.5.39)

$$u(x) = \sqrt{\frac{2}{\lambda}} \frac{1}{\beta x}, \quad \beta^2 = -1.$$
 (1.5.45)

After application to (1.5.45) the formula of generating solutions by means of translational transformations

$$u_{II}(x) = u_I(x'), \quad x'_{\mu} = x_{\mu} + a_{\mu}, \qquad (a_{\mu} = \text{const}),$$

we get the C(1,3)-ungenerative family of solutions of Equation (1.5.39)

$$u(x) = \sqrt{\frac{2}{\lambda}} \frac{1}{\beta x + \omega}, \quad \beta^2 = -1, \qquad (1.5.46)$$

where æ is an arbitrary constant.

1.6. Reduction and solutions of the Liouville equation

The Liouville equation

$$\Box u + \lambda e^u = 0 \tag{1.6.1}$$

arises in the problems of differential geometry, the theory of nonlinear waves, and quantum field theory [18].

In the two-dimensional case the general solution of Equation (1.6.1) is

$$u(x_0, x_1) = \ln \left\{ -\frac{8}{\lambda} \frac{\dot{f}(x_0 + x_1) \dot{g}(x_0 - x_1)}{\left[f(x_0 + x_1) + g(x_0 - x_1) \right]^2} \right\},$$
 (1.6.2)

where f, g are arbitrary differentiable functions and \dot{f}, \dot{g} are derivatives with respect to the corresponding argument $(\dot{f}\dot{g} < 0)$, was constructed in 1853 by

Liouville. The singular solutions of Equation (1.6.1) were obtained in [127, 128].

Below we shall consider the three-dimensional Equation (1.6.1). As shown in §1.1 its maximal (in Lie's sense) invariance group when n > 2 is the extended Poincare group $\tilde{P}(1, n - 1)$ and the general form of the symmetry IFO is as follows:

$$X = (\mathfrak{a} x^{\mu} + c^{\mu\nu} x_{\nu} + d^{\mu}) \frac{\partial}{\partial x_{\mu}} - 2\mathfrak{a} \frac{\partial}{\partial u} \,. \tag{1.6.3}$$

Because of the addend $-2\omega\partial_u$ the operator (1.6.3) doesn't have the structure of (1.4.6). But if we make the transformation (see (1.3.20))

$$u = -2 \ln W,$$
 (1.6.4)

then (1.6.3) coincides with the operator (1.5.2) with k = -1 and it is possible to use ansatz (1.5.3). Substituting ansatze 1°-10° (1.5.3) with k = -1 into (1.6.4) instead of W, we obtain the ansatz for Equation (1.6.1) [88]:

$$1^{\circ} - 5^{\circ}. \quad u(x) = \varphi(w_1, w_2) - 2\ln(\alpha y),$$

$$6^{\circ} - 10^{\circ}. \quad u(x) = \varphi(w_1, w_2).$$
(1.6.5)

Formulas (1.6.5) are to be considered together with Table 1.4.1 where the corresponding values of the invariant variables w_1, w_2 are adduced and the conditions for $\alpha_{\nu}, \beta_{\nu}$ et al. are indicated.

Substituting ansatz $1^{\circ} - 10^{\circ}$ (1.6.5) into Equation (1.6.1) we obtain, correspondingly

1°.
$$a^2 w_1^2 \varphi_{11} + 4w_1 (w_2 + a + 1)\varphi_{12} + 4w_2 (w_2 - 1)\varphi_{22} + a(a - 1)w_1 \varphi_1 + 2(3w_2 - 1)\varphi_2 + 2 + \lambda e^{\varphi} = 0;$$

2°.
$$\varphi_{11} + 4\varphi_{12} - 4w_2\varphi_{22} - 2\varphi_2 + \lambda e^{\varphi} = 0;$$

3°.
$$[1 - (w_2 - 1)^{-1}] \varphi_{11} - 4(w_2 - 1)\varphi_{12} + 4w_2(w_2 - 1)\varphi_{22} - \varphi_1 + +2(3w_2 - 1)\varphi_2 + 2 + \lambda e^{\varphi} = 0;$$

4°.
$$\varphi_{11} + 2(2w_2 + b)\varphi_{12} + 4w_2^2\varphi_{22} - \varphi_1 + bw_2\varphi_2 + 2 + \lambda e^{\varphi} = 0;$$

5°.
$$(w_1^2+1)\varphi_{11}+2w_1w_2\varphi_{12}+(w_2^2+1)\varphi_{22}+2w_1\varphi_1+2w_2\varphi_2+2+\lambda e^{\varphi}=0;$$

6°.
$$4w_1\varphi_{11} + 4a\varphi_{12} - \varphi_{22} + 4\varphi_1 + \lambda e^{\varphi} = 0;$$
 (1.6.6)

7°.
$$-4w_1\varphi_{11} + (1+a^2w_1^{-1})\varphi_{22} + 4\varphi_1 + \lambda e^{\varphi} = 0$$

8°.
$$-\varphi_{11} + 2(w_1^2 + a^2)\varphi_{22} + \frac{\lambda}{a^2b}e^{\varphi} = 0;$$

9°.
$$\varphi_{11} + 4w_1\varphi_{12} + 4\omega_2\varphi_{22} + \lambda e^{\varphi} = 0;$$

Chapter 1. Poincare-Invariant Scalar Equations

 $10^{\circ}. \quad \alpha \varphi_{11} + 2\alpha \beta \varphi_{12} + \beta \varphi_{22} + \lambda e^{\varphi} = 0.$

Here $\varphi_k = \frac{\partial \varphi}{\partial w_k}$, and $\varphi_{kl} = \frac{\partial^2 \varphi}{\partial w_k \partial w_l}$; N 1° – 10° correspond to the numeration in Tables 1.6.1 and 1.4.1.

First of all we note that if $\alpha^2 = \alpha\beta = 0$, $\beta^2 = -\lambda/2$, then Equation 10°(1.6.6) takes the form

$$\varphi_{22} - 2e^{\varphi} = 0$$

and its general solution is

$$\varphi = \begin{cases} F(\omega_1) - 2\ln \operatorname{ch} [\omega_2 + G(\omega_1)], \\ F(\omega_1) - 2\ln \cos[\omega_2 + G(\omega_1)], \\ - 2\ln [-\omega_2 + G(\omega_1)], \end{cases}$$

where F, G are arbitrary differentiable functions. Further, if $\alpha^2 = -\beta^2 = 1$, $\alpha\beta = 0$ then 10°(1.6.6) coincides with the two-dimensional Liouville equation.

It turns out that in the case n = 2 the symmetry of Equation (1.6.1) is essentially more rich than the extended Poincare group $\tilde{P}(1,1)$ [88].

Theorem 1.6.1. The maximal (in Lie's sense) invariance group of Equation (1.6.1) with n = 2 is infinite-dimensional and generated by the operators

$$X = (f+g)\partial_0 + (f-g)\partial_1 - 2(\dot{f} + \dot{g})\partial_u, \qquad (1.6.7)$$

where $f = f(x_0 + x_1)$ and $g = g(x_0 - x_1)$ are arbitrary differentiable functions; and \dot{f}, \dot{g} are derivatives with respect to the corresponding arguments.

Proof. From the invariance condition

$$\left| \begin{array}{c} X_{2}(u_{00} - u_{11} + \lambda e^{u}) \end{array} \right|_{u_{00} - u_{11} + \lambda e^{u} = 0} = 0,$$

where X is defined in (1.1.7), we have

$$\begin{split} \left[\eta_{00} - \eta_{11} + \lambda \eta e^{u} + 2(u_{0}\eta_{0u} - u_{1}\eta_{1u}) + (u_{00} - u_{11})(\eta_{u} - u_{\sigma}\xi_{u}^{\sigma}) - \\ -2(u_{0\sigma}\xi_{0}^{\sigma} - u_{1\sigma}\xi_{1}^{\sigma}) - 2(u_{0\sigma}u_{0} - u_{1\sigma}u_{1})\xi_{u}^{\sigma} - u_{\sigma}(\xi_{00}^{\sigma} - \xi_{11}^{\sigma}) - \\ -(u_{0}^{2} - u_{1}^{2})u_{\sigma}\xi_{uu}^{\sigma} \right] \Big|_{u_{00} - u_{11} + \lambda e^{u} = 0} = 0, \end{split}$$

whence we obtain

$$\begin{aligned} \xi_{u}^{\sigma} &= 0, \quad \xi_{0}^{1} = \xi_{1}^{0}, \quad \xi_{0}^{0} = \xi_{1}^{1}, \\ \eta_{u0} &= \eta_{u1}, \quad \eta_{00} - \eta_{11} + \lambda \eta e^{u} - \lambda \eta_{u} + 2\lambda e^{u} \xi_{0}^{0} = 0 \end{aligned}$$

30

The general solution of this system has the form

$$\begin{aligned} \xi^0 &= f(x_0 + x_1) + g(x_0 - x_1), \\ \xi^1 &= f(x_0 + x_1) - g(x_0 - x_1), \\ \eta &= -2\xi_0^0 \end{aligned}$$

and determines the operator (1.6.7). The theorem is proved.

Solution (1.6.2) can be easily generalized to the *n*-dimensional Equation (1.6.1) (certainly for n > 2 it is not general):

$$u(x) = \ln \left\{ -\frac{8}{\lambda} \frac{\dot{f}(\alpha x)\dot{g}(\beta x)}{\left[f(\alpha x) + g(\beta x)\right]^2} \right\}, \qquad \alpha^2 = \beta^2 = 0, \qquad \alpha\beta = 2, \ (1.6.8)$$

where α_{ν} and β_{ν} ($\nu = \overline{0, n-1}$) are arbitrary variables satisfying the mentioned conditions, and f, g, \dot{f}, \dot{g} are arbitrary functions and their derivatives.

Putting in 1° (1.6.6) $\varphi_2 = 0$, we obtain the ODE

$$a^2 w_1^2 \varphi_{11} + a(a-1)w_1 \varphi_1 + \lambda e^u = 0$$
 (1.6.9)

whose general solution has the form

$$u(x) = \begin{cases} -2\ln\left[\frac{\sqrt{-\lambda}}{2c_1}w_1^{-1/a}\operatorname{sh}(c_1w_1^{1/a} + c_2)\right], & \lambda < 0, \\ -2\ln\left[\frac{\sqrt{\lambda}}{2c_1}w_1^{-1/a}\operatorname{ch}(c_1w_1^{1/a} + c_2)\right], & \lambda > 0, \\ -2\ln\left[\frac{\sqrt{-\lambda}}{2c_1}w_1^{-1/a}\operatorname{cos}(c_1w_1^{1/a} + c_2)\right], & \lambda < 0, \\ -2\ln\left[\frac{\sqrt{-\lambda}}{2c_1}w_1^{-1/a}(c_1w_1^{1/a} + c_2)\right], & \lambda < 0, \end{cases}$$
(1.6.10)

The ansatz 1° (1.6.5) and the formulas (1.6.10) result in the following solutions for Equation (1.6.1):

$$u(x) = \begin{cases} -2\ln \left[\rho \mathcal{P}(x) \operatorname{sh} (c_1 Q(x) + c_2)\right], \\ -2\ln \left[\delta \mathcal{P}(x) \operatorname{ch} (c_1 Q(x) + c_2)\right], \\ -2\ln \left[\rho \mathcal{P}(x) \cos(c_1 Q(x) + c_2)\right], \\ -2\ln \left[\rho \mathcal{P}(x)(Q(x) + c_2)\right], \end{cases}$$
(1.6.11)

where

$$-\rho^{2} = \delta^{2} = \frac{\lambda}{4c_{1}^{2}}, \quad \mathcal{P}(x) = (\alpha y)^{-1/a},$$

$$Q(x) = (\beta y)(\alpha y)^{1/a}, \quad \alpha^{2} = \alpha \beta = 0, \quad \beta^{2} \neq 0.$$
 (1.6.12)

We list some more solutions for Equation (1.6.1) having the structure of (1.6.11) but with other functions P(x) and Q(x):

Chapter 1. Poincare-Invariant Scalar Equations

$$\begin{aligned} \mathcal{P}(x) &= \alpha y, \quad Q(x) = \sqrt{y^2} / \alpha y; \\ \mathcal{P}(x) &= (\beta y)^2 + y^2, \quad Q(x) = \ln P(x); \\ \mathcal{P}(x) &= 1, \quad Q(x) = \beta y + a \ln (\alpha y); \\ \mathcal{P}(x) &= 1, \quad Q(x) = \beta y; \\ \mathcal{P}(x) &= F^{-1}(\alpha y), \quad Q(x) = (\beta y)F(\alpha y); \\ \mathcal{P}(x) &= F^{-1}(\alpha y), \quad Q(x) = \beta yF(\alpha y) - \ln F(\alpha y); \\ \mathcal{P}(x) &= 1, \quad Q(x) = \beta y + F(\alpha y). \end{aligned}$$
(1.6.13)

In the formulas (1.6.13) α_{ν} , β_{ν} are arbitrary parameters satisfying the conditions $\alpha^2 = \alpha\beta = 0$, $\beta^2 \neq 0$; F is an arbitrary smooth function.

From the partial solution of Equation 9° (1.6.6) and the ansatz 9° (1.6.5) we obtain one more solution for Equation (1.6.1):

$$u(x) = -\ln\left(\frac{\lambda}{2}x_{\nu}x^{\nu}\right). \qquad (1.6.14)$$

The solutions of Equation (1.6.1) can evidently be generalized for an arbitrary number of variables.

It should be stressed that all solutions of the Liouville Equation (1.6.1) obtained here have a singularity at the point $\lambda = 0$. This means that with the regular methods of perturbation theory it is impossible to obtain the approximate solutions near the exact solutions (1.6.2), (1.6.11)-(1.6.14).

1.7. Reduction and solutions of d'Alembert's equation with the nonlinearities $\sin u$, sh u

The wave equation

$$\Box u + F(u) = 0, \quad F(u) = \{ \operatorname{sh} u, \sin u \}$$
(1.7.1)

describes a wide class of physical phenomena: dislocation propagation through a crystal lattice, Bloch wall movement in magnetic crystals, propagation of a "skewed wave" along a lipid membrane, unitary theory of elementary particles, propagation of magnetic fluxes in Josephson line, *et al.* (See [1, 26, 210] and references therein.)

For the two-dimensional (n = 2) Equation (1.7.1) there are well-known soliton, soliton-antisoliton, and two-soliton solutions [170, §2.5].

As shown in §1.1 the maximal (in Lie's sense) invariance algebra of Equation (1.7.1) is the algebra AP(1, n-1) whose basis generators have the form (1.1.2). Thus we seek solutions for Equation (1.7.1) in the form

$$u(x) = \varphi(w) \tag{1.7.2}$$

In the present paragraph we construct the families of solutions of the 4dimensional (n = 4) Equation (1.7.1).

Let us use the invariance variables listed in Table 2.1.1. This gives 13 unequivalent ansatze (1.7.2). Substituting into Equation (1.7.1) we obtain 13 reduced PDEs, correspondingly:

$$\begin{split} 1^{\circ} & -\varphi_{11} - \varphi_{22} - \varphi_{33} + F(\varphi) = 0; \\ 2^{\circ} & \varphi_{11} - \varphi_{22} - \varphi_{33} + F(\varphi) = 0; \\ 3^{\circ} & -\varphi_{22} - \varphi_{33} + F(\varphi) = 0; \\ 4^{\circ} & -\varphi_{11} - \frac{1}{w_{1}}\varphi_{1} + \varphi_{22} - \varphi_{33} + F(\varphi) = 0; \\ 5^{\circ} & \varphi_{11} + \frac{1}{w_{1}}\varphi_{1} - \varphi_{22} - \varphi_{33} + F(\varphi) = 0; \\ 6^{\circ} & \varphi_{22} + \frac{2}{w_{2}}\varphi_{2} - \varphi_{33} + 2\frac{w_{1}}{w_{2}}\varphi_{12} + F(\varphi) = 0; \\ 7^{\circ} & \varphi_{11} + \frac{1}{w_{1}}\varphi_{1} - \frac{1}{w_{3}}\varphi_{22} - \varphi_{33} + \frac{1}{w_{3}}\varphi_{3} + \frac{2}{w_{1}}\varphi_{12} + F(\varphi) = 0; \\ 8^{\circ} & -\frac{1}{w_{3}^{2}}\varphi_{22} - \varphi_{33} - \frac{1}{w_{3}}\varphi_{3} + 4\epsilon\varphi_{12} + F(\varphi) = 0; \\ 9^{\circ} & -\varphi_{11} + \frac{1}{w_{1}}\varphi_{1} + (1 - \alpha^{2}w_{1}^{-2})\varphi_{12} - \varphi_{33} + F(\varphi) = 0; \\ 10^{\circ} & -\varphi_{11} + \frac{1}{w_{1}}\varphi_{1} + (1 + \alpha^{2}w_{1}^{-2})\varphi_{12} + \varphi_{33} + F(\varphi) = 0; \\ 11^{\circ} & \varphi_{11} + \frac{1}{w_{1}}\varphi_{1} - \varphi_{22} - \varphi_{33} + \frac{2\alpha}{w_{1}}\varphi_{12} + F(\varphi) = 0; \\ 12^{\circ} & 4w_{2}\varphi_{11} - \varphi_{22} - \varphi_{33} + F(\varphi) = 0; \\ 13^{\circ} & \varphi_{22} + \frac{2}{w_{2}}\varphi_{2} - (1 + w_{1}^{2})\varphi_{33} + 2\frac{w_{3}}{w_{2}}\varphi_{23} + 2\frac{w_{1}}{w_{2}}\varphi_{12} + F(\varphi) = 0; \\ \end{array}$$

Equations $1^{\circ}-13^{\circ}$ (1.7.3) correspond to the ansatz (1.7.2) with the invariants 1-13 from Table 2.1.1. The direct reduction of Equations $1^{\circ}-12^{\circ}$ (1.7.3) results in ODEs of the form

$$\ddot{\varphi} + \frac{k}{w}\dot{\varphi} = \epsilon F(u), \quad k = 0, 1, 2; \quad \epsilon = \pm 1$$
(1.7.4)

Putting in 13° (1.7.3) $\varphi_2 = 0$ we get the ODE

$$-(1+w_1^2)\,\varphi_{33}+F(\varphi)=0,$$

which, by the change of variables

$$w_3 \to w = rac{w_3}{\sqrt{1+w_1^2}} + f(w_1),$$

where $f(w_1)$ is an arbitrary differentiable function, results again in Equation (1.7.4) with k = 0 and $\epsilon = -1$:

$$-\ddot{\varphi} + F(\varphi) = 0, \quad w = \frac{x_2 + \epsilon(x_0 + x_1)x_3}{\sqrt{1 + (x_0 + x_1)^2}} + f(x_0 + x_1). \tag{1.7.5}$$

When k = 0, Equation (1.7.4) can be solved by quadrature:

$$w + c_1 = \int \frac{d\varphi}{\sqrt{2\epsilon \int F(\tau)d\tau + c}}$$
(1.7.6)

where c_1, c are arbitrary constants.

When k = 1 Equation (1.7.4) may be reduced to one of Painleve's Equations [114, 139]. If k > 1 then (1.7.4) cannot be solved by quadrature.

From (1.7.6), for $F(\varphi) = \sin \varphi$, we obtain [63, 114, 88]

$$\varphi = 4 \arctan(\alpha e^{\epsilon_0 w}) - \frac{1}{2}(1-\epsilon)\pi, \quad \epsilon_0 = \pm 1, \quad \epsilon = \pm 1, \quad \alpha = \text{const};$$

$$\varphi = 2 \arccos[\operatorname{dn}(w+\alpha,m)] + \frac{1}{2}(1+\epsilon)\pi, \quad 0 < m < 1; \quad (1.7.7)$$

$$\varphi = 2 \arccos[\operatorname{cn}(w+\alpha,m)] + \frac{1}{2}(1+\epsilon)\pi, \quad 0 < m < 1,$$

where α is an arbitrary constant, and dn(x, y), cn(x, y) are Jacobi elliptic functions (see Appendix 1).

Analogously, for $F(\varphi) = \operatorname{sh} \varphi$ from (1.7.6) we have [83] (1.7.8)

$$\begin{split} \varphi &= 2 \operatorname{arcth} \left[\operatorname{sn} (z, x) \right], \quad z = \frac{1}{2} \sqrt{c + 2} w, \quad k^2 = \frac{c - 2}{c + 2}, \quad c > 2; \\ \varphi &= 2 \operatorname{arcth} (\operatorname{sn} z'), \quad z' = \sqrt{2} w, \quad c = 2; \\ \varphi &= \operatorname{arccos} \left[\frac{\frac{1}{2} - \operatorname{sn}^2(z, k)}{\operatorname{cn}^2(w, k)} \right], \quad z = \frac{1}{2} \sqrt{c + 2} w, \quad k = \frac{4}{c + 2}, \quad c > 2; \\ \varphi &= \operatorname{arccos} \left[\frac{2 - c \operatorname{sn} (w, k)}{2 \operatorname{cn}^2(w, k)} \right], \quad k^2 = \frac{c + 2}{4}, \quad 0 < c < 2; \\ \varphi &= \operatorname{arcch} (e^w), \quad c = 2; \\ \varphi &= \operatorname{arcth} [\operatorname{cn} (w, k)]^{-1}, \quad c = 0, \quad k^2 = \frac{1}{2}; \\ \varphi &= 2 \operatorname{arcch} \left[\frac{c}{2} \operatorname{cn}^2(z, k) + \operatorname{sn}^2(z, x) \right], \quad z = \frac{1}{2} \sqrt{c + 2} w, \quad k^2 = \frac{c - 2}{c + 2}, \quad c > 2. \end{split}$$

In formulas (1.7.7) and (1.7.8) it follows from the previous analysis that w may take the values

$$x_0, x_1, x_2 + f(x_0 + x_1), \quad \frac{x_2 + \epsilon(x_0 + x_1)x_3}{\sqrt{1 + (x_0 + x_1)^2}} + f(x_0 + x_1), \quad (1.7.9)$$

where $f(x_0 + x_1)$ is an arbitrary differentiable function.

Applying to (1.7.7) and (1.7.8) the operation of group generation it is easy to construct P(1,3)-ungenerative families of solutions for Equation (1.7.1). These solutions have the form (1.7.7), (1.7.8) where the variable ω takes one of the following values:

$$dy, ay, by + f(ay + dy), \quad \frac{by + \epsilon(ay + dy)cy}{\sqrt{1 + (ay + dy)^2}} + f(ay + dy),$$

where a_{ν} , b_{ν} , d_{ν} , d_{ν} are arbitrary constants satisfying the conditions (2.1.27), $y_{\nu} = x_{\nu} + \delta_{\nu}$, δ_{ν} are arbitrary constants.

1.8. Solutions of eikonal equations

In the present paragraph we shall obtain families of exact solutions of equations whose symmetry was studied in §1.2.

1. Let us consider Equation (1.2.18)

$$u_{\nu}u^{\nu} \equiv \frac{\partial u}{\partial x^{\nu}} \frac{\partial u}{\partial x_{\nu}} = 0.$$
 (1.8.1)

Equation (1.8.1) admits the infinite-dimensional algebra generated by the operators (see Theorem 1.2.3)

$$X = \eta(u) \,\partial_u$$

with an arbitrary differentiable function $\eta(u)$, and therefore it possesses a remarkable property: an arbitrary differentiable function of a solution of Equation (1.8.1) is also a solution, which can be easily confirmed.

Let us seek the solution of Equation (1.8.1) in the form

$$u(x) = \varphi(w) + g(x), \qquad (1.8.2)$$

where the variables w are listed in Table 1.4.1 and the corresponding expressions for the function g(x) have the form

$$g(x) = \left\{egin{array}{ll} \ln{(lpha y)}, & ext{for } N1-5, \ 0, & ext{for } N6-10. \end{array}
ight.$$

Substituting ansatz (1.8.2) into Equation (1.8.1) we obtain the following reduced PDE for the function $\varphi(w)$:

1°.
$$a^2 w_1^2 \varphi_1^2 - 4w_1 (aw_2 - a - 1)\varphi_1 \varphi_2 + 4w_2 (w_2 - 1)\varphi_2^2 + 2aw_1\varphi_1 - 4(w_2 - 1)\varphi_2 + 1 = 0;$$

Chapter 1. Poincare-Invariant Scalar Equations

2°.
$$\varphi_1^2 + 4\varphi_1\varphi_2 - 4w_2\varphi_2^2 + 4\varphi_2 = 0;$$

3°. $[1 - (w_2^2 - 1)^{-1}]\varphi_1^2 - 4(w_2 - 1)(\varphi_1 - w_2\varphi_2)\varphi_2 + 2\varphi_1 - 4(w_2 - 1)\varphi_2 + 1 = 0;$
4°. $\varphi_1^2 - 2(2w_2 - b)\varphi_1\varphi_2 + 4w_2\varphi_2^2 + 2\varphi_1 - 4w_2\varphi_2 + 1 = 0;$
5°. $(w_1^2 + 1)\varphi_1^2 + 2w_1w_2\varphi_1\varphi_2 + (w_2^2 + 1)\varphi_2^2 - 2(w_1\varphi_1 + w_2\varphi_2) + 1 = 0;$
6°. $4w_1\varphi_1^2 + 4a\varphi_1\varphi_2 - \varphi_2^2 - 2\varphi_2 = 1;$ (1.8.3)
7°. $-4w_1\varphi_1^2 + (1 - a^2w_1^{-1})\varphi_2^2 - 2\varphi_2 + 1 = 0;$
8°. $-\varphi_1^2 + (2w_1 + a^2)\varphi_2^2 + a^2\varphi_2 = 0;$
9°. $\varphi_1^2 + 4w_1\varphi_1\varphi_2 + 4w_2\varphi_2^2 + 4\varphi_2 = 0;$
10°. $a\varphi_1^2 + 2a\beta\varphi_1\varphi_2 + \beta^2\varphi_2^2 = 0.$

Here $\varphi_k = \frac{\partial \varphi}{\partial w_k}$. Equations $1^\circ - 10^\circ$ correspond to the reduced Equations (1.8.1) appearing as a result of substitution by ansatz $1^\circ - 10^\circ$ from Table 1.4.1, (1.8.2).

Integrating (1.8.3) we obtain with (1.8.2) the following solutions of Equation (1.8.1) [88]:

$$u(x) = F(\alpha x), \quad \alpha^2 = 0;$$
 (1.8.4)

$$u(x) = F\left(\frac{\alpha x}{x^2}\right), \quad \alpha^2 = 0; \tag{1.8.5}$$

$$u(x) = F\left(\beta x \pm \sqrt{(\beta x)^2 - x^2}\right), \quad \beta^2 = 1,$$
 (1.8.6)

where F is an arbitrary function of the mentioned arguments (this arbitrary function appears as a consequence of the property of solutions of Equation (1.8.1) stressed above).

The arbitrary dependence on u of the coefficients $x, c_{\mu\nu}, d_{\nu}$ of the infinitesimal operator (1.2.19) admitted by Equation (1.8.1) permits us to seek solutions of Equation (1.8.1) in the form

$$u = \varphi(v), \tag{1.8.7}$$

where $v = v(x, u) = \{v_1(x, u), v_2(x, u)\}.$

Substituting (1.8.7) into (1.8.1) we have

$$v_{1\nu}v_1^{\nu}\varphi_{v_1}^2 + 2v_{1\nu}v_2^{\nu}\varphi_{v_1}\varphi_{v_2} + v_{2\nu}v_2^{\nu}\varphi_{v_2}^2 = 0.$$
(1.8.8)

Owing to the fact that v_1 and v_2 are invariants with respect to some subgroup of the symmetry group of Equation (1.8.1) we can rewrite (1.8.8) in the form

$$A(v)\varphi_{v_1}^2 + 2B(v)\varphi_{v_1}\varphi_{v_2} + C(v)\varphi_{v_2}^2, \qquad (1.8.9)$$

where the functions A(v), B(v), C(v) are found from the conditions

$$\frac{v_{1\nu}v_1^{\nu}}{A(v)} = \frac{v_{1\nu}v_2^{\nu}}{B(v)} = \frac{v_{2\nu}v_2^{\nu}}{C(v)}$$
$$\varphi(v) = \Phi(J_1), \quad \varphi(v) = \Phi(J_2),$$

The set of solutions of Equation (1.8.9) is equivalent to the set of solutions of two equations

$$A(v)\varphi_{v_1} + \left[B(v) + \sqrt{B^2(v) - A(v)C(v)}\right]\varphi_{v_2} = 0,$$

$$A(v)\varphi_{v_1} + \left[B(v) - \sqrt{B^2(v) - A(v)C(v)}\right]\varphi_{v_2} = 0,$$

whose general solution may be represented in the form

$$\varphi(v) = \Phi(J_1), \quad \varphi(v) = \Phi(J_2),$$

where J_1 and J_2 are the first integrals of the equations

$$\frac{dv_1}{A(v)} = \frac{dv_2}{B(v) \pm \sqrt{B^2(v) - A(v)C(v)}},$$

It is possible to use for v_1 and v_2 the expressions for w_1 and w_2 from Table 1.4.1; however, in the present case the coefficients $\alpha_{\nu} \beta_{\nu} \delta_{\nu}$ should be considered as arbitrary functions of u. Let us list several solutions of Equation (1.8.1) obtained by this method:

$$lpha_{
u}(u)x^{
u} = x^2, \quad lpha_{
u}(u)lpha^{
u}(u) = 0;$$

 $lpha_{
u}(u)x^{
u} = x^2 - [eta_{
u}(u)x^{
u}]^2, \quad lpha^2 = lphaeta = 0, \quad eta^2 = 1;$

$$\phi(x, u) + \beta_{\nu}(u) x^{\nu} \ln \phi(x, u) = 0, \quad \beta^2 = -1,$$

where $\phi(x, u) = \beta_{\nu}(u)x^{\nu} \pm \sqrt{x^2 + (\beta_{\nu}(u)x^{\nu})^2}$

$$\alpha_{\nu}(u)x^{\nu} - \ln \frac{\sqrt{\delta_{\nu}(u)x^{\nu} + (\beta_{\nu}(u)x^{\nu})^2} \pm \beta_{\nu}(u)x^{\nu}}{(\beta_{\nu}(u)x^{\nu})^2},$$

Chapter 1. Poincare-Invariant Scalar Equations

$$\begin{aligned} \alpha^{2} &= \delta = \alpha\beta = \beta\delta = 0, \quad \alpha\delta = 2, \quad \beta^{2} = 1; \\ (\alpha_{\nu}(u)x^{\nu})^{3} + 3\alpha_{\nu}(u)x^{\nu}\beta_{\sigma}(u)x^{\sigma} + 3\delta_{\nu}(u)x^{\nu} \pm [(\alpha_{\nu}(u)x^{\nu})^{2} + 2\beta_{\nu}(u)x^{\nu} + 1]^{3/2} = 0, \\ \alpha^{2} &= \alpha\beta = \beta\delta = 0, \quad \frac{1}{2}\alpha\delta = -\beta^{2} = \delta^{2} = 1; \\ \alpha_{\nu}(u)x^{\nu} &= x^{2} + 1, \quad \alpha_{\nu}(u)\alpha^{\nu}(u) = 4. \end{aligned}$$

2. The solutions of the relativistic Hamilton equation

$$u_{\nu}u^{\nu} \equiv \frac{\partial u}{\partial x^{\nu}}\frac{\partial u}{\partial x_{\nu}} = 1$$
(1.8.10)

are sought in the form (1.5.3) when k = -1, i.e.,

$$u(x) = \begin{cases} (\alpha y)\varphi(w_1, w_2), & N \ 1^\circ - 5^\circ, \\ \varphi(w_1, w_2), & N \ 6^\circ - 10^\circ \end{cases}$$
(1.8.11)

The corresponding variables w_1 and w_2 are listed in Table 1.4.1.

As a result of substituting ansatz $1^{\circ} - 10^{\circ}$ from (1.8.11) into (1.8.10) we obtain the following reduced PDEs for the function φ :

$$\begin{split} 1^{\circ} & a^{2}w_{1}^{2}\varphi_{1}^{2} - 4w_{1}(aw_{2} - a - 1)\varphi_{1}\varphi_{2} - 4w_{2}(w_{2} - 1)\varphi_{2}^{2} + \\ & + 2aw_{1}\varphi\varphi_{1} - 4(w_{2} - 1)\varphi\varphi_{2} + \varphi^{2} = 1; \\ 2^{\circ} & \varphi_{1}^{2} + 4\varphi_{1}\varphi_{2} - 4w_{2}\varphi_{2}^{2} + 4\varphi\varphi_{2} = 1; \\ 3^{\circ} & [1 - (w_{2} - 1)^{-1}]\varphi_{1}^{2} + 4(1 - w_{2})\varphi_{1}\varphi_{2} + 4w_{2}(w_{2} - 1)\varphi_{2}^{2} + \\ & + 2\varphi\varphi_{1} + 4(1 - w_{2})\varphi\varphi_{2} + \varphi^{2} = 1; \\ 4^{\circ} & \varphi_{1}^{2} + 2(b - 2w_{2})\varphi_{1}\varphi_{2} + 4w_{2}^{2}\varphi_{2}^{2} + 2\varphi\varphi_{1} - 4w_{2}\varphi\varphi_{2} + \varphi^{2} = 1; \\ 5^{\circ} & (w_{1}^{2} + 1)\varphi_{1}^{2} + 2w_{1}w_{2}\varphi_{1}\varphi_{2} + (w_{2}^{2} + 1)\varphi_{2}^{2} - 2\varphi(w_{1}\varphi_{1} + w_{2}\varphi_{2}) + \varphi^{2} = 1; \\ 6^{\circ} & 4w_{1}\varphi_{1}^{2} + 4a\varphi_{1}\varphi_{2} - \varphi_{2}^{2} - 2\varphi_{2} = 2; \\ 7^{\circ} & -4w_{1}\varphi_{1}^{2} + (1 + a^{2}w_{1}^{-1})\varphi_{2}^{2} + 2\varphi_{2} = 0; \\ 8^{\circ} & -\varphi_{1}^{2} + (2w_{1} + a^{2})\varphi_{2}^{2} + \varphi_{2} = a^{-2}b^{-1}; \\ 9^{\circ} & \varphi_{1}^{2} + 4w_{1}\varphi_{1}\varphi_{2} + 4w_{2}\varphi_{2}^{2} + 4\varphi_{2} = 1; \\ 10^{\circ} & \alpha\varphi_{1}^{2} + 2\alpha\beta\varphi_{1}\varphi_{2} + \beta^{2}\varphi_{2}^{2} = 1. \end{split}$$

Having determined some partial solutions of Equations (1.8.12) we construct, with the corresponding formulas (1.8.11), the solutions of Equation (1.8.10). Below we list some of these solutions [88, 92]:

$$u(x) = \beta y \sin c + \sqrt{y^2 - (\beta)^2} \cos c, \quad \beta^2 = 1, \quad c = \text{const};$$
$$u(x) = (2c\alpha y)^{-1} (c^2 y^2 + (\alpha y)^2), \quad \alpha^2 = 0;$$

$$u(x) = \sqrt{y^2 + 1} - \ln \frac{\sqrt{y^2 + 1} + 1}{\alpha y}, \quad \alpha^2 = 0;$$

$$u(x) = -\beta y + \ln |(\beta y)^2 - y^2| - 2 \arctan \frac{\delta y}{\alpha y}, \alpha^2 = \alpha \beta = \beta \delta = 0, \alpha \delta = 0;$$

$$u(x) = \left[(\alpha y)^2 + 2a(\beta y)^+ a^2 \right]^{3/2} + (\alpha y)^3 + 3a(\alpha y)(\beta y) + 3a^2(\delta y) + \alpha y,$$

where $\alpha^2 = \alpha \beta = \beta \delta = 0, \alpha \delta = -\beta^2 = \delta = (6a^2)^{-1};$

$$u(x) = F(\alpha y) + \beta y,$$

where $\alpha^2 = \alpha\beta = 0$, $\beta^2 = 1$, and F is an arbitrary differentiable function;

$$\begin{split} u(x) &= \sqrt{(\alpha y)^2 + y^2} , \qquad \alpha^2 = -1; \\ u(x) &= \sqrt{y_0^2 + (\vec{\alpha} \vec{y})^2} , \qquad \vec{\alpha}^2 = 1; \\ u(x) &= \sqrt{y^2} \end{split}$$

It is possible to apply the generating formulas (see §1.4) to the family of solutions (1.8.13) in such a way as to construct new families of solutions for Equation (1.8.10). Let us adduce two such solutions. For this purpose we write down the finite transformations generated by operators K_A (1.3.2) [92]:

$$x'_{\mu} = \frac{x_{\mu} - c_{\mu}s^{2}}{1 - 2cx + 2c_{3}u + c^{A}c_{A}s^{2}},$$

$$u'(x') = \frac{u(x) - c_{3}s^{2}}{1 - 2cx + 2c_{3}u + c^{A}c_{A}s^{2}}$$
(1.8.14)

where $\mu, \nu = 0, 1, 2; s^2 = x^2 - u^2, A = 0, 1, 2, 3; x_3 \equiv u$. We take as initial solutions $u_I(x)$ the simplest of the solutions of the set (1.8.13)

$$u_I(x) = \beta x, \quad \beta^2 = 1$$

and

$$u_I(x) = \sqrt{x^2}$$

Solving the functional Equation (1.4.14) for the transformation (1.8.14) we have correspondingly [92]

$$u(x) = (2a)^{-1} \left(-1 \pm \sqrt{1 + 4a\beta x + 4a^2 x^2} \right),$$

$$a = c_3 - \beta c \neq 0, \quad \beta^2 = 1;$$
(1.8.15)

Chapter 1. Poincare-Invariant Scalar Equations

$$u(x) = (c^{A}c_{A})^{-1} \left[c_{3} \pm \left(c_{3} + c^{A}c_{A} - 2c_{A}c^{A}(cx) + (c_{A}c^{A})^{2}x^{2} \right)^{1/2} \right],$$

$$c_{A}c^{A} \neq 0, \quad A = 0, 1, 2, 3.$$
(1.8.16)

A solution of Equation (1.8.10) may be also be sought in implicit form

$$v_1 = \varphi(v) \tag{1.8.17}$$

Let us consider several simple cases.

a).
$$v_1(x,u) = \beta_A x^A, \quad v(x,u) = \alpha_A x^A.$$
 (1.8.18)

Substitution of (1.8.17) and (1.8.18) into (1.8.10) gives

$$\alpha_A \alpha^A \varphi_v^2 - 2\alpha_A \beta^A \varphi_v + \beta_A \beta^A = 0$$

and if $\alpha_A \alpha^A = \alpha_A \beta^A = \beta_A \beta^A = 0$ then

$$\beta_A x^A = \varphi(\alpha_A x^A). \tag{1.8.19}$$

Formula (1.8.19) defines the solution of Equation (1.8.10) with an arbitrary differentiable function φ .

b).
$$v_1(x,u) = x_A x^A, \quad v(x,u) = \alpha_A x^A.$$

In this case the equation for the function $\varphi(v)$ has the form

$$\alpha_A \alpha^A \varphi_v^2 - 4v \varphi_v + 4v = 0.$$

Integrating this equation and returning to (1.8.17) we obtain the solution of Equation (1.8.10):

$$x_A x^A - 2\alpha_A x^A + \alpha_A \alpha^A = 0.$$

c).
$$v_1(x,u) = \frac{x_A x^A}{\beta_A x^A}$$
, $v(x,u) = \alpha_A x^A$; $\alpha_A \alpha^A = \alpha_A \beta^A = 0$, $\beta_A \beta^A = 1$.

The equation for the function $\varphi(v)$ has the form

$$v\varphi_v + \varphi = \varphi^2.$$

Integrating this equation we obtain the solution of Equation (1.8.10)

$$\frac{(\beta_A x^A)^2}{x_A x^A} = \alpha_A x^A + 1.$$

Solutions of Equation (1.2.17) are obtained from solutions of (1.8.10) by changing in the latter the signature (1,-1,-1) to (1,-1,1).

40

The generalization of the solutions obtained here in the case of an arbitrary number of independent variables is quite evident.

1.9. Symmetry and exact solutions of the Euler-Lagrange-Born-Infeld equation

The equation

$$L(u) \equiv (1 - u_{\nu}u^{\nu})\Box u + u^{\mu}u^{\nu}u_{\mu\nu} = 0, \qquad (1.9.1)$$

where u = u(x), $x = (x_0, ..., x_{n-1})$, and

$$u_{\mu} \equiv \frac{\partial u}{\partial x_{\mu}}, \quad u_{\mu\nu} \equiv \frac{\partial^2 u}{\partial x_{\mu} \partial x_{\nu}}, \quad u^{\mu} = g^{\mu\nu} u_{\nu}, \ g^{\mu\nu} = g_{\mu\nu} = (1, -1, \dots, -1) \, \delta_{\mu\nu}$$

which we considered in §1.1 while solving the problem of group classification of second-order PDEs invariant under the group P(1,n). Equation (1.9.1) in Euclidean space generalizes the *n*-dimensional case of the equation of minimal surfaces which was first obtained by Lagrange from the Euler-Lagrange variational principle.

In the present paragraph we shall determine the maximal (in Lie's sense) invariance group of Equation (1.9.1) and then obtain some of its solutions.

Theorem 1.9.1. [87] The maximal local invariance group of Equation (1.9.1) is the extended Poincare group $\widetilde{P}(1, n)$

Proof. From the invariance condition

$$\left. \begin{array}{c} X L(u) \\ 2 \end{array} \right|_{L(u)=0} = 0$$

where X is written in (1.1.7), we obtain the defining equations for the coordinates $\xi^{\mu}(x, u)$ and $\eta(x, u)$:

$$\partial_b \xi^a + \partial_a \xi^b = 0, \quad a \neq b; \quad \partial_a \xi^0 = \partial_0 \xi^a,$$
$$\partial_u \xi^0 = \partial_0 \eta, \quad \partial_u \xi^a + \partial_a \eta = 0,$$
$$\partial_0 \xi^0 = \partial_1 \xi^1 = \dots = \partial_{n-1} \xi^{n-1} = \partial_u \eta$$

whose general solution has the form

$$\xi^{A} = c^{AB}x_{B} + d^{A}, \quad A = 0, 1, 2, \dots, n; \quad x_{n} \equiv u,$$
$$c^{AB} = -c^{BA}, \quad d^{A} = \text{const}$$

and establishes the basis of the algebra $A\widetilde{P}(1,n)$ operators. Thus, the theorem is proved.

Let us seek the exact solutions of Equation (1.9.1). We consider first the case where n = 2. By the way, let us note that the particular class of exact solutions of the two-dimensional Equation (1.9.1) had been obtained by Barbashov and Chernikov (see [17]); these solutions, as shown in [207], may be found by the godograph method.

Following $\S1.4$, we seek solutions for Equation (1.9.1) in the form

$$u(x) = f(x)\varphi(w) + g(x) \tag{1.9.2}$$

There are several cases [87].

a).
$$w = \alpha x, \ f(x) = 1, \ g(x) = \beta x; \ \alpha_{\nu}, \beta_{\nu} - \text{const}; \ \nu = 0, 1$$

In this case, for the function $\varphi(w)$ we obtain the equation

$$\left[\alpha^{2} + (\alpha_{0}\beta_{1} - \alpha_{1}\beta_{0})^{2}\right]\ddot{\varphi} = 0$$
 (1.9.3)

If $\alpha^2 + (\alpha_0\beta_1 - \alpha_1\beta_0)^2 \neq 0$, then the solution of (1.9.3) is a linear function, and consequently

$$u(x) = a_{\nu} x^{\nu} + c, \qquad (1.9.4)$$

where a_{ν} and c are arbitrary constants. When $\alpha^2 + (\alpha_0 \beta_1 - \alpha_1 \beta_0)^2 = 0$, the solution of Equation (1.9.1) has the form

$$u(x) = \varphi(\alpha x) + \beta x \tag{1.9.5}$$

where φ is an arbitrary twice-differentiable function. The solution to (1.9.5) with $\beta_0 = \beta_1 = 0$ was also obtained in [17].

b).
$$w = x^2 \equiv x_0^2 - x_1^2$$
, $f(x) = 1$, $g(x) = a \ln (x_0 + x_1)$. (1.9.6)

In this case the ansatz (1.9.2) reduced (1.9.1) to the ODE

$$(w+a^2)\ddot{\varphi} - 2w\dot{\varphi}^3 - 3a\dot{\varphi}^2 + \dot{\varphi} = 0$$

whose general solution has the form

$$\varphi(w) = \begin{cases} \frac{1}{2} \ln \left[c \left(\frac{b\sqrt{w+a^2} - a\sqrt{w+b^2}}{bw\sqrt{w+a^2} + aw\sqrt{w+b^2}} \right)^a \left(\frac{\sqrt{w+a^2} + \sqrt{w+b^2}}{\sqrt{w+a^2} - \sqrt{w+b^2}} \right)^b \right], \\ \frac{a}{2} \ln \left[c \frac{b\sqrt{w+a^2} - a\sqrt{w+b^2}}{bw\sqrt{w+a^2} + aw\sqrt{w+b^2}} \right] + b \arctan \frac{\sqrt{w+a^2}}{\sqrt{b^2 - w}}, \\ \sqrt{w + a^2} - a \ln \left[c (\sqrt{w + a^2} + a) \right]. \end{cases}$$

Whence, using (1.9.6) and (1.9.2), we obtain the solution of Equation (1.9.1):

$$u(x) = \begin{cases} \frac{1}{2} \ln \left\{ c \left(\frac{x_0 + x_1}{x_0 - x_1} \right)^a \ \text{th}^a \left[\frac{1}{4} \ln \left(\frac{a^2}{b^2} \frac{x^2 + b^2}{x^2 + a^2} \right) \right] \ \text{cth}^b \left(\frac{1}{4} \ln \frac{x^2 + b^2}{x^2 + a^2} \right) \right\}, \\ \frac{1}{2} \ln \left\{ c \left(\frac{x_0 + x_1}{x_0 - x_1} \right)^a \ \text{th}^a \left[\frac{1}{4} \ln \left(\frac{a^2}{b^2} \frac{x^2 + b^2}{x^2 + a^2} \right) \right] + b \ \arctan \sqrt{\frac{a^2 + x^2}{b^2 - x^2}} \right\} (1.9.7) \\ a \ln \left(c \frac{x_0 + x_1}{\sqrt{x^2 + a^2} + a} \right) + \sqrt{x^2 + a^2}, \end{cases}$$

where a, b, and c are arbitrary constants.

c).
$$w = \frac{x_1}{x_0}, \quad f(x) = x_0, \quad g(x) = c = \text{const.}$$
 (1.9.8)

Substitution of (1.9.8) and (1.9.2) into (1.9.1) gives the equation

$$(\varphi^2+w^2-1)\ddot{\varphi}=0$$

by integration we get the solutions of Equation (1.9.1) in the form (1.9.4) and

$$u(x) = \pm \sqrt{x_0^2 - x_1^2} + c \tag{1.9.9}$$

d). $w = x_0 + x_1$, $f(x) = \sqrt{x_0 - x_1}$, g(x) = c = const (1.9.10)

The corresponding equation for the function $\varphi(w)$ has the form

$$\varphi^2 \ddot{\varphi} - 3\varphi \dot{\varphi}^2 + 2\dot{\varphi} = 0. \tag{1.9.11}$$

Changing variables in (1.9.11),

$$\dot{\varphi} = \phi(y), \quad y = \varphi$$
 (1.9.12)

we obtain the linear equation

$$y^2\dot{\phi} - 3y\phi + 2 = 0$$

whose general solutions are given by the formula

$$\phi = \frac{c_1 y^4 + 1}{2y}.$$

Using this result and formulas (1.9.12), (1.9.10), and (1.9.2) we construct the solutions of Equation (1.9.1):

$$u(x) = \begin{cases} \pm \left[\frac{x_0 - x_1}{c_1} \operatorname{th} \left(c_1(x_0 + x_1) + c_2\right)\right]^{1/2} + c_3, \\ \pm \left[\frac{x_0 - x_1}{c_1} \operatorname{cth} \left(c_1(x_0 + x_1) + c_2\right)\right]^{1/2} + c_3, \\ \pm \left[\frac{x_0 - x_1}{c_1} \operatorname{tan} \left(c_1(x_0^2 + x_1^2) + c_2\right)\right]^{1/2} + c_3, \\ \pm \left[x_0 - x_1 + c_2(x_0 - x_1)\right]^{1/2} + c_3, \end{cases}$$
(1.9.13)

where c_1, c_2, c_3 are arbitrary constants.

e).
$$w = x_0 - x_1 + a$$
, $\ln (x_0 + x_1)$, $f(x) = \sqrt{x_0 + x_1}$,
 $g(x) = c = \text{const}$ (1.9.14)

Substituting (1.9.14) and (1.9.2) into (1.9.1) results in the ODE

$$(\varphi + 4a)\ddot{\varphi} - 4a\dot{\varphi}^3 - 3\varphi\dot{\varphi}^2 + 2\dot{\varphi} = 0$$

which, by transformation (1.9.12), may be transformed into the Riccati equation:

$$(y^2 + 4a)\dot{\phi} - 4a\phi^2 - 3y\phi + 2 = 0 \tag{1.9.15}$$

The general solution of Equation (1.9.15) is

$$\phi(y) = \frac{2a - y(c_1\sqrt{y^2 + 4a} - y)}{2a(c_1\sqrt{y^2 + 4a} - y)}.$$

Substituting this expression into (1.9.12) we obtain the following equation:

$$\frac{d\varphi}{dw} = \frac{2a - \varphi \left(c_1 \sqrt{\varphi^2 + 4a} - \varphi\right)}{2a \left(c_1 \sqrt{\varphi^2 + 4a} - \varphi\right)},$$

and integrating this we obtain, via (1.9.14) and (1.9.2), solutions for Equation (1.9.1):

$$u(x) = \pm \left[c_2 e^{c_3(x_0 - x_1)} + \frac{2}{c_3} (x_0 + x_1) \right]^{1/2} + c_4, \quad c_1 = 0;$$

$$c_2 = \left(u - \sqrt{u^2 + 4a(x_0 + x_1)} \right)^{-1} \times \\ \times \exp \left[\frac{u}{u - \sqrt{u^2 + 4a(x_0 + x_1)}} + \frac{x_0 + x_1}{2a} \right], \quad c_1 = 1;$$

$$c_{2} = \frac{(v-1)^{s}(v+1)^{1/s}(v-k_{+})^{b-}(v-k_{-})^{b+}}{(x_{0}+x_{1})\exp(\frac{x_{0}+x_{1}}{a})}, \quad c_{1}^{2} > 0;$$
(1.9.16)

$$c_{2} = \frac{(v-1)^{s}(v+1)^{1/s} \left(v^{2} - 2c_{1}v + 1\right)^{(s^{2}+1)/2s}}{(x_{0}+x_{1}) \exp\left\{\frac{2c_{1}}{\sqrt{1-c_{1}^{2}}} \arctan\frac{v-c_{1}}{\sqrt{1-c_{1}^{2}}} + \frac{x_{0}-x_{1}}{a}\right\}}, \quad c_{1}^{2} - 1 < 0,$$

where $v = u[u^2 + 4a(x_0 + x_1)]^{-1/2}$, $s = (1 - c_1)(1 + c_1)^{-1}$,

and
$$b_{\pm} = \frac{(c_1^2 + 1)(c_1^2 - 1) \pm (c_1^3 - 1)}{(c_1^2 - 1)^{3/2}}, \quad k_{\pm} = c_1 \pm \sqrt{c_1^2 - 1}.$$

Let us consider Equation (1.9.1) when $n \ge 3$. As in the two-dimensional case we seek solutions in the form (1.9.2). Let

$$u(x) = (\beta x)\varphi(w), \quad w = \frac{(\alpha x)^2}{\beta x}, \quad \alpha^2 = \alpha\beta = 0, \quad \beta^2 \neq 0 \quad (1.9.17)$$

where a, $\alpha_{\nu} \beta_{\nu}$ are arbitrary constants satisfying the indicated conditions. Substituting (1.9.17) into (1.9.1) results in the ODE

 $\ddot{\varphi}=0$

whence we obtain, if we return to (1.9.17), the solution of Equation (1.9.1)

$$u(x) = (\alpha x)^2 + \beta x, \quad \alpha^2 = \alpha \beta = 0, \quad \beta^2 \neq 0$$
(1.9.18)

One can directly verify that

$$u(x) = (\beta x)\varphi(\alpha x), \quad \alpha^2 = \alpha\beta = 0, \quad \beta^2 \neq 0$$
 (1.9.19)

where φ is an arbitrary twice-differentiable function, is also a solution of Equation (1.9.1).

Let

$$u(x) = \varphi(w) + \beta x, \quad w = \alpha x,$$

where α_{ν} , β_{ν} are arbitrary constants. Substituting this ansatz into (1.9.1) results in the solution

$$u(x) = \varphi(\alpha x) + \beta x, \quad (\alpha x)^2 + \alpha^2 (1 - \beta^2) = 0$$
 (1.9.20)

with an arbitrary twice-differentiable function φ .

Let us consider the ansatz

$$u(x) = \varphi(w), \quad w = x^2$$

For the function $\dot{\varphi}(w)$ from (1.9.1) follows the Bernoulli equation

$$2w\ddot{\varphi} + n\dot{\varphi} - 4(n-1)w\dot{\varphi}^3 = 0$$

integration of which we yields the solution of Equation (1.9.1)

$$u(x) = c_1 \int_0^{\sqrt{x^2}} \frac{d\tau}{\sqrt{1 + c_2 \tau^{2n-2}}}$$
(1.9.21)

Chapter 1. Poincare-Invariant Scalar Equations

where c_1 , c_2 are constants. When $c_1 = 1$ or $c_2 = 0$ we find from (1.9.21)

$$u(x) = \sqrt{x^2}$$

Owing to the fact that Equation (1.9.1) is invariant under transformations from the group P(1,n), where the variables $x_0, x_1, \ldots, x_{n-1}, u$ enter equally, solutions may be sought in implicit form. If, for example, we use the invariants

a).
$$w_1 = \alpha_A x^A \equiv \alpha x - \alpha_n u, \quad w_2 = \beta_A x^A \equiv \beta x - \beta_n u,$$
 (1.9.22)

or

b).
$$w_1 = x_A x^A \equiv x^2 - u^2, \quad w_2 = \beta_A x^A$$

where α_A, β_A are arbitrary constants, and the substitution

$$w_1 = \phi(w_2) \tag{1.9.23}$$

then in case a) we obtain the solution

$$\alpha_A x^A = \phi(\beta_A x^A), \quad (\alpha_A \alpha^A)(\beta_A \beta^A) - (\alpha_A \beta^A)^2 = 0 \tag{1.9.24}$$

In case b) (1.9.22) the substitution (1.9.23) reduces Equation (1.9.1) to the ODE

$$2(w_2^2 - \beta^2 \phi)\ddot{\phi} + n(4\phi - 4w_2\dot{\phi} + \beta_A \beta^A \dot{\phi}^2) = 0$$
(1.9.25)

where n is the number of independent variables x.

The substitution

$$\beta_A \beta^A \phi(w_2) = \psi(w_2) + w_2^2, \quad \beta_A \beta^A \neq 0$$

in (1.9.25) yields the equation

$$2\psi\ddot{\psi} - n\dot{\psi}^2 - 4(n-1)\psi = 0$$

whose general solution is given by the formula

$$\int_0^{\sqrt{\psi}} \frac{d\tau}{\sqrt{c_1 \tau^{2n-2} - 1}} = w_2 + c_2 \text{ or } \psi = 0.$$

From here we obtain the solution of Equation (1.9.1) in the form of the functional relation

$$(\beta_A x^A)^2 - (\beta_A \beta^A) x_A x^A + \phi(\beta_A x^A) = 0, \quad \beta_A \beta^A \neq 0.$$
(1.9.26)

46

If $\beta_A \beta^A = 0$ then the substitutions (1.9.23) and (1.9.22) reduce Equation (1.9.1) to the linear Euler equation

$$w_2^2\ddot{\phi} - 2nw_2\dot{\phi} + 2n\phi = 0$$

whose solution is

$$\phi = c_1 w_2 + c_2 w_2^{2n}$$

The corresponding solution of Equation (1.9.1) is found from the algebraic relation

$$x_A x^A = c_1 \beta_A x^A + c_2 (\beta_A x^A)^{2n}, \quad \beta_A \beta^A = 0$$

1.10. Symmetry and exact solutions of the Monge-Ampere equation

In this section we study the symmetry and construct several classes of solutions of the multi-dimensional MA equation

$$\det (u_{\mu\nu}) = 0, \quad u_{\mu\nu} \equiv \frac{\partial^2 u}{\partial x_{\mu} \partial x_{\nu}}; \quad \mu, \nu = \overline{0, n-1}$$
(1.10.1)

and also make group classifications of equations like

$$\det (u_{\mu\nu}) = F(x, u, \underline{u}), \quad \underline{u} = \{u_{\mu}\}.$$
(1.10.2)

The MA equation was generalized in the *n*-dimensional case by Pogorelov and had been used by him to solve the multi-dimensional Minkowsky problem [165]. Nowadays Equation (1.10.1) is widely used in quantum field theory.

As shown in §1.1, Equation (1.10.1) is invariant under the group P(1,n). More complete information on its symmetry gives the following statement.

Theorem 1.10.1. [86]. The maximal local invariance group of the MA equation (1.10.1) is the group $G = {IGL(n + 1, R), C(n + 1)}$, containing the group of general linear inhomogeneous transformations IGL(n + 1, R) of the space $R^{n+1} = R^n(x) \times R^1(u)$ and the group of conformal transformations C(n + 1). The basis elements of the corresponding Lie algebra have the form

$$P_{A} = \frac{\partial}{\partial x_{A}}, \quad \widetilde{J}_{AB} = x_{A}P_{B}; \quad A, B = \overline{0, n};$$

$$K_{A} = x_{A}x_{B}P_{B}; \quad (x_{n} \equiv u).$$
(1.10.3)

Proof. From the invariance condition (1.1.6)

$$\left. \underset{2}{X} \det \left(u_{\mu\nu} \right) \, \right|_{\det \left(u_{\mu\nu} \right) = 0} = 0$$

where the operator X_2 is given in (1.1.7), we find the system of defining equations for the coordinates $\xi^{\mu}(x, u)$, $\eta(x, u)$ of the infinitesimal operator $X = \xi^{\mu}\partial_{\mu} + \eta\partial_{u}$:

$$\eta_{\mu\nu} = \xi^0_{uu} = 0, \quad 2\xi^\mu_{u\nu} = \delta_{\mu\nu}\eta_{uu}$$
$$\xi^0_{\mu\nu} = \delta_{\mu0}\eta_{\nuu} + \delta_{\nu0}\eta_{\muu}$$

The general solution of this system is rather easy to derive and has the form

$$\xi^A = x^A c^B x_B + c^{AB} x_B + d^A, \quad \xi^n \equiv \eta.$$

The solution yields the formulae (1.10.3). The theorem is proved.

As the group of general linear transformations IGL contains as subgroups the Galilei group and the Lorentz group, it is possible to state that for the MA Equation (1.10.1) both the relativity principles of Galilei and Lorentz-Poincare-Einstein are true. The relativistic or the nonrelativistic symmetry of the MA equation can be singled out as follows.

Let us consider Equation (1.10.2).

Theorem 1.10.2. [86] Equation (1.10.2) is invariant under algebra AP(1, n)

$$P_A = \frac{\partial}{\partial x^A}, \quad J_{AB} = x_A P_B - x_B P_A \tag{1.10.4}$$

iff

$$F(x, u, u_{1}) = \lambda (1 - u_{\nu} u^{\nu})^{\frac{n+2}{2}}, \quad \lambda = \text{const.}$$
(1.10.5)

Equation (1.10.2) is invariant under the Galilei algebra AG(2, n-1)

$$P_A = \frac{\partial}{\partial x^A}, \ J_{AB} = x_a P_b - x_b P_a; \ G_{1a} = x_0 P_a + m x_a P_n,$$

$$G_{2a} = x_n P_a + m x_a P_0; \quad a, b = \overline{1, n-1}$$
(1.10.6)

iff

$$F(x, u, u) = \lambda (u_0 + \frac{u_a u_a}{2m})^{\frac{n+2}{2}}; \quad \lambda, m = \text{const}$$
(1.10.7)

Proof. Necessity. As in the proof of the previous theorem we start from the invariance condition (1.10.8)

$$X_{2}\left[\det(u_{\mu\nu}) - F(x, u, u)\right] \Big|_{\det(u_{\mu\nu}) = F} = 0$$
(1.10.8)

where the operator X_2 is constructed via the formulae (1.1.7), ξ^{μ} and η are defined in (1.10.4) for the Poincare group or in (1.10.6) for the Galilei group. For the function $F(x, u, \psi)$ from (1.10.8) we obtain the equation

$$\xi^{\mu} \left(\frac{\partial F}{\partial u_{\mu}} + \frac{\partial F}{\partial x_{\mu}} \right) + \eta \frac{\partial F}{\partial u} + \left[2 \left(\frac{\partial \xi^{\mu}}{\partial x^{\mu}} + u_0 \frac{\partial \xi^0}{\partial u} \right) - n \left(\eta_u - u_0 \frac{\partial \xi^0}{\partial u} \right) \right] F = 0 \quad (1.10.9)$$

In the case of the algebra (1.10.4) it follows that

$$F(x, u, u_{1}) = F(w), \quad w = u_{\nu}u^{\nu},$$
$$2(1-w)\frac{dF}{dw} + (n+2)F = 0.$$

The general solution of these relations is given in (1.10.5). Analogously, for the set of operators (1.10.6) we obtain from (1.10.9)

$$F(x, u, u) = F(W), \quad W = u_0 + \frac{1}{2m}u_a u^a,$$
$$2W\frac{dF}{dW} + (n+2)F = 0$$

whence follows the formula (1.10.7).

Sufficiency. It is easy to confirm directly that Equation (1.10.2) with the function F from (1.10.5) and (1.10.7) is invariant under the groups P(1,n) and G(2,n-1), respectively. The theorem is proved.

Let us note that the equation

$$\det (u_{\mu\nu}) = \lambda \left(1 - u_{\nu} u^{\nu} \right)^{\frac{n+2}{2}}$$

is no longer Galilei-invariant and the equation

$$\det(u_{\mu\nu}) = \lambda \left(u_0 + \frac{u_a u_a}{2m} \right)^{\frac{n+2}{2}}$$
(1.10.10)

is no longer Lorentz-invariant.

We seek the solutions of Equation (1.10.1) in the form [86]

$$u(x) = \varphi(w) \tag{1.10.11}$$

where w = w(x) is some differentiable function. Ansatz (1.10.11) reduces (1.10.1) to the linear ODE

$$M(w)\ddot{\varphi} + N(w)\dot{\varphi} = 0,$$
 (1.10.12)

where $N(w) = \det(w_{\mu\nu})$,

M(w) =

 $= \begin{vmatrix} w_0^2 & w_{01} & \dots & w_{0 n-1} \\ w_0 w_1 & w_{11} & \dots & w_{1 n-1} \\ \dots & \dots & \dots & \dots \\ w_0 w_{n-1} & w_{n-1 1} & \dots & w_{n-1 n-1} \end{vmatrix} + \begin{vmatrix} w_{00} & w_0 w_1 & \dots & w_{0 n-1} \\ w_{10} & w_1^2 & \dots & w_{1 n-1} \\ \dots & \dots & \dots & \dots \\ w_{n-1 0} & w_{n-1} w_1 & \dots & w_{n-1 n-1} \end{vmatrix} + \begin{vmatrix} w_{00} & w_0 w_1 & \dots & w_{0 n-1} \\ w_{10} & w_1^2 & \dots & w_{1 n-1} \\ \dots & \dots & \dots & \dots \\ w_{n-1 0} & w_{n-1} w_1 & \dots & w_{n-1 n-1} \end{vmatrix} + \begin{vmatrix} w_{00} & w_0 w_1 & \dots & w_{0 n-1} \\ w_{10} & w_1^2 & \dots & w_{1 n-1} \\ \dots & \dots & \dots & \dots \\ w_{n-1 0} & w_{n-1} w_1 & \dots & w_{n-1 n-1} \end{vmatrix} + \begin{vmatrix} w_{00} & w_0 w_1 & \dots & w_{0 n-1} \\ w_{10} & w_1^2 & \dots & w_{1 n-1} \\ \dots & \dots & \dots & \dots \\ w_{n-1 0} & w_{n-1} w_1 & \dots & w_{n-1 n-1} \end{vmatrix} + \begin{vmatrix} w_{00} & w_0 w_1 & \dots & w_{0 n-1} \\ w_{10} & w_1^2 & \dots & w_{1 n-1} \\ \dots & \dots & \dots & \dots \\ w_{n-1 0} & w_{n-1} w_1 & \dots & w_{n-1 n-1} \end{vmatrix} + \begin{vmatrix} w_{10} & w_{10} & w_{10} \\ w_{10} & w_{10} & \dots & \dots \\ w_{n-1 0} & w_{n-1} w_{1} & \dots & w_{n-1 n-1} \end{vmatrix} + \begin{vmatrix} w_{10} & w_{10} & \dots & \dots \\ w_{n-1 0} & w_{n-1} w_{1} & \dots & w_{n-1 n-1} \end{vmatrix} + \begin{vmatrix} w_{10} & w_{10} & w_{10} \\ w_{10} & w_{10} & \dots & \dots \\ w_{n-1 0} & w_{n-1} w_{1} & \dots & w_{n-1 n-1} \end{vmatrix} + \begin{vmatrix} w_{10} & w_{10} & \dots & \dots \\ w_{n-1 0} & w_{n-1} & \dots & \dots \\ w_{n-1 0}$

$$+\ldots+\begin{vmatrix} w_{00} & w_{01} & \ldots & w_{0}w_{n-1} \\ w_{10} & w_{11} & \ldots & w_{1}w_{n-1} \\ \vdots \\ w_{n-10} & w_{n-11} & \ldots & w_{n-1}^2 \end{vmatrix} \equiv \frac{d}{d\tau}N(w) \mid_{\frac{d}{d\tau}w_{\mu\nu}} \stackrel{\text{def}}{=} w_{\mu}w_{\nu}$$

The proof of this statement can be carried out by the method of mathematical induction. In the two-dimensional case, using direct calculation, we obtain

$$\det \left(\ddot{\varphi}w_{\mu}w_{\nu} + \dot{\varphi}w_{\mu\nu}\right) = \dot{\varphi}\ddot{\varphi}(w_{0}^{2}w_{11} + w_{1}^{2}w_{00} - 2w_{0}w_{1}w_{01}) + \dot{\varphi}^{2}(w_{00}w_{11} - w_{01}^{2}) \equiv \\ \equiv \dot{\varphi}\left[\dot{\varphi}\det\left(w_{\mu\nu}\right) + \ddot{\varphi}(w_{0}^{2}w_{11} + w_{1}^{2}w_{00} - 2w_{0}w_{1}w_{01})\right]$$

The calculations for n > 2 are carried out analogously. Let

$$w = \alpha x, \quad \alpha_{\mu} = \text{const}$$
 (1.10.13)

In this case it follows from (1.10.2) that the solution of Equation (1.10.1) is an arbitrary differentiable function φ , i.e.,

$$u(x) = \varphi(\alpha x) \tag{1.10.14}$$

Solution (1.10.14) can be generalized. It is easy to verify that the function

$$u(x) = \varphi(w_1, w_2, \dots, w_{n-1}) \tag{1.10.15}$$

where $w_k = \alpha_{\nu}^{(k)} x^{\nu}$, $\alpha_{\nu}^{(k)}$ are arbitrary constants and $k = \overline{1, n-1}$ satisfies (1.10.1).

Let us put $w = x^2$ in (1.10.11). Then

$$N(w) = 2^n, \quad M(w) = 2^{n+1}w$$

and Equation (1.10.12) takes the form

$$2w\ddot{\varphi} + \dot{\varphi} = 0 \tag{1.10.16}$$

The general solution of (1.10.16) is given by the expression

$$\varphi(\omega) = c_1 \sqrt{\omega} + c_2, \qquad (c_1, c_2) = \text{const.}$$

Thus we obtain the solution of Equation (1.10.1)

$$u(x) = c_1 \sqrt{x_\nu x^\nu} + c_2$$

Let us list some more families of solutions for Equation (1.10.1) in explicit and implicit form:

$$u(x) = (\alpha x)^2 - \alpha^2 x^2,$$

$$u(x) = x^2 / \alpha x,$$

$$u^2 - x^2 = c(\alpha_n u - \alpha x)^2,$$

$$\alpha x - \alpha_n u = \varphi(\beta x - \beta_n u).$$

(1.10.17)

In the formulae (1.10.17) $c, \alpha_{\nu}, \beta_{\nu}, \alpha_n, \beta_n$ are arbitrary constants, with $\alpha x \equiv \alpha_{\nu} x^{\nu}, \ \nu = \overline{0, n-1}$, and φ is an arbitrary twice-differentiable function.

1.11.* Symmetry of the scalar wave equation with interaction

The standard approach of describing interaction of scalar particle (spin s = 0) with external electromagnetic field consists in the following: one has to substitute $\pi_{\mu} = \partial_{\mu} - eA_{\mu}$ (A_{μ} is the vector-potential of the electromagnetic field) for ∂_{μ} in the equation of motion of a free scalar particle. As a result one gets the equation

$$\pi_{\mu}\pi^{\mu}u = m^{2}u, \quad \text{that is}$$

$$\partial_{\mu}\partial^{\mu}u - eu\partial_{\mu}A^{\mu} - 2eA_{\mu}\partial^{\mu}u + e^{2}A_{\mu}A^{\mu}u = m^{2}u.$$
(1.11.1)

Let us generalize this equation as follows

$$\partial_{\mu}\partial^{\mu}u + \lambda_{1}u\partial_{\mu}A^{\mu} + \lambda_{2}A_{\mu}\partial^{\mu}u + \lambda_{3}A_{\mu}A^{\mu}u = m^{2}u, \qquad (1.11.2)$$

where λ_1 , λ_2 , λ_3 are arbitrary constants.

Equations (1.11.1), (1.11.2) are invariant neither under the Poincare group P(1,3) nor under the Lorentz group O(1,3), the vector-potential A_{μ} being given as arbitrary functions of x. However, if we treat these equations as nonlinear ones, the vector-potential being considered as an arbitrary vector field equal in rights with the scalar field u, we get that the equations possess nontrivial symmetry. From this point of view, Equations (1.11.2) actually mean an infinite set of Equations (1.11.2) for different A_{μ} [59^{*}]. The most physically interesting case, when $\lambda_2 = 2\lambda_1 = -2e$, $\lambda_3 = e^2$, is selected from the set of Equations (1.11.2) due to its symmetry properties.

^{*} This section is written in collaboration with R.Z.Zhdanov.

Let us consider first the case m = 0. The symmetry operators are looked for in the form

$$Q = \xi^{\mu}(x, u, A) \frac{\partial}{\partial x^{\mu}} + \eta(x, u, A) \frac{\partial}{\partial u} + \eta^{\mu}(x, u, A) \frac{\partial}{\partial A^{\mu}}.$$
 (1.11.3)

Theorem 1.11.1 Depending on λ_1 , λ_2 , λ_3 Equation (1.11.2) with m = 0 is invariant under the following Lie algebras:

1) if

$$\lambda_1 \neq 0, \quad \lambda_3 \neq 0, \quad \lambda_2^2 - 4\lambda_3 = 0, (\lambda_1\lambda_2 - 2\lambda_3)^2 + (\lambda_2 - 2\lambda_1)^2 \neq 0,$$
(1.11.4)

then IA is $\{AP(1, n), A_1^{\infty}\}$ with elements

$$P_{\mu} = \partial_{\mu} \equiv \frac{\partial}{\partial x^{\mu}}, \quad J_{\mu\nu} = x_{\mu}P_{\nu} - x_{\nu}P_{\mu} + S_{\mu\nu},$$

$$D = x^{\nu}P_{\nu} - A^{\nu}\partial_{A^{\nu}} + ku\partial_{u} \qquad (1.11.5)$$

$$(S_{\mu\nu} \equiv A_{\mu}\partial_{A^{\nu}} - A_{\nu}\partial_{A^{\mu}})$$

and

$$Q = \lambda_2 F(x) u \partial_u - 2 \frac{\partial F}{\partial x_\mu} \frac{\partial}{\partial A^\mu}, \qquad (1.11.6)$$

where F(x) is an arbitrary solution of the wave equation $\Box F = 0$, k is an arbitrary constant;

2) if

$$\lambda_1 = -e, \quad \lambda_2 = -2e, \quad \lambda_3 = e^2, \quad e \neq 0$$
 (1.11.7)

then IA is the conformal algebra AC(1,n) with basis elements (1.11.5), k = (1-n)/2 and

$$K_{\mu} = 2x_{\mu}D - x^2 P_{\mu} + 2S_{\mu\nu}x^{\nu}, \qquad (1.11.8)$$

and A_1^{∞} generated by operators

$$Q = eF(x)u\partial_u + \frac{\partial F}{\partial x_\mu}\frac{\partial}{\partial A^\mu}$$
(1.11.9)

where F = F(x) is an arbitrary smooth function;

$$\lambda_1 \neq 0, \quad \lambda_3 \neq 0, \quad \lambda_2^2 - 4\lambda_3 \neq 0, \tag{1.11.10}$$

then IA is AC(1,n) with basis elements (1.11.5), where
$$k = (1 - n) \frac{\lambda_1 \lambda_2 - 2\lambda_3}{\lambda_2^2 - 4\lambda_3}, \text{ and}$$

$$\widetilde{K}_{\mu} = K_{\mu} + \lambda \partial_{A^{\mu}}, \quad E = u \partial_u,$$

$$(\lambda = 2(1 - n)(\lambda_2 - 2\lambda_1)/(\lambda_2^2 - 4\lambda_3))$$

(1.11.11)

4) if

$$\lambda_1 \neq 0, \quad \lambda_2 = \lambda_3 = 0, \tag{1.11.12}$$

then IA is $A\widetilde{P}(1,n)$ (1.11.5) and A_3^{∞} generated by operators

$$Q_1 = a(x)u\partial_u - \frac{2}{\lambda_1}\frac{\partial a}{\partial x_\mu}\ln u\frac{\partial}{\partial A^\mu},$$
(1.11.13)

$$Q_2 = F^{\mu}(x) \frac{\partial}{\partial A^{\mu}}, \quad Q_3 = \partial_u + \frac{1}{u} A^{\mu} \frac{\partial}{\partial A^{\mu}},$$

where a, F^{μ} satisfy conditions

$$\Box a = 0, \quad \partial_{\mu} F^{\mu}(x) = 0;$$

5) if

$$\lambda_1 \neq 0, \quad \lambda_2 \neq 0, \quad \lambda_3 = 0, \tag{1.11.14}$$

then IA is AC(1,n) with basis elements (1.11.5), where $k = (1 - n)\lambda_1/\lambda_2$, (1.11.11) with $\lambda = 2(1 - n)\lambda_2^{-2}(\lambda_2 - 2\lambda_1)$ and A_1^{∞} generated by operators

$$Q = u^{-\lambda_2/\lambda_1} F^{\mu}(x) \frac{\partial}{\partial A^{\mu}}, \qquad \partial_{\mu} F^{\mu} = 0; \qquad (1.11.15)$$

6) if

$$\lambda_1 = \lambda_2 = 0, \qquad \lambda_3 \neq 0, \tag{1.11.16}$$

then IA is AC(1,n) with basis elements (1.11.5), (1.11.8), where k = (1 - n)/2, and A_2^{∞} generated by operators

$$E = u\partial_u, \qquad Q = b(x)\partial_u + B^{\mu}(x, u, A)\frac{\partial}{\partial A^{\mu}}, \qquad (1.11.17)$$

where b, B^{μ} are arbitrary solutions of the equations

$$\Box b + \lambda_3 b A_{\nu} A^{\nu} + 2\lambda_3 u A_{\nu} B^{\nu} = 0; \qquad (1.11.18)$$

7) if

$$\lambda_1 = 0, \quad \lambda_2 \neq 0, \quad \lambda_3 \neq 0, \quad \lambda_2^2 - 4\lambda_3 = 0,$$
 (1.11.19)

Chapter 1. Poincare-Invariant Scalar Equations

then IA is $\widetilde{\operatorname{AP}}(1,n)$ (1.11.5) and A_3^{∞}

$$Q_1 = a(x)u\partial_u - \frac{2}{\lambda_2}\frac{\partial a}{\partial x_\mu}\frac{\partial}{\partial A^\mu},$$
(1.11.20)

$$Q_2 = B^{\mu}(x, u, A) \frac{\partial}{\partial A^{\mu}}, \quad \Box a = 0, \quad A_{\mu}B^{\mu} = 0$$

Proof. Using the Lie algorithm one finds that coefficients of operator (1.11.3) must satisfy the following (defining) equations:

$$\xi_u^{\mu} = \xi_{A^{\nu}}^{\mu} = \eta_{A^{\nu}} = \eta_{uu} = 0; \qquad (1.11.21)$$

$$\xi^{\mu}_{\nu} + \xi^{\nu}_{\mu} = \delta^{\mu}_{\nu} f(x) \tag{1.11.22}$$

(f(x) is an arbitrary differentiable function);

$$\lambda_1 \left[(\eta^{\mu}_{A^{\nu}} + 2\xi_0^0 \delta^{\mu}_{\nu} - \xi^{\mu}_{\nu}) u - \delta^{\nu}_{\mu} (\eta - u\eta_u) \right] = 0; \qquad (1.11.23)$$

$$\lambda_2 \left[\left(\eta^{\mu} + (2\xi_0^0 \delta_{\mu\nu} - \xi_{\nu}^{\mu}) A^{\nu} \right] - \lambda u \eta_u^{\mu} + 2\eta_{u\mu} - \Box \xi^{\mu} = 0; \qquad (1.11.24)$$

$$2\lambda_{3}u \Big[\eta^{\mu} + (2\xi_{0}^{0}\delta_{\nu}^{\mu} - \xi_{\nu}^{\mu})A^{\nu}\Big]A_{\mu} + \lambda_{1}u\eta_{u}^{\mu} +$$

$$+\lambda_{2}A_{\mu}\eta^{\mu} + \Box \eta + \lambda_{3}A_{\mu}A^{\mu}(\eta - u\eta_{u}) = 0.$$
(1.11.25)

Equation (1.11.21) yields

$$\xi^{\mu} = \xi^{\mu}(x), \quad \eta = a(x)u + b(x).$$
 (1.11.26)

The general solution of Killing equations (1.11.22) is given in (1.2.7).

Now it is easy to find the general solution of Equations (1.11.23)-(1.11.25)and thereby the general form of ξ^{μ} , η , η^{μ} . Depending on λ_1 , λ_2 , λ_3 there are seven different cases which are stated above in the theorem. So, the theorem is proved.

Theorem 1.11.2 Equation (1.11.2) with $m \neq 0$ is invariant under AP(1,n) with basis elements

$$P_{\mu} = \frac{\partial}{\partial x^{\mu}}, \quad \mu = \overline{0, n},$$

$$J_{\mu\nu} = x_{\mu}P_{\nu} - x_{\nu}P_{\mu} + A_{\mu}\partial_{A^{\nu}} - A_{\nu}\partial_{A^{\mu}}.$$
(1.11.27)

So, the treatment of Equation (1.11.2) as nonlinear system gives us the possibility to study its symmetry properties in detail.

54

Chapter 2

Systems of Poincare-invariant Nonlinear PDEs

In the present chapter we will consider systems of nonlinear PDEs that are invariant under the Poincare group P(1,3), extended Poincare group $\tilde{P}(1,3)$ and conformal group C(1,3). Ansatze for spinor fields will be constructed. The formula of generating solutions (GS) by conformal transformations is obtained for fields of arbitrary spin. Wide classes of exact solutions of nonlinear generalizations of the Dirac equation are found as well as solutions of the quantum electrodynamics nonlinear equations, coupled nonlinear equations for vector and scalar fields, the Yang-Mills equations, and some others.

2.1. Reduction and exact solutions of the nonlinear massive Dirac Equation

Let us consider a nonlinear Dirac equation for a massive spinor field

$$\left[i\gamma\cdot\partial - m - \lambda(\psi\psi)^k\right] = 0, \qquad (2.1.1)$$

where $\gamma \cdot \partial = \gamma_0 \partial_0 + \gamma_1 \partial_1 + \gamma_2 \partial_2 + \gamma_3 \partial_3$; $\partial_{\nu} = \frac{\partial}{\partial x_{\nu}}$, $\nu = \overline{0,3}$; $\psi = \psi(x)$ is a four-component spinor (column); $\psi = \psi^+ \gamma_0$, $x \in \mathbb{R}(1,3)$; and m, k, λ are arbitrary real constants; γ_{ν} are Dirac matrices

$$\gamma_0 = \begin{pmatrix} \sigma_0 & 0\\ 0 & -\sigma_0 \end{pmatrix} \qquad \gamma_a = \begin{pmatrix} 0 & \sigma_a\\ -\sigma_a & 0 \end{pmatrix}, \quad (a = 1, 2, 3) \qquad (2.1.2)$$

 σ_0 and σ_a are 2×2 unit and Pauli matrices

$$\sigma_0 = \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}; \quad \sigma_1 = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}, \quad \sigma_2 = \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix}, \quad \sigma_3 = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} (2.1.3)$$

Equation (2.1.1) under $\lambda = 0$ coincides with the linear Dirac equation discovered in 1928, which describes free particles and antiparticles with mass m

and spin s = 1/2. An interest in generalizing the Dirac equation was inspired by Louis de Broglie's ideas about the possibility of describing particles (fields) with spin s = 0, 1, 3/2, ... by virtue of the field with spin s = 1/2. de Broglie developed for this purpose the method of "fusion" [35] according to which the wave function is represented as a product of "fusable" functions. From the point of view of unitary field theory, if we introduce only one field, then the field equations should be nonlinear, as the intersection generating exited states in the form of diverse particles can only be in this case a self-intersection. One of the first attempts at a nonlinear realization of de Broglie's idea was the work of Ivanenko and his collaborators dated to 1938–1953 (see, for example, [156], Introduction). In the early fifties Heisenberg [118] put forward a wide program on the development of unitary field theory based on the following nonlinear spinor equation

$$[i\gamma \cdot \partial + \lambda(\psi\gamma_{\mu}\gamma_{5}\psi)\gamma^{\mu}\gamma_{5}]\psi = 0, \quad \gamma_{5} = i\gamma_{0}\gamma_{1}\gamma_{2}\gamma_{3} \tag{2.1.4}$$

The simplest conformally invariant nonlinear spinor equation has been proposed by Gursey [115]

$$\left[i\gamma \cdot \partial + \lambda(\psi\psi)^{1/3}\right]\psi = 0 \tag{2.1.5}$$

A wide class of conformally invariant nonlinear generalizations of the Dirac equation different from (2.1.4) and (2.1.5) was suggested in [101] and [102] (see §2.4). A simple method of constructing nonlinear spinor equations which essentially differ from (2.1.1) and (2.1.4) was pointed out in [65]. The simplest equation of this kind follows from (2.1.1) under m = 0 with the help of the replacement $\gamma_{\mu} \rightarrow \psi \gamma_{\mu} \psi$:

$$(\psi \gamma_{\mu} \psi) \partial^{\mu} \psi = 0.$$

This equation is invariant under the infinite-dimensional group of point transformations.

There are some works on exact solutions of nonlinear spinor equations: [5, 6, 19, 20, 132, 138, 149, 150]; all of them use the Heisenberg ansatz [119] only

$$\psi(x) = [f(\omega) + (\gamma \cdot x)g(\omega)]\chi, \qquad (2.1.6)$$

where $w = x^{\mu}x_{\mu}$; χ is a constant spinor; and f and g are real, differentiable functions.

Below we construct new ansatze for spinor fields and then obtain multiparameter families of exact solutions of Equation (2.1.1) and, in the next sections, exact solutions of Equation (2.1.1) under m = 0, Equation (2.1.5), and a conformally invariant generalization of the Dirac-Heisenberg equation, (2.1.4).

To find exact solutions of the nonlinear Dirac equation (2.1.1) we use the ansatz suggested in [63]

$$\psi(x) = A(x)\varphi(\omega), \qquad (2.1.7)$$

where A(x) is 4×4 matrix; $\varphi(\omega)$ is an unknown four-component function (column) depending on new variables $\omega = \omega(x) = \{\omega_1, \omega_2, \omega_3\}$. After substituting ansatz (2.1.7) into (2.1.1) we obtain

$$(i\gamma \cdot \partial A)\varphi + i\gamma_{\nu}\frac{\partial\omega_{a}}{\partial x_{\nu}}A\frac{\partial\varphi}{\partial\omega_{a}} - \left[m + \lambda(\overline{\varphi}\overline{A}A\varphi)^{k}\right]A\varphi = 0 \qquad (2.1.8)$$

Furthermore we require Equation (2.1.8) to contain addends depending on the new variables ω only, whence follows

$$A^{-1}\gamma_{\nu}\frac{\partial A}{\partial x_{\nu}} = F(\omega), \qquad A^{-1}\gamma_{\nu}\frac{\partial \omega_{a}}{\partial x_{\nu}} = F_{a}(\omega), \qquad \overline{A}A = G(\omega) \qquad (2.1.9)$$

where F, F_a , and G are arbitrary 4×4 matrices. If one succeeds in finding a partial solution A = A(x), $\omega_a = \omega_a(x)$ of system (2.1.9), then to determine the function we obtain from (2.1.8) the reduced equation

$$F(\omega)\varphi + F_a(\omega)\frac{\partial\varphi}{\partial\omega_a} + i\left[m + \lambda(\overline{\varphi}G(\omega)\varphi)^k\right]\varphi = 0.$$
 (2.1.10)

At first sight the system of PDEs (2.1.9) is not in the least simpler than the initial equation. Nevertheless algebraic-theoretic techniques give an effective algorithm for constructing vast classes of its exact solutions. Later, to determine the explicit form of ansatz (2.1.7), we shall proceed in the same way as in §1.4.

Let us use the fact that the maximal invariance algebra of Equation (2.1.1) is AP(1,3), with basis elements

$$P_0 = i\partial_0, \quad P_a = -i\partial_a,$$

$$J_{\mu\nu} = x_{\mu}P_{\nu} - x_{\nu}P_{\mu} + \frac{i}{2}(\gamma_{\mu}\gamma_{\nu} - \gamma_{\nu}\gamma_{\mu}).$$
(2.1.11)

If A(x) and $\omega(x)$ satisfy the conditions

$$(a^{\mu}P_{\mu} + c^{\mu\nu}J_{\mu\nu})A(x) = 0, \qquad (2.1.12)$$

$$[a^{\mu}P_{\mu} + c^{\mu\nu}(x_{\mu}P_{\nu} - x_{\nu}P_{\mu})]\omega(x) = 0, \qquad (2.1.13)$$

where a_{μ} , $c_{\mu\nu} = -c_{\nu\mu}$ are arbitrary real constants, they also satisfy system (2.1.9). So the problem of describing ansatze (2.1.7) for spinor fields invariant under AP(1,3) (2.1.11) is reduced to the solution of systems (2.1.12) and (2.1.13). Using nonequivalent one-dimensional subalgebras of the Poincare algebra AP(1,3) and corresponding invariant variables constructed in [163] and [114], we write down the P(1,3)-nonequivalent ansatze (2.1.7) [100].

Ν	Algebra	Invar. var. $\omega = \{\omega_1, \omega_2, \omega_3\}$	Ansatze $\psi(x) =$
1.	P_0	x_1, x_2, x_3	$\overline{\varphi(\omega)}$
2.	P_3	x_0, x_1, x_2	$arphi(\omega)$
3.	$P_0 + P_1$	$x_0 + x_1, x_1, x_3$	$arphi(\omega)$
4.	J_{12}	$(x_1^2+x_2^2)^{1/2}, x_0, x_3$	$\exp\{-\frac{1}{2}\gamma_1\gamma_2\cdot$
			$\cdot \arctan \frac{x_1}{x_2} \} \varphi(\omega)$
5.	J_{03}	$(x_0^2 - x_3^2)^{1/2}, x_1, x_2$	$\exp\{+\frac{1}{2}\gamma_0\gamma_3\cdot$
			$\cdot \ln (x_0 + x_3) \varphi(\omega)$
6.	$J_{02} + J_{12}$	$x_0 + x_1,$	$\exp\{-\tfrac{x_2}{2(x_0+x_1)}\cdot$
		$(x_0^2 - x_1^2 - x_2^2)^{1/2}, x_3$	$\cdot \gamma_2(\gamma_0+\gamma_1) \} arphi(\omega)$
7.	$\alpha J_{23} - J_{01}$	$(x_0^2 - x_1^2)^{1/2},$	$\exp\left\{\frac{1}{2}\gamma_0\gamma_1\cdot\right.$
		$\alpha \ln (x_0 + x_1) +$	$-\ln(x_0+x_1)-$
		$+ \arctan \frac{x_2}{x_2}, (x_2^2 + x_3^2)^{1/2}$	$-rac{1}{2}\gamma_2\gamma_3\cdot$
		-3	$\cdot \arctan \frac{x_2}{x_3} \Big\} \varphi(\omega)$
8.	$J_{23}-$	$x_0 + x_1, \epsilon(x_0 - x_1) +$	$\exp\{-\frac{1}{2}\gamma_2\gamma_3\cdot$
	$-\frac{\epsilon}{2}(P_0+P_1)$	$+ \arctan \frac{x_2}{x_3}, (x_2^2 + x_3^2)^{1/2}$	$\cdot \arctan \frac{x_2}{x_3} \varphi(\omega)$
9.	$\overline{J_{12}} + \alpha P_0$	$(x_1^2 + x_2^2)^{1/2},$	$\exp\{-\frac{1}{2}\gamma_1\gamma_2\cdot$
		$x_0 + lpha \arctan rac{x_1}{x_2}, x_3$	$\cdot \arctan \frac{x_1}{x_2} \Big\} \varphi(\omega)$
10.	$J_{12} + \alpha P_3$	$(x_1^2 + x_2^2)^{1/2},$	$\exp\left\{-\frac{1}{2}\gamma_1\gamma_2\cdot\right.$
		$x_3 + \alpha \arctan \frac{x_1}{x_2}, x_0$	$\cdot \arctan \frac{x_1}{x_2} \Big\} \varphi(\omega)$
11.	$J_{01} + \alpha P_2$	$(x_0^2 - x_1^2)^{1/2},$	$\exp\left\{\frac{1}{2}\gamma_0\gamma_1\cdot\right\}$
		$x_2 + \alpha \ln (x_0 + x_1), x_3$	$\cdot \ln (x_0 + x_1) \} \varphi(\omega)$
12.	$J_{02} + J_{12} +$	$x_0 - x_1 + (x_0 + x_1)x_2 +$	$\exp\{\tfrac{1}{4}(x_0+x_1)\cdot$
	$+P_0 - P_1$	$+\frac{1}{6}(x_0+x_1)^3,$	$\cdot \gamma_2(\gamma_0+\gamma_1)ig\} arphi(\omega)$
		$x_2 + \frac{1}{4}(x_0 + x_1)^2, x_3$	
13.	$J_{02} + J_{12} -$	$x_0 + x_1, (x_0^2 - x_1^2 - x_2^2)^{1/2},$	$= \exp\{-\tfrac{x_2}{2(x_0+x_1)}\cdot$
	$-\epsilon P_3$	$x_2 + \epsilon (x_0 + x_1) x_3$	$\cdot \gamma_2(\gamma_0+\gamma_1) \} \varphi$

Table 2.1.1. P(1,3)-nonequivalent ansatze for a spinor field.

Here $\alpha \neq 0$ is an arbitrary constant; $\epsilon = \pm 1$.

To illustrate how to construct the ansatze of Table 2.1.1 we consider, as an example, operator J_{12} (the fourth position of the table). In this case Equations (2.1.12) and (2.1.13) take the form

$$(x_1\partial_2 - x_2\partial_1 - \frac{1}{2}\gamma_1\gamma_2)A(x) = 0, \qquad (2.1.14)$$

$$(x_1\partial_2 - x_2\partial_1)\omega(x) = 0. (2.1.15)$$

We look for a solution of (2.1.14) in the form

$$A(x) = \exp\left\{\frac{1}{2}\gamma_1\gamma_2 f(x)\right\},\,$$

where f(x) is a scalar function. Substitution of this expression into (2.1.14) gives

$$(x_1\partial_2 - x_2\partial_1 - \frac{1}{2}\gamma_1\gamma_2)\exp\left\{\frac{1}{2}\gamma_a\gamma_2 f(x)\right\} =$$

= $\frac{1}{2}\gamma_1\gamma_2\left[(x_1\partial_2 - x_2\partial_1)f - 1\right]\exp\{\frac{1}{2}\gamma_1\gamma_2\} = 0,$

that is

$$x_1\frac{\partial f}{\partial x_2} - x_2\frac{\partial f}{\partial x_1} = 1.$$

Integrating this equation one obtains

$$f(x) = -\arctan\frac{x_1}{x_2},$$

and therefore

$$A(x) = \exp\left\{-\frac{1}{2}\gamma_1\gamma_2 \arctan\frac{x_1}{x_2}\right\}$$

Equation (2.1.15) is equivalent to the following Euler-Lagrange system

$$\frac{dx_0}{0} = \frac{dx_1}{-x_2} = \frac{dx_2}{x_1} = \frac{dx_3}{0},$$

which has first integrals of the form

$$x_0, \qquad x_1^2 + x_2^2, \qquad x_3$$

On putting

$$\omega_1 = (x_1^2 + x_2^2)^{1/2}, \qquad \omega_2 = x_0, \qquad \omega_3 = x_3$$

we finally obtain ansatz N4. For the rest of the cases calculations can be done in much the same way.

Ansatze 1-13 of Table 2.1.1 are the complete set of P(1,3)-nonequivalent ansatze for spinor fields. This means that one of them cannot be changed into another by means of a group generating procedure.

Let us substitute ansatze 1–13 from Table 2.1.1 into Equation (2.1.1). After lengthy but elementary calculations we obtain the following reduced PDEs for the function $\varphi(\omega)$ [100]:

(1)
$$\gamma_1\varphi_1 + \gamma_2\varphi_2 + \gamma_3\varphi_3 + iF\varphi = 0$$
,

(2)
$$\gamma_0\varphi_1 + \gamma_1\varphi_2 + \gamma_2\varphi_3 + iF\varphi = 0$$
,

$$(3) (\gamma_{0} + \gamma_{1})\varphi_{1} + \gamma_{2}\varphi_{2} + \gamma_{3}\varphi_{3} + iF\varphi = 0,$$

$$(4) \gamma_{2} \left(\varphi_{1} + \frac{1}{2\omega_{1}}\varphi\right) + \gamma_{0}\varphi_{2} + \gamma_{3}\varphi_{3} + iF\varphi = 0,$$

$$(5) \frac{1}{2}(\gamma_{0} + \gamma_{3})\varphi + \frac{1}{2} \left[(\gamma_{0} + \gamma_{3})\omega_{1} + \frac{1}{\omega_{1}}(\gamma_{0} - \gamma_{3}) \right] \varphi_{1} + \gamma_{1}\varphi_{2} + \gamma_{2}\varphi_{3} + iF\varphi = 0;$$

$$(6) (\gamma_0 + \gamma_1) \left(\varphi_1 + \frac{1}{2\omega_1}\varphi\right) + \left[(\gamma_0 + \gamma_1)\frac{\omega_1^2 + \omega_2^2}{2\omega_1\omega_2} - \gamma_1\frac{\omega_1}{\omega_2}\right]\varphi_2 + \gamma_3\varphi_3 + iF\varphi = 0,$$

$$(7) \frac{1}{2}(\gamma_0 + \gamma_1)\varphi + \frac{1}{2}\left[(\gamma_0 + \gamma_1)\omega_1 + \frac{1}{\omega_1}(\gamma_0 - \gamma_1)\right]\varphi_1 + \left[\alpha(\gamma_0 + \gamma_1) + \frac{1}{\omega_3}\gamma_2\right]\varphi_2 + \gamma_3\left(\frac{1}{2\omega_3}\varphi + \varphi_3\right) + iF\varphi = 0,$$

(8)
$$(\gamma_0 + \gamma_1)\varphi_1 + \left[\epsilon(\gamma_0 - \gamma_1) + \frac{1}{\omega_3}\gamma_2\right]\varphi_2 + \gamma_3\left(\frac{1}{2\omega_3}\varphi + \varphi_3\right) + iF\varphi = 0,$$

(9) $\gamma_2\left(\frac{1}{2\omega_1}\varphi + \varphi_1\right) + \left(\frac{\alpha}{\omega_1}\gamma_1 + \gamma_0\right)\varphi_2 + \gamma_3\varphi_3 + iF\varphi = 0,$

(10)
$$\gamma_2 \left(\frac{1}{2\omega_1}\varphi + \varphi_1\right) + \left(\frac{\alpha}{\omega_1}\gamma_1 + \gamma_3\right)\varphi_2 + \gamma_0\varphi_3 + iF\varphi = 0,$$

$$(11) \ \frac{1}{2}(\gamma_0 + \gamma_1)\varphi + \frac{1}{2} \left[(\gamma_0 + \gamma_1)\omega_1 + \frac{1}{\omega_1}(\gamma_0 - \gamma_1) \right] \varphi_1 + \frac{1}{\omega_1}(\gamma_0 - \gamma_1) = 0$$

+ $[\gamma_2 + \alpha(\gamma_0 + \gamma_1)] \varphi_2 + \gamma_3 \varphi_3 + iF \varphi = 0,$

$$(12) \left[(\gamma_0 + \gamma_1)\omega_2 + \gamma_0 - \gamma_1 \right] \varphi_1 + \gamma_2 \varphi_2 + \gamma_3 \varphi_3 + iF\varphi$$

$$(13) \left(\gamma_0 + \gamma_1 \right) \left(\varphi_1 - \frac{1}{2\omega_1} \varphi \right) + \left[(\gamma_0 + \gamma_1) \frac{\omega_1^2 + \omega_2^2}{2\omega_1 \omega_2} - \gamma_1 \frac{\omega_1}{\omega_2} \right] \varphi_2 + \left[(\gamma_0 + \gamma_1) \frac{\omega_3}{\omega_1} + \gamma_2 + \epsilon_1 \omega_1 \gamma_3 \right] \varphi_3 + iF\varphi = 0,$$

Here $\varphi_a \equiv \frac{\partial \varphi}{\partial \omega_a}$, $a = \overline{1,3}$; $F = [m + \lambda (\overline{\varphi} \varphi)^k]$; the equation of number n, $n = \overline{1,13}$ is obtained with the help of the *n*th ansatz of Table 2.1.1.

The next step in finding invariant solutions of Equation (2.1.1) should have been the investigation of symmetry properties of the reduced Equations 1-13(2.1.16). But the problem of establishing the maximal group of invariance of these equations is very tedious and cumbersome in itself. Naturally a question arises as to whether there is a procedure for obtaining information about the symmetry of the reduced equations from the symmetry of the initial equation. An affirmative answer to this question is given by the following statement (as for its proof, see [161]).

Theorem 2.1.1. Let G be an invariance group of some PDE and $H \subset G$ be a one parameter subgroup of G, (H being a normal divisor in G). Then the factor group G/H is an invariance group of the equation obtained through reduction of the initial equation by H-invariant solutions.

It will be noted that this theorem does not guarantee the maximality of the

group G/H, that is, the reduced equation can admit a wider symmetry group than G/H.

In our case we have G = P(1,3), and H is one of 13 subgroups with generators listed Table 2.1.1. Generally speaking, these subgroups are not normal divisors in P(1,3) and for this reason one has to construct normalizers of subgroups before using Theorem 2.1.1, that is, to find maximal subgroups of P(1,3) which normal divisors are $H_1 - H_{13}$. The next step is the construction of factor groups Nor $H_1 - Nor H_{13}$, where Nor H_i (i = 1, ..., 13) is a normalizer of a group H_i . According to Theorem 2.1.1 these factor-groups are symmetry groups of the reduced equations.

The complete realization of this program is rather cumbersome, making it the subject of a separate investigation. Therefore we will now simply make a direct reduction of Equations (2.1.16) to systems of ODEs or two-dimensional PDEs (if possible). This means we suppose that φ is dependent on one or two variables of the ω_1 , ω_2 , ω_3 . As a result we obtain:

(1)
$$\gamma_1 \varphi_1 + iF\varphi = 0$$
,

(2)
$$\gamma_0 \varphi_1 + iF \varphi = 0,$$

(2) $(\gamma_0 \varphi_1 + \gamma_0) \varphi_2 + \gamma_0 \varphi_1$

(3)
$$(\gamma_0 + \gamma_1)\varphi_1 + \gamma_2\varphi_2 + iF\varphi = 0,$$

$$(4) \gamma_{2} \left(\varphi_{1} + \frac{1}{2\omega_{1}}\varphi\right) + iF\varphi = 0, \qquad (2.1.17)$$

$$(5) \frac{1}{2}(\gamma_{0} + \gamma_{3})\varphi + \gamma_{1}\varphi_{2} + iF\varphi = 0, \qquad (2.1.17)$$

$$(5') \frac{1}{2}(\gamma_{0} + \gamma_{3})\varphi + \frac{1}{2} \left[(\gamma_{0} + \gamma_{3})\omega_{1} + \frac{1}{\omega_{1}}(\gamma_{0} - \gamma_{3})\right]\varphi_{1} + iF\varphi = 0, \qquad (6) (\gamma_{0} + \gamma_{1}) \left(\varphi_{1} + \frac{1}{2\omega_{1}}\varphi\right) + \gamma_{3}\varphi_{3} + iF\varphi = 0, \qquad (6') (\gamma_{0} + \gamma_{1}) \left(\varphi_{1} + \frac{1}{2\omega_{1}}\varphi\right) + \left[(\gamma_{0} + \gamma_{1})\frac{\omega_{1}^{2} + \omega_{2}^{2}}{2\omega_{1}\omega_{2}} - \gamma_{1}\frac{\omega_{1}}{\omega_{2}}\right]\varphi_{2} + iF\varphi = 0, \qquad (7) \frac{1}{2}(\gamma_{0} + \gamma_{1})\varphi + \gamma_{3} \left(\varphi_{3} + \frac{1}{2\omega_{3}}\varphi\right) + iF\varphi = 0, \qquad (8) \left[\epsilon(\gamma_{0} - \gamma_{1}) + \frac{1}{\omega_{3}}\gamma_{2}\right]\varphi_{2} + \left(\varphi_{3} + \frac{1}{2\omega_{3}}\varphi\right) + iF\varphi = 0, \qquad (9)$$

(8')
$$(\gamma_0 + \gamma_1)\varphi_1 + \gamma_3 \left(\varphi_3 + \frac{1}{2\omega_3}\varphi\right) + iF\varphi = 0,$$

(9) $\gamma_2 \left(\varphi_1 + \frac{1}{2\omega_1}\varphi\right) + \left(\frac{\alpha}{\omega_1}\gamma_1 + \gamma_0\right)\varphi_2 + iF\varphi = 0,$

$$(10) \gamma_{2} \left(\varphi_{1} + \frac{1}{2\omega_{1}}\varphi\right) + \left(\frac{\alpha}{\omega_{1}}\gamma_{1} + \gamma_{3}\right)\varphi_{2} + iF\varphi = 0,$$

$$(11) \frac{1}{2}(\gamma_{0} + \gamma_{1})\varphi + [\gamma_{2} + \alpha(\gamma_{0} + \gamma_{1})]\varphi_{2} + iF\varphi = 0,$$

$$(12) \gamma_{2}\varphi_{2} + iF\varphi = 0,$$

$$(13) (\gamma_{0} + \gamma_{1}) \left(\varphi_{1} + \frac{1}{2\omega_{1}}\varphi\right) + \left[(\gamma_{0} + \gamma_{1})\frac{\omega_{3}}{\omega_{1}} + \gamma_{2} + \epsilon\omega_{1}\gamma_{3}\right]\varphi_{3} + iF\varphi = 0.$$

Finding the general solutions of Equations (1), (2) from (2.1.17) presents no difficulties (see formulae (1) and (2) from (2.1.24). Consider (3) from (2.1.17) at greater length. We will rewrite it as follows:

$$\frac{\partial \varphi}{\partial \omega_2} = i\gamma_2 \frac{\partial f}{\partial \omega_2} \varphi + \gamma_2(\gamma_0 + \gamma_1)\Phi,$$

$$\frac{\partial f}{\partial \omega_2} = F \equiv \lambda (\overline{\varphi}\varphi)^k + m, \quad f = f(\omega_1, \omega_2)$$

$$\Phi = \Phi(\omega_1, \omega_2) = \frac{\partial \varphi}{\partial \omega_1}$$
(2.1.18)

The first equation from (2.1.18) is a linear inhomogeneous system of ODEs and its general solution has the form

$$\varphi = e^{i\gamma_2 f} \left[\chi(\omega_1) + \gamma_2(\gamma_0 + \gamma_1) \int^{\omega_2} e^{i\gamma_2 f} \Phi d\omega_2 \right], \qquad (2.1.19)$$

where $\chi(\omega_1)$ is a four-component spinor depending on ω_1 . Substituting (2.1.19) into (2.1.18), one can determine the explicit form of functions f and φ , but they are so unwieldy that we do not present them here. It will be noted that the maximal invariance group of Equation (3) from (2.1.17) is infinite-dimensional, and that is why one can find its general solution [212].

By the change of variables

$$\varphi(\omega_1) = \frac{1}{\sqrt{\omega_1}} \Phi(\omega_1) \tag{2.1.20}$$

Equation (4) from (2.1.17) is reduced to the form

$$\gamma_2 \dot{\Phi} + i \left[m + \lambda \omega_1^{-k} (\overline{\phi} \phi)^k \right] \Phi = 0,$$

or

$$\dot{\Phi} \equiv \frac{d\Phi}{d\omega_1} = i\gamma_2 \left[m + \lambda \omega_1^{-k} (\overline{\phi}\phi)^k\right] \Phi = 0, \qquad (2.1.21)$$

62

whence follows

$$\dot{\overline{\phi}}\phi + \overline{\phi}\dot{\phi} \equiv rac{d}{d\omega_1}(\overline{\phi}\phi) = 0,$$

that is $\overline{\phi}\phi = \text{const.}$ This allows us to look for the general solution of Equation (2.1.21) in the form

$$\phi(\omega_1) = \exp\{i\gamma_2 g(\omega_1)\}\chi,\tag{2.1.22}$$

where χ is a constant spinor. Substituting (2.1.22) into (2.1.21) we obtain the ODE for the function $g = g(\omega_1)$

$$\dot{g} = \lambda \omega^{-k} (\overline{\chi}\chi)^k + m, \qquad (2.1.23)$$

whose general solution is given in (2.1.25). The rest of the equations from (2.1.17) are solved in a similar way. Below we list their solutions.

$$(1) \varphi(\omega) = \exp \{i \approx \gamma_1 \omega_1\} \chi,$$

$$(2) \varphi(\omega) = \exp \{-i \approx \gamma_0 \omega_1\} \chi,$$

$$(3) \varphi(\omega) = \exp \{i \approx \gamma_2\} \exp \{i(\gamma_0 + \gamma_1)f(\omega_1)\} \chi,$$

$$(3') \varphi(\omega) = \exp \{i m \gamma_2 \omega_2 + f(\omega_1)\} (\gamma_0 + \gamma_1) \chi,$$

$$(4) \varphi(\omega) = \frac{1}{\sqrt{\omega_1}} \exp \{i \gamma_2 g(\omega_1)\} \chi,$$

$$(5) \varphi(\omega) = \exp \left\{i (\gamma_1 \left[\approx -\frac{i}{2}(\gamma_0 + \gamma_3) \right] \omega_2 \right\} \chi,$$

$$(6) \varphi(\omega) = \exp \left\{i (\gamma_3 \omega_3 m + f(\omega_1))\right\} (\gamma_0 + \gamma_1) \chi,$$

$$(7) \varphi(\omega) = \frac{1}{\sqrt{\omega_3}} (\gamma_0 + \gamma_1) \exp \left\{-i m \gamma_3 \omega_3\right\} \chi,$$

$$(8) \varphi(\omega) = \frac{1}{\sqrt{\omega_3}} \exp \left\{i \gamma_3 g(\omega_3)\right\} \exp \left\{i (\gamma_0 + \gamma_1) f(\omega_1)\right\} \chi,$$

$$(11) \varphi(\omega) = \exp \left\{i (\gamma_2 + \alpha(\gamma_0 + \gamma_1)) \left(\approx -\frac{i}{2}(\gamma_0 + \gamma_1) \right) \omega_2 \right\} \chi,$$

$$(12) \varphi(\omega) = \exp \left\{i \exp_2 \omega_2\right\} \chi.$$

The solutions in (2.1.24) are enumerated in the same way as the ansatze in Table 2.1.1 and Equations (2.1.16) and (2.1.17); χ is a constant spinor; $\alpha = \lambda(\overline{\chi}\chi)^k + m$; and $f(\omega)$ is an artibrary differentiable function;

$$g(\omega) = \begin{cases} \frac{\lambda}{1-k} \omega^{1-k} (\overline{\chi}\chi)^k + m\omega, & k \neq 1; \\ \lambda(\overline{\chi}\chi) \ln \omega + m\omega, & k = 1. \end{cases}$$
(2.1.25)

One can apply the procedure of generating solutions to (2.1.24) (see Introduction and §1.4). Using the explicit form of the final transformations generated by operators (2.1.11) (see Appendix 3) we present the corresponding formulae of GS in Table 2.1.2.

Table 2.1.2	Final transformations of $P(1,3)$ group and formulae of generating
	solutions.

Ν	Operator	Transformations		Formulae of GS
		x ightarrow x'	$\psi(x) ightarrow \psi'(x') =$	$\psi_{II} =$
1-4	P_{μ}	$x'_{\mu} = x_{\mu} + \delta_{\mu}$	$=\psi(x)$	$=\psi_I(x')$
5-7	$\alpha_a J_a \equiv$	$x_0'=x_0,$	$= \exp\left\{\frac{1}{2}ec{lpha}\cdotec{S} ight\}\psi(x)$	$=(\cos\frac{\alpha}{2}-\vec{lpha}\vec{S}\cdot$
	$\equiv \frac{1}{2} \epsilon_{abc} J_{bc}$	$\vec{x}' = \vec{x} \cos \alpha +$	$=\left(\cos\frac{\alpha}{2}+\right)$	$\cdot \sin rac{lpha}{2} \psi_I(x')$
	-	$+\frac{\vec{x}\times\vec{\alpha}}{\alpha}\sin\alpha+$	$+i\vec{\alpha}\vec{S}\sin\frac{\alpha}{2}\psi(x)$	
		$+\frac{ec{lpha}(ec{lpha}\cdotec{x})}{lpha^2}(1-\coslpha)$	2	
8-10	$\beta_a J_{0a}$	$x'_0 = x_0 \operatorname{ch} \beta +$	$= \exp\{-\frac{1}{2} \cdot \gamma_0 \vec{\gamma} \cdot \vec{\beta}\}.$	$= (ch \beta +$
		$+ \frac{ec{eta} \cdot ec{x}}{eta} \sh eta$	$\psi(x)$	$\left +\gamma_0\frac{\vec{\gamma}\cdot\vec{\beta}}{\beta}\operatorname{sh}\beta\right)\cdot$
		$ert ec x' = ec x + rac{ec eta(ec eta \cdot ec x)}{eta^2} \cdot$	$= (ch \beta -$	$\cdot \psi_I(x')$
		$(\operatorname{ch}\beta-1)+$	$-\gamma_0rac{ec{\gamma}\cdotec{eta}}{eta} \shetaig)\psi(x)$	
		$+x_0\frac{\vec{\beta}}{\beta}\operatorname{sh}\beta$		

Here δ_{μ} ; $\vec{\alpha} = (\alpha_1, \alpha_2, \alpha_3)$; $\vec{\beta} = (\beta_1, \beta_2, \beta_3)$ are arbitrary constants;

$$\alpha = \left(\alpha_1^2 + \alpha_2^2 + \alpha_3^2\right)^{1/2}, \qquad \beta = \left(\beta_1^2 + \beta_2^2 + \beta_3^2\right)^{1/2}; \qquad \vec{S} = (S_1, S_2, S_3)$$

are 4×4 matrices, with $S_a = \frac{i}{2} \epsilon_{abc} \gamma_b \gamma_c$. Inserting (2.1.24) into the corresponding ansatze of Table 2.1.1, and having applied the formulae of GS from Table 2.1.2, we obtain the solutions of Equation (2.1.1).

$$\psi(x) = \exp \{i \mathscr{E}(\gamma a)(ay)\} \chi; \qquad (2.1.26)$$

$$\psi(x) = \exp \{-i \mathscr{E}(\gamma d)(dy)\} \chi; \qquad (2.1.26)$$

$$\psi(x) = \exp \{i \mathscr{E}(\gamma b)(by)\} \exp \{i(\gamma a + \gamma d)f(ay + dy)\} \chi; \qquad (\psi(x) = \exp \{i (m(\gamma b)(by) + f(ay + dy))\} (\gamma a + \gamma d)\chi; \qquad (\psi(x) = \frac{1}{\sqrt{\omega}} \exp \{i (m(\gamma b)(by) + f(ay + dy))\} (\gamma a + \gamma d)\chi; \qquad (\psi(x) = \frac{1}{\sqrt{\omega}} \exp \{i (-\frac{1}{2}(\gamma a)(\gamma b) \arctan \frac{ay}{by}\} \exp \{i(\gamma a)g(\omega)\} \chi; \qquad (2.1.26)$$

$$\begin{split} \omega &= \left((ay)^2 + (by)^2 \right)^{1/2}; \\ \psi(x) &= \exp\left\{ -\frac{1}{2} (\gamma c) (\gamma d) \ln (cy + dy) \right\} \exp\left\{ (\gamma a) \left(i \varkappa + \frac{1}{2} (\gamma c + \gamma d) ay \right) \right\} \chi; \\ \psi(x) &= \exp\left\{ -\frac{by}{2(ay + dy)} (\gamma b) (\gamma a + \gamma d) \right\} \exp\left\{ i \left(m(\gamma c) (cy) + f(ay + dy) \right) \right\} (\gamma a + \gamma d) \chi; \end{split}$$

$$\psi(x) = \frac{1}{\sqrt{\omega}} \exp\left\{-\frac{1}{2}(\gamma b)(\gamma c) \arctan\frac{by}{cy} - \frac{1}{2}(\gamma a + \gamma d) \times \\ \times \ln\left(ay + dy\right)\right\} \exp\left\{i\left(m(\gamma c)\,\omega\right\}(\gamma a + \gamma d)\chi,\right\}$$

$$\omega = \left((by)^2 + (cy)^2\right)^{1/2};$$

$$\psi(x) = \frac{1}{\sqrt{\omega}} \exp\left\{-\frac{1}{2}(\gamma b)(\gamma c) \arctan\frac{by}{cy}\right\} \exp\left\{i(\gamma c)g(\omega)\right\} \times \exp\left\{i(\gamma a + \gamma d)f(ay + dy)\right\}\chi,$$

$$\begin{split} \omega &= \left((by)^2 + (cy)^2 \right)^{1/2}; \\ \psi(x) &= \exp\left\{ \frac{1}{2} (\gamma d) (\gamma a) \ln \left(ay + dy \right) \right\} \exp\left\{ i \left((\gamma b) + \alpha (\gamma a + \gamma d) \right) \times \right. \\ &\left. \left. \left(\frac{a}{2} \left(\gamma a + \gamma d \right) (by + \alpha \ln \left(ay + dy \right) \right) \right\} \right\} \\ \psi(x) &= \exp\left\{ -\frac{1}{4} (\gamma a + \gamma d) (\gamma b) (ay + dy) \right\} \exp\left\{ i \frac{a}{2} (\gamma b) \left(by + \frac{1}{4} (ay + dy)^2 \right) \right\} \\ \end{split}$$

In (2.1.26) the following notations have been used: χ is a constant spinor, $x = m + \lambda (\overline{\chi}\chi)^k$; $y_\mu = x_\mu + \delta_\mu$; $\delta_\mu, a_\mu, b_\mu, c_\mu, d_\mu$ are arbitrary real constants satisfying the conditions

$$a^{2} = a_{\nu}a^{\nu} = b^{2} = c^{2} = -d^{2} = -1;$$

$$ab = bc = dc = ac = da = db = 0$$
(2.1.27)

f(w) is arbitrary differentiable function; and g(w) is determined in (2.1.25). It will be noted that functions (2.1.27) are analytic in the constants λ and m. So, on setting $\lambda = 0$ one obtains from (2.1.26) solutions of the linear Dirac equation.

It is useful to know the covariant form of ansatze for the spinor fields represented in Table 2.1.1. Having applied to the ansatze from Table 2.1.1 the formulae of GS from Table 2.1.2, we obtain covariant P(1,3)-nonequivalent ansatze for spinor fields and list them in the following table.

N	Invariant variables ω	Ansatze $\psi(x) =$
1.	ax, bx, cx	$\varphi(\omega)$
2.	dx, ax, bx	$arphi(\omega)$
3.	ax + dx, bx, cx	$arphi(\omega)$
4.	$((ax)^2 + (bx)^2)^{1/2}, dx, cx$	$\exp\{-\frac{1}{2}(\gamma a)(\gamma b)\cdot$
5.	$\left((dx)^2 - (cx)^2\right)^{1/2}, ax, bx$	$\exp\left\{\frac{1}{2}(\gamma d)(\gamma c)\cdot \\ \cdot \ln (dx + cx)\right\}\varphi(\omega)$
6.	$ax + dx$, $\left[(dx)^2 - (ax)^2 - (bx)^2 \right]^{1/2}$,	$\exp\{-\frac{bx}{2(ax+dx)}(\gamma b)\cdot$
7.	$\frac{cx}{\left[(dx)^2 - (ax)^2\right]^{1/2}}, \alpha \ln (dx + ax) +$	$ \cdot ((\gamma a) + (\gamma d)) \} \varphi(\omega) $ $ \exp\{\frac{1}{2}(\gamma d)(\gamma a) \ln (ax +$
	+ $\arctan(bx/cx)$, $[(bx)^2 + (cx)^2]^{1/2}$	$+dx) - \frac{1}{2}(\gamma b)(\gamma c) \cdot$
8.	$ax + dx$, $\epsilon(dx - ax) + \arctan \frac{bx}{cx}$,	$\exp\{-\frac{1}{2}(\gamma b)(\gamma c)\cdot\right\}$
	$\left[(bx)^2 + (cx)^2 \right]^{1/2}$	$\cdot \arctan rac{bx}{cx} ig\} arphi(\omega)$
9.	$\left[(ax)^2 + (bx)^2\right]^{1/2}, dx + \alpha \arctan \frac{ax}{bx},$	$\expig\{-rac{1}{2}(\gamma a)(\gamma b)\cdot$
	cx	$\cdot \arctan rac{ax}{bx} ig arphi(\omega)$
10.	$\left[(ax)^2 + (bx)^2 \right]^{1/2}, cx + \alpha \arctan \frac{ax}{bx},$	$\expig\{-rac{1}{2}(\gamma a)(\gamma b)\cdot$
	dx	$\cdot \arctan rac{ax}{bx} \big\} arphi(\omega)$
11.	$\left[(dx)^2 - (ax)^2 \right]^{1/2}, bx + \alpha \ln (ax + dx),$	$\exp\{rac{1}{2}(\gamma d)(\gamma a)\cdot$
		$-\ln{(dx+ax)}\varphi(\omega)$
12.	$dx - ax + (dx + ax)bx + \frac{1}{6}(dx + ax)^3,$	$\exp\{\frac{1}{4}(dx+ax)(\gamma b)\cdot$
	$bx + \frac{1}{4}(dx + ax)^2, cx$	$\cdot (\gamma a + \gamma d) \} \varphi(\omega)$
13.	$dx + ax, \left[(dx)^2 - (ax)^2 - (bx)^2 \right]^{1/2},$	$\exp\{-\frac{bx}{2(ax+dx)}(\gamma b)\cdot$
	$bx + \epsilon (dx + ax)cx$	$\cdot (\gamma a + \gamma d) \big\} \varphi(\omega)$

Table 2.1.3. Covariant P(1,3)-nonequivalent ansatze for spinor fields.

In this table $\epsilon = \pm 1$; $\alpha, a_{\nu}, b_{\nu}, c_{\nu}, d_{\nu}$ are arbitrary constants satisfying conditions (2.1.27).

The ansatze given in Tables 2.1.1 and 2.1.3 do not exhaust all possible ansatze for reducing the nonlinear Dirac equation (2.1.1). For example, ansatz $[100, 212, 31^*]$

$$\psi(x) = \left[f(\omega) + i(\gamma^{\nu} \frac{\partial \omega}{\partial x_{\nu}})g(\omega)\right]\chi, \qquad \overline{\chi}\chi = 1, \qquad (2.1.28)$$

where f and g are real scalar differentiable functions, reduces (2.1.1) to the

system of ODEs

$$\dot{f} = gF, \qquad F \equiv m + \lambda (f^2 + \epsilon g^2)^k, \epsilon \dot{g} + \frac{N}{\omega}g = -fF.$$
(2.1.29)

In (2.1.29) dot means differentiation with respect to ω ; $\epsilon = \pm 1$; N = -2, -1, ..., 3; the variable $\omega = \omega(x)$ is determined from the system of PDEs of Collins type [41]

$$\frac{\partial \omega}{\partial x_{\nu}} \frac{\partial \omega}{\partial x^{\nu}} = \epsilon$$

$$\Box \omega = \frac{N}{\omega}$$
(2.1.30)

Solutions of this overdetermined system have the form $[100, 30^*, 31^*, 60^*, 122^*, 130^*]$

$$w(x) = \begin{cases} \left[(ay)^2 + (by)^2 + (cy)^2 \right]^{1/2}, & \epsilon = -1, \quad N = -2; \\ \left[(ay)^2 + (by)^2 \right]^{1/2}, & \epsilon = -1, \quad N = -1; \\ ay + v(by + dy), & \epsilon = -1, \quad N = 0; \\ dy, & \epsilon = +1, \quad N = 0; \\ \left[(dy)^2 - (ay)^2 \right]^{1/2}, & \epsilon = +1, \quad N = 1; \\ \left[(dy)^2 - (ay)^2 - (by)^2 \right]^{1/2}, & \epsilon = +1, \quad N = 2; \\ (y_{\nu} y^{\nu})^{1/2}, & \epsilon = +1, \quad N = 3; \end{cases}$$
(2.1.31)

where $y_{\nu} = x_{\nu} + \delta_{\nu}$, and $\delta_{\nu}, a_{\nu}, b_{\nu}, c_{\nu}, d_{\nu}$ are arbitrary constants satisfying (2.1.27); v is an arbitrary differentiable function. When ω is chosen as $\sqrt{x_{\nu}x^{\nu}}$ then ansatz (2.1.28) coincides with Heisenberg's [119] (2.1.6).

Let us obtain some solutions of system (2.1.29). Multiplying the first equation by f and the second by g, and then summing them we obtain

$$\frac{d}{d\omega}\left(f^2 + \epsilon g^2\right) = -\frac{2N}{\omega}g^2. \qquad (2.1.32)$$

In particular, with N = 0, (2.1.32) yields

$$f^2 + \epsilon g^2 = \text{const}$$

and in this case system (2.1.29) can be easily integrated, and its general solution has the form

$$\epsilon = -1: \quad f(\omega) = \alpha \operatorname{sh} F\omega + \beta \operatorname{ch} F\omega, \quad F = m + \lambda (\alpha^2 - \beta^2)^k$$
$$g(\omega) = \alpha \operatorname{ch} F\omega + \beta \operatorname{sh} F\omega; \qquad (2.1.33)$$

$$\epsilon = +1: \quad f(\omega) = \alpha \sin F\omega + \beta \cos F\omega, \quad F = m + \lambda (\alpha^2 + \beta^2)^k$$
$$g(\omega) = \alpha \cos F\omega - \beta \sin F\omega; \qquad (2.1.34)$$

where α and β are arbitrary constants, and ω , as follows from (2.1.31), is given by the expressions $\omega = ay + v(by + dy)$ for (2.1.33) and $\omega = dy$ for (2.1.34). When $N \neq 0$ and $m \neq 0$ we did not succeed in integrating the system (2.1.29). If $m = 0, N \neq 0$ one can seek solutions of system (2.1.29) in the form

$$f(\omega) = \frac{\alpha}{\omega^s}, \quad g(\omega) = \frac{\beta}{\omega^s}; \quad (\alpha, \beta, s \text{ are constants})$$
(2.1.35)

Substituting these expressions into (2.1.32) and (2.1.29) gives

$$s (\alpha^{2} + \epsilon \beta^{2}) = N \beta^{2},$$

$$2sk = 1, \qquad s\alpha + \lambda (\alpha^{2} + \epsilon \beta^{2})^{k} = 0,$$

$$\beta(N - \epsilon s) + \lambda \alpha (\alpha^{2} + \epsilon \beta^{2})^{k} = 0,$$

whence follows

$$s = \frac{1}{2k}, \quad 2kN - \epsilon > 0, \quad \alpha = -\beta\sqrt{2kN - \epsilon}$$

$$\beta = \frac{(2kN - \epsilon)^{1/4k}}{(2k\lambda)^{1/2k}\sqrt{2kN}}$$
(2.1.36)

It is seen from (2.1.36) that solutions of Equation (2.1.1), determined by formulae (2.1.28) and (2.1.35), unlike solutions (2.1.26) are not analytical in the coupling constant λ . Therefore it is of interest to consider the case when $\lambda = 0, m \neq 0$, that is, the standard Dirac equation. System (2.1.29) under $\lambda = 0$ yields

$$\ddot{f} + \frac{\epsilon N}{\omega} \dot{f} + \epsilon m^2 f = 0,$$

$$\epsilon \ddot{g} + \frac{N}{\omega} \dot{g} + \left(m^2 - \frac{N}{\omega^2}\right) g = 0$$
(2.1.37)

Equations (2.1.37), when $\epsilon = 1$, are reduced by the change of variables

$$f(\omega) = c_1 \omega^{(1-N)/2} Z_{\pm \frac{1-N}{2}}(m\omega),$$

$$g(\omega) = c_2 \omega^{(1-N)/2} Z_{\pm \frac{1+N}{2}}(m\omega)$$
(2.1.38)

 $(c_1, c_2 \text{ are arbitary constants})$ to the Bessel equation for cylindrical function $Z_j(m\omega)$. Under $\epsilon = -1$ solutions of Equations (2.1.37) can be expressed in terms of modified cylindrical functions.

68

To conclude this section we write down P(1,3)-nonequivalent ansatze for vector fields calculated according to formula $A_{\mu} = \overline{\psi} \gamma_{\mu} \psi$ [100]. Invariant variables ω and other notations are the same as in Table 2.1.3.

Table 2.1.4. P(1,3)-nonequivalent ansatze for vector fields.

N	Ansatze $A_{\mu}(x) =$
1-3	$B_{\mu}(\omega)$
4	$\left[g_{\mu\sigma} + \left(1 - \frac{bx}{\omega_1}\right)(b_{\mu}b_{\sigma} + a_{\mu}a_{\sigma}) + \frac{ax}{\omega_1}(b_{\mu}a_{\sigma} - b_{\sigma}a_{\mu})\right]B^{\sigma}(\omega)$
5	$\left[g_{\mu\sigma}+\frac{(dx+cx)^2-1}{2(dx+cx)}(d_{\mu}c_{\sigma}-c_{\mu}d_{\sigma})+\right.$
	$+rac{(dx+cx-1)^2}{2(dx+cx)}(d_\mu d_\sigma-c_\mu c_\sigma)\Big]B^\sigma(\omega)$
6	$\left[g_{\mu\sigma} + \frac{bx}{ax+dx}\left((d_{\mu} + a_{\mu})b_{\sigma} - b_{\mu}(d_{\sigma} + a_{\sigma})\right) + \frac{1}{2}\left(\frac{bx}{ax+dx}\right)^{2}\right]$
	$\cdot (d_\mu + a_\mu) (d_\sigma + a_\sigma) \Big] B^\sigma(\omega)$
7	$\left[g_{\mu\sigma} + \frac{(dx + ax - 1)^2}{2(dx + ax)}(d_{\mu}d_{\sigma} - a_{\mu}a_{\sigma}) + \frac{(dx + ax)^2 - 1}{2(dx + ax)}(d_{\mu}a_{\sigma} - d_{\sigma}a_{\mu}) + \right]$
	$+ \left(1 - \frac{cx}{\omega_3}\right) \left(b_\mu b_\sigma + c_\mu c_\sigma\right) + \frac{bx}{\omega_3} (c_\mu b_\sigma - b_\mu c_\sigma) \Big] B^\sigma(\omega)$
8	$\left[g_{\mu\sigma} + \left(1 - \frac{cx}{\omega_3}\right)(b_{\mu}b_{\sigma} + c_{\mu}c_{\sigma}) + \frac{bx}{\omega_3}(c_{\mu}b_{\sigma} - b_{\mu}c_{\sigma})\right]B^{\sigma}(\omega)$
9	$\left[g_{\mu\sigma} + \left(1 - \frac{bx}{\omega_1}\right)(b_{\mu}b_{\sigma} + a_{\mu}a_{\sigma}) + \frac{ax}{\omega_1}(b_{\mu}a_{\sigma} - a_{\mu}b_{\sigma})\right]B^{\sigma}(\omega)$
10	$\left[g_{\mu\sigma} + \left(1 - \frac{bx}{\omega_1}\right)(b_{\mu}b_{\sigma} + a_{\mu}a_{\sigma}) + \frac{ax}{\omega_1}(b_{\mu}a_{\sigma} - a_{\mu}b_{\sigma})\right]B^{\sigma}(\omega)$
11	$\left[g_{\mu\sigma}+rac{(dx+ax-1)^2-1}{2(dx+ax)}(d_\mu a_\sigma-d_\sigma a_\mu)+ ight.$
	$+rac{(dx+ax-1)^2}{2(dx+ax)}(d_\mu d_\sigma - a_\mu a_\sigma)\Big]B^\sigma(\omega)$
12	$\Big[g_{\mu\sigma} + \frac{1}{2}(dx + ax)\big(b_{\mu}(a_{\sigma} + d_{\sigma}) - (a_{\mu} + d_{\mu})b_{\sigma} +$
	$+rac{1}{8}(dx+ax)^2(a_\mu+d_\mu)(a_\sigma+d_\sigma)\Big]B^\sigma(\omega)$
13	$\left[g_{\mu\sigma}+rac{bx}{ax+dx}\left(b_{\sigma}(a_{\mu}+d_{\mu})-(a_{\sigma}+d_{\sigma})b_{\mu} ight)+ ight.$
	$+rac{1}{2}\left(rac{bx}{dx+ax} ight)^2(a_{\mu}+d_{\mu})(a_{\sigma}+d_{\sigma})\Big]B^{\sigma}(\omega)$

Below we consider, as an example, how to calculate ansatz N5 from Table 2.1.4. Using ansatz N5 from Table 2.1.3 and the identity $(\gamma d\gamma c)^2 = 1$ it is easy to find

$$\psi(x) = \exp\left\{\frac{1}{2}\gamma d\gamma c \ln\left(dx + cx\right)\right\}\varphi(\omega) = \left[\sum_{n=0}^{\infty} \frac{1}{(2n)!} \left(\frac{\ln\left(dx + cx\right)}{2}\right)^{2n} + \right]$$

$$+\gamma d\gamma c \sum_{n=0}^{\infty} \frac{1}{(2n+1)!} \left(\frac{\ln(dx+cx)}{2}\right)^{2n+1} \bigg] \varphi(\omega) = (\operatorname{ch} \tau + \gamma d\gamma c \operatorname{sh} \tau) \varphi(\omega),$$

where $\tau \equiv \frac{1}{2} \ln (dx + cx)$.

Next, we substitute $\psi(x)$ into the formula $A_{\mu} = \overline{\psi} \gamma_{\mu} \psi$. This gives

$$\begin{aligned} A_{\mu}(x) &= \overline{\varphi}(\omega)(\operatorname{ch}\tau - \gamma d\gamma c\operatorname{sh}\tau)\gamma_{\mu}(\operatorname{ch}\tau + \gamma d\gamma c\operatorname{sh}\tau)\varphi(\omega) = \\ &= \overline{\varphi}(\omega)\gamma_{\mu}\varphi(\omega)\operatorname{ch}^{2}\tau + \operatorname{sh}\tau\operatorname{ch}\tau\overline{\varphi}(\omega)[\gamma_{\mu},\gamma d\gamma c]\varphi(\omega) + \\ &\quad + \overline{\varphi}(\omega)\gamma c\gamma d\gamma_{\mu}\gamma d\gamma c\varphi(\omega)\operatorname{sh}^{2}\tau \end{aligned}$$

Further, we use the identities

$$\begin{aligned} [\gamma_{\mu}, \gamma d\gamma c] &= 2(d_{\mu}\gamma c - c_{\mu}\gamma d);\\ \gamma c\gamma d\gamma_{\mu}\gamma d\gamma c &= \gamma c\gamma d\left(\gamma d\gamma c\gamma_{\mu} + [\gamma_{\mu}, \gamma d\gamma c]\right) = \\ &= -\gamma_{\mu} + 2(d_{\mu}\gamma d - c_{\mu}\gamma c); \end{aligned}$$

 $\operatorname{ch}^2 \tau - \operatorname{sh}^2 \tau = 1;$

$$2 \operatorname{sh} \tau \operatorname{ch} \tau = \operatorname{sh} 2\tau = \frac{1}{2} \left(dx + cx - \frac{1}{dx + cx} \right) = \frac{(dx + cx)^2 - 1}{2(dx + cx)};$$

$$\operatorname{sh}^2 \tau = \operatorname{ch} 2\tau - 1 = \frac{1}{2} \left(dx + cx + \frac{1}{dx + cx} \right) - 1 = \frac{(dx + cx - 1)^2}{2(dx + cx)};$$

Denoting $B_{\sigma}(\omega) = \overline{\varphi}(\omega)\gamma_{\sigma}\varphi(\omega)$ we finally obtain ansatz N5 from Table 2.1.4.

In much the same way one can construct ansatz for tensor fields $F_{\mu\nu}$ via the formula $F_{\mu\nu} = \frac{i}{2}\overline{\psi}(\gamma_m u\gamma_\nu - \gamma_\nu\gamma_\mu)\psi$.

2.2 Reduction and exact solutions of the nonlinear massless Dirac equation

Provided m = 0, the symmetry of Equation (2.1.1) becomes wider. The equation

$$\left[i\gamma\partial + \lambda(\overline{\psi}\psi)^{1/2k}\right]\psi = 0 \tag{2.2.1}$$

with an arbitrary $k \neq 0$ is invariant under the extended Poincare group $\tilde{P}(1,3)$, the generators of the corresponding Lie algebra having the form of (2.1.11) and

$$D = x^{\mu} P_{\mu} + ik. \tag{2.2.2}$$

N	Algebra	Invar. var. $\omega = \{\omega_1, \omega_2, \omega_3\}$	Ansatze $\psi(x) =$
1.	P_0	x_1, x_2, x_3	$\varphi(\omega)$
2.	P_1	x_0, x_1, x_3	$arphi(\omega)$
3.	$P_0 + P_1$	$x_0 + x_1, x_2, x_3$	$arphi(\omega)$
4.	D	$x_1/x_0, x_2/x_0, x_3/x_0$	$x_0^{-k}\varphi(\omega)$
5.	$J_{01} + P_2$	$x_0^2 - x_1^2,$	$\exp\{\frac{1}{2}\gamma_0\gamma_1\cdot$
		$\ln{(x_0 + x_1)} - x_2, x_3$	$\cdot \ln (x_0 + x_1) \} \varphi(\omega)$
6.	$J_{01} + J_{12} +$	$rac{x_0^2-x_1^2-x_2^2}{x_3^2},rac{x_0-x_2}{x_3},$	$(x_0-x_2)^{-k}\exp\left\{\frac{1}{2lpha}\gamma_1\cdot\right\}$
	$+\alpha D, (\alpha \neq 0)$	$lpha rac{x_1}{x_0 - x_2} - \ln \left(x_0 - x_2 ight)$	$\cdot(\gamma_2-\gamma_0)\ln(x_0-x_2)\Big\}$
-			$\varphi(\omega)$
(.	$J_{01} + J_{12}$	$x_0 - x_2, x_3,$	$\exp\{\frac{x_1}{2(x_0-x_2)}$
0	7 . 7 .	$x_0^2 - x_1^2 - x_2^2$	$(\gamma_1(\gamma_0 - \gamma_2))\varphi(\omega)$
0.	$J_{01} + J_{12} + J$	$x_0 - x_2, x_0 - x_1 - x_2,$	$\exp\left\{\frac{1}{2\beta}\gamma_1(\gamma_2-\gamma_0)\right\}.$
	$+\beta P_3, (\beta \neq 0)$	$\beta x_1 - x_3(x_0 - x_2)$	$\varphi(\omega)$
9.	$J_{01} + J_{12} -$	$x_3+\beta(x_0-x_2),$	$\exp\left\{\frac{1}{2}\gamma_1(\gamma_2-\gamma_0)\right\}$
	$-P_0 + \beta P_3$	$2x_1 + (x_0 - x_2)^2,$	$\cdot (x_2 - x_0) \Big\} \varphi(\omega)$
		$3x_3 + 3x_1(x_0 - x_2) +$	
		$+(x_0-x_2)^3$	
10.	$J_{23} + \alpha D$	$\frac{x_0}{x_1}, \ln(x_2^2 + x_3^2) +$	$(x_2^2 + x_3^2)^{-\kappa/2}$.
		$+2\alpha \arctan \frac{x_2}{x_3},$	$\cdot \exp\{-\frac{1}{2}\gamma_1\gamma_2\cdot$
		$(x_2^2 + x_3^2)/x_0x_1$	$\cdot \arctan \frac{x_2}{x_3} \big\} \varphi(\omega)$
11.	$J_{01} + \beta J_{23} +$	$(x_0 + x_1)^{2\alpha}$	$(x_0^2 - x_1^2)^{-k/2}$.
	$+\alpha D,$	$(x_0^2 - x_1^2)^{-(\alpha+1)},$	$\cdot \exp\{\frac{1}{2}\gamma_0\gamma_1\cdot$
		$(x_0^2 - x_1^2)/(x_2^2 + x_3^2),$	$\left \cdot\frac{\ln\left(x_{0}+x_{1}\right)}{\alpha+1}-\frac{1}{2}\gamma_{2}\gamma_{3}\right $
	$(\alpha \neq -1)$	$\beta \ln (x_2^2 + x_3^2) +$	$\cdot \arctan \frac{x_2}{x_3} \big\} \varphi(\omega)$
		$+2\alpha \arctan \frac{x_2}{x_3}$	
12.	$\beta J_{23} + J_{01} -$	$x_0 + x_1, \frac{(x_0^2 - x_1^2)}{(x_0^2 + x_0^2)},$	$(x_0^2 - x_1^2)^{-k/2} \exp\{-\frac{1}{4}$
	-D	$\beta \ln (x_2^2 + x_3^2) - \beta \ln (x_2^2 + x_3^2)$	$\cdot \gamma_0 \gamma_1 \ln (x_0 + x_1) -$
		$-2 \arctan \frac{x_2}{x_2}$	$-\frac{1}{2}\gamma_2\gamma_3 \arctan \frac{x_2}{x_2} \left\{ \varphi(\omega) \right\}$
13.	$J_{01} + \alpha J_{23} +$	$[2(x_0 + x_1) + \beta]e^{2(x_1 - x_0)/\beta},$	$[2(x_0 + x_1) + \beta]^{-k/2}$.
	$D + \beta P_0$	$[2(x_0 + x_1) + \beta](x_2^2 + x_3^2)^{-1},$	$\cdot \exp\{+\frac{1}{4}\gamma_0\gamma_1\cdot$
		$\alpha \ln (x_2^2 + x_3^2) +$	$-\ln\left[2(x_0+x_1)+\beta\right] -$
		$+2 \arctan \frac{x_2}{x_3}$	$\left -\frac{1}{2}\gamma_2\gamma_3 \arctan \frac{x_2}{x_3} \right\} \varphi(\omega)$
14.	$J_{23} + \alpha_0 P_0 +$	$x_2^2 + x_3^2, \arctan \frac{x_2}{x_3} +$	$\exp\left\{-\frac{1}{2}\gamma_2\gamma_3\cdot\right\}$
	$+\alpha_1 P_1$	$+\beta_0 x_0 + \beta_1 x_1,$	$-\operatorname{arctan} \frac{x_2}{x_3} \varphi(\omega)$
		$\alpha_1 x_0 + \alpha_0 x_1,$	- U
		$\alpha_1\beta_1 - \alpha_0\beta_0 = 1$	

Table 2.2.1. $\widetilde{\mathsf{P}}(1,3)\text{-nonequivalent}$ ansatze for spinor fields

In the same way as in the previous section we shall construct the $\tilde{P}(1,3)$ nonequivalent ansatze for spinor fields [109] ($\tilde{P}(1,3)$ -nonequivalent one-dimensional subalgebras of $A\tilde{P}(1,3)$ are found in [109, 11*]; see also Appendix 2) and list them in the Table 2.2.1., in which $\alpha, \beta, \alpha_0, \beta_0, \alpha_1, \beta_1$ are arbitrary constants satisfying the indicated conditions.

Let us substitute ansatze 1–14 from Table 2.2.1 into Equation (2.2.1). Omitting some rather cumbersome calculations, we write down the resulting reduced system of PDEs for the function $\varphi(\omega)$:

$$(1) \gamma_{a}\varphi_{a} = iF\varphi,$$

$$(2.2.3)$$

$$(2) \gamma_{0}\varphi_{1} + \gamma_{2}\varphi_{2} + \gamma_{3}\varphi_{3} + iF\varphi = 0,$$

$$(3) (\gamma_{0} + \gamma_{1})\varphi_{1} + \gamma_{2}\varphi_{2} + \gamma_{3}\varphi_{3} = iF\varphi,$$

$$(4) -k\gamma_{0}\varphi + (\gamma_{a} - \gamma_{0}\omega_{a})\varphi_{a} = iF\varphi$$

$$(5) \frac{1}{2}(\gamma_{0} + \gamma_{1})\varphi + [(\gamma_{0}(\omega_{1} + 1) + \gamma_{1}(\omega_{1} - 1)]\varphi_{1} + (\gamma_{0} + \gamma_{1} - \gamma_{2})\varphi_{2} + + \gamma_{3}\varphi_{3} = iF\varphi;$$

$$(6) (k\gamma_{2} - \gamma_{0})\varphi + [(\gamma_{0} - \gamma_{2})(\omega_{1} + \alpha^{-2}\omega_{2}^{2}\omega_{3}^{2}) + (\gamma_{0} + \gamma_{2})\omega_{2}^{2} - 2\alpha^{-1}\gamma_{1}\omega_{3}\omega_{2}^{2} - 2\gamma_{3}\omega_{1}\omega_{2}]\varphi_{1} + [(\gamma_{0} - \gamma_{2})\omega_{2} - \gamma_{3}\omega_{2}^{2}]\varphi_{2} + [\alpha\gamma_{1} + (\gamma_{2} - \gamma_{0})(\omega_{3} + 1)]\varphi_{3} = iF\varphi;$$

$$(7) \frac{1}{2\omega_{1}}(\gamma_{0} - \gamma_{2})\varphi + (\gamma_{0} - \gamma_{2})\varphi_{1} + \gamma_{3}\varphi_{2} + [(\gamma_{0} + \gamma_{2})\omega_{1} - (\gamma_{0} - \gamma_{2})\omega_{3}\omega_{1}^{-1}]\varphi_{3} = iF\varphi;$$

$$(8) \frac{1}{2\beta}\gamma_{4}(\gamma_{2} - \gamma_{0})\varphi + (\gamma_{0} - \gamma_{2})\varphi_{1} + [(\gamma_{0} + \gamma_{2})\omega_{1} - 2\beta^{-1}\gamma_{1}\omega_{3} + (\gamma_{0} - \gamma_{2})(\beta^{-2}\omega_{3}^{2} + \omega_{2})\omega_{1}^{-1}]\varphi_{2} + (\beta\gamma_{1} - \gamma_{3}\omega_{1})\varphi_{3} = iF\varphi;$$

$$(9) [\gamma_{3} + \beta(\gamma_{0} - \gamma_{2})]\varphi_{1} + 2\gamma_{1}\varphi_{2} + \frac{3}{2}(2\gamma_{2} + (\gamma_{0} - \gamma_{2})\omega_{2})\varphi_{3} = iF\varphi;$$

$$(10) \frac{1}{2}(1 - 2k)\gamma_{3}\varphi + \sqrt{\omega_{1}\omega_{3}}(\gamma_{0} - \gamma_{1}\omega_{1})\varphi_{1} + 2(\gamma_{3} + \alpha\gamma_{2})\varphi_{2} + [2\gamma_{3} - (\gamma_{0} + \gamma_{1}\omega_{1})\sqrt{\frac{\omega_{3}}{\omega_{1}}}]\omega_{3}\varphi_{3} = iF\varphi;$$

$$(11) \left[-k\left(\gamma_{0} \operatorname{ch} \ln \omega_{1}^{\frac{1}{2}(\alpha+1)} - \gamma_{1} \operatorname{sh} \ln \omega_{1}^{\frac{1}{2}(\alpha+1)}\right) + \frac{1}{2}(\alpha+1)^{-1}(\gamma_{0} + \gamma_{1})\omega_{1}^{-1/2(\alpha+1)} + \frac{1}{2}\gamma_{3}\sqrt{\omega_{2}}\right]\varphi - 2(\alpha+1)\omega_{1}\left(\gamma_{0} \operatorname{ch} \ln \omega_{1}^{1/2(\alpha+1)} - \gamma_{1} \operatorname{sh} \ln \omega_{1}^{\frac{1}{2}(\alpha+1)}\right)\varphi_{1} + 2(\alpha\gamma_{2} + \beta\gamma_{3})\sqrt{\omega_{2}}\varphi_{3} = iF\varphi;$$

$$(12) \left[-k \left(\gamma_{0} \operatorname{ch} \ln \sqrt{\omega_{1}} - \gamma_{1} \operatorname{sh} \ln \sqrt{\omega_{1}} \right) + \frac{1}{4} (\gamma_{0} - \gamma_{1}) \sqrt{\omega_{1}} + \frac{1}{2} \gamma_{3} \sqrt{\omega_{2}} \right] \varphi + + (\gamma_{0} + \gamma_{1}) \sqrt{\omega_{1}} + 2\omega_{2} (\gamma_{0} \operatorname{ch} \ln \sqrt{\omega_{1}} \varphi_{1} - \gamma_{1} \operatorname{sh} \ln \sqrt{\omega_{1}} - 2\gamma_{3} \sqrt{\omega_{2}}) \varphi_{2} + + 2(\beta \gamma_{3} - \gamma_{2}) \sqrt{\omega_{2}} \varphi_{3} = iF \varphi_{3} + (\beta \gamma_{3} - \gamma_{2}) \sqrt{\omega_{2}} \varphi_{3} = iF \varphi_{3} + (\gamma_{1} - 2k)(\gamma_{0} + \gamma_{1}) + \gamma_{3} \omega_{2}] \varphi + 2 \left[(\beta - 1)\gamma_{0} + (\beta + 1)\gamma_{1} \right] \omega_{1} \varphi_{1} + + 2\omega_{2} (\gamma_{0} + \gamma_{1} - \sqrt{\omega_{2}} \gamma_{3}) \varphi_{2} + 2(\gamma_{2} + \alpha \gamma_{3}) \sqrt{\omega_{2}} \varphi_{3} = iF \varphi_{3} + (\gamma_{1} - \sqrt{\omega_{2}} \gamma_{3}) \varphi_{2} + 2(\gamma_{2} + \alpha \gamma_{3}) \sqrt{\omega_{2}} \varphi_{3} = iF \varphi_{3} + (\gamma_{1} - \gamma_{2} + \beta_{0} \gamma_{0} + \beta_{1} \gamma_{1}) \varphi_{2} + (\gamma_{1} - \gamma_{0} + \gamma_{0} \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{1} - \gamma_{0} + \gamma_{0} \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{1} - \gamma_{0} + \gamma_{0} \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{1} - \gamma_{0} + \gamma_{0} \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{1} - \gamma_{0} + \gamma_{0} \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{1} - \gamma_{0} + \gamma_{0} \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{1} - \gamma_{0} + \gamma_{0} \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} + (\gamma_{0} + \gamma_{0} - \gamma_{1}) \varphi_{3} = iF \varphi_{3}$$

In these formulae $\varphi_a \equiv \partial \varphi / \partial \omega_a$; $F \equiv \lambda (\overline{\varphi} \varphi)^{1/2k}$; an equation for the number n, where n = 1, 2, ..., 14, is obtained by the ansatz with the same number n from Table 2.2.1.

Next we perform direct reduction everywhere possible of Equations (2.2.3) to ODEs or two-dimensional PDEs. This means that we will suppose that φ is dependent on one or two variables from $\omega_1, \omega_2, \omega_3$. Omitting cases similar to those considered in the previous section we have

$$(4) -k\gamma_{0}\varphi + (\gamma_{a} - \omega_{a}\gamma_{0})\varphi_{a} = iF\varphi; \qquad (2.2.4)$$

$$(6) k(\gamma_{2} - \gamma_{0})\varphi + \omega_{2}(\gamma_{0} - \gamma_{2} - \omega_{2}\gamma_{3})\varphi_{2} = iF\varphi;$$

$$(6') k(\gamma_{2} - \gamma_{0})\varphi + [(\gamma_{2} - \gamma_{0})(1 + \omega_{3}) + \alpha\gamma_{1}])\varphi_{3} = iF\varphi;$$

$$(9) [\gamma_{3} + \beta(\gamma_{0} - \gamma_{2})]\varphi_{1} = iF\varphi;$$

$$(9') 2\gamma_{1}\varphi_{2} = iF\varphi;$$

$$(9'') [\gamma_{3} + \beta(\gamma_{0} - \gamma_{2})]\varphi_{1} + 2\gamma_{1}\varphi_{2} = iF\varphi;$$

$$(10) \frac{1}{2}(1 - 2k)\gamma_{3}\varphi + 2(\gamma_{3} + \alpha\gamma_{2})\varphi_{2} = iF\varphi.$$

Here we have used the same notations as in (2.2.3).

The general solution of Equation (9') from (2.2.4) has the form

$$\varphi(\omega_2) = \exp\left\{-\frac{i}{2} \varkappa \gamma_1 \omega_2\right\} \chi, \qquad \varkappa \equiv \lambda \left(\overline{\chi} \chi\right)^{1/2k},$$

whence follows a solution of the initial Equation (2.2.1)

$$\psi(x) = \exp\left\{\frac{\gamma_1}{2}(\gamma_0 - \gamma_2)(x_0 - x_2)\right\} \times$$

$$\times \exp\left\{-\frac{i}{2}\varpi\gamma_1\left[2x_1 + (x_0 - x_2)^2\right]\right\} \chi$$
(2.2.5)

Next we consider Equation (10) from (2.2.4). Under k = 1/2 its general solution has the form

$$\varphi(\omega_2) = \exp\left\{-\frac{1}{2}\lambda(\overline{\chi}\chi)(1+\alpha^2)^{-1}(\gamma_3+\alpha\gamma_2)\omega_2\right\}\chi$$
(2.2.6)

Under $k \neq \frac{1}{2}$ and $\alpha \neq 0$ we did not succeed in integrating Equation (10) from (2.2.4). However, if $\alpha = 0$, then making a change of variables

$$\phi(\omega_2) = \exp\left\{\frac{1}{4}(2k-1)\omega_2\right\}\phi(\omega_2)$$

we obtain

$$2\exp\left\{\frac{1}{4}(1-2k)k^{-1}\omega_2\right\}\gamma_3\phi_2=i\lambda(\overline{\phi}\phi)^{1/2k}\phi$$

The general solution of the last equation has the form

$$\phi(\omega_2) = \exp\left\{\frac{2i\lambda k}{1-2k}(\overline{\chi}\chi)^{1/2k}\exp\left\{\frac{1}{4}\frac{2k-1}{k}\omega_2\gamma_3\right\}\right\}\chi.$$

Substituting the above results into ansatz N10 from Table 2.2.1, we obtain the following solutions of Equation (2.2.1): if k = 1/2:

$$\psi(x) = (x_2^2 + x_2^2)^{-1/4} \exp\left\{-\frac{1}{2}\gamma_2\gamma_3 \arctan\frac{x_2}{x_3}\right\} \times$$
(2.2.7)

$$\times \exp\left\{-\frac{i}{2}\lambda\overline{\chi}\chi(1+\alpha^2)^{-1}(\gamma_3+\alpha\gamma_2)\left[\ln\left(x_2^2+x_2^2\right)+2\alpha\arctan\frac{x_2}{x_3}\right]\right\}\chi;$$

if $k \neq 1/2$:

$$\psi(x) = (x_2^2 + x_2^2)^{-1/4} \exp\left\{-\frac{1}{2}\gamma_2\gamma_3 \arctan\frac{x_2}{x_3}\right\} \times$$
(2.2.8)

$$\times \exp\left\{\frac{2i\lambda k}{1-2k}(\overline{\chi}\chi)^{1/2k}(x_2^2+x_2^2)^{(2k-1)/4}\gamma_3\right\}\chi.$$

Next we consider Equation (9'') from (2.2.4). This two-dimensional PDE can be reduced to the two-dimensional Dirac equation. Having made a change of variables

$$z_1 = \omega_1, \quad z_2 = \frac{1}{2}\omega_2$$

and denoting

$$\Gamma_1 = \gamma_3 + \beta(\gamma_0 - \gamma_2), \qquad \Gamma_2 = \gamma_1,$$

we obtain

$$\Gamma_a \frac{\partial \varphi}{\partial z_a} = i\lambda (\overline{\varphi}\varphi)^{1/2k}\varphi; \qquad \Gamma_a \Gamma_b + \Gamma_b \Gamma_a = 2g_{ab}; \qquad a, b = 1, 2. \ (2.2.9)$$

We look for a solution of (2.2.9) in the form

$$\varphi(z) = \left[\Gamma_a z_a f(z_b z_b) + ig(z_b z_b)\right] \chi, \qquad (2.2.10)$$

where f and g are unknown scalar functions. Substituting (2.2.10) into (2.2.9) gives the system of ODEs

$$f + \omega \dot{f} = \frac{1}{2} \lambda (\overline{\chi} \chi)^{1/2k} \left(g^2 - \omega f^2 \right)^{1/2k} g,$$
$$\dot{g} = \frac{1}{2} \lambda (\overline{\chi} \chi)^{1/2k} \left(g^2 - \omega f^2 \right)^{1/2k} f,$$

where dot means differentiation with respect to $\omega = z_a z_a$. A partial solution of this system is given by the formulae:

$$k < 0,$$

$$f \equiv f(\omega) = \sqrt{|k|} \left(\mp \frac{\left(k^2 + |k|\right)^{1/2}}{\lambda(\overline{\chi}\chi)^{1/2k}} \right)^k \omega^{-(k+1)/2},$$

$$g \equiv g(\omega) = \mp \left(1 + |k|^{-1}\right)^{-1/2} \left(\mp \frac{\left(k^2 + |k|\right)^{1/2}}{\lambda(\overline{\chi}\chi)^{1/2k}} \right)^k \omega^{-(k/2)}$$
(2.2.11)

Since Equation (2.2.9) with k = 1/2 is conformally invariant with respect to AC(1,1) (see Paragraph 2.3) one can use the ansatz

$$\varphi(z) = \frac{\Gamma_a z_a}{z_b z_b} \phi(\omega), \qquad \omega = \frac{\beta_a z_a}{z_b z_b}; \qquad a, b = 1, 2, \qquad (2.2.12)$$

which reduces (2.2.9) to the system of ODEs

$$\Gamma_a \beta_a \dot{\phi} = i\lambda(\overline{\phi}\phi)\phi,$$

whose general solution has the form

$$\phi(\omega) = \exp\left\{-i\lambda(\overline{\chi}\chi)(\beta_1^2 + \beta_2^2)(\Gamma_a\beta_a)\omega\right\}\chi$$
(2.2.13)

So, ansatz N9 from the Table 2.2.1 and formulae (2.2.10)-(2.2.13) give the following solutions of Equation (2.2.1): if k < 0,

$$\psi(x) = \exp\left\{\frac{1}{2}\gamma_1(\gamma_0 - \gamma_2)(x_0 - x_2)\right\} \left\langle\left\{[\gamma_3 + \beta(\gamma_0 - \gamma_2)] \times (2.2.14)\right\} \right\rangle$$

×
$$[x_3 + \beta(x_0 - x_2)] + \frac{1}{2}\gamma_1 [2x_1 + (x_0 - x_2)^2] f(\omega) + ig(\omega) \rangle \chi;$$

if
$$k = \frac{1}{2}$$
,
 $\psi(x) = \exp\left\{\frac{1}{2}\gamma_1(\gamma_0 - \gamma_2)(x_0 - x_2)\right\}\left\{\left[\gamma_3 + \beta(\gamma_0 - \gamma_2)\right] \times (2.2.15) \times [x_3 + \beta(x_0 - x_2)] + \frac{1}{2}\gamma_1 \left[2x_1 + (x_0 - x_2)^2\right]\right\}\omega^{-1} \times (2.2.15)$

$$\times \exp\left\{-i\lambda(\overline{\chi}\chi)(\beta_1^2+\beta_2^2)^{-1}\left\langle\beta_1\left[\gamma_3+\beta(\gamma_0-\gamma_2)\right]+\frac{1}{2}\beta_2\gamma_1\right\rangle\times\right. \\ \left.\times\left\langle\beta_1x_3+\beta(x_0-x_2)\right]+\frac{1}{2}\beta_2\left[2x_1+(x_0-x_2)^2\right]\right\rangle\omega^{-1}\right\}\chi.$$

In (2.2.14) and (2.2.15) f, g are given in (2.2.11) and

$$\omega = [x_3 + \beta(x_0 - x_2)]^2 + \frac{1}{4} \left[2x_1 + (x_0 - x_2)^2 \right]^2, \qquad (2.2.16)$$

Consider the ansatze [31*]

$$\psi(x) = \exp\left\{\frac{1}{2}\gamma_0\gamma_3\ln\left(x_0 + x_3\right)\right\}\varphi(x_0^2 - x_3^2).$$
(2.2.17)

$$\psi(x) = \exp\left\{\frac{x_1}{2(x_0 + x_3)}(\gamma_0 + \gamma_3)\gamma_1\right\} \exp\left\{\frac{1}{2}\gamma_0\gamma_3\ln(x_0 + x_3)\right\}\varphi(x_0^2 - x_1^2 - x_2^2);$$
(2.2.18)

$$\psi(x) = \exp\left\{\frac{1}{2(x_0 + x_3)}(\gamma_0 + \gamma_3)(\gamma_1 x_1 + \gamma_2 x_2\right\} \times \\ \times \exp\left\{\frac{1}{2}\gamma_0\gamma_3\ln(x_0 + x_3)\right\}\varphi(x_\nu x^\nu)$$
(2.2.19)

They reduce Equation (2.2.1) to the systems of ODEs $4\omega\dot{\varphi} = -\left\{s(1+\gamma_0\gamma_3) - i\lambda(\overline{\varphi}\varphi)^{1/2k}\left[\omega(\gamma_0+\gamma_3)+\gamma_0-\gamma_3\right]\right\}\varphi, \qquad (2.2.20)$

with s = 1, 2, 3; $\omega = \{x_0^2 - x_3^2; x_0^2 - x_1^2 - x_2^2; x_\nu x^\nu\}$ in accord with (2.2.17)-(2.2.19) and $\dot{\varphi} \equiv d\varphi/d\omega$. From (2.2.20) it follows that

$$4\omega \frac{d}{d\omega} \left(\overline{\varphi}\varphi\right) = -2s \left(\overline{\varphi}\varphi\right).$$

Hence

$$\overline{\varphi}\varphi = c\omega^{-s/2}, \quad c = \text{const}$$
 (2.2.21)

and thereby system (2.2.20) becomes the linear system

$$4\omega\dot{\varphi} = -\left\{s(1+\gamma_0\gamma_3) - i\lambda\left(c\omega^{-s/2}\right)^{1/2k}\left[\omega(\gamma_0+\gamma_3) + \gamma_0 - \gamma_3\right]\right\}\varphi.$$

Writing it out we obtain

$$\begin{aligned} &2\omega\dot{\varphi}^{0} = i\lambda'\omega^{\alpha+1}\varphi^{2}, \quad \lambda' = \lambda c^{1/2k}, \, \alpha = -s/4k, \\ &2\omega\dot{\varphi}^{1} = -s\varphi^{1} + i\lambda'\omega^{\alpha}\varphi^{3}, \\ &2\omega\dot{\varphi}^{2} = -s\varphi^{2} + i\lambda'\omega^{\alpha}\varphi^{0}, \\ &2\omega\dot{\varphi}^{3} = is\omega^{\alpha+1}\varphi^{1} \end{aligned}$$

whence follows

76

$$\begin{split} \omega^2 \ddot{\varphi}^0 &+ \frac{s - 2\alpha}{2} \omega \dot{\varphi}^0 + \frac{\lambda'^2}{4} \omega^{2\alpha + 1} \varphi^0 = 0, \\ \omega^2 \ddot{\varphi}^3 &+ \frac{s - 2\alpha}{2} \omega \dot{\varphi}^3 + \frac{\lambda'^2}{4} \omega^{2\alpha + 1} \varphi^3 = 0, \\ \varphi^2 &= -\frac{2i}{\lambda'} \omega^{-\alpha} \dot{\varphi}^0, \\ \varphi^1 &= -\frac{2i}{\lambda'} \omega^{-\alpha} \dot{\varphi}^3. \end{split}$$

$$(2.2.22)$$

The general solution of Equations (2.2.22) has the form

$$\begin{split} \varphi^{0} &= \omega^{(1+\alpha-s/2)/2} \left(J_{\ell}(z)\chi^{0} + Y_{\ell}(z)\chi^{2} \right), \\ \varphi^{3} &= \omega^{(1+\alpha-s/2)/2} \left(J_{e}(z)\chi^{3} + Y_{e}(z)\chi^{1} \right), \\ \varphi^{2} &= -i \Big\{ \omega^{-(1+\alpha-s/2)/2} (1+\alpha-s/2)(\lambda')^{-1} \left(J_{\ell}(z)\chi^{0} + Y_{\ell}(z)\chi^{2} \right) + \\ &+ \omega^{(\alpha-s/2)/2} \left(\dot{J}_{\ell}(z)\chi^{0} + \dot{Y}_{\ell}(z)\chi^{1} \right) \Big\}, \\ \varphi^{1} &= -i \Big\{ \omega^{-(1+\alpha+s/2)/2} (1+\alpha-s/2)(\lambda')^{-1} \left(J_{\ell}(z)\chi^{3} + Y_{\ell}(z)\chi^{1} \right) + \\ &+ \omega^{(\alpha-s/2)/2} \left(\dot{J}_{\ell}(z)\chi^{3} + \dot{Y}_{\ell}(z)\chi^{1} \right) \Big\}, \end{split}$$

where $J_{\ell}(z)$, $Y_{\ell}(z) = \frac{1}{\sin \ell \pi} \left[J_{\ell}(z) \cos \ell \pi - J_{-\ell}(z) \right]$ are Bessel functions, and

$$z = \frac{\lambda'}{2\alpha + 1} \omega^{(2\alpha + 1)/2}; \qquad \lambda' = \lambda c^{1/2k}, \alpha = -\frac{s}{4k}; \qquad l = \frac{1 + \alpha - s/2}{2\alpha + 1}.$$
(2.2.24)

Using (2.2.23) and the identity

$$J_\ell \dot{Y}_\ell - \dot{J}_\ell Y_\ell = \frac{2}{\pi z}$$

one can find the constant c from (2.2.21) explicitly.

Solutions (2.2.5)-(2.2.8) and (2.2.14)-(2.2.16), as well as (2.2.17)-(2.2.19) and (2.2.23) can be made $\tilde{P}(1,3)$ -ungenerative by means of the formulae of GS from Table 2.1.2 and the formula of GS by the scale transformations

$$\psi_{II}(x) = e^{+k\theta}\psi_I(x'), \quad x'_{\mu} = e^{\theta}x_{\mu}, \quad \theta = \text{const.}$$
 (2.2.25)

For example, solution (2.2.5) results in the following $\tilde{P}(1,3)$ -ungenerative solution of Equation (2.2.1)

$$\psi(x) = \exp\left\{\frac{\theta}{2}\gamma a(\gamma d - \gamma b)(dy - by)\right\} \exp\left\{-\frac{1}{2}i\lambda(\overline{\chi}\chi)^{1/2k}(\gamma a) \times \left[2ay + \theta(dy - by)^2\right]\right\}\chi.$$
 (2.2.26)

As another example, the $\tilde{P}(1,3)$ -ungenerative solution of Equation (2.2.1) with k = 1/2 resulting from (2.2.7) is:

$$\psi(x) = \left[(by)^2 + (cy)^2 \right]^{-1/4} \exp\left\{ -\frac{1}{2}\gamma b\gamma c \arctan\frac{by}{cy} \right\} \times \\ \times \exp\left\{ -\frac{i\lambda}{1+\theta^2} (\overline{\chi}\chi)(\gamma c + \theta\gamma b) \times \\ \times \left[\ln\left((by)^2 + (cy)^2 \right) + 2\theta \arctan\frac{by}{cy} \right] \right\} \chi;$$

In formulae (2.2.26) and (2.2.27) $y_{\mu} = x_{\mu} + \delta_{\mu}; \delta_{\mu}, \theta, a_{\mu}, b_{\mu}, c_{\mu}, d_{\mu}$ are arbitrary constants satisfying relations (2.1.27).

We do not list $\tilde{P}(1,3)$ -ungenerative solutions of Equation (2.2.1) following from solutions (2.2.8), (2.2.14), (2.2.15),(2.2.17)-(2.2.19) because the astute reader will no doubt be able to do so alone. For the same reason, and for the sake of brevity, we do not present covariant counterparts of $\tilde{P}(1,3)$ nonequivalent ansatze for spinor fields, which can be constructed using the procedure of GS applied to the ansatze from Table 2.2.1.

2.3. Conformal symmetry and formula of generating solutions for fields of arbitrary spin. C(1,3)-ungenerative ansätze.

1. Conformal transformations are just a superposition of the inversion

$$x_{\mu} \to \widetilde{x}_{\mu} = \frac{x_{\mu}}{x^{\nu} x_{\nu}},\tag{2.3.1}$$

followed by translation and another inversion

$$\begin{aligned} x_{\mu} \to x'_{\mu} &= \frac{\widetilde{x}_{\mu} - c_{\mu}}{(\widetilde{x}_{\nu} - c_{\nu})(\widetilde{x}^{\nu} - c^{\nu})} = \frac{x_{\mu} - c_{\mu}x^{2}}{\sigma(c, x)}, \\ \sigma(c, x) &= 1 - 2cx + c^{2}x^{2}; \qquad (c^{2} \equiv c^{\nu}c_{\nu}, \ c_{\mu} = \text{const}). \end{aligned}$$
(2.3.2)

Transformations of inversion had been used in potential theory long ago to construct solutions of the Laplace equation (the so-called Kelvin transformation discovered in 1847 (see [205])). In 1909, Bateman and Cunningham [25, 43] had discovered that Maxwell's equations were conformally invariant. Later, it

became clear that many important equations of mathematical and theoretical physics possessed such a symmetry.

The extended Poincare group supplemented by conformal transformations form the fifteen-parameter conformal group C(1,3). The general form of the basis elements of the corresponding Lie algebra AC(1,3) (up to nilpotent matrices which can be added to D and K_{μ} ; see [145]) in the case of linear representation are as follows:

$$P_{0} = i\partial_{0}, \quad P_{a} = -i\partial_{a}, \quad (a = 1, 2, 3)$$

$$J_{\mu\nu} = x_{\mu}P_{\nu} - x_{\nu}P_{\mu} + S_{\mu\nu}$$

$$D = x^{\nu}P_{\nu} + ik,$$

$$K_{\mu} = 2x_{\mu}D - x^{2}P_{\mu} + 2S_{\mu\nu}x^{\nu},$$
(2.3.3)

where the constant k is the so-called conformal degree; $S_{\mu\nu} = -S_{\nu\mu}$ are matrices that provide a finite-dimensional representation of AO(1,3) (on representations of AO(1,3) see [111]).

The operators (2.3.3) satisfy the following commutation relations of AC(1,3):

$$[P_{\mu}, P_{\nu}] = 0, [P_{\sigma}, J_{\mu\nu}] = i(g_{\sigma\mu}P_{\nu} - g_{\sigma\nu}P_{\mu}),$$

$$[J_{\mu\nu}, J_{\rho\sigma}] = i(g_{\nu\rho}J_{\mu\sigma} + g_{\mu\sigma}J_{\nu\rho} - g_{\nu\sigma}J_{\mu\rho} - g_{\mu\rho}J_{\nu\sigma}),$$

$$[P_{\mu}, D] = iP_{\mu}, \qquad [J_{\mu\nu}, D] = 0,$$

$$[K_{\mu}, K_{\nu}] = 0, \qquad [K_{\sigma}, J_{\mu\nu}] = i(g_{\sigma\mu}K_{\nu} - g_{\sigma\nu}K_{\mu}),$$

$$[K_{\mu}, D] = -iK_{\mu},$$

$$[K_{\mu}, P_{\nu}] = 2i(g_{\mu\nu}D - J_{\mu\nu}).$$

(2.3.4)

It will be noted that the algebra AC(1,3) is locally isomorphic to AO(2,4) (see, for example, [21]). The isomorphism is achieved by means of introducing additional "Lorentz" generators

$$J_{\mu 4} = \frac{1}{2}(P_{\mu} - K_{\mu}), \quad K_{\mu 5} = \frac{1}{2}(P_{\mu} + K_{\mu}), \quad J_{45} = -D,$$

which satisfy, together with $J_{\mu\nu}$, the commutation relations of AO(2,4)

$$[J_{AB}, J_{CD}] = i(g_{BC}J_{AD} + g_{AD}J_{BC} - g_{AC}J_{BD} - g_{BD}J_{AC}),$$

where $A, B, C, D = \overline{0,5}$, $g_{AB} = \{1, -1, -1, -1, -1, 1\}$. Besides that one can choose from AC(1,3) two Poincare algebras: the standard one, AP₁(1,3) \equiv AP(1,3) = $\{P_{\mu}, J_{\mu\nu}\}$ and AP₂(1,3) = $\{K_{\mu}, J_{\mu\nu}\}$. Using subalgebraic classification of the Poincare algebra [163] one can construct, by analogy with Sections 2.1 and 2.2, new sets of AP₂(1,3)-inequivalent ansatze.

It can be seen from (2.3.3) that the algebra AC(1,3) is uniquely determined by the conformal degree k and the representation of AO(1,3), which is provided by the matrices $S_{\mu\nu}$. We point out these characteristics for some PDEs in the following table. As will become clear from what follows, the conformal degree may be considered a fundamental characteristic of a field, like spin.

Table 2.3.1. Conformally invariant PDEs and characteristics of their AC(1,3)(2.3.3)

N	Equation	Conformal degree	Matrix representaions
1	Wave $\Box u + \lambda u^3 = 0$	1	D(0,0)
2	Polywave $\square^{s}u + \lambda u^{\frac{2+s}{2-s}} = 0$	2-s	D(0,0)
3	Eikonal $(\partial_{\nu} u)(\partial^{\nu} u) = 0$	0	D(0,0)
4	Continuity $\partial_{\mu}j^{\mu} = 0$	3	D(1/2, 1/2)
5	Weyl $i\sigma_{\mu}\partial^{\mu}\varphi = 0$	3/2	D(0, 1/2), D(1/2, 0)
6	Dirac-Gursey	3/2	$D(0, 1/2) \oplus D(1/2, 0)$
	$\left[i\gamma\partial + \lambda(\overline{\psi}\psi)^{1/3}\right]\psi = 0$		
7	Generalized Dirac-Gursey	k	$D(0, 1/2) \oplus D(1/2, 0)$
	$\left[i\gamma \mathcal{D} + \lambda (\overline{\psi}\psi)^{1/2k}\right]\psi = 0,$		
	$\mathcal{D}_{\mu} = \partial_{\mu} + \frac{1}{2k} \left(\frac{3}{2} - k \right) \partial_{\mu} \ln \left(\overline{\psi} \psi \right)$		
8	Maxwell electromagnetic field	2	$D(1,0)\oplus D(0,1)$
	$\partial_{\mu}F^{\mu u} = 0, \partial_{\mu}\widetilde{F}^{\mu u},$		
	$(F^{0a} = E_a, F^{ab} = \epsilon_{abc} H_c,$		
	$\widetilde{F}^{\mu\nu} = \frac{1}{2} \epsilon_{\mu\nu\rho\sigma} F^{\rho\sigma} $		
9	Generalized Maxwell equation	k	$D(1,0)\oplus D(0,1)$
	for electromagnetic field		
	$\mathcal{D}_{\mu} F^{\mu\nu} = 0, \mathcal{D}_{\mu} F^{\mu\nu} = 0$		
	$\mathcal{D}_{\mu} = \partial_{\mu} + \partial_{\mu} \ln \left(\mathcal{T}_{1}^{\alpha} \mathcal{T}_{2}^{\beta} \right),$		
	$\mathcal{T}_1 = -\frac{1}{2} F_{\mu\nu} F^{\mu\nu} = \vec{E}^2 - \vec{H}^2,$		
	$T_2 = \frac{1}{4} F_{\mu\nu} F^{\mu\nu} = \vec{E} \cdot \vec{H}, \alpha + \beta = \frac{2-k}{2k},$		
10	Maxwell vector potential $2k$	1	D(1/2, 1/2)
	$\Box A_{\mu} - \partial_{\mu} \partial^{\nu} A_{\nu} = \lambda A_{\mu} A^{\nu} A_{\nu}$		
		1 1	$\begin{bmatrix} 3 \\ 0 \end{bmatrix} D(1/2, 1/2)$
11	rang-mills		$\bigcup_{i=1}^{D(1/2, 1/2)}$
	$\partial^{\nu} G^{a}_{\mu\nu} = e\epsilon_{abc} G^{a}_{\mu\nu} Y^{c\nu}$		
	$G^a_{\mu\nu} = \partial_\mu Y^a_\nu - \partial_\nu Y^a_\mu + e\epsilon_{abc} Y^b_\mu Y^c_\nu$		

Remark 2.3.1. Eikonal (N3), Weyl (N5), and Maxwell's (N8) equations do not admit generalization by nonlinear addends constructed from the fields (without derivatives).

Remark 2.3.2. Of course, the equations listed in Table 2.3.1 do not exhaust

all conformally invariant equations. There are many PDEs which are invariant under nonlinear representations of conformal algebra. For example, the relativistic Hamilton-Jacobi equation (1.2.1) is invariant under AC(1,4), with basis elements (1.2.2) containing nonlinear addends on u; see also Remark 1.3.1. (page 13). Another example of representation AC(1,3) different from (2.3.3) is given at the end of Section 2.8.

As distinct from transformations of the Poincare group (see Table 2.1.2) the special conformal transformations (2.3.2) are nonlinear. As will be shown below, this circumstance leads to highly nontrivial formulae of generating solutions [188].

To find an explicit form of transformations generated by the operator

$$\mathcal{K} = 2cxx\partial - x^2c\partial - 2(iS_{\mu\nu}x^{\nu} - kx_{\mu})c^{\mu}, \qquad (2.3.5)$$

which is a linear combination of operators K_{μ} from (2.3.3), one has to solve the following Cauchy problem (the system of Lie equations)

$$\frac{\partial x'_{\mu}}{\partial c^{\nu}} = 2x'_{\mu}x'_{\nu} - x'_{\sigma}x'^{5}g_{\mu\nu}, \qquad x'_{\mu} \Big|_{c^{\nu}=0} = x_{\mu}; \qquad (2.3.6)$$

$$\frac{\partial \psi'(x')}{\partial c^{\nu}} = 2(iS_{\mu\nu}x'_{\nu} - kx'_{\mu})\psi'(x'), \qquad \psi'(x') \Big|_{c^{\nu}=0} = \psi(x), \qquad (2.3.7)$$

or calculate the operator expressions (see Paragraph 5.3)

$$x'_{\mu} = \exp\left\{2cx\,x\partial - x^2c\partial\right\} x_{\mu} \exp\left\{-(2cx\,x\partial - x^2c\partial)\right\} \psi'(x') = \exp\left\{2cx\,x\partial - x^2c\partial\right\} \exp\left\{-\mathcal{K}\right\}\psi(x)$$
(2.3.8)

which are the formal solutions of the Lie equations (2.3.6) and (2.3.7). It is easy to confirm that formulae (2.3.2) give the solution of Equations (2.3.6). The solution of Equations (2.3.7) will be sought in the form

$$\psi'(x') = R(x,c)\psi(x), \qquad (2.3.9)$$

where R(x,c) is a nonsingular matrix, and R(x,0) = 1. Substituting (2.3.9) into (2.3.7) and using (2.3.2), we get

$$\frac{\partial R}{\partial c^{\mu}} = 2(iS_{\mu\nu}x'^{\nu} - kx'_{\mu})R =$$

$$= \frac{2}{\sigma(c,x)} \left[(iS_{\mu\nu}(x^{\nu} - c^{\nu}x^{2}) - k(x_{\mu} - c_{\mu}x^{2}) \right]R.$$
(2.3.10)

Since

$$\frac{\partial\sigma}{\partial c^{\mu}} = -2(x_{\mu} - c_{\mu}x^2),$$

(2.3.10) can be rewritten as follows:

$$\frac{\partial R}{\partial c^{\mu}} - \frac{k}{\sigma} \frac{\partial \sigma}{\partial c^{\mu}} R = 2i S_{\mu\nu} \frac{x^{\nu} - c^{\nu} x^2}{\sigma} R.$$
(2.3.11)

It can be seen from (2.3.11) that it is convenient to represent the matrix R(x,c) in the form

$$R(x,c) = \sigma^{k}(x,c)T(x,c),$$
(2.3.12)

where T(x,c) is a matrix. Substituting (2.3.12) into (2.3.11) gives the following for the matrix T:

$$\frac{\partial T}{\partial c^{\mu}} = 2iS_{\mu\nu}\frac{x^{\nu} - c^{\nu}x^2}{\sigma}T.$$
(2.3.13)

Multiplying (2.3.13) by c^{μ} and by x^{μ} we get, respectively

$$c^{\mu}\frac{\partial T}{\partial c^{\mu}} = 2i\sigma^{-1}S_{\mu\nu}c^{\mu}x^{\nu}T, \qquad (2.3.14)$$

$$x^{\mu}\frac{\partial T}{\partial c^{\mu}} = 2i\sigma^{-1}x^{2}S_{\mu\nu}c^{\mu}x^{\nu}T.$$
 (2.3.15)

One can seek solutions of these equations in the form

$$T = \exp\left\{2iS_{\mu\nu}c^{\mu}x^{\nu}\tau(x,c)\right\},$$
(2.3.16)

where $\tau(x,c)$ is a scalar differentiable function. Substituting (2.3.16) into (2.3.14), (2.3.15) gives

$$c^{\mu}\frac{\partial T}{\partial c^{\mu}} = 2iS_{\mu\nu}c^{\mu}x^{\nu}\left(\tau + c^{\alpha}\frac{\partial\tau}{\partial c^{\alpha}}\right)T = \frac{2i}{\sigma}S_{\mu\nu}c^{\mu}x^{\nu}T,$$
$$x^{\mu}\frac{\partial T}{\partial c^{\mu}} = 2iS_{\mu\nu}c^{\mu}x^{\nu}\left(x^{\alpha}\frac{\partial\tau}{\partial c^{\alpha}}\right)T = \frac{2i}{\sigma}x^{2}S_{\mu\nu}c^{\mu}x^{\nu}T,$$

(here we used formulae (5.3.18), (5.3.20)), whence follows that the function $\tau(x,c)$ should satisfy the equations

$$\tau + c^{\alpha} \frac{\partial \tau}{\partial c^{\alpha}} = \frac{1}{\sigma},$$

$$x^{\alpha} \frac{\partial \tau}{\partial c^{\alpha}} = \frac{x^{2}}{\sigma}$$

(2.3.17)

Suppose that the function $\tau(x,c)$ depends, like $\sigma(x,c)$, only on the generalized variables

$$w_1 = cx, \qquad w_2 = c^2 x^2.$$
 (2.3.18)

In this case Equations (2.3.17) take the form

$$\tau + w_1 \frac{\partial \tau}{\partial w_1} + w_2 \frac{\partial \tau}{\partial w_2} = (1 - 2w_1 + w_2)^{-1},$$
$$\frac{\partial \tau}{\partial w_1} + 2w_1 \frac{\partial \tau}{\partial w_2} = (1 - 2w_1 + w_2)^{-1}.$$

The general solution of these equations is given by the formula

$$\tau = \frac{1}{2} \left(w_1^2 - w_2 \right)^{-1/2} \ln \left(\frac{1 - w_1 + \sqrt{w_1^2 - w_2}}{1 - w_1 - \sqrt{w_1^2 - w_2}} \right),$$

and, once again using (2.3.18), we find

$$\tau(x,c) = \frac{1}{2} \left[(cx)^2 - c^2 x^2 \right]^{-1/2} \ln \left(\frac{1 - w_1 + \sqrt{w_1^2 - w_2}}{1 - w_1 - \sqrt{w_1^2 - w_2}} \right), \quad (2.3.19)$$

Thereby we obtain the formula of final conformal transformations for fields of arbitrary spin (see [82], [173], [188]):

$$\psi'(x') = R(x,c)\psi(x) = \sigma^k \exp\left\{2iS_{\mu\nu}c^{\mu}x^{\nu}\tau\right\}\psi(x), \qquad (2.3.20)$$

where σ and τ are given in (2.3.2) and (2.3.19).

Now it's easy to write the formula of generating solutions for field equations invariant under conformal transformations (2.3.2) and (2.3.20). According to (17) we find (see [188])

$$\psi_{II}(x) = \sigma^{-k} \exp\left\{-2iS_{\mu\nu}c^{\mu}x^{\nu}\tau\right\}\psi_{I}(x'), \qquad (2.3.21)$$

where σ, x' and τ are determined in (2.3.2) and (2.3.19).

Let us apply formulae (2.3.20) and (2.3.21) to a spinor Dirac field. In this case we have

$$S_{\mu\nu} = \frac{i}{4} (\gamma_{\mu} \gamma_{\nu} - \gamma_{\nu} \gamma_{\mu}), \qquad k = \frac{3}{2},$$
 (2.3.22)

where γ_{μ} are Dirac matrices (2.1.2). It is easy to calculate

$$2iS_{\mu\nu}c^{\mu}x^{\nu} = cx - \gamma c\gamma x, \qquad (2.3.23)$$

$$(2iS_{\mu\nu}c^{\mu}x^{\nu})^{2} = (cx)^{2} - c^{2}x^{2} \stackrel{\text{def}}{=} \theta^{2},$$

 $(2iS_{\mu\nu}c^{\mu}x^{\nu})^3 = \theta^2(cx - \gamma c\gamma x),$

.....

Hence

 $\exp\left\{2iS_{\mu\nu}c^{\mu}x^{\nu}\tau\right\} \equiv \exp\left\{(cx-\gamma c\gamma x)\tau\right\} =$

$$= \operatorname{ch} \tau \theta + (cx - \gamma c \gamma x) \frac{\operatorname{sh} \tau \theta}{\theta} \qquad (2.3.24)$$

From (2.3.19), (2.3.24) we find

$$\operatorname{sh} \tau \theta = \frac{\theta}{\sqrt{\sigma}}, \qquad \operatorname{ch} \tau \theta = \frac{1 - cx}{\sqrt{\sigma}}.$$
 (2.3.25)

Thus, finally, we obtain the formula of conformal transformations for the Dirac spinor field

$$\psi'(x') = \sigma^{3/2} \exp\left\{ (cx - \gamma c \gamma x)\tau \right\} \psi(x) = \sigma(1 - \gamma c \gamma x)\psi(x).$$
(2.3.26)

Using (2.3.23)-(2.3.25) one can easily obtain from (2.3.21) the corresponding formula of generating solutions

$$\psi_{II}(x) = \frac{1 - \gamma x \gamma c}{\sigma^2} \psi_I(x'), \qquad (2.3.27)$$

where x' and σ are determined in (2.3.2).

Remark 2.3.3. In the case of a spinor field with arbitrary conformal degree k formulae (2.3.20) and (2.3.21), as follows from (2.3.22)-(2.3.25), have the form

$$\psi'(x') = \sigma^{k-1/2} (1 - \gamma c \gamma x) \psi(x), \qquad (2.3.28)$$

$$\psi_{II}(x) = \sigma^{-(k+1/2)} (1 - \gamma x \gamma c) \psi_{I}(x'), \qquad (2.3.29)$$

where x' and σ are given in (2.3.2).

Remark 2.3.4. In the case of n spacial variables, the conformally invariant Dirac equation, with nonlinearity of the Gursey type, has the form (see [93, 95])

$$\left[i\gamma\partial - \lambda(\overline{\psi}\psi)^{1/n}\right]\psi = 0 \tag{2.3.30}$$

(here γ -matrices have appropriate structure, see [2*], §8.3). The conformal degree of its AC(1,n) is equal to n/2, the conformal transformations and corresponding formula of generating solutions having the form

$$\psi'(x') = \sigma^{(n-1)/2} (1 - \gamma c \gamma x) \psi(x), \qquad (2.3.31)$$

84

$$\psi_{II}(x) = \sigma^{-(n+1)/2} (1 - \gamma x \gamma c) \psi_{I}(x'), \qquad (2.3.32)$$

where x' and σ are given in (2.3.2).

Now we write the explicit form of formulae (2.3.20) and (2.3.21) for scalar, vector, and tensor fields. For scalar fields u with k = 1 we have

$$u'(x') = \sigma u(x),$$
 (2.3.33)

$$u_{II} = \sigma^{-1} u_I(x'). \tag{2.3.34}$$

For scalar fields u with k = 2 - s we have

$$u'(x') = \sigma^{2-s}u(x), \qquad (2.3.35)$$

$$u_{II}(x) = \sigma^{s-2} u_I(x'). \tag{2.3.36}$$

For vector fields A_{μ} with k = 1 we have [97]

$$A'_{\mu}(x') = \left[g_{\mu\nu}\sigma + 2(x_{\mu}c_{\nu} - x_{\nu}c_{\mu} + 2cxc_{\mu}x_{\nu} - x^{2}c_{\mu}c_{\nu} - c^{2}x_{\mu}x_{\nu})\right]A^{\nu}(x),$$

$$(2.3.37)$$

$$A_{\mu II}(x) = \left[\frac{g_{\mu\nu}}{\sigma} + \frac{2}{\sigma^2}(c_{\mu}x_{\nu} - c_{\nu}x_{\mu} + 2cxx_{\mu}c_{\nu} - x^2c_{\mu}c_{\nu} - c^2x_{\mu}x_{\nu})\right]A_I^{\nu}(x').$$
(2.3.38)

For vector fields A_{μ} with arbitrary k we have

$$A'_{\mu}(x') = \left[g_{\mu\nu}\sigma^{k} + 2\sigma^{k-1}(x_{\mu}c_{\nu} - x_{\nu}c_{\mu} + 2cxc_{\mu}x_{\nu} - x^{2}c_{\mu}c_{\nu} - c^{2}x_{\mu}x_{\nu})\right]A^{\nu}(x),$$

$$(2.3.39)$$

$$A_{\mu\Pi}(x) = \left[g_{\mu\nu}\sigma^{-k} + 2\sigma^{-(k+1)}(c_{\mu}x_{\nu} - c_{\nu}x_{\mu} + 2cxx_{\mu}c_{\nu} - x^{2}c_{\mu}c_{\nu} - c^{2}x_{\mu}x_{\nu})\right]A_{I}^{\nu}(x').$$
(2.3.40)

For tensor fields $F_{\mu\nu}$ with k = 2 we have

$$F'_{\mu\nu}(x') = \sigma^2 F_{\mu\nu} + 2\sigma \Big\{ \Big[(2cx - 1)(c_{\mu}F_{\beta\nu} - c_{\nu}F_{\beta\mu}) - (2.3.41) \\ -c^2 (x_{\mu}F_{\beta\nu} - x_{\nu}F_{\beta\mu}) \Big] x^{\beta} + \Big[x_{\mu}F_{\alpha\nu} - x_{\nu}F_{\alpha\mu} - (2.3.41) \\ -x^2 (c_{\mu}F_{\alpha\nu} - c_{\nu}F_{\alpha\mu}] c^{\alpha} + 2(c_{\mu}x_{\nu} - c_{\nu}x_{\mu})F_{\alpha\beta}c^{\alpha}x^{\beta} \Big\}$$

85

$$F_{\mu\nu}^{II}(x) = \sigma^{-2} F_{\mu\nu}^{I}(x') + 2\sigma^{-3} \Big\{ \Big[(2cx - 1) \big(x_{\mu} F_{\beta\nu}^{I}(x') - - x_{\nu} F_{\beta\mu}^{I}(x') \big) - x^{2} (c_{\mu} F_{\beta\nu}^{I}(x') - c_{\nu} F_{\beta\mu}^{I}(x')] c^{\beta} + + \big[c_{\mu} F_{\alpha\nu}^{I}(x') - c_{\nu} F_{\alpha\mu}^{I}(x') - c^{2} (x_{\mu} F_{\alpha\nu}^{I}(x') - - x_{\nu} F_{\alpha\mu}^{I}(x')) \Big] x^{\alpha} + 2 (x_{\mu} c_{\nu} - x_{\nu} c_{\mu}) x^{\alpha} c^{\beta} F_{\alpha\beta}^{I}(x') \Big\}.$$
(2.3.42)

From expressions (2.3.41) and (2.3.42) one can easy derive the explicit form of the conformal transformations and corresponding formula of generating solutions for electric \vec{E} and magnetic \vec{H} fields, satisfying Maxwell equations, by virtue of equalities

$$E_a = F^{0a}, \qquad H_a = \frac{1}{2} \epsilon_{abc} F^{bc}.$$
 (2.3.43)

Let us write the conformal transformations and formula of generating solutions for the antisymmetric tensor $F_{\mu\nu}$ with arbitrary conformal degree k

$$\begin{aligned} F'_{\mu\nu}(x') &= \sigma^{k} F_{\mu\nu}(x') + 2\sigma^{k-1} \Big\{ \Big[(2cx - 1) \big(c_{\mu} F_{\beta\nu} - \\ &- c_{\nu} F_{\beta\mu} \big) \big) - c^{2} \big(x_{\mu} F_{\beta\nu} - x_{\nu} F_{\beta\mu} \big) \Big] x^{\beta} + \\ &+ \big[x_{\mu} F_{\alpha\nu} - x_{\nu} F_{\alpha\mu} - x^{2} \big(c_{\mu} F_{\alpha\nu} - c_{\nu} F_{\alpha\mu} \big) \Big] c^{\alpha} + \\ &+ 2 \big(c_{\mu} x_{\nu} - c_{\nu} x_{\mu} \big) F_{\alpha\beta} c^{\alpha} x^{\beta} \Big\}, \end{aligned}$$

$$\begin{aligned} F^{II}_{\mu\nu}(x) &= \sigma^{-k} F^{I}_{\mu\nu}(x') + 2\sigma^{-(k+1)} \Big\{ \big[(2cx - 1) \times \\ & (2.3.45) \\ &\times \big(x_{\mu} F^{I}_{\beta\nu}(x') - x_{\nu} F^{I}_{\beta\mu}(x') \big) - x^{2} \big(c_{\mu} F^{I}_{\beta\nu}(x') - c_{\nu} F^{I}_{\beta\mu}(x') \big) \big] c^{\beta} + \\ &+ \big[c_{\mu} F^{I}_{\alpha\nu}(x') - c_{\nu} F^{I}_{\alpha\mu}(x') - c^{2} \big(x_{\mu} F^{I}_{\alpha\nu}(x') - x_{\nu} F^{I}_{\alpha\mu}(x') \big) \big] x^{\alpha} + \\ &+ 2 \big(x_{\mu} c_{\nu} - x_{\nu} c_{\mu} \big) x^{\alpha} c^{\beta} F^{I}_{\alpha\beta}(x') \Big\}. \end{aligned}$$

In formulae (2.3.33)-(2.3.45) x' and σ are given in (2.3.2).

Remark 2.3.5. As we have already noted, the spinor field plays a crucial role in unified field theory because it allows us to construct fields with any spin. Here we would like to point out that to represent a given field via a spinor field ψ one must construct an appropriate combination from ψ and take ψ with the proper conformal degree. For example, scalar (u), vector (A_{μ}) , and tensor $(F_{\mu\nu} = -F_{\nu\mu})$ fields with conformal degree \tilde{k} can be constructed as

$$u = \overline{\psi}\psi, \qquad A_{\mu} = \overline{\psi}\gamma_{\mu}\psi, \qquad F_{\mu\nu} = \frac{i}{2}\overline{\psi}(\gamma_{\mu}\gamma_{\nu} - \gamma_{\nu}\gamma_{\mu})\psi \quad (2.3.46)$$

from the spinor field ψ with conformal degree $\tilde{k}/2$. Spinor field ψ with arbitrary conformal degree k can be constructed, in turn, by means of the Dirac spinor field $\psi_{\mathcal{D}}$ with conformal degree 3/2:

$$\psi = \left(\overline{\psi}_{\mathcal{D}}\psi_{\mathcal{D}}\right)^{-\left(\frac{3}{2}-k\right)/3}\psi_{\mathcal{D}}, \qquad \overline{\psi}_{\mathcal{D}}\psi_{\mathcal{D}} \neq 0.$$
(2.3.47)

Using (2.3.46), (2.3.28), and (2.3.29) and what has been said above, we find another way of obtaining the formulae (2.3.33)-(2.3.45).

Now we shall construct conformally invariant ansatze. First we do this for a spinor field. Starting from the general expression (2.1.7), we determine the matrix A(x) and the invariant variables ω as solutions of the equations

$$(2cx x\partial)\omega(x) = 0,$$

$$[2cx x\partial - x^2 c\partial + \gamma c\gamma x + (2k-1)cx]A(x) = 0$$
(2.3.48)

(here we used (2.3.5) and (2.3.23)).

One can easy verify that the functions

$$\omega = \frac{\beta x}{x^{\nu} x_{\nu}}, \qquad A(x) = \gamma x / (x^{\nu} x_{\nu})^{(2k+1)/2}$$
(2.3.49)

 $(\beta_{\nu} \text{ are arbitrary constants}, \beta c = 0)$ satisfy Equations (2.3.48). Hence a conformally invariant ansatz for a spinor field with conformal degree k has the form

$$\psi(x) = \left[\gamma x \middle/ (x^{\nu} x_{\nu})^{(2k+1)/2}\right] \varphi(\omega), \qquad \omega = \frac{\beta x}{x^{\nu} x_{\nu}} \qquad (2.3.50)$$

 $(\beta_{\nu} \text{ are arbitrary constants}, \beta c = 0)$

In the particular case of the Dirac spinor field, when k = 3/2, from (2.3.50) we obtain the ansatz (see [93], [95])

$$\psi(x) = \frac{\gamma x}{\left(x^{\nu} x_{\nu}\right)^{2}} \varphi(\omega), \qquad \omega = \frac{\beta x}{x^{\nu} x_{\nu}}$$
(2.3.51)

Using (2.3.50) and (2.3.46) one can easily construct conformally invariant ansatze for scalar, vector, and tensor fields in the case of arbitrary conformal degree k. Without going into the details we simply list them:

$$u(x) = (x_{\nu}x^{\nu})^{-k}v(\omega); \qquad (2.3.52)$$

$$A_{\mu}(x) = \frac{B_{\mu}(\omega)}{(x^{\nu}x_{\nu})^{k}} - 2x_{\mu}\frac{x^{\nu}B_{\nu}(\omega)}{(x_{\alpha}x^{\alpha})^{k+1}};$$
(2.3.53)

$$F_{\mu\nu}(x) = \frac{f_{\mu\nu}(\omega)}{(x^{\alpha}x_{\alpha})^{k}} - 2\frac{(x_{\mu}f_{\beta\nu}(\omega) - x_{\nu}f_{\beta\mu}(\omega))x^{\beta}}{(x^{\alpha}x_{\alpha})^{k+1}}.$$
 (2.3.54)

In formulae (2.3.52)–(2.3.54), $\omega = \beta x/(x^{\nu} x_{\nu})$.

Having applied transformations of translations to ansatzes (2.3.50)-(2.3.54), we obtain the C(1,3)-ungenerative ansatze. For example, from (2.3.51) we obtain the following C(1,3)-ungenerative ansatz for the Dirac spinor field:

$$\psi(x) = \frac{\gamma y}{(y_{\nu}y^{\nu})^2}\varphi(\Omega), \qquad \Omega = \frac{\beta y}{y_{\nu}y^{\nu}} + \mathfrak{A}$$
(2.3.55)

where $y_{\nu} = x_{\nu} + \delta_{\nu}$; $\delta_{\mu\nu}$, β_{ν} , æ are arbitrary constants.

Ungenerativity of ansatz (2.3.55) under transformations from the group $\widetilde{P}(1,3)$ (the corresponding formulae are given in Table 2.1.2 and Equation (2.2.25)) is quite evident. Furthermore, one can confirm in a straight-forward way that applying the formula of GS (2.3.27) to (2.3.55) is the same as changing the parameters $\beta_{\nu}, \delta_{\nu}, \infty$, and the function φ in (2.3.55) in such a manner that

$$\varphi \to \frac{1 - \gamma c \gamma \delta}{\sigma^2(\delta, c)} \varphi, \quad \sigma(\delta, c) \equiv 1 - 2\delta c + \delta^2 c^2,$$

$$\delta_\mu \to \delta'_\mu = \frac{\delta_\mu - c_\mu \delta^2}{\sigma(\delta, c)}, \qquad (2.3.56)$$

$$\begin{split} \beta_{\mu} &\to \beta_{\mu}' = \frac{\beta_{\mu}}{\sigma(\delta,c)} + \frac{2}{\sigma^{2}(\delta,c)} (\delta_{\mu}c_{\nu} - \delta_{\nu}c_{\mu} + 2\delta cc_{\mu}\delta_{\nu} - \delta^{2}c_{\mu}c_{\nu} - c^{2}\delta_{\mu}\delta_{\nu})\beta^{\nu}, \\ \mathfrak{w} &\to \mathfrak{w}' = \frac{\beta\delta c^{2} - \beta c}{\sigma(\delta,c)}. \end{split}$$

This means that the family of solutions like (2.3.55) is C(1,3)-ungenerative. Ungenerativity of ansatz constructed from (2.3.52)-(2.3.54), analogously to ansatz (2.3.55), is proved in much the same way.

It will be noted that any Lorentz-invariant solution of a conformally invariant equation can be made C(1,3)-ungenerative if formulae of GS by scale, conformal, and translational transformations are applied successively.

It will also be noted that the conformally invariant ansatz (2.3.51) (as well as ansatze (2.3.52)-(2.3.54)) can be obtained from a plane-wave ansatz $\psi(x) = \varphi(\beta x)$ by applying to it formula of GS (2.3.27) under $c_0 = 1$, $c_1 = c_2 = c_3 = 0$ and then making the translation transformation $x_0 \to x_0 + 1$.

In conclusion, let us show how to construct from a given solution of a linear PDE an infinite sequence of solutions (see $[24^*]$). From the definition of invariance (see the Introduction) it follows that if operator Q is admitted by a linear system of PDEs, then

$$\psi_1 = a_1 Q \psi_0, \quad \psi_2 = a_2 Q \psi_1, \dots \psi_n = a_n Q \psi_{n-1} \dots$$
(2.3.57)

(where a_1, a_2, \ldots are some normalization factors) will be solutions of this system as soon as $\psi_0(x)$ is a solution. For systems of PDEs invariant under
AC(1,3) (2.3.3) the procedure (2.3.57) makes nontrivial sense only for operators K_{μ} (as a matter of convenience we shall use operator \mathcal{K} (2.3.5)). Below we construct, with the help of operator (2.3.5), sequences of solutions to the wave equation, the massless Dirac equation, and Maxwell's equations.

In the case of a scalar field with conformal degree k = 1, operator (2.3.5) takes the form

$$\mathcal{K} = 2cxx\partial - x^2\partial + 2cx. \tag{2.3.58}$$

Then, starting from $u_0 = 1$ and taking $a_n = 1/2n$ one obtains the sequence of solutions (2.3.57) of the wave equation

$$\Box u = 0 \tag{2.3.59}$$

in the form

$$u_{0} = 1$$

$$u_{1} = cx,$$

$$u_{2} = (cx)^{2} - \frac{1}{4}c^{2}x^{2},$$

$$u_{3} = (cx)^{3} - \frac{1}{2}c^{2}x^{2},$$

$$u_{4} = (cx)^{4} - \frac{3}{4}(cx)^{2}c^{2}x^{2} - \frac{1}{16}(c^{2}x^{2})^{2},$$

(2.3.60)

One can easily make sure that u_1, u_2, u_3, \ldots satisfy Equation (2.3.59). It will be noted that u_2, u_3, \ldots satisfy the following recurrence relations

$$u_n = \frac{1}{2n} \mathcal{K} u_{n-1} = (cx)u_{n-1} - \frac{1}{4}c^2 x^2 u_{n-2}, \quad n \ge 2.$$
 (2.3.61)

One can make solutions (2.3.60) C(1,3)-ungenerative, generating them by conformal and translation transformations.

For the massless Dirac equation

$$i\gamma\partial\psi = 0 \tag{2.3.62}$$

we have

$$\mathcal{K} = 2cxx\partial - x^2c\partial + 2cx + \gamma c\gamma x \tag{2.3.63}$$

and starting from $\psi_0 = \chi$, where χ is a constant spinor, we obtain

$$\psi_0 = \chi$$

 $\psi_1 = (cx - \frac{1}{4}\gamma x\gamma c)\chi,$

Chapter 2. Systems of Poincare-invariant Nonlinear PDEs

$$\psi_{2} = \left[(cx)^{2} - \frac{1}{6}c^{2}x^{2} - \frac{1}{3}(cx)\gamma x\gamma c \right] \chi, \qquad (2.3.64)$$

$$\psi_{3} = \left[(cx)^{3} - \frac{3}{8}c^{2}x^{2}(cx) + \left(\frac{1}{16}c^{2}x^{2} - \frac{3}{8}(cx)^{2} \right)\gamma x\gamma c \right] \chi,$$

It is easy to show that functions (2.3.64) satisfy the Dirac equation (2.3.62). Solutions (2.3.64) can also be made C(1,3)-ungenerative if they are generated by conformal and translation transformations.

For Maxwell's equations for vector-potential

$$\Box A_{\mu} - \partial_{\mu} \partial^{\nu} A_{\nu} = 0 \tag{2.3.65}$$

we analogously find the sequence of solutions $A_{\mu}^{(1)}, A_{\mu}^{(2)}, \ldots$ starting from a solution $A_{\mu}^{(0)}$:

$$A_{\mu}^{(n)} = a_n \left[\left(2cx(x\partial + 1) - x^2 c\partial \right) A_{\mu}^{(n-1)} + 2c_{\mu} \left(A_{\nu}^{(n-1)} x^{\nu} \right) - 2x_{\mu} \left(A_{\nu}^{(n-1)} c^{\nu} \right) \right]$$

$$n = 1, 2, \dots$$
(2.3.66)

where a_n are some normalization constants. In particular, if we start with $A^{(0)}_{\mu} = a_n$ (a_n are arbitrary constants), then (2.3.66) gives

$$A_{\mu}^{(1)} = a_{\mu}cx + c_{\mu}ax - x_{\mu}ac,$$

$$A_{\mu}^{(2)} = a_{\mu} \left[(cx)^{2} - \frac{1}{4}c^{2}x^{2} \right] + c_{\mu} \left(2(ax)(cx) - \frac{1}{2}acx^{2} \right) - -x_{\mu} \left((ac)(cx) + \frac{1}{2}c^{2}ax \right),$$

$$(2.3.67)$$

Solutions (2.3.67) have the property $F_{\mu\nu} \equiv \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu} = 0$. But if we start with $A_{\mu}^{(0)} = a_{\mu}bx$ $(a_{\mu}, b_{\nu}$ are arbitrary constants) then we obtain from (2.3.66) another sequence of solutions of Equations (2.3.65):

$$A_{\mu}^{(1)} = a_{\mu}(cxbx - \frac{1}{4}x^{2}bc) + \frac{1}{2}bx(c_{\mu}ax - x_{\mu}ac), \qquad (2.3.68)$$

and in this case $F_{\mu\nu} \neq 0$.

Lastly we write down formula (2.3.57) for the tensor of the electromagnetic field $F_{\mu\nu}$ (2.3.43). So, if $F^{(0)}_{\mu\nu}$ is a solution of Maxwell's equations

$$\partial_{\mu}F^{\mu\nu} = 0, \qquad \epsilon^{\mu\nu\rho\sigma}\partial_{\nu}F_{\rho\sigma} = 0 \qquad (2.3.69)$$

then

$$F_{\mu\nu}^{(n+1)} = (c_{\mu}F_{\alpha\nu}^{(n)} - c_{\nu}F_{\alpha\mu}^{(n)})x^{\alpha} + (x_{\nu}F_{\alpha\mu}^{(n)} - x_{\mu}F_{\alpha\nu}^{(n)})c^{\alpha} + 2cxF_{\mu\nu}^{(n)} + (cxx^{\alpha} - \frac{1}{2}x^{2}c^{\alpha})\partial_{\alpha}F_{\mu\nu}^{(n)}; \quad n = 0, 1, \dots$$
(2.3.70)

90

will be also solutions of Maxwell's Equations (2.3.69).

2.4. Conformally invariant nonlinear equations for spinor fields and their solutions

1. Let us consider the following Poincare invariant nonlinear PDE for a spinor field * [101], [102]):

$$\left\{\gamma^{\mu}\left[i\partial_{\mu}+(\overline{\psi}\gamma_{\mu}\psi)F_{1}+(\overline{\psi}\gamma_{5}\gamma_{\mu}\psi)F_{2}+(\overline{\psi}\gamma_{\mu}\psi)\gamma_{5}F_{3}+(\overline{\psi}\gamma_{5}\gamma_{\mu}\psi)\gamma_{5}F_{4}\right]+\left(\overline{\psi}\sigma_{\mu\nu}\psi)\sigma^{\mu\nu}F_{5}+(\overline{\psi}\sigma_{\mu\nu}\psi)\gamma_{5}\sigma^{\mu\nu}F_{6}+F_{7}+\gamma_{5}F_{8}\right\}\psi=0,\qquad(2.4.1)$$

where

$$\gamma_5 = i\gamma_0\gamma_1\gamma_2\gamma_3 = \begin{pmatrix} 0 & \sigma_0\\ \sigma_0 & 0 \end{pmatrix}; \qquad \sigma_{\mu\nu} = \frac{i}{2}(\gamma_\mu\gamma_\nu - \gamma_\nu\gamma_\mu); \quad (2.4.2)$$

 F_1, \ldots, F_8 are arbitrary scalar functions depending on $\overline{\psi}\psi$ and $\overline{\psi}i\gamma_5\psi$. The rest of the notations are given in (2.1.1)-(2.1.3).

From the set of Equations (2.4.1) we select those which are invariant under scale transformations

$$x_{\mu} \to x'_{\mu} = e^{\theta} x_{\mu}; \quad \psi(x) \to \psi'(x') = e^{-\theta k} \psi(x) \tag{2.4.3}$$

and under conformal transformations (2.3.2), (2.3.26).

Theorem 2.4.1. [101, 102]. Equation (2.4.1) is invariant under the scale transformations (2.4.3) iff

$$F_{i} = (\overline{\psi}\psi)^{(1-2k)/2k}\phi_{i}, \quad i = \overline{1,6}; F_{j} = (\overline{\psi}\psi)^{1/2k}\phi_{j}, \quad j = 7,8,$$
(2.4.4)

where ϕ_1, \ldots, ϕ_8 are arbitrary scalar functions depending on $(\overline{\psi}\psi)/(\overline{\psi}i\gamma_5\psi)$.

Proof. The necessary and sufficient condition of the invariance of equation (2.4.1) under the transformation (2.4.3) is the fulfilment of the relations

^{*} There is no nonlinear Poincare-invariant equation of such a type for a twocomponent spinor field φ (Weyl field) because it is impossible to add to the Weyl equation a nonlinear term constructed from φ and φ^+ and to conserve even Lorentz invariance. The same statement holds true for Maxwell's equations for electromagnetic fields in a vacuum.

Chapter 2. Systems of Poincare-invariant Nonlinear PDEs

$$\exp \left\{ \theta(1-2k) \right\} F_i \left(\exp\{-2\theta k\} \overline{\psi} \psi, \exp\{-2\theta k\} \overline{\psi} i \gamma_5 \psi \right) =$$

$$= F_i (\overline{\psi} \psi, \overline{\psi} i \gamma_5 \psi); \quad i = \overline{1,6};$$
(2.4.5)
$$(2.4.5)$$

$$\exp \left\{ \theta(1-k) \right\} F_j \left(\exp\{-2\theta k\} \overline{\psi} \psi, \exp\{-2\theta k\} \overline{\psi} i \gamma_5 \psi \right) =$$
$$= F_j (\overline{\psi} \psi, \overline{\psi} i \gamma_5 \psi); \quad j = 7, 8;$$

Differentiation of (2.4.5) with respect to θ followed by setting $\theta = 0$ gives

$$(1-2k)F_{i} = 2ku\frac{\partial F_{i}}{\partial u} + 2kv\frac{\partial F_{i}}{\partial v}, \quad i = \overline{1,6}$$

$$(1-k)F_{j} = 2ku\frac{\partial F_{j}}{\partial u} + 2kv\frac{\partial F_{i}}{\partial v}, \quad j = 7,8,$$

$$(2.4.6)$$

where $u = \overline{\psi}\psi$, $v = \overline{\psi}i\gamma_5\psi$.

After integrating (2.4.6) we get (2.4.4). One can directly confirm that Equation (2.4.1) with functions F(2.4.4) is invariant under the scale transformations (2.4.3). The theorem is proved.

Remark 2.4.1. Obviously, Equation (2.4.1) with functions

$$F_{i_{1}} = \left[(\overline{\psi} \gamma_{\mu} \psi) (\overline{\psi} \gamma^{\mu} \psi) \right]^{(1-2k)/4k} \phi_{i_{1}}, \qquad i_{1} = 1, 3;$$

$$F_{i_{2}} = \left[(\overline{\psi} \gamma_{5} \gamma_{\mu} \psi) (\overline{\psi} \gamma_{5} \gamma^{\mu} \psi) \right]^{(1-2k)/4k} \phi_{i_{2}}, \qquad i_{2} = 2, 4;$$

$$F_{i_{3}} = \left[(\overline{\psi} \sigma_{\mu\nu} \psi) (\overline{\psi} \sigma^{\mu\nu} \psi) \right]^{(1-2k)/4k} \phi_{i_{3}}, \qquad i_{3} = 5, 6;$$

$$F_{j} = \left(\overline{\psi} \psi \right)^{1/2k} \phi_{j}, \qquad j = 7, 8$$

$$(2.4.7)$$

where ϕ_1, \ldots, ϕ_8 are arbitrary scalar functions depending on $\frac{\overline{\psi}\psi}{\overline{\psi}i\gamma_5\psi}$, is also invariant under the scale transformations (2.4.3).

Remark 2.4.2. From four-component spinor fields ψ and $\overline{\psi}$ one can construct 16 real quadratic forms $\rho_j(j=\overline{1,16})$

$$\begin{array}{lll} u = \psi \psi & - \text{ scalar,} \\ v = \overline{\psi} i \gamma_5 \psi & - \text{ pseudoscalar,} \\ j_{\mu} = \overline{\psi} \gamma_{\mu} \psi & - \text{ vector,} \\ n_{\mu} = \overline{\psi} \gamma_5 \gamma_{\mu} \psi & - \text{ pseudovector,} \\ f_{\mu\nu} = \overline{\psi} \sigma_{\mu\nu} \psi & - \text{ antisymmetric tensor.} \end{array}$$
(2.4.8)

These 16 bispinor densities are not independent since the spinor wave function ψ is composed of four independent complex functions. Furthermore, as

the overall phase of the spinor has no effect on the bispinor densities ρ_j , we conclude that these 16 functions must satisfy a total of nine algebraic equations. These equations are known as identities of Fierz-Pauli and have the form (see [42])

$$\begin{aligned} j_{\mu}j^{\mu} &= -n_{\mu}n^{\mu} = u^{2} + v^{2}; \\ j_{\mu}n^{\mu} &= 0; \\ f_{\mu\nu} &= (u^{2} + v^{2})^{-1} \left[u\epsilon_{\mu\nu\rho\sigma}j^{\rho}n^{\sigma} - v(j_{\mu}n_{\nu} - j_{\nu}n_{\mu}) \right], \qquad u^{2} + v^{2} \neq 0. \end{aligned}$$
(2.4.9)

From this one can conclude that Equations (2.4.1) with functions (2.4.4) and (2.4.7) are in a sense equivalent.

Theorem 2.4.2. (See [101]). Equation (2.4.1) is invariant under the conformal group C(1,3) iff functions F have the form (2.4.4) and k = 3/2.

Proof. Since the conformal group contains the extended Poincare group we can use the previous theorem. Then one can confirm that conformal transformations (2.3.2), (2.3.26) leave equation (2.4.1) with functions F from (2.4.4) invariant iff k = 3/2. The theorem is proved.

As a consequence of Theorem 2.4.2 we get the Dirac-Gursey Equation (2.1.5)and the following conformally invariant generalization of the Dirac-Heisenberg Equation (2.1.4)

$$\left\{i\gamma\partial + \lambda \frac{(\overline{\psi}\gamma_{\nu}\psi)\gamma^{\nu}}{\left[(\overline{\psi}\gamma_{\mu}\psi)(\overline{\psi}\gamma^{\mu}\psi)\right]^{1/3}} + \varkappa \frac{(\overline{\psi}\gamma_{5}\gamma_{\mu}\psi)\gamma_{5}\gamma^{\mu}}{\left[(\overline{\psi}\gamma_{5}\gamma_{\nu}\psi)(\overline{\psi}\gamma_{5}\gamma^{\nu}\psi)\right]^{1/3}}\right\}\chi$$
(2.4.10)

 (λ, x) are arbitrary constants).

2. Now we describe some exact solutions of the nonlinear conformally invariant spinor wave Equations (2.1.5) and (2.4.10). For the first time exact solutions of the Dirac-Gursey Equation (2.1.5) were obtained by Kortel (see [132]) with the help of the Heisenberg ansatz (see (2.1.28) with $\omega = \sqrt{x_{\nu}x^{\nu}}$). Later these results, slightly generalized, were expounded in [5],[6], and [150]. We list the Kortel solutions as they are given in [150]:

$$\psi(x) = \left(\frac{4\alpha}{\lambda}\right)^{3/2} \frac{i\gamma x + \alpha}{\left(\omega^2 + \alpha^2\right)^2} \chi, \qquad (2.4.11)$$

$$\psi(x) = \frac{1}{4} \left(\frac{3}{\lambda}\right)^{3/2} \omega^{-5/2} (i\gamma x + \omega) \chi, \qquad (2.4.12)$$

where $\omega = \sqrt{x_{\nu}x^{\nu}}$, α is a constant, and $\overline{\chi}\chi = 1$. Note that solutions (2.4.11) and (2.4.12) are nonanalytic in the coupling constant λ . New solutions of

Equation (2.1.5) can be obtained from (2.1.26) under m = 0 and k = 1/3. Let us list some of them:

$$\psi(x) = \left[(ax)^2 + (bx)^2 \right]^{-1/4} \exp\left\{ -\frac{1}{2} (\gamma a) (\gamma b) \times \right.$$
(2.4.13)

$$\times \arctan \frac{ax}{bx} \Big\} \exp \Big\{ -\frac{3}{2} i \lambda(\overline{\chi}\chi)^{1/3} (\gamma b) \left[(ax)^2 + (bx)^2 \right]^{2/3} \Big\} \chi;$$

$$\psi(x) = \exp\left\{-\frac{1}{4}(\gamma a + \gamma d)\gamma b(ax + dx)\right\} \times$$

$$\times \exp\left\{-i\lambda(\overline{\chi}\chi)^{1/3}\gamma b\left[bx + \frac{1}{4}(ax + dx)^2\right]\right\}\chi,$$
(2.4.14)

where $a_{\nu}, b_{\nu}, d_{\nu}$ are arbitrary constants satisfying (2.1.27).

One can apply the formulae of generating solutions (2.3.27) to obtain further solutions from (2.4.11)-(2.4.14). Having made in these solutions transformation of translations one obtains C(1,3)-ungenerative families of solutions. For example, solution (2.4.11) yields the following multiparameter C(1,3)ungenerative family of solutions of Equation (2.1.5):

$$\psi(x) = \left(\frac{4\alpha}{\lambda}\right)^{3/2} \frac{i\gamma y + \alpha(1 - \gamma y\gamma c)}{\left[y^2 \left(1 + \alpha^2 c^2\right) + \alpha^2(1 - 2cy)\right]^2} \chi, \qquad (2.4.15)$$

where $y_{\nu} = x_{\nu} + \delta_{\nu}$; c_{ν}, δ_{ν} are arbitrary constants; $\overline{\chi}\chi = 1$.

The substitution of conformally invariant ansatz (2.3.51) into (2.1.5) yields the following ODE

$$\frac{d\varphi}{d\omega} = -i\lambda(\beta_{\nu}\beta^{\nu})^{-1}(\overline{\varphi}\varphi)^{1/3}(\gamma\beta)\varphi$$

the general solution of which has the form

$$\varphi(\omega) = \exp\left\{-i\lambda(\beta_{\nu}\beta^{\nu})^{-1}(\overline{\chi}\chi)^{1/3}(\gamma\beta)\omega\right\}\chi$$

This gives one more solution of Equation (2.1.5) (see [93, 95])

$$\psi(x) = \frac{\gamma x}{(x^{\nu} x_{\nu})^2} \exp\left\{-i\lambda(\beta_{\nu}\beta^{\nu})^{-1}(\overline{\chi}\chi)^{1/3}(\gamma\beta)\frac{\beta x}{x^{\nu} x_{\nu}}\right\}\chi.$$
 (2.4.16)

The corresponding C(1,3)-ungenerative solution has the form

$$\psi(x) = \frac{\gamma y}{(y^{\nu} y_{\nu})^2} \exp\left\{-i\lambda(\beta_{\nu}\beta^{\nu})^{-1}(\overline{\chi}\chi)^{1/3}(\gamma\beta)\left(\frac{\beta y}{y^{\nu} y_{\nu}} + \alpha\right)\right\}\chi.$$
 (2.4.17)

where $y_{\nu} = x_{\nu} + \delta_{\nu}$; δ_{ν} , α are arbitrary constants.

Following [101, 102] we construct solutions of the modified Dirac-Heisenberg Equation (2.4.10) with x = 0

$$\left\{i\gamma\partial - \lambda \frac{(\overline{\psi}\gamma_{\mu}\psi)\gamma^{\mu}}{\left[(\overline{\psi}\gamma_{\nu}\psi)(\overline{\psi}\gamma^{\nu}\psi)\right]^{1/3}}\right\}\psi = 0.$$
 (2.4.18)

Ansatz (2.3.51) reduces (2.4.18) to the system of ODEs

$$i(\gamma\beta)\frac{d\varphi}{d\omega} - \lambda \frac{(\overline{\varphi}\gamma_{\mu}\varphi)\gamma^{\mu}\varphi}{\left[(\overline{\varphi}\gamma_{\nu}\varphi)(\overline{\varphi}\gamma^{\nu}\varphi)\right]^{1/3}} = 0.$$
(2.4.19)

Consider the case of a complex coupling constant, that is

$$\lambda = \lambda_1 + i\lambda_2$$
, where $\lambda_1 = \operatorname{Re}\lambda$, $\lambda_2 = \operatorname{Im}\lambda$

Then we get the following solutions of Equation (2.4.19):

- if $\lambda_2 = 0$, then $\varphi(\omega) = \exp\{i\lambda\omega\}\chi$;
- if $\lambda_1 = 0$, then $\varphi(\omega) = \left(c + \frac{2}{3}\lambda_2\omega\right)^{-3/2}\chi;$

if $\lambda_1 \lambda_2 \neq 0$, then $\varphi(\omega) = (f_1 + if_2) \chi$;

$$f_{1} = \pm \left[(w - 2v)^{1/2} + (w + 2v)^{1/2} \right],$$

$$f_{2} = \mp \left[(w - 2v)^{1/2} - (w + 2v)^{1/2} \right],$$
(2.4.20)

$$w = \left(c_1 - 2\frac{\lambda_2}{\lambda_1}v^2\right)^{1/2}, \qquad \int \frac{dv}{w^{4/3}} = 2\lambda_2\omega + c_2.$$

In formulae (2.4.20)

$$\beta_{\mu} = \frac{\overline{\chi}\gamma_{\mu}\chi}{\left[(\overline{\chi}\gamma_{\nu}\chi)(\overline{\chi}\gamma^{\nu}\chi)\right]^{1/3}},$$

$$\omega = \frac{\beta x}{x_{\nu}x_{\nu}}.$$
(2.4.21)

The C(1,3)-ungenerative solution of Equation (2.4.18) with real coupling constant has the form

$$\psi(x) = \frac{\gamma y}{(y_{\nu}y^{\nu})^2} \exp\left\{i\lambda\left(\frac{\beta y}{y_{\nu}y^{\nu}} + \varpi\right)\right\}\chi,$$
(2.4.22)

where $y_{\nu} = x_{\nu} + \delta_{\nu}$; β_{ν} are given in (2.4.21); \mathfrak{B} , δ_{ν} are arbitrary constants.

3. Consider a conformally invariant equation for a spinor field with arbitrary conformal degree k [149]

$$\begin{bmatrix} i\gamma \mathcal{D} + \lambda (\overline{\psi}\psi)^{1/2k} \end{bmatrix} \psi = 0,$$

$$\mathcal{D}_{\mu} = \partial_{\mu} + \frac{1}{2k} \left(\frac{3}{2} - k\right) \partial_{\mu} \ln \left(\overline{\psi}\psi\right)$$
(2.4.23)

As we have already noted, spinor fields with arbitrary conformal degree play a fundamental role in unified field theory because they allow us to construct fields with any spin and with a given conformal degree (see Remark 2.3.5).

Equation (2.4.23) is merely the Dirac-Gursey Equation (2.1.5) when written in terms of the spinor

$$\psi_{\mathcal{D}} = (\overline{\psi}\psi)^{\frac{1}{2k}(\frac{3}{2}-k)}\psi, \qquad (2.4.24)$$

where $\psi_{\mathcal{D}}$ denotes a Dirac spinor with conformal degree k = 3/2, and ψ is a spinor field with arbitrary conformal degree k (the inverse formula expressing ψ via $\psi_{\mathcal{D}}$ is given in (2.3.47)).

In order to construct a scalar field with conformal degree 1 from a Dirac spinor field (it is this conformal degree which makes the scalar field satisfy the wave equation (see Table 2.3.1)) one must set k = 1/2 in (2.3.47) and then use the corresponding formula from (2.3.46). This gives

$$u(x) = (\overline{\psi}_{\mathcal{D}}\psi_{\mathcal{D}})^{1/3}.$$
(2.4.25)

Let us substitute solutions (2.4.11) and (2.4.12) of the Dirac-Gursey Equation (2.1.5) in (2.4.25). The result is

$$u(x) = \frac{3}{2\lambda} (x_{\nu} x^{\nu})^{-1/2}, \qquad (2.4.26)$$

$$u(x) = \frac{4}{\lambda} \frac{\alpha}{x_{\nu} x^{\nu} + \alpha^2}.$$
 (2.4.27)

Functions (2.4.26) and (2.4.27), as one can easily check, satisfy the nonlinear wave equation

$$\Box u + \lambda_1 u^3 = 0 \tag{2.4.28}$$

provided $\lambda = \frac{3}{2}\sqrt{\lambda_1}$ for (2.4.26) and $\lambda = \sqrt{2\lambda_1}$ for (2.4.27), $\lambda_1 > 0$. It is noteworthy [61^{*}] that these solutions of Equation (2.4.28) lead by means of the t'Hooft-Corrigan-Fairlie-Wilczek ansatz to the well-known meron solution of de Alfaro-Fubini-Furlan [4^{*}] and to the instanton solution of Belavin-Polyakov-Schwartz-Tyupkin [1^{*}] of the SU(2) Yang-Mills field equations (see Section 2.10).

4. As we have considered how to construct other fields from the spinor field (see Remark 2.3.5), it is natural to consider the inverse problem: how to construct the spinor field itself by means of other fields.

It is obvious that any spinor wave function can be represented as an expansion in the basis of the 4×4 matrices

$$\psi(x) = e^{i\varphi} \Big(U - i\gamma_5 V + \gamma_\mu J^\mu - \gamma_5 \gamma_\mu N^\mu + \frac{1}{2} \sigma_{\mu\nu} F^{\mu\nu} \Big) \chi \equiv e^{-i\varphi} \sum_{\alpha=1}^{16} R^\alpha \Gamma_\alpha \chi,$$
(2.4.29)

where

$$\varphi, \ R^{\alpha} = \{U, V, J^{\mu}, N^{\mu}, F^{\mu\nu}\}$$
(2.4.30)

are real scalar functions; χ is a constant spinor; and the rest of the notations are given in (2.1.2) and (2.4.2). Clearly, due to the Fierz-Pauli relations (2.4.9), which hold for any spinor field, the 16 functions R^{α} (2.4.30) are not independent. The set of functions R^{α} can always be chosen such that they satisfy the bispinor algebra (2.4.9), that is

$$J_{\mu}J^{\mu} = -N_{\mu}N^{\mu} = U^{2} + V^{2};$$

$$N_{\mu}J^{\mu} = 0;$$

$$F_{\mu\nu} = (U^{2} + V^{2})^{-1} \left[U\epsilon_{\mu\nu\rho\sigma}J^{\rho}N^{\sigma} - V(J_{\mu}N_{\nu} - J_{\nu}N_{\mu}) \right], \qquad U^{2} + V^{2} \neq 0.$$

(2.4.31)

This fact allows us to prove the factorization theorem (see [42]), according to which the general form of the spinor field (2.4.29) can be factored in the following manner:

$$\psi(x) = e^{-i\varphi} (U - i\gamma_5 V + \gamma_\mu J^\mu) \times$$

$$\times \left[1 - (U^2 + V^2)^{-1} (U + i\gamma_5 V) N_\nu \gamma_5 \gamma^\nu \right] \chi.$$
(2.4.32)

Let

$$\rho^{\alpha} = \{u, v, j_{\mu}, n_{\mu}, f_{\mu\nu} = -f_{\nu\mu}\}; \qquad j_0 > 0, \qquad u^2 + v^2 \neq 0 \qquad (2.4.33)$$

be a given set of real functions forming a bispinor algebra (2.4.9). Then the following statement (the inversion theorem) holds true.

Theorem 2.4.3 [42]. The spinor ψ which generates the given bispinor algebra (2.4.33), (2.4.8), (2.4.9) is determined by Equations (2.4.29) or (2.4.32), functions R^{α} (2.4.30) having the form

$$R^{\alpha} = \rho^{\alpha} / (4J), \qquad (2.4.34)$$

where

Chapter 2. Systems of Poincare-invariant Nonlinear PDEs

$$J^{2} = \frac{1}{4}\overline{\chi}(u - i\gamma_{5}v + j_{\mu}\gamma^{\mu}) \left[1 - (u^{2} + v^{2})^{-1}(u + i\gamma_{5}v)n_{\nu}\gamma_{5}\gamma^{\nu}\right]\chi \qquad (2.4.35)$$

So, to construct a spinor field ψ one needs, in general, the set of 16 real functions (2.4.33) which represent themselves as scalar, pseudoscalar, vector, pseudovector, and antisymmetric tensor of rank-two fields satisfying relations (2.4.9). In such a case spinor ψ is determined by use of formulae (2.4.29) or (2.4.32), (2.4.34), and (2.4.35). If it is necessary to construct a spinor field with conformal degree k one has to take fields ρ^{α} (2.4.33) with conformal degree 2k.

One can construct a spinor field in the following manner. Let U and J_{α} be scalar and vector fields. Then formulae

$$\psi_1(x) = \left(\gamma^{\mu} \frac{\partial U}{\partial x^{\mu}}\right) \chi,$$

$$\psi_2(x) = \gamma_{\mu} J^{\mu} \chi$$
(2.4.36)

determine spinor fields (in sense of ansatze) ψ_1 and ψ_2 . But in this case the connection between U and $u = \psi_1 \psi_1$, and between J_{μ} and $j_{\mu} = \overline{\psi}_2 \gamma_{\mu} \psi_2$ is not so simple as it was in (2.4.34). In particular, for U and u we find

$$u = \overline{\psi}_1 \psi_1 = \frac{\partial U}{\partial x^{\nu}} \frac{\partial U}{\partial x_{\nu}} (\overline{\chi}\chi)$$
(2.4.37)

It is clear that there are many ways of constructing spinor fields by use of formulae like (2.4.36). The Heisenberg ansatz (2.1.6) and its generalization (2.1.28) is an example of such a construction of a spinor field by means of three real scalar fields (functions) ω , f, g.

2.5. Reduction and exact solutions of coupled nonlinear PDEs for spinor and scalar fields

Consider the following system of PDEs

$$\left\{ i\gamma\partial - \left[\lambda_1 |u|^{k_1} + \lambda_2 (\overline{\psi}\psi)^{k_2}\right] \right\} \psi = 0,$$

$$\Box u + \left[\mu_1 |u|^{k_1} + \mu_2 (\overline{\psi}\psi)^{k_2}\right]^2 u = 0,$$
(2.5.1)

where $\psi = \psi(x)$ is a four-component spinor field; u = u(x) is a complex scalar field; $|u| = \sqrt{u^* u}$; $x \in R(1,3)$; λ_i, μ_i, k_i (i = 1,2) are arbitrary real parameters; γ_{μ} are Dirac matrices (2.1.2).

System (2.5.1) is invariant under the conformal group C(1,3) if $k_1 = 1$, $k_2 = 1/3$, conformal degrees being 3/2 for spinor fields and 1 for scalar fields; with arbitrary numbers k_1, k_2 , system (2.5.1) is invariant under $A\tilde{P}(1,3)$.

98

Solutions of system (2.5.1) under $k_1 = 1$, $k_2 = 1/3$ are sought by using the conformally invariant ansatze (2.3.51) and (2.3.52). These ansatze reduce (2.5.1) to the following system of ODEs:

$$i(\gamma\beta)\dot{\varphi} + \left[\lambda_1|v| + \lambda_2(\overline{\varphi}\varphi)^{1/3}\right]\varphi = 0,$$

$$\beta^2\ddot{v} + \left[\mu_1|v|^{k_1} + \mu_2(\overline{\varphi}\varphi)^{1/3}\right]^2 v = 0,$$

(2.5.2)

where dot indicates derivative with respect to $\omega = \frac{\beta x}{x^{\nu} x_{\nu}}$.

The simplest solutions of Equations (2.5.2) are as follows

$$\varphi(\omega) = \exp \{i\mathfrak{a}(\gamma\beta)\omega\} \chi, \qquad (2.5.3)$$
$$v(\omega) = \rho \exp \{i\omega\},$$

where $\mathfrak{B}, \rho, \beta_{\nu}$ are arbitrary constants satisfying relations

$$\begin{split} &\mathfrak{w}\beta^2 = \lambda_1 \rho + \lambda_2 (\overline{\chi}\chi)^{1/3} \\ &\beta^2 = \left[\mu_1 \rho + \mu_2 (\overline{\chi}\chi)^{1/3}\right]^2. \end{split} \tag{2.5.4}$$

Formulae (2.5.3), (2.3.51), (2.3.52) give solutions of Equations (2.5.1):

$$\psi(x) = \frac{\gamma y}{(y^{\nu} y_{\nu})^2} \exp\left\{i \hat{w}(\gamma \beta) \left(\frac{\beta y}{y^{\nu} y_{\nu}}\right) + \alpha\right) \chi,$$

$$u(x) = \frac{\rho}{y^{\nu} y_{\nu}} \exp\left\{i \left(\frac{\beta y}{y^{\nu} y_{\nu}}\right) + \alpha\right),$$

(2.5.5)

where the constants x, ρ, β_{ν} satisfy relations (2.5.4). Having made in (2.5.5) translation transformations we obtain, according to Section 2.3, the multiparameter family of C(1,3)-ungenerative solutions of Equations (2.5.1):

$$\psi(x) = \frac{\gamma y}{(y^{\nu} y_{\nu})^2} \exp\left\{i\varpi(\gamma\beta)\left(\frac{\beta y}{y_{\nu} y^{\nu}} + \alpha\right)\right\}\chi,$$
$$u(x) = \frac{\rho}{y_{\nu} y^{\nu}} \exp\left\{i\left(\frac{\beta y}{y_{\nu} y^{\nu}} + \alpha\right)\right\},$$

where $y_{\nu} = x_{\nu} + \delta_{\nu}$; α, δ are arbitrary real constants.

For finding solutions of Equations (2.5.1) with arbitrary k_1 and k_2 we use the ansatz (see [212], [13*])

$$\psi(x) = \left\{ ig_1(\omega) + \gamma_4 g_2(\omega) - \left[if_1(\omega) + \gamma_4 f_2(\omega) \right] i\gamma \partial \omega \right\} \chi,$$

$$u(x) = v(\omega)$$
(2.5.6)

where g_1, g_2, f_1, f_2 are some real scalar functions; v is a complex scalar function; $\omega = \omega(x)$ are new independent variables determined in (2.1.31); $\gamma_4 = -i\gamma_5$, and γ_5 is determined in (2.4.2).

By substituting (2.5.6) into (2.5.1) and taking into account (2.1.30) we obtain

$$\epsilon \ddot{v} + \frac{N}{\omega} \dot{v} = -\left[\mu_1 |v|^{k_1} + \widetilde{\mu}_2 F\right]^2 v,$$

$$\epsilon \dot{f}_1 + \frac{N}{\omega} f_1 = \left[\lambda_1 |v|^{k_1} + \widetilde{\lambda}_2 F\right] g_1,$$

$$\dot{g}_1 = -\left[\lambda_1 |v|^{k_1} + \widetilde{\lambda}_2 F\right] f_1,$$

$$\dot{g}_2 = \left[\lambda_1 |v|^{k_1} + \widetilde{\lambda}_2 F\right] f_2,$$

$$\epsilon \dot{f}_2 + \frac{N}{\omega} f_2 = -\left[\lambda_1 |v|^{k_1} + \widetilde{\lambda}_2 F\right] g_1,$$

(2.5.7)

where dot means derivative with respect to ω as defined in (2.1.31);

 $\widetilde{\lambda}_2 = \lambda_2(\overline{\chi}\chi)^{k_2}, \ \widetilde{\mu}_2 = \mu_2(\overline{\chi}\chi)^{k_2}, \ \text{and} \ F = \left[g_1^2 - g_2^2 + \epsilon(f_1^2 - f_2^2)\right]^{k_2}.$

We succeeded in constructing the general solution of system (2.5.7) only in the case where N = 0; in the rest of the cases particular solutions were found. Without going into details we cite the results obtained.

When $\epsilon = -1$, N = -2, -1 we have

$$f_{j}(\omega) = c_{j}\omega^{-1/2k_{2}}, \qquad j = 1, 2;$$

$$g_{j}(\omega) = \mp (-1)^{j}(1 + 2k_{2}N)^{1/2}c_{j}\omega^{-1/2k_{2}}; \qquad (2.5.8)$$

$$v(\omega) = E\omega^{-1/k_{1}}$$

and the conditions on arbitrary constants c_1, c_2 (real), and E (complex) hold

$$[(m-1)k_{1}-1]k_{1}^{-2} + \{\mu_{1}|E|^{k_{1}} + \widetilde{\mu}_{2} \left[2mk_{2}(c_{1}^{2}-c_{2}^{2})\right]^{k_{2}}\}^{2} = 0, \qquad (2.5.9)$$

$$\pm (1-2mk_{2})^{1/2} - 2k_{2}\{\lambda_{1}|E|^{k_{1}} + \widetilde{\lambda}_{2} \left[2mk_{2}(c_{1}^{2}-c_{2}^{2})\right]^{k_{2}}\} = 0,$$
with $k_{2} < \frac{1}{2}$, $k_{1} < \frac{1}{2}$, $m = -N$

with $k_2 < \frac{1}{2m}$, $k_1 < \frac{1}{m-1}$; $m \equiv -N$.

When $\epsilon = -1$, N = 0 we have

$$f_1 = c_1 \operatorname{sh} \left\{ -\lambda_1 \int |v(\omega)|^{k_1} d\omega + \widetilde{\lambda}_2 (c_3^2 - c_1^2)^{k_2} \omega + c_2 \right\},$$

$$f_2 = c_3 \operatorname{ch} \left\{ \lambda_1 \int |v(\omega)|^{k_1} d\omega + \widetilde{\lambda}_2 (c_3^2 - c_1^2)^{k_2} \omega + c_4 \right\},$$

$$g_{1} = c_{1} \operatorname{sh} \left\{ -\lambda_{1} \int |v(\omega)|^{k_{1}} d\omega + \widetilde{\lambda}_{2} (c_{3}^{2} - c_{1}^{2})^{k_{2}} \omega + c_{2} \right\}, \qquad (2.5.10)$$

$$g_{2} = c_{3} \operatorname{ch} \left\{ \lambda_{1} \int |v(\omega)|^{k_{1}} d\omega + \widetilde{\lambda}_{2} (c_{3}^{2} - c_{1}^{2})^{k_{2}} \omega + c_{4} \right\}, \qquad (2.5.10)$$

$$v(\omega) = \rho(\omega) \exp\{i\theta(\omega)\},$$

where

$$\int^{\rho(\omega)} [a_{-}(z) + c_{6}]^{-1/2} dz = \omega + c_{5},$$

$$\theta(\omega) = \int \rho^{-1/2}(\omega) d\omega + c_{8},$$

$$a_{-}(z) = \frac{\mu_{1}^{2}}{k_{1} + 1} z^{2(k_{1} + 1)} + \tilde{\mu}_{2}^{2} (c_{3}^{2} - c_{1}^{2})^{2k_{2}} z^{2} + \frac{4\mu_{1}\tilde{\mu}_{2}}{k_{1} + 2} (c_{3}^{2} - c_{1}^{2})^{k_{2}} z^{k_{1} + 2} + 2c_{5}^{2} z.$$

When $\epsilon = 1, N = 0$ we have

$$f_{1} = c_{1} \sin \left\{ \lambda_{1} \int |v(\omega)|^{k_{1}} d\omega - \tilde{\lambda}_{2} (c_{3}^{2} - c_{1}^{2})^{k_{2}} \omega + c_{2} \right\},$$

$$f_{2} = c_{3} \cos \left\{ \lambda_{1} \int |v(\omega)|^{k_{1}} d\omega - \tilde{\lambda}_{2} (c_{3}^{2} - c_{1}^{2})^{k_{2}} \omega + c_{4} \right\},$$

$$g_{1} = c_{1} \cos \left\{ -\lambda_{1} \int |v(\omega)|^{k_{1}} d\omega - \tilde{\lambda}_{2} (c_{3}^{2} - c_{1}^{2})^{k_{2}} \omega + c_{2} \right\},$$

$$g_{2} = c_{3} \sin \left\{ \lambda_{1} \int |v(\omega)|^{k_{1}} d\omega - \tilde{\lambda}_{2} (c_{3}^{2} - c_{1}^{2})^{k_{2}} \omega + c_{4} \right\},$$

$$v = \rho(\omega) \exp\{i\theta(\omega)\},$$
(2.5.11)

where

$$\begin{split} \int^{\rho(\omega)} \left[a_{+}(z) + c_{6} \right]^{-1/2} dz &= \omega + c_{6}, \\ \theta(\omega) &= \int \sqrt{\rho(\omega)} \, d\omega + c_{8}, \\ a_{+}(z) &= -\frac{\mu_{1}^{2}}{k_{1} + 1} z^{2(k_{1} + 1)} - \widetilde{\mu}_{2}^{2} (c_{1}^{2} - c_{3}^{2})^{2k_{2}} z^{2} - \frac{4\mu_{1} \widetilde{\mu}_{2}}{k_{1} + 2} (c_{1}^{2} - c_{3}^{2})^{k_{2}} z^{k_{1} + 2} + 2c_{5} z. \end{split}$$

When $\epsilon = 1, N = 1, 2, 3$ we have

$$f_{j} = c_{j}\omega^{-1/2k_{2}}, \quad j = 1, 2;$$

$$g_{k} = \mp (-1)^{j} (2Nk_{2} - 1)^{1/2} c_{j}\omega^{-1/2k_{2}};$$

$$v = E\omega^{-1/k_{1}},$$

(2.5.12)

101

where the arbitrary constants c_1, c_2 (real), and E (complex) satisfy the relations

$$(k_{1}+1)k_{1}^{-2} - Nk_{1}^{-1} + \left\{ \mu_{1}|E|^{k_{1}} + \widetilde{\mu}_{2} \left[2Nk_{2}(c_{1}^{2} - c_{2}^{2}) \right]^{k_{2}} \right\} = 0 \qquad (2.5.13)$$

$$\mp (2Nk_{2}-1)^{1/2} - 2k_{2} \left\{ \lambda_{1}|E|^{k_{1}} + \widetilde{\lambda}_{2} \left[2Nk_{2}(c_{1}^{2} - c_{2}^{2}) \right]^{k_{2}} \right\} = 0,$$

$$k_{1} > \frac{1}{N-1}, \qquad k_{2} > \frac{1}{2N}.$$

When $\epsilon = 1$; N = 2, 3; $K_1 = 2/(N - 1)$; $K_2 = 1/N$ we have

$$f_{j} = (-1)^{j} \theta \omega g_{j}(\omega), \qquad j = 1, 2;$$

$$g_{j} = c_{j}(1 + \theta^{2}w)^{-(N+1)/2}; \qquad (2.5.14)$$

$$v = E(1 + \theta^{2}\omega^{2})^{(1-N)/2},$$

where the arbitrary constants c_1, c_2, θ (real), and E (complex) satisfy the relations

$$\theta^{2}(N^{2}-1) = \left[\mu_{1}|E|^{2/(N-1)} + \widetilde{\mu}_{2}(c_{1}^{2}-c_{2}^{2})^{1/N}\right]^{2},$$

$$\theta(N+1) = \lambda_{1}|E|^{2/(N-1)} + \widetilde{\lambda}_{2}(c_{1}^{2}-c_{2}^{2})^{1/N}$$
(2.5.15)

By substituting the explicit form of functions $f_1(\omega)$, $f_2(\omega)$, $g_1(\omega)$, and $g_2(\omega)$ given in (2.5.8)-(2.5.14) and the corresponding ω written in (2.1.31) into ansatz (2.5.6), we obtain solutions for Equations (2.5.1). Under $k_1 = 1$ and $k_2 = 1/3$ one can apply to these solutions formulae of generating solutions (2.3.27), (2.3.34) and in this way obtain new families of solutions of Equations (2.5.1) when $k_1 = 1$, $k_2 = 1/3$.

2.6. Exact solutions of systems of nonlinear equations of quantum electrodynamics *

Consider nonlinear coupled PDEs which describe interacting electron (Dirac spinor) and electromagnetic fields

$$[\gamma^{\mu}(i\partial_{\mu} - eA_{\mu}) - m_{1}]\psi = 0,$$

$$\Box A_{\mu} - \partial_{\mu}\partial^{\nu}A_{\nu} + m_{2}^{2}A_{\mu} + \lambda A_{\mu}A^{\nu}A_{\nu} = e\overline{\psi}\gamma_{\mu}\psi,$$

(2.6.1)

where $\psi = \psi(x)$ is a four-component Dirac spinor; $A_{\mu} = A_{\mu}(x)$ is a fourvector potential of the electromagnetic field; $\mu, \nu = 0, 3; e, \lambda, m_1, m_2$ are real

^{*} Some results stated in this section were obtained in collaboration with R.Z. Zhdanov.

constants; γ_{μ} are Dirac matrices (2.1.2). When $\lambda = m_2 = 0$, system (2.6.1) coincides with the well-known standard equations of quantum electrodynamics [2, 31]. When $m_1 = m_2 = 0$, system (2.6.1) takes the form

$$\gamma^{\mu}(i\partial_{\mu} - eA_{\mu})\psi = 0,$$

$$\Box A_{\mu} - \partial_{\mu}\partial^{\nu}A_{\nu} + \lambda A_{\mu}A^{\nu}A_{\nu} = e\overline{\psi}\gamma_{\mu}\psi,$$
(2.6.2)

When $\lambda = 0$ system (2.6.1) takes the form

$$[\gamma^{\mu}(i\partial_{\mu} - eA_{\mu}) - m_{1}]\psi = 0,$$

$$\Box A_{\mu} - \partial_{\mu}\partial^{\nu}A_{\nu} + m_{2}^{2}A_{\mu} = e\overline{\psi}\gamma_{\mu}\psi.$$
(2.6.3)

And lastly, when $m_1 = m_2 = \lambda = 0$ system (2.6.1) takes the form

$$\gamma^{\mu}(i\partial_{\mu} - eA_{\mu})\psi = 0,$$

$$\Box A_{\mu} - \partial_{\mu}\partial^{\nu}A_{\nu} = e\overline{\psi}\gamma_{\mu}\psi.$$

(2.6.4)

Systems (2.6.1)-(2.6.4) are invariant under the following groups: (2.6.1) under P(1,3); (2.6.1) with $\lambda = m_2 = 0, m_1 \neq 0$ under P(1,3) \oplus U(1); (2.6.2) under C(1,3); (2.6.3) under P(1,3); (2.6.4) under P(1,3) \oplus U(1). Below we shall obtain exact solutions for these equations.

By substituting conformally invariant ansatze (2.3.51), (2.3.53) into (2.6.2) we get the system of ODEs

$$i(\gamma\beta)\dot{\varphi} = e(\gamma B)\varphi$$

$$\beta^{2}\ddot{B}_{\mu} - \beta_{\mu}\beta^{\nu}\ddot{B}_{\nu} + \lambda B_{\mu}B^{\nu}B_{\nu} = e\overline{\varphi}\gamma_{\mu}\varphi$$
(2.6.5)

where dots indicate derivatives with respect to $\omega = \beta x/(x_{\nu}x^{\nu})$. The simplest solution of Equations (2.6.5) is

$$\varphi(\omega) = \exp\{-ie\omega\}\chi, \qquad B_{\mu}(\omega) = \beta_{\mu}$$
 (2.6.6)

where constant spinor χ and arbitrary constants β_{μ} satisfy the conditions

$$\lambda \beta^2 \beta_\mu = e \overline{\chi} \gamma_\mu \chi. \tag{2.6.7}$$

From (2.6.6), returning to (2.3.51), (2.3.53), we get the following solution of Equations (2.6.2) (see [97], [98]):

$$\psi(x) = \frac{\gamma x}{(x_{\nu} x^{\nu})^2} \exp\left\{-ie\frac{\beta x}{x^{\nu} x_{\nu}}\right\} \chi,$$

$$A_{\mu}(x) = \frac{\beta_{\mu}}{x_{\nu} x^{\nu}} - 2x_{\mu} \frac{\beta x}{(x^{\nu} x_{\mu})^2} \equiv \partial_{\mu}\omega$$
(2.6.8)

whence follows (see formulae (2.3.55), (2.3.56)) the C(1,3)-ungenerative solution of Equations (2.6.2)

$$\psi(x) = \frac{\gamma y}{(y_{\nu}y^{\nu})^2} \exp\left\{-ie\left(\frac{\beta y}{y^{\nu}y_{\nu}} + \varkappa\right)\right\}\chi,$$

$$A_{\mu}(x) = \frac{\beta_{\mu}}{y_{\nu}y^{\nu}} - 2y_{\mu}\frac{\beta y}{(y^{\nu}y_{\mu})^2}$$
(2.6.9)

where $y_{\nu} = x_{\nu} + \delta_{\nu}$; δ_{ν} , æ are arbitrary real constants; β_{ν} , χ satisfy relations (2.6.7).

Solutions (2.6.8), (2.6.9) have the properties: $F_{\mu\nu} \equiv \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu} = 0$; and

$$\overline{\psi}\psi = \begin{cases} (x_{\nu}x^{\nu})^{-3}(\overline{\chi}\chi) & \text{for } (2.6.8)\\ (y_{\nu}y^{\nu})^{-3}(\overline{\chi}\chi) & \text{for } (2.6.9) \end{cases}$$

constant e (electron charge) tends to zero as soon as the coupling constant of self-interaction of the electromagnetic field $\lambda \to 0$.

Now we'll try to find solutions of system (2.6.2) with the help of the ansatz

$$\psi(x) = (\gamma b) \exp\{if(ax)\},\$$

$$A_{\mu}(x) = b_{\mu}g_{1}(ax) + a_{\mu}g_{2}(ax);\qquad b^{2} = 0$$
(2.6.10)

where a_{μ} , b_{μ} are arbitrary real constants; f, g_1, g_2 real scalar functions to be determined; χ , as usual, is a constant spinor.

Substitution of (2.6.10) into (2.6.2) gives rise to the following system of ODEs:

$$\dot{f} + eg_2 = 0,$$

$$a^2 \ddot{g}_1 = 2eb^{\nu} \overline{\chi} \gamma_{\nu} \chi - \lambda g_1 \left(a^2 g_2^2 + 2abg_1 g_2 \right),$$

$$ab \ddot{g}_1 = \lambda g_2 \left(a^2 g_2^2 + 2abg_1 g_2 \right)$$
(2.6.11)

where dot means differentiation with respect to $\omega = ax$.

System (2.6.11) can be integrated when $a^2 = 0$, $ab \neq 0$. In this case it takes the form

$$\dot{f} + eg_2 = 0,$$

 $eb^{
u}\overline{\chi}\gamma_{
u}\chi - \lambda abg_1^2g_2 = 0,$
 $\ddot{g}_1 - 2\lambda g_1g_2^2 = 0,$

whence follows

$$g_2 = \frac{x}{\lambda} g_1^{-2}, \qquad (2.6.12)$$

$$\ddot{g}_1 = \frac{2\varpi^2}{\lambda}g_1^{-3},$$

$$\dot{f} + eg_2 = 0,$$

where

$$\mathfrak{x} = \frac{e}{ab} b^{\mu} \overline{\chi} \gamma_{\mu} \chi.$$

The general solution of system (2.6.12), depending on sign of λ is given by the formulae:

when $\lambda > 0$, $c_1 \neq 0$ we have

$$g_{1}(\omega) = \pm c_{1}^{-1/2} \left[(c_{1}\omega + c_{2})^{2} + \frac{2\varpi^{2}}{\lambda} \right]^{1/2},$$

$$g_{2}(\omega) = \frac{\varpi c_{1}}{\lambda} \left[(c_{1}\omega + c_{2})^{2} + \frac{2\varpi^{2}}{\lambda} \right]^{-1},$$

$$f(\omega) = -\frac{e}{\sqrt{2\lambda}} \arctan\left[\frac{\sqrt{\lambda}}{\sqrt{2\omega}} (c_{1}\omega + c_{2}) \right];$$
(2.6.13)

when $\lambda < 0, c_1 \neq 0$ we have

$$g_{1}(\omega) = \pm c_{1}^{-1/2} \left[(c_{1}\omega + c_{2})^{2} + \frac{2\varpi^{2}}{\lambda} \right]^{1/2},$$

$$g_{2}(\omega) = \frac{\varpi c_{1}}{\lambda} \left[(c_{1}\omega + c_{2})^{2} + \frac{2\varpi^{2}}{\lambda} \right]^{-1},$$

$$f(\omega) = -\frac{e}{2\sqrt{2|\lambda|}} \ln \left| \frac{\sqrt{|\lambda|}(c_{1}\omega + c_{2}) + \sqrt{2}\varpi}{\sqrt{|\lambda|}(c_{1}\omega + c_{2}) - \sqrt{2}\varpi} \right|;$$
(2.6.14)

when $c_1 = 0, \lambda < 0$ we have

$$g_{1}(\omega) = \pm \left[\frac{2\sqrt{2}\omega}{\sqrt{|\lambda|}} + c_{3}\right]^{1/2},$$

$$g_{2}(\omega) = \left[-2\sqrt{2|\lambda|}\omega + \frac{\lambda}{\omega}c_{3}\right]^{-1},$$

$$f(\omega) = \frac{e}{2\sqrt{2|\lambda|}} \ln \left(2\sqrt{2|\lambda|}\omega - \frac{\lambda}{\omega}c_{3}\right);$$
(2.6.15)

where c_1, c_2, c_3 are arbitrary real constants. Inserting f, g_1, g_2 from (2.6.13) – (2.6.15) into (2.6.10) one obtains families of solutions of system (2.6.2):

Chapter 2. Systems of Poincare-invariant Nonlinear PDEs

 $\lambda > 0, \quad c_1 \neq 0:$

 $\lambda < 0, \quad c_1 \neq 0:$

$$\psi(x) = (\gamma b) \exp\left\{-\frac{ie}{\sqrt{2\lambda}} \arctan\left[\frac{\sqrt{\lambda}}{\sqrt{2\omega}} (c_1 a x + c_2)\right]\right\} \chi, \qquad (2.6.16)$$

$$A_{\mu}(x) = \pm b_{\mu}c_{1}^{-1/2} \left[(c_{1}ax + c_{2})^{2} + \frac{2a^{2}}{\lambda} \right]^{1/2} + a_{\mu} a_{\mu}c_{1} \left[\lambda (c_{1}ax + c_{2})^{2} + 2a^{2} \right]^{-1};$$

$$\psi(x) = (\gamma b) \exp\left\{-\frac{ie}{2\sqrt{2|\lambda|}} \ln\left|\frac{\sqrt{|\lambda|}(c_1ax + c_2) + \sqrt{2}\omega}{\sqrt{|\lambda|}(c_1ax + c_2) - \sqrt{2}\omega}\right|\right\}\chi,\tag{2.6.17}$$

$$A_{\mu}(x) = \pm b_{\mu}c_{1}^{-1/2} \left[(c_{1}ax + c_{2})^{2} + \frac{2\varpi^{2}}{\lambda} \right]^{1/2} + a_{\mu}\varpi c_{1} \left[\lambda (c_{1}ax + c_{2})^{2} + 2\varpi^{2} \right]^{-1};$$

$$\lambda < 0, \quad c_1 = 0:$$

$$\psi(x) = (\gamma b) \exp\left\{\frac{ie}{2\sqrt{2|\lambda|}} \ln\left(2\sqrt{|\lambda|}ax - \frac{\lambda}{a}c_3\right)\right\}\chi,$$

$$A_{\mu}(x) = \pm b_{\mu}\left[\frac{2\sqrt{2a}}{\sqrt{|\lambda|}}ax + c_3\right]^{1/2} + a_{\mu}\left[-2\sqrt{2|\lambda|}ax + \frac{\lambda}{a}c_3\right]^{-1}.$$
(2.6.18)

Let us recall that in (2.6.16)-(2.6.18) $b^2 = a^2 = 0$, $ab \neq 0$ (without loss of generality we can choose ab = 1);

$$\mathfrak{x}=\frac{e}{ab}b^{\nu}(\overline{\chi}\gamma_{\nu}\chi);$$

 c_1, c_2, c_3 are arbitrary real constants.

One can easily check that solutions (2.6.16)-(2.6.18) give $\overline{\psi}\psi = 0$ (due to the condition $b^2 = 0$); the tensor $F_{\mu\nu} \equiv \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu}$ is given by the expressions, respectively

$$F_{\mu\nu} = \pm (b_{\mu}a_{\nu} - b_{\nu}a_{\mu})\sqrt{c_{1}}(c_{1}ax + c_{2}) \left[(c_{1}ax + c_{2})^{2} + \frac{2\varpi^{2}}{\lambda} \right]^{-1/2};$$

$$F_{\mu\nu} = \pm (b_{\mu}a_{\nu} - b_{\nu}a_{\mu})\sqrt{c_{1}}(c_{1}ax + c_{2}) \left[(c_{1}ax + c_{2})^{2} + \frac{2\varpi^{2}}{\lambda} \right]^{-1/2};$$

$$F_{\mu\nu} = \pm (b_{\mu}a_{\nu} - b_{\nu}a_{\mu})\frac{\varpi}{\sqrt{2|\lambda|}} \left[\frac{2\varpi}{\sqrt{2|\lambda|}}ax + c_{3} \right]^{-1/2};$$
(2.6.19)

106

One can obtain from solutions (2.6.16)-(2.6.18) new families of solutions for Equations (2.6.2) if one applies to them the formulae of generating solutions (2.3.27) and (2.3.38).

Solutions for system (2.6.3) will be sought in the form

$$\psi(x) = (\gamma b) \exp \left\{-im_1(\gamma a)\omega + if(bx)\right\} \chi,$$

$$A_\mu(x) = b_\mu g(\omega), \qquad \omega = ax;$$

$$b^2 = ab = 0, \qquad a^2 = -1.$$

(2.6.20)

Substituting (2.6.20) into (2.6.3) gives rise to ODEs for the scalar function $g(\omega)$:

$$\ddot{g} - m_2^2 g + 2e(\theta_1 \operatorname{ch} 2m_1 \omega + \theta_2 \operatorname{sh} 2m_1 \omega) = 0, \qquad (2.6.21)$$

where

$$\theta_1 = \overline{\chi}\gamma b\chi, \quad \theta_2 = i\overline{\chi}\gamma a\gamma b\chi$$
 (2.6.22)

(one can easily confirm that θ_1 and θ_2 are real constants). With $m_2^2 > 0$, and $m_2^2 \neq 4m_1^2$ the general solution of Equation (2.6.21) has the form

$$g(\omega) = c_1 \operatorname{ch} m_2 \omega + c_2 \operatorname{sh} m_2 \omega + \frac{2e}{m_2^2 - 4m_1^2} (\theta_1 \operatorname{ch} 2m_1 \omega + \theta_2 \operatorname{sh} 2m_1 \omega). (2.6.23)$$

With $m_2^2 = 4m_1^2$ the general solution of Equation (2.6.21) has the form

$$g(\omega) = \left(-\frac{e\theta_1}{2m_1}\omega + c_1\right) \operatorname{ch} 2m_1\omega + \left(-\frac{e\theta_2}{2m_1}\omega + c_2\right) \operatorname{sh} 2m_1\omega.$$
(2.6.24)

And, lastly, under $m_2^2 = -m^2 < 0$ the general solution of Equation (2.6.21) has the form

$$g(\omega) = c_1 \cos m\omega + c_2 \sin m\omega - \frac{2e}{m^2 + 4m_1^2} (\theta_1 \operatorname{ch} 2m_1\omega + \theta_2 \operatorname{sh} 2m_1\omega). \quad (2.6.25)$$

In formulae $(2.6.23)-(2.6.25) c_1, c_2$ are arbitrary real constants.

Inserting expressions (2.6.23)-(2.6.25) into ansatz (2.6.20) we obtain three families of solutions for Equations (2.6.3): when $m_2^2 > 0$, $m_2^2 \neq 4m_1^2$ we have

$$A_{\mu}(x) = b_{\mu} \bigg[c_1 \operatorname{ch} m_2 a x + c_2 \operatorname{sh} m_2 a x + (2.6.26) \bigg]$$
(2.6.26)

$$+\frac{2e}{m_2^2-4m_1^2}\left(\theta_1\operatorname{ch} 2m_1ax+\theta_2\operatorname{sh} 2m_1ax\right)\bigg];$$

when $m_2^2 = 4m_1^2$ we have

Chapter 2. Systems of Poincare-invariant Nonlinear PDEs

$$A_{\mu}(x) = b_{\mu} \left[\left(-\frac{e\theta_1}{2m_1}ax + c_1 \right) \operatorname{ch} 2m_1ax + \left(-\frac{e\theta_2}{2m_1}ax + c_2 \right) \operatorname{sh} 2m_1ax \right];$$
(2.6.27)

when $m_2^2 = -m^2 < 0$ we have

$$A_{\mu}(x) = b_{\mu} \bigg[c_1 \cos max + c_2 \sin max -$$
(2.6.28)
$$2e - (a_1 + b_2) \bigg]$$

$$-\frac{2e}{m^2+4m_1^2}\left(\theta_1\operatorname{ch} 2m_1ax+\theta_2\operatorname{sh} 2m_1ax\right)\Big],$$

where θ_1, θ_2 are defined in (2.6.22); c_1, c_2, c_3 are arbitrary constants; for all three cases the spinor $\psi(x)$ has the form stated in (2.6.20) with the arbitrary differentiable function f(x); $a^2 = -1$, $ab = b^2 = 0$. It will be noted that solution (2.6.26) has resonance nature when $m_2^2 \to 4m_1^2$.

Let us select from solutions (2.6.27) a subfamily

$$\psi(x) = (\gamma b) \exp\left\{-im_1(\gamma a)ax\right\} \chi,$$

$$A_{\mu}(x) = \frac{e\theta}{2m_1} b_{\mu}(ax)(\operatorname{sh} 2m_1ax - \operatorname{ch} 2m_1ax),$$
(2.6.29)

where $\theta = \theta_1 = -\theta_2$. One can easily confirm that for these functions the following relations hold:

$$\Box \psi + m_1^2 \psi = 0,$$

$$\Box A_{\mu} + 4m_1 \left(m_1 - \frac{1}{ax} \right) A_{\mu} = 0$$
(2.6.30)

which apparently can be treated in such a manner. As a result of the interaction of the fields ψ and A_{μ} a particle comes into being. This particle has variable mass M = M(x)

$$M^2 = 4m_1 \left(m_1 - \frac{1}{ax} \right)$$
 (2.6.31)

So $M^2 > 0$ when $(ax)^{-1} < m_1$, M = 0 when $(ax)^{-1} = m_1$, and lastly $M^2 < 0$ when $(ax)^{-1} > m$. In the case of non-resonance interactions such an effect does not take place.

Let us try to find solutions for system (2.6.3) with the help of the ansatz

$$\psi(x) = (\gamma a) \exp\left\{-ie \int^{\omega_3} v(\omega_0, \tau) d\tau\right\} \varphi(\omega_0, \omega_1, \omega_2, \omega_3),$$

$$A_\mu(x) = a_\mu u(\omega_0, \omega_1, \omega_2, \omega_3) + d_\mu v(\omega_0, \omega_3)$$
(2.6.32)

where

$$\begin{aligned}
\omega_0 &= ax, & \omega_1 = bx, \\
w_2 &= cx, & \omega_3 = dx;
\end{aligned}$$
(2.6.33)

108

 $a_{\mu}, b_{\mu}, c_{\mu}, d_{\mu}$ are arbitrary real constants satisfying relations

$$a^{2} = d^{2} = ab = bc = cd = ac = bd = 0,$$

 $b^{2} = c^{2} = -1, \quad ad = 2;$
(2.6.34)

where $u = u(\omega_0, \omega_1, \omega_2, \omega_3)$, $v = v(\omega_0, \omega_3)$ are scalar real functions, and $\varphi = \varphi(\omega_0, \omega_1, \omega_2, \omega_3)$ is a four-component spinor.

After substituting (2.6.32) into (2.6.3) we get the following system of PDEs

$$i(\gamma b)\varphi_{1} + i(\gamma c)\varphi_{2} + m_{1}\varphi = 0,$$

$$u_{11} + u_{22} = m_{2}^{2}u - 2e\overline{\varphi}\gamma a\varphi + 4u_{03},$$

$$u_{3} + v_{0} = 0,$$

$$4v_{03} + m_{2}^{2}v = 0,$$

(2.6.35)

where $\varphi_{\mu} \equiv \frac{\partial \varphi}{\partial \omega_{\mu}}; \qquad u_{\mu} \equiv \frac{\partial u}{\partial \omega_{\mu}}; \qquad \mu = \overline{0,3}.$

Since $v = v(\omega_0, \omega_3)$, equation $v_0 + u_3 = 0$ from (2.6.35) yields

$$u = U(\omega_0, \omega_3) + U(\omega_0, \omega_1, \omega_2).$$
 (2.6.36)

Inserting this expression into (2.6.35) we get system of PDEs

$$\begin{split} 1^{\circ} & i\gamma b\varphi_{1} + i\gamma c\varphi_{2} + m_{1}\varphi = 0, \\ 2^{\circ} & \widetilde{U}_{11} + \widetilde{U}_{22} = m_{2}^{2}\widetilde{U} - 2e\overline{\varphi}\gamma a\varphi, \\ 3^{\circ} & m_{2}^{2}U + 4U_{03} = 0, \\ 4^{\circ} & m_{2}^{2}v + 4v_{03} = 0, \\ 5^{\circ} & v_{0} + U_{3} = 0. \end{split}$$
 (2.6.37)

System (2.6.37) is formally a nonlinear one, but if we solve the first equation 1° and insert the result into the second equation 2°, we will reduce it to a linear inhomogeneous system of PDEs which can be solved using the classical method of separation of variables.

Let us present two partial solutions of Equation 1° (2.6.37):

$$\varphi(\omega) = \exp\left\{-im_1(\gamma b)\omega_1\right\}\chi(w_0),\tag{2.6.38}$$

$$\varphi(\omega) = \left(\omega_1^2 + \omega_2^2\right)^{-1/4} \exp\left\{-\frac{1}{2}(\gamma b)(\gamma c) \arctan\frac{\omega_1}{\omega_2}\right\} \times$$

$$\times \exp\left\{-im_1(\gamma c) \left(\omega_1^2 + \omega_2^2\right)^{1/2}\right\} \chi(\omega_0),$$
(2.6.39)

where χ is an arbitrary spinor depending on $\omega_0 = ax$.

The general solution of Equation 2° (2.6.37) has the form

$$\widetilde{U} = \widetilde{U}^{(1)} + \widetilde{U}^{(2)},$$
 (2.6.40)

where $\widetilde{U}^{(1)}$ is the general solution of the Helmholtz equation

$$\widetilde{U}_{11}^{(1)} + \widetilde{U}_{22}^{(1)} = \widetilde{m}_2^2 U^{(1)}, \qquad (2.6.41)$$

and $\widetilde{U}^{(2)}$ is a partial solution of Equation 2° (2.6.37). Under φ from (2.6.38) a partial solution of Equation 2° (2.6.37) is given by the expression

$$\widetilde{U}^{(2)} = \begin{cases} \frac{2e}{m_2^2 - 4m_1^2} \left(\theta_1 \operatorname{ch} 2m_1\omega_1 + \theta_2 \operatorname{sh} 2m_1\omega_1\right), & m_2^2 \neq 4m_1^2; \\ -\frac{e}{2m_1} \left(\theta_1\omega_1 \operatorname{sh} 2m_1\omega_1 + \theta_2\omega_1 \operatorname{ch} 2m_1\omega_1\right), & m_2^2 = 4m_1^2; \end{cases}$$
(2.6.42)

where

$$\theta_{1} = \overline{\chi}(\omega_{0})\gamma a\chi(\omega_{0}),$$

$$\theta_{2} = i\overline{\chi}(\omega_{0})\gamma a\gamma b\chi(\omega_{0}).$$
(2.6.43)

Under φ from (2.6.39) a partial solution of Equation 2° (2.6.37) is expressed via the Bessel functions. We do not write it because it is rather cumbersome.

The general solution of Equations $3^{\circ} - 5^{\circ}$ (2.6.37) can easily be found by means of the method of separation of variables. It has the form

$$U = c_1 \exp\left\{4c_2\omega_0 - \frac{m_2^2}{c_2}\omega_3\right\},$$

$$v = \frac{m_2^2c_1}{4c_2^2} \exp\left\{4c_2\omega_0 - \frac{m_2^2}{c_2}\omega_3\right\},$$
(2.6.44)

where c_1, c_2 are arbitrary real constants, $c_2 \neq 0$.

Using (2.6.32)-(2.6.34), (2.6.36), (2.6.38), (2.6.40)-(2.6.44) it is not difficult to find corresponding solutions of system (2.6.3), but we do not do it here for the sake of brevity. Let us only note that these solutions, due to the term $\widetilde{U}^{(2)}$, (2.6.42) have the resonance nature analogous to that of solution (2.6.26).

Solutions for system (2.6.4) are sought in the form

$$\psi(x) = (\gamma b) \exp \{if(bx)\} \chi,$$

$$A_{\mu}(x) = b_{\mu}g(\omega), \qquad \omega = ax,$$
(2.6.45)

where a_{μ}, b_{μ} are arbitrary constants and $a^2 = -1$, $b^2 = ab = 0$. Substituting (2.6.45) into (2.6.4) gives rise to the ODE for a scalar real function

$$\ddot{g} + 2e\overline{\chi}\gamma b\chi = 0. \tag{2.6.46}$$

110

The general solution of Equation (2.6.46) is given by

$$g(\omega) = -e\overline{\chi}\gamma b\chi\omega^2 + c_1\omega + c_2$$

where c_1, c_2 are arbitrary constants. After substituting this expression into (2.6.45) we get a partial solution of system (2.6.4):

$$\psi(x) = (\gamma b) \exp\left\{if(bx)\right\}\chi,$$

$$A_{\mu}(x) = b_{\mu}\left[-e\overline{\chi}\gamma b\chi(ax)^{2} + c_{1}ax + c_{2}\right]$$
(2.6.47)

where f is an arbitrary differentiable real function; $a_{\mu}, b_{\mu}, c_1, c_2$ are arbitrary constants, $a^2 = -1$, $b^2 = ab = 0$. Having applied the formulae of generating solutions (2.3.27) and (2.3.38) to (2.6.47) one obtains a new family of solutions for system (2.6.4).

Now we shall demonstrate how to do a partial linearization of the standard equations of quantum electrodynamics which coincide with (2.6.1) under $\lambda = m_2 = 0$ and thereby construct wide families of their solutions.

Let us write the standard equations of quantum electrodynamics

$$[\gamma^{\mu}(i\partial_{\mu} - eA_{\mu}) - m]\psi = 0,$$

$$\Box A_{\mu} - \partial_{\mu}\partial^{\nu}A_{\nu} = e\overline{\psi}\gamma_{\mu}\psi.$$
(2.6.48)

Solutions of these equations are sought in the form

$$\psi(x) = (\gamma\beta)\varphi(\omega),$$

$$A_{\mu}(x) = \beta_{\mu}g(\omega),$$
(2.6.49)

where $g(\omega)$ is a scalar real function, $\varphi(\omega)$ is a spinor to be determined; ω are three new variables

$$\omega_1 = ax, \quad \omega_2 = bx, \quad \omega_3 = \beta x;
a^2 = b^2 = -1, \quad \beta^2 = ab = a\beta = b\beta = 0.$$
(2.6.50)

By substituting (2.6.49) into (2.6.48) we get the system of PDEs

$$(\gamma a)\varphi_1 + (\gamma b)\varphi_2 - im\varphi = 0,$$

$$g_{11} + g_{22} + 2e\overline{\varphi}\gamma\beta\varphi = 0,$$
(2.6.51)

where $\varphi_1 = \partial \varphi / \partial \omega_1$, $\varphi_2 = \partial \varphi / \partial \omega_2$ and so on.

Let $g = \tilde{g}(\omega, \varphi)$ be a partial solution of the second equation of system (2.6.51). Then on making the change of variables

$$\begin{aligned}
\phi(\omega) &= \varphi(\omega), \\
G(\omega) &= g(\omega) - \widetilde{g}(\omega, \varphi)
\end{aligned}$$
(2.6.52)

we reduce Equations (2.6.51) (and thereby system (2.6.48)) to the linear system of PDEs

$$(\gamma a)\phi_1 + (\gamma b)\phi_2 - im\phi = 0,$$

 $G_{11} + G_{22} = 0.$ (2.6.53)

This allows us to construct wide classes of solutions for system (2.6.48).

Suppose that the function φ in (2.6.51) does not depend on ω_2 . In such a case we have for $\varphi = \varphi(\omega_1, \omega_2)$ the ODE

 $(\gamma a)\varphi_1 - im\varphi = 0;$

the general solution of which has the form

$$\varphi = \exp\left\{-im_1(\gamma a)\omega_1\right\}\chi(\omega_3). \tag{2.6.54}$$

Inserting (2.6.54) into the second equation of system (2.6.51) we get

$$g_{11} + g_{22} + 2e \left[\theta_1 \operatorname{ch} 2m\omega_1 + \theta_2 \operatorname{sh} 2m\omega_1\right] = 0, \qquad (2.6.55)$$

where

$$\theta_1 = \overline{\chi}(\omega_3)\gamma\beta\chi(\omega_3), \qquad \theta_2 = i\overline{\chi}(\omega_3)\gamma a\gamma\beta\chi(\omega_3)$$

The general solution of Equation (2.6.55) can be written as

$$g(\omega) = f(z, \omega_3) + f(z^*, \omega_3) - \frac{e}{2m^2} (\theta_1 \operatorname{ch} 2m\omega_1 + \theta_2 \operatorname{sh} 2m\omega_1), \qquad (2.6.56)$$

where $z = \omega_1 + i\omega_2$; f is an analytic function of z. By substituting (2.6.56) and (2.6.54) into (2.6.49) we obtain a family of exact solutions of system (2.6.48)

$$\psi(x) = (\gamma\beta) \exp \{-im(\gamma a)ax\} \chi(\beta x),$$

$$A_{\mu}(x) = \beta_{\mu} \Big[f(z,\beta x) + f(z^*,\beta x) - \frac{e}{2m^2} \left(\theta_1 \operatorname{ch} 2max + \theta_2 \operatorname{sh} 2max \right) \Big], \quad (2.6.57)$$

where z = ax + ibx; θ_1, θ_2 are the same as in (2.6.55); a_μ, b_μ, β_μ are defined in (2.6.50).

It will be noted that solution (2.6.57) is analytic in the constant e and singular in the constant m.

2.7. On the linearization and general solution of two-dimensional Dirac-Heisenberg-Thirring and quantum electrodynamics equations

Following [106] and [212] we consider two two-dimensional nonlinear systems of PDEs: equations of the Dirac-Heisenberg-Thirring type

$$i\Gamma_{\mu}\partial_{\mu}\psi = \lambda(\bar{\psi}\Gamma_{\mu}\psi)\Gamma^{\mu}\psi, \qquad (2.7.1)$$

where $\mu = 0, 1;$

$$\Gamma_0 = \begin{pmatrix} 0 & i\sigma_2 \\ i\sigma_2 & 0 \end{pmatrix}, \qquad \Gamma_1 = \begin{pmatrix} 0 & \sigma_3 \\ \sigma_3 & 0 \end{pmatrix}$$
(2.7.2)

and the equations of quantum electrodynamics

$$\begin{bmatrix} i\gamma_{\mu}\partial_{\mu} - e\gamma_{\mu}A_{\mu} + \lambda(\bar{\psi}\gamma_{\mu}\psi)\gamma^{\mu} \end{bmatrix} \psi, \Box A_{\mu} - \partial_{\mu}\partial^{\nu}A_{\nu} = -e\bar{\psi}\gamma_{\mu}\psi$$
(2.7.3)

Here $\psi = \psi(x_0, x_1)$ is a four-component spinor; γ_{μ} are Dirac matrices (2.1.2); $A_{\mu} = A_{\mu}(x_0, x_1)$ is a vector-potential of electromagentic field; σ_2, σ_3 are Pauli matrices (2.1.3); λ, e are arbitrary real constants; $\mu, \nu = 0, 1$.

Systems (2.7.1) and (2.7.3) are linearized by means of reversible nonlocal transformations. This allows us to construct the general solutions of the systems.

1. Let us rewrite system (2.7.1) by introducing the two-component spinors φ and χ such that $\psi = \text{column}(\varphi, \chi)$:

$$i(\sigma_{2}\varphi_{0} + \sigma_{3}\varphi_{1}) = \lambda \left[i \left(|\varphi|^{2} + |\chi|^{2} \right) \sigma_{2} - i(\varphi^{+}\sigma_{2}\sigma_{3}\varphi + \chi^{+}\sigma_{2}\sigma_{3}\chi)\sigma_{3} \right] \varphi,$$

$$i(i\sigma_{2}\chi_{0} + \sigma_{3}\chi_{1}) = \lambda \left[i \left(|\varphi|^{2} + |\chi|^{2} \right) \sigma_{2} - i(\varphi^{+}\sigma_{2}\sigma_{3}\varphi + \chi^{+}\sigma_{2}\sigma_{3}\chi)\sigma_{3} \right] \chi,$$

$$(2.7.4)$$

In light-cone variables

$$\xi = x_0 - x_1, \qquad \eta = x_0 + x_1 \tag{2.7.5}$$

system (2.7.4) takes the form

$$\begin{split} i\varphi_{\xi}^{0} &= -\lambda \left(|\chi^{1}|^{2} + |\varphi^{1}|^{2} \right) \varphi^{0}, \\ i\varphi_{\eta}^{1} &= \lambda \left(|\chi^{0}|^{2} + |\varphi^{0}|^{2} \right) \varphi^{1}, \\ i\chi_{\xi}^{0} &= -\lambda \left(|\chi^{1}|^{2} + |\varphi^{1}|^{2} \right) \chi^{0}, \\ i\chi_{\eta}^{1} &= \lambda \left(|\chi^{0}|^{2} + |\varphi^{0}|^{2} \right) \chi^{1}, \end{split}$$

$$(2.7.6)$$

where superscripts mean components, and subscripts mean differentiation with respect to the corresponding index variable (ξ or η).

By means of the following nonlocal reversible change of variables

$$\varphi^{0} = v^{0} \exp\left\{i\lambda \int \left(|u^{1}|^{2} + |v^{1}|^{2}\right) d\xi\right\},\$$

$$\varphi^{1} = v^{1} \exp\left\{-i\lambda \int \left(|u^{0}|^{2} + |v^{0}|^{2}\right) d\eta\right\},\qquad(2.7.7)$$

Chapter 2. Systems of Poincare-invariant Nonlinear PDEs

$$\chi^{0} = u^{0} \exp\left\{i\lambda \int \left(|u^{1}|^{2} + |v^{1}|^{2}\right) d\xi\right\},\$$
$$\chi^{1} = u^{1} \exp\left\{-i\lambda \int \left(|u^{0}|^{2} + |v^{0}|^{2}\right) d\eta\right\},\$$

where v^0, v^1, u^0, u^1 are complex scalar functions depending on ξ and η , system (2.7.6) is reduced to the decomposed linear system of PDEs

$$v_{\xi}^{0} = u_{\xi}^{0} = 0, \qquad v_{\eta}^{1} = u_{\eta}^{1} = 0.$$
 (2.7.8)

The reader can easily confirm that substituting (2.7.7) into (2.7.6) results in (2.7.8). Therefore the problem of integrating Equations (2.7.1) is reduced to that of integrating Equations (2.7.8). The general solution of Equations (2.7.8) is not difficult to find. It has the form

$$u^{0} = F^{0}(\eta), \qquad u^{1} = F^{1}(\xi),$$

$$v^{0} = G^{0}(\eta), \qquad v^{1} = G^{1}(\xi),$$
(2.7.9)

where F^0, G^0, F^1, G^1 are arbitrary complex differentiable functions. After substituting (2.7.9) and (2.7.5) into (2.7.7) we obtain the general solution of the system (2.7.1)

$$\begin{split} \varphi^{0} &= F^{0}(x_{0} + x_{1}) \exp\left\{i\lambda \int^{x_{0} - x_{1}} (|F^{1}|^{2} + |G^{1}|^{2}) d\xi\right\}, \\ \varphi^{0} &= F^{1}(x_{0} - x_{1}) \exp\left\{-i\lambda \int^{x_{0} + x_{1}} (|F^{0}|^{2} + |G^{0}|^{2}) d\eta\right\}, \end{split}$$
(2.7.10)
$$\chi^{0} &= G^{0}(x_{0} + x_{1}) \exp\left\{i\lambda \int^{x_{0} - x_{1}} (|F^{1}|^{2} + |G^{1}|^{2}) d\xi\right\}, \\ \chi^{1} &= G^{1}(x_{0} - x_{1}) \exp\left\{i\lambda \int^{x_{0} + x_{1}} (|F^{0}|^{2} + |G^{0}|^{2}) d\eta\right\}, \end{split}$$

Remark 2.7.1. Whether or not it is possible to linearize the Dirac-Heisenberg-Thirring system (2.7.1) is closely related to the fact that this system admits infinite-dimensional local groups of invariance. For that reason there is another way of obtaining the general solution (2.7.10). It consists of the following. One must find a partial solution to system (2.7.1) and then multiply it by means of formulae of generating solutions (which are constructed according to (17)) until it becomes ungenerative. This latter solution will be the general solution of (2.7.10).

Using Lie's algorithm one can prove that the maximal invariance group of Equations (2.7.4) is $O(4) \otimes O(4) \otimes G_{\infty}$, where G_{∞} is the infinite-dimensional group of transformations

$$\begin{aligned} x_0 &\to x_0' = \frac{1}{2} \left[\int_{-\infty}^{x_0 - x_1} f_1^{-2}(\xi) d\xi + \int_{0}^{x_0 + x_1} f_0^{-2}(\eta) d\eta \right], \\ x_1 &\to x_1' = \frac{1}{2} \left[\int_{-\infty}^{x_0 + x_1} f_0^{-2}(\eta) d\eta - \int_{0}^{x_0 - x_1} f_1^{-2}(\xi) d\xi \right], \\ \varphi^0 &\to \varphi^{0\prime} = f_0(x_0 + x_1) \varphi^0, \\ \chi^0 &\to \chi^{0\prime} = f_0(x_0 + x_1) \chi^0, \\ \varphi^1 &\to \varphi^{1\prime} = f_1(x_0 - x_1) \varphi^1, \\ \chi^1 &\to \chi^{1\prime} = f_1(x_0 - x_1) \chi^1, \end{aligned}$$

where f_0, f_1 are arbitrary real functions.

Note that system (2.7.4) also admits the following nonlocal group of transformations

$$\begin{split} \chi^{0} &\to \chi^{0}{}' = a \chi^{0} \exp\left\{i\lambda \int \left[\left(|b|^{2}-1\right)|\chi^{1}|^{2}+\left(|d|^{2}-1\right)|\varphi^{1}|^{2}\right] d\xi\right\},\\ \chi^{1} &\to \chi^{1}{}' = b \chi^{1} \exp\left\{-i\lambda \int \left[\left(|a|^{2}-1\right)|\chi^{0}|^{2}+\left(|c|^{2}-1\right)|\varphi^{0}|^{2}\right] d\eta\right\},\\ \varphi^{0} &\to \varphi^{0}{}' = c \varphi^{0} \exp\left\{i\lambda \int \left[\left(|b|^{2}-1\right)|\chi^{1}|^{2}+\left(|d|^{2}-1\right)|\varphi^{1}|^{2}\right] d\xi\right\},\\ \varphi^{1} &\to \varphi^{1}{}' = b \varphi^{1} \exp\left\{-i\lambda \int \left[\left(|a|^{2}-1\right)|\chi^{0}|^{2}+\left(|c|^{2}-1\right)|\varphi^{0}|^{2}\right] d\eta\right\}, \end{split}$$

where a, b, c, d are arbitrary complex parameters.

2. Let us consider the two-dimensional equations of quantum electrodynamics (2.7.3). Rewriting (2.7.3) in terms of light-cone variables (2.7.5) and representing ψ as a column vector with four components $\psi^0, \psi^1, \psi^2, \psi^3$ we get the equivalent system of PDEs:

$$\begin{split} &i\psi_{\xi}^{0} = -\left[\frac{1}{2}e(A^{1} - A^{0}) + \lambda\left(|\psi^{1}|^{2} + |\psi^{3}|^{2}\right)\right]\psi^{0}, \\ &i\psi_{\eta}^{1} = \left[\frac{1}{2}e(A^{1} + A^{0}) + \lambda\left(|\psi^{0}|^{2} + |\psi^{2}|^{2}\right)\right]\psi^{1}, \\ &i\psi_{\xi}^{2} = -\left[\frac{1}{2}e(A^{1} - A^{0}) + \lambda\left(|\psi^{1}|^{2} + |\psi^{3}|^{2}\right)\right]\psi^{2}, \end{split}$$
(2.7.11)

Chapter 2. Systems of Poincare-invariant Nonlinear PDEs

$$\begin{split} &i\psi_{\eta}^{3} = \left[\frac{1}{2}e(A^{1} + A^{0}) + \lambda\left(|\psi^{0}|^{2} + |\psi^{2}|^{2}\right)\right]\psi^{3},\\ &\left(\partial_{\xi} - \partial_{\eta}\right)\left(A_{\eta}^{0} - A_{\xi}^{0} + A_{\eta}^{1} + A_{\xi}^{1}\right) = e\left(|\psi^{0}|^{2} + |\psi^{1}|^{2} + |\psi^{2}|^{2} + |\psi^{3}|^{2}\right),\\ &\left(\partial_{\xi} + \partial_{\eta}\right)\left(A_{\eta}^{0} - A_{\xi}^{0} + A_{\eta}^{1} + A_{\xi}^{1}\right) = e\left(-|\psi^{0}|^{2} + |\psi^{1}|^{2} - |\psi^{2}|^{2} + |\psi^{3}|^{2}\right). \end{split}$$

System (2.7.11) is linearized by means of the following nonlocal reversible change of variables:

$$\begin{split} \psi^{0} &= u^{0} \exp\left\{i\lambda \int \left(|u^{1}|^{2} + |u^{3}|^{2}\right) d\xi + \frac{ie}{2} \int (A^{1} - A^{0}) d\xi\right\}, \\ \psi^{1} &= u^{1} \exp\left\{-i\lambda \int \left(|u^{0}|^{2} + |u^{2}|^{2}\right) d\eta + \frac{ie}{2} \int (A^{1} + A^{0}) d\eta\right\}, \end{split}$$
(2.7.12)
$$\psi^{2} &= u^{2} \exp\left\{i\lambda \int \left(|u^{1}|^{2} + |u^{3}|^{2}\right) d\xi + \frac{ie}{2} \int (A^{1} - A^{0}) d\xi\right\}, \\ \psi^{3} &= u^{3} \exp\left\{-i\lambda \int \left(|u^{0}|^{2} + |u^{2}|^{2}\right) d\eta + \frac{ie}{2} \int (A^{1} + A^{0}) d\eta\right\}, \end{split}$$

where u^0, u^1, u^2, u^3 are complex functions. Substituting (2.7.12) into (2.7.11) gives rise to the equations for the functions $u^0, u^1, u^2, u^3, A^0, A^1$:

$$\begin{split} u_{\xi}^{0} &= u_{\xi}^{2} = 0, \qquad u_{\eta}^{1} = u_{\eta}^{3} = 0, \\ \left(\partial_{\xi} - \partial_{\eta}\right) \left(A_{\eta}^{0} - A_{\xi}^{0} + A_{\eta}^{1} + A_{\xi}^{1}\right) &= e\left(|u^{0}|^{2} + |u^{1}|^{2} + |u^{2}|^{2} + |u^{3}|^{2}\right), \quad (2.7.13) \\ \left(\partial_{\xi} + \partial_{\eta}\right) \left(A_{\eta}^{0} - A_{\xi}^{0} + A_{\eta}^{1} + A_{\xi}^{1}\right) &= e\left(-|u^{0}|^{2} + |u^{1}|^{2} - |u^{2}|^{2} + |u^{3}|^{2}\right). \end{split}$$

After integrating this system and substituting the result into (2.7.12), we obtain the general solution of system (2.7.3):

$$\begin{split} A^{0} &= e^{\sum_{x_{0}+x_{1}}^{x_{0}+x_{1}} \int_{x_{0}}^{z} \left(|u^{0}|^{2} + |u^{2}|^{2} \right) d\eta dz + \frac{\partial f}{\partial x_{0}}, \\ A^{0} &= -e^{\sum_{x_{0}-x_{1}}^{x_{0}-x_{1}} \int_{x_{0}}^{z} \left(|u^{1}|^{2} + |u^{3}|^{2} \right) d\xi dz - \frac{\partial f}{\partial x_{1}}, \\ \psi^{0} &= u^{0}(x_{0} + x_{1}) \exp\left\{ i \int_{x_{0}-x_{1}}^{x_{0}-x_{1}} \left[\lambda \left(|u^{1}|^{2} + |u^{3}|^{2} \right) + \frac{e}{2} \left(A^{1} - A^{0} \right) \right] d\xi \right\}, \\ \psi^{1} &= u^{1}(x_{0} - x_{1}) \exp\left\{ -i \int_{x_{0}-x_{1}}^{x_{0}+x_{1}} \left[\lambda \left(|u^{0}|^{2} + |u^{2}|^{2} \right) + \frac{e}{2} \left(A^{1} + A^{0} \right) \right] d\eta \right\}, \\ \psi^{2} &= u^{2}(x_{0} + x_{1}) \exp\left\{ i \int_{x_{0}-x_{1}}^{x_{0}-x_{1}} \left[\lambda \left(|u^{1}|^{2} + |u^{3}|^{2} \right) + \frac{e}{2} \left(A^{1} + A^{0} \right) \right] d\eta \right\}, \end{split}$$

$$u^{3}(x_{0}-x_{1})\exp\left\{-i\int_{x_{0}+x_{1}}^{x_{0}+x_{1}}\left[\lambda\left(|u^{0}|^{2}+|u^{3}|^{2}\right)+\frac{e}{2}\left(A^{1}-A^{0}
ight)\right]d\eta
ight\},$$

where u^0, u^1, u^2, u^3 are arbitrary complex functions, and $f(x_0, x_1)$ is an arbitrary real function.

2.8. Symmetry analysis of nonlinear equations of classical electrodynamics

In describing the electromagnetic field in various media, an investigator commonly uses Maxwell's equations in the form

$$\frac{\partial \vec{D}}{\partial t} = \operatorname{rot} \vec{H}, \qquad \operatorname{div} \vec{D} = 0,$$

$$\frac{\partial \vec{B}}{\partial t} = -\operatorname{rot} \vec{E}, \qquad \operatorname{div} \vec{B} = 0,$$
(2.8.1)

where \vec{E} and \vec{H} are vectors of intensity and \vec{D} and \vec{B} are vectors of the induction of electric and magnetic fields. The system of Equations (2.8.1) is undetermined, and it is therefore necessary to add to it the constitutive equations (equations connecting \vec{D} , \vec{B} , \vec{E} , and \vec{H}) which reflect the properties of the medium. As a rule, the constitutive equations considerably restrict the symmetry of the whole system of field equations. From our point of view [107] one can use symmetry as a guide-line in constructing and selecting constitutive equations. Below we exploit this idea to describe constraints between \vec{D} , \vec{B} , \vec{E} , and \vec{H} which conserve Poincare and conformal symmetry of the field equations (2.8.1). Note, that in [143*] it is proposed another approach to nonlinear generalization of Maxwell's equations. It is based on the idea of changing the speed of light c onto $v = f(\vec{E}^2 - \vec{H}^2, \vec{E} \cdot \vec{H})$, the latter being the speed of electromagnetic field energy transfer.

Symmetry properties of Maxwell's equations in vacua were investigated by Lorentz, Poincare, and Einstein, and in full detail by Bateman and Cunningham [25, 43]. The two latter authors established that Maxwell's equations possess only a 16-parameter (in Lie's sense, of course) group of invariance which contains as a subgroup the 15-parameter conformal group C(1,3). As mentioned above, the essential difference between Equations (2.8.1) and Maxwell's equations for an electromagnetic field in vacuum is the strong undetermination of the former. Therefore it is natural to expect system (2.8.1) to have wider symmetry than Maxwell's equations in vacuum. It is appropriate to point out that Maxwell's equations in vacuum represent an overdetermined system of eight equations for six unknown quantities. The symmetry properties of Equations (2.8.1) are summarized by the following statement [107]. **Theorem 2.8.1.** Equations (2.8.1) are invariant under infinite-dimensional Lie algebra, with basis elements having the form

$$X_{1} = \xi^{\mu}(x)\partial_{\mu} + \eta_{\mu\nu}\partial_{F_{\mu\nu}} + \tilde{\tau}_{\mu\nu}\partial_{H_{\mu\nu}},$$

$$X_{2} = \frac{1}{2}F_{\mu\nu}\partial_{F_{\mu\nu}},$$

$$X_{3} = \frac{1}{2}\tilde{H}_{\mu\nu}\partial_{\tilde{H}_{\mu\nu}},$$

$$X_{4} = \frac{1}{2}F_{\mu\nu}\partial_{\tilde{H}_{\mu\nu}},$$

$$X_{5} = \frac{1}{2}\tilde{H}_{\mu\nu}\partial_{F_{\mu\nu}},$$
(2.8.2)

where $\xi^{\mu}(x)$ are arbitrary differentiable functions; $\mu, \nu = \overline{0,3}$;

$$F_{\mu\nu} = -F_{\mu\nu}, \qquad F_{0a} = E_a, \qquad F_{ab} = \epsilon_{abc} B_c$$

$$\widetilde{F}_{\mu\nu} = -\frac{1}{2} \epsilon_{\mu\nu\rho\sigma} F^{\rho\sigma}; \qquad H_{\mu\nu} = -H_{\nu\mu},$$
(2.8.3)

$$H_{0a} = D_a, \qquad H_{ab} = -\epsilon_{abc}H_c, \qquad \widetilde{H}_{\mu\nu} = -\frac{1}{2}\epsilon_{\mu\nu\rho\sigma}H^{\rho\sigma};$$

$$\eta_{\mu\nu} = -F_{\mu\alpha}\xi^{\alpha}_{\nu} - F_{\alpha\nu}\xi^{\alpha}_{\mu} \qquad \left(\xi^{\alpha}_{\nu} \equiv \frac{\partial\xi^{\alpha}}{\partial x^{\nu}}\right)$$
$$\tilde{\tau}_{\mu\nu} = -\tilde{H}_{\mu\alpha}\xi^{\alpha}_{\nu} - \tilde{H}_{\alpha\nu}\xi^{\alpha}_{\mu}.$$
(2.8.4)

Proof. Using tensors $F_{\mu\nu}$ and $\tilde{H}_{\mu\nu}$ from (2.8.3) we rewrite Equations (2.8.1) as follows:

$$L_{1} \equiv \partial_{\alpha}F_{\mu\nu} + \partial_{\mu}F_{\nu\alpha} + \partial_{\nu}F_{\alpha\mu} = 0,$$

$$L_{2} \equiv \partial_{\alpha}\widetilde{H}_{\mu\nu} + \partial_{\mu}\widetilde{H}_{\nu\alpha} + \partial_{\nu}\widetilde{H}_{\alpha\mu} = 0.$$
(2.8.5)

The proof of the theorem is reduced to the application of Lie's algorithm to system (2.8.5). This means that one has to construct all differential operators

$$X = \xi^{\mu}\partial_{\mu} + \eta^{\mu\nu}\partial_{F^{\mu\nu}} + \tilde{\tau}^{\mu\nu}\partial_{\tilde{H}^{\mu\nu}}$$
(2.8.6)

satisfying the conditions of invariance

$$\begin{array}{c|c} X_{1}L_{1} \\ L_{1}=0 \\ L_{2}=0 \end{array} = 0, \qquad \begin{array}{c} X_{1}L_{2} \\ L_{2}=0 \end{array} = 0, \qquad (2.8.7) \end{array}$$

where the operator X_1 is constructed according to the prologation formulae (4).

From (2.8.7) we get the defining equations

$$\begin{aligned} \xi^{\mu}_{F_{\alpha\beta}} &= \xi^{\mu}_{\widetilde{H}_{\alpha\beta}} = 0, \\ \partial_{\alpha}\eta_{\mu\nu} + \partial_{\mu}\eta_{\nu\alpha} + \partial_{\nu}\eta_{\alpha\mu} &= 0, \\ \partial_{\alpha}\widetilde{\tau}_{\mu\nu} + \partial_{\mu}\widetilde{\tau}_{\nu\alpha} + \partial_{\nu}\widetilde{\tau}_{\alpha\mu} &= 0, \end{aligned}$$
(2.8.8)
$$\eta^{\mu\nu}_{F_{\rho\sigma}} &= \widetilde{\tau}^{\mu\nu}_{\widetilde{H}_{\rho\sigma}} = \xi^{\rho}_{\nu}\delta_{\mu\sigma} - \xi^{\sigma}_{\nu}\delta_{\mu\rho} + \xi^{\sigma}_{\mu}\delta_{\nu\rho} - \xi^{\rho}_{\mu}\delta_{\nu\sigma}, \\ \widetilde{\tau}^{\mu\nu}_{\widetilde{H}_{\mu\nu}} &= c_{1}, \quad \eta^{\mu\nu}_{F_{\mu\nu}} = c_{2}, \quad \eta^{\mu\nu}_{\widetilde{H}_{\mu\nu}} = c_{3}, \quad \eta^{\mu\nu}_{\widetilde{H}_{\mu\nu}} = c_{4} \qquad \text{(no sum over } \mu, \nu), \end{aligned}$$

where c_1, c_2, c_3, c_4 are arbitrary constants.

The general solution of Equations (2.8.8) determines the maximal invariance algebra of system (2.8.1). The theorem is proved.

Theorem 2.8.2 [107]. System (2.8.5) is invariant under 20-dimensional Lie algebra AIGL(4,R) which contains as subalgebras Poincare algebra AP(1,3) and Galilei algebra AG(1.3).

Proof. According to the previous theorem, Equations (2.8.5) admit infinitesimal operators (2.8.6) with arbitrary functions $\xi^{\mu}(x)$. Consider $\xi^{\mu}(x) = c^{\mu\nu}x_{\nu} + a^{\mu}$ ($c^{\mu\nu}$, a^{μ} are arbitrary constants). If $c_{\mu\nu} = -c_{\nu\mu}$ then the operator X_1 (2.8.2) yields basis elements of AP(1,3)

$$P_{0} = i\partial_{0}, \qquad P_{a} = -\partial_{a}, \qquad a = 1, 2, 3, J_{\mu\nu} = x_{\mu}P_{\nu} - x_{\nu}P_{\mu} + (S_{\mu\nu}\psi)^{l}\partial_{\psi^{l}}$$
(2.8.9)

where the summation is assumed over l from 1 to 12; ψ is a 12-component column $(E_1, \ldots, B_1, \ldots, D_1, \ldots, H_1, \ldots, H_3)$; $S_{\mu\nu} = -S_{\nu\mu}$ are matrices of the form

$$S_{ab} = \begin{pmatrix} \hat{S}_{ab} & \hat{0} & \hat{0} & \hat{0} \\ \hat{0} & \hat{S}_{ab} & \hat{0} & \hat{0} \\ \hat{0} & \hat{0} & \hat{S}_{ab} & \hat{0} \\ \hat{0} & \hat{0} & \hat{0} & \hat{S}_{ab} \end{pmatrix} \quad S_{0a} = \frac{1}{2} \epsilon_{abc} \begin{pmatrix} \hat{0} & \hat{0} & \hat{0} & \hat{S}_{bc} \\ \hat{0} & \hat{0} & -\hat{S}_{bc} & \hat{0} \\ \hat{0} & \hat{S}_{bc} & \hat{0} & \hat{0} \\ -\hat{S}_{bc} & \hat{0} & \hat{0} & \hat{0} \end{pmatrix},$$
(2.8.10)

$$\widehat{S}_{12} = \begin{pmatrix} 0 & -i & 0 \\ i & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}, \quad \widehat{S}_{23} = \begin{pmatrix} 0 & 0 & 0 \\ 0 & 0 & -i \\ 0 & i & 0 \end{pmatrix}, \quad \widehat{S}_{31} = \begin{pmatrix} 0 & 0 & i \\ 0 & 0 & 0 \\ -i & 0 & 0 \end{pmatrix},$$

where $\hat{0}$ is a three-row zero matrix. If we take $c_{0\mu} = 0$, $c_{ab} = -c_{ba}$ then the operator X_1 (2.8.2) yields basis elements of AG(1,3)

$$P_{0} = i\partial_{0}, \qquad P_{a} = -i\partial_{a},$$

$$J_{ab} = x_{a}P_{b} - x_{b}P_{a} + (S_{ab}\psi)^{l}\partial_{\psi^{l}} \qquad (2.8.11)$$

$$G_{a} = x_{0}P_{a} + (M_{a}\psi)^{l}\partial_{\psi^{l}},$$

where

$$M_a = \frac{1}{2} \epsilon_{abc} \begin{pmatrix} \widehat{0} & \widehat{S}_{bc} & \widehat{0} & \widehat{0} \\ \widehat{0} & \widehat{0} & \widehat{0} & \widehat{0} \\ \widehat{0} & \widehat{0} & \widehat{0} & \widehat{0} \\ \widehat{0} & \widehat{0} & \widehat{S}_{bc} & \widehat{0} \end{pmatrix}$$

The theorem is proved.

In a similar way one can confirm that operator X_1 (2.8.2) also includes basis elements of AC(1,3) as well as basis elements of ASch(1,3). They have the form (2.8.9) and

$$D = x^{\nu} P_{\nu} + 2i\psi^{l}\partial_{\psi^{l}},$$

$$K_{\mu} = 2x_{\mu}D - x^{2}P_{\mu} + 2(x^{\nu}S_{\mu\nu}\psi)^{l}\partial_{\psi^{l}};$$
(2.8.12)

(2.8.11) and

$$D = 2x_0 P_0 - x_a P_a + (\lambda_0 \psi)^i \partial_{\psi^i},$$

$$\Pi = x_0^2 P_0 + x_0 (\lambda_0 \psi)^l \partial_{\psi^i} - x_a G_a,$$
(2.8.13)

where

$$\lambda_0 = i \begin{pmatrix} 3 & 0 & 0 & 0 \\ 0 & 2 & 0 & 0 \\ 0 & 0 & 2 & 0 \\ 0 & 0 & 0 & 3 \end{pmatrix} \times I_3,$$

with I_3 the three-dimensional unit matrix.

The above results mean that system (2.8.1) without constraints (constitutive equations) satisfies both Lorentz-Poincare-Einstein and Galilei principles of relativity. As was indicated in [103] the same holds for the nonlinear Euler equation of an ideal fluid.

Let us write down the final Lorentz and Galilei transformations which are admitted by Equations (2.8.1) (they are generated by operators J_{0a} (2.8.9) and G_a (2.8.11), respectively):

$$\begin{aligned} x'_{0} &= \gamma (x_{0} - \vec{x} \cdot \vec{v}) & \vec{x}' = \vec{x} + \frac{\gamma - 1}{v^{2}} (\vec{x} \cdot \vec{v}) \vec{v} - \gamma \vec{v} x_{0}, \qquad \left(\gamma = \frac{1}{\sqrt{1 - v^{2}}}\right) \\ \vec{E}' &= \gamma \vec{E} + \frac{1 - \gamma}{v^{2}} \vec{v} (\vec{E} \cdot \vec{v}) - \gamma \vec{v} \times \vec{B}, \\ \vec{B}' &= \gamma \vec{B} + \frac{1 - \gamma}{v^{2}} \vec{v} (\vec{B} \cdot \vec{v}) + \gamma \vec{v} \times \vec{E}, \qquad (2.8.14a) \\ \vec{H}' &= \gamma \vec{H} + \frac{1 - \gamma}{v^{2}} \vec{v} (\vec{H} \cdot \vec{v}) + \gamma \vec{v} \times \vec{D}, \vec{D}' = \gamma \vec{D} + \frac{1 - \gamma}{v^{2}} \vec{v} (\vec{D} \cdot \vec{v}) - \gamma \vec{v} \times \vec{H}; \\ x'_{0} &= x_{0}, \qquad \vec{x}' = \vec{x} + \vec{v} x_{0}, \\ \vec{E}' &= \vec{E} - \vec{v} \times \vec{B}, \qquad \vec{B}' = \vec{B}, \end{aligned}$$

120

 $\vec{H}' = \vec{H} + \vec{v} \times \vec{D}, \qquad \vec{D}' = \vec{D}.$

Let us consider constitutive equations of the form

$$H_{\mu\nu} = \phi_{\mu\nu}(F_{01}, F_{02}, F_{03}, F_{12}, F_{23}, F_{31}) \tag{2.8.15}$$

where $\phi_{\mu\nu} = -\phi_{\nu\mu}$ are smooth functions of components of the tensor $F_{\mu\nu}$.

Theorem 2.8.3 [107]. The system of Equations (2.8.5), (2.8.15) is invariant under the Poincare group if and only if

$$H_{\mu\nu} = MF_{\mu\nu} + N\widetilde{F}_{\mu\nu} \tag{2.8.16}$$

where $M = M(J_1, J_2)$, $N = N(J_1, J_2)$ are arbitrary differentiable functions of invariants of the electromagnetic field

$$J_1 = -\frac{1}{4}F_{\mu\nu}F^{\mu\nu} = \vec{E}^2 - \vec{B}^2, \qquad J_2 = \frac{1}{4}\widetilde{F}_{\mu\nu}F^{\mu\nu} = \vec{B} \cdot \vec{E}. \quad (2.8.17)$$

Proof. Since Equations (2.8.15) do not contain field derivatives, then to prove the theorem it is sufficient to find the functions $\phi_{\mu\nu}$ which ensure the Poincare invariance of Equations (2.8.15). The conditions of Poincare invariance have the form

$$P_{\rho} \left[H_{\mu\nu} - \phi_{\mu\nu}(F) \right] \Big|_{H_{\mu\nu} = \phi_{\mu\nu}(F)} = 0$$
 (2.8.18)

$$J_{\alpha\beta} \left[H_{\mu\nu} - \phi_{\mu\nu}(F) \right] \bigg|_{H_{\mu\nu} = \phi_{\mu\nu}(F)} = 0$$
 (2.8.19)

Using expressions (2.8.9), we rewrite (2.8.18) and (2.8.19) as follows:

$$(S_{\alpha\beta}\psi)^{l}\frac{\partial}{\partial\psi^{l}}\left[H_{\mu\nu}-\phi_{\mu\nu}(F)\right]\Big|_{H_{\mu\nu}=\phi_{\mu\nu}(F)}=0 \qquad (2.8.20)$$

The general solution of the determined Equations (2.8.20) is given by (2.8.16). The theorem is proved.

In terms of field strengths \vec{D} , \vec{B} , \vec{E} , and \vec{H} , the constitutive Equations (2.8.16) take the form

$$\vec{D} = M\vec{E} + N\vec{B}, \qquad \vec{H} = M\vec{B} - N\vec{E}$$
 (2.8.21)

Note that if $M = \alpha$, $N = \beta$, $(\alpha.\beta \text{ constants})$ then constraints (2.8.21) together with (2.8.1) lead to Maxwell's equations in vacuum. If we take

$$m = L^{-1}, \qquad N = (\vec{B} \cdot \vec{E})L^{-1},$$

$$L = \left[1 + (\vec{B}^2 - \vec{E}^2) - (\vec{B} \cdot \vec{E})^2\right]^{1/2} \qquad (2.8.22)$$

then Equations (2.8.1) together with (2.8.22) coincide with the nonlinear equations for the electromagnetic field suggested by Born and cited in the literature as Born-Infeld equations.

Let us present one more example of constitutive Equations (2.8.21). Letting $M = \epsilon$, $N = -\mu \vec{B} \cdot \vec{E}$ (ϵ, μ constants) one can resolve relations (2.8.21) with respect to \vec{D} and \vec{B} to obtain the following example of Poincare invariant constitutive equations:

$$\vec{D} = \epsilon \left\{ 1 + \frac{\mu^2 (\vec{E} \cdot \vec{H})^2}{\epsilon^2 (\epsilon + \mu \vec{E}^2)} \right\} \vec{E} - \frac{\mu (\vec{E} \cdot \vec{H})}{\epsilon (\epsilon + \mu \vec{E}^2)} \vec{H},$$

$$\vec{B} = \frac{\vec{H}}{\epsilon} - \frac{\mu (EH)}{\epsilon (\epsilon + \mu \vec{E}^2)} \vec{E}.$$
(2.8.23)

Consider constitutive equations of the form

$$\vec{D} = \epsilon(\vec{E}, \vec{H})\vec{E}, \qquad \vec{B} = \mu(\vec{E}, \vec{H})\vec{H}.$$
 (2.8.24)

Such equations are widely used in describing the electromagnetic field in various real media. The following statements are consequences of Theorem 2.8.3.

Consequence 2.8.1. The system of Equations (2.8.1) and (2.8.24) is Poincare invariant if and only if

$$\epsilon(\vec{E}, \vec{H}) \cdot \mu(\vec{E}, \vec{H}) = 1.$$
 (2.8.25)

Consequence 2.8.2. If $\vec{B} = \vec{\varphi}(\vec{H})$, $\vec{D} = \vec{f}(\vec{E}, \vec{H})$ then the requirement of Poincare invariance of Equations (2.8.1) together with these constraints results in a linear dependence of \vec{D} and \vec{B} on \vec{E} and \vec{H} , that is

$$\vec{D} = \epsilon \vec{E}, \qquad \vec{B} = \frac{1}{\epsilon} \vec{H}.$$
 (2.8.26)

Now we shall try to find which of the constitutive Equations (2.8.21) are conformally invariant. The answer is given by the following theorem.

Theorem 2.8.4 [107]. The system of Equations (2.8.1) and (2.8.21) is invariant under the conformal group C(1,3) if and only if

$$M = M\left(\frac{J_1}{J_2}\right), \quad N = N\left(\frac{J_1}{J_2}\right)$$
 (2.8.27)

with arbitrary differentiable functions M and N depending on the ratio of invariants J_1 and J_2 determined in (2.8.17).

Proof. One can obtain the proof analogously to that of Theorem 2.8.3. So, it is easy to establish that the requirement of scale invariance restricts the form of constitutive Equations (2.8.21) or (2.8.16) as follows:

$$H_{\mu\nu} = M\left(\frac{J_1}{J_2}\right)F_{\mu\nu} + N\left(\frac{J_1}{J_2}\right)\widetilde{F}_{\mu\nu}$$
(2.8.28)

The next requirement, the invariance of (2.8.16) under the pure conformal transformations, doesn't impose any additional restrictions on (2.8.28). Therefore, the theorem is proved.

Letting

-

$$M=\mu\frac{\vec{E}^2-\vec{B}^2}{\vec{E}\cdot\vec{B}},\qquad N=0$$

we get an example of conformally invariant constitutive equations in the resolved form

$$\vec{D} = \left(\frac{\mu \vec{H}^2}{\mu \vec{E}^2 - \vec{E} \cdot \vec{H}}\right)^{1/2} \vec{E}, \qquad \vec{B} = \left(\frac{\mu \vec{E}^2 - \vec{E} \cdot \vec{H}}{\mu \vec{H}^2}\right)^{1/2} \vec{H}.$$
 (2.8.29)

Consequence 2.8.3. The nonlinear Born-Infeld Equations (2.8.1) and (2.8.22) are not conformally invariant.

Now we turn to equations of electrodynamics for a vector-potential. Consider a nonlinear system

$$\Box A_{\mu} - \partial_{\mu} \partial^{\nu} A_{\nu} = A_{\nu} F(A^{\nu} A_{\nu}), \qquad (2.8.30)$$

Theorem 2.8.5. The system of PDEs (2.8.30) is invariant under the conformal group C(1,3) if and only if

$$F(A_{\nu}A^{\nu}) = \lambda A_{\nu}A^{\nu}, \qquad \lambda = \text{const.}$$
(2.8.31)

Proof. Equations (2.8.30) under F = 0 are conformally invariant, with generators of AC(1,3) determined in (2.3.3) and Table 2.3.1. In Lie's approach these operators can be written as follows:

$$P_{0} = i\partial_{0}, \qquad P_{a} = -i\partial_{a},$$

$$J_{\mu\nu} = x_{\mu}P_{\nu} - x_{\nu}P_{\mu} + A_{\mu}\mathcal{P}_{\nu} - A_{\nu}\mathcal{P}_{\mu},$$

$$D = x^{\nu}P_{\nu} - A^{\nu}\mathcal{P}_{\nu},$$

$$K_{\mu} = 2x_{\mu}D - x^{2}P_{\mu} + 2x^{\nu}(A_{\mu}\mathcal{P}_{\nu} - A_{\nu}\mathcal{P}_{\mu}),$$

$$(\mathcal{P}_{0} = i\partial/\partial A_{0}, \qquad \mathcal{P}_{a} = -i\partial/\partial A_{a}).$$

$$(2.8.32)$$

The relativistic invariance of Equations (2.8.30) with arbitrary function F is obvious. So the next step is to clear up what functions F do not break down scale invariance of the free Equations (2.8.30). The scale (dilatation) transformation

$$x'_{m}u = e^{\theta}x_{\mu}, \qquad A'_{\mu}(x') = e^{-\theta}A_{\mu}(x), \qquad \theta = \text{const}$$
 (2.8.33)

generated by operator D from (2.8.32), transforms system (2.8.31) as follows:

$$e^{-3\theta} (\Box A_{\mu} - \partial_{\mu} \partial^{\nu} A_{\nu}) = e^{-\theta} F \left(e^{-2\theta} A_{\nu} A^{\nu} \right) A_{\nu}$$

whence, if we require invariance, we get the law of transformation of the function F:

$$F(A_{\nu}A^{\nu}) = e^{2\theta}F\left(e^{-2\theta}A_{\nu}A^{\nu}\right)$$
(2.8.34)

Differentiation of (2.8.34) with respect to θ followed by setting $\theta = 0$ gives rise to the determining equation

$$F - u \frac{\partial F}{\partial u} = 0, \qquad u \equiv A_{\nu} A^{\nu},$$

the general solution of which is given in (2.8.31).

Using infinitesimal conformal transformations

$$\begin{aligned} x'_{\mu} &= (1+2cx)x_{\mu} - c_{\mu}x^{\nu}x_{\nu}, \\ A'_{\mu}(x') &= \left[(1-2cx)g_{\mu\nu} + 2(x_{\mu}c_{\nu} - x_{\nu}c_{\mu}) \right] A^{\nu}(x) \end{aligned}$$

it is not difficult to complete the proof.

Let us present some more statements which can be proved in much the same way as by means of Lie's algorithm.

Theorem 2.8.6. The system of coupled equations

$$(\partial_{\mu} - eA_{\mu})A^{\mu} = 0,$$

$$\Box A_{\mu} - \partial_{\mu}\partial^{\nu}A_{\nu} = 0,$$
(2.8.35)

where e is a constant (electron charge), is invariant under AC(1,3) with infinitesimal generators P_{μ} , $J_{\mu\nu}$, and D given by (2.8.32), and

$$\widetilde{K}_{\mu} = K_{\mu} + \frac{2}{e} \mathcal{P}_{\mu} \equiv 2x_{\mu}D - x^{2}P_{\mu} + 2x^{\nu}(A_{\mu}\mathcal{P}_{\nu} - A_{\nu}\mathcal{P}_{\mu}) + \frac{2}{e}\mathcal{P}_{\mu}.$$
 (2.8.36)

It is to be pointed out that generators \tilde{K}_{μ} (2.8.36) are non-analytic in the parameter of the nonlinearity e. One can make sure that under e = 0, system
(2.8.35) is not conformally invariant, although the first equation $\partial_{\mu}A_{\mu} = 0$, as well as the second one $\Box A_{\mu} - \partial_{\nu}\partial^{\nu}A_{\nu} = 0$ are separately conformally invariant. The point is that these equations have different conformal degrees: k = 3 for the first, and k = 1 for the second (see Table 2.3.1).

Another peculiarity of generators (2.8.36) is the impossibility of representing these operators as matrix-differential ones of type (7).

System (2.8.35) is a rather rare example of what is essentially an extension of the symmetry group of a linear system of PDEs by means of a nonlinear addend without derivatives.

Final transformations generated by operators (2.8.36) have the form (2.3.2) and

$$A'_{\mu}(x') = \left[g_{\mu\nu}\sigma + 2(x_{\mu}c_{\nu} - x_{\nu}c_{\mu} + 2cxc_{\mu}x_{\nu} - x^{2}c_{\mu}c_{\nu} - c^{2}x_{\mu}x_{\nu})\right]A^{\nu}(x) + \frac{2}{e}\left[c_{\mu}(1 - 2cx) + x_{\mu}c^{2}\right]$$
(2.8.37)

(compare with (2.3.37)). Transformations (2.8.37), just as expected, are non-analytic in the parameter e.

In conclusion we show how to construct equations like Maxwell's equations for the tensor $\mathcal{F}_{\mu\nu}$ with arbitrary conformal degree k.

From the standard tensor $F_{\mu\nu}$

$$F^{0a} = E_a, \qquad F^{ab} = \epsilon_{abc} H_c \tag{2.8.38}$$

with conformal degree equal to 2 (see Table 2.3.1) one can construct the new tensor

$$\mathcal{F}^{\mu\nu} = \left(I_1^{\alpha_1} I_2^{\beta_1}\right) F^{\mu\nu}, \qquad (2.8.39)$$

where

$$I_{1} = -\frac{1}{2}F_{\mu\nu}F^{\mu\nu} = \vec{E}^{2} - \vec{H}^{2}$$

$$I_{2} = \frac{1}{8}\epsilon_{\mu\nu\rho\sigma}F^{\mu\nu}F^{\rho\sigma} \equiv \frac{1}{4}F^{\mu\nu}\widetilde{F}_{\mu\nu} = \vec{E}\cdot\vec{H}, \qquad (2.8.40)$$

 α_1 , β_1 are arbitrary constants, $\alpha_1 + \beta_1 = (k-2)/4$. It can be easily confirmed that tensor (2.8.39) possesses conformal degree k.

The inverse relation

$$F^{\mu\nu} = (J_1^{\alpha} J_2^{\beta}) \mathcal{F}^{\mu\nu}$$
 (2.8.41)

where

$$J_1 = -\frac{1}{2} \mathcal{F}_{\mu\nu} \mathcal{F}^{\mu\nu}, \qquad J_2 = \frac{1}{8} \epsilon_{\mu\nu\rho\sigma} \mathcal{F}^{\mu\nu} \widetilde{\mathcal{F}}^{\rho\sigma}, \qquad \alpha + \beta = \frac{2-k}{2k}, \quad (2.8.42)$$

expresses the tensor $F^{\mu\nu}$ in terms of the tensor $\mathcal{F}^{\mu\nu}$.

Inserting (2.8.41) into the Maxwell's equations we get

$$\begin{split} \partial_{\mu}F^{\mu\nu} &= \partial_{\mu}\left[(J_{1}^{\alpha}J_{2}^{\beta})\mathcal{F}^{\mu\nu}\right] = 0,\\ \partial_{\mu}\widetilde{F}^{\mu\nu} &= \partial_{\mu}\left[(J_{1}^{\alpha}J_{2}^{\beta})\widetilde{\mathcal{F}}^{\mu\nu}\right] = 0. \end{split}$$

These equations can be rewritten in the form

$$D_{\mu}\mathcal{F}^{\mu\nu} = 0, \qquad D_{\mu}\widetilde{\mathcal{F}}^{\mu\nu} = 0 \qquad (2.8.43)$$

with

$$D_{\mu} = \partial_{\mu} + \partial_{\mu} \ln \left(J_1^{\alpha} J_2^{\beta} \right), \qquad \alpha + \beta = \frac{2-k}{2k.}$$
(2.8.44)

So we obtain the system of nonlinear PDEs (2.8.43) which is invariant under AC(1,3) with arbitrary conformal degree k (see also Table 2.3.1, Equations N9).

2.9. Solutions of nonlinear equations for vector fields

Let us consider the system of PDEs

$$\Box A_{\mu} - \partial_{\mu}\partial^{\nu}A_{\nu} + m^{2}A_{\mu} + \lambda A_{\mu}A^{\nu}A_{\nu} = 0, \qquad (2.9.1)$$

where $A_{\mu} = A_{\mu}(x)$; $x \in \mathbb{R}(1,3)$; $\mu, \nu = \overline{0,3}$; m, λ are arbitrary constants (to be definite we choose $\lambda > 0$). The maximal group (MG), in Lie's sense, of system (2.9.1) depending on m and λ is as follows: 1) $m = \lambda = 0$, MG = $\mathbb{C}(1,3) \otimes \mathbb{U}(1)$; 2) m = 0. $\lambda \neq 0$, MG = $\mathbb{C}(1,3)$; 3) $m \neq 0$, $\lambda = 0$, MG = $\mathbb{P}(1,3)$; 4) $m \neq 0$, $\lambda \neq 0$, MG = $\mathbb{P}(1,3)$.

To find exact solutions of system (2.9.1) we use the ansatze listed in the Table 2.1.4, preliminarily simplified.

Consider the ansatz

$$A_{\mu}(x) = a_{\mu}f(\omega_1), \qquad \omega_1 = bx, \qquad (2.9.2)$$

where f is a scalar real function, and a_{μ}, b_{μ} are arbitrary real constants satisfying relations (2.1.27).

Substitution of Equation (2.9.2) into (2.9.1) gives rise to ODEs for the function $f = f(\omega_1)$

$$\ddot{f} - m^2 + \lambda f^3 = 0.$$
 (2.9.3)

Solutions of Equation (2.9.3) are expressed by means of Jacobi elliptic functions (see Appendix 1):

$$f(\omega_1) = m \sqrt{\frac{2}{\lambda(2-k^2)}} \operatorname{dn}\left(\frac{m}{\sqrt{2-k^2}}\omega_1, k\right), \qquad 0 \le k < \sqrt{2}.$$
 (2.9.4)

Under k = 0 and k = 1 the elliptic function (2.9.4) degenerates:

$$f(\omega_1) = \frac{m}{\sqrt{\lambda}}, \qquad k = 0; \qquad (2.9.5)$$

$$f(\omega_1) = \sqrt{\frac{2}{\lambda}} \frac{m}{\operatorname{ch}(m\omega_1)}, \qquad k = 1.$$
 (2.9.6)

Consider the ansatz

$$A_{\mu}(x) = d_{\mu}f(\omega_2), \qquad \omega_2 = ax \tag{2.9.7}$$

Where the arbitrary real constants d_{μ} and $a_{\mu}, b_{\mu}, c_{\mu}$ (to be introduced soon) satisfy relations (2.1.27). After substituting (2.9.7) into (2.9.1) we get the ODE for the function

$$\ddot{f} - m^2 - \lambda f^3 = 0. (2.9.8)$$

which has a solution

$$f(\omega_2) = m \sqrt{\frac{2}{\lambda(2-k^2)}} \operatorname{cs}\left(\frac{m}{\sqrt{2-k^2}}\omega_2, k\right), \qquad 0 < \sqrt{2}.$$
 (2.9.9)

Under k = 0 and k = 1 the elliptic function (2.9.9) degenerates:

$$f(\omega_1) = \frac{m}{\sqrt{\lambda}} \operatorname{cotan}\left(\frac{m}{\sqrt{2}}\omega_2\right), \qquad k = 0;$$
 (2.9.10)

$$f(\omega_2) = \sqrt{\frac{2}{\lambda}} \frac{m}{\operatorname{sh}(m\omega_2)}, \qquad k = 1.$$
(2.9.11)

Consider another ansatz

$$A_{\mu}(x) = a_{\mu}f(\omega_3), \qquad \omega_3 = dx.$$
 (2.9.12)

It reduces system (2.9.1) to the following ODE for the function $f = f(\omega_3)$

$$\ddot{f} + m^2 f - \lambda f^3 = 0. (2.9.13)$$

Equation (2.9.13) has the solution

$$f(\omega_3) = mk \sqrt{\frac{2}{\lambda(1+k^2)}} \operatorname{sn}\left(\frac{m}{\sqrt{1+k^2}}\omega_3, k\right), \qquad (2.9.14)$$

Under k = 0 and k = 1 the elliptic function (2.9.14) degenerates:

$$f(\omega_3)=0, \qquad k=0;$$

Chapter 2. Systems of Poincare-invariant Nonlinear PDEs

$$f(\omega_3) = \frac{m}{\sqrt{\lambda}} \operatorname{th}\left(\frac{m}{\sqrt{2}}\omega_3\right), \qquad k = 1.$$
 (2.9.15)

Solutions (2.9.6), (2.9.11), and (2.9.15) present themselves as solitary waves. Solution (2.9.15) is known as a *kink* (see, for example, [170]).

Consider the ansatz

$$A_{\mu}(x) = a_{\mu}f(\omega_4) + (d_{\mu} + b_{\mu})g(\omega_4), \qquad \omega_4 = cx, \qquad (2.9.16)$$

where f and g are scalar real functions. Substituting (2.9.16) into (2.9.1) gives rise to the following system of ODEs:

$$\ddot{f} - m^2 f + \lambda f^3 = 0.$$

 $\ddot{g} - m^2 g + \lambda f^2 g = 0.$
(2.9.17)

The first equation from (2.9.17) coincides with (2.9.3) and therefore one can use expression (2.9.4) for the function $f = f(\omega_4)$. Inserting this result into the second equation of system (2.9.17) we get for the function $g(\omega_4)$ the Lame equation, which has Lame functions [24] as solutions.

Consider another ansatz

$$A_{\mu}(x) = (a_{\mu}bx - b_{\mu}ax)f(\omega_{5}) + (a_{\mu}ax + b_{\mu}bx)g(\omega_{5}),$$

$$\omega_{5} = \left[(ax)^{2} + (bx)^{2}\right]^{1/2}$$
(2.9.18)

After substituting (2.9.18) into (2.9.1) we get

$$\omega \ddot{f} + 3\dot{f} + \lambda \omega^2 f^3 - m^3 \omega f = 0, \qquad g = 0.$$
 (2.9.19)

Under m = 0 Equation (2.9.19) has a solution

$$f(\omega_5) = \frac{1}{2} \sqrt{\frac{3}{\lambda w_5}} \tag{2.9.20}$$

Let us present solutions of the massless Equations (2.9.3), (2.9.8). For equation $\ddot{f} + \lambda f^3 = 0$ we have

$$f(\omega) = \frac{1}{\sqrt{2}} \operatorname{cn} \left(\omega, \frac{1}{\sqrt{2}} \right),$$

$$f(\omega) = \sqrt{\frac{2}{\lambda}} \operatorname{dn} \left(\omega, \sqrt{2} \right).$$
(2.9.21)

For equation $\ddot{f} - \lambda f^3 = 0$ we obtain solutions

$$f(\omega) = \sqrt{\frac{2}{\lambda}} \frac{1}{\omega},$$

$$f(\omega) = \frac{1}{\sqrt{\lambda}} \operatorname{nc} \left(\omega, \frac{1}{\sqrt{2}}\right),$$

$$f(\omega) = \sqrt{\frac{2}{\lambda}} \operatorname{dn} \left(\omega, \sqrt{2}\right).$$

(2.9.22)

Under m = 0 one can look for solutions of system (2.9.1) by means of the conformally invariant ansatz (2.3.53). Substituting (2.3.53) into (2.9.1) with m = 0 gives rise to the system of ODEs for $B_{\mu} = B_{\mu}(\omega_6)$, $\omega_6 = (\beta x)/(x_{\nu}x^{\nu})$:

$$\beta^{2}\ddot{B}_{\mu} - \beta_{\mu}\beta^{\nu}\ddot{B}_{\nu} + \lambda B_{\mu}B^{\nu}B_{\nu} = 0.$$
 (2.9.23)

Letting $\beta_{\mu} = b_{\mu}$, $B_{\mu} = a_{\mu}f(\omega_6)$ we reduce (2.9.23) to the equation $\ddot{f} + \lambda f^3 = 0$, and letting $\beta_{\mu} = d_{\mu}$, $B_{\mu} = a_{\mu}f(\omega_6)$ we reduce it to $\ddot{f} - \lambda f^3 = 0$. Therefore in these cases one can make use of solutions given by (2.9.21) and (2.9.22). Letting $\beta_{\nu} = c_{\nu}$, $B_{\mu} = a_{\mu}f(\omega_6) + (d_{\mu} + b_{\mu})g(\omega_6)$ we get (2.9.17) under m = 0

$$\ddot{f} + \lambda f^3 = 0,$$

$$\ddot{g} + \lambda f^2 g = 0.$$
(2.9.24)

A simple solution of system (2.9.24) is obtained under $\lambda < 0$:

$$f(\omega) = \sqrt{\frac{2}{|\lambda|}} \frac{1}{\omega}, \qquad g(\omega) = \alpha_1 \omega^2 + \frac{\alpha_2}{\omega}$$
(2.9.25)

where α_1, α_2 are arbitrary constants, and $\omega = \omega_6$

Using ansatz (2.3.53) and the solutions of system (2.9.23) described above, we obtain conformally invariant solutions to the massless system (2.9.1). Let us write down an example of such solutions:

$$A_{\mu}(x) = \frac{1}{\sqrt{\lambda}} \left(\frac{a_{\mu}}{x_{\nu} x^{\nu}} - 2x_{\mu} \frac{ax}{(x^{\nu} x_{\nu})^2} \right) \operatorname{cn} \left(\frac{bx}{x_{\nu} x^{\nu}}, \frac{1}{\sqrt{2}} \right).$$
(2.9.26)

Another way of obtaining new solutions of the massless system (2.9.1) is achieved by means of the formula of generating solutions (2.3.38).

In conclusion let us note that all of the solutions of system (2.9.1) described above are non-analytic in the coupling constant λ . Such solutions, as is known from [170], lead to many interesting consequences in quantum field theory.

2.10. Some exact solutions of SU(2) Yang-Mills field theory

Recently, non-Abelian gauge field theories took a central role in describing the interactions of elementary particles. In particular, many works are devoted to the simplest non-Abelian gauge field theory SU(2), Yang-Mills (YM) gauge theory. An excellent review of classical solutions of SU(2) YM field equations is given in [4].

In the present paragraph new multiparameter exact solutions of SU(2) YM equations are obtained. These solutions are real, non-Abelian, and C(1,3)-ungenerative.

1. First of all let us give a brief review of the foundations of the SU(2) YM gauge theory. The field equations have the form

$$\partial_{\mu}\partial^{\mu}\vec{Y}_{\nu} - \partial_{\nu}\partial^{\mu}\vec{Y}_{\mu} + e\left[(\partial^{\mu}\vec{Y}_{\mu}) \times \vec{Y}_{\nu} - 2(\partial^{\mu}\vec{Y}_{\nu}) \times \vec{Y}_{\mu} + (\partial^{\nu}\vec{Y}_{\mu}) \times \vec{Y}_{\mu}\right] + e^{2}\vec{Y}^{\mu} \times (\vec{Y}_{\mu} \times \vec{Y}_{\nu}) = 0, \qquad (2.10.1)$$

where $\vec{Y}_{\mu} = \{Y_{\mu}^{1}, Y_{\mu}^{1}, Y_{\mu}^{1}\}$ is the YM potential; $\mu, \nu = \overline{0, 3}$; *e* is a constant. Equations (2.10.1) follow from the Lagrangian

$$\mathcal{L} = -\frac{1}{4} G^a_{\mu\nu} G^{a\mu\nu}, \qquad (2.10.2)$$

where $G^{a}_{\mu\nu}$ is the field strength tensor

$$G^a_{\mu\nu} = \partial_\mu Y^a_\nu - \partial_\nu Y^a_\mu + e\epsilon_{abc} Y^b_\mu Y^c_\nu.$$
(2.10.3)

By means of the tensor (2.10.3), YM equations (2.10.1) are written as follows:

$$\partial^{\nu}G^{a}_{\mu\nu} = e\epsilon_{abc}G^{b}_{\mu\nu}Y^{c\nu}, \qquad (2.10.4)$$

The YM equations (2.10.1) possess rich symmetry. As shown in [182] their maximal Lie symmetry group is the group $SU(2)\otimes C(1,3)$. Basis elements of AC(1,3) are written in (2.3.3) and also

$$k = 1, \qquad S_{\mu\nu} = \widehat{S}_{\mu\nu} \otimes I_3, \qquad (2.10.5)$$

where $\hat{S}_{\mu\nu}$ are 4×4 matrices realizing irreducible representation D(1/2,1/2) of AO(1,3); I_3 is the unit 3×3 matrix; and the notation \otimes means direct matrix multiplication.

Generators of the SU(2) gauge group have the form [4,182]

$$X = \left[\epsilon_{abc}Y^b_{\mu}\theta^c(x) + \frac{1}{e}\partial_{\mu}\theta^a(x)\right] \frac{\partial}{\partial Y^a_{\mu}}$$
(2.10:6)

with arbitrary differentiable functions $\theta^a(x)$. It follows from (2.10.5) that conformal transformations of the YM potential have the form (compare with (2.3.37)) [96]

$$Y^{a}_{\mu}(x) = [\sigma(x,c)\delta^{\nu}_{\mu} + 2(x_{\mu}c^{\nu} - x^{\nu}c_{\mu} + 2cxc_{\mu}x^{\nu} - x^{2}c_{\mu}c^{\nu} - c^{2}x_{\mu}x^{\nu})]Y^{a}_{\nu}(x).$$
(2.10.7)

The corresponding formula of generating solutions is [96]

$$Y^{a}_{\mu I\!I}(x) = [\sigma^{-1}(x,c)\delta^{\nu}_{\mu} + 2\sigma^{-2}(x,c)(c_{\mu}x^{\nu} - x_{\mu}c^{\nu} + (2.10.8)) + 2cxx_{\mu}c^{\nu} - c^{2}x_{\mu}x^{\nu} - x^{2}c_{\mu}c^{\nu}]Y^{a}_{\nu I}(x').$$

The gauge transformations generated by operators (2.10.6) have the form [4]

$$Y_{\mu}^{\prime a}(x) = \cos\theta Y_{\mu}^{a} + \sin\theta \epsilon_{abc} Y_{\mu}^{b} n^{c} + 2\sin^{2}\frac{\theta}{2} n^{a} (n^{b}Y_{\mu}^{b}) +$$
(2.10.9)

$$+\frac{1}{e}\left[\frac{1}{2}n^{a}\partial_{\mu}\theta+\frac{1}{2}\sin\theta\,\partial_{\mu}n^{a}+\sin^{2}\frac{\theta}{2}\,\epsilon_{abc}(\partial_{\mu}n^{b})n^{c}\right],$$

where $n^a = n^a(x)$ is a unit vector defined by

$$\theta^a(x) \equiv n^a(x)\theta(x) \tag{2.10.10}$$

It will be noted that the tensor of the YM field (2.10.3) is not invariant under gauge transformations unlike the tensor of the electromagnetic field. This is an important difference between Abelian and non-Abelian gauge theories. The tensor changes under gauge transformations (2.10.9) as follows:

$$G^a_{\mu\nu} \to G^{\prime a}_{\mu\nu} = G^a_{\mu\nu} \cos\theta + \left(\sin\theta\epsilon_{abc}n^c + 2\sin^2\frac{\theta}{2}n^a n^b\right)G^b_{\mu\nu}.$$
 (2.10.11)

By analogy with classical electrodynamics, "electric" and "magnetic" YM fields

$$E_k^a = G_{0k}^a, \qquad B_k^a = -\frac{1}{2}\epsilon_{kij}G_{ij}^a$$
 (2.10.12)

are introduced.

The energy-momentum tensor is defined by

$$\theta_{\mu\nu} = -G^a_{\mu\lambda}G^{a\lambda}_{\nu} + \frac{1}{4}g_{\mu\nu}G^a_{\alpha\beta}G^{a\alpha\beta}$$
(2.10.13)

The components of $\theta_{\mu\nu}$ are

$$\theta_{00} = \frac{1}{2} (E_k^a E_k^a + B_k^a B_k^a) = \sum_i \theta_{ii},$$

$$\theta_{0j} = -\epsilon_{jmn} E_m^a B_n^a,$$

$$\theta_{ij} = -E_i^a E_j^a - B_i^a B_j^a + \delta_{ij} \frac{1}{2} (E_k^a E_k^a + B_k^a B_k^a).$$

(2.10.14)

One can confirm that the tensor $\theta_{\mu\nu}$ is gauge invariant.

It is appropriate to note some essential differences between the YM equations (2.10.1) and the classical electrodynamics equations

$$\partial^{\mu}F_{\mu\nu} \equiv \partial^{\mu}(\partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu}) = \Box A_{\nu} - \partial_{\nu}\partial^{\mu}A_{\mu} = 0.$$

First of all, YM equations are essentially nonlinear. Secondly, they contain both the field tensor $G^a_{\mu\nu}$ and the potential Y^a_{μ} (see writing (2.10.4)). It seems that the YM potential plays a more basic role than the potential in an Abelian gauge theory. In electromagnetism one can work exclusively with the field strengths \vec{E} and \vec{B} . Things are not so simple when the gauge group is non-Abelian. Thirdly, in Abelian gauge theories the field strengths locally determine the gauge potential up to an arbitrary gauge transformation. The same is not true for non-Abelian gauge theories: two YM potentials that are gauge inequivalent can provide the same YM field strengths. This leads to an interesting problem, specifically, the determination of all possible gauge potentials that yield a given field-strength tensor (see [4] and literature cited therein).

We shall note one more peculiarity of SU(2) gauge field theory. If the potential Y^a_{μ} satisfies the so-called self-duality conditon

$$\widetilde{G}^a_{\mu\nu} \equiv \frac{1}{2} \epsilon_{\mu\nu\rho\sigma} G^{a\rho\sigma} = \pm i G^a_{\mu\nu}, \qquad (2.10.15)$$

or equivalently, in terms of field strengths \vec{E} and \vec{B} (2.10.12),

$$\pm iE_n^a = B_n^a \tag{2.10.16}$$

then it automatically satisfies the equations of motion (2.10.1). Note that the factor i in this definition of self-duality is unavoidable because we are working in Minkowski space where $\widetilde{\tilde{G}}^a_{\mu\nu} = -G^a_{\mu\nu}$. Clearly, in Minkowski space any self-dual field configuration contains complex fields. (In Euclidean space E^4 the factor i is absent and self-dual fields can be real.)

One can try to make use of the property of self-dual fields to satisfy the equations of motion (2.10.1) by searching for solutions of the self-dual equations (2.10.15), which are first order, rather than trying to solve the second-order equations of motion. As is shown in [182], the self-dual equations (2.10.15) possess the same symmetry as the field equations (2.10.1), that is their maximal Lie group of invariance is $SU(2) \otimes C(1,3)$. Self-duality is a very special property which most solutions do not have. Nevertheless, interesting solutions can be found in this way.

Any self-dual solution in Minkowski space has a vanishing energy-momentum tensor $\theta_{\mu\nu}$ (2.10.13). Indeed, representing it as

$$\theta_{\mu\nu} = -\frac{1}{4} \left(G^a_{\mu\alpha} + i \widetilde{G}^a_{\mu\alpha} \right) \left(G^{a\alpha}_{\nu} - i \widetilde{G}^{a\alpha}_{\nu} \right)$$

we get $\theta_{\mu\nu} = 0$ for any field configuration which satisfies self-dual conditions (2.10.15) or (2.10.16).

2. By now the interest in classical solutions of the YM field equations has become so widespread that many workers are involved in the search for new solutions. A largely complete review of the known classical solutions of SU(2) gauge theory until 1979 is given in [4].

A well-known ansatz is the 't Hooft-Corrigan-Fairlie-Wilczek ansatz

$$eY_0^a = \pm i\partial_a \ln \varphi,$$

$$eY_j^a = (\epsilon_{jak}\partial_k \pm \delta_{aj}\partial_0) \ln \varphi,$$
(2.10.17)

where $\varphi = \varphi(x)$ is a scalar function. Ansatz (2.10.17) can also be written in the form

$$Y^a_\mu = \eta_{a\mu\nu} \partial^\nu \ln \varphi \tag{2.10.18}$$

with 't Hooft tensor

$$\eta_{a\mu\nu} = \epsilon_{0a\mu\nu} \mp i g_{a\mu} g_{0\nu} \pm i g_{a\nu} g_{0\mu}. \tag{2.10.19}$$

Ansatz (2.10.17) reduces system (2.10.1) to the equation

$$\partial_{\mu}\Box\varphi - (3/\varphi)(\partial_{\mu}\varphi)\Box\varphi \equiv \varphi^{3}\partial_{\mu}(\varphi^{-3}\Box\varphi) = 0.$$
(2.10.20)

Equation (2.10.20) yields

$$\Box \varphi + \lambda \varphi^3 = 0, \qquad (2.10.21)$$

where λ is an arbitrary constant. Solutions of Equation (2.10.21) have been considered in Paragraph 5.1 (see also Appendix 1), and in (2.10.17) these lead automatically to explicit solutions of the YM equations (2.10.1).

As seen from (2.10.17) the potentials Y^a_{μ} are complex for real φ . This is an unfortunate property of the ansatz, as one would like to find real solutions of the SU(2) YM theory. But complex solutions are also interesting, and furthermore there exists the possibility that for a particular solution φ the SU(2) potential (2.10.17) can be made real by a suitable complex SU(2) gauge transformation. For arbitrary φ this is not possible.

Let us write down several quantities of interest that follow from ansatz (2.10.17). The YM field strengths are

$$eE_{n}^{a} \equiv eG_{0n}^{a} = \epsilon_{nam} \left(\frac{1}{\varphi}\partial_{0}\partial_{m}\varphi - \frac{2}{\varphi^{2}}\partial_{0}\varphi\partial_{m}\varphi\right) \pm i\delta_{na} \left[\frac{1}{\varphi}\partial_{0}^{2}\varphi - \frac{1}{\varphi^{2}}\left(\partial_{0}\varphi\partial_{0}\varphi + \partial_{m}\varphi\partial_{m}\varphi\right)\right] \mp i \left[\frac{1}{\varphi}\partial_{n}\partial_{a}\varphi - \frac{2}{\varphi^{2}}\partial_{n}\varphi\partial_{a}\varphi\right], \quad (2.10.22)$$
$$B_{n}^{a} \equiv -\frac{1}{2}e\epsilon_{nij}G_{ij}^{a} = \pm ieE_{n}^{a} + \delta_{an}\left(\frac{1}{\varphi}\right)\Box\varphi$$

The self-duality condition (2.10.16) implies $\Box \varphi = 0$. The field strengths E_n^a and B_n^a are in general complex. Remarkably, however, their squares are both real, and this means that the energy and Lagrangian densities obtained from ansatz (2.10.17) are real, even though the potential is complex.

It is easy to show that the energy-momentum tensor (2.10.13) following from ansatz (2.10.17) has the form

$$e^{2}\theta_{\mu\nu} = \frac{\Box\varphi}{\varphi} \left[\frac{4}{\varphi^{2}} \partial_{\mu}\varphi \partial_{\nu}\varphi - \frac{2}{\varphi} \partial_{\mu}\partial_{\nu}\varphi + g_{\mu\nu} \left(\frac{1}{2\varphi} \Box\varphi - \frac{1}{\varphi^{2}} \partial^{\alpha}\varphi \partial_{\alpha}\varphi \right) \right].$$
(2.10.23)

A self-dual solution has $\theta_{\mu\nu} = 0$ because $\Box \varphi = 0$.

The total energy is

$$\mathcal{E} = \int \theta_{00} d^3 x = -\frac{6\lambda}{e^2} \int \left[\frac{1}{2} (\partial_0 \varphi)^2 + \frac{1}{2} (\vec{\nabla} \varphi)^2 + \frac{1}{4} \lambda \varphi^4 \right] d^3 x, \qquad (2.10.24)$$

where we have neglected surface terms at infinity.

The Euclidean counterpart of ansatz (2.10.17)

$$eY_0^a = \mp \partial_a \ln \varphi,$$

$$eY_j^a = (\epsilon_{jan} \partial_n \pm i \delta_{aj} \partial_0) \ln \varphi,$$
(2.10.25)

is connected with such fundamental objects of non-Abelian gauge theories as merons and instantons. Below we consider these solutions of YM equations in Euclidean space.

As was shown in paragraph 2.4 (see formulae (2.4.26), (2.4.27)) the functions

$$\varphi(x) = \frac{1}{\sqrt{\lambda x_{\nu} x^{\nu}}},\tag{2.10.26}$$

$$\varphi(x) = \sqrt{\frac{8}{\lambda}} \frac{\alpha}{x_{\nu} x^{\nu} + \alpha^2}.$$
(2.10.27)

(α is an arbitrary constant) are solutions of Equation (2.10.21). In Euclidean space these solutions lead [61*], by means of ansatz (2.10.25), to one meron solution of deAlfaro-Fubini-Furlan [4*]

$$eY_0^a = \pm \frac{x_a}{x^2},$$

$$eY_j^a = -\epsilon_{jan} \frac{x_n}{x^2} \mp \delta_{aj} \frac{x_0}{x^2}$$
(2.10.28)

and to Belavin-Polyakov-Schwartz-Tyupkin instanton solution [1*]

$$eY_0^a = \mp \frac{2x_a}{x^2 + \alpha^2}, eY_j^a = -\epsilon_{jan} \frac{2x_n}{x^2} \pm \delta_{aj} \frac{2x_0}{x^2 + \alpha^2}$$
(2.10.29)

of the YM equations in Euclidean space, respectively. Here we would like to pay attention to the following point. Both solutions (2.10.28) and (2.10.29) follow from solutions (2.4.11) and (2.4.12) of the nonlinear Dirac-Gursey equation (2.1.5) by means of formulae (2.4.25) and (2.10.25). Therefore one can consider meron (2.10.28) and instanton (2.10.29) as objects generated by the spinor field ψ obeying the nonlinear Dirac-Gursey equation (2.1.5) [61*]. It is a principal point which can lead to a new and deeper understanding of YM field theory. The very names meron and instanton are connected with topological properties of the solutions.

In Euclidean gauge theory a useful four-vector is defined:

$$J_{\mu} = \frac{1}{4} \epsilon_{\mu\nu\rho\sigma} \left(Y^a_{\nu} \partial_{\rho} Y^a_{\sigma} + \frac{e}{3} \epsilon_{abc} Y^a_{\nu} Y^b_{\rho} Y^c_{\sigma} \right)$$
(2.10.30)

(let us remember that in Euclidean space one does not need to distinguish upper and lower indices). The identity

$$\partial_{\mu}J_{\mu} = \frac{1}{8}\tilde{G}^{a}_{\mu\nu}G^{a}_{\mu\nu}.$$
 (2.10.31)

can easily be confirmed.

The integral

$$q = \frac{e^2}{4\pi^2} \int d^4x \partial_\mu J_\mu \equiv \frac{e^2}{32\pi^2} \int d^4x \tilde{G}^a_{\mu\nu} G^a_{\mu\nu}$$
(2.10.32)

defines the topological index or charge of the Euclidean field configuration.

A meron is a localized, singular solution of the Euclidean gauge theory with one-half unit of topological charge. The name meron derives from a Greek word meaning part (of unit of topological charge). The topological charge of the meron is concentrated at the point where the solution is singular. This is to be contrasted with the instanton's nonsingular cloud of topological charge. The charge of the instanton is one unit. The name instanton is derived from its localization in E^4 which corresponds to finite duration as well as spacial extension in Minkowski space, i.e., "event." All instantons are self-dual and meron solutions are not self-dual. An instanton is also called a pseudoparticle.

Solutions of Equation (2.10.21) expressed by means of Jacobi elliptic functions lead to elliptic solutions of the YM equations. The elliptic solution depends on a continuous parameter k (see Appendix 1), and for k = 1 it reduces to the one-instanton solution, and for k = 0 it becomes the two-meron solution. Note that the two-meron solution is obtained by means of the ansatz (2.10.25) under

$$\varphi(x) = \left[\frac{(a_{\nu} - b_{\nu})(a_{\nu} - b_{\nu})}{\lambda(x - a)^2(x - b)^2}\right]^{1/2}$$
(2.10.33)

with arbitrary constants a_{ν}, b_{ν} .

This solution of Equation (2.10.21) follows from solution (2.10.26) by means of the formulae of generating solutions (2.3.2) and (2.3.34), and then letting $x_{\mu} \rightarrow x_{\mu} - a_{\mu}$ and taking $c_{\mu} = (b_{\mu} - a_{\mu})/(a - b)^2$ [61*].

The corresponding spinor field ψ which gives, via (2.4.25), the scalar field (2.10.33), can be easily obtained from (2.4.12) by means of formulae (2.3.2) and (2.3.27) and then letting $x_{\mu} \to x_{\mu} - a_{\mu}$ and taking $c_{\mu} = (b_{\mu} - a_{\mu})/(a - b)^2$.

Note that the same procedure of generating solutions applied to (2.10.27) results in another solution of Equation (2.10.21):

$$\varphi(x) = \sqrt{\frac{8(a-b)^2}{\lambda}} \frac{1}{(x-a)^2 + (x-b)^2}$$
(2.10.27*a*)

under $\alpha^2 = (a - b)^2$. For more detail see [61*].

3. Now we will construct real C(1,3)-ungenerative non-Abelian solutions (when $\vec{Y}_{\mu} \times \vec{Y}_{\nu} \neq 0$) of the YM equations (2.10.1). For this aim we use the conformally invariant ansatz (compare with (2.3.53))

$$Y_{\nu}^{a}(x) = (x_{\alpha}x^{\alpha})^{-1}\varphi_{\nu}^{a}(\omega) - \frac{2x_{\nu}x^{\sigma}\varphi_{\sigma}^{a}(\omega)}{(x_{\alpha}x^{\alpha})^{2}}, \qquad \omega = \frac{\beta x}{x^{\nu}x_{\nu}}, \qquad (2.10.34)$$

where β_{ν} are arbitrary constants.

Substitution of the ansatz (2.10.34) into (2.10.1) gives rise to the following system of ODEs

$$\beta^{2}\ddot{\vec{\varphi}}_{\nu} - \beta_{\nu}\beta^{\nu}\ddot{\vec{\varphi}}_{\mu} + e\left[\beta^{\mu}\left(\dot{\vec{\varphi}}\times\vec{\varphi}_{\nu} - 2\dot{\vec{\varphi}}_{\nu}\times\vec{\varphi}_{\mu}\right) + \beta_{\nu}\dot{\vec{\varphi}}_{\mu}\times\vec{\varphi}^{\mu}\right] + e^{2}\vec{\varphi}^{\mu}\times\left(\vec{\varphi}_{\mu}\times\vec{\varphi}_{\nu}\right) = 0, \qquad (2.10.35)$$

where dot means differentiation with respect to ω ; $\vec{\varphi} = \{\varphi_{\nu}^{a}\}$. We look for solutions of the system (2.10.35) in the form

$$\vec{\varphi}(\omega) = \alpha_{\nu} \vec{n} f(\omega) + \rho_{\nu} \vec{p} g(\omega), \qquad (2.10.36)$$

where f and g are scalar real functions to be defined; α_{ν}, ρ_{ν} are arbitrary constants satisfying relations

$$\alpha \rho = \alpha \beta = \rho \beta = 0, \qquad (2.10.37)$$

 \vec{n}^2, \vec{p} are mutually orthonormal constant vectors of isospin

$$\vec{n}^2 = \vec{p}^2 = 1, \qquad \vec{n} \cdot \vec{p} = 0.$$
 (2.10.38)

Inserting (2.10.36) into (2.10.35) we get coupled nonlinear ODE for two unknown functions f and g

$$\beta^2 \ddot{f} - e^2 \rho^2 g^2 f = 0,$$

$$\beta^2 \ddot{g} - e^2 \alpha^2 f^2 g = 0.$$
(2.10.39)

The system (2.10.39) has nontrivial solutions only if $\beta^2 \neq 0$. Further, there are 6 different cases.

Under $\rho^2 = 0$, $\alpha^2 = \beta^2 = -1$ the system (2.10.39) takes the form

$$\ddot{f} = 0,$$

 $\ddot{g} - e^2 f^2 g = 0.$
(2.10.40)

Under $\alpha^2 = 0$, $\rho^2 = \beta^2 = -1$ the system (2.10.39) takes the form

$$\ddot{g} = 0,$$

 $\ddot{f} - e^2 g^2 f = 0.$ (2.10.41)

Under $\rho^2 = 1$, $\alpha^2 = \beta^2 = -1$ and under $\alpha^2 = 1$, $\rho^2 = \beta^2 = -1$ the system (2.10.39) takes the form

$$f \pm e^2 g^2 f = 0,$$

 $\ddot{g} \mp e^2 f^2 g = 0$
(2.10.42)

respectively.

Under $\beta^2 = 1$, $\alpha^2 = \rho^2 = -1$ the system (2.10.39) takes the form

$$\ddot{f} + e^2 g^2 f = 0,$$

 $\ddot{g} + e^2 f^2 g = 0.$
(2.10.43)

And finally, under $\alpha^2 = \beta^2 = \rho^2 = -1$ the system (2.10.39) takes the form

$$\ddot{f} - e^2 g^2 f = 0,$$

$$\ddot{g} - e^2 f^2 g = 0.$$
(2.10.44)

Now we try to look for solutions of systems (2.10.40)-(2.10.44). From (2.10.40) it is easy to find the general solution for function f:

$$f(\omega) = c_0 \omega + c \tag{2.10.45}$$

 $(c_0, c \text{ are constants of integration})$ and after substituting it into the second equation of the system we get the linear ODE for function $g(\omega)$

$$\ddot{g} = e^2 (c_0 \omega + c)^2 g.$$
 (2.10.46)

Under $c_0 = 0$ the general solution of Equation (2.10.46) is given by

$$g(\omega) = \begin{cases} c_1 \operatorname{sh} ec\omega + c_2 \operatorname{sh} ec\omega, & c \neq 0; \\ c_1\omega + c_2 \end{cases}$$
(2.10.47)

 $(c_1, c_2 \text{ are arbitrary constants})$. Under $c_0 \neq 0$ Equation (2.10.46) is reduced by the substitution

$$\omega \to y = \omega + \frac{c}{c_0}$$

to the ODE

$$\frac{d^2g}{dy^2} = e^2 c_0^2 y^2 g,$$

whose solutions are expressed in terms of Bessel functions

$$g(y) = \sqrt{y} Z_{1/4} \left(\frac{i}{2} e c_0 y^2\right)$$

By interchanging $f \leftrightarrow g$ the system (2.10.41) is transformed to the system (2.10.40).

Equations (2.10.42) have the first integral

$$\dot{f}^2 + \dot{g}^2 = E^2 = \text{const},$$

which allows the reduction of (2.10.42) to one second-order ODE. Letting

$$g = g(f)$$

we find

$$\dot{g} = \frac{dg}{df}\dot{f} \equiv g'\dot{f}, \quad \ddot{g} = g''\dot{f}^2 + g'\ddot{f}, \quad \dot{f}^2 = E^2[1 + (g')^2]^{-1}.$$

Inserting these expressions into (2.10.42) we get the ODE

$$g'' \frac{E^2}{1+(g')^2} \mp e^2 g f(g'g+f) = 0, \quad g' \equiv \frac{dg}{df}.$$

Further consider the system (2.10.43) as well as the system (2.10.44). Letting f = g we get Equations (2.9.3), (2.9.8) under $m = 0, \lambda = e^2$, respectively. Therefore one can use solutions (2.9.21), (2.9.22), whence follows solutions of the system (2.10.43)

$$f = g = \frac{1}{e} \operatorname{cn} \left(\omega, \frac{1}{\sqrt{2}} \right),$$

$$f = g = \frac{\sqrt{2}}{e} \operatorname{dn} \left(\omega, \sqrt{2} \right)$$
(2.10.48)

and solutions of the system (2.10.44)

$$f = g = \frac{\sqrt{2}}{e} \frac{1}{\omega},$$

$$f = g = \frac{1}{e} \operatorname{nc} \left(\omega, \frac{1}{\sqrt{2}} \right),$$

$$f = g = \frac{\sqrt{2}}{e} \operatorname{nd} (\omega, \sqrt{2}).$$

(2.10.49)

Under g = 0 we find from (2.10.43), (2.10.44)

$$f = c_1 \omega + c_2, \tag{2.10.50}$$

where c_1, c_2 are arbitrary constants. Analogously, letting f = 0 one finds

$$g = c_1 \omega + c_2, \tag{2.10.51}$$

Equations (2.10.43), (2.10.44) have the first integral (see also [169])

$$\dot{f}^2 + \dot{g}^2 + \epsilon e^2 f^2 g^2 = E, \qquad \epsilon = \pm 1$$
 (2.10.52)

which allows the reduction of systems (2.10.43), (2.10.44) to one second-order ODE. Letting

$$f = \frac{1}{e}F, \qquad g = \frac{1}{e}G(F)$$
 (2.10.53)

we calculate

$$\dot{G} = \frac{dG}{dF}\dot{F} \equiv G'\dot{F},$$

$$\ddot{G} = G''\dot{F}^2 + G'\ddot{F},$$
(2.10.54)

From (2.10.52) - (2.10.54) we have

$$\dot{F}^2 = \frac{E - \epsilon F^2 G^2}{1 + (G')^2},$$
(2.10.55)

so

$$\int d\omega = \int \frac{dF}{\sqrt{(E - \epsilon F^2 G^2)/(1 + (G')^2)}}.$$
(2.10.56)

From (2.10.43), (2.10.44), (2.10.53) - (2.10.55) we get

$$(E - \epsilon F^2 G^2) G'' + \epsilon F G (F - G G') \left(1 + (G')^2 \right) = 0.$$
 (2.10.57)

Under $\epsilon = -1$ and E = 0, Equation (2.10.57) takes the form

$$F^{2}G^{2}G'' - (F^{2}G - FG^{2}G')\left(1 + (G')^{2}\right) = 0.$$

After the change of variables

$$G = Fx(\tau), \qquad \tau = \ln F$$

we obtain

$$x\frac{d^2x}{d\tau^2} - \left(\frac{dx}{d\tau}\right)^2 + x\left(x + \frac{dx}{d\tau}\right)^2 - 1 = 0.$$
 (2.10.58)

Taking in (2.10.58) $\frac{dx}{d\tau}$ as a function of x, i.e.,

$$\frac{dx}{d\tau} = y(x), \qquad \frac{d^2x}{d\tau^2} = y\frac{dy}{dx}$$

we obtain the first-order equation on y(x), namely the Abel equation of the second kind [130, 169]

$$xy\frac{dy}{dx} = -xy^3 + (1 - 3x^2)y^2 - 3x^3y + 1 - x^4.$$
 (2.10.59)

Substituting solutions of the system (2.10.39) into (2.10.36), (2.10.34) we find solutions of the YM equations (2.10.1). Let us write down a couple of such solutions following from (2.10.47), (2.10.49):

$$\vec{Y}_{\nu}(x) = \vec{n} \frac{c}{x^2} \left(\alpha_{\nu} - 2x_{\nu} \frac{\alpha x}{x^2} \right) + \frac{\vec{p}}{x^2} \left(c_1 \operatorname{ch} ec \frac{\beta x}{x^2} + c_2 \operatorname{sh} ec \frac{\beta x}{x^2} \right) \times \left(\rho_{\nu} - 2x_{\nu} \frac{\rho x}{x^2} \right), \qquad (2.10.60)$$

$$\rho^2 = 0, \qquad \alpha^2 = \beta^2 = -1, \qquad \alpha \rho = \alpha \beta = \rho \beta = 0;$$

$$\vec{Y}_{\nu}(x) = \frac{\sqrt{2}}{e(\alpha x)} \left[\vec{n} \left(\alpha_{\nu} - 2x_{\nu} \frac{\alpha x}{x^2} \right) + \vec{p} \left(\rho_{\nu} - 2x_{\nu} \frac{\rho x}{x^2} \right) \right], \qquad (2.10.61)$$
$$\alpha^2 = \beta^2 = \rho^2 = -1, \qquad \alpha \rho = \rho \beta = \alpha \beta = 0.$$

Replacing x_{μ} on $y_{\mu} = x_{\mu} + \delta_{\mu}$ (δ_{μ} are constants) we obtain from (2.10.60), (2.10.61) two families of C(1,3)-ungenerative solutions of YM system (2.10.1). Note, that solution (2.10.60) is analytic in constant *e* but the solution (2.10.61) is not.

It will be also noted that solutions of YM equations described in Point 2 can be generated by means of formulae (2.10.8), (2.3.2) and to obtain in such a way new families of solutions.

In conclusion we present the reduced system of ODEs which is obtained when using the ansatz

$$\vec{Y}_{\nu}(x) = (a_{\nu}bx - b_{\nu}ax)f(\omega)\vec{n} + c_{\mu}\vec{p}g(\omega) + d_{\mu}\vec{q}h(\omega), \qquad (2.10.62)$$

where $\omega = [(ax)^2 + (bx)^2]^{1/2}$; $a_{\nu}, b_{\nu}, c_{\nu}, d_{\nu}$ are arbitrary constants satisfying relations (2.1.27); f, g, h are real scalar functions; $\vec{n}, \vec{p}, \vec{q}$ are mutually orthonormal isospin vectors. Substitution of the ansatz (2.10.62) into (2.10.1) gives rise to the ODE

$$\begin{split} \omega \ddot{f} + 3\dot{f} + e^2 \omega f(h^2 - g^2) &= 0, \\ \omega \ddot{g} + \dot{g} + e^2 \omega g(h^2 - \omega^2 f^2) &= 0, \\ \omega \ddot{h} + \dot{h} - e^2 \omega h(\omega^2 h^2 + g^2) &= 0. \end{split}$$
(2.10.63)

Unfortunately we do not succeeded in finding solutions of system (2.10.63).

2.11. On connection between solutions of Dirac and Maxwell equations, dual Poincare invariance and superalgebras of invariance of Dirac equation

Consider the massless Dirac equation

$$i\gamma\partial\psi = 0 \tag{2.11.1}$$

(we use notation given in Section 2.1). There is a connection between solutions of Equations (2.11.1) and Maxwell's equations for vacuum

$$\dot{\vec{E}} \equiv \frac{\partial \vec{E}}{\partial t} = \operatorname{rot} \vec{H}, \qquad \operatorname{div} \vec{E} = 0,$$

$$\dot{\vec{H}} \equiv \frac{\partial \vec{H}}{\partial t} = -\operatorname{rot} \vec{E}, \qquad \operatorname{div} \vec{H} = 0,$$
(2.11.2)

where $\vec{E} = (E_1, E_2, E_3)$, $\vec{H} = (H_1, H_2, H_3)$ are vectors of electric and magnetic fields. To establish this connection let us decompose an arbitrary solution of Equation (2.11.1) into real and imaginary parts using notation of Ljolje [67*]

$$\psi = \psi_{\text{real}} + \psi_{\text{imag}} = \begin{pmatrix} -D_1 \\ D_3 \\ -B_2 \\ -G \end{pmatrix} + i \begin{pmatrix} D_2 \\ -F \\ -B_1 \\ B_3 \end{pmatrix}.$$
 (2.11.3)

Theorem 2.11.1 [65*,64*] Let ψ , defined by (2.11.3), be an arbitrary solution of the massless Dirac equation (2.11.1). Then the functions

$$\vec{E} = \vec{D} + \vec{\nabla} \left(\int_{t_0}^t G(\tau, \vec{x}) d\tau + \widetilde{G}(t_0, \vec{x}) \right),$$

$$\vec{H} = \vec{B} + \vec{\nabla} \left(\int_{t_0}^t F(\tau, \vec{x}) d\tau + \widetilde{F}(t_0, \vec{x}) \right),$$
(2.11.4)

where $\widetilde{G}(t_0, \vec{x})$ and $\widetilde{F}(t_0, \vec{x})$ satisfy the Poisson equations

$$\Delta \widetilde{G}(t_0, \vec{x}) = \partial G(\tau, \vec{x}) \tau \Big|_{\tau = t_0}, \qquad \Delta \widetilde{F}(t_0, \vec{x}) = \partial F(\tau, \vec{x}) \tau \Big|_{\tau = t_0}, \qquad (2.11.5)$$

 t_0 is an arbitrary constant, are solutions of Maxwell equations (2.11.2).

Proof. First of all we note that after substitution of ψ (2.11.3) into (2.11.1) and separation into real and imaginary parts we get the following Maxwell equations with currents

$$\dot{\vec{D}} = \operatorname{rot} \vec{B} - \vec{\nabla}G, \qquad \operatorname{div} \dot{\vec{D}} = -\dot{G}, \\ \dot{\vec{B}} + \operatorname{rot} \vec{D} = -\vec{\nabla}F, \qquad \operatorname{div} \vec{B} = -\dot{F}$$

$$(2.11.6)$$

where $\vec{D} = (D_1, D_2, D_3)$, $\vec{B} = (B_1, B_2, B_3)$, dot means differentiation with respect to t. So, the Dirac equation (2.11.1) and the system (2.11.6) are fully equivalent. Therefore, taking into account (2.11.6) and the well-known fact that every component of ψ -function (2.11.3) of the Dirac equation (2.11.1) satisfies wave equation $\Box \psi = 0$ (in particular, $\Delta G(\tau, \vec{x}) = \frac{\partial^2}{\partial \tau^2} G(\tau, \vec{x})$, we find after substitution of (2.11.4) into (2.11.2)

$$\begin{aligned} \dot{\vec{E}} &- \operatorname{rot} \vec{H} = \dot{\vec{D}} + \vec{\nabla}G - \operatorname{rot} \vec{B} = 0, \\ \operatorname{div} \vec{E} &= \operatorname{div} \vec{D} + \int_{t_0}^t \Delta G(\tau, \vec{x}) d\tau + \Delta \widetilde{G}(t_0, \vec{x}) = \\ &= \operatorname{div} \vec{D} + \int_{t_0}^t \left[\frac{\partial^2}{\partial \tau^2} G(\tau, \vec{x}) \right] d\tau + \Delta \widetilde{G}(t_0, \vec{x}) = \\ &= \operatorname{div} \vec{D} + \dot{G} - \frac{\partial G(\tau, \vec{x})}{\partial \tau} \Big|_{\tau = t_0} + \Delta \widetilde{G}(t_0, \vec{x}) = 0 \end{aligned}$$

In the last equality we have used (2.11.5). In the same spirit one can prove the validity of the theorem for the second part of the Maxwell equations (2.11.2). Thus, the theorem is proved. And in addition, the inverse statement also holds.

Theorem 2.11.2 Let there be given a solution \vec{E} , \vec{H} of Maxwell's equations (2.11.2) and two solutions F and G of scalar wave equation

$$\Box F = 0, \qquad \Box G = 0. \tag{2.11.7}$$

Then the ψ -function (2.11.3) with components F, G and

$$D_{a} = E_{a} - \partial_{a} \left(\int_{t_{0}}^{t} G(\tau, \vec{x}) d\tau + \widetilde{G}(t_{0}, \vec{x}) \right),$$

$$B_{a} = H_{a} - \partial_{a} \left(\int_{t_{0}}^{t} F(\tau, \vec{x}) d\tau + \widetilde{F}(t_{0}, \vec{x}) \right),$$
(2.11.8)

where a = 1, 2, 3 and $\tilde{G}(t_0, \vec{x})$, $\tilde{F}(t_0, \vec{x})$ are determined from (2.11.5), is a solution of the massless Dirac equation (2.11.1).

Proof. Let us use the equivalence between the Dirac equation (2.11.1) and the system (2.11.6). Having substituted (2.11.8) into (2.11.6) and taking into account (2.11.2), (2.11.7), and (2.11.5), we get

$$ec{D} - \operatorname{rot} ec{B} + ec{
abla} G = ec{E} - ec{
abla} G + ec{
abla} G - \operatorname{rot} ec{H} = 0,$$

 $\operatorname{div} ec{D} + \dot{G} = \operatorname{div} ec{E} - \int_{t_0}^t \Delta G(\tau, ec{x}) d\tau - \Delta \widetilde{G}(t_0, ec{x}) + \dot{G} = 0.$

Analogously, one proves the theorem for the remaining equations of system (2.11.6).

Theorem 2.11.2 has an important corollary: choosing F = G = 0 we get from (2.11.8) $\vec{D} = \vec{E}, \vec{B} = \vec{H}$ and in this case the formula (2.11.3) takes especially simple form

$$\psi = \begin{pmatrix} -E_1 + iE_2 \\ E_3 \\ -H_2 - iH_1 \\ iH_3 \end{pmatrix}$$
(2.11.9)

So, if \vec{E} and \vec{H} satisfy Maxwell equations (2.11.2), then ψ given by (2.11.9) automatically satisfies Dirac equation (2.11.1), and one can consider relation (2.11.9) as representation of spinor field ψ by electromagnetic field \vec{E} , \vec{H} . It is appropriate to note that if \vec{E} and \vec{H} are transformed under Lorentz boost as electromagnetic field, the ψ -function (2.11.9) is not transformed like a Dirac spinor (this question will be discussed in detail below). It will also be noted that, according to Theorem 2.11.1, the procedure of obtaining solutions of the vacuum Maxwell equations (2.11.2) from those of the massless Dirac equations (2.11.1) and associated Poisson equations (2.11.5) is unique to within a gauge transformation, whereas the inverse procedure (Maxwell \rightarrow Dirac) involves ambiguities due to the arbitrary choice of additional scalar fields F, G satisfying (2.11.7).

Consider an example. Let us take solutions of Maxwell equations (2.11.2) and wave equations (2.11.7) in the form

$$\vec{E} = \vec{\alpha} \times \vec{x}, \quad \vec{H} = -2\vec{\alpha}t, \quad F = G = 3t^2 + \vec{x}^2, \quad \vec{\alpha} = \text{const.}$$

Then, by means of (2.11.8), (2.11.3) one easily finds the following solution of the Dirac equation (2.1.1)

$$\psi = \begin{pmatrix} -[(\vec{\alpha} \times \vec{x})_1 - 2tx_1] + i[(\vec{\alpha} \times \vec{x})_2 - 2tx_2] \\ [(\vec{\alpha} \times \vec{x})_3 - 2tx_3] - i(3t^2 + \vec{x}^2) \\ 2t(\alpha_2 + x_2) + 2it(\alpha_1 + x_1) \\ -(3t^2 + \vec{x}^2) - 2it(\alpha_3 + x_3) \end{pmatrix}$$

Chapter 2. Systems of Poincare-invariant Nonlinear PDEs

In terms of \vec{D} , \vec{B} , F, G from (2.11.3)

$$\overline{\psi}\psi = \vec{D}^2 - \vec{B}^2 + F^2 - G^2 \tag{2.11.10}$$

and in the considered case we have

$$\overline{\psi}\psi = \vec{\alpha}^2 \vec{x}^2 - (\vec{\alpha} \cdot \vec{x})^2 - 4t^2 (\vec{\alpha}^2 + 2\vec{\alpha} \cdot \vec{x}).$$

Let us make a four-component ψ -function as

$$\psi = i\gamma \partial \begin{pmatrix} \varphi_0 \\ \varphi_1 \\ \varphi_2 \\ \varphi_3 \end{pmatrix}$$
(2.11.11)

where $\varphi_0, \ldots, \varphi_3$ are arbitrary solutions of the wave equation, that is $\Box \varphi_{\mu} = 0$. So, (2.11.11), (2.11.3), and (2.11.4) give the following chain of solutions: scalar wave equation \rightarrow massless Dirac equation \rightarrow vacuum Maxwell equations. Infinite series of solutions of scalar wave equation is given in (2.3.60).

For further analysis it is convenient to consider Dirac equation (2.11.1) together with its conjugation and to write it uniformly as

$$i\Gamma^{\mu}\partial_{\mu}\psi = 0, \qquad (2.11.12)$$

where $\psi = \psi(x) = \text{column}(\psi, \tilde{\psi}), \ \tilde{\psi} = \gamma_0 \psi^*, \ \Gamma^{\mu} \text{ are } 8 \times 8 \text{ matrices}$

$$\Gamma^{\mu} = \begin{pmatrix} \gamma^{\mu} & 0_4 \\ 0_4 & -(\gamma^{\mu})^T \end{pmatrix}$$
(2.11.13)

where 0_4 is the 4×4 zero matrix.

Symmetry properties of Equation (2.11.12) were studied first by Dirac, who had shown that it is comformally invariant (see §2.3). Afterwards, Pauli and Touschek found that this equation admits as well the eight-parameter group G_8 of component transformations. And finally, Ibragimow [66*] had proved that the 23-parameter group $G_{23}=C(1,3)\oplus G_8$ is maximal in the sense of the Lieinvariance group of the equation. Relativistic invariance of Equation (2.11.12) is usually understood as invariance with respect to the spinor representation

$$D(\frac{1}{2}, 0) \oplus D(0, \frac{1}{2}) \oplus D(\frac{1}{2}, 0) \oplus D(0, \frac{1}{2})$$
 (2.11.14)

of the Poincare group P(1,3) (it means that ψ is transformed as a spinor under the Lorentz boost). However, the invariance of Equation (2.11.12) under the Pauli-Touschek 8-parameter group allows for two additional representations of AP(1,3), which are realized on the set of solutions of the equation, namely

$$D(1,0) \oplus D(0,1) \oplus D(0,0) \oplus D(0,0)$$
 (2.11.15)

and

$$D(\frac{1}{2},\frac{1}{2}) \oplus D(\frac{1}{2},\frac{1}{2}).$$
 (2.11.16)

So, it is natural to call this dual Poincare invariance. The explicit form for the basis elements of AP(1,3) for representations (2.11.14)-(2.11.16) is

$$AP^{k}(1,3) = \left\langle P_{\mu} = \frac{\partial}{\partial x^{\mu}}, \quad J^{(k)}_{\mu\nu} = x_{\mu}P_{\nu} - x_{\nu}P_{\mu} + S^{(k)}_{\mu\nu} \right\rangle, \qquad (2.11.17)$$

where k = 1, 2, 3 corresponds to (2.11.14) - (2.11.16), respectively;

$$S_{\mu\nu}^{(1)} = -\frac{1}{4} [\Gamma_{\mu}, \Gamma_{\nu}];$$

$$S_{\mu\nu}^{(2)} = S_{\mu\nu}^{(1)} + Q_{\mu\nu};$$

$$S_{01}^{(3)} = S_{01}^{(2)}, \qquad S_{02}^{(3)} = S_{02}^{(2)}, \qquad S_{03}^{(3)} = S_{03}^{(2)} - 2Q_{03}$$

$$S_{12}^{(3)} = S_{12}^{(2)}, \qquad S_{13}^{(3)} = S_{13}^{(2)} - 2Q_{13}, \qquad S_{23}^{(3)} = S_{23}^{(2)} - 2Q_{23};$$
(2.11.18)

 Γ_{μ} are given in (2.11.13); $Q_{\mu\nu}$ are six basis elements of the Pauli-Touschek algebra:

$$Q_{01} = \frac{1}{2} \begin{pmatrix} 0_4 & -i\gamma_0\gamma_2 \\ i\gamma_0\gamma_2 & 0_4 \end{pmatrix}, \qquad Q_{02} = \frac{1}{2} \begin{pmatrix} 0_4 & -\gamma_0\gamma_2 \\ -\gamma_0\gamma_2 & 0_4 \end{pmatrix},$$
$$Q_{03} = \frac{1}{2} \begin{pmatrix} -\gamma_5 & 0_4 \\ 0_4 & \gamma_5 \end{pmatrix}, \qquad Q_{12} = \frac{i}{2} \begin{pmatrix} I_4 & 0_4 \\ 0_4 & -I_4 \end{pmatrix}, \qquad (2.11.19)$$
$$Q_{13} = \frac{1}{2} \begin{pmatrix} 0_4 & -\gamma_1\gamma_3 \\ -\gamma_1\gamma_3 & 0_4 \end{pmatrix}, \qquad Q_{23} = \frac{i}{2} \begin{pmatrix} 0_4 & \gamma_1\gamma_3 \\ -\gamma_1\gamma_3 & 0_4 \end{pmatrix}.$$

It will be noted that the action of operators (2.11.17) is defined in the space of 8-component functions introduced in (2.11.12). Invariance of Equation (2.11.12) under AP⁽²⁾(1,3) results in the possibility to represent it in the form (2.11.6), and invariance of (2.11.12) under AP⁽³⁾(1,3) allows us to rewrite it as $[67^*]$

$$\partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu} - \frac{1}{2}\epsilon_{\mu\nu\rho\sigma}\left(\partial^{\rho}B^{\sigma} - \partial^{\sigma}B^{\rho}\right) = 0, \qquad (2.11.20)$$
$$\partial_{\nu}A^{\nu} = \partial_{\nu}B^{\nu} = 0,$$

where

$$\psi = \psi_{\text{real}} + i\psi_{\text{imag}} = \begin{pmatrix} -A^2 \\ -B^0 \\ -B^1 \\ B^3 \end{pmatrix} + i \begin{pmatrix} -A^1 \\ A^3 \\ B^2 \\ -A^0 \end{pmatrix}$$
(2.11.21)

Now consider the following three states of symmetry operators of Equation (2.11.12)

$$SA^{(k)} = \left\langle P_{\mu}, J^{(k)}_{\mu\nu}, \Gamma_{4}, I; Q_{\mu\nu} \right\rangle$$
(2.11.22)

where P_{μ} , $J_{\mu\nu}^{(k)}$ and $Q_{\mu\nu}$ are defined in (2.11.17) and (2.11.19), Γ_{μ} are given in (2.11.13), $\Gamma_4 = \Gamma_0 \Gamma_1 \Gamma_2 \Gamma_3$. These sets of operators form Lie algebras as well as superalgebras [64*]. Operators P_{μ} , $J_{\mu\nu}^{(k)}$, Γ_4 , *I* are even, and $Q_{\mu\nu}$ are odd in corresponding superalgebras. To prove this statement we write down commutation and anticommutation relations for these operators.

Operators P_{μ} and $J_{\mu\nu}^{(k)}$ satisfy standard commutation relations of the Poincare algebra (1.1.4); Γ_4 and *I* commute with all elements of SA^(k). Further, it is convenient to introduce notations

$$R_{a} = Q_{0a}, \qquad T_{a} = \frac{1}{2} \epsilon_{abc} Q_{bc}$$

$$N_{a}^{(k)} = J_{0a}^{(k)}, \qquad M_{a}^{(k)} = \frac{1}{2} \epsilon_{abc} J_{bc}^{(k)}$$
(2.11.23)

It is easy to verify that

$$\{R_a, R_b\} \equiv R_a R_b + R_b R_a = \frac{1}{2} \delta_{ab} I$$

$$\{T_a, T_b\} = -\frac{1}{2} \delta_{ab} I, \qquad \{R_a, T_b\} = \delta_{ab} \Gamma_4$$

$$(2.11.24)$$

Operators R_a , T_a from SA⁽¹⁾ commute with all even operators of SA⁽¹⁾. For SA⁽²⁾ we have

$$[P_{\mu}, R_{a}] = [P_{\mu}, T_{a}] = 0,$$

$$[N_{a}^{(2)}, R_{b}] = [R_{a}, R_{b}] = \epsilon_{abc}T_{c},$$

$$[N_{a}^{(2)}, T_{b}] = [R_{a}, T_{b}] = -\epsilon_{abc}R_{c},$$

$$[M_{a}^{(2)}, R_{b}] = -\epsilon_{abc}R_{c},$$

$$[M_{a}^{(2)}, T_{b}] = [T_{a}, T_{b}] = -\epsilon_{abc}T_{c},$$

(2.11.25)

Superalgebra $SA^{(3)}$ is isomorphic to $SA^{(2)}$. The isomorphism is achieved by the transformation

$$R_3 \to R'_3 = -R_3, \qquad T_1 \to T'_1 = -T_1, \qquad T_2 \to T'_2 = -T_2 \qquad (2.11.26)$$

So, the structure of superalgebras (2.11.22) is fully described.

In conclusion, it will be noted that dual Poincare invariance analogous to that considered above has system of two Dirac equations with masses m and -m. For more detail see [64^{*}].

In [64*] nonlinear generalizations of the Dirac system are constructed which possess dual Poincare invariance. Therein we construct the complete set of P(1,3)-inequivalent ansatze of codimension 1 for all representations of Poincare algebra discussed. These ansatze are used for reduction and for finding exact solutions of some nonlinear Dirac equations.

Chapter 3

Euclid and Galilei Groups and Nonlinear PDEs for Scalar Fields

In the present chapter we describe a wide class of nonlinear PDEs for scalar fields invariant under Euclid, Galilei, or larger groups. For some of such equations we construct multiparameter families of exact solutions.

3.1. Second-order PDEs invariant under the extended Euclid group $\widetilde{E}(1, n-1)$

The aim of the present paragraph is to describe equations of the form

$$\Box u + F(u, u)u_0 = 0$$
 (3.1.1)

where u = u(x), $x = (x_0, x_1, \ldots, x_{n-1})$, $u_1 = \left\{\frac{\partial u}{\partial x_a}\right\}$, $\mu = \overline{1, n-1}$. F is a differentiable function, which is invariant under the extended Euclid group $\widetilde{E}(1, n-1)$, i.e., the Euclid group E(1, n-1) (E(1, n-1) contains translation and rotation in \mathbb{R}^{n-1} added by time translation) augmented by the one-parameter group of scale transformation. The corresponding Lie algebra $A\widetilde{E}(1, n-1)$ is defined by the following commutation relations

$$[P_0, P_a] = 0, \quad [P_0, J_a] = 0, \quad [P_a, J_b] = i\epsilon_{abc}P_c, \quad a, b = \overline{1, n - 1},$$

$$[J_a, J_b] = i\epsilon_{abc}J_c, \quad [P_0, D] = iP_0, \quad [P_a, D] = iP_a. \quad [J_a, D] = 0$$
(3.1.2)

Below we consider the standard representation of $A\widetilde{E}(1, n-1)$

$$P_0 = i\partial_0, \qquad P_a = -i\partial_a, J_a = (\vec{x} \times \vec{P})_a = \epsilon_{abc} x_b P_c.$$
(3.1.3)

The most general form of the operator D which satisfies relations (3.1.2) is as follows

$$D = x_{\nu}P^{\nu} + i(au + a)\partial_u \equiv x_0P_0 - x_aP_a + i(au + a)\partial_u, \qquad (3.1.4)$$

where a, x are arbitrary constants.

Theorem 3.1.1. [91] Equation (3.1.1) is invariant with respect to the group $\widetilde{E}(1, n-1)$ iff

$$1^{\circ} \quad F = u^{k} f\left(\frac{u_{a}u_{a}}{u^{2(k+1)}}\right),$$

$$2^{\circ} \quad F = e^{u} f\left(\frac{u_{a}u_{a}}{e^{4u}}\right),$$

$$3^{\circ} \quad F = \sqrt{u_{a}u_{a}}f(u),$$

$$(3.1.5)$$

where k is an arbitrary constant, f is an arbitrary differentiable function.

Proof. In order to prove the theorem we use the Lie algorithm in much the same way as in $\S1.1$. The infinitesimal operator may be written in the form

$$X = \xi^{\mu}(x)\partial_{\mu} + \eta(u)\partial_{u} = \alpha^{\mu}P_{\mu} + \beta^{a}J_{a} + \omega D, \qquad \mu = \overline{0, n-1}, \qquad (3.1.6)$$

where α_{μ} , β_a , æ are arbitrary constants.

Using the condition of invariance (necessary and sufficient)

$$\frac{X}{2}(\Box u + Fu_0) \Big|_{\Box u + Fu_0 = 0} = 0,$$

where X_2 is the second extension of the operator X (3.1.6) which is defined according to formulae (1.1.7), we obtain

$$u_b F_{u_a} - u_a F_{u_b} = 0,$$

(a-\approx) u_a F_{u_a} + \eta F_u + \approx F = 0. (3.1.7)

The first equation of (3.1.7) implies that

$$F(u, \underline{u}) = F(u, w), \qquad w = u_a u_a,$$

which allows us to rewrite the second equation from (3.1.7) in the form

$$2(a - x)wF_w + \eta F_u + xF = 0. (3.1.8)$$

There are three possibilities

$$1^{\circ} a \neq 0, a = 0, 2^{\circ} a = 0, a \neq 0, 3^{\circ} a = a = 0.$$

According to them we have three different solutions of Equation (3.1.8) (see (3.1.5)) and obtain three different forms of the operator D (3.1.4)

$$1^{\circ} \quad D = x^{\nu} P_{\nu} + \frac{i}{k},$$

$$2^{\circ} \quad D = x^{\nu} P_{\nu} - i\partial_{u},$$

$$3^{\circ} \quad D = x^{\nu} P_{\nu}.$$

$$(3.1.9)$$

It is easy to verify that Equation (3.1.1) with F from (3.1.5) is invariant under $\widetilde{E}(1, n - 1)$. The invariance with respect to translations and rotations generated by operator (3.1.3) is obvious. The scale transformations generated by dilation operators D (3.1.9) have the form

1°
$$x'_{\mu} = e^{\theta} x_{\mu}, \qquad u'(x') = e^{-\theta/k} u(x), \qquad \theta = \text{const},$$

2° $x'_{\mu} = e^{\theta} x_{\mu}, \qquad u'(x') = u(x) - \theta$ (3.1.10)
3° $x'_{\mu} = e^{\theta} x_{\mu}, \qquad u'(x') = u(x).$

One can make sure that these transformations leave Equation (3.1.1) with corresponding function F invariant. The theorem is proved.

Remark. Equation (3.1.1) with function F = F(u) is invariant under $\widetilde{E}(1, n-1)$ iff

1°
$$F = \lambda_1 u^k, \quad D = x^{\nu} P_{\nu} + \frac{i}{k},$$

2° $F = \lambda_2 \exp(u), \quad D = x^{\nu} P_{\nu} - i \partial_u,$ (3.1.11)
3° $F = 0, \quad D = x^{\nu} P_{\nu}.$

where λ_1 , λ_2 , k are arbitrary constants.

If $F = F(\underline{u})$ (3.1.8) yields for $a = (1 - \frac{1}{k})$

$$F = \lambda (u_a u_a)^{k/2}, \qquad D = x^{\nu} P_{\nu} + \left(1 - \frac{1}{k}\right) i, \qquad \lambda, k = \text{const.} \tag{3.1.12}$$

In addition, Equation (3.1.1) with function F(3.1.12) is also invariant with respect to the transformations

$$u(x) \rightarrow u'(x') = u(x) + \text{const}, \qquad x'_{\mu} = x_{\mu}$$

150 Chapter 3. Euclid and Galilei Groups and Nonlinear PDEs for Scalar Fields

3.2. Reduction and exact solutions of nonlinear PDEs of the type $\Box u + F(u, u)u_0 = 0$

In this paragraph we reduce and find exact solutions of $\widetilde{E}(1,3)$ -invariant nonlinear Equations (3.2.4), (3.2.42), (3.2.51), following the algorithm expounded in §1.4. With that end in view the complete set of $\widetilde{E}(1,3)$ -nonequivalent ansatze is built.

1. Invariants of the Euclid group $\widetilde{E}(1,3)$.

To obtain invariants of E(1,3) group we act as in §1.4, where invariants of $\widetilde{P}(1,2)$ are described. Let us proceed from the equation

$$\xi^{\mu}(x)\partial_{\mu}\omega(x) = 0, \qquad (3.2.1)$$

where $\xi^{\mu}(x)$ are defined in (3.1.6), and go on to the equivalent characteristic equations

$$\frac{dx_0}{\xi^0} = \frac{dx_1}{\xi^1} = \frac{dx_2}{\xi^2} = \frac{dx_3}{\xi^3} = d\tau$$
(3.2.2)

which can be rewritten as system of ODEs

$$\frac{dx_{\mu}}{d\tau} = \xi^{\mu}(x(\tau)). \tag{3.2.3}$$

Since $\xi^{\mu}(x)$ are linear functions of x_{μ} (see (3.1.5)), so the system (3.2.3) is linear. The general solution of this system after eliminating τ by means of Equation (3.2.2) yields the unknown invariants. Depending on the correlation between coefficients in (3.1.5) we shall have several different cases.

Without going into details of lengthy though elementary calculations we list in Table 3.2.1 the invariants of $\widetilde{E}(1,3)$.

2. Let us apply the above results for obtaining solutions of the equation

$$\Box u + \lambda u u_0 = 0 \tag{3.2.4}$$

This equation is used in field theory and gas dynamics. Solutions of twodimensional Equation (3.2.4) are obtained in [172]. Following §1.4 [178] we seek solutions of Equation (3.2.4) in the form [61]

$$u(x) = f(x)\varphi(\omega), \qquad (3.2.5)$$

where $\omega(x) = \{\omega_1(x), \omega_2(x), \omega_3(x)\}$ are given in the Table 3.2.1 and functions f(x) are to be determined from the equation

$$\frac{du}{-\alpha u} = d\tau \tag{3.2.6}$$

and (3.2.2) or from the condition of splitting.

Invariant variables	Conditions on
$\omega(x)=\{\omega_1(x),\omega_2(x),\omega_3(x)\}$	the parameters
$\frac{\alpha_a y_a}{y_0}, \frac{y_a y_a}{y_0^2},$	$\alpha_a^2 = \alpha_a \alpha_a = \alpha^2 \neq 0,$
$y_0 \exp\left\{b_1 \arctan rac{\gamma_a y_a}{eta_a y_a} ight\}$	$\beta_a^2 \neq 0, \gamma_a^2 \neq 0,$
	$egin{aligned} &lpha_aeta_a=lpha_a\gamma_a=eta_a\gamma_a=0,\ &(eta_ay_a)^2+(\gamma_ay_a)^2= \end{aligned}$
Gelle Velle	$=rac{b_{1}^{2}}{lpha^{2}}(lpha^{2}\omega_{2}-w_{1}^{2}).$
$\frac{-uy_1}{y_0}, \frac{y_1y_1}{y_0^2},$	$\beta_a^z \neq 0,$
$rac{eta_a y_a}{lpha_a y_a} - b_2 \ln lpha_a y_a$	$\alpha_a^2 = \alpha_a \beta_a = 0.$
$lpha_a y_a/y_0, \; y_a y_a/y_0^2, \; eta_a y_a/y_0^{b_3}$	$\alpha_a^2 \neq 0, \beta_a^2 = \alpha_a \beta_a = 0.$
$\frac{\alpha_a y_a}{y_0}, \frac{\sigma_a y_a}{y_0},$	$\alpha_a^2 \neq 0, \delta_a^2 = \sigma_a^2 = 0,$
$\delta_a x_a - b_4 \ln y_0$	$\alpha_a \delta_a = \alpha_a \sigma_a = 0,$
	$\sigma_a \delta_a = a \neq 0.$
$eta_ u x^ u, rac{(lpha_a y_a)}{-lpha_a^2} + y_a y_a,$	$\alpha_a^2 = -\alpha_\nu \beta^\nu = \alpha^2 \neq 0,$
	$\ell_a^2 = \delta_a^2 = \ell \neq 0;$
$x_0 - b_5 \arctan \ell_a y_a / \delta_a y_a$	$\alpha_a \ell_a = \alpha_a \delta_a = \ell_a \delta_a = \beta_\nu \delta^\nu = 0,$
	$\beta_{\nu}\beta^{\nu} = \beta \neq 0,$
$Q_{a} = \nu \left(\alpha_a y_a \right)^2$	$\frac{(\ell_a y_a)^2}{2} + \frac{(\delta_a y_a)^2}{2} = \ell \omega_2.$
$\rho_{\nu}x^{\nu}, \frac{1}{-\alpha^{2}} + y_{a}y_{a},$	$\alpha_a^2 = -\alpha_\nu \beta^\nu \neq 0,$
$x_0 + o_6 \ln \sigma_a z_a$	$\beta_{\nu}\beta^{\nu} = \beta \neq 0,$
$Q = \left(\alpha_a y_a \right)^2 + L q$	$\alpha_a \sigma_a = \sigma_a = \rho_\nu \sigma = 0.$
$p_{\nu}x$, $\frac{1}{2} + b_7 \ell_a y_a$, $\frac{1}{2} (\alpha_a y_a)^3 + b_7 (\alpha_a y_a) (\ell_a y_a) - b^2 \delta_a y_a$	$\begin{array}{c} \rho_{\nu}\rho^{*} \neq 0, \ell_{a}^{*} = \alpha_{a}\delta_{a} = \delta_{a}^{*} = \\ -\ell \neq 0, \ \alpha^{2} = \delta, \ell = 0 \end{array}$
$3(\neg a g a) (\neg a g a)(\neg a g a) (\neg a g a)$	$a^2 - a^2 \neq 0$ $\ell^2 - \delta^2 - \ell \neq 0$
$x_0 - b_8 \arctan(\ell_a x_a / \delta_a x_a)$	$\alpha_a = \alpha \neq 0, \ \alpha_a = \alpha_a = c \neq 0, \\ \alpha_a \ell_a = \alpha_a \delta_a = \ell_a \delta_a = 0.$
$\frac{y_1}{y_2}, \frac{y_2}{y_3}.$	
$a_{\nu}x^{\nu}, \beta_{a}x_{a}, \gamma_{a}x_{a}$	$\alpha_{\nu}\alpha^{\nu} = \alpha^2 \neq 0, \ \beta_{\tau}^2 = \beta^2 \neq 0.$
	$\gamma_a^2 = \gamma^2 \neq 0, \ \alpha_a \beta_a = b_9,$
	$\alpha_a \gamma_a = b_{10}, \ \beta_a \gamma_a = b_{11}$
	Invariant variables

Table 3.2.1. The invariants of $\widetilde{E}(1,3)$.

Here $y_{\nu} = x_{\nu} + a_{\nu}$, $z_{\nu} = x_{\nu} + \frac{1}{2}a_{\nu}$; $a_{\nu}, b_{\nu}, c_a, \delta_a, \sigma_a, \alpha_{\nu}, \beta_{\nu}, \gamma_{\nu}$ are arbitrary constants satisfying conditions stated in the table.

So, we have 10 ansatze of the form (3.2.5)

$$u(x) = \begin{cases} y_0^{-1/2} \varphi(\omega), & \text{for } \omega \text{ N1-4} \\ \varphi(\omega), & \text{for } \omega \text{ N5-10} \end{cases}$$
(3.2.7)

The numeration of ansatze used here and below corresponds to the numeration of invariants of the Table 3.2.1.

Inserting ansatze (3.2.5) into the Equation (3.2.4) we obtain 10 reduced PDEs for function $\varphi(\omega)$

$$\begin{array}{ll} (1) \quad (\omega_{1}^{2}-\alpha^{2})\varphi_{11}+4\omega_{2}(\omega_{2}-1)\varphi_{22}-\omega_{3}^{2}\left(1-\frac{\alpha^{2}b_{1}^{2}}{\alpha^{2}\omega_{2}-\omega_{1}^{2}}\right)\varphi_{33}-\\ -4\omega_{1}(1-\omega_{1})\varphi_{12}-4\omega_{2}\omega_{3}\varphi_{23}+(10\omega_{2}-6)\varphi_{2}+4\omega_{1}\varphi_{1}+\\ +\left(\frac{\alpha^{2}b_{1}^{2}}{\alpha^{2}\omega_{2}-\omega_{1}^{2}}-2\right)\omega_{3}\varphi_{3}-\lambda\omega_{1}\varphi\varphi_{1}-2\lambda\omega_{2}\varphi\varphi_{2}+\lambda\omega_{3}\varphi\varphi_{3}+2\varphi-\lambda\varphi^{2}=0;\\ (2) \quad \omega_{1}^{2}\varphi_{11}+4\omega_{2}(\omega_{2}-1)\varphi_{22}+\frac{b_{2}}{\omega_{1}^{2}}\varphi_{33}+4\omega_{1}(\omega_{2}-1)\varphi_{12}-4b_{2}\varphi_{23}+\\ +4\omega_{1}\varphi_{1}+(10\omega_{2}-6)\varphi_{2}-\lambda\omega_{1}\varphi\varphi_{1}-2\lambda\omega_{2}\varphi\varphi_{2}+2\varphi-\lambda\varphi^{2}=0;\\ (3) \quad (\omega_{1}^{2}+\alpha^{2})\varphi_{11}+4\omega_{2}(\omega_{2}-1)\varphi_{22}+b_{3}^{2}\omega_{3}^{2}\varphi_{33}+4\omega_{1}(\omega_{2}-1)\varphi_{12}+\\ +2b_{3}\omega_{1}\omega_{3}\varphi_{13}+4\omega_{3}(b_{3}\omega_{2}-1)\varphi_{23}+4\omega_{1}\varphi_{1}+(10\omega_{2}-6)\varphi_{2}+\\ +b_{3}(b_{3}+3)\omega_{3}\varphi_{3}-\lambda\omega_{1}\varphi\varphi_{1}-\lambda\omega_{2}\varphi\varphi_{2}-\lambda b_{3}\varphi\varphi_{3}+2\varphi-\lambda\varphi^{2}=0;\\ (4) \quad (\omega_{1}^{2}+\alpha^{2})\varphi_{11}+4\omega_{2}^{2}\varphi_{22}+b_{4}^{2}\varphi_{33}+4\omega_{1}\omega_{2}\varphi_{12}+2b_{4}\varphi_{13}+11b_{4}\omega_{2}\varphi_{23}+\\ +4\omega_{1}\varphi_{1}+10\omega_{2}\varphi_{2}+3b_{4}\varphi_{3}-\lambda\omega_{1}\varphi\varphi_{1}-2\lambda\omega_{2}\varphi\varphi_{2}-2b_{4}\varphi\varphi_{3}+2\varphi-\lambda\varphi^{2}=0;\\ (5) \quad \beta\varphi_{11}+\omega_{2}\varphi_{22}+\left(1-\frac{b_{5}}{\omega_{2}}\right)\varphi_{33}+2\beta_{0}\varphi_{13}-4\varphi_{2}+\lambda\beta_{0}\varphi\varphi_{1}+\lambda\varphi\varphi_{3}=0;\\ (6) \quad \beta\varphi_{11}+\omega_{2}\varphi_{22}+2b_{0}\varphi_{13}+\varphi_{33}-4b_{6}\varphi_{23}-4\varphi_{2}+\lambda\beta_{0}\varphi\varphi_{1}+\lambda\varphi\varphi_{3}=0;\\ (7) \quad \beta\varphi_{11}-b_{7}^{2}\varphi_{22}+2b_{7}^{2}(\omega_{2}^{2}-b_{7}^{2})\varphi_{33}-2\beta_{7}^{2}\ell^{2}\varphi_{13}+\lambda\beta_{0}\varphi\varphi_{1}=0;\\ (8) \quad -\alpha^{2}\varphi_{11}-4\omega_{2}\varphi_{22}+\left(1-\frac{b_{8}}{\omega_{2}}\right)\varphi_{33}+4\omega_{1}\varphi_{12}-6\varphi_{2}+\lambda\varphi\varphi_{3}=0;\\ (9) \quad (\omega_{1}^{2}-1)\varphi_{11}+(\omega_{2}^{2}-1)\varphi_{22}+(\omega_{3}^{2}-1)\varphi_{33}+2(\omega_{1}\omega_{2}-1)\varphi_{12}+\\ +2(\omega_{1}\omega_{3}-1)\varphi_{13}+2(\omega_{2}\omega_{3}-1)\varphi_{23}+4\omega_{1}\varphi_{1}+4\omega_{2}\varphi_{2}+4\omega_{3}\varphi_{3}-\\ -\lambda\varphi(\omega_{1}\varphi_{1}+\omega_{2}\varphi_{2}+\omega_{3}\varphi_{3})+2\varphi-\lambda\varphi^{2}=0;\\ (10) \quad \alpha^{2}\varphi_{11}-\beta^{2}\varphi_{22}-\gamma^{2}\varphi_{33}+2b_{9}\varphi_{12}+2b_{10}\varphi_{13}+2b_{11}\varphi_{23}+\lambda\alpha_{0}\varphi\varphi_{1}=0.\\ \text{Let in } (3.2.8) \varphi=\varphi(\omega_{1}), \text{ so we have the ODE}\\ (\omega_{1}^{2}-\alpha^{2})\varphi_{11}+4\omega_{1}\varphi_{1}-\lambda\omega_{1}\varphi_{1}+2\varphi-\lambda\varphi^{2}=0,\\ (3.2.9) \\ +2(\omega_{1}^{2}-\alpha^{2})\varphi_{11}+4\omega_{1}\varphi_{1}-\lambda\omega_{1}\varphi\varphi_{1}+2\varphi-\lambda\varphi^{2}=0,\\ (3.2.9) \\ +2(\omega_{1}^{2}-\alpha^{2})\varphi_{11}+4\omega_{1}\varphi_{1}-\lambda\omega_{1}\varphi\varphi_{1}+2\varphi-\lambda\varphi^{2}=0,\\ (3.2.9) \\ +2(\omega_{1}^{2}-\alpha^{2})\varphi_{11}+4\omega_{1}\varphi_{1}-\lambda\omega_{1}\varphi\varphi_{1}+2\varphi-\lambda\varphi^{2}=0,\\ (3.2.9) \\ +2(\omega_{1}^{2}-\alpha^{2})\varphi_{1}$$

which is reduced after first integration to the Riccati equation

$$\omega_1(\omega_1^2 - \alpha^2)\varphi_1 + (\omega_1^2 + \alpha^2)\varphi - \frac{\lambda}{2}\omega_1^2\varphi^2 = c_1, \quad c_1 = \text{const} \quad (3.2.10)$$

We seek solution of Equation (3.2.10) in the form

$$\varphi = c_1 \left[\omega_1 \psi(\omega_1) + \alpha^2 \right]^{-1} \tag{3.2.11}$$

Inserting (3.2.11) into (3.2.10) we obtain the equation

$$\frac{d\psi}{\psi^2 - \alpha^2 + \frac{1}{2}\lambda c_1} = \frac{d\omega_1}{-(\omega_1^2 - \alpha^2)}$$

which, depending on the constant $\alpha^2 - \frac{1}{2}\lambda c_1$, has the following solutions

$$\psi = \begin{cases} \frac{(\alpha - \omega_1)^{a/\alpha} - c_2(\alpha + \omega_1)^{a/\alpha}}{(\alpha - \omega_1)^{a/\alpha} + c_2(\alpha + \omega_1)^{a/\alpha}}, & \alpha^2 - \frac{1}{2}\lambda c_1 = a^2 > 0; \\ 2\alpha \left(\ln \left| \frac{\alpha - \omega_1}{\alpha + \omega_1} \right| + c_2 \right)^{-1}, & \alpha^2 - \frac{1}{2}\lambda c_1 = 0; \\ a \tan \left(\frac{a}{2\alpha} \ln \left| \frac{\alpha + \omega_1}{\alpha - \omega_1} \right| + c_2 \right), & \alpha^2 - \frac{1}{2}\lambda c_1 = -a^2 < 0. \end{cases}$$
(3.2.12)

From (3.2.12) and (3.2.11) we have the following solutions of Equation (3.2.10)

$$\varphi = \begin{cases} c_1 \left[a\omega_1 \frac{(\alpha - \omega_1)^{a/\alpha} - c_2(\alpha + \omega_1)^{a/\alpha}}{(\alpha - \omega_1)^{a/\alpha} + c_2(\alpha + \omega_1)^{a/\alpha}} + \alpha^2 \right]^{-1}, \\ c_1 \left[2\alpha\omega_1 \left(\ln \left| \frac{\alpha - \omega_1}{\alpha + \omega_1} \right| + c_2 \right)^{-1} + \alpha^2 \right]^{-1}, \\ c_1 \left[a\omega_1 \tan \left(\frac{a}{2\alpha} \ln \left| \frac{\alpha + \omega_1}{\alpha - \omega_1} \right| + c_2 \right) + \alpha^2 \right]^{-1}. \end{cases}$$
(3.2.13)

Equation (3.2.10) is reduced to Bernoulli one when $c_1 = 0$, and has the following general solution

$$\varphi = \left[c_2 \frac{\omega_1^2 - \alpha^2}{\omega_1} - \frac{\lambda}{8\alpha} \frac{\omega_1^2 - \alpha^2}{\omega_1} \ln \left| \frac{\omega_1 - \alpha}{\omega_1 + \alpha} \right| + \frac{\lambda}{4} \right]^{-1}.$$
 (3.2.14)

Functions (3.2.13) and (3.2.14) give the general solution of the Riccati equation (3.2.10).

The ansatz N1 from (3.2.7) and formulaes (3.2.13), (3.2.14) lead to the following solutions of Equation (3.2.4)

$$u(x) = c_1 \left[a \alpha_a y_a \frac{(\alpha y_0 - \alpha_a y_a)^{a/\alpha} - c_2 (\alpha y_0 + \alpha_a y_a)^{a/\alpha}}{(\alpha y_0 - \alpha_a y_a)^{a/\alpha} + c_2 (\alpha y_0 + \alpha_a y_a)^{a/\alpha}} + \alpha^2 y_0 \right]^{-1},$$

$$u(x) = c_1 \left[2\alpha \alpha_a y_a \left(\ln \left| \frac{\alpha y_0 - \alpha_a y_a}{\alpha y_0 + \alpha_a y_a} \right| + c_2 \right)^{-1} + \alpha^2 y_0 \right]^{-1}, \qquad (3.2.15)$$

Chapter 3. Euclid and Galilei Groups and Nonlinear PDEs for Scalar Fields

$$u(x) = c_1 \left[a\alpha_a y_a \tan\left(\frac{a}{2\alpha} \ln\left|\frac{\alpha y_0 + \alpha_a y_a}{\alpha y_0 - \alpha_a y_a}\right| + c_2\right) + \alpha^2 y_0 \right]^{-1},$$

$$u(x) = \left[c_2 \frac{(\alpha_a y_a)^2 - \alpha^2 y_0^2}{\alpha_a y_a} - \frac{\lambda}{8\alpha} \frac{(\alpha_a y_0)^2 - \alpha^2 y_0^2}{\alpha_a y_a} \ln\left|\frac{\alpha y_0 - \alpha_a y_a}{\alpha y_0 + \alpha_a y_a}\right| + \frac{\lambda}{4} y_0 \right]^{-1}.$$

If $\varphi = \varphi(\omega_2)$ then Equation (3.2.8(1)) takes the form

$$4\omega_2(\omega_2-1)\varphi_{22}+(10\omega_2-6)\varphi_2-2\lambda\omega_2\varphi\varphi_2+2\varphi-\lambda\varphi^2=0.$$

After integrating this ODE is reduced to the Riccati equation

$$4\omega_2(\omega_2 - 1)\varphi_2 + 2(\omega_2 - 1)\varphi - \lambda\omega_2\varphi^2 = c_1, \qquad c_1 = \text{const.}$$
(3.2.16)

The substitution

$$\psi(t) = t\varphi(t), \qquad t = \sqrt{\omega_2}$$

into Equation (3.2.16) yields the equation with separated variables. The general solution of the Riccati equation (3.2.16) is

$$\varphi = a\sqrt{\omega_2} \tan\left(\frac{\lambda a}{4} \ln \left| c_2 \frac{1 - \sqrt{\omega_2}}{1 + \sqrt{\omega_2}} \right| \right), \qquad c_1 = \frac{1}{2}\lambda a^2;$$
$$\varphi = \frac{1}{\sqrt{\omega_2}} \left(-\frac{\lambda}{4} \ln \left| \frac{\sqrt{\omega_2} - 1}{\sqrt{\omega_2} + 1} \right| + c_2 \right)^{-1}, \qquad c_1 = 0; \qquad (3.2.17)$$
$$\varphi = \frac{a}{\sqrt{\omega_2}} \frac{c_2(1 + \sqrt{\omega_2})^{\lambda a/2} - (1 - \sqrt{\omega_2})^{\lambda a/2}}{c_2(1 + \sqrt{\omega_2})^{\lambda a/2} + (1 - \sqrt{\omega_2})^{\lambda a/2}}, \qquad c_1 = -\frac{1}{2}\lambda a^2.$$

The ansatz N1 from (3.2.7) and functions (3.2.17) give the following solution of Equation (3.2.4)

$$u(x)=a\sqrt{y_ay_a} an\left(rac{\lambda a}{4}\ln\left|c_2rac{y_0-\sqrt{y_ay_a}}{y_0+\sqrt{y_ay_a}}
ight|
ight),$$

$$u(x) = (y_a y_a)^{-\frac{1}{2}} \left(-\frac{\lambda}{4} \ln \left| \frac{y_0 - \sqrt{y_a y_a}}{y_0 + \sqrt{y_a y_a}} \right| + c_2 \right)^{-1}, \qquad (3.2.18)$$

$$u(x) = a(y_a y_a)^{-\frac{1}{2}} \frac{c_2(y_0 + \sqrt{y_a y_a})^{\lambda a/2} - (y_0 - \sqrt{y_a y_a})^{\alpha a/2}}{c_2(y_0 + \sqrt{y_a y_a})^{\lambda a/2} + (y_0 - \sqrt{y_a y_a})^{\alpha a/2}}.$$

If $\varphi = \varphi(\omega_1)$, then Equation (3.2.8(2)) takes the form

$$\omega_1^2 \varphi_{11} + 4\omega_1 \varphi_1 - \lambda \omega_1 \varphi \varphi_1 + 2\varphi - \lambda \varphi^2 = 0.$$
 (3.2.19)

With the help of the substitution

$$\varphi = t\psi(t), \qquad t = \frac{1}{\omega_1}, \qquad (3.2.20)$$

the ODE (3.2.19) is reduced to

$$\psi_{tt} + \lambda \psi \psi_t = 0. \tag{3.2.21}$$

This latter equation is easily integrated and has the general solution as follows

$$\psi = c_1 \operatorname{th} \left(\frac{\lambda c_1}{2} t + c_2 \right), \qquad \psi = c_1 \operatorname{cth} \left(\frac{\lambda c_1}{2} t + c_2 \right),$$
$$\psi = -c_1 \operatorname{tan} \left(\frac{\lambda c_1}{2} t + c_2 \right), \qquad \psi = \left(\frac{\lambda}{2} t + c_2 \right)^{-1},$$

where c_1, c_2 are constants. Using these results and ansatz N2 from (3.2.7), we obtain the following solutions of Equation (3.2.4)

$$u(x) = \frac{c_1}{\alpha_a y_a} \operatorname{th} \left(\frac{\lambda c_1}{2} \frac{y_0}{\alpha_a y_a} + c_2 \right), \quad u(x) = \frac{-c_1}{\alpha_a y_a} \operatorname{tan} \left(\frac{\lambda c_1}{2} \frac{y_0}{\alpha_a y_a} + c_2 \right)$$
$$u(x) = \frac{c_1}{\alpha_a y_a} \operatorname{cth} \left(\frac{\lambda c_1}{2} \frac{y_0}{\alpha_a y_a} + c_2 \right), \quad u(x) = \left(\frac{\lambda}{2} y_0 + c_2 \alpha_a y_a \right)^{-1}. \quad (3.2.22)$$

If $u(x) = \varphi(\omega_3)h(x)$, $h(x) = (\alpha_a y_a)^{-1}$, ω_3 is from N2 Table 3.2.1 then Equation (3.2.4) is reduced to $\varphi_{\omega_3\omega_3} = 0$, i.e., $\varphi = c_1\omega_3 + c_2$, where c_1, c_2 are constants of integration. The corresponding solution of Equation (3.2.4) is

$$u(x) = c_1 \left(\frac{\beta_a y_a}{(\alpha_a y_a)^2} - b_3 \ln \left| \frac{\beta_a y_a}{\alpha_a y_a} \right| \right) + c_2 (\alpha_a y_a)^{-1}.$$
(3.2.23)

If $\varphi = \varphi(\omega_1)$, then Equation (3.2.8(3)) is reduced to the ODE

$$(\omega_1^2 + \alpha^2)\varphi_{11} + 4\omega_1\varphi_1 - \lambda\omega_1\varphi\varphi_1 + 2\varphi - \lambda\varphi^2 = 0.$$
(3.2.24)

Note, that Equation (3.2.24) coincides with Equation (3.2.9) to within sign, so taking into account (3.2.24) we can present at once its general solution

$$\varphi = c_1 \left[a\omega_1 \operatorname{th} \left(\frac{a}{\alpha} \arctan \frac{\omega_1}{\alpha} + c_1 \right) - \alpha^2 \right]^{-1}, \quad \alpha^2 + \frac{\lambda c_1}{2} = -a^2 < 0;$$

$$\varphi = c_1 \left[a\omega_1 \left(\arctan \frac{\omega_1}{\alpha} + c_2 \right)^{-1} - \alpha^2 \right]^{-1}, \quad \alpha^2 + \frac{\lambda c_1}{2} = 0; \quad (3.2.25)$$

$$\varphi = c_1 \left[a\omega_1 \tan \left(-\frac{a}{\alpha} \arctan \frac{\omega_1}{\alpha} + c_2 \right) - \alpha^2 \right]^{-1}, \quad \alpha^2 + \frac{\lambda c_1}{2} = a^2 > 0;$$

$$\varphi = \left[c_2 \frac{\omega_1^2 + \alpha^2}{\omega_1} - \frac{\lambda}{4\alpha} \frac{\omega_1^2 + \alpha^2}{\omega_1} \arctan \frac{\omega_1}{\alpha} + \frac{\lambda}{4}\right]^{-1}, \quad c_1 = 0;$$

where c_1 , c_2 are arbitrary constants.

Formulae (3.2.25) and ansatz N3 (3.2.7) give the following solutions of Equation (3.2.4)

$$u(x) = c_1 \left[a\alpha_a y_a \operatorname{th} \left(\frac{a}{\alpha} \arctan \frac{\alpha_a y_a}{\alpha y_0} + c_2 \right) - \alpha^2 \right]^{-1},$$

$$u(x) = c_1 \left[a\alpha_a y_a \operatorname{tan} \left(-\frac{a}{\alpha} \arctan \frac{\alpha_a y_a}{\alpha y_0} + c_2 \right) - \alpha^2 y_0 \right]^{-1},$$

$$u(x) = c_1 \left[\alpha\alpha_a y_a \left(\arctan \frac{\alpha_a y_a}{\alpha y_0} + c_2 \right)^{-1} - \alpha^2 y_0 \right]^{-1},$$

$$u(x) = \frac{\alpha_a y_a}{(\alpha_a y_a)^2 + \alpha^2 y_0^2} \left(\frac{\lambda}{\alpha} y_0 \frac{\alpha_a y_a}{(\alpha_a y_a)^2 + \alpha^2 y_0^2} - \frac{\lambda}{4\alpha} \arctan \frac{\alpha_a y_a}{\alpha y_0} + c_2 \right)^{-1},$$
(3.2.26)

If $\varphi = \varphi(\omega_3)$, then Equation (3.2.8(3)) is reduced to the ODE

$$b_3^2\omega_3^2\varphi_{33} + b_3(b_3+3)\varphi_3 - \lambda b_3\omega_3\varphi\varphi_3 + 2\varphi - \lambda\varphi^2 = 0$$

which, in turn, by substitution

$$\omega_3 = z^{b_3} \tag{3.2.27}$$

is reduced to Equation (3.2.19). So we can use formulae (3.2.20), (3.2.19), (3.2.27) and then ansatz N3 (3.2.7) to obtain solutions of Equation (3.2.4)

$$u(x) = \frac{c_1}{(\beta_a y_a)^{1/b_3}} \operatorname{th} \left(\frac{\lambda c_1}{2} \frac{y_0}{(\beta_a y_a)^{1/b_3}} + c_2\right),$$

$$u(x) = \frac{c_1}{(\beta_a y_a)^{1/b_3}} \operatorname{cth} \left(\frac{\lambda c_1}{2} \frac{y_0}{(\beta_a y_a)^{1/b_3}} + c_2\right),$$

$$u(x) = \frac{-c_1}{(\beta_a y_a)^{1/b_3}} \operatorname{tan} \left(\frac{\lambda c_1}{2} \frac{y_0}{(\beta_a y_a)^{1/b_3}} + c_2\right),$$

$$u(x) = \left(c_2(\beta_a y_a)^{1/b_3} + \frac{\lambda}{y_0}\right)^{-1}.$$

(3.2.28)

If $\varphi = \varphi(\omega_3)$, then Equation (3.2.8(4)) is reduced to the ODE

$$b_4^2\varphi_{33} + 3b_4\varphi_3 - \lambda b_4\varphi\varphi_3 + 2\varphi - \lambda\varphi^2 = 0,$$

which via the substitution

$$w_3 = \ln t \tag{3.2.29}$$

is reduced to the Equation (3.2.21). Using the general solution of Equation (3.2.21) and formulae (3.2.29) and ansatz N4 from (3.2.7) we get solutions of Equation (3.2.4)

$$u(x) = c_1 \exp\left\{\frac{-\delta_a x_a}{b_4}\right\} \operatorname{th}\left(\frac{\lambda c_1}{2} y_0 \exp\left\{\frac{-\delta_a x_a}{b_4}\right\} + c_2\right),$$

$$u(x) = c_1 \exp\left\{\frac{-\delta_a x_a}{b_4}\right\} \operatorname{cth}\left(\frac{\lambda c_1}{2} y_0 \exp\left\{\frac{-\delta_a x_a}{b_4}\right\} + c_2\right),$$

$$u(x) = -c_1 \exp\left\{\frac{-\delta_a x_a}{b_4}\right\} \operatorname{tan}\left(\frac{\lambda c_1}{2} y_0 \exp\left\{\frac{-\delta_a x_a}{b_4}\right\} + c_2\right),$$

$$u(x) = \left(c_2 \exp\left\{\frac{\delta_a x_a}{b_4}\right\} + \frac{\lambda}{2} y_0\right)^{-1}.$$
(3.2.30)

Below we list some more particular solutions of Equation (3.2.4) which are succeeded in finding solutions of the rest of Equations (3.2.8):

$$u(x) = c_{1} \operatorname{th} \left(\frac{\mu c_{1}}{2} \beta_{\nu} x^{\nu} + c_{2} \right)$$

$$u(x) = c_{1} \operatorname{cth} \left(\frac{\mu c_{1}}{2} \beta_{\nu} x^{\nu} + c_{2} \right)$$

$$u(x) = -c_{1} \operatorname{tan} \left(\frac{\mu c_{1}}{2} \beta_{\nu} x^{\nu} + c_{2} \right)$$

$$u(x) = \left(\frac{\mu c_{1}}{2} \beta_{\nu} x^{\nu} + c_{2} \right)^{-1}$$

(3.2.31)

where c_1 , c_2 are arbitrary constants $\mu = \frac{\lambda b_0}{b_5}$, $\beta_{\nu}\beta^{\nu} \neq 0$. ((3.2.7) N5)

$$u(x) = c_{1} \operatorname{th} \left[\frac{\lambda c_{1}}{2} (x_{0} + b_{6} \ln \sigma_{a} z_{a}) + c_{2} \right],$$

$$u(x) = c_{1} \operatorname{cth} \left[\frac{\lambda c_{1}}{2} (x_{0} + b_{6} \ln \sigma_{a} z_{a}) + c_{2} \right],$$

$$u(x) = -c_{1} \tan \left[\frac{\lambda c_{1}}{2} (x_{0} + b_{6} \ln \sigma_{a} z_{a}) + c_{2} \right],$$

$$u(x) = \left[\frac{\lambda}{2} (x_{0} + b_{6} \ln \sigma_{a} z_{a}) + c_{2} \right]^{-1},$$

(3.2.32)

where c_1 , c_2 are arbitrary constants, $\sigma_a^2 = 0$ ((3.2.7) N6). Note that functions Chapter 3. Euclid and Galilei Groups and Nonlinear PDEs for Scalar Fields

$$\begin{split} u(x) &= F(\beta_a y_a) \operatorname{th} \left(\frac{\lambda y_0}{2} F(\beta_a y_a) + c_2 \right), \\ u(x) &= F(\beta_a y_a) \operatorname{cth} \left(\frac{\lambda y_0}{2} F(\beta_a y_a) + c_2 \right), \\ u(x) &= -F(\beta_a y_a) \operatorname{tan} \left(\frac{\lambda y_0}{2} F(\beta_a y_a) + c_2 \right), \\ u(x) &= \left(F(\beta_a y_a) + \frac{\lambda y_0}{2} \right), \end{split}$$
(3.2.33)

which are generalizations of solutions (3.2.28), (3.2.30) also satisfy Equation (3.2.4).

One more generalization. If in (3.2.32) $w_3 = \sigma_a z_a$ (see N6, Table 3.2.1) change into arbitrary differentiable function $G(\sigma_a z_a)$, then we obtain from (3.2.32) new family of solutions of Equation (3.2.4):

$$u(x) = c_{1} \operatorname{th} \left[\frac{\lambda c_{1}}{2} (x_{0} + G(\sigma_{a} z_{a})) + c_{2} \right],$$

$$u(x) = c_{1} \operatorname{cth} \left[\frac{\lambda c_{1}}{2} (x_{0} + G(\sigma_{a} z_{a})) + c_{2} \right],$$

$$u(x) = -c_{1} \operatorname{tan} \left[\frac{\lambda c_{1}}{2} (x_{0} + G(\sigma_{a} z_{a})) + c_{2} \right],$$

$$u(x) = \left[\frac{\lambda}{2} (x_{0} + G(\sigma_{a} z_{a})) + c_{2} \right]^{-1},$$

(3.2.34)

where $\sigma_a \sigma_a = 0$.

If $\varphi_{\omega_3} = 0$, then (3.2.8(5)) is reduced to PDE

$$\beta\varphi_{11} + \omega_3\varphi_{22} - 4\varphi_2 + \lambda\beta_0\varphi\varphi_1 = 0. \tag{3.2.35}$$

Using the substitution

$$y_0 = \omega_1, \qquad y_1 = 2\sqrt{-\beta\omega_2}$$
 (3.2.36)

we transform Equation (3.2.35) to the canonical form

$$\varphi_{00} - \varphi_{11} - \frac{\varphi_1}{y_1} + \frac{\lambda\beta_0}{\beta}\varphi\varphi_0 = 0, \qquad (3.2.37)$$

which is called the nonlinear Darboux equation.

One can prove with the help of Lie method the following statement.

Theorem 3.2.1. The maximal local invariance group of nonlinear Darboux equation (3.2.37) is 2-parameter Lie group of scale and translation transformations. The basis elements of corresponding Lie algebra are

$$P_0 = i\partial_0, \qquad D = i(y_0\partial_0 + y_1\partial_1 + i).$$

We seek solutions of Equation (3.2.37) in the form

$$\varphi = \Phi(w_1)g(y), \tag{3.2.38}$$

where $w_1 = \frac{z_1^2}{z_0}$, $g(y) = z_0^{-1}$, $z_0 = y_0 + b_0$, $z_1 = y_1$, $b_0 = \text{const}$, Inserting ansatz (3.2.38) into (3.2.37), we obtain ODE, which is reduced after first integration to Riccati one

$$4w_1(w_1 - 1)\Phi_1 + 2w_1\Phi - \frac{\lambda\beta_0}{\beta}w_1\Phi^2 = c, \qquad c = \text{const}$$
(3.2.39)

A particular solution of this equation is easily obtained when c = 0:

$$\Phi = \left[c_1\sqrt{|w_1-1|} + \frac{\lambda\beta_0}{\beta}\right]^{-1}.$$

Corresponding solution of Equation (3.2.35) is constructed by (3.2.38) and it has the form

$$\varphi = \left[c_1 \sqrt{|y_0^2 - (y_0 + b_0)^2|} + \frac{\lambda \beta_0}{\beta} (y_0 + b_0)\right]^{-1}, \qquad (3.2.40)$$

Using formulae (3.2.40), (3.2.36) and ansatz N5 from (3.2.7)) we find solution of Equation (3.2.4)

$$u(x) = \left[c_1 \sqrt{4\beta \left(\frac{(\alpha_a y_a)^2}{-\alpha^2} + y_a y_a\right)^2 + (\beta_\nu x^\nu + b_0)^2} + \frac{\lambda\beta_0}{\beta} (\beta_\nu x^\nu + b_0)\right]^{-1}.$$
(3.2.41)

In conclusion of this point it is worth while to note that above obtained solutions (3.2.15), (3.2.18), (3.2.22), (3.2.28), (3.2.30)-(3.2.34), (3.2.41) of Equation (3.2.4) one can easily generalize to the case of arbitrary number of independent variables.

3. Let us consider equation

$$\Box u + \lambda e^u u_0 = 0. \tag{3.2.42}$$

Since invariance algebra of Equation (3.2.42) is given by operators (3.1.2), 2° (3.1.8), its solutions we seek in the form [91]

$$u(x) = \begin{cases} \varphi(\omega) - \ln y_0, & \text{for } \omega \text{ N1-4} \\ \varphi(\omega), & \text{for } \omega \text{ N5-10} \end{cases}$$
(3.2.43)

 $+ 2\omega_1\varphi_1 + 6(\omega_2 - 1)\varphi_2 - \lambda(\omega_1\varphi_1 + 2\omega_2\varphi_2 + 1)e^{\varphi} + 1 = 0;$

Ansatz (3.2.43) is obtained from (3.2.5), (3.2.7) with the help of transformation (1.6.4).

Below we present the first six reduced equations which are obtained as a result of substitution of ansatz (3.2.43) into Equation (3.2.42)

(1)
$$(w_1^2 - \alpha^2)\varphi_{11} + 4w_2(w_2 - 1)\varphi_{22} + w_3^2 \left(1 - \frac{\alpha^2 b_1^2}{\alpha^2 w_2 - \omega_1^2}\right)\varphi_{33} + 4\omega_1(\omega_2 - 1)\varphi_{12} - 4\omega_2\omega_3\varphi_{23} + 6(\omega_2 - 1)\varphi_2 + 2\omega_1\varphi_1 - \frac{\alpha^2 b_1^2\omega_3}{\alpha^2\omega_2 - \omega_1^2}\varphi_3 + \lambda(-\omega_1\varphi_1 + 2\omega_2\varphi_2 + \omega_3\varphi_3 - 1)e^{\varphi} + 1 = 0;$$

(2)
$$\omega_1^2 \varphi_{11} + 4\omega_2(\omega_2 - 1)\varphi_{22} + \frac{b_2}{\omega_1^2}\varphi_{33} + 4\omega_1(\omega_2 - 1)\varphi_{12} - 4b_2\varphi_{23} +$$

(3)
$$(\omega_1^2 + \alpha^2)\varphi_{11} + 4\omega_2(w_2 - 1)\varphi_{22} + b_3^2\omega_3^2\varphi_{33} + 4\omega_1(\omega_2 - 1)\varphi_{12} + 2b_3\omega_1\omega_3\varphi_{13} + 4\omega_3(b_3w_2 - 1)\varphi_{23} + 2\omega_1\varphi_1 + 6(\omega_2 - 1)\varphi_2 +$$

$$+ b_3(b_3 + 1)\omega_3\varphi_3 - \lambda(\omega_1\varphi_1 + 2\omega_2\varphi_2 + b_3\omega_3\varphi_3 + 1)e^{\varphi} + 1 = 0; \quad (3.2.44)$$

(4)
$$(\omega_1^2 + \alpha^2)\varphi_{11} + 4\omega_2^2\varphi_{22} + b_4^2\varphi_{33} + 4\omega_1\omega_2\varphi_{12} + 2b_4\omega_1\varphi_{13} + 4b_4\omega_2\varphi_{23} + 2\omega_1\varphi_1 + 6\omega_2\varphi_2 + b_4\varphi_3 - \lambda(\omega_1\varphi_1 + 2\omega_2\varphi_2 + b_4\varphi_3 + 1)e^{\varphi} + 1 = 0;$$

(5)
$$\beta \varphi_{11} + \omega_2 \varphi_{22} + \left(1 - \frac{b_5}{\omega_2}\right) \varphi_{33} + 2\beta_0 \varphi_{13} - 4\varphi_2 + \lambda (\beta_0 \varphi_1 + \varphi_3) e^{\varphi} = 0;$$

(6)
$$\beta \varphi_{11} + \omega_2 \varphi_{22} + 2\beta_0 \varphi_{13} + \varphi_{33} - 4b_6 \varphi_{23} - 4\varphi_2 + \lambda (\beta_0 \varphi_1 + \varphi_3) e^{\varphi} = 0.$$

Now we put $\varphi = \varphi(\omega_1)$, then (3.2.44(1)) is reduced to the ODE

$$(\omega_1^2 - \alpha^2)\varphi_{11} - 2\omega_1\varphi_1 - \lambda(\omega_1\varphi_1 + 1)e^{\varphi} + 1 = 0, \qquad (3.2.45)$$

which after first integration takes the form

$$(\omega_1^2 - \alpha^2)\varphi_1 - \lambda\omega_1 e^{\varphi} = -\omega_1 + c_1, \qquad c_1 = \text{const}$$

This latter equation is reduced by the substitution

$$\varphi = \ln v \tag{3.2.46}$$
to the Bernoulli equation

$$(\omega_1^2 - \alpha^2)v' + (\omega_1 - c_1)v - \lambda\omega_1 v^2 = 0,$$

which is easily integrated and has the general solution as follows

$$v = \left[(\omega_1 - \alpha)^{\frac{\alpha - c_1}{2\alpha}} (\omega_1 + \alpha)^{\frac{\alpha + c_1}{2\alpha}} \int \frac{-\lambda \omega_1 d\omega_1}{(w_1 - \alpha)^{(3\alpha - c_2)/(2\alpha)} (\omega_1 + \alpha)^{(3\alpha + c_2)/(2\alpha)}} \right]^{-1}$$
(3.2.47)

We get from (3.2.47), (3.2.46) two families of solutions (depending upon constant c_1) of Equation (3.2.45)

$$\varphi = -\ln\left(\frac{\lambda}{4\alpha}(\omega_1 \pm \alpha)\ln\left|c_2\frac{\omega_1 + \alpha}{\omega_1 - \alpha}\right| + \frac{\lambda}{2}\right), \qquad c_1 = \pm\alpha;$$

$$\varphi = -\ln\left(c_2\sqrt{\omega_1^2 - \alpha^2} + \lambda\right), \qquad c_1 = 0.$$

(3.2.48)

In turn, formulae (3.2.48), (3.2.43) give solutions of Equation (3.2.42)

$$u(x) = -\ln\left(\frac{\lambda}{4\alpha}(\alpha_a y_a \pm \alpha y_a)\ln\left|c_2\frac{\alpha_a y_a + \alpha y_a}{\alpha_a y_a - \alpha y_0}\right| + \frac{\lambda}{2}\right),$$

$$u(x) = -\ln\left(c_2\sqrt{(\alpha_a y_a)^2 - \alpha^2 y_0^2} + \lambda\right), \qquad c_1 = 0.$$
 (3.2.49)

Below we present some more solutions of Equation (3.2.42), which were obtained in much the same way when considering the rest of reduced Equations (2)-(6) from (3.2.44)

$$\begin{split} u(x) &= -\ln \left[\frac{1}{c_1} (\beta_a y_a)^{1/b_3} (c_2 \exp\left\{ \frac{c_1 y_0}{(\beta_a y_a)^{1/b_3}} \right\} - \lambda \right], \quad \beta_a \beta_a = 0; \\ u(x) &= -\ln \left[c_2 \sqrt{(\alpha_a y_a)^2 + \alpha^2 y_0^2} \exp\left\{ -\frac{c_1}{\alpha} \arctan \frac{\alpha_a y_a}{a y_0} \right\} - \right. \\ &- \frac{\lambda c_1}{c_1^2 + \alpha^2} \left(\alpha_a y_a - \frac{\alpha^2}{c_1} y_0 \right) \right], \qquad \alpha^2 < 0; \\ u(x) &= -\ln \left[c_2 \exp\left\{ -c_1 \beta_\nu x^\nu \right\} + \frac{\lambda \beta_0}{\beta c_1} \right], \qquad \beta = \beta_\nu \beta^\nu \neq 0; \\ u(x) &= -\ln \left[\frac{1}{c_1} \exp\left\{ \frac{\delta_a x_a}{b_4} \right\} \left(c_2 \exp\left\{ \frac{c_1 b_4 y_0}{\delta_a x_a} \right\} - \lambda \right) \right], \qquad \delta_a \delta_a = 0; \\ u(x) &= -\ln \left[c_2 (\sigma_a z_a)^{-c_1 b_6} \exp\left\{ -c_1 x_0 \right\} + \frac{\lambda}{c_1} \right], \qquad \sigma_a \sigma_a = 0; \end{split}$$
(3.2.50)

Chapter 3. Euclid and Galilei Groups and Nonlinear PDEs for Scalar Fields

$$u(x) = -\ln \left[F(\beta_a y_a)(c_2 \exp\left\{\frac{y_0}{F(\beta_a y_a)}\right\} - \lambda) \right], \qquad \beta_a \beta_a = 0;$$
$$u(x) = -\ln \left[c_2 \left(G(\sigma_a z_a) e^{x_0} \right)^{-c_1} + \frac{\lambda}{c_1} \right], \qquad \sigma_a \sigma_a = 0.$$

where F, G are arbitrary differentiable functions, β_a , α_a , δ_a , σ_a , c_1 , c_2 are constants, satisfying conditions stated above.

4. Let us consider once more $\widetilde{E}(1,3)$ -invariant nonlinear PDEs

$$\Box u + \lambda \sqrt{u_a u_a} u_0 = 0. \tag{3.2.51}$$

As for as invariance algebra of Equation (3.2.51) is given by operators (3.1.3), we seek its solutions in the form [91]

$$u(x) = \varphi(\omega) \tag{3.2.52}$$

where new variables are written in Table 3.2.1.

Omitting cumbersome calculations connected with substitution of ansatze (3.2.52) into Equation (3.2.51), we present the final result: solutions of Equation (3.2.51)

$$\begin{split} u(x) &= \frac{1}{ac_1} \arctan \frac{\alpha_a y_a}{ay_0} + c_2, \quad -\alpha^2 + \frac{\alpha \lambda}{2c_1} = a^2 > 0; \quad (3.2.53) \\ u(x) &= \frac{1}{2ac_1} \ln \left| c_2 \frac{\alpha_a y_a - ay_0}{\alpha_a y_a + ay_0} \right|, \quad -\alpha^2 + \frac{\lambda \alpha}{2c_1} = -a^2 < 0, \quad \alpha_a^2 = \alpha^2 \neq 0; \\ u(x) &= \frac{2}{\lambda} \int \frac{dt}{\ln ((1 - t^2)/c_1)}, \quad t = \frac{y_0}{\sqrt{y_a y_a}}; \\ u(x) &= b \ln (\beta_a y_a)^{1/b_3} - \frac{c_1 y_0}{(\beta_a y_a)^{1/b_3}} + c_2, \qquad \beta_a \beta_a = 0; \end{split}$$

$$u(x) = c_1 y_0 \exp\left\{-\frac{\delta_a x_a}{b_4}\right\}, \qquad \delta_a \delta_a = 0; \qquad (3.2.54)$$

$$\begin{split} u(x) &= \frac{-\beta^2}{\sqrt{\beta_a \beta_a} \lambda \beta_0} \ln \left| c_1 \exp\{\beta_0 x_0\} + c_2 \exp\{\beta_a x_a\} \right|, \qquad \beta_\nu \beta^\nu = \beta \neq 0; \\ u(x) &= F(\beta_a y_a) + G(\beta_a y_a) x_0, \qquad \beta_a \beta_a = 0, \end{split}$$

where F, G are arbitrary differentiable functions, $\alpha_a, \beta_{\nu}, \delta_a$ are arbitrary real constants satisfying conditions stated.

3.3. PDEs admitting Galilei algebras

One of the best known Galilei-invariant equation is the heat (or diffusion) equation

$$u_0 + \lambda \Delta u = 0, \tag{3.3.1}$$

where $u_0 = \frac{\partial u}{\partial x_0}$, $\Delta u = \frac{\partial^2 u}{\partial x_1^2} + \dots + \frac{\partial^2 u}{\partial x_{n-1}^2}$.

In two-dimensional case the symmetry of Equation (3.3.1) had been studied by S.Lie. This result of Lie is easy generalized on *n*-dimensional case.

Theorem 3.3.1. Maximal (in sense of Lie) invariance algebra of the heat equation (3.3.1) has the following basis elements:

$$P_{0} = i\partial_{0}, \qquad P_{a} = -i\partial_{a}, \qquad I = u\partial_{u},$$

$$J_{ab} = x_{a}P_{b} - x_{b}P_{a}, \qquad G_{a} = x_{0}P_{a} + \frac{i}{2\lambda}x_{a},$$

$$D = 2x_{0}P_{0} - x_{a}P_{a},$$

$$\Pi = x_{0}^{2}P_{0} - x_{0}x_{a}P_{a} - \frac{in}{2}x_{0} + \frac{i}{4\lambda}\vec{x}^{2}.$$

To prove this statement one can use Lie's method (see $\S1.1$).

Unfortunately, most of real phenomena of diffusion and heat transmission are not described satisfactorily with the help of the linear Equation (3.3.1). A well-known nonlinear generalization of Equation (3.3.1)

$$u_0 + \vec{\nabla}(F(u)\vec{\nabla}u) = 0 \tag{3.3.2}$$

has an essential shortcoming: it is Galilei invariant iff F(u) = const, so that there is no one nonlinear equation of the form (3.3.2) which would satisfy relativistic principle of Galilei.

Below we describe Galilei-invariant nonlinear generalizations of the heat equation (3.3.1) and Schrödinger equation (3.3.24).

1. Consider equation

$$u_0 + F(x_0, \vec{x}, u, \eta, \eta) = 0, \qquad (3.3.3)$$

where $u_1 = (u_1, u_2, \ldots, u_{n-1}), u_2 = (u_{11}, u_{12}, \ldots, u_{n-1,n-1})$, are first and second derivatives of function u in spacial variables x_a . If function F does not depend on second derivatives u_2 , so that Equation (3.3.3) is a first-order PDE, then the following statement holds true [181].

Theorem 3.3.2. If equation

$$u_0 + F(x_0, \vec{x}, u, u) = 0, \qquad (3.3.4)$$

is invariant under Galilei algebra AG(1, n - 1) with operators

$$P_0 = i\partial_0, \qquad P_a = -i\partial_a,$$

$$J_{ab} = x_a P_b - x_b P_a, \qquad (3.3.5)$$

$$G_a = x_0 P_a + x_a a(x, u)\partial_u$$

so it is locally equivalent to the Hamilton-Jacobi equation

$$u_0 + \frac{1}{2m} \left(\vec{\nabla} u \right)^2 = 0. \tag{3.3.6}$$

Proof. Coordinates ξ^0 , ξ^a , η of infinitesimal operator

$$X = \xi^0 \partial_0 + \xi^a \partial_a + \eta \partial_u$$

of AG(1, n-1) (3.3.5) have the form

$$\xi^{0} = d_{0}, \qquad \xi^{a} = g_{a}x_{0} + c_{ab}x_{b} + d_{a}, \eta = a(x, u)g_{a}x_{a} + c(x, u), \qquad a, b = \overline{1, n - 1},$$
(3.3.7)

where $d_0, d_a, c_{ab} = -c_{ba}, g_a$ are parameters.

To meet the demands of translation and rotation invariance of Equation (3.3.4), it is obvious that function F should be

$$F(x_0, \vec{x}, u, u_1) = \Phi(u, (\vec{\nabla}u)^2).$$
(3.3.8)

Further, from invariance with respect to operator G_a we obtain

$$\eta_0 = 0 \qquad 2\eta_a \Phi_2 = g_a, 2\eta_u W_2 \Phi_2 + \eta \Phi_1 - \eta_u \Phi = 0,$$
(3.3.9)

where $\Phi_1 = \frac{\partial \Phi}{\partial W_1}$, $\Phi_2 = \frac{\partial \Phi}{\partial W_2}$, $W_1 = u$, $W_2 = (\vec{\nabla}u)^2$. The general solution of the system (2.2.0) has the form

The general solution of the system (3.3.9) has the form

$$\begin{split} \Phi &= A'(u)\frac{1}{2m}(\vec{\nabla}u)^2 + \frac{\lambda_1}{A'(u)},\\ \eta &= \frac{1}{A'(u)}(mg_a x_a + \lambda_2), \end{split}$$

where λ_a, λ_2, m are arbitrary constants, A(u) is an arbitrary differentiable function.

So we see that Equation (3.3.4) to be invariant under AG(1, n - 1) (3.3.5) should have the form

$$u_0 + A'(u)\frac{1}{2m}(\vec{\nabla}u)^2 + \frac{\lambda_1}{A'(u)} = 0.$$
 (3.3.10)

It is easy to show that Equation (3.3.10) is equivalent to the Hamilton-Jacobi equation (3.3.6) (to do it one has to perform the transformation $A(u) + \lambda_1 x_0 \rightarrow u$), so the theorem is proved.

Remark 3.3.1. The theorem yields that operators $G_a \subset AG(1, n-1)$ have the form

$$G_a = x_0 P_a - im x_a \partial_u. \tag{3.3.11}$$

Below we use the following notations. Consider matrix A constructed from second derivatives u_{ab} of function u

$$A = \begin{pmatrix} u_{11} & u_{12} & \dots & u_{1n} \\ u_{21} & u_{22} & \dots & u_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ u_{n1} & u_{n2} & \dots & u_{nn} \end{pmatrix}$$

Let symbol $\langle K \rangle$ denotes the sum of various minors of order k of main diagonal of the matrix A, so that

$$<1> = u_{11} + u_{22} + \dots + u_{nn},$$

$$<2> = \begin{vmatrix} u_{11} & u_{12} \\ u_{21} & u_{22} \end{vmatrix} + \begin{vmatrix} u_{11} & u_{13} \\ u_{31} & u_{33} \end{vmatrix} + \dots + \begin{vmatrix} u_{n-1n-1} & u_{n-1n} \\ u_{nn-1} & u_{nn} \end{vmatrix},$$

$$\ldots$$

$$= \begin{vmatrix} u_{11} & u_{12} & \dots & u_{1n} \\ u_{21} & u_{22} & \dots & u_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ u_{n1} & u_{n2} & \dots & u_{nn} \end{vmatrix} = \det A$$

Galilei algebra AG(1,n) (3.3.5), (3.3.11) extended by generator of scale transformation

$$D = 2x_0 P_0 - x_a P_a \tag{3.3.12}$$

and generator of projective transformation

$$\Pi = x_0^2 P_0 - x_0 x_a P_a + \frac{im}{2} \vec{x}^2 \partial_u$$
(3.3.13)

we shall call extended Galilei algebra (denote $A\widetilde{G}(1,n) = \{AG(1,n), D\}$) and Schrödinger algebra (denote $ASch(1,n) = \{A\widetilde{G}(1,n), \Pi\}$).

Theorem 3.3.3. Eqn. (3.3.3) is invariant with respect to algebras AG(1,n), $A\widetilde{G}(1,n)$, ASch(1,n), iff it has the form

$$u_0 + \frac{1}{2m} \left(\vec{\nabla} u \right)^2 + \Phi(\langle 1 \rangle, \langle 2 \rangle, \dots, \langle n \rangle) = 0, \qquad (3.3.14)$$

$$u_0 + \frac{1}{2m} (\vec{\nabla} u)^2 + \Delta u \varphi(W_2, W_3, \dots, W_n) = 0, \qquad (3.3.15)$$

$$u_0 + \frac{1}{2m} (\vec{\nabla} u)^2 + \sqrt{(\Delta u)^2 + \frac{2n}{1-n}} < 2 > \psi(J_3, J_4, \dots, J_n) = 0, \quad (3.3.16)$$

where Φ, φ, ψ are arbitrary differentiable functions of corresponding arguments, $W_k = \frac{\langle k \rangle}{[\langle 1 \rangle]^k}, \quad k = \overline{2, n}, J_k, k = \overline{3, n} \text{ are first integrals of the ODE}$ $\frac{dW_2}{2nW_2 - (n-1)} = \frac{dW_3}{3nW_3 - (n-1)W_2} = \dots = \frac{dW_k}{nkW_k - (n-1)W_{k-1}} =$ $= \dots \frac{dW_n}{n^2W_n - (n-1)W_{n-1}}$

Proof. From condition of invariance of Equation (3.3.3) under AG(1,n) it follows that function F should satisfy the following system of PDEs

$$d_0 \frac{\partial F}{\partial x_0} = d_a \frac{\partial F}{\partial x_a} = d_{n+1} \frac{\partial F}{\partial u} = 0, \qquad (3.3.17)$$

$$g_a \left(m \frac{\partial F}{\partial u_a} - u_a \right) = 0, \tag{3.3.18}$$

$$c_{ab}u_b\frac{\partial F}{\partial u_a} + (c_{bc}u_{ac} + c_{ac}u_{bc})\frac{\partial F}{\partial u_{ab}} = 0, \qquad (3.3.19)$$

$$x \left(2u_{ab} \frac{\partial F}{\partial u_{ab}} + u_a \frac{\partial F}{\partial u_a} - 2F \right) = 0,$$
 (3.3.20)

$$a\delta_{ab}\frac{\partial F}{\partial u_{ab}} = 0, \tag{3.3.21}$$

where $d_0, d_a, d_{n+1}, g_a, \mathfrak{X}, c_{ab} = -c_{ba}$ are group parameters, δ_{ab} is the Kronecker symbol.

Function F, as it follows from (3.3.17), does not depend on variables x_0, x_a, u . Equation (3.3.18) yields

$$F = \frac{1}{2m} \left(\vec{\nabla} u \right)^2 + \Phi(\underline{u})$$
 (3.3.22)

Substituting (3.3.22) into (3.3.18)-(3.3.21) we get

$$(c_{ab}u_{ac} + c_{ac}u_{bc})\frac{\partial\Phi}{\partial u_{ab}} = 0,$$

$$\approx \left(u_{ab}\frac{\partial\Phi}{\partial u_{ab}} - \Phi\right) = 0,$$

$$a\delta_{ab}\frac{\partial\Phi}{\partial u_{ab}} = 0.$$
(3.3.23)

After solving Equations (3.3.23) we obtain formulae (3.3.14)-(3.3.16). The theorem is proved.

Remark 3.3.2. Equations (3.3.14)-(3.3.16) are nonlinear generalizations of Equation (3.3.1). It is obvious to make the transformation

$$u(x) \to \exp\left\{\frac{u}{2\lambda m}\right\}$$

which allows us to pass from Equation (3.3.1) to the equivalent one

$$u_0 + \frac{1}{2m} \left(\vec{\nabla} u \right)^2 + \lambda \Delta u = 0.$$

2. The Schrödinger equation for complex function $u(x_0, x)$ has the form

$$\left(P_0 + \frac{\vec{P}^2}{2m}\right)u = 0, \qquad (3.3.24)$$

where $P_0 = i\partial_0$, $\vec{P} = -i\vec{\nabla}$.

It is well-known that maximal Lie-invariant algebra of the Schrödinger equation is ASch(1, n), basis elements having the form

$$X_{0} = i\partial_{0}, \qquad X_{a} = -i\partial_{a}, \qquad I = i(u\partial_{u} - u^{*}\partial_{u}),$$

$$J_{ab} = x_{a}\partial_{b} - x_{b}\partial_{a}, \qquad G_{a} = x_{0}\partial_{a} - mx_{a}I$$

$$D = 2x_{0}\partial_{0} + x_{a}\partial_{a} - \frac{n}{2}(u\partial_{u} + u^{*}\partial_{u^{*}}),$$

$$\Pi = x_{0}^{2}\partial_{0} + x_{0}x_{a}\partial_{a} - \frac{m\vec{x}^{2}}{2}I - \frac{n}{2}x_{0}(u\partial_{u} + u^{*}\partial_{u^{*}}).$$
(3.3.25)

Invariance algebra ASch(1,3) of three-dimensional Equation (3.3.24) has been established in [157].

Consider a quasilinear generalization of Equation (3.3.24)

$$iu_0 + f^{ab}(x_0, \vec{x}, u, u^*)u_{ab} + F(x_0, \vec{x}, u, u^*, u, u^*) = 0, \qquad (3.3.26)$$

where u^* is the complex conjugate to u.

The solution of the problem analogous to that considered above is the following statement.

Theorem 3.3.4. Equation (3.3.24) is invariant under the algebras AG(1, n), $A\widetilde{G}(1,n)$, ASch(1,n) iff it has the form

$$\left(P_0 + \frac{\vec{P}^2}{2m}\right)u + f(|u|, (\vec{\nabla}|u|)^2)u = 0, \qquad (3.3.27)$$

$$\left(P_0 + \frac{\vec{P}^2}{2m}\right)u + |u|^{-2/k}F_1\left[|u|^{-2+2/k}(\vec{\nabla}|u|)^2\right]u = 0,$$
(3.3.28)

$$\left(P_0 + \frac{\vec{P}^2}{2m}\right)u + \left(\vec{\nabla}|u|\right)^2 F_2(|u|)u = 0, \qquad (3.3.29)$$

$$\left(P_0 + \frac{\vec{P}^2}{2m}\right)u + |u|^{4/n}\varphi\left(|u|^{-2-4/n}\left(\vec{\nabla}|u|\right)^2\right)u = 0, \qquad (3.3.30)$$

where f, F_1, F_2, φ are arbitrary differentiable functions of corresponding arguments.

One can obtain the proof of the theorem by proceeding in much the same way as in previous case (see Theorem 3.3.3).

Theorem 3.3.5. The nonlinear Schrödinger equation

$$\left(P_0 + \frac{\vec{P}^2}{2m}\right)u = F(u, u^*)$$
(3.3.31)

is invariant under AG(1, n), $A\widetilde{G}(1, n)$, ASch(1, n) if it has the form

$$\left(P_0 + \frac{\vec{P}^2}{2m}\right)u = F(|u|)u$$
 (3.3.32)

$$\left(P_0 + \frac{\vec{P}^2}{2m}\right)u = \lambda_1 |u|^k u \tag{3.3.33}$$

$$\left(P_0 + \frac{\vec{P}^2}{2m}\right)u = \lambda |u|^{4/n}u \qquad (3.3.34)$$

respectively. The nonlinear equation

$$\left(P_0 + \frac{\vec{P}^2}{2m}\right)u = \left(\varphi(|u|) + i\ln u\right)u \tag{3.3.35}$$

is invariant under 11-dimensional Lie algebra with basis elements

$$P_{0}, P_{a}, J_{ab}, G_{a} = e^{x_{0}}P_{a} + mx_{a}I, I = e^{x_{0}}(u\partial_{u} - u^{*}\partial_{u^{*}})$$
(3.3.36)

Theorem 3.3.6 [82*]. The nonlinear Schrödinger equation for a particle with variable mass

$$\left(P_0 + \frac{\vec{P}^2}{2M(x_0)}\right)u = \frac{M'}{M}(\varphi(x_0, |u|) + i\ln u)u$$
(3.3.37)

is invariant under the 10-dimensional Lie algebra with basis elements

$$P_{a}, J_{ab}, G_{a} = x_{0}P_{a} + M(x_{0})x_{a}I,$$

$$E = M(x_{0})I.$$
(3.3.38)

3.4. The Galilean relativistic principle and nonlinear PDEs

We shall describe, following [71], nonlinear PDEs which are invariant with respect to the Galilean transformations

$$t \to t' = t, \qquad x_a \to x'_a = x_a + v_a t$$

$$(3.4.1)$$

We consider two essentially different representations of Galilean transformations: the so-called projective Galilean transformations (PGT), when dependent variable is also transformed together with independent variables (for example, the heat equation (3.3.1) is invariant under transformations (3.4.1) iff $u(x) \rightarrow u'(x') = \exp\{-\frac{1}{2}v_a(x_a + v_at)\}u(x)\}$, and Galilean transformations (GT), when the dependent variable remains unchanged.

Theorem 3.4.1. Equation

$$F(t, x, u, \underbrace{u}_{1}, \underbrace{u}_{2}) \equiv -\Delta u + A(t, x, u)u_{t} + B(t, x, u, \underbrace{u}_{1})$$
(3.4.2)

where u = u(t, x), $x \in \mathbb{R}^n$ $u_1 = \left\{\frac{\partial u}{\partial x_a}\right\}$, $u_2 = \left\{\frac{\partial^2 u}{\partial x_a \partial x_b}\right\}$; $a, b = \overline{1, n}$;

A, B are differentiable functions, is invariant under PGT iff

$$A(t, x, u) = f(t, w)$$
 (3.4.3)

Chapter 3. Euclid and Galilei Groups and Nonlinear PDEs for Scalar Fields

$$B(t, x, u, u) = ug(t, w_1, \dots, w_n) + (f(t, w) - \lambda) \left(\frac{x_a u_a}{t} + \frac{\lambda |x|^2}{4t^2}u\right), \quad (3.4.4)$$

where

$$w = u \exp\left\{\frac{\lambda |x|^2}{4t}\right\}, \qquad |x|^2 \equiv x_a x_a, \qquad (3.4.5)$$

$$w_a = \left(u_a + \frac{\lambda}{2t} x_a u\right) \exp\left\{\frac{\lambda |x|^2}{4t}\right\}$$
(3.4.6)

f, g are arbitrary differentiable functions.

Proof. According to the Lie method, from condition of invariance

$$\left. \begin{array}{c} XF \\ 2 \end{array} \right|_{F=0} = 0,$$

where operator X is constructed by formulae (1.1.7), and

$$\xi^0 = 0, \quad \xi^a = g^a t, \quad \eta = -\frac{1}{2}\lambda g_a x_a u$$

we obtain system to define functions A and B

$$t\frac{\partial A}{\partial x_a} - \frac{1}{2}\lambda x_a u\partial A u = 0, \qquad (3.4.7)$$
$$\frac{2}{\lambda}t\frac{\partial B}{\partial x_a} - x_a u\frac{\partial B}{\partial u} - u\frac{\partial B}{\partial u_a} - x_a u_b\frac{\partial B}{\partial u_b} + x_a B - \frac{2}{\lambda}u_a(A-\lambda) = 0, \qquad (3.4.8)$$

Rewriting (3.4.7) in equivalent form

$$\frac{dx_a}{t} = \frac{du}{-\frac{1}{2}\lambda x_a u}$$

we get invariants

$$w_0 = t, \qquad w = u \exp\left\{\frac{\lambda |x|^2}{4t}\right\},$$

which give the general solution of Equation (3.4.7) (see formula (3.4.3)).

Analogously, passing from (3.4.8) to corresponding characteristic system

$$-\frac{dx_{a}}{(2/\lambda)t} = \frac{du}{x_{a}u} = \frac{du_{a}}{u + x_{a}u_{a}} = \frac{du_{1}}{x_{a}u_{1}} = \dots = \frac{du_{a-1}}{x_{a}u_{a-1}} = \frac{du_{a+1}}{x_{a}u_{a+1}} = \dots =$$
$$= \frac{du_{n}}{x_{a}u_{n}} = \frac{dB}{x_{a}B + (2/\lambda)(\lambda - f(w_{0}, w))}, \quad a = \overline{1, n}$$
(3.4.9)

(there is no sum over repeated indices) we find the invariants

$$w = u \exp\left\{\frac{\lambda |x|^2}{4t}\right\}, \quad w_0 = t$$

$$w_a = \left(u_a + \frac{\lambda x_a}{2t}u\right) \exp\left\{\frac{\lambda |x|^2}{4t}\right\}, \quad (3.4.10)$$

$$I = \left[B + \left(\lambda - f(w_0, w)\right) \left(\frac{x_a u_a}{t} + \frac{\lambda |x|^2}{4t^2}u\right)\right] \exp\left\{\frac{\lambda |x|^2}{4t}\right\}$$

and then, from the functional equation

$$\phi(w, w_0, w_1, \ldots, w_n, I) = 0$$

function B as in (3.4.4). The theorem is proved.

Consequence 3.4.1. Suppose the coefficient B in (3.4.2) to be independent on ψ , then equation

$$\Delta u = \lambda u_0 + ug(w, t) \tag{3.4.11}$$

is the most general one, invariant under PGT, g being arbitrary differentiable function.

Theorem 3.4.2. Equation (3.4.2) is invariant under the algebra of operators

$$G_{a} = t\partial_{a} - \frac{1}{2}\lambda x_{a}u\partial_{u}, \qquad \left(\partial_{a} \equiv \frac{\partial}{\partial x_{a}}, \quad \partial_{u} \equiv \frac{\partial}{\partial u}\right)$$
$$J_{ab} = x_{a}\partial_{b} - x_{b}\partial_{a} \qquad (3.4.12)$$

iff it has the form

$$\Delta u = f(w,t)u_t + ug(w, w_a w_a, t) + (f(w,t) - \lambda) \left(\frac{x_a x_a}{t} + \frac{\lambda |x|^2}{4t}u\right), \quad (3.4.13)$$

where

$$w_a w_a = \left[u_a u_a + \lambda x_a u_a \frac{u}{t} + \left(\frac{\lambda |x| u}{2t} \right)^2 \right] \exp\left\{ \frac{\lambda |x|^2}{2t} \right\}.$$

This theorem is proved in the same way as the first one.

It should be noted that Equation (3.4.11) can be obtained from (3.4.13) when function B in (3.4.2) is independent of u_{1} . Invariance under PGT automatically implies invariance under rotation group.

The futher restriction of the class of Equations (3.4.11) is achieved by the requirement for the equations to be invariant under the operator of scale transformations

$$D = 2t\partial_t + x_a\partial_a + ku\partial_u, \qquad k = \text{const}$$
(3.4.14)

172 Chapter 3. Euclid and Galilei Groups and Nonlinear PDEs for Scalar Fields and under the projective operator

$$\Pi = t^2 \partial_t + t x_a \partial_a - \left(\frac{1}{4}\lambda |x|^2 + \frac{1}{2}nt\right) u \partial_u, \qquad (3.4.15)$$

The following two theorems are proved in much the same way as previous ones.

Theorem 3.4.3. Among Equations (3.4.11) the only one

$$\Delta u = \lambda u_t + \frac{u}{t^2} g\left(t^{n/2} w\right) \tag{3.4.16}$$

admits the operator Π (3.4.15), g being an arbitrary differentiable function.

Theorem 3.4.4. Among Equations (3.4.11) the only one

$$\Delta u = \lambda u_t + \lambda_1 u t^{-2} \left(\frac{u}{\epsilon(t,x)}\right)^{\beta}, \qquad t^{n/2} w = \frac{u}{\epsilon} \cdot \text{const}$$
(3.4.17)

admits operators D (3.4.14) with $k = 2/\beta - n$ and Π (3.4.15),

$$\epsilon = \epsilon(t, x) = \left(\frac{\lambda}{\sqrt{2\pi}t}\right)^{n/2} \exp\left\{-\frac{\lambda|x|^2}{4t}\right\}$$
(3.4.18)

being the fundamental solution of the heat equation

$$\Delta v = \lambda v_t. \tag{3.4.19}$$

Note 3.4.1. Setting $\beta = 0$ in (3.4.17), we obtain equation

$$\Delta u = \lambda u_t + \lambda_1 t^{-2} u$$

which is reduced to (3.4.19) by means of local substitution

$$u = v(t, x) \exp\left\{\frac{\lambda_1}{\lambda t}\right\}, \qquad \lambda \neq 0.$$

Note 3.4.2. All equations considered above contain (explicitly or implicitly) the fundamental solution (3.4.18) of the heat equation (3.4.19). This is apparently because $\epsilon(t, x)$ is the general solution of the system

$$\Delta u = \lambda u_t,$$

$$G_a u \equiv t u_a + \frac{1}{2} \lambda x_a u = 0, \qquad a = \overline{1, n}$$
(3.4.20)

2. Let us consider equations of the form

$$u_t = C(t, x, u)\Delta u + K\left(t, x, u, \frac{u}{l}\right)$$
(3.4.21)

where C and K are differentiable functions, and find out conditions which ensure invariance of (3.4.21) under GT generated by operators

$$\widetilde{G}_a = t\partial_a \tag{3.4.22}$$

Theorem 3.4.5. Equation (3.4.21) admits operators (3.4.22) iff

$$C(t, x, u) = f(t, u)$$

$$K(t, x, u, u) = g(t, x, u, u) - t^{-1}x_a u_a,$$
(3.4.23)

where f and g are arbitrary differentiable functions.

To prove this theorem one has to repeat arguments used in proving previous theorems.

Let us present without proof some more statements dealing with Equations (3.4.21) which admit operators \tilde{G}_a (3.4.22), J_{ab} (3.4.12) and

$$\Pi = t^2 \partial_t + t x_a \partial_a \tag{3.4.24}$$

$$D = 2t\partial_t + x_a\partial_a \tag{3.4.25}$$

Theorem 3.4.6. Equations (3.4.21) are invariant under the operators \tilde{G}_a and J_{ab} iff they have the form

$$u_t = f(t, u)\Delta u + g(t, u, w_{n+1}) - \frac{x_a u_a}{t}, \qquad (3.4.26)$$

where f and g are arbitrary differentiable functions, $w_{n+1} = u_a u_a$.

Theorem 3.4.7. Equation (3.4.26) admits operator (3.4.24) iff

$$f(t,u) = f(u), \qquad g(t,u,w_{n+1}) = t^{-2}g(u,t^2w_{n+1})$$
(3.4.27)

Theorem 3.4.8. Equation (3.4.26) admits operators (3.4.24), (3.4.25) iff it has the form

$$u_t = f(u)\Delta u + u_a u_a g(u) - \frac{x_a u_a}{t}$$
 (3.4.28)

Now consider a two-dimensional equation

$$F(x, u, u_0, u_1, u_{00}, u_{11}, u_{01}) = 0. (3.4.29)$$

Theorem 3.4.9. Amongst the set of Equations (3.4.29) only equations given by

$$F(w^{(1)}, w^{(2)}, u, u_1, u_{11}) = 0 (3.4.30)$$

are invariant under operators of GT (3.4.22).

In (3.4.30) we use the following notations:

$$w^{(1)} = \det \begin{pmatrix} u_0 & u_1 \\ u_{10} & u_{11} \end{pmatrix}, \qquad w^{(2)} = \det \begin{pmatrix} u_{00} & u_{01} \\ u_{10} & u_{11} \end{pmatrix}$$
 (3.4.31)

Theorem 3.4.9 can be easy generalized on the case of (n+1)-dimensional space.

Theorem 3.4.10. Equation

$$F\left(u, \underbrace{u}_{1}, \underbrace{u}_{2}\right) = 0 \tag{3.4.32}$$

is invariant under operators ∂_0 , ∂_a and (3.4.22) iff it has the form

$$F_1\left(w^{(1)}, w^{(2)}, u, u, u, u\right) = 0$$
(3.4.33)

where

$$w^{(1)} = \det \begin{pmatrix} u_0 & u_1 & \dots & u_n \\ u_{10} & u_{11} & \dots & u_{1n} \\ \vdots & \vdots & \ddots & \vdots \\ u_{n0} & u_{n1} & \dots & u_{nn} \end{pmatrix}$$

$$w^{(2)} = \det \begin{pmatrix} u_{00} & u_{01} & \dots & u_{0n} \\ u_{10} & u_{11} & \dots & u_{1n} \\ \vdots & \vdots & \ddots & \vdots \\ u_{n0} & u_{n1} & \dots & u_{nn} \end{pmatrix}$$
(3.4.34)

Note 3.4.3. In the specific case when

$$F_1 = w^{(2)} = 0$$

it yields multi-dimensional Monge-Ampere equation, which has been studied in 1.10.

Note 3.4.4. The maximal invariance algebra of equation

$$w^{(1)} - \lambda = 0, \qquad \lambda = \text{const}$$
 (3.4.35)

is infinite-dimensional and is given by operators

$$X = \xi^{\mu} \partial_{\mu} + \eta \partial_{u}, \qquad \mu = \overline{0, n}$$

$$\xi^{0} = c_{00}t + d_{0}, \qquad \xi^{a} = c_{ab}x_{b} + f_{a}(t), \qquad (3.4.36)$$

$$\eta = cu + d, \qquad c = (n+1)^{-1} (c_{00} + 2(c_{11} + \ldots + c_{nn}))$$

where c_{00} , c_{ab} , d_0 , d are arbitrary constants, and $f_a(t)$ are arbitrary differentiable functions.

Note 3.4.5. It is easy to construct the general solution of the two-dimensional equation

$$w^{(1)} = \det \begin{pmatrix} u_0 & u_1 \\ u_{01} & u_{11} \end{pmatrix} = 0$$
 (3.4.37)

To do it we represent (3.4.37) as follows

$$\frac{\partial}{\partial x_1}\left(\frac{u_1}{u_0}\right)=0,$$

whence the general solution is obtained

$$u = F(x_1 + G(x_0)) \tag{3.4.38}$$

(F and G are arbitrary differentiable functions). Direct verification shows that

$$u = F(l_a x_a + G(x_0)), \qquad a = \overline{1, n}, \qquad l_a = \text{const}$$

is a particular solution of (n + 1)-dimensional Equation (3.4.35) with $\lambda = 0$.

Note 3.4.6. Equation

$$w^{(1)} = \det \begin{pmatrix} u_0 & u_1 & \dots & u_n \\ u_{01} & u_{11} & \dots & u_{n1} \\ \vdots & \vdots & \ddots & \vdots \\ u_{0n} & u_{1n} & \dots & u_{nn} \end{pmatrix} = F(u), \quad (3.4.39)$$

where F(u) is an arbitrary twice differentiable function, can be reduced to (3.4.35) at $\lambda = 1$ with the help of substitution

$$u \to v = \int \frac{du}{\left[F(u)\right]^{1/(n+1)}}.$$
 (3.4.40)

In conclusion, we note that among Galilei invariant equations (3.4.33) one can distinguish a class of equations

$$u_0 = \lambda(u, u_1) \Delta u + Q(u, u_1) - w^{(3)} / w^{(2)},$$

$$w^{(3)} = \det \begin{pmatrix} 0 & u_1 & \dots & u_n \\ u_{10} & u_{11} & \dots & u_{1n} \\ \vdots & \vdots & \ddots & \vdots \\ u_{n0} & u_{n1} & \dots & u_{nn} \end{pmatrix}$$

$$w^{(2)} = \det \begin{pmatrix} u_{11} & u_{12} & \dots & u_{1n} \\ u_{21} & u_{22} & \dots & u_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ u_{n1} & u_{n2} & \dots & u_{nn} \end{pmatrix}$$
(3.4.41)

 λ , Q being arbitrary functions, which are diffusive type nonlinear PDE with a strong nonlinearity. Note that in [112*] another approach is developed to describe diffusive processes with final rate.

3.5 Reduction and exact solutions of nonlinear Schrödinger equation

In this paragraph we consider a particular case of Sch(1,3)-invariant nonlinear generalization of Schrödinger equation, namely the equation

$$\left(i\frac{\partial}{\partial x_0} + \frac{\Delta}{2m}\right)u + \lambda|u|^{4/3}u = 0, \qquad |u| = \sqrt{u^*u}.$$
(3.5.1)

Solutions of Equation (3.5.1) we seek for in the form

$$u(x) = f(x)\varphi(\omega) \tag{3.5.2}$$

functions f(x) and new variables $\omega = \{w_1(x), \omega_2(x), \omega_3(x)\}$ being determined with the help of method expounded in §1.4. So, it is necessary to make up a linear combination of symmetry operators (they are given in (3.3.25))

$$Q = \sum_{\ell=1}^{13} \theta_{\ell} Q_{\ell} = \xi^{\mu}(x) \partial_{\mu} + \eta(x) u \partial_{u}$$
(3.5.3)

and then, to find f(x) and $\omega(x)$, to solve system of ODE

$$\frac{dx_0}{\xi^0} = \frac{dx_1}{\xi^1} = \dots = \frac{dx_3}{\xi^3} = \frac{du}{u\eta}.$$
 (3.5.4)

Without going into details we present in Table 3.5.1 the final result: explicit form of Ansatze (3.5.2).

Ν	Invariants $\omega_1, \omega_2, \omega_3$	Ansatze $u(x) = f(x)\varphi(x)$
1.	$ec{lpha} \cdot ec{x} \ / \ \sqrt{1-x_0^2}, \ ec{x}^2 \ / \ (1-x_0^2),$	$u = (1 - x_0^2)^{-3/4} \cdot \exp\{rac{1}{2}im\cdot$
	$\operatorname{arcth} x_0 + \operatorname{arctan} (ec{eta} \cdot ec{x} / ec{\gamma} \cdot ec{x})$	$+\left(x_0ec{x}^2 \ / \ (1-x_0^2) ight) ight\} arphi(\omega)$
2.	$ec{lpha}\cdotec{x}/x_0, ec{x}^2/x_0^2,$	$u = x_0^{-3/2}.$
	$1/x_0 + \arctan\left(ec{eta}\cdotec{x} \ / \ ec{\gamma}\cdotec{x} ight)$	$\cdot \exp\left\{-rac{1}{2}imec{x}^2/x_0 ight\}arphi(\omega)$
3.	$\vec{\alpha} \cdot \vec{x} / (1 + x_0^2)^{1/2}, \vec{x}^2 / (1 + x_0^2),$	$u = (1 + x_0^2)^{-3/4} \cdot$
	$- \arctan x_0 + \arctan \left(ec{eta} \cdot ec{x} \ / \ ec{\gamma} \cdot ec{x} ight)$	$\left \cdot \exp\left\{-rac{1}{2}imx_0ec{x}^2/(1+x_0^2) ight\}arphi(\omega) ight.$
4.	$ec{lpha}\cdotec{x}\ /\ \sqrt{x_0},\ ec{x^2}/x_0,$	$u = x_0^{-3/4} \varphi(\omega)$
	$-\ln x_0 + \arctan ec eta \cdot ec x / ec \gamma \cdot ec x$	
5.	$ec{lpha}\cdotec{x}/\sqrt{x_0},ec{eta}\cdotec{x}/\sqrt{x_0},ec{\gamma}\cdotec{x}/\sqrt{x_0}$	$u = x_0^{-3/4} \varphi(\omega)$
6.	$ec{lpha}\cdotec{x},ec{x^2},$	$u = \varphi(\omega)$
	$-x_0 + \arctan ec eta \cdot ec x / ec \gamma \cdot ec x$	
7.	$ec{lpha}\cdotec{x},ec{x}^2,x_0$	$u = \varphi(\omega)$
8.	$ec{lpha}\cdotec{x}+x_0ec{eta}\cdotec{x},ec{lpha}\cdotec{x}+x_0ec{\gamma}\cdotec{x},x_0$	$u = \exp\{-\frac{1}{2}im(\vec{\alpha}\cdot\vec{x})^2/x_0\}\varphi(\omega)$
9.	$ec{lpha}\cdotec{x},ec{eta}\cdotec{x},x_0$	$u = \varphi(\omega)$

Table 3.5.1. Ansatze invariant under some subgroups of Sch(1,3)-group.

Note 3.5.1. Table 3.5.1 does not contain some cumbersome cases.

2. Having substituted Ansatze from Table 3.5.1 into Equation (3.5.1), we get reduced PDEs as follows:

- (1) $L_2(\varphi) + 6\varphi_2 2im\varphi_3 + m^2\omega_2\varphi 2\lambda m|\varphi|^{4/3}\varphi = 0$
- (2) $L_2(\varphi) + 6\varphi_2 + 2im\varphi_3 2\lambda m|\varphi|^{4/3}\varphi = 0$

(3)
$$L_2(\varphi) + 6\varphi_2 + 2im\varphi_3 - m^2\omega_2\varphi - 2\lambda m|\varphi|^{4/3}\varphi = 0$$

(4)
$$L_2(\varphi) + im\omega_1\varphi_1 + 2(im\omega_2 + 3)\varphi_2 + 2im\varphi_3 + \frac{3}{2}im\varphi - 2\lambda m|\varphi|^{4/3}\varphi = 0$$

(5)
$$\Delta \varphi + im\omega_a \varphi_a + \left(\frac{3}{2}im - 2\lambda m|\varphi|^{4/3}\right)\varphi = 0$$

(6) $L_2(\varphi) + 6\varphi_2 + 2im\varphi_3 - 2\lambda m |\varphi|^{4/3} \varphi = 0$

(7)
$$i\varphi_3 - \frac{1}{2m}(\varphi_{11} + 4\omega_1\varphi_{12} + 4\omega_2\varphi_{22}) + \lambda|\varphi|^{4/3}\varphi = 0$$

(8)
$$(\omega_3^2 + 1)(\varphi_{11} + \varphi_{22}) + \varphi_{22} - \frac{2im}{\omega_3}(\omega_a\varphi_a + \frac{1}{2}\varphi) - 2\lambda m|\varphi|^{4/3}\varphi = 0$$

(9)
$$i\varphi_3 - \frac{1}{2m}(\varphi_{11} + \varphi_{22}) + \lambda |\varphi|^{4/3} \varphi = 0$$

Chapter 3. Euclid and Galilei Groups and Nonlinear PDEs for Scalar Fields

where $L_2(\varphi) = \varphi_{11} + 4\omega_2\varphi_{22} + (\omega_2 - \omega_1^2)^{-1}\varphi_{33} + 4\omega_1\varphi_{12}; \quad \varphi_a \equiv \frac{\partial\varphi}{\partial\omega_a}$

Having solved some equations from (3.5.5), we obtain solutions of Equation (3.5.1):

$$u(x) = x_0^{-3/2} \exp\left\{-\frac{im\,\vec{x}^2}{2\,x_0}\right\} \varphi(\omega_1), \qquad \omega_1 = \frac{\vec{\alpha} \cdot \vec{x}}{x_0} \tag{3.5.6}$$

where function $\varphi(\omega_1)$ is determined from quadrature

$$\int_{0}^{\varphi} \frac{d\tau}{\sqrt{c_2 + \tau^{10/3}}} = \sqrt{\frac{6}{5}\lambda m} \left(\omega_1 + c_1\right)$$
(3.5.7)

 c_1 and c_2 are arbitrary constants.

$$u(x) = x_0^{-3/2} \exp\left\{-\frac{im\,\vec{x}^2}{2\,x_0}\right\}\varphi(\omega_2), \qquad \omega_2 = \frac{\vec{x}^2}{x_0} \tag{3.5.8}$$

where real function $\varphi(\omega_2)$ satisfies the Emden-Fauler equation

$$2\omega_2\varphi_{22} + 3\varphi_2 - \lambda m\varphi^{7/3} = 0 \tag{3.5.9}$$

If $\lambda = \frac{3}{2}i$ then

$$u(x) = (1 - x_0^2)^{-3/4} \exp\left\{\frac{im}{2}\frac{\vec{x}^2}{1 - x_0}\right\}.$$

Some more solutions of Equation (3.5.1):

$$u(x) = (\alpha_0 x_0 - \vec{\alpha} \cdot \vec{x})^{-3/2} \exp\left\{-\frac{im}{2}\frac{\vec{x}^2}{x_0}\right\} \varphi(\omega_2), \qquad (3.5.10)$$

where α_0 , α_a are arbitrary constants, and $\vec{\alpha}^2 = \frac{8}{15} \lambda m$.

$$u(x) = x_0^{-3/4} \exp\left\{-\frac{im}{2}\frac{\vec{x}^2 - \vec{\beta} \cdot \vec{x}}{x_0}\right\},$$
(3.5.11)

where $\vec{\beta}$ are arbitrary constants, $\vec{\beta}^2 = -\frac{8\lambda}{m}$.

$$u(x) = \left(\frac{8}{3}\lambda m \vec{x}^2\right)^{-3/4} \exp\left\{-\frac{im}{2}\frac{\vec{x}^2}{x_0}\right\},$$

$$u(x) = \left(\frac{c}{3\lambda}\right)^{3/4} \frac{1}{\sqrt{x_0}} \exp\left\{i\left(cx_0^{-1/3} - \frac{m}{2}\frac{(\vec{\alpha} \cdot \vec{x})^2}{x_0}\right)\right\},$$
(3.5.12)

where $c, \vec{\alpha}^2$ are arbitrary constants, $\vec{\alpha} = 1$.

$$u(x) = \left(\frac{3i}{4\lambda x_0}\right)^{3/4}, \qquad u(x) = \left(\frac{c}{\lambda}\right)^{3/4} \exp\{icx_0\}$$
(3.5.13)

$$u(x) = (\vec{\alpha} \cdot \vec{x})^{-3/2}, \qquad u(x) = \left(\frac{8}{3}\lambda m \vec{x}^2\right)^{-3/4}, \qquad u(x) = \exp\{i\vec{\beta} \cdot \vec{x}\} \quad (3.5.14)$$

where $\vec{\alpha}$, $\vec{\beta}$ are arbitrary constants, $\vec{\alpha}^2 = \frac{8}{15} \lambda m$, $\beta^2 = -2\lambda m$.

Note 3.5.2. When $\lambda = 0$ (3.5.10), (3.5.11) give the fundamental solution of Schrödinger equation

$$u(x) = x_0^{-3/2} \exp\left\{-\frac{im}{2}\frac{\vec{x}^2}{x_0}\right\}$$
(3.5.15)

Now we present some soliton-like solutions of Equation (3.5.1) obtained in $[70, 68^*, 69^*]$. Let

$$u(x) = \exp\left\{\frac{ix_0}{\beta}\right\}\varphi(\omega), \qquad \omega = \alpha_a x_a$$
 (3.5.16)

where $\varphi(\omega)$ is a real function, α_a , β are arbitrary real constants.

Having substituted (3.5.16) into (3.5.1) we get

$$\frac{\vec{\alpha}^2}{2m}\ddot{\varphi} + \frac{1}{\beta}\varphi + \lambda\varphi^{7/3} = 0 \tag{3.5.17}$$

Partial solutions of Equation (3.5.17) we seek for in the form

$$\varphi(\omega) = a(2 \operatorname{sh} b\omega)^{k_1} (2 \operatorname{ch} b\omega)^{k_2}, \qquad (3.5.18)$$

where a, b, k_1, k_2 are constants to be defined. After substituting (3.5.18) into (3.5.17) we obtain

$$\frac{\alpha^2 b^2}{2m} \left[k_1 + k_2 + 2k_1 k_2 + k_1 (k_2 - 1) \left(\frac{\operatorname{ch} b\omega}{\operatorname{sh} b\omega} \right)^2 + k_2 (k_2 - 1) \left(\frac{\operatorname{sh} b\omega}{\operatorname{ch} b\omega} \right)^2 \right] + \lambda \left[a (2 \operatorname{sh} b\omega)^{k_1} (2 \operatorname{ch} b\omega)^{k_2} \right]^{4/3} = 0$$
(3.5.19)

In (3.5.19) constants a, b, k_1 , k_2 , can be selected so as to satisfy it identically. Let $k_1 = 0$, $k_2 = -3/2$ then (3.5.19) takes the form

$$\frac{1}{\beta} - \frac{3}{2}\alpha^2 b^2 + \frac{15}{4}\frac{\alpha^2}{2m}b^2 \left(\frac{\operatorname{sh}b\omega}{\operatorname{ch}b\omega}\right)^2 + \frac{\lambda a^{4/3}}{4(\operatorname{ch}b\omega)^2} = 0$$

This equality holds true iff

$$a = \pm 2\sqrt{2} \left(\frac{5}{-3\lambda\beta}\right)^{3/4}, \qquad b = \pm \frac{2\sqrt{2m}}{3\sqrt{-\alpha^2\beta}}, \qquad \beta < 0 \quad \lambda > 0$$

It yields the solution of Equation (3.5.17)

$$\varphi(\omega) = a_{-}(\operatorname{ch} b\omega)^{-3/2}, \qquad a_{-} = \pm \left(\frac{-5}{3\lambda\beta}\right)^{3/4}$$
(3.5.20)

and then, returning to (3.5.16), the following solution of Equation (3.5.1)

$$u(x) = a_{-} \exp\left\{\frac{ix_{0}}{\beta}\right\} (\operatorname{ch} b\alpha_{a}x_{a})^{-3/2}$$
 (3.5.21)

where a_{-} is defined in (3.5.20).

The solution (3.5.21) is naturally to consider as soliton-like by analogy with that of [209]

$$\psi(t,x_1) = \left(-\frac{\lambda}{4}\right)^{1/4} \frac{\exp\left\{\frac{1}{2}iv\left[x_1 + \left(\frac{\epsilon}{2} + \frac{\lambda^2}{8v}\right)t\right]\right\}}{\operatorname{ch}\sqrt{-\frac{\lambda}{4}}(x_1 + vt)}$$

which is a solution of one-dimensional nonlinear Schrödinger equation

$$i\frac{\partial\Psi}{\partial x_0}+rac{\partial^2\Psi}{\partial x_1^2}=\lambda(\Psi^*\Psi)\Psi$$

Setting in (3.5.19) $k_1 = -\frac{3}{2}, k_2 = 0$ we find

$$a=\pm\left(rac{20}{3\lambdaeta}
ight)^{3/4},\qquad b=\pmrac{2}{3}\sqrt{-rac{lpha^2eta}{2m}},\qquadeta<0\quad\lambda<0$$

The Corresponding solution of Equation (3.5.1) has the form

$$u(x) = a_{+} \exp\left\{\frac{ix_{0}}{\beta}\right\} (\operatorname{sh} b\vec{\alpha} \cdot \vec{x})^{-3/2}, \qquad a_{+} \equiv \pm \left(\frac{5}{3\lambda\beta}\right)^{3/4} \qquad (3.5.22)$$

Using formulae of generating solutions (see Table 4.1.2) one can construct other solutions of Equation (3.5.1) from those presented here.

It will be noted that using ansatz (3.5.2) one can determine functions f(x) and new variables $\omega_a(x)$ from splitting conditions (see §1.4). The substitution of (3.5.2) into (3.5.1) gives rise to the following equation

$$i\left(\frac{\partial f}{\partial t}\varphi + f\frac{\partial \omega_a}{\partial t}\frac{\partial \varphi}{\partial \omega_a}\right) + \frac{1}{2m}\left[\varphi\Delta f + 2\frac{\partial f}{\partial x_a}\right]$$
(3.5.23)

$$\cdot \frac{\partial \omega_j}{\partial x_a} \frac{\partial \varphi}{\partial \omega_j} + f\left(\frac{\partial \varphi}{\partial \omega_j} \Delta \omega_j + \frac{\partial \omega_j}{\partial x_a} \frac{\partial \omega_k}{\partial x_a} \frac{\partial^2 \varphi}{\partial \omega_j \partial \omega_k}\right) + \lambda (f^2 \varphi^* \varphi)^{2/3} f \varphi = 0,$$

whence follows the splitting conditions

$$\frac{1}{2m}\Delta f + i\frac{\partial f}{\partial t} = f^{7/3}F_0(\omega),$$

$$if\frac{\partial \omega_a}{\partial t} + \frac{1}{m}\left(\frac{\partial f}{\partial x_k}\frac{\partial \omega_a}{\partial x_k} + \frac{1}{2}f\Delta\omega_a\right) = f^{7/3}F_a(\omega),$$

$$\frac{\partial \omega_i}{\partial x_a}\frac{\partial \omega_j}{\partial x_a} = f^{7/3}G_{ij}(\omega),$$
(3.5.24)

where F_0 , F_a , G_{ij} are smooth functions of ω_a . If conditions (3.5.24) are fulfilled, then Equation (3.5.1) (see also (3.5.23)) takes the form

$$\left[F_0(\omega) + \lambda(\varphi^*\varphi)^{2/3}\right]\varphi + F_a(\omega)\frac{\partial\varphi}{\partial\omega_a} + G_{ij}(\omega)\frac{\partial^2\varphi}{\partial\omega_i\partial\omega_j} = 0.$$
(3.5.25)

Note that system (3.5.24) may have to possess wide symmetry, and it opens new ways of constructing solutions of the initial equation (3.5.1).

In conclusion we present, following [68^{*}], in Table 3.5.2 the complete set of Sch(1,3)-inequivalent ansatze (3.5.2) for a Schrödinger-invariant equation (the basis elements of ASch(1,3) are given in (3.3.25)). In this table α_0 , α , β are arbitrary real constants, $t \equiv x_0$. In [68^{*},69^{*}] ansatze of Table 3.5.2 are used for reduction and finding exact solutions of some nonlinear Schrödinger equation.

As a concluding remark we note that the wave equation $\Box u = 0$ is reduced to the heat equation $\varphi_{\tau} = \varphi_{11} + \varphi_{22}$ by means of the ansatz

$$u = \exp\{-(x_3 + t)\}\varphi(\tau, x_1, x_2), \qquad \tau = \frac{1}{4}(x_3 - t),$$

which is invariant under the operator

$$Q = \partial_t + \partial_3 - 2u\partial_u.$$

N	Algebra	Invar. var. ω	Ansatze $u(x) = f(x)\varphi(\omega)$
1.	P_1	t, x_2, x_3	f = 1
2.	$X_0 + \alpha_0 I$	x_1, x_2, x_3	$f = \exp\{ilpha_0 t\}$
3.	$J_{12} + \alpha I$	$t, x_1^2 + x_2^2, x_3$	$f = \exp\{ilpha \arctan{(x_2/x_1)}\}$
4.	$J_{12} + G_3$	$t, x_1^2 + x_2^2,$	$f=\exp\{-imx_3^2/2t\}$
		$x_3 - t \arctan\left(x_2/x_1\right)$	
5.	$J_{12} - X_0$	$t + \arctan(x_2/x_1),$	$\int f = \exp\{-ilpha_0 t\}$
	$+\alpha_0 I$	$x_1^2 + x_2^2, x_3$	
6.	$G_1 + X_2$	$t, x_1 - tx_2, x_3$	$\int f = \exp\{-imx_1^2/2t\}$
7.	$G_1 - X_0$	$2x_1 + t^2, x_2, x_3$	$f = \exp\{imt(x_1 + \frac{1}{3}t^2)\}$
0	I LOG		(1, 2, 2, 2, 3, 3, 3, 3, 3, 3, 3, 3, 3, 3, 3, 3, 3,
8.	$J_{12} + \beta G_3 - V$	$t + \arctan(x_2/x_1),$	$J = \exp\{i\beta mi(x_3 + \frac{1}{3}\beta i^2)\}$
	$-\Lambda_0$	$\begin{array}{c} x_1 + x_2, 2x_3 + \rho t^* \\ x_1 & x_2 & x_3 \end{array}$	$t = -3/4 \pm i\alpha m/2$
9.	$D + \alpha I$	$\overline{\sqrt{t}}, \overline{\sqrt{t}}, \overline{\sqrt{t}}$	$f = t^{-0/4 + \tan 2}$
10	У П	x_1 x_2	$f = (t^2 + 1)^{-3/4} \text{ or } \int 1 dt$
10.	$\Lambda_0 + \Pi -$	$\overline{\sqrt{1+t^2}}, \overline{\sqrt{1+t^2}},$	$\int -(i+1) + \exp \left(-\frac{1}{2}im\right)$
	$-\alpha I$	x_3	$\left(\frac{\vec{x}^2 t}{1+2\alpha \arctan t}\right)$
	ar	$\sqrt{1+t^2}$	$\left(1+t^2+2t^2 \operatorname{dir} \operatorname{dir} \operatorname{dir} t\right)\right\}$
11.	$J_{12} + \beta D$	$\ln t + 2\beta \arctan{(x_1/x_2)},$	$\int f = t^{-3/4 + i\alpha m/2\beta}$
[$x_1^2 + x_2^2 x_3$	
	$+\alpha I$	$\frac{1}{t}, \frac{1}{\sqrt{t}}$	
12	$\mathbf{Y}_{i} + \mathbf{\Pi}_{i}$	β arctant arctan $\begin{pmatrix} x_1 \end{pmatrix}$	$\int f - (t^2 + 1)^{-3/4} \exp \int -\frac{1}{4} im$
12.	A0 + II-	p arctail t – arctail $\left(\frac{x_2}{x_2}\right)$,	$\int -(i+1) = \exp \left\{-\frac{1}{2}im^{2}\right\}$
	$-\beta \ln - \alpha I$	$\frac{x_1^2 + x_2^2}{x_3}$	$\left(\frac{\vec{x}^2 t}{1 + 2\alpha \arctan t} \right)$
		$t^2 + 1$ ' $\sqrt{t^2 + 1}$	$\left(1+t^2+2\pi \tan \theta\right)$
13.	$X_0 + \Pi -$	$\frac{tx_1+x_2}{2}+\beta \arctan t.$	$\int f = (t^2 + 1)^{-3/4} \exp \left\{ -\frac{1}{2} im \right\}$
	-0	t^2+1	
	$-J_{12}-$	$\frac{tx_2 + x_1}{t^2 + 1}, \frac{x_3}{\sqrt{2}}$	$\cdot \left(\frac{x^{-t}}{1+t^{2}} + \right)$
		$t^2 + 1 \sqrt{t^2 + 1}$	$\left \begin{array}{c} \left(1 + t^{2} \\ t(tx_{0} - x_{0}) \right) \right\rangle$
	$-\beta(G_1+X_2)$		$\left +2\beta \arctan \frac{i(ix_2-x_1)}{i(1-x_1)} \right $
L	l		

Table 3.5.2. Sch(1,3)-inequivalent ansatze for a complex scalar field.

3.6 Symmetry and some exact solutions of the Hamilton-Jacobi equation

Here we investigate point and contact symmetry and construct exact solutions of Hamilton-Jacobi equation of free particle

$$u_t + \frac{1}{2m} \left(\vec{\nabla} u \right)^2 = 0,$$
 (3.6.1)

where function $u = u(t, \vec{x})$ means action, m is a constant (mass of particle).

Theorem 3.6.1. [36] Maximal local (point) invariance group of the HJ equation (3.6.1) is the 21-parameter group, basis elements of corresponding Lie algebra having the form

$$p_{0} = \partial_{t}, \qquad p_{a} = \partial_{a}, \qquad p_{4} = \partial_{u}; \qquad a = 1, 2, 3$$

$$J_{ab} = x_{a}p_{b} - x_{b}p_{a}$$

$$G_{a}^{(1)} = tp_{a} + mx_{a}p_{a}, \qquad D^{(1)} = tp_{0} + \frac{1}{2}x_{a}p_{a},$$

$$\Pi^{(1)} = t^{2}p_{0} + tx_{a}p_{a} + \frac{m}{2}\vec{x}^{2}p_{4}, \qquad (3.6.2)$$

$$G_{a}^{(2)} = up_{a} + mx_{a}p_{0}, \qquad D^{(2)} = up_{4} + \frac{1}{2}x_{a}p_{a},$$

$$\Pi^{(2)} = u^{2}p_{4} + ux_{a}p_{a} + \frac{m}{2}\vec{x}^{2}p_{0},$$

$$K_{a} = 2x_{a}(D^{(1)} + D^{(2)}) + s^{2}p_{a}, \qquad (s^{2} \equiv \frac{2}{m}tu - \vec{x}^{2}).$$

Proof. We shall act in the same spirit as in $\S1.2$ (see Theorem 1.2.1). From condition of invariance (3) we find the following defining system for coordinates of infinitesimal operator (2)

$$\begin{aligned} \xi_{u}^{t} &= 0, & \xi_{a}^{t} = m\xi_{u}^{a}; & a, b = 1, 2, 3 \\ \xi_{b}^{a} &+ \xi_{a}^{b} = 0, & a \neq b; \\ \eta_{t} &= 0, & \eta_{a} = m\xi_{t}^{a}, \\ \eta_{u} &+ \xi_{t}^{t} = 2\xi_{a}^{a} & (\text{no sum over } a). \end{aligned}$$
(3.6.3)

The general solution of Equations (3.6.3) has the form

$$\xi^{t} = 2t\vec{K}\cdot\vec{x} + a_{1}t^{2} + a_{2}\frac{m\vec{x}^{2}}{2} + b_{1}t + g_{2a}mx_{a} + d_{0},$$

$$\xi^{a} = 2x_{a}\vec{K}\cdot\vec{x} + K_{a}s^{2} + (a_{1}t + a_{2}u)x_{a} + g_{1a}t + g_{2a}u + \frac{1}{2}(b_{1} + b_{2})x_{a} + c_{ab}x_{b} + d_{a},$$

$$\vec{x} = a \qquad m\vec{x}^{2}$$
(3.6.4)

$$\eta = 2u\vec{K} \cdot \vec{x} + a_2u^2 + a_1\frac{m\vec{x}^2}{2} + b_2u + g_{1a}mx_a + d_4,$$

 K_a , a_1 , a_2 , b_1 , b_2 , g_{1a} , g_{2a} , $c_{ab} = -c_{ba}$, d_0 , ..., d_4 are arbitrary constants. Thus follows (3.6.2). The theorem is proved.

Note 3.6.1. Looking closely at (3.6.2) one notices an interesting automorphism

of the vector fields given by interchanging t and u. It allows us to pick out from algebra (3.6.2) the following subalgebras:

$$\begin{aligned} \operatorname{ASch}(1(t),3) &= \left\{ p_0, p_a, p_4, J_{ab}, G_a^{(1)}, D^{(1)}, \Pi^{(1)} \right\}; \\ \operatorname{ASch}(1(u),3) &= \left\{ p_4, p_a, p_0, J_{ab}, G_a^{(2)}, D^{(1)}, \Pi^{(2)} \right\}; \\ \operatorname{AG}(2,3) &= \left\{ p_0, p_4, p_a, J_{ab}, G_a^{(1)}, G_a^{(2)}, \right\}; \\ \operatorname{AG}(2,3) &= \operatorname{AG}(2,3) \oplus \left\{ D^{(1)}, D^{(2)} \right\}; \\ \operatorname{ASch}(2,3) &= \left\{ (3.6.2) \right\}. \end{aligned}$$
(3.6.5)

Theorem 3.6.2. Algebra ASch(2,3) (3.6.2) is locally isomorphic to conformal algebra AC(1,4).

Proof. Due to symmetrical role of variables t and u in ASch(2,3), pointed out above, we can introduce the variables

$$x^{0} = \frac{1}{\sqrt{2}} \left(t + \frac{u}{m} \right), \qquad x^{4} = \frac{1}{\sqrt{2}} \left(t - \frac{u}{m} \right)$$
 (3.6.6)

and the covariant notation $\alpha = 0, 1, ..., 4$, with the metric $g_{00} = -g_{11} = ... = -g_{44} = 1$; $g_{ab} = 0$ ($\alpha \neq \beta$). Then upon introducing

$$P_{\alpha} = \frac{\partial}{\partial x^{\alpha}}, \qquad J_{ab} = x_{\alpha} P_{\beta} - x_{\beta} P_{\alpha}$$

$$D = x^{\alpha} P_{\alpha}, \qquad K_{\alpha} = 2x_{\alpha} D - x^{\beta} x_{\beta} P_{\alpha}$$

(3.6.7)

we find that the generators

$$P_{a} = P_{a}, \qquad P_{0} = \frac{1}{\sqrt{2}}(P_{0} + mP_{4})$$

$$P_{4} = \frac{1}{\sqrt{2}}(P_{0} - mP_{4})$$

$$J_{ab} = J_{ab}, \qquad J_{04} = -D^{(1)} + D^{(2)},$$

$$J_{0a} = -\frac{1}{\sqrt{2}}(G_{a}^{(1)} + \frac{1}{m}G_{a}^{(2)}),$$

$$J_{4a} = -\frac{1}{\sqrt{2}}(G_{a}^{(1)} - \frac{1}{m}G_{a}^{(2)}),$$

$$D = D^{(1)} + D^{(2)}$$
(3.6.8)

$$\begin{split} K_a &= -\mathcal{K}_a, \\ K_0 &= \sqrt{2} \left(\Pi^{(1)} + \frac{1}{m} \Pi^{(2)} \right), \\ K_4 &= -\sqrt{2} \left(\Pi^{(1)} - \frac{1}{m} \Pi^{(2)} \right), \end{split}$$

satisfy the commutation rules of AC(1,4) (1.2.3). The theorem is proved.

Now, to find the group action of Sch(2,3) group, we use the final transformations of C(1,4) group (see §2.3):

1) translations: generated by P_{α}

$$x'_{\alpha} = x_{\alpha} + a\alpha,$$

2) Lorentz transformations: generated by $J_{\alpha\beta}$

$$x'_{\alpha} = \Lambda^{\beta}_{\alpha} x_{\beta}, \qquad \Lambda^{\beta}_{\alpha} \in O(1,4)$$

3) dilatation: generated by D

$$x'_{\alpha} = e^{\theta} x_{\alpha},$$

4) special conformal transformations: generated by \mathcal{K}

$$x'_{\alpha} = \frac{x_{\alpha} - c_{\alpha} x^{\beta} x_{\beta}}{1 - 2c^{\beta} x_{\beta} + (c^{\beta} c_{\beta})(x^{\beta} x_{\beta})}$$

Writing these transformations in terms of the light-cone variables t and u we obtain the symmetry group of the HJ equation:

$$\begin{aligned} x'^{a} &= \Lambda_{b}^{a} x^{b} + \Lambda_{+}^{a} t + \frac{1}{m} \Lambda_{-}^{a} u, \\ t' &= \Lambda_{+}^{+} t + \frac{1}{m} \Lambda_{-}^{+} u + \Lambda_{a}^{+} x^{a}, \\ u' &= m \Lambda_{+}^{-} t + \Lambda_{-}^{-} u + m \Lambda_{a}^{-} x^{a}; \\ x'_{a} &= x_{a} + a^{a}, \qquad t' = t + a^{+}, \qquad u' = u + a^{-} \\ x'_{a} &= e^{\theta} x_{a}, \qquad t' = e^{\theta} t \qquad u' = e^{\theta} u; \\ x'^{a} &= \sigma^{-1} \left(x^{a} - c^{a} \vec{x}^{2} - \frac{2}{m} c^{a} t u \right), \\ t' &= \sigma^{-1} \left(t - c^{+} \left(\frac{2}{m} t u - \vec{x}^{2} \right) \right), \\ u' &= \sigma^{-1} \left(u - c^{-} \left(2tu - m \vec{x}^{2} \right) \right), \end{aligned}$$

where

$$\sigma = \sigma(t, x^{a}, u) = 1 - 2c^{+}t - \frac{2}{m}c^{-}u + 2\vec{c}\vec{x} + \vec{c}^{2}\left(\frac{2}{m}tu - \vec{x}^{2}\right)$$

and the parameters indicated with + and - can easily be expressed in terms of their covariant counterparts.

Now consider contact symmetry of HJ equation.

Theorem 3.6.3. Maximal invariance algebra of contact symmetry of HJ equation (3.6.1) is infinite-dimensional and is given by operators (1.2.10) with the characteristic function

$$W = W(u_t, u_a, w_0, w_a), \qquad w_0 = 2u - x_a u_a, \qquad w_a = t u_a - m x_a \quad (3.6.10)$$

The proof is absolutely analogous to that of Theorem 1.2.2.

Note 3.6.2. Contact transformations of HJ equation described above contain homogeneous ones, so that transformations for which Lie equations have a Hamiltonian structure (see (1.2.13)). For example, when $W = W(w_a)$, we let

$$W = w_a w_a = t^2 (\vec{\nabla} u)^2 - 2mt x_a u_a + m^2 \vec{x}^2.$$
(3.6.11)

To find final transformations generated by infinitesimal operator (1.2.10) with characteristic function (3.6.11) one has to solve the following set of first-order ODEs (Lee equations, see (1.2.13)):

$$\begin{split} t' &= 0, \qquad t'(\theta = 0) = t \\ \dot{x}'_a &= 2t'(mx'_a - t'u'_a), \qquad x'_a(\theta = 0) = x_a \\ \dot{u}' &= 0, \qquad u'(\theta = 0) = u, \\ \dot{u}'_t &= 2(t'u'_au'_a - mx'_au'_a), \qquad u'_{t'}(\theta = 0) = u_t \\ \dot{u}'_a &= 2m(mx'_a - t'u'_a), \qquad u'_a(\theta = 0) = u_a \end{split}$$

where dot means differentiations with respect to parameter θ . After integrating this system we find

$$t' = t, x'_{a} = x_{a} + 2\theta m t (x_{a} - u_{a}/m)$$

$$u' = u, u'_{a} = u_{a} + 2m\theta (mx_{a} - tu_{a})$$

$$u'_{t} = u_{t} - 2\theta u_{a} (mx_{a} - tu_{a}) - 2m\theta^{2} (mx_{a} - tu_{a})^{2}$$
(3.6.12)

One can make sure that transformations (3.6.12) leave Equation (3.6.1) invariant. Putting $u_a = mv_a$ and $u_t = -E$, where v_a and E are velocity and energy of particle, we can rewrite (3.6.12) as follows

$$t' = t, x'_{a} = x_{a} + 2\theta m t (x_{a} - v_{a} t)$$

$$v'_{a} = v_{a} + 2m\theta (x_{a} - v_{a} t)$$

$$E' = E + 2m^{2}\theta \vec{v} \cdot (\vec{x} - \vec{v}t) + 2m^{3}\theta^{2} (\vec{x} - \vec{v}t)^{2}$$
(3.6.13)

When $\vec{x} = \vec{v}t + \vec{x}^{(0)}$, $\vec{x}^{(0)}$ is a constant vector (uniform motion), (3.6.13) coincide with Galilean transformations. In other cases transformations (3.6.13) describe passing to uniformly accelerated frame of reference.

Before to construct solutions of the HJ equation (3.6.1) we note that the substitution

$$t = \frac{1}{\sqrt{2}}(x_0 + v), \qquad u = \frac{m}{\sqrt{2}}(x_0 - v)$$
 (3.6.14)

transforms Equation (3.6.1) into the relativistic HJ equation (1.2.1)

$$\frac{4m}{(m+u_t)^2} \left(u_t + \frac{1}{2m} (\vec{\nabla} u)^2 \right) = 0 \to 1 - v_\nu v^\nu = 0 \tag{3.6.15}$$

and vice versa, the substitution

$$x_0 = \frac{1}{\sqrt{2}} \left(t + \frac{u}{m} \right), \qquad v = \frac{1}{\sqrt{2}} \left(t - \frac{u}{m} \right)$$
 (3.6.16)

transforms the relativistic HJ equation into nonrelativistic

$$\frac{m}{(1+v_0)^2}(1-v_{\nu}v^{\nu}) = 0 \to u_t + \frac{1}{2m}(\vec{\nabla}u)^2 = 0$$
(3.6.17)

However, this equivalence, as it is seen from (3.6.15), (3.6.17), breaks down when

$$m + u_t = 0$$
 or $1 + v_0 = 0$.

It means that if to cast away from manifold of solutions of HJ equation (3.6.1) the solutions

$$u = -mt + \varphi(\vec{x}) \tag{3.6.18}$$

and from manifold of solutions of the relativistic HJ equation the solutions

$$v = -x_0 + \text{const} \tag{3.6.19}$$

then the remaining manifolds will be locally equivalent and this equivalence is given by (3.6.14), (3.6.16).

It will be noted that substitution (3.6.16) being applied to an arbitrary P(1,4)-invariant PDE results in at least Galilean invariant equation. For example, the Monge-Ampere equation is invariant under (3.6.16) (it is due to its IGL(4,R) invariance (see Theorem 1.10.1)). Another example: ELBI equation (1.1.17) having been applied (3.6.16) results in the equation

Chapter 3. Euclid and Galilei Groups and Nonlinear PDEs for Scalar Fields

$$m^{2}u_{tt} - 2m\left(u_{t} + \frac{1}{2m}(\vec{\nabla}u)^{2}\right)\Delta u + 2mu_{a}u_{ta} + u_{a}u_{b}u_{ab} = 0$$
(3.6.20)

Theorem 3.6.4. Maximal local invariance group of Equation (3.6.20) is the extended Galilei group $\tilde{G}(2,3)$, basis elements of corresponding Lie algebra $A\tilde{G}(2,3)$ written in (3.6.5), (3.6.2). The proof is analogous to that of Theorem 3.6.1.

So, one can use Equation (3.6.20) as a nonrelativistic counterpart of ELBI equation to describe physical phenomena which satisfy Galilean relativistic principle.

Now let us list some exact solutions of HJ equation (3.6.1) obtained in [186]

$$u = -\frac{t}{2m} + (\vec{a} \cdot \vec{x}),$$

$$u = \frac{m}{2t}\vec{x}^{2}, \qquad u = \frac{m}{2t}(\vec{a} \cdot \vec{x})^{2},$$

$$u = \frac{4m}{3}(t^{2} - \vec{a} \cdot \vec{x})^{3/2} + 2mt\vec{a} \cdot \vec{x} - \frac{4m}{3}t^{3},$$

(3.6.21)

where $\vec{a} = \{a_1, a_2, a_3\}$ are arbitrary constants, $\vec{a}^2 = 1$.

One can use solutions (3.6.21) to generate with the help of generating formulae other solutions of Equation (3.6.1). Several such formulae are presented below:

$$\begin{split} u_{II}(x) &= u_{I}(t+a_{0},\vec{x}+\vec{a}) + a_{4}, \\ u_{II}(x) &= u_{I}\left(\frac{t}{1-\theta t},\frac{\vec{x}}{1-\theta t}\right) - \frac{m\vec{x}^{2}}{2}\frac{\theta}{1-\theta t}, \\ \frac{u_{II}(x)}{1-\theta u_{II}(x)} &= u_{I}\left(t+\frac{m}{2}\vec{x}^{2}\frac{\theta}{1-\theta u_{II}(x)},\frac{\vec{x}}{1-\theta u_{II}(x)}\right), \\ u_{II}(x) &= u_{I}\left(t,\vec{x}+\vec{v}t\right) - m\vec{v}\vec{x} - \frac{m\vec{v}^{2}}{2}t \end{split}$$

and so on; a_0 , a_4 , \vec{a} , \vec{v} , θ are arbitrary constants.

3.7 Symmetry and some exact solutions of the Boussinesq equation

Below we shall study symmetry and construct exact solutions of the Boussinesq equation

$$v_0 = \frac{\lambda}{2} \Delta v^2, \qquad \lambda = \text{const}$$
 (3.7.1)

where v = v(x), $x = \{x_0, \vec{x}\} \in \mathbb{R}^{n+1}$. Solutions of the two-dimensional Boussiness equation were obtained in [180].

For the sake of convenience we put

 $v^2 = u$

and consider instead of (3.7.1) the equation

$$u_0 = \lambda \sqrt{u} \Delta u \tag{3.7.2}$$

and more general equation

$$u_0 = F(u)\Delta u \tag{3.7.3}$$

where F(u) is arbitrary differentiable function.

Theorem 3.7.1. The widest invariance algebra admitted by Equation (3.7.3) is $A\widetilde{E}(1,n) \oplus AC(n)$ and it is achieved iff

$$F(u) = \lambda u^{4/(2-n)}$$
(3.7.4)

basis elements having the form

$$P_0 = i\partial_0, \quad P_a = i\partial_a, \quad J_{ab} = x_a P_b - x_b P_a, \quad D_1 = 2x_0 P_0 - x_a P_a \quad (3.7.5)$$

$$D_2 = x_a P_a + \frac{2-n}{2}i, (3.7.6)$$

$$K_a = 2x_a D_2 - \vec{x}^2 P_a \tag{3.7.7}$$

where n is the number of spacial variables \vec{x} , $a, b = \overline{1, n}$. In other cases we have: if

$$F(u) = \lambda u^k, \qquad k \neq 0, \ \frac{4}{2-n}$$
 (3.7.8)

or

$$F(u) = \lambda e^u \tag{3.7.9}$$

then the maximal IA of Equation (3.7.3) will be $A\widetilde{E}(1,n)$ provided

$$D_2 = x_a P_a + \frac{2}{k}i ag{3.7.10}$$

for (3.7.8) and

$$D_2 = x_a P_a - 2i \frac{\partial}{\partial u} \tag{3.7.11}$$

for (3.7.9).

In the case of arbitrary function F(u) the maximal admitted IA is $A\tilde{E}(1,n)$ with basis elements (3.7.5).

Proof. According to the Lie method, from condition of invariance (3) one can obtain the following system to define coordinates of infinitesimal symmetry operators (2):

$$\xi_u^0 = \xi_u^a = \xi_a^0 = \eta_{uu} = 0, \qquad \xi_b^a + \xi_a^b = 0, \qquad a \neq b$$
(3.7.12)

$$\xi_0^a + F(2\eta_{ua} - \Delta\xi^a) = 0, \qquad (3.7.13)$$

$$\eta \frac{F'}{F} + \xi_0^0 = 2\xi_1^1 = \dots = 2\xi_n^n \tag{3.7.14}$$

If $F(u) \neq \text{const}$, Equations (3.7.12)–(3.7.14) result in

$$\xi^{0} = 2\mathfrak{E}_{1}x_{0} + d_{0}$$

$$\xi^{a} = b_{a}\vec{x}^{2} - 2x_{a}\vec{b}\cdot\vec{x} + (\mathfrak{E}_{1} - \mathfrak{E}_{2})x_{a} + c_{ab}x_{b} + d_{a}$$

$$\eta = a(x)u + b(x) = ((n-2)\vec{b}\cdot\vec{x} + c_{1})u + b(\vec{x})$$

(3.7.15)

where c_1 , \mathfrak{X}_1 , \mathfrak{X}_2 , $c_{ab} = -c_{ba}$, b_a , d_0 , d_a are group parameters, and function $b(\vec{x})$ is a solution of Laplace equation $\Delta b = 0$. For function F(u) we have the ODE

$$\frac{F'}{F}(a(x)u + b(x)) = 2\xi_1^1 - \xi_0^0 \tag{3.7.16}$$

Here we have three irreducible cases

1)
$$a(\vec{x}) \neq 0$$
, $b(\vec{x}) = 0$

2) $a(\vec{x}) = 0, \quad b(\vec{x}) \neq 0$

3)
$$a(\vec{x}) = b(\vec{x}) = 0$$

Let us consider the first case. Substituting (3.7.15) into (3.7.16) we get

$$u\frac{F'}{F} = \frac{-4b(\vec{x}) - 2\omega}{(n-2)b(\vec{x}) + c_1} = k = \text{const.}$$
(3.7.17)

If $b \neq 0$, then equality (3.7.17) holds true when $c_1 = \frac{1}{2}(n-2)\omega_2$, k = 4/(2-n), and after integrating it results in (3.7.4). In turn (3.7.15) gives (3.7.5)-(3.7.7). If b = 0, (3.7.17) yields $k \neq 0$, $c_1 = \frac{2}{k}\omega_2$, $F(u) = \lambda u^k$.

In the second case (3.7.16) takes the form

$$\frac{F'}{F} = -2\mathfrak{E}_2/b(\vec{x})$$

and it makes sense only if $b(\vec{x}) = \text{const.}$ Putting $b(\vec{x}) = -2\omega_2$ we get (3.7.8), (3.7.11).

Supposing a = b = 0 we find that (3.7.16) holds true with arbitrary function F(u) provided

$$2\xi_1^1 = \xi_0^0 \tag{3.7.18}$$

It follows that $b = x_2 = 0$ and, hence, Equation (3.7.3) is invariant under algebra (3.7.5). The theorem is proved.

Remark 3.7.1. Theorem 3.7.1 is proved on the assumption that $F(u) \neq \text{const.}$ Otherwise, Equation (3.7.3) coincides with the linear heat equation and symmetry of this latter one is well-known.

Consequence 3.7.1. The maximal IA of the Boussinesq equation (3.7.2) is the extended Euclid algebra with basis elements (3.7.5), (3.7.10).

Solutions of the Boussinesq equation we seek in the form (3.2.6), where invariant variables ω are given in Table 3.2.1 and

$$f(x) = \begin{cases} x_0^2, & N \ 1-4\\ 1, & N \ 5-10 \end{cases}$$

Below we present reduction equations which succeed in integrating. In the 6th case (the enumeration corresponds to that of Table 3.2.1), on the assumption $\varphi_{\omega_2} = 0$, we have

$$\beta_0 \varphi_{11} + \varphi_3 = \lambda \sqrt{\varphi} \beta^2 \varphi_{11} \tag{3.7.19}$$

If $\beta_0 = 0$, $\beta^2 = -i/12\lambda$ then one can separate variables ω_1 , ω_3 so that

$$\varphi = A(\omega_1)B(\omega_3) \tag{3.7.20}$$

Substituting (3.7.20) into (3.7.19) we find

$$\frac{A^{\prime\prime}}{A} = -\frac{6B^{\prime}}{B^{3/2}} = k = \text{const}$$

whence $A = \left(\frac{k}{12}\right)^2 (\omega_1 + c_1)^4$, $A = \left(\frac{12}{k}\right)^2 (\omega_3 + c_3)^{-2}$; c_1 , c_2 are constants. It follows the solution of Equation (3.7.2)

$$u(x) = \frac{(\vec{\beta} \cdot \vec{x} + c_1)^4}{(x_0 + c_2 \ln \vec{\sigma} \vec{x} + c_3)^2},$$
(3.7.21)

where $c_1, c_2, c_3, \vec{\beta}, \vec{\sigma}$ are arbitrary constants, $\beta^2 = -1/12\lambda, \vec{\beta} \cdot \vec{\sigma} = \vec{\sigma}^2 = 0$. Solution (3.7.21) can be generalized as follows

$$u(x) = \frac{(\vec{\beta} \cdot \vec{x} + c_1)^4}{[x_0 + F(\vec{\sigma} \cdot \vec{x})]^2},$$
(3.7.22)

where F is an arbitrary differentiable function.

Let $\varphi_3 = 0$, then (3.7.19) takes the form

$$\lambda\beta^2\varphi_{11}\sqrt{\varphi}=\beta_0\varphi_1$$

and has the general solution

$$\sqrt{\varphi} + c_2 \ln \left(\sqrt{\varphi} - c_2\right) = a(\omega_1 + c_1)$$

where $c_1, c_2, a = \beta_0 / \lambda \beta^2$ are arbitrary constants. So we have one more solution of Equation (3.7.2)

$$\sqrt{u} + c_2 \ln \left(\sqrt{u} - c_2\right) = a(\beta_{\nu} x^{\nu} + c_1) \tag{3.7.23}$$

Let $\varphi_1 = 0$ then the 4th ansatz reduces (3.7.2) to the ODE

$$-2\omega_2\varphi_2 + \varphi_3 + 2\varphi = 2\lambda a\sqrt{\varphi}\varphi_{23} \tag{3.7.24}$$

We seek for solutions of Equation (3.7.24) in the form

$$\varphi = A(\omega_3)\omega_2^2 \tag{3.7.25}$$

The substitution (3.7.25) into (3.7.24) gives

$$A'(1-4\lambda a\sqrt{A})=2A$$

or

$$A\exp\{-4\lambda\sqrt{A}a\} = c_1\exp\{\omega_3\}$$
(3.7.26)

whence we find solution of Equation (3.7.2)

$$\ln\sqrt{u} - 4\frac{\vec{\sigma}\cdot\vec{\delta}}{\vec{\sigma}\cdot\vec{x}}x_0\sqrt{u} = \vec{\delta}\cdot\vec{x} - \ln\vec{\sigma}\vec{x},$$

$$\vec{\sigma}^2 = \vec{\delta}^2 = 0, \qquad \vec{\sigma}\cdot\vec{\delta} \neq 0.$$
 (3.7.27)

Two more solutions of the Boussinesq equation (3.7.2) obtained with the help of ansatz N8:

$$\sqrt{u} = -\frac{(\vec{\alpha} \cdot \vec{x})^2}{6\lambda\alpha^2 x_0} + c_1 x_0^{-1/3}$$
(3.7.28)

$$\sqrt{u} = -\frac{\vec{x}^2}{10\lambda x_0} + c_2 x_0^{-5/3}, \qquad (3.7.29)$$

where $c_1, c_2, \vec{\alpha}$ are arbitrary constants.

3.8 Symmetry properties of Fokker-Planck equations

Following [76*], we consider symmetry properties of one- and two-dimensional Fokker-Planck (FP) equations.

The one-dimensional FP equation has the form $[77^*]$

$$\frac{\partial u}{\partial t} = -\frac{\partial}{\partial x} \left[A(x,t)u \right] + \frac{1}{2} \frac{\partial^2}{\partial x^2} \left[B(x,t)u \right], \qquad (3.8.1)$$

where u = u(x,t) is the probability density; A and B are differentiable functions. This is the basic equation in the theory of continuous Markovian processes. The following FP equations are of special interest [77*,78*].

(a) diffusion in a gravitational field

$$\frac{\partial u}{\partial t} = \frac{\partial}{\partial x}(gu) + \frac{1}{2}D\frac{\partial^2 u}{\partial x^2};$$
(3.8.2)

(b) the Ornstein-Uhlenbeck process

$$\frac{\partial u}{\partial t} = \frac{\partial}{\partial x}(kxu) + \frac{1}{2}D\frac{\partial^2 u}{\partial x^2}; \qquad (3.8.3)$$

(c) the Rayleigh-type process

$$\frac{\partial u}{\partial t} = \frac{\partial}{\partial x} \left[\left(\gamma x - \frac{\mu}{x} \right) u \right] + \frac{1}{2} \mu \frac{\partial^2 u}{\partial x^2}; \qquad (3.8.4)$$

(d) models in population genetics [78*]

$$\frac{\partial u}{\partial t} = \frac{\alpha}{2} \frac{\partial^2}{\partial x^2} [(x-c)^2 u] + \beta \frac{\partial}{\partial x} [(x-c)u], \qquad (3.8.5)$$

$$\frac{\partial u}{\partial t} = \frac{\partial^2}{\partial x^2} [(1 - x^2)^2 u], \qquad (3.8.6)$$

$$\frac{\partial u}{\partial t} = \frac{\alpha}{2} \frac{\partial^2}{\partial x^2} [x^2 (1 - x^2)^2 u]; \qquad (3.8.7)$$

(e) the Rayleigh process

$$\frac{\partial u}{\partial t} = \frac{\partial}{\partial x} \left[\left(\gamma x - \frac{\mu}{x} \right) u \right] + \mu \frac{\partial^2 u}{\partial x^2}, \qquad (3.8.8)$$

where $D, g, k, \gamma, \alpha, b, c$ are arbitrary constants.

Using Lie's method one can make sure that the maximal invariance group of Equations (3.8.2)-(3.8.7) is a six-parameter one. The same dimension has the invariance group of the heat equation. Although these six-parameter groups are different, they are locally isomorphic. That is why one can reduce Equations (3.8.2)-(3.8.7) to the heat equation.

Theorem 3.8.1. The change of variables

$$u(x,t) = f(x,t)w(y(x,t),\tau(x,t)),$$
 (3.8.9)

where the function f and new independent variables y and τ are as follows:

$$f = \exp\left\{-\frac{g}{D}x - \frac{g^2}{2D}t\right\}, \qquad y = x, \qquad \tau = \frac{D}{2}t;$$
 (3.8.10)

$$f = \exp\{kt\}, \quad y = \exp\{kt\}x, \quad \tau = \frac{D}{4k}\exp\{2kt\}$$
 (3.8.11)

$$f = \exp\{2\gamma t\} x, \qquad y = \exp\{\gamma t\} x, \qquad \tau = \frac{\mu}{4\gamma} \exp\{2\gamma t\}$$
(3.8.12)

$$f = \exp\left\{-\left(\frac{\beta^2}{2\alpha} + \frac{\beta}{2} + \frac{\alpha}{8}\right)t\right\}(x-c)^{-(3/2+\beta/\alpha)}$$
(3.8.13)

$$y = \sqrt{\frac{2}{\alpha}} \ln (x - c), \qquad \tau = t;$$

$$f = \exp\{-t\}(1 - x^2)^{-3/2}, \qquad y = \frac{1}{2} \ln \frac{1 + x}{1 - x}, \qquad \tau = t; \qquad (3.8.14)$$

$$f = \exp\left\{-\frac{\alpha t}{8}\right\} [x(1-x)]^{-3/2}, \qquad y = \ln\frac{x}{1-x} \qquad \tau = \frac{\alpha t}{2} \qquad (3.8.15)$$

reduce Equations (3.8.2) - (3.8.7), correspondingly, to the heat equation

$$w_{\tau} = w_{yy} \tag{3.8.16}$$

The proof can be easily obtained by inspection.

Remark 3.8.1. One can prove a more general statement. Equation (3.8.1) with coefficients

$$A(x,t) = A(x),$$
 $B(x,t) = B = \text{const}$ (3.8.17)

is reduced to the heat equation if and only if

$$B\frac{\partial A}{\partial x} + A^2 = c_2 x^2 + c_1 x + c_0, \qquad (3.8.18)$$

where c_0 , c_1 , c_2 are arbitrary constants. Note that Equations (3.8.2)-(3.8.4) satisfy condition (3.8.18) and Equation (3.8.8) does not. The general solution of Equation (3.8.18), which is a Riccati one, cannot be obtained in quadrature [130]. It will be also noted that FP equation (3.8.1) with coefficients

$$A(x,t) = A(x),$$
 $B(x,t) = B(x)$ (3.8.20)

is reduced to the case (3.8.17) by means of the change of variables (3.8.9), where

$$f = \frac{1}{\sqrt{B(x)}}, \qquad y = \int \frac{dx}{\sqrt{B(x)}}, \qquad \tau = t.$$
 (3.8.21)

Remark 3.8.2. The FP equation of the form

$$\frac{\partial u}{\partial t} = -\frac{\partial}{\partial x} \left[\left(a(t)x + b(t) \right) u \right] + c(t) \frac{\partial^2 u}{\partial x^2}$$
(3.8.22)

was considered in $[28^*, 79^*]$, where a class of solutions of it was obtained by means of rather complicated algebraic method. This result can be easily obtained if we note that Equation (3.8.22) is reduced to the heat Equation (3.8.16) by the substitution (3.8.9) with

$$f = \exp\left\{-\int_{0}^{t} a(s)ds\right\},\$$

$$y = \exp\left\{-\int_{0}^{t} a(s)ds\right\}x - \int_{0}^{t} b(s)\exp\left\{-\int_{0}^{s} a(\xi)d\xi\right\}ds,\qquad(3.8.23)$$

$$\tau = \int_{0}^{t} c(s)\exp\left\{-2\int_{0}^{s} a(\xi)d\xi\right\}ds.$$

Some group-theoretic aspects of FP equations are considered in [27,80*].

Now consider the two-dimensional FP equation which describes the motion of a particle in a fluctuating medium (so-called Brownian movement)

$$\frac{\partial u}{\partial t} = -\frac{\partial}{\partial x}(yu) + \frac{\partial}{\partial y}\left(V'(x)u\right) + \gamma \frac{\partial}{\partial y}\left(yu + \frac{\partial u}{\partial y}\right), \qquad (3.8.24)$$

where u = u(t, x, y), γ is a constant and V(x) is the potential (its gradient defines the exterior force). Equation (3.8.24) is known as the *Kramers* equation [77*].

Theorem 3.8.2. [76*] The maximal invariance group of the free Kramers equation

$$\frac{\partial u}{\partial t} = -\frac{\partial}{\partial x}(yu) + \gamma \frac{\partial}{\partial y}\left(yu + \frac{\partial u}{\partial y}\right)$$
(3.8.25)

is a six-dimensional Lie group generated by the following operators

$$P_{0} = \partial_{t}, \qquad P_{1} = \partial_{x}, \qquad I,$$

$$G_{1} = t\partial_{x} + \partial_{y} + \frac{1}{2}(y + \gamma x)I,$$

$$S_{1} = e^{\gamma t} \left(\frac{1}{\gamma}\partial_{x} + \partial_{y} + yI\right), \qquad T_{1} = e^{-\gamma t} \left(\frac{1}{\gamma}\partial_{x} - \partial_{y}\right),$$
(3.8.26)

which satisfy the commutation relations

$$[P_0, G_1] = P_1, \qquad [P_0, S_1] = \gamma S_1, [P_0, T_1] = -\gamma T_1, \qquad [P_1, G_1] = \frac{1}{2}\gamma I, \qquad [T_1, S_1] = -I$$
 (3.8.27)

(the rest of the commutators are equal to zero). The proof can be obtained by Lie's method.

Remark 3.8.3. One can prove a more general statement: the widest symmetry group of Equation (3.8.24) is achieved when $V'(x) = c_1 x + c$ (c_1 and c are arbitrary constants) and it is a six-parameter group.

Remark 3.8.4. The change of variables

$$u = w(\tau, \xi, \eta), \qquad \tau = t, \qquad \xi = x - \frac{c}{\gamma}t, \qquad \eta = y - \frac{c}{\gamma}$$
 (3.8.28)

reduces Equation (3.8.24) with V'(x) = c to the free Kramers equation (3.8.25).

Let us write down the final transformations generated by operators (3.8.26). Operators P_0 and P_1 generate translations on variables t and x; I generates the identical transformation;

 G_1 generates

$$t' = t, x' = x + at, y' = y + a, u'(x') = \exp\left\{-\frac{1}{2}\left[ay + \frac{a^2}{2}(1 + \gamma t) + \gamma ax\right]\right\}u(x); (3.8.29)$$

 S_1 generates

$$t' = t, x' = x + \frac{b}{\gamma} e^{\gamma t}, y' = y + b e^{\gamma t}, (3.8.30)$$
$$u'(x') = \exp\left\{by e^{\gamma t} - \frac{b^2}{2} e^{2\gamma t}\right\} u(x);$$

 T_1 generates

$$t' = t, \qquad x' = x + \frac{\theta}{\gamma} e^{-\gamma t}, \qquad y' = y - \theta e^{-\gamma t},$$

$$u'(x') = u(x); \qquad (3.8.31)$$

where a, b, θ are group parameters. It is appropriate to write the corresponding formulae of generating solutions which follow from (3.8.29)-(3.8.31):

$$u_{II}(t, x, y) = \exp\left\{\frac{a}{2}\left[y + \frac{a}{2}(1 + \gamma t) + \gamma x\right]\right\}u_{I}(t', x', y')$$
(3.8.32)
$$u_{II}(t,x,y) = \exp\left\{-bye^{\gamma t} + \frac{b^2}{2}e^{2\gamma t}\right\}u_I(t',x',y')$$
(3.8.33)

$$u_{II}(t, x, y) = u_{I}(t', x', y')$$
(3.8.34)

where t', x', y' are given in (3.8.29)-(3.8.31) respectively.

It will be noted that transformations (3.8.29) are just the Galilean ones as soon as the variable y in the Kramers equation is taken to be the velocity of the particle.

A well-known solution of the Kramers equation (3.8.24) is the Boltzmann distribution

$$u(x,y) = \mathcal{N} \exp\left\{-V(x) - \frac{1}{2}y^2\right\}$$
(3.8.35)

(\mathcal{N} is a normalization constant). It is a stationary solution. Applying to (3.8.35) with V = 0 formulae (3.8.32)-(3.8.34), one can easily obtain new non-stationary solutions of Equation (3.8.25).

Let us consider the ansatz invariant with respect to the operator S_1 from (3.8.26)

$$u(t, x, y) = \exp\left\{-\frac{y^2}{2}\right\}\varphi(\omega_1, \omega_2), \qquad \omega_1 = t, \qquad \omega_2 = \gamma x - y \qquad (3.8.36)$$

Substituion of (3.8.36) into (3.8.25) gives rise to the heat equation

$$\frac{\partial\varphi}{\partial\omega_1} - \gamma \frac{\partial^2\varphi}{\partial\omega_2^2} = 0 \tag{3.8.37}$$

The simplest solution of (3.8.37) is $\varphi = \text{const}$, but it is the solution that leads, together with the ansatz (3.8.36), to the Boltzmann distribution (3.8.35). It is clear that by using solutions of the heat equation (3.8.37) and the ansatz (3.8.36) one can construct many partial solutions of Equation (3.8.25). For example, the fundamental solution of (3.8.37) and (3.8.36) results in the following solution of Equation (3.8.25)

$$u(t, x, y) = \frac{1}{\sqrt{4\pi\gamma t}} \exp\left\{-\frac{y^2}{2} - \frac{(\gamma x - y)^2}{4\gamma t}\right\}$$
(3.8.38)

The operator T_1 from (3.8.26) leads to the ansatz

$$u = \widetilde{\varphi}(\omega_1, \omega_2), \qquad \omega_1 = t, \qquad \omega_2 = \gamma x + y$$
 (3.8.39)

which reduces Equation (3.8.25) to the heat equation (3.8.37), where, $\varphi = \exp\{\gamma \omega_1\}\widetilde{\varphi}(\omega_1, \omega_2)$.

A great number of partial solutions of Equation (3.8.25) can be found by means of the method described in [24^{*}] (see (2.3.58)). For example, starting from $u_0 = e^{\gamma t}$, we find

$$u_1 = 2G_1 e^{\gamma t} = e^{\gamma t} (\gamma x + y),$$

$$u_2 = G_1 u_1 = e^{\gamma t} [(\gamma t + 1) + \frac{1}{2} (\gamma x + y)^2], \dots$$
(3.8.40)

Analogously, by means of the operator T_1 from (3.8.26), we find, starting from $u_0 = \exp \left\{-y^2/2\right\}$,

$$u_{1} = T_{1}u_{0} = y \exp\left\{-\left(\gamma t + \frac{y^{2}}{2}\right)\right\},$$

$$u_{2} = T_{1}u_{1} = (y^{2} - 1) \exp\left\{-\left(2\gamma t + \frac{y^{2}}{2}\right)\right\}, \dots$$
(3.8.41)

Solutions (3.8.38), (3.8.40), and (3.8.41) can be multiplied by the formulae of generating solutions (3.8.32)-(3.8.34).

Chapter 4

Systems of PDEs Invariant Under Galilei Group

In the present chapter we consider linear and nonlinear systems of PDEs invariant under various representations of the Galilei group and its generalizations (such as the extended Galilei group, the Schrödinger group). Sets of Sch(1,3)and G(1,3)-nonequivalent ansatze are constructed. A wide class of linear and nonlinear Sch(1,3)-invariant systems of PDEs is described. Lame equations are studied: superalgebra of symmetry is found and a Galilei-invariant generalization is constructed. Gas dynamics and Navier-Stokes equations are considered. Exact solutions of some enumerated above equations are found.

4.1. The Schrödinger group Sch(1,3): nonequivalent ansatze and final transformations for fields of arbitrary spin

The maximal group of point transformations of the Schrodinger equation

$$\overset{\,\,{}_\circ}{S}\psi \equiv \left(P_0 - \frac{\vec{P}^2}{2m}\right)\psi = 0,\tag{4.1.1}$$

where $\psi = \psi(x_0, \ldots, x_3)$ is complex wave function, is called the Schrödinger group. It is 13-dimensional and containes the Galilei group, scale and projective transformations. The basis elements of the corresponding algebra ASch(1,3) are written in (3.3.25). Further, we will consider the case of the multi-component ψ -function.

From the group-theoretic point of view an analogy between relativistic and nonrelativistic mechanics can be treated as follows:

Basic symmetry group of			
relativistic mechanics	nonrelativistics mechanics		
Poincare group $P(1,3)$	Galilei group $G(1,3)$		
extended Poincare group	extented Galilei group		
$\widetilde{P}(1,3) = \{P(1,3), D\}$	$\widetilde{G}(1,3) = \{G(1,3), D\}$		
conformal group	Schrödinger group		
$C(1,3) = \{\widetilde{P}(1,3), K_{\mu}\}$	Sch $(1,3) = {\widetilde{G}(1,3), \Pi}$		

The most general form of basis operators realizing linear representation of the Schrödinger algebra ASch(1,3) is [78, 87*]:

$$P_{0} = i\partial_{0}, \qquad P_{a} = -i\partial_{a}, \qquad M = im, J_{a} = (\vec{x} \times \vec{p})_{a} + \hat{S}_{a}, \qquad G_{a} = x_{0}P_{a} - mx_{a} + \lambda_{a}, D = 2x_{0}P_{0} - \vec{x} \cdot \vec{P} + \lambda_{0}, \Pi = x_{0}D - x_{0}^{2}P_{0} + \frac{1}{2}m\vec{x}^{2} - \vec{\lambda} \cdot \vec{x},$$
(4.1.2)

where m is a constant; \hat{S} , λ_a , λ_0 are numerical matrices. Operators (4.1.2) satisfy commutation rules

$$\begin{split} & [P_0, P_a] = [P_0, J_a] = [P_a, P_b] = [G_a, G_b] = [D, J_a] = 0 \\ & [\Pi, J_a] = [\Pi, G_a] = [P_0, M] = [P_a, M] = [G_a, M] = 0 \\ & [D, M] = [\Pi, M] = 0 \\ & [J_a, J_b] = i\epsilon_{abc}J_c, \qquad [P_a, J_b] = i\epsilon_{abc}P_c, \\ & [J_a, G_b] = i\epsilon_{abc}G_c, \qquad [P_a, G_b] = M\delta_{ab}, \\ & [P_0, G_a] = iP_a, \qquad [P_0, D] = 2iP_0, \qquad [P_a, D] = iP_a, \\ & [D, G_a] = iG_a \qquad [P_0, \Pi] = iD, \qquad [P_a, \Pi] = iG_a, \qquad [D, \Pi] = 2i\Pi. \end{split}$$

When $\lambda_0 = \frac{3}{2}i$, $\lambda_a = 0$ then (4.1.2), (4.1.3) coincide with the maximal IA of the Schrödinger equation (4.1.1) (2.3.25).

As it follows from (4.1.3) matrices \hat{S}_a , λ_0 , λ_a should satisfy the commutation relations

$$\begin{split} [\widehat{S}_{a}, \widehat{S}_{b}] &= i\epsilon_{abc}\widehat{S}_{c}, \\ [\lambda_{a}, \lambda_{b}] &= 0, \qquad [\lambda_{a}, \widehat{S}_{b}] = i\epsilon_{abc}\widehat{S}_{c}, \\ [\lambda_{0}, \widehat{S}_{a}] &= 0, \qquad [\lambda_{0}, \lambda_{a}] = i\lambda_{a} \end{split}$$
(4.1.4)

About finite-dimensional representations of algebra (4.1.4) see the following paragraph.

Solutions invariant under algebra (4.1.2) one can seek for in the form (2.1.7), matrices A(x) and invariant variables $\omega(x)$ being determined from the equations like (2.1.12), (2.1.13). Without going into details of these calculations we present in the Table 4.1.1 ansatze invariant under one-dimensional subalgebras of ASch(1,3). It will be noted that these subalgebras are Sch(1,3)nonequivalent on the field of complex numbers.

Ν	Algebra	Invariant variables ω	Ansatze $\psi(x) =$
1.	М	x_0, x_1, x_2, x_3	0
2.	P_1	x_0, x_2, x_3	$arphi(\omega)$
3.	D	$rac{x_1}{\sqrt{x_0}},rac{x_2}{\sqrt{x_0}},rac{x_3}{\sqrt{x_0}}$	$x_0^{i\lambda_0/2}arphi(\omega)$
4.	J_3	$x_1, (x_1^2 + x_2^2)^{1/2}, x_3$	$\expig\{i\widehat{S}_3 \arctan(x_1/x_2)ig\} arphi(\omega)$
5.	$J_3 + \alpha M$	$x_0, (x_1^2 + x_2^2)^{1/2}, x_3$	$\exp\Bigl\{i(\widehat{S}_3+lpha m)\cdot$
			$\cdot \arctan rac{x_1}{x_2} \Big\} arphi(\omega)$
6.	$J_3 + \alpha P_3$	$x_0, (x_1^2 + x_2^2)^{1/2},$	$\exp\left\{i\widehat{S}_3 \arctan(x_1/x_2)\right\} arphi(\omega)$
		$x_3 - lpha \arctan rac{x_1}{x_2}$	
7.	$J_3 + \alpha D$	$\ln (x_1^2 + x_2^2) - 2\alpha \arctan \frac{x_1}{x_2},$	$(x_1^2 + x_2^2)^{i\lambda_0/2} \exp\{i\widehat{S}_3\cdot$
		$\frac{x_1^2 + x_2^2}{\sqrt{x_0}x_2}, \frac{x_3}{\sqrt{x_0}}$	$\cdot \arctan(x_1/x_2) ig arphi(\omega)$
8.	$J_3 + \alpha D +$	$\ln (x_1^2 + x_2^2) - 2\alpha \arctan \frac{x_1}{x_2},$	$x_0^{i\beta m/2} (x_1^2 + x_2^2)^{i\lambda_0/2} \exp\{i\widehat{S}_3\cdot$
	$+\beta M$	$rac{x_1^2+x_2^2}{\sqrt{x_0}x_3},rac{x_3}{\sqrt{x_0}}$	$\cdot \arctan(x_1/x_2) ig arphi(\omega)$
9.	$M + \beta D$	$\frac{x_1}{\sqrt{x_0}}, \frac{x_2}{\sqrt{x_0}}, \frac{x_3}{\sqrt{x_0}}$	$x_0^{rac{1}{2}(i\lambda_0+eta m)}arphi(\omega)$
10.	$G_1 + \alpha P_2$	$x_0, \alpha x_1 - x_0 x_2, x_3$	$\exp\left\{im\left(rac{x_1^2}{2x_0}-rac{x_2}{lpha m}\lambda_1 ight) ight\}\cdot$
			$\cdot arphi(\omega)$

 Table 4.1.1. One-dimensional subalgebras of ASch(1,3) and corresponding ansätze.

Here $\alpha > 0$, $\beta \neq 0$ are arbitrary constants; an entry x_0 means, for example, if

$$\lambda_0 = rac{i}{2} egin{pmatrix} n & 0 \ 0 & k \end{pmatrix},$$

where n and k are some numbers, then

$$x_0^{i\lambda_0/2} = \begin{pmatrix} x_0^{-n/4} & 0\\ 0 & x_0^{-k/4} \end{pmatrix}$$

Transformations generated by operators (4.1.2) can be used for obtaining other solutions of Sch(1,3)-invariant PDEs in the same spirit as it was done in Paragraphs 1.4, 2.3. In Table 4.1.2 we present corresponding forulae.

Table	4.1.2 .	Final	transformations	of	Sch(1,3)	group	\mathbf{and}	formulae	of
		genera	ating solutions.						

N	Operator	Transformations		Formulae of GS
		$x \rightarrow x'$	$\psi(x) \rightarrow \psi'(x') =$	$\psi_{II} =$
1-4	$P_{\mu} = \{P_0, P_a\}$	$x'_{\mu} = x_{\mu} + a_{\mu}$	$=\psi(x)$	$=\psi_I(x')$
5-7	$J = \alpha_a J_a$	$x'_0 = x_0,$	$= \exp \left\{ -i ec lpha \cdot ec S ight\} \cdot$	$= \exp \left\{ i \vec{\alpha} \vec{\hat{S}} \right\} \cdot$
		$\vec{x}' = \vec{x} \cos \alpha +$	$\dot{\psi}(x)$	$\dot{\psi}_I(x')$
ļ		$+\frac{\vec{x}\times\vec{\alpha}}{1}\sin\alpha+$		
		$+\frac{\vec{\alpha}(\vec{\alpha}\cdot\vec{x})}{\alpha^2}$		
		$(1 - \cos \alpha)$		
8-10	$G = v_a G_a$	$x'_0 = x_0,$	$=\exp\{im\cdot$	$=\exp\{-im\cdot$
			$\cdot (\vec{v}\vec{x} + \frac{1}{2}\vec{v}^2x_0) -$	$\cdot (\vec{v}\vec{x} + \frac{1}{2}\vec{v}^2x_0) +$
		$\vec{x}' = \vec{x} + \vec{v}x_0$	$-i\vec{\lambda}\vec{v}\}\psi(x)$	$+\vec{\lambda}\vec{v}\}\psi_I(x')$
11	M	$x'_{\mu} = x_{\mu}$	$=\exp\{im heta\}\psi(x)$	$=\exp\{-im\theta\}\cdot$
10		1 28		$\psi_I(x')$
		$\begin{aligned} x_0' &= e^{2\rho} x_0, \\ \vec{z}_0' &= \theta \end{aligned}$	$=\exp\{i\beta\lambda_0\}\psi(x)$	$= \exp\{-i\beta\lambda_0\}$
		$x = e^{\mu}$		$\psi_I(x')$
13	П	$x_0'=\frac{x_0}{1-\theta x_0},$	$=(1- heta x_0)^{-i\lambda_0}$.	$=\exp\left\{-\frac{1}{1-\theta x_0}\right\}$
		$\vec{x}' = \frac{\vec{x}}{1 - \theta x_0}$	$\cdot \exp\Big\{rac{i heta}{1- heta x_0}\cdot$	$\left \cdot\left(\frac{mec{x}^2}{2}-ec{\lambda}\cdotec{x} ight) ight)\right\}$
			$\Big \cdot\left(rac{mec{x}^2}{2}-ec{\lambda}\cdotec{x} ight)\Big\}\cdot$	$(1- heta x_0)^{i\lambda_0}$
			$\dot{\psi}(x)$	$\cdot \psi_I(x')$

Here a_{μ} , $\vec{\alpha} = (\alpha_1, \alpha_2, \alpha_3)$, \vec{v} , β , θ are arbitrary real constants.

One can make sure in validity of the final transformations formulae of Table 4.1.2 by straightforward verification of corresponding Lie equations. In particular, Lie equations for G_a (see (4.1.2)) have the form

$$\begin{aligned} \frac{\partial x'_b}{\partial v_a} &= x'_0 \delta_{ab}, \qquad \frac{\partial x'_0}{\partial v_a} = 0; \qquad x'_0 \left|_{v_a = 0} = x_0, \qquad x'_a \right|_{v_b = 0} = x_a; \\ \frac{\partial \psi'(x')}{\partial v_a} &= (imx'_a - i\lambda_a)\psi'(x'), \qquad \psi'(x') \left|_{v_a = 0} = \psi(x) \right. \end{aligned}$$

and it is easy to see that they are identically satisfied by $x_a', \,\psi'(x')$ N8–10 of

Table 4.1.2. Indeed,

$$\begin{split} \frac{\partial}{\partial v_a}(x_b + v_b x_0) &\equiv \delta_{ab} x_0, \qquad x'_0 = x_0; \\ \frac{\partial}{\partial v_a} \psi'(x') &\equiv \frac{\partial}{\partial v_a} \left[\exp\left\{ im\left(\vec{v} \cdot \vec{x} + \frac{1}{2}\vec{v}^2 x_0\right) - i\vec{\lambda}\vec{v} \right\} \psi(x) \right] = \\ &= [im(x_a + v_a x_0) - i\lambda_a] \psi'(x') \equiv i(mx'_a - \lambda_a) \psi'(x'). \end{split}$$

In the case of operator Π Lie equations have the form

$$\frac{dx'_0}{d\theta} = x'^2_0, \qquad \frac{d\vec{x}'}{d\theta} = x'_0\vec{x}'; \qquad x'_0 \Big|_{\theta=0} = x_0, \qquad \vec{x}' \Big|_{\theta=0} = \vec{x};$$
$$\frac{\psi'(x')}{d\theta} = i\left(x'_0\lambda_0 + \frac{m\vec{x}'^2}{2} - \vec{\lambda} \cdot \vec{x}'\right)\psi'(x'), \qquad \psi'(x') \Big|_{\theta=0} = \psi(x).$$

Let us make sure that x', $\psi'(x')$ N13 of Table 4.1.2 are solutions of these equations. So, we calculate

$$\frac{d}{d\theta}x'_{0} \equiv \frac{d}{d\theta}\left(\frac{x_{0}}{1-\theta x_{0}}\right) = \left(\frac{x_{0}}{1-\theta x_{0}}\right)^{2} \equiv x'_{0}^{2}; \qquad \frac{d}{d\theta}\vec{x}' \equiv \frac{d}{d\theta}\left(\frac{\vec{x}}{1-\theta x_{0}}\right) = \\ = \frac{x_{0}}{1-\theta x_{0}} \cdot \frac{\vec{x}}{1-\theta x_{0}} \equiv x'_{0}\vec{x}'; \\ \frac{d}{d\theta}\psi'(x') \equiv \frac{d}{d\theta}\left[(1-\theta x_{0})^{-i\lambda_{0}}\exp\left\{\frac{i\theta}{1-\theta x_{0}}\left(\frac{m\vec{x}^{2}}{2}-\vec{\lambda}\cdot\vec{x}\right)\right\}\psi(x)\right] = \\ = (1-\theta x_{0})^{-i\lambda_{0}}\left[i\lambda_{0}\frac{x_{0}}{1-\theta x_{0}}+\frac{i}{(1-\theta x_{0})^{2}}\left(\frac{m\vec{x}'^{2}}{2}-\vec{\lambda}\cdot\vec{x}\right)\right]\cdot \\ \cdot \exp\left\{\frac{i\theta}{1-\theta x_{0}}\left(\frac{m\vec{x}'^{2}}{2}-\vec{\lambda}\cdot\vec{x}\right)\right\} \equiv i\left(\lambda_{0}x'_{0}+\frac{m\vec{x}'^{2}}{2}-\vec{\lambda}\cdot\vec{x}'\right)\psi'(x')\right\}$$

Here we have used the equality $(\lambda_0 \neq 0)$

$$(1-\theta x_0)^{-i\lambda_0-1}(\vec{\lambda}\cdot\vec{x})=(\vec{\lambda}\cdot\vec{x})(1-\theta x_0)^{-i\lambda_0}.$$

Now we show a simple way of constructing Sch(1,3)-ungenerative solutions. Let us start from a projective invariant ansatz

$$\psi(x) = x_0^{i\lambda_0} \exp\left\{i\left(\frac{m\vec{x}^2}{2x_0} - \vec{\lambda}\vec{x}\right)\right\}\varphi(\omega), \qquad \omega = \frac{\vec{\beta}\cdot\vec{x}}{x_0} + \beta_0, \tag{4.1.5}$$

which results from the equations

$$\Pi A(x) \equiv \left(x_0^2 P_0 - x_0 \vec{x} \cdot \vec{P} + \frac{m \vec{x}^2}{2} + \lambda_0 x_0 - \vec{\lambda} \vec{x} \right) A(x) = 0,$$

Chapter 4. Systems of PDEs Invariant Under Galilei Groups

$$(x_0^2 P_0 - x_0 \vec{x} \cdot \vec{P}) \omega(x) = 0; \qquad \psi(x) = A(x) \varphi(\omega).$$

Having done gauge and translation transformations (see Table 4.1.2, N11 and N1-4) on the ansatz (4.1.5) we obtain as a result a Sch(1,3)-ungenerative ansatz

$$\psi(x) = (x_0 + a_0)^{i\lambda_0} \exp\left\{im\left(\frac{(\vec{x} + \vec{a})^2}{2(x_0 + a_0)} + \alpha\right) - i\vec{\lambda}(\vec{x} + \vec{a})\right\}\varphi(y),$$

$$y = \frac{\vec{\beta} \cdot (\vec{x} + \vec{a})}{x_0 + a_0} + \beta_0; \qquad \beta_0, \ \beta_a, \ a_0, \ a_a, \ \alpha \text{ are constants.}$$
(4.1.6)

Ungenerativity of (4.1.6) with the help of GS formulae N1-12 of Table 4.1.2 is quite obvious. One can easy prove ungenerativity of (4.1.6) with respect to GS formulae N13 if use is made of the identity

$$\exp\left\{i\frac{\theta}{1-\theta x_0}(\vec{\lambda}\cdot\vec{x})\right\}(1-\theta x_0)^{i\lambda_0}=(1-\theta x_0)^{i\lambda_0}\exp\{i\theta\vec{\lambda}\cdot\vec{x}\}$$

and to note that this procedure of GS is reduced to transformations in the space of parameters:

$$\begin{split} \vec{\beta} &\to \vec{\beta}' = \frac{\vec{\beta}}{1 - \theta a_0}, \qquad \beta_0 \to \beta'_0 = \beta_0 - \frac{\theta}{1 - \theta a_0} (\vec{\beta} \cdot \vec{a}), \\ \vec{\alpha} \to \vec{\alpha}' = \frac{\vec{a}}{1 - \theta a_0}, \qquad a_0 \to a'_0 = \frac{a_0}{1 - \theta a_0}, \\ \alpha \to \alpha' = \alpha + \frac{\theta}{1 - \theta a_0} \cdot \frac{\vec{a}^2}{2}, \\ \varphi \to \varphi' = (1 - \theta a_0)^{i\lambda_0} \varphi. \end{split}$$

In conclusion, we present in Table 4.1.3 G(1,3)-nonequivalent ansatze. One dimensional subalgebras of AG(1,3) are found in [12].

4.2. Linear and nonlinear systems of PDEs invariant under the Schrödinger group Sch(1,3).

Following [94] we shall describe linear and nonlinear systems of PDEs for complex multi-component wave function ψ invariant with respect to ASch(1,3)

N	Algebra	Invariant variables	Ansatze
		$\omega = \{\omega_1, \omega_2, \omega_3\}$	$\psi(x) =$
1.	P_0	x_1, x_2, x_3	$= \varphi(\omega)$
2.	$P_0 + \beta m$	x_1, x_2, x_3	$=\exp\{imeta x_0\}arphi(\omega)$
3.	P_1	x_0, x_2, x_3	$=\varphi(\omega)$
4.	G_1	x_0, x_2, x_3	$= \exp\left\{\frac{i}{x_0}(\frac{1}{2}mx_1^2 - \lambda_1 x_1)\right\}\varphi(\omega)$
5.	$G_1 + \alpha P_0$	$2\alpha x_1 + x_0^2, x_2, x_3$	$=\exp\left\{-\frac{im}{\alpha}(x_0x_1+\frac{1}{3\alpha}x_0^3-\right)$
			$-\frac{1}{m}\lambda_1 x_0\}\varphi(\omega)$
6.	$G_1 + \alpha P_2$	$\alpha x_1 - x_0 x_2, x_0, x_3$	$= \exp \left\{ im \left(rac{x_1^2}{2x_0} - rac{\lambda_1 x_2}{lpha m} ight) ight\} arphi(\omega)$
7.	J_3	$(x_1^2+x_2^2)^{1/2}, x_0, x_3$	$=\exp\left\{i\widehat{S}_{3}\arctanrac{x_{1}}{x_{2}} ight\}arphi(\omega)$
8.	$J_3 + \alpha m$	$(x_1^2+x_2^2)^{1/2}, x_0, x_3$	$= \exp \left\{ i(\widehat{S}_3 + \alpha m) \cdot \right.$
			$\cdot \arctan \frac{x_1}{x_2} \Big\} \varphi(\omega)$
9.	$J_3 + \alpha P_0$	$(x_1^2 + x_2^2)^{1/2}, x_3,$	$= \exp\left\{i\widehat{S}_{3}\arctan\frac{x_{1}}{x_{2}}\right\}\varphi(\omega)$
		$x_0 + \alpha \arctan \frac{x_1}{x_2}$	_
10.	$J_3 + \alpha P_0 +$	$(x_1^2+x_2^2)^{1/2}, x_3,$	$= \exp \Big\{ i(\widehat{S}_3 + \beta m) \arctan rac{x_1}{x_2} \Big\} arphi(\omega)$
	$+\beta m$	$x_0 + \alpha \arctan \frac{x_1}{x_2}$	2
11.	$J_3 + \alpha P_3$	$(x_1^2 + x_2^2)^{1/2}, x_0,$	$=\exp\left\{i\widehat{S}_{3}\arctanrac{x_{1}}{x_{2}} ight\}arphi(\omega)$
		$x_3 - lpha \arctan rac{x_1}{x_2}$	~ 2
12.	$J_3 + \alpha G_3$	$(x_1^2 + x_2^2)^{1/2}, x_0,$	$= \exp \Big\{ i ig(\widehat{S}_3 + lpha \lambda_3 ig) rctan rac{x_1}{x_2} + ig)$
		$x_3-lpha x_0 rctan {x_1\over x_2}$	$+imx_3^2/2x_0\Big\}arphi(\omega)$
13.	$J_3 + \alpha P_0 +$	$(x_1^2+x_2^2)^{1/2},$	$=\exp\left\{i(\widehat{S}_3+eta\lambda_3)\arctanrac{x_1}{x_2}- ight.$
	$+\beta G_3$	$\beta x_0^2 + 2\alpha x_3,$	$-rac{imeta}{lpha}\left(x_0x_3+rac{eta x_0^3}{3lpha} ight)^2 ight\}arphi(\omega)$
		$x_0 + lpha \arctan rac{x_1}{x_2}$	× /

Table 4.1.3. One-dimensional subalgebras of AG(1,3) and corresponding ansatze.

(4.1.2) and satisfying the Schrödinger Equation (4.1.1). In other words, we shall find additional conditions having been putted on ψ , select from the manifold of solutions of the Schrödinger equation (4.1.1) submanifold on which it is realized a representation of Sch(1,3)-group with nonzero mass m and spin s.

Theorem 4.2.1. [94]. System of Equations (4.1.1) for multi-component complex function ψ is invariant under ASch(1,3) (4.1.2) iff

$$L\psi \equiv \left(\lambda_0 - \frac{3}{2}i - \frac{1}{m}\vec{\lambda} \cdot \vec{P}\right)\psi = 0 \tag{4.2.1}$$

$$\Lambda \psi \equiv \vec{\lambda}^2 \psi \equiv \left(\lambda_1^2 + \lambda_2^2 + \lambda_3^2\right) \psi = 0 \tag{4.2.2}$$

Proof. Necessity. Let us use condition of invariance in the form (10), (11), so that

$$\left[\overset{\circ}{S}, Q \right] \psi \left| \underset{\overset{\circ}{S}\psi=0}{\overset{\circ}{=} 0} \right.$$

$$(4.2.3)$$

where Q is any operator from (4.1.2), \check{S} is the Schrödinger operator (4.1.1). As a result we have

$$[\check{S}, D] = 2i\check{S},$$

 $[\check{S}, \Pi] = 2ix_0\check{S} + iL,$ (4.2.4)
 $[\check{S}, Q_\ell] = 0, \qquad Q_\ell = \{P_0, P_a, J_a, G_a, M\},$

where operator L is given in (4.2.1), whence it follows (4.2.1).

Sufficiency. One has to calculate commutators of operator L (4.2.1) with operators Q (4.1.2). It yields

$$\begin{split} [L,Q_s] &= 0, \qquad Q_s = \{P_0,\,P_a,\,J_a,G_a,\,M,D\},\\ [L,\Pi] &= \frac{1}{im}\Lambda, \qquad \Lambda = \lambda_1^2 + \lambda_2^2 + \lambda_3^2 \end{split}$$

and to complete the proof it remains to make sure that

$$\begin{split} [\Lambda,Q_s] &= 0, \qquad Q_\ell = \{P_0,\,P_a,\,J_a,\,G_a,\,M\},\\ [\Lambda,D] &= -2i\Lambda, \qquad [\Lambda,\Pi] = -2ix_0\Lambda \end{split}$$

So, the system of PDEs (4.1.1), (4.2.1), (4.2.2), which for the sake of convenience we write down separately

$$\dot{S}\psi \equiv \left(P_0 - \frac{\vec{P}^2}{2m}\right)\psi = 0,$$

$$L\psi \equiv \left(\lambda_0 - \frac{3}{2}i - \frac{1}{m}\vec{\lambda}\cdot\vec{P}\right)\psi = 0$$

$$\Lambda\psi \equiv \left(\lambda_1^2 + \lambda_2^2 + \lambda_3^2\right)\psi = 0$$
(4.2.5)

is invariant under ASch(1,3) (4.1.2), that is, on the manifold of solutions of the system (4.2.5) a representation of Sch(1,3) group with nonzero mass mand spin s is realized. In other words, Schrödinger equation (4.1.1) for multicomponent function is conditionally invariant under ASch(1,3) and conditions ensuring this invariance are (4.2.1), (4.2.2). (About conditional symmetry see §5.7).

It is clear from the stated above that the problem of describing explicit form of systems (4.2.5) is reduced to the problem of describing finite-dimensional representations of AE(1,3) (4.1.4), which in turn is reduced to the problem of describing finite-dimensional representations of AE(3):

$$\begin{split} & [\widehat{S}_a, \widehat{S}_b] = i\epsilon_{abc}\widehat{S}_c, \\ & [\lambda_a, \lambda_b] = 0, \qquad [\lambda_a, \widehat{S}_b] = i\epsilon_{abc}\widehat{S}_c \end{split} \tag{4.2.6}$$

Algebra AE(3) (4.2.6) has two invariant operators [78]

$$C_1 = \widehat{S}_a \lambda_a, \qquad C_2 = \Lambda \equiv \lambda_a \lambda_a, \qquad (4.2.7)$$

which in the case of finite-dimensional representations are nilpotent matrices, that is

$$C_1^{N_1} = 0, \qquad C_2^{N_2} = 0 \tag{4.2.8}$$

where N_1 , N_2 are some integers.

Following [78] we shall consider only those representations of algebra (4.2.6) which include not more than two non-equivalent representations when reduced on AO(3). In this case matrices \hat{S}_a can be taken in the form

$$\widehat{S}_{a} \equiv \widehat{S}_{a}^{(n\,m)} = \begin{pmatrix} I_{n} \otimes S_{a} & \widehat{0}^{+} \\ \widehat{0} & I_{m} \otimes \Sigma_{a} \end{pmatrix}$$
(4.2.9)

where S_a and Σ_a are generators of irreducible representations of AO(3), that is matrices satisfying relations

$$[S_a, S_b] = i\epsilon_{abc}S_c, \qquad S_aS_a = s(s+1),$$

$$[\Sigma_a, \Sigma_b] = i\epsilon_{abc}\Sigma_c, \qquad \Sigma_a\Sigma_a = s'(s'+1);$$

(4.2.10)

 I_n and I_m are unit matrices of dimension $n \times n$, $m \times m$, and $\hat{0}$ is the zero-matrix of dimension $m \times n$; symbol \otimes means direct (Kronecker) product of matrices.

Theorem 4.2.2. [78]. Any (to within equivalence) indecomposable finitedimensional representation of AE(3) (4.2.6) including not more than two nonequivalent representations of subalgebra AO(3) and satisfying additional condition such that invariant operator C_1 (4.2.7) has not more than two orthogonal eigensubspaces, can be enumerated by the set of integers (n, m, α) , provided $\alpha = 1, 2; n \le 4, m \le 4; |n - m| \le 2, n \cdot m \ne 9.$

Explicit form of corresponding matrices \hat{S}_a and λ_a is given by

$$\widehat{S}_{a} = \widehat{S}_{a}^{(n\,m\,\alpha)} = \widehat{S}_{a}^{(n\,m\,\alpha)}; \qquad \lambda_{a} = \lambda_{a}^{(n\,m\,\alpha)}$$

$$\lambda_{a}^{(n\,m\,1)} = \frac{1}{2s} \begin{pmatrix} a_{1}^{n\,m} \otimes S_{a} & a_{2}^{n\,m} \otimes K_{a} \\ a_{3}^{n\,m} \otimes K_{a}^{+} & a_{4}^{n\,m} \otimes \Sigma_{a} \end{pmatrix}; \qquad \lambda_{a}^{(n\,m\,2)} = \left[\lambda_{a}^{(n\,m\,2)}\right]^{+},$$
(4.2.11)

where matrices $\widehat{S}_{a}^{(n\,m)}$ are determined in (4.2.9); s' = s - 1; K_{a} are matrices of dimension $(2s - 1) \times (2s + 1)$ satisfying relations

$$K_a S_b - \Sigma_b K_a = i\epsilon_{abc} K_c,$$

$$S_a S_b + K_a^+ K_b = i\epsilon_{abc} S_c + s^2 \delta_{ab};$$
(4.2.12)

 a_{1-4}^{nm} are matrices with matrix elements

$$(a_{1}^{n\,m})_{ij} = \begin{cases} \frac{\delta_{i-1\,j}}{s-1}, & n \ge m, \quad i,j \le n \\ \frac{s-1}{s+1} \delta_{i-1\,j}, & n < m \end{cases}$$

$$(a_{2}^{n\,m})_{ij} = \begin{cases} -(2s-1)^{-1/2} \delta_{i-2\,j}, & n > m, \quad i \le n, \quad j \le m \\ (2s+1)^{-1/2} \delta_{i\,j}, & n < m \\ k\delta_{i\,j+1}, & n = m \end{cases}$$

$$(a_{3}^{n\,m})_{ij} = \begin{cases} (2s-1)^{-1/2} \delta_{i\,j}, & n > m, \quad i \le m, \quad j \le n \\ (2s+1)^{-1/2} \delta_{i-2\,j}, & n \le m \end{cases}$$

$$(a_{4}^{n\,m})_{ij} = \begin{cases} \frac{s+1}{s-1} \delta_{i-1\,j}, & n \ge m \\ \delta_{i-1\,j}, & n < m \end{cases}$$

$$(4.2.13)$$

where k is an arbitrary parameter. Representations $D(n, m, \alpha = 1)$ and $D(n, m, \alpha = 2)$ are equivalent if and only if |n - m| = 1.

Proof. One can show that (4.2.11) defines the general form of matrices λ_a satisfying commutations relations of AE(3) (4.2.6) when matrices \hat{S}_a have the form (4.2.9). From the commutativity of matrices λ_a and making use of relations

$$K_{a}S_{b} - K_{b}S_{a} = i(s+1)\epsilon_{abc}K_{c},$$

$$\Sigma_{a}K_{b} - \Sigma_{b}K_{a} = i(1-s)\epsilon_{abc}K_{c},$$

$$K_{a}K_{b}^{+} - K_{b}K_{a}^{+} = -i(2s+1)\epsilon_{abc}\Sigma_{c},$$

$$K_{a}^{+}K_{b} - K_{b}^{+}K_{a} = i(2s-1)\epsilon_{abc}S_{c},$$
(4.2.14)

we find the following system to define matrices $a_{1-4}^{n\,m}$:

$$(a_1^{n\,m})^2 + (2s-1)a_2^{n\,m}a_3^{n\,m} = 0,$$

$$(s+1)a_1^{n\,m}a_2^{n\,m} - (s-1)a_2^{n\,m}a_4^{n\,m} = 0,$$

$$(s+1)a_3^{n\,m}a_1^{n\,m} - (s-1)a_4^{n\,m}a_3^{n\,m} = 0,$$

$$(a_4^{n\,m})^2 - (2s+1)a_3^{n\,m}a_2^{n\,m} = 0.$$

(4.2.15)

The condition concerning the operator C_1 (4.2.7) (that is, C_1 should have not more than two invariant eigensubspaces) means that matrices a_1^{nm} , a_3^{nm} should not be reducible.

All nonequivalent solution of Equations (4.2.15) are given in (4.2.13), the greater number from the set (n, m) coinciding with the index of nilpotentness N_1 of the operator C_1 (4.2.7). Below we present the explicit form of matrices S_a and K_a in the basis $|s, s_3 \rangle$, that is in the basis in which operators $S^2 \equiv S_a S_a$ and S_3 are diagonal [120, 78, 87*]:

$$S^{2}|s, s_{3} \rangle = s(s+1)|s, s_{3} \rangle, \qquad S_{3}|s, s_{3} \rangle = s_{3}|s, s_{3} \rangle,$$

$$S_{1}|s, s_{3} \rangle = a_{s_{3}}^{s}{}_{s_{3}+1}|s, s_{3} + 1 \rangle + a_{s_{3}}^{s}{}_{s_{3}-1}|s, s_{3} - 1 \rangle,$$

$$S_{2}|s, s_{3} \rangle = ia_{s_{3}}^{s}{}_{s_{3}+1}|s, s_{3} + 1 \rangle - ia_{s_{3}}^{s}{}_{s_{3}-1}|s, s_{3} - 1 \rangle, \qquad (4.2.16)$$

$$K_{1}|s, s_{3} \rangle = x_{s}^{s-1}|s - 1, s_{3} \rangle - ix_{s_{3}}^{s-1}{}_{s_{3}-2}|s - 1, s_{3} - 2 \rangle,$$

$$K_{2}|s, s_{3} \rangle = ix_{s_{3}}^{s-1}|s - 1, s_{3} \rangle - ix_{s_{3}}^{s-1}{}_{s_{3}-2}|s - 1, s_{3} - 2 \rangle,$$

$$K_{3}|s, s_{3} \rangle = f_{s_{3}}^{s-1}|s - 1, s_{3} \rangle$$

where:

$$s_{3} = -s, -s + 1, \dots, s; \qquad a_{s_{3} s_{3} \pm 1}^{s} = \frac{1}{2}\sqrt{s_{3}(s_{3} \pm 1) - s(s + 1)},$$

$$f_{s_{3}}^{s s-1} = \sqrt{s_{3}(2s - s_{3})}; \qquad \mathfrak{a}_{s_{3}}^{s s-1} = \frac{1}{2}\sqrt{(2s - s_{3})(2s + 1 - s_{3})},$$

$$\mathfrak{a}_{s_{3} s_{3} - 2}^{s s-1} = \frac{1}{2}\sqrt{s_{3}(s_{3} + 1)}, \qquad (4.2.17)$$

Explicit form of matrices Σ_a can be obtained from (4.2.16), (4.2.17) by replacement $s \to s' = s - 1$.

Theorems 4.2.1, 4.2.2 allow us to write down explicitely systems (4.2.5). Below we present some of such systems. It is convenient to enumerate these systems by indicating representation which is realized by matrices \hat{S}_a (4.2.9). Below, φ and χ denote, unless otherwise stated, 2s + 1 and (2s - 1)-component column functions.

1) Representation $D(s) \oplus D(s)$:

$$\widehat{S}_{a} = \begin{pmatrix} S_{a} & 0\\ 0 & S_{a} \end{pmatrix}, \qquad \lambda_{a} = \begin{pmatrix} 0 & 0\\ S_{a} & 0 \end{pmatrix}, \qquad \lambda_{0} = \frac{i}{2} \begin{pmatrix} 3 & 0\\ 0 & 5 \end{pmatrix}$$

Chapter 4. Systems of PDEs Invariant Under Galilei Groups

$$\dot{S}\psi = 0, \qquad \psi = \begin{pmatrix} \varphi_1 \\ \varphi_2 \end{pmatrix}$$

$$im\varphi_2 - (\vec{S} \cdot \vec{P})\varphi_1 = 0$$
(4.2.18)

In particular, for spin s = 1/2 we have $S_a = \frac{1}{2}\sigma_a$, σ_a are the Pauli matrices, and Equations (4.2.18) take the form

$$\overset{\,\,{}_\circ}{S}\psi=0,\qquad\psi=\left(egin{array}{c} arphi_1\ arphi_2\ \end{array}
ight); ext{ where } arphi_1,\,arphi_2 ext{ are two-component functions},$$

$$2im\varphi_2 - (\vec{\sigma}\cdot\vec{P})\varphi_1 = 0$$

or equivalently

$$P_0\varphi_1 - i(\vec{\sigma} \cdot \vec{P})\varphi_2 = 0,$$

$$2im\varphi_2 - (\vec{\sigma} \cdot \vec{P})\varphi_1 = 0$$
(4.2.19)

Equations (4.2.19) are known as Levi-Leblond ones for nonrelativistic particle with spin s = 1/2 [144].

2) Representation $D(s) \oplus D(s-1)$:

a)
$$\hat{S}_{a} = \begin{pmatrix} S_{a} & 0\\ 0 & \Sigma_{a} \end{pmatrix}, \qquad \lambda_{a} = \begin{pmatrix} 0 & 0\\ K_{a} & 0 \end{pmatrix}, \qquad \lambda_{0} = \frac{i}{2} \begin{pmatrix} 3 & 0\\ 0 & 5 \end{pmatrix}$$

 $\dot{S}\psi = 0, \qquad \psi = \begin{pmatrix} \varphi\\ \chi \end{pmatrix}$ (4.2.20)

$$im\chi - (\vec{K}\cdot\vec{P})\varphi = 0$$

In particular, for spin s = 1 we have $(S_a)_{bc} = i\epsilon_{abc}$:

$$S_{1} = \begin{pmatrix} 0 & 0 & 0 \\ 0 & 0 & -i \\ 0 & i & 0 \end{pmatrix}, \quad S_{2} = \begin{pmatrix} 0 & 0 & i \\ 0 & 0 & 0 \\ -i & 0 & 0 \end{pmatrix}, \quad S_{3} = \begin{pmatrix} 0 & -i & 0 \\ i & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix},$$
$$K_{1} = (i \ 0 \ 0), \quad K_{2} = (0 \ i \ 0), \quad K_{1} = (0 \ 0 \ i)$$
(4.2.21)

and Equations (4.2.20) take the form

$$\begin{aligned} \widehat{S}\psi &= 0, \\ m\chi - (\vec{P}\cdot\vec{\varphi}) &= 0 \end{aligned} \tag{4.2.22}$$

 $\psi = \text{column}(\vec{\varphi}, \chi), \chi$ is a scalar function. Equations (4.2.22) can be considered as the Galilean counterpart of the Proca equations [78].

b)
$$\widehat{S}_a = \begin{pmatrix} S_a & 0\\ 0 & \Sigma_a \end{pmatrix}$$
, $\lambda_a = \begin{pmatrix} 0 & K_a^+\\ 0 & 0^- \end{pmatrix}$, $\lambda_0 = \frac{i}{2} \begin{pmatrix} 5 & 0\\ 0 & 3 \end{pmatrix}$

 $\cdot 210$

$$\check{S}\psi = 0, \qquad \psi = \begin{pmatrix} \varphi \\ \chi \end{pmatrix}$$
(4.2.23)

$$im\varphi - (K^+ \cdot P)\chi = 0$$

3) Representation $D(s) \oplus D(s) \oplus D(s-1)$:

a)

$$\widehat{S}_{a} = \begin{pmatrix} S_{a} & 0 & 0\\ 0 & S_{a} & 0\\ 0 & 0 & \Sigma_{a} \end{pmatrix}, \quad \lambda_{a} = -\frac{i}{2s} \begin{pmatrix} 0 & 0 & 0\\ S_{a} & 0 & 0\\ K_{a} & 0 & 0 \end{pmatrix}, \quad \lambda_{0} = \frac{i}{2} \begin{pmatrix} 3 & 0 & 0\\ 0 & 5 & 0\\ 0 & 0 & 5 \end{pmatrix}, \\
\overset{\circ}{S}\psi = 0, \qquad \psi = \text{column}(\varphi_{1} \varphi_{2} \chi) \\
2ms\varphi_{2} + (\vec{S} \cdot \vec{P})\varphi_{1} = 0 \tag{4.2.24} \\
2ms\chi + (\vec{K} \cdot \vec{P})\varphi_{1} = 0$$

Making use (4.2.12) one can rewrite Equations (4.2.24) as follows

$$sP_0\varphi_1 + (\vec{S} \cdot \vec{P})\varphi_2 + (\vec{K}^+ \vec{P})\chi = 0$$

$$2ms\varphi_2 + (\vec{S} \cdot \vec{P})\varphi_1 = 0$$

$$2ms\chi + (\vec{K} \cdot \vec{P})\varphi_1 = 0$$
(4.2.25)

System (4.2.25) is known as Hagen-Herley equations [120] and describes a nonrelativistic particle with mass m and arbitrary spin s.

b)

$$\widehat{S}_{a} = \begin{pmatrix} S_{a} & 0 & 0\\ 0 & S_{a} & 0\\ 0 & 0 & \Sigma_{a} \end{pmatrix}, \quad \lambda_{a} = \frac{i}{2s} \begin{pmatrix} 0 & S_{a} & K_{a}^{+}\\ 0 & 0 & 0\\ 0 & 0 & 0 \end{pmatrix},$$

$$\lambda_{0} = \frac{i}{2} \begin{pmatrix} 5 & 0 & 0\\ 0 & 3 & 0\\ 0 & 0 & 3 \end{pmatrix},$$

 $\check{S}\psi = 0, \qquad \psi = \operatorname{column}(\varphi_1 \, \varphi_2 \, \chi)$

(4.2.26)

$$2ms\varphi_1 - (\vec{S}\cdot\vec{P})\varphi_2 + (\vec{K}_a^+\vec{P})\chi = 0.$$

4) Representation $D(s) \oplus D(s) \oplus D(s) \oplus D(s-1)$:

$$\mathbf{a}) \quad \widehat{S}_{a} = \begin{pmatrix} S_{a} & 0 & 0 & 0\\ 0 & S_{a} & 0 & 0\\ 0 & 0 & S_{a} & 0\\ 0 & 0 & 0 & \Sigma_{a} \end{pmatrix}, \quad \lambda_{0} = \frac{i}{2} \begin{pmatrix} 1 & 0 & 0 & 0\\ 0 & 3 & 0 & 0\\ 0 & 0 & 5 & 0\\ 0 & 0 & 0 & 3 \end{pmatrix}$$

Chapter 4. Systems of PDEs Invariant Under Galilei Groups

$$\lambda_a = \begin{pmatrix} 0 & 0 & 0 & 0 \\ S_a & 0 & 0 & 0 \\ 0 & S_a & -K_a^+/\sqrt{2s-1} & 0 \\ K_a/\sqrt{2s-1} & 0 & 0 & 0 \end{pmatrix}$$
(4.2.27)

In this case $\Lambda = \lambda_a \lambda_a \neq 0$. To find Λ one has to use the relations which follow from (4.2.12), (4.2.14) [94]:

$$K_a^+ K_a = s(2s-1), \qquad K_a K_a^+ = s(2s+1)$$

$$\Sigma_a \Sigma_b + K_a K_b^+ = -is\epsilon_{abc}\Sigma_c + s^2\delta_{ab}$$

$$\Sigma_a K_a = 0, \qquad K_a S_a = 0 \qquad (4.2.28)$$

$$K_a K_b^+ = -\frac{1}{2}(\Sigma_a \Sigma_b + \Sigma_b \Sigma_a) - \frac{i}{2}(2s+1)\epsilon_{abc}\Sigma_c + s^2\delta_{ab}$$

$$K_a^+ K_b = -\frac{1}{2}(S_a S_b + S_b S_a) + \frac{i}{2}(2s-1)\epsilon_{abc}\Sigma_c + s^2\delta_{ab}$$
Having used (4.2.28) we find A for A for

Having used (4.2.28) we find Λ for λ_a from (4.2.27):

$$\Lambda = \lambda_a \lambda_a = \begin{pmatrix} 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 \\ s^2 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 \end{pmatrix}$$
(4.2.29)

Finally, system (4.2.5) in this case takes the form

$$\begin{split} \overset{\circ}{S}\psi &= 0, \qquad \psi = \operatorname{column}(\varphi_1 \ \varphi_2 \ \varphi_3, \ \chi) \\ im\varphi_3 &- (\vec{S} \cdot \vec{P})\varphi_2 + \frac{1}{\sqrt{2s-1}} (\vec{K}^+ \cdot \vec{P})\chi = 0 \\ \varphi_1 &= 0 \end{split}$$
(4.2.30)

In much the same way one can continue constructing Sch(1,3)-invariant systems of PDEs of the type (4.2.5).

Now, let us consider the following problem: describe the most general form of functions F, G, R depending on ψ and ψ^+ so that the system of PDEs

$$\overset{\circ}{S}\psi \equiv \left(P_0 - \frac{1}{2m}\vec{P}^2\right)\psi = F$$

$$L\psi \equiv \left(\lambda_0 - \frac{3}{2}i - \frac{1}{m}\vec{\lambda}\cdot\vec{P}\right)\psi = G$$

$$\Lambda\psi \equiv \left(\lambda_1^2 + \lambda_2^2 + \lambda_3^2\right)\psi = R$$
(4.2.31)

would be invariant under the Schrödinger group Sch(1,3). To solve this problem we shall use the final transformations of Sch(1,3) group written in Table 4.1.2. We restrict ourselves by considering several particular cases.

Theorem 4.2.3. [185, 94]. Equations

$$\overset{\diamond}{S}\varphi_1 = F_1, \qquad \overset{\diamond}{S}\varphi_2 = F_2$$

$$im\varphi_2 - (\vec{S} \cdot \vec{P})\varphi_1 = G$$

$$(4.2.32)$$

where $\varphi_1 = \varphi_1(x), \varphi_2 = \varphi_2(x)$ are (2s + 1)-component functions; F_1, F_2, G are (2s + 1)-component functions depending on $\varphi_1, \varphi_1^+, \varphi_2, \varphi_2^+$ are invariant under the Schrödinger group Sch(1,3) iff

$$F_{1} = \mathscr{E}F\left(\frac{\omega_{1}}{\omega_{2}}\right)\omega_{2}^{2}\varphi_{1}, \qquad \mathscr{E} = \text{const}$$

$$F_{2} = \mathscr{E}F\left(\frac{\omega_{1}}{\omega_{2}}\right)\omega_{2}\varphi_{2}, \qquad (4.2.33)$$

$$G = 0,$$

where F is an arbitrary smooth function;

$$\omega_1 = (\varphi_1^+ \varphi_2 + \varphi_2^+ \varphi_1)^{1/4}, \qquad \omega_2 = (\varphi_1^+ \varphi_1)^{1/3}. \tag{4.2.34}$$

Proof. First of all we note that system (4.2.32) is a nonlinear generalization of Equations (4.2.18) and, as easy to see, it admits operators $\{P_0, P_a, J_a, M\}$ from ASch(1,3) (4.1.2) if functions F_1, F_2, G depend on $\varphi_1^+\varphi_1, \varphi_2^+\varphi_2, \varphi_1^+\varphi_2, \varphi_2^+\varphi_1$. Hence, to prove the theorem it is sufficient to consider the case with operator Π only.

Using formulae N8 – 10, 13 from Table 4.1.2 we write down Galilean and projective transformations for functions φ_1 and φ_2 :

$$\begin{aligned}
\varphi_1'(x') &= \exp\left\{im\left(\vec{v}\cdot\vec{x} + \frac{1}{2}\vec{v}^2x_0\right)\right\}\varphi_1(x), \\
\varphi_2'(x') &= \exp\left\{im\left(\vec{v}\cdot\vec{x} + \frac{1}{2}\vec{v}^2x_0\right)\right\}\left(\varphi_2(x) - i(\vec{S}\cdot\vec{v})\varphi_1(x)\right), \\
x_0' &= x_0 \qquad \vec{x}' = \vec{x} + \vec{v}x_0;
\end{aligned}$$
(4.2.35)

$$\begin{split} \varphi_1'(x') &= (1 - \theta x_0)^{3/2} \exp\left\{\frac{im\vec{x}^2}{2} \frac{\theta}{1 - \theta x_0}\right\} \varphi_1(x), \quad (4.2.36) \\ \varphi_1'(x') &= (1 - \theta x_0)^{5/2} \exp\left\{\frac{im\vec{x}^2}{2} \frac{\theta}{1 - \theta x_0}\right\} \left(\varphi_2(x) - i\frac{\theta}{1 - \theta x_0}(\vec{S} \cdot \vec{x})\varphi_1(x)\right), \\ x_0' &= \frac{x_0}{1 - \theta x_0}, \quad \vec{x}' = \frac{\vec{x}}{1 - \theta x_0}. \end{split}$$

Under the influence of projective transformations we find that

Chapter 4. Systems of PDEs Invariant Under Galilei Groups

$$P_0 \rightarrow P'_0 \equiv i \frac{\partial}{\partial x'_0} = (1 - \theta x_0)^2 P_0 + \theta (1 - \theta x_0) \vec{x} \cdot \vec{P}$$

$$(4.2.37)$$

$$P \rightarrow P' = -i\nabla' = (1 - \theta x_0)P$$

and system (4.2.32) takes the form

$$(1 - \theta x_0)^{7/2} e^{if} \overset{\checkmark}{S} \varphi_1 = F_1', \qquad f \stackrel{\text{def}}{=} \frac{m\vec{x}^2}{2} \frac{\theta}{1 - \theta x_0};$$

$$(1 - \theta x_0)^{9/2} e^{if} \left[\overset{\lor}{S} \varphi_2 - \frac{\theta}{1 - \theta x_0} \frac{1}{m} \left(im\varphi_2 - (\vec{S} \cdot \vec{P})\varphi_1 + im(\vec{S} \cdot \vec{x}) \overset{\lor}{S} \varphi_1 \right) \right] = F_2',$$

$$(1 - \theta x_0)^{5/2} e^{if} \left[im\varphi_2 - (\vec{S} \cdot \vec{P})\varphi_1 \right] = G'$$

whence it follows that

$$F_{1}' = (1 - \theta x_{0})^{7/2} e^{if} F_{1}, \qquad f \equiv \frac{m\vec{x}^{2}}{2} \frac{\theta}{1 - \theta x_{0}};$$

$$F_{2}' = (1 - \theta x_{0})^{9/2} e^{if} \left[F_{2} - i \frac{\theta}{1 - \theta x_{0}} (\vec{S} \cdot \vec{x}) F_{1} - \frac{\theta}{m(1 - \theta x_{0})} G \right] \qquad (4.2.38)$$

$$G' = (1 - \theta x_{0})^{5/2} e^{if} G$$

Comparing (4.2.38) with (4.2.36) one can conclude that functions F_1 , F_2 , G should have the structure

$$F_1 = \phi \varphi_1, \qquad F_2 = \phi \varphi_2, \qquad G = 0$$
 (4.2.39)

where ϕ is a scalar function which is transformed as follows

$$\phi' = (1 - \theta x_0)^2 \phi \tag{4.2.40}$$

From (4.2.36) it is easy to see that function ϕ can be constructed only from two arguments $\varphi_1^+\varphi_1$ and $\varphi_1^+\varphi_2 + \varphi_2^+\varphi_1$. Since

$$(\varphi_1^+\varphi_1)' = (1-\theta x_0)^3(\varphi_1^+\varphi_1), \qquad (\varphi_1^+\varphi_2+\varphi_2^+\varphi_1)' = (1-\theta x_0)^4(\varphi_1^+\varphi_2+\varphi_2^+\varphi_1)$$

function ϕ should have the form

$$\phi = xF\left(\frac{\omega_1}{\omega_2}\right)\omega_2^2$$

where \mathfrak{X} is a constant; ω_1 , ω_2 are given in (4.2.34), F is an arbitrary smooth function. The theorem is proved.

Below we present some more Sch(1,3)-invariant nonlinear systems of PDEs constructed in the same spirit as (4.2.32) [94]:

$$P_{0}\varphi_{1} - i(\vec{\sigma} \cdot \vec{P})\varphi_{2} = \mathscr{B}F\omega_{2}\varphi_{2} + \mu G\omega_{2}^{2}\varphi_{1}$$

$$2im\varphi_{2} - (\vec{\sigma} \cdot \vec{P})\varphi_{1} = i\mathscr{B}F\omega_{2}\varphi_{1}$$

$$(4.2.41)$$

where \mathfrak{X} , μ are constants; φ_1 and φ_2 are two-component functions, $\vec{\sigma}$ are Pauli matrices, F and G are arbitrary smooth functions of ω_1/ω_2 ; ω_1 and ω_2 are given in (4.2.34). When $\mu = \mathfrak{X} = 0$ Equations (4.2.41) coincide with those of Levi-Leblond (4.2.19).

System

$$\overset{\vee}{S}\psi = \mathfrak{a}(\varphi^{+}\varphi)^{2/3}\psi, \qquad \psi = \begin{pmatrix} \varphi \\ \chi \end{pmatrix}$$

$$im\chi - (\vec{K} \cdot \vec{P})\varphi = 0$$

$$(4.2.42)$$

coincides with (4.2.20) when x = 0.

System

$$\overset{\vee}{S}\psi = xF\left(\frac{(\varphi_{1}^{+}\varphi_{2} - \varphi_{2}^{+}\varphi_{1})^{1/2}}{(\varphi_{1}^{+}\varphi_{1})^{2/3}}\right)(\varphi_{1}^{+}\varphi_{1})^{2/3}\psi, \quad \psi = \begin{pmatrix}\varphi_{1}\\\varphi_{2}\\\chi\end{pmatrix}$$

$$2ms\varphi_{2} + (\vec{S} \cdot \vec{P})\varphi_{1} = 0$$

$$2ms\xi + (\vec{K} \cdot \vec{P})\varphi_{1} = 0$$

$$(4.2.43)$$

coincides with (4.2.24) when $\mathfrak{w} = 0$.

System

$$sP_{0}\varphi_{1} + (\vec{S} \cdot \vec{P})\varphi_{2} + (\vec{K}^{+}\vec{P})\chi = \&F\left(\frac{(\varphi_{1}^{+}\varphi_{2} - \varphi_{2}^{+}\varphi_{1})^{1/2}}{(\varphi_{1}^{+}\varphi_{1})^{2/3}}\right)(\varphi_{1}^{+}\varphi_{1})^{2/3}\psi,$$

$$2ms\varphi_{2} + (\vec{S} \cdot \vec{P})\varphi_{1} = 0,$$

$$(4.2.44)$$

$$2ms\chi + (\vec{K} \cdot \vec{P})\varphi_{1} = 0$$

,

coincides with the equations of Hagen-Herley (4.2.25) when x = 0.

4.3. Systems of second-order PDEs invariant under the Galilei group

Following [78] we shall obtain two classes of Galilean invariant systems of second-order PDEs. The essential difference between these equations and those

studied in the previous paragraph consists in the following. Equations to be considered give more complete description of interection of a particle having spin with an external electromagnetic field. Equations of §4.2 describe spin-orbit coupling but fail to take into account quadrupole and Darwin interections. It is generally accepted to think that such interections are truly relativistic effects, and, for instance, if the particle spin s = 1/2, only the Dirac relativistic equation describes them adequately. Equations considered below refute this widespread opinion.

As was stated above (see §4.1) the Schrödinger equation (4.1.1) is invariant under the Galilei group and describes a free spinless particle with mass m. Naturally the question arises: are there equations of the form

$$P_0\psi \equiv i\frac{\partial}{\partial x_0}\psi = H_s(\vec{P})\psi, \qquad (4.3.1)$$

where $H_s(\vec{P})$ is a differential operator, $\psi = \psi(x_0, \vec{x})$ is a complex multicomponent function, on the manifold of solutions of which a representation of the Galilei group G(1,3) with nonzero spin would be realized?

We seek for Galilean-invariant equations (4.3.1) in the space of 2(2s + 1)component quadratic integrable function

$$\psi = \left\{ \psi_1(x_0, \vec{x}), \dots, \psi_{2(2s+1)}(x_0, \vec{x}) \right\}$$
(4.3.2)

The problem of description of such equations we solve in two, generally speaking, nonequivalent approaches. In the first one the problem is to find (within equivalence) all operators H_s^I satisfying the conditions

$$[P_0 - H_s, Q_A]\psi = 0, (4.3.3)$$

where $Q_A = \{P_0, P_a, J_a, G_a, M\}$ are given in (4.1.2), provided

$$\widehat{S}_a = \begin{pmatrix} S_a & 0\\ 0 & S_a \end{pmatrix}, \qquad \lambda_a = k(\widehat{\sigma}_1 + i\widehat{\sigma}_2)S_a, \qquad (4.3.4)$$

 S_a are matrices of irreducible representation D(s) of AO(3); $\hat{\sigma}_a$ are 2(2s+1)-dimensional Pauli matrices

$$\widehat{\sigma}_0 = \begin{pmatrix} I & 0 \\ 0 & I \end{pmatrix} \quad \widehat{\sigma}_1 = \begin{pmatrix} 0 & I \\ I & 0 \end{pmatrix} \quad \widehat{\sigma}_2 = i \begin{pmatrix} 0 & -I \\ I & 0 \end{pmatrix}, \quad \widehat{\sigma}_3 = \begin{pmatrix} I & 0 \\ 0 & -I \end{pmatrix} (4.3.5)$$

I and 0 are unit and zero matrices of dimension $(2s + 1) \times (2s + 1)$; k is an arbitrary complex parameter.

Having substituted operators from AG(1,3) (4.1.2) into (4.3.3) we find that it is fulfilled if

$$[H_s^I, P_a] = [H_s^I, J_a] = 0 (4.3.6)$$

$$[H_s^I, G_a] = iP_a \tag{4.3.7}$$

The second approach to the problem is to find all operators $H^{I\!I}_a$ so that the operators

$$P_0^{II} = H_s^{II}, \qquad P_a^{II} = P_a = -i\partial_a, \qquad M^{II} = \hat{\sigma}_1 m$$

$$J_a^{II} = J_a = (\vec{x} \times \vec{P})_a + \hat{S}_a, \qquad (4.3.8)$$

$$G_a^{II} = x_0 P_a - M x_a + \lambda_a^{II}$$

will be the generators of AG(1,3). In (4.3.8) λ_a^{II} are operators to be found; \hat{S}_a are given in (4.3.4).

We require the operators (4.3.8) to be Hermitian with respect to the usual scalar product

$$(\psi_1, \psi_2) = \int \psi_a^+(x_0, \vec{x}) \psi_2(x_0, \vec{x}) d^3x.$$
(4.3.9)

As distinguished from (4.1.2) operators H_a^I , G_a^I are non-Hermitian with respect to (4.3.9) but Hermitian with respect to

$$(\psi_1,\psi_2) = \int \psi_a^+(x_0,\vec{x}) M \psi_2(x_0,\vec{x}) d^3x.$$
 (4.3.10)

where M is a positively defined operator which will be found below. We require the Hamiltonian H_a^{II} to satisfy the conditon

$$\left(H_{s}^{I\!I}\right)^{2} = \left(m + \frac{\vec{P}^{2}}{2m}\right)^{2}$$
 (4.3.11)

It means that the intrinsic energy of particle (the eigenvalues of the invariant operator $C_1 = 2mP_0 - P_aP_a$) coincides with its mass.

Theorem 4.3.1. [78]. The most general form of the Hamiltonian H_s^I satisfying together with the generators of AG(1,3) (4.1.2), (4.3.4) commutation relations (4.3.6), (4.3.7) is as follows

$$H_s^I = \hat{\sigma}_1 m a + 2ik\hat{\sigma}_3 \hat{S}_a P_a + \frac{1}{2m} C_{ab} P_a P_b$$
(4.3.12)

$$\widetilde{H}_{s}^{I} = \frac{a}{2}(\widehat{\sigma}_{1} - i\widehat{\sigma}_{2})m + \widehat{\sigma}_{3}\widetilde{a}m + 2\widetilde{a}k(\widehat{\sigma}_{2} - i\widehat{\sigma}_{1})\widehat{S}_{a}P_{a} + \frac{1}{2m}C_{ab}P_{a}P_{b}$$
(4.3.13)

where

$$C_{ab} = \delta_{ab} - 2ak^2(\widehat{\sigma}_1 + i\widehat{\sigma}_2)(S_a S_b + S_b S_a)$$

$$(4.3.14)$$

 a, \tilde{a}, k are arbitrary parameters.

Proof. It is convenient to seek for H_s^I in a representation where $\lambda_a = 0$. It is achieved by the transformation

$$H_s^I \to (H_s^I)' = V H_s^I V^{-1}, \qquad P_a \to P_a' = V P_a V^{-1} = P_a,$$
 (4.3.15)

$$J_a^I \to (J_a^I)' = V J_a^I V^{-1} = J_a^I = J_a, \qquad (G_a^I)' = V G_a V^{-1} = x_0 P_a - m x_a,$$

where

$$V = \exp\left\{\frac{i}{m}\vec{\lambda}\cdot\vec{P}\right\} = 1 + \frac{i}{m}\vec{\lambda}\cdot\vec{P}$$
(4.3.16)

From (4.3.6), (4.3.7), (4.3.15) we easy find the general form of operator $(H_s^I)'$

$$(H_s^I)' = \frac{\vec{P}^2}{2m} + A, \qquad A = \hat{\sigma}_{\mu} a^{\mu} m, \qquad (4.3.17)$$

where a^{μ} are arbitrary complex coefficients.

So, Equation (4.3.1) has in the representation (4.3.15) the form

$$P_0\psi' = \left(\frac{\vec{P}^2}{2m} + m\hat{\sigma}_{\mu}a^{\mu}\right)\psi', \qquad \psi' = V\psi \qquad (4.3.18)$$

Now we show that matrix A from (4.3.17) can be reduced to

$$A = \hat{\sigma}_3 \tilde{a} m + \frac{a}{2} \left(\hat{\sigma}_1 - i \hat{\sigma}_2 \right) m, \qquad (4.3.19)$$

or

$$A = \hat{\sigma}_1 am, \tag{4.3.20}$$

where a, \tilde{a} are arbitrary coefficients. Indeed, one can always turn coefficient a_0 into zero:

$$(H_s^I)' \to \exp\{ia_0 m x_0\} (H_s^I)' \exp\{-ia_0 m x_0\} + + \exp\{ia_0 m x_0\} P_0 \exp\{-ia_0 m x_0\} = (H_s^I)' - a_0 m$$
(4.3.21)

Further, there are three possibilities

$$A \equiv 0, \qquad a_b = 0; \tag{4.3.22}$$

$$A^{2} = a_{1}^{2} + a_{2}^{2} + a_{3}^{2} = 0, \qquad a_{b} \neq 0;$$
(4.3.23)

$$A^{2} = a_{1}^{2} + a_{2}^{2} + a_{3}^{2} = a^{2} \neq 0;$$
(4.3.24)

The first case gives (4.3.19) provided $a = \tilde{a} = 0$. The second case (4.3.23) corresponds to non-unitary representation of the Galilei group because the invariant operator $C_1 = 2mP_0 - P_aP_a = 2m^2A$ is a nilpotent matrix, but

such cases we do not consider. Let in (4.3.24) $a_1^2 + a_2^2 \neq 0$. Having made transformation

$$A \to V_1 A V_1^{-1},$$

$$V_1 = b + i \hat{\sigma}_3 c + (\hat{\sigma}_1 + i \hat{\sigma}_2) d; \qquad b = \cos \varphi, \quad c = \sin \varphi \qquad (4.3.25)$$

$$V_1^{-1} = b - i \hat{\sigma}_3 - (\hat{\sigma}_1 + i \hat{\sigma}_2) d.$$

$$\varphi = \frac{1}{2} \arctan\left(\frac{a_1 + 2d^2}{a_2 - 2id}\right), \qquad d = \sqrt{\frac{d_3^2 (a_1^2 - ia^2)}{4a(a_1^2 + a_2^2)}} \qquad (4.3.26)$$

we obtain (4.3.20). If $a_1^2 + a_2^2 = 0$, then matrix A is transformed by virtue of the operator

$$V_{2} = 1 + (\hat{\sigma}_{1} + i\hat{\sigma}_{2}) \cdot f/2, \qquad V_{2}^{-1} = 1 - (\hat{\sigma}_{1} + i\hat{\sigma}_{2}) \cdot f/2$$

$$f = \begin{cases} a_{1}/a_{3}, & a_{2} = ia_{1} \\ 0, & a_{2} = -ia_{1} \end{cases}$$
(4.3.27)

to (4.3.19). Operators (4.3.25), (4.3.27) satisfy conditions

$$V_{\alpha}\lambda_{a}V_{\alpha}^{-1} = \mathfrak{A}_{\alpha}\lambda_{a}, \qquad \alpha = 1,2 \tag{4.3.28}$$

where matrices λ_a are given in (4.3.4); $\mathfrak{X}_1 = \exp\{2i\varphi\}$, $\mathfrak{X}_2 = 1$, parameter φ is determined in (4.3.26). One can make sure that there is no operator satisfying (4.3.28) and transforming (4.3.19) to (4.3.20).

Acting on (4.3.17), (4.3.19), (4.3.20) by transformation inverse to (4.3.15) we get Hamiltonians (4.3.12), (4.3.13) which evidently satisfy conditions (4.3.6), (4.3.7). The theorem is proved.

One can easy make sure that Equations (4.3.1), (4.3.12), (4.3.13) are Galilean invariant using transformations (see Table 4.1.2, N8-10)

$$x'_{0} = x_{0}, \qquad \vec{x}' = \vec{x} + \vec{v}x_{0}$$

$$\psi'(x') = \exp\left\{im\left(\vec{v}\cdot\vec{x} + \frac{\vec{v}^{2}}{2}x_{0}\right)\right\}(1 - \vec{\lambda}\cdot\vec{v})\psi(x)$$
(4.3.29)

generated by G_a (4.1.2), (4.3.4). Since

$$P'_0 = P_0 + \vec{v} \cdot \vec{P}, \qquad \vec{P}' = \vec{P}$$
(4.3.30)

it follows that equation

$$P_0'\psi'(x') - H_s^I(\vec{P}')\psi'(x') = \left(P_0 + \vec{v} \cdot \vec{P} - H_s^I(\vec{P})\right)\psi'(x') = 0$$
(4.3.31)

are identically satisfied on sets of solutions of Equations (4.3.1), (4.3.12), (4.3.13).

It is not difficult to calculate (the most simple it may be done in the representation (4.3.15)) that Casimir operators of AG(1,3) (4.1.2), (4.3.4) have the following eigenvalues

$$C_{1} = 2MP_{0} - P_{a}P_{a} = \pm m; \qquad C_{2} = M = m$$

$$C_{3} = W_{a}W_{a} = (MJ_{a} - \epsilon_{abc}P_{b}J_{c})^{2} = m^{2}s(s+1). \qquad (4.3.32)$$

It allows us to interpret Equations (4.3.1), (4.3.12), (4.3.13) as those which describe a free nonrelativistic particle of mass m and spin s and intrinsic energy $\pm m$.

Now we shall find Hamiltonians H_s^{II} which are Hermitian in the metrics (4.3.9) and satisfy together with generators (4.3.8) relations (4.1.3), (4.3.11).

Theorem 4.3.2. [78]. The most general form of the Hamiltonian H_s^{II} (within equivalence) Hermitian in the metrics (4.3.9) and satisfying relations (4.3.8), (4.1.3), (4.3.11) is as follows

$$H_{s}^{II} = \hat{\sigma}_{1} \left(m + \frac{\vec{P}^{2}}{2m} - \frac{\tilde{\vec{S}} \cdot \vec{P}}{s^{2}m} \sin^{2} \theta_{s} \right) + \hat{\sigma}_{2} \frac{\sqrt{2} \sin \theta_{s}}{s} \left(\vec{\vec{S}} \cdot \vec{P} \right) - \\ - \hat{\sigma}_{3} \left(a_{s} \frac{\vec{P}^{2}}{2m} + b_{s} \left(\vec{\vec{S}} \cdot \vec{P} \right)^{2} / 2ms^{2} \right),$$

$$(4.3.33)$$

where

$$a_{1/2} = \sin 2\theta_{1/2}, \qquad b_{1/2} = 0; \qquad a_1 = 1, \qquad b_1 = \sin 2\theta_1; \qquad (4.3.34)$$
$$a_{3/2} = b_{3/2} - \frac{5}{4} \sin \theta_{3/2} = -\frac{1}{8} \sin 2\theta_{3/2} - \frac{3}{4} \sin \theta_{3/2} \left(1 - \frac{1}{9} \sin \theta_{3/2}\right)^{1/2};$$

 $a_s = b_s = \theta_s = 0, s > 3/2$ and $\theta_{1/2}, \theta_1, \theta_{3/2}$ are arbitrary parameters.

Proof. First of all, we prove that H_s^{II} does not include differential operators of more than second order. To do it we assume that $H_s^{II} = \sum_{i=0}^{N} H_i$, where H_i contains derivatives of *i*th order only. Then (4.3.11) results in

$$H_N H_N = H_N^+ H_N = 0, \quad H_N = 0, \text{ or } N > 2$$
 (4.3.35)

It is convenient to represent operator H_s^{II} as an expansion on spin matrices \hat{S}_a (4.3.4) and 2(2s + 1)-dimensional Pauli matrices (4.3.5)

$$H_s^{II} = \left(a_\mu m + b_\mu \frac{\vec{P}^2}{2m} + c_\mu \left(\vec{\hat{S}} \cdot \vec{P}\right) + d_\mu \frac{\left(\vec{\hat{S}} \cdot \vec{P}\right)^2}{2m}\right) \hat{\sigma}^\mu \tag{4.3.36}$$

where a_{μ} , b_{μ} , c_{μ} , d_{μ} are arbitrary real coefficients.

If to introduce the operators of orthogonal projection

$$\Lambda_r = \prod_{r \neq r'} \frac{(\vec{\hat{S}} \cdot \vec{P})/P - r'}{r - r'}; \qquad r, r' = -s, -s + 1, \dots, s$$

which satisfy conditions of orthonormality and completeness

$$\Lambda_{\tau}\Lambda_{r'} = \delta_{\tau\tau'}\Lambda_{r}, \qquad \sum_{r}\Lambda_{r} = 1, \qquad \sum_{r}r^{k}\Lambda_{r} = \left(\frac{\vec{\hat{S}}\cdot\vec{P}}{P}\right)^{k}$$

then (4.3.36) can be rewritten as follows

$$H_{s}^{II} = \sum_{r=-s'}^{r=+s'} \left(a_{\mu}m + (b_{\mu} + r^{2}d_{\mu})\frac{\vec{P}^{2}}{2m} + rpc_{\mu} \right) \hat{\sigma}^{\mu}\Lambda_{r}$$
(4.3.37)

It is obvious that operator H_s^{II} (4.3.37) satisfies commutation relations (4.3.6) and we require (4.3.37) to satisfy (4.3.11). Having substituted (4.3.37) into (4.3.11) and using properties of Λ_r we find after equating independent addends that coefficients a_{μ} , b_{μ} , c_{μ} , d_{μ} should satisfy one of the following systems of algebraic equations

$$\sum_{i=1}^{3} a_i^2 = 1; \qquad \sum_{i=1}^{3} \left[r^2 c_i^2 + a_i (b_i + r^2 d_i) \right] = 1;$$

$$\sum_{i=1}^{3} c_i r(b_i + r^2 d_i) = 0; \qquad \sum_{i=1}^{3} r c_i a_i = 0; \qquad \sum_{i=1}^{3} (b_i + r^2 d_i)^2 = 1;$$
(4.3.38)

or

$$a_0 = b_0 = 1;$$
 $d_0 = c_0 = a_i = b_i = c_i = d_i = 0;$ $i = 1, 2, 3$ (4.3.39)

The general solution of Equations (4.3.38) has the form (within linear transformations of equivalence):

$$a_{1} = 1, \qquad a_{0} = a_{2} = a_{3} = 0;$$

$$b_{1} = 1; \qquad b_{3} = a_{s}; \qquad b_{0} = b_{2} = 0;$$

$$c_{2} = \frac{\sqrt{2}}{s^{2}} \sin \theta_{s}, \qquad c_{0} = c_{1} = c_{3} = 0$$

$$d_{1} = -c_{2}^{2}, \qquad d_{3} = b_{s}/s^{2}, \qquad d_{0} = d_{2} = 0$$

(4.3.40)

where a_s , b_s , θ_s are given in (4.3.34). One can make sure that Equations (4.3.39) are inconsistent with (4.1.2); (4.3.40), (4.3.36) result in Hamiltonian (4.3.33).

To conclude the proof it is sufficient to show the explicit form of operators λ_a^{II} which ensure fulfilment of relations of AG(1,3) (4.1.3) by operators (4.3.8). The simplest way of obtaining λ_a^{II} is as follows:

$$\lambda_a^{I\!I} = [U, \,\widehat{\sigma}_1 x_a m] \, U^{\dagger}, \qquad (4.3.41)$$

where operator

$$U = \frac{E + H_s^{II} \hat{\sigma}_1}{\sqrt{2E \left[2E - \left(\frac{Pr}{m_s} \sin \theta_s\right)^2\right]}} \Lambda_r; \qquad E = m + \frac{\vec{P}^2}{2m}$$
(4.3.42)

diagonalizes Hamiltonian (4.3.33) and generators (4.3.8):

$$U^{\dagger}H_{s}^{II}U = \widehat{\sigma}_{1}E, \qquad U^{\dagger}G_{a}^{II}U = x_{0}P_{a} - \widehat{\sigma}_{1}mx_{a},$$

$$U^{\dagger}J_{a}^{II}U = J_{a}^{II}, \qquad U^{\dagger}P_{a}^{II}U = P_{a}^{II}$$

$$(4.3.43)$$

The theorem is proved.

Now we shall find out whether the Equations (4.3.1) are invariant under the scale and projective transformations. For the equations of the first type (this is (4.3.12), (4.3.13)) it can be done quite simple in the representation (4.3.15)-(4.3.17). So, we calculate

$$D' = VDV^{-1} = \left(1 + \frac{i}{m}\vec{\lambda}\cdot\vec{P}\right)\left(2x_0P_0 - \vec{x}\cdot\vec{P} + \lambda_0\right)\left(1 - \frac{i}{m}\vec{\lambda}\cdot\vec{P}\right) =$$
$$= 2x_0P_0 - \vec{x}\cdot\vec{P} + \lambda_0 + \frac{1}{m}\vec{\lambda}\cdot\vec{P} = D + \frac{1}{m}\vec{\lambda}\cdot\vec{P}$$
(4.3.44)

Note that matrix λ_0 satisfying commutation relations (4.1.4) with matrices λ_a (4.3.4) has the form

$$\lambda_0 = (i/2)\widehat{\sigma}_3 \tag{4.3.45}$$

We find

$$[P_{0} - (H_{s}^{I})', D'] = \left[P_{0} - \frac{\vec{P}^{2}}{2m} - \hat{\sigma}_{\mu}a^{\mu}m, 2x_{0}P_{0} - \vec{x}\vec{P} + \lambda_{0} + \frac{1}{m}\vec{\lambda}\cdot\vec{P}\right] = 2i\left(P_{0} - \frac{\vec{P}^{2}}{2m}\right) - \left[\hat{\sigma}_{\mu}, \lambda_{0} + \frac{1}{m}\vec{\lambda}\cdot\vec{P}\right]a^{\mu}m \neq 2i\left(P_{0} - (H_{s}^{I})'\right)$$

Hence Equations (4.3.1): (4.3.12), (4.3.13) do not admit a Lie generator of scale transformations D (4.1.2). It follows in turn that these equations are non-invariant with respect to the local projective transformations generated by II (see (4.1.2)). One can make sure in the same way that Equation (4.3.1): (4.3.33) is also non-invariant under the local scale and projective transformations. Nevertheless, there is a possibility to extend invariance algebra of these equations from AG(1,3) to ASch(1,3) with the help of the operators

$$D = \frac{1}{2m}(G_a P_a + P_a G_a), \qquad \Pi = \frac{1}{2m}G_a G_a$$
(4.3.46)

which, however, are differential operators of the second and third order and generate nonlocal transformations which are to be calculated according to (5.3.6).

It will be noted that in the case when s = 1/2, $\theta_{1/2} = \pi/4$, k = -i, a = 1Equations (4.3.1): (4.3.12) and (4.3.33) have the form

$$\left(\gamma_{\mu}P^{\mu} - m - i\gamma_5\frac{\vec{P}^2}{2m}\right)\psi = 0 \tag{4.3.47}$$

and

$$\left[\gamma_{\mu}P^{\mu} - m + (1 + \gamma_5 - \gamma_0)\frac{\vec{P}^2}{2m}\right]\psi = 0$$
 (4.3.48)

where γ_{μ} are Dirac matrices (2.1.2), $\gamma_5 = i\gamma_0\gamma_1\gamma_2\gamma_3$. Equations (4.3.47), (4.3.48) differ from the standard Dirac equation in terms with \vec{P}^2 but it is these addends which change the Poincare invariance of the Dirac equation for Galilean one (local for (4.3.48) and nonlocal for (4.3.47)).

Let us write down Equation (4.3.1): (4.3.12) for spinning particle when s = 1/2, 1 at large: $s = 1/2, \psi = \text{column}(\varphi, \chi)$; φ, χ are two- component functions

$$\begin{pmatrix} P_0 - \frac{\vec{P}^2}{2m} \end{pmatrix} \varphi = am\chi + iak(\vec{\sigma} \cdot \vec{P})\varphi - \frac{a^2k^2}{2m}\vec{P}^2\chi; \begin{pmatrix} P_0 - \frac{\vec{P}^2}{2m} \end{pmatrix} \chi = am\varphi - iak(\vec{\sigma} \cdot \vec{P})\chi.$$

$$(4.3.49)$$

 $s = 1, \psi = \text{column}(\vec{\varphi} \vec{\chi}); \vec{\varphi}, \vec{\chi} \text{ are three-component functions}$

$$\begin{pmatrix}
P_0 - \frac{\vec{P}^2}{2m}
\end{pmatrix} \vec{\varphi} = am\vec{\chi} - 2ak\vec{P} \times \vec{\varphi} - \frac{ak^2}{m} \left(\vec{P}^2\vec{\chi} - \vec{P}(\vec{P} \cdot \vec{\chi})\right);$$

$$\begin{pmatrix}
P_0 - \frac{\vec{P}^2}{2m}
\end{pmatrix} \vec{\chi} = am\vec{\varphi} + 2ak\vec{P} \times \vec{\chi}$$
(4.3.50)

matrices S_a are given in (4.2.21).

Equation (4.3.48) can be rewritten as follows

$$P_0\psi = H\psi,$$

$$H = \hat{\sigma}_3 m + \hat{\sigma}_1 (\vec{\sigma} \cdot \vec{P}) + \left[1 - (\hat{\sigma}_3 + i\hat{\sigma}_2)\right] \frac{\vec{P}^2}{2m}$$
(4.3.51)

It is rewarding to study its maximal Lie symmetry.

Theorem 4.3.3. [184]. The maximal group of point transformations admitted by Equation (4.3.51) is the Galilei group G(1,3), basis elements of corresponding AG(1,3) having the form

$$P_{0} = i \frac{\partial}{\partial x_{0}}, \qquad P_{a} = -\frac{\partial}{\partial x_{a}}, \qquad M = m$$

$$J_{a} = (\vec{x} \times \vec{P})_{a} + \hat{S}_{a}, \qquad (4.3.52)$$

$$G_{a} = x_{0}P_{a} - mx_{a} - \frac{1}{2}(\hat{\sigma}_{3} + i\hat{\sigma}_{2})\sigma_{a}$$

where

$$\hat{S}_a = \frac{1}{2} \begin{pmatrix} \sigma_a & 0\\ 0 & \sigma_a \end{pmatrix}$$
(4.3.53)

Proof. Use of the standard Lie alogrithm requires some cumbersome calculations. To avoid this we shall use the criterion of invariance in the form (10)(see also §5.3) so that

$$[P_0 - H, Q]\psi = 0 \tag{4.3.54}$$

where Q is the first-order differential operator

$$Q = \xi^{\mu}(x)\partial_{\mu} + \eta(x) \tag{4.3.55}$$

 ξ^{μ} are scalar functions; $\eta(x)$ is a matrix of dimension 4×4 . So we find

$$R \equiv [P_{0} - H, Q] = i(\xi_{0}^{a}P_{a} + \eta_{0}) - m[\hat{\sigma}_{3}, \eta] - - [\hat{\sigma}_{1}\sigma_{a}, \eta]P_{a} + i\hat{\sigma}_{1}\sigma_{b}(\xi_{b}^{a}P_{a} + \eta_{b}) + \frac{i}{m}(\xi_{a}^{b}P_{a}P_{b} + \eta_{a}P_{a}) + + \frac{1}{2m}((\Delta\xi^{a})P_{a} + \Delta\eta)\Gamma - [\Gamma, \eta]\frac{\vec{P}^{2}}{2m} + + i\left(\xi_{0}^{0} + \hat{\sigma}_{1}\sigma_{a}\xi_{a}^{0} + \frac{1}{m}\xi_{a}^{0}P_{a}\Gamma - \frac{i}{2m}(\Delta\xi^{0})\Gamma\right)P_{0},$$
(4.3.56)

where $\xi_{\nu}^{\mu} = \partial \xi^{\mu} / \partial x_{\nu}$, $\eta_{\mu} = \partial \eta / \partial x_{\nu}$; $\mu, \nu = \overline{1,3}$; $\Gamma = 1 - (\hat{\sigma}_3 + i\hat{\sigma}_2)$. The transition on the manifold of solutions of Equation (4.3.51) in (4.3.54) can be done as follows

$$[P_0 - H, Q] \Big|_{P_0 = H} \equiv R \Big|_{P_0 = H} = 0$$
(4.3.57)

that is in the expression (4.3.56) P_0 should be replaced by H (4.3.51) everywhere. Equating the coefficients of the identity (4.3.57) yields a set of equations to determine $\xi^{\mu}(x)$ and $\eta(x)$:

$$\begin{split} \xi_0^a + \xi_0^0 \widehat{\sigma}_1 \sigma_a + i [\widehat{\sigma}_1 \sigma_a, \eta] + \widehat{\sigma}_1 \sigma_a \xi_b^a + \frac{1}{m} \eta_a \Gamma - \frac{i}{2m} (\Delta \xi^a) \Gamma = 0; \\ \widehat{\sigma}_3 m \xi_0^0 + \eta_0 + i m [\widehat{\sigma}_3, \eta] + \widehat{\sigma}_1 \sigma_a \eta_a - \frac{i}{2m} (\Delta \eta) \Gamma = 0; \\ (\xi_0^0 + 2\xi_a^a) \Gamma + i [\Gamma, \eta] = 0, \text{ (no sum over } a) \\ \xi_a^b + \xi_b^a = 0, \qquad a \neq b. \end{split}$$

The general solution of this system has the form

$$\xi^{0} = c^{0}, \qquad \xi^{a} = c^{ab}x_{b} + g^{a}x_{0} + c^{a}, \eta(x) = -\frac{1}{4}\epsilon_{abc}\sigma_{a}c_{bc} - mg_{a}x_{a} - \frac{i}{2}(\hat{\sigma}_{3} + i\hat{\sigma}_{2})\sigma_{a}g_{a} + c_{4}$$
(4.3.58)

where c^0 , c^a , c_4 , $c_{ab} = -c_{bc}$, g_a are arbitrary constants. So, the theorem is proved.

4.4. Solutions of the nonlinear Levi-Leblond equation

Here we shall obtain some exact solutions of Sch(1,3)-invariant coupled nonlinear equations (see §4.2, (4.2.41))

$$P_{0}\varphi - i(\vec{\sigma} \cdot \vec{P})\chi = \mathscr{E}F\left(\frac{\omega_{1}}{\omega_{2}}\right)\omega_{2}\chi + \mu G\left(\frac{\omega_{1}}{\omega_{2}}\right)\omega_{1}^{2}\varphi,$$

$$2im\chi - (\vec{\sigma} \cdot \vec{P})\varphi = i\mathscr{E}F\left(\frac{\omega_{1}}{\omega_{2}}\right)\omega_{2}\varphi.$$
(4.4.1)

For this aim we use the projective invariant ansatz (4.1.5) which in this case takes the form

$$\varphi(x) = x_0^{-3/2} \exp\left\{i\frac{m\vec{x}^2}{2x_0}\right\} \widetilde{\phi}(\omega), \qquad \omega = \frac{\vec{\beta}\vec{x}}{x_0},$$

$$\chi(x) = x_0^{-5/2} \exp\left\{i\frac{m\vec{x}^2}{2x_0}\right\} \left(\widetilde{\psi}(\omega) - \frac{i}{2}(\vec{\sigma} \cdot \vec{x})\widetilde{\phi}(\omega)\right)$$
(4.4.2)

Having substituted (4.4.2) into (4.4.1) we obtain the following system of ODEs

$$2m\widetilde{\psi} + (\vec{\sigma} \cdot \vec{\beta})\widetilde{\phi} = \mathscr{E}F\delta_2\widetilde{\phi},$$
$$-(\vec{\sigma} \cdot \vec{\beta})\dot{\widetilde{\psi}} = \mathscr{E}F\delta_2\widetilde{\psi} + \mu G\delta_2^2\widetilde{\phi}$$

where F, G are arbitrary smooth functions of δ_1/δ_2 , $\delta_1 = (\tilde{\phi}^{\dagger}\tilde{\psi} + \tilde{\psi}^{\dagger}\tilde{\phi})^{1/4}$, $\delta_2 = (\phi^{\dagger}\phi)^{1/3}$. It is easy to check that this system has a solution

$$\begin{split} \widetilde{\phi}(\omega) &= \exp\left\{im(\vec{\sigma}\cdot\vec{\beta})\omega\right\}\phi,\\ \widetilde{\psi}(\omega) &= \exp\left\{im(\vec{\sigma}\cdot\vec{\beta})\omega\right\}\chi, \end{split} \tag{4.4.3}$$

where ϕ, χ are constant spinors satisfying the conditions

$$-(m\vec{\beta}^{2})^{2} = (\varpi F \delta_{2})^{2} + 2\mu m G \delta_{2}^{2},$$

$$2m\chi = (\varpi F \delta_{2} - im\vec{\beta}^{2})\phi.$$
(4.4.4)

From (4.4.4)-(4.4.2) we obtain a projective invariant solution of Equation (4.4.1)

$$\varphi(x) = x_0^{-3/2} \exp\left\{im\frac{\vec{x}^2}{2x_0}\right\} \exp\left\{im\omega(\vec{\sigma}\cdot\vec{\beta})\right\}\phi,$$

$$\chi(x) = x_0^{-5/2} \exp\left\{im\left(\frac{\vec{x}^2}{2x_0} + \omega(\vec{\sigma}\cdot\vec{\beta})\right)\right\}\cdot$$

$$\cdot \left(\chi - \frac{i}{2} \exp\left\{-im\omega(\vec{\sigma}\cdot\vec{\beta})\right\}(\vec{\sigma}\cdot\vec{x}) \exp\left\{im\omega(\vec{\sigma}\cdot\vec{\beta})\right\}\phi\right)$$
(4.4.5)

and with the help of (4.1.6) from here it follows a Sch(1,3)-ungenerative solution

$$\begin{split} \varphi(x) &= (x_0 + a_0)^{-3/2} \exp\left\{ im \left[\frac{(\vec{x} + \vec{a})^2}{2(x_0 + a_0)} + y(\vec{\sigma} \cdot \vec{\beta}) + \alpha \right] \right\} \phi, \\ \chi(x) &= (x_0 + a_0)^{-5/2} \exp\left\{ im \left[\frac{(\vec{x} + \vec{a})^2}{2(x_0 + a_0)} + y(\vec{\sigma} \cdot \vec{\beta}) + \alpha \right] \right\} \cdot \\ &\quad \cdot \left(\chi - \exp\left\{ -imy(\vec{\sigma} \cdot \vec{\beta}) \right\} (\vec{\sigma} \cdot \vec{x}) \exp\left\{ imy(\vec{\sigma} \cdot \vec{\beta}) \right\} \phi \right); \\ y &= \frac{\vec{\beta} \cdot \vec{x} + \vec{\beta} \vec{a}}{x_0 + a_0} + \beta_0 \end{split}$$
(4.4.6)

Galilean-invariant ansatz

$$\varphi(x) = \exp\left\{im\frac{\vec{x}^2}{2x_0}\right\} \widetilde{\phi}(\omega), \qquad \omega = \vec{k} \cdot \vec{x} + k_0 x_0$$

$$\chi(x) = \exp\left\{im\frac{\vec{x}^2}{2x_0}\right\} \left(\widetilde{\psi}(\omega) - \frac{i}{2x_0}(\vec{\sigma} \cdot \vec{x})\widetilde{\phi}(\omega)\right)$$
(4.4.7)

reduces (4.4.1) to the following system of ODEs

$$2m\tilde{\psi} + (\vec{\sigma} \cdot \vec{k})\dot{\vec{\phi}} = \approx F\delta_2 \widetilde{\phi}, \qquad \delta_1 = \left(\widetilde{\phi}^{\dagger}\widetilde{\psi} + \widetilde{\psi}^{\dagger}\widetilde{\phi}\right)^{1/4},$$
$$-(\vec{\sigma} \cdot \vec{k})\dot{\vec{\psi}} = \approx F\delta_2 \widetilde{\phi} + \mu G\delta_2 \widetilde{\phi}, \qquad \delta_2 = \left(\widetilde{\phi}^{\dagger}\widetilde{\phi}\right)^{1/3}$$
$$\omega_1\dot{\vec{\phi}} + \frac{3}{2}\widetilde{\phi} = 0,$$

which has the solution

$$\widetilde{\phi}(\omega) = \omega^{-3/2}\phi, \qquad \widetilde{\psi}(\omega) = \omega^{-5/2}\chi$$
(4.4.8)

where ϕ and χ are two-component constant columns satisfying conditions

$$\left[\mathscr{E}F\delta_{2}(\vec{\sigma}\cdot\vec{k}) + \frac{15}{4}k^{2} - (\mathscr{E}F\delta_{2})^{2} - 2\mu mG\delta_{2}^{2} \right]\phi = 0,$$
(4.4.9)

$$2m\chi = \left(\varpi F\delta_2 + \frac{3}{2}(\vec{\sigma} \cdot \vec{k})\right)\phi.$$

Formulae (4.4.7)-(4.4.9) result in a Galilean-invariant solution of the system (4.4.1):

$$\begin{aligned} \varphi(x) &= (\vec{k} \cdot \vec{x} + k_0 x_0)^{-3/2} \exp\left\{\frac{im\vec{x}^2}{2x_0}\right\} \phi, \end{aligned} \tag{4.4.10} \\ \chi(x) &= (\vec{k} \cdot \vec{x} + k_0 x_0)^{-5/2} \exp\left\{\frac{im\vec{x}^2}{2x_0}\right\} \left(\chi - \frac{i}{2x_0} (\vec{k} \cdot \vec{x} + k_0 x_0) (\vec{\sigma} \cdot \vec{x}) \phi\right). \end{aligned}$$

Consider another version of nonlinear equation for Galilean particle with spin s = 1/2, namely Equation [29*, 31*]

$$\left[(i\gamma_0 + \gamma_5)\partial_0 + i\gamma_a\partial_a + \lambda(\overline{\psi}\psi)^k\right]\psi = 0, \qquad (4.4.11)$$

where $\psi = \psi(x)$ is a four-component complex function, γ -matrices are defined in (2.1.2), (2.4.2), λ , k are arbitrary real constants.

Theorem 4.4.1. $[29^*, 31^*]$. The maximal point IA of the system (4.4.11) is given by the following basis IFO:

Under
$$k = 1/3$$
,
 $P_0 = i\partial_0$, $P_a = -i\partial_a$
 $J_{ab} = x_a P_b - x_b P_a + \frac{i}{4}[\gamma_a, \gamma_b],$ (4.4.12)
 $G = \phi^a(x_0) P_a - \frac{1}{2} \dot{\phi}^a(x_0) \gamma_a(i\gamma_0 + \gamma_5),$

$$\widetilde{\Pi} = \phi^0(x_0) P_0 - \dot{\phi}^0(x_0) x^a P_a + \frac{3}{2} \ddot{\phi}^0(x_0) \gamma_a x_a (\gamma_0 - i\gamma_5).$$

Under $k \neq 1/3, k \neq 0$

$$P_{0} = i\partial_{0}, \qquad P_{a} = -i\partial_{a}$$

$$J_{ab} = x_{a}P_{b} - x_{b}P_{a} + \frac{i}{4}[\gamma_{a}, \gamma_{b}], \qquad (4.4.13)$$

$$G = \phi^{a}(x_{0})P_{a} - \frac{1}{2}\dot{\phi}^{a}(x_{0})\gamma_{a}(i\gamma_{0} + \gamma_{5}),$$

$$D = x^{\mu}P_{\mu} + 2i/k,$$

where ϕ^{μ} , $\mu = \overline{0,3}$ are arbitrary smooth functions of x_0 , $\dot{\phi}^{\mu} \equiv d\phi^{\mu}/dx_0$.

One can prove the theorem by means of Lie's method. Note that under k = 1/3 Equation (4.4.11) is invariant under the Schrödinger algebra ASch(1,3), basis elements having the form

$$P_{0} = i\partial_{0}, \qquad P_{a} = -i\partial_{a}$$

$$J_{ab} = x_{a}P_{b} - x_{b}P_{a} + \frac{i}{4}[\gamma_{a}, \gamma_{b}], \qquad (4.4.14)$$

$$G_{a} = x_{0}P_{a} - \frac{1}{2}\gamma_{a}(i\gamma_{0} + \gamma_{5}),$$

$$D = x^{\mu}P_{\mu} + \frac{3}{2}i$$

$$\Pi = x_{0}^{2}P_{0} - 2x_{0}x_{a}P_{a} + 3ix_{0} + \gamma_{a}x_{a}(\gamma_{0} + i\gamma_{5}).$$

The above described symmetry of Equation (4.4.11) can be used for constructing nontrivial formulae of generating solutions. In particular, we have (k = 1/3):

$$\psi_{II}(x) = \phi_0^{3/2} \exp\left\{-\frac{1}{2} \left[\dot{\phi}_a \gamma_a + \dot{\phi}_0 \phi_0^{-1} \gamma_a (x_a + \phi_a)\right] \cdot (4.4.15) \right.$$

$$\cdot (\gamma_0 - i\gamma_5) \left\{\psi_{I} \left(\int \phi_0 dx_0, (x_a + \phi_a) \phi_0^{-1}\right) \right\}$$

Taking as $\psi_I(x)$ the following partial solution of the Equation (4.4.11) under k = 1/3

$$\psi_I(x) = \exp\left\{-i\lambda(\gamma_a\theta_a)(\theta_b x_b)(\overline{\chi}\chi)^{1/3}\right\}\chi,\tag{4.4.16}$$

(θ_a are arbitrary constants) we get by means of (4.4.15) another family of

solutions of the equation in question

$$\psi(x) = \phi_0^{3/2} \exp\left\{-\frac{1}{2} \left(\gamma_a \dot{\phi}_a + \dot{\phi}_0 \phi_0^{-1} \cdot \right. \\\left. \cdot \gamma_a(x_a + \phi_a) \right) (\gamma_0 - i\gamma_5) \right\} \exp\left\{-i\lambda(\gamma_a \theta_a) \cdot \right. \\\left. \cdot \left(\theta_b x_b + \theta_b \phi_b\right) \phi_0^{-1}(\overline{\chi}\chi)^{1/3} \right\} \chi.$$

$$(4.4.17)$$

It is interesting to note that Equation (4.4.11) under k = 1/3 can be considered as nonrelativistic counterpart of the nonlinear Dirac-Gursey equation (2.1.5).

4.5. Symmetry analysis of gas dynamics equations

The aim of this paragraph is to study the symmetry of the basic equations of gas dynamics

$$\frac{\partial \vec{u}}{\partial x_0} + \left(\vec{u} \cdot \vec{\nabla}\right) \vec{u} + \frac{1}{\rho} \vec{\nabla} P = 0,$$

$$\frac{\partial \rho}{\partial x_0} + \operatorname{div}\left(\rho \vec{u}\right) = 0,$$

$$P = f(\rho),$$
(4.5.1)

where $\vec{u} = \vec{u}(x) = \{u^1, \ldots, u^n\}$ is the velocity vector of the gas spreading; $x = (x_0, \vec{x}) = (x_0, x_1, \ldots, x_n) \in \mathbb{R}^{n+1}$; $\rho = \rho(x)$ and P = P(x) are the density and pressure of the gas, and to obtain their exact solutions.

At the beginning we establish that the one-dimensional Equations (4.5.1) in the special case of isentropic and polytropic gas possess an infinite-dimensional group of point transformations. It allows us to construct the general solution of the one-dimensional Equations (4.5.1) [90]. We also study symmetry of multi-dimensional Equations (4.5.1) in some special cases of gas motion and construct exact solutions of these equations [90,91].

1. One-dimensional isentropic gas motion.

When number of spatial variables n = 1, Equations (4.5.1) take the form

$$\frac{\partial \rho}{\partial x_0} + \rho \frac{\partial u}{\partial x_1} + u \frac{\partial \rho}{\partial x_1} = 0,$$

$$\rho \frac{\partial u}{\partial x_0} + \rho u \frac{\partial u}{\partial x_1} + c^2(\rho) \frac{\partial \rho}{\partial x_1} = 0,$$
(4.5.2)

where $c^2(\rho)$ is called sound velocity and is determined by the formula

$$c(\rho) = \left[\frac{\partial f(\rho)}{\partial \rho}\right]^{1/2} > 0,$$

 $p = f(\rho)$ is a given monotonic function.

Equations (4.5.2) were studied as far back as the end of the 17th century. In 1809, Poisson obtained a solution of these equations in the form of ordinary wave

$$u = F(x - (u + c)t)$$

where F is arbitrary differentiable function. Challis [37] noted that Equations (4.5.2) have not always a unique solution for velocity u. To obtain the unique solution Stokes suggested [197] a criterion of a breakdown of the velocity u (when the derivative of u aims at the infinity) and obtained two conditions on the breakdown. Earnshaw [47] found a solution of Equations (4.5.2) in the form of ordinary wave for arbitrary dependence $p = f(\rho)$. Independently, Riemann [171] developed theory of ordinary wave and gave the general solution of the Equations (4.5.2) by virtue of Riemann invariants.

A great contribution to the theory of gas dynamics was made by Hugoniot [122], Sedov, Hristianovich, Stanyukovich [196], Zeldovich, Landau and Lifshits [140].

Kurant [137] had shown that Equations (4.5.2) can be linearized by means of hodograph transformation

$$V^{\mu} = V^{\mu}(y_0, y_1), \qquad V^{\mu} = x_{\mu}, \qquad y_{\nu} = u^{\nu}, \qquad V^{\mu}_{\nu} = \frac{\partial V^{\mu}}{\partial y_{\nu}}, \qquad (4.5.3)$$
$$x_0 = t, \qquad x_1 = x, \qquad u^0 = \rho, \qquad u^1 = u; \qquad \mu, \nu = 0, 1.$$

As a result of application of (4.5.3) to (4.5.2) one obtains the system of linear equations

$$V_1^1 + y_0 V_0^0 - y_1 V_1^0 = 0,$$

$$V_0^1 - y_1 V_0^0 + h(y_0) V_1^0 = 0,$$
(4.5.4)

where $h(y_0) = h(\rho) = \frac{1}{2}c^2(\rho)$.

In case of polytropic gas, when $p = A\rho^{\gamma}$, with A, γ constants and $\gamma = \frac{2N+1}{2N-1}$, $N = 0, 1, 2, \ldots$, Equations (4.5.4) are reduced to the following second-order system of PDEs

$$V_{rs}^{0} + N \frac{V_{r}^{0} + V_{s}^{0}}{r+s} = 0, \qquad (4.5.5)$$

$$V_{r}^{1} = \left(r - s - \frac{r + s}{2N - 1}\right) V_{r}^{0},$$

$$V_{s}^{1} = \left(r - s + \frac{r + s}{2N - 1}\right) V_{s}^{0},$$
(4.5.6)

where

$$r = \frac{1}{2} \left(y_1 + \sqrt{A(4N^2 - 1)} y_0^{1/(2N-1)} \right),$$

$$s = \frac{1}{2} \left(-y_1 + \sqrt{A(4N^2 - 1)} y_0^{1/(2N-1)} \right)$$
(4.5.7)

are Riemannian invariants.

One can easy recognize in (4.5.5) the Darboux equation. Its general solution has the form

$$V^0 = \varphi(r) + g(s), \qquad N = 0$$
 (4.5.8)

$$V^{0} = k + \frac{\partial^{N-1}}{\partial r^{N-1}} \left(\frac{\varphi(r)}{(r+s)^{N}} \right) + \frac{\partial^{N-1}}{\partial s^{N-1}} \left(\frac{g(s)}{(r+s)^{N}} \right), \qquad N \ge 1$$
(4.5.9)

where φ , g are arbitrary differentiable functions, and k is a constant.

If one succeeds in finding the general solution of Equations (4.5.6), thereby making use of the inverse hodograph transformation, one constructs the general solution of Equations (4.5.2). When $\gamma = 3$ (N = 1) it is succeeded to do and in such a case the general solution of Equations (4.5.2) has the form (see [196])

$$x_1 - (u^1 + 3Au^0)x_0 = F^1(u^1 + 3Au^0),$$

$$x_1 - (u^1 - 3Au^0)x_0 = F^2(u^1 - 3Au^0),$$
(4.5.10)

where F^1 , F^2 are arbitrary differentiable functions.

It will be noted that under $\gamma = -1$ (N = 0) the explicit form of solution of the system (4.5.2) can be also easy found. By substituting (4.5.8) into (4.5.6) and denoting $\varphi(r) = F'(r)$, g(s) = G'(s) we get the general solution of Equations (4.5.6) as follows

$$V^{1} = rF'(r) - F(r) - sG'(s) + x.$$
(4.5.11)

Then by means of the inverse hodograph transformation we obtain from (4.5.8), (4.5.11) the general solution of Equations (4.5.2)

$$t \equiv x_0 = F'(u - \sqrt{-A\rho^{-1}}) + G'(u + \sqrt{-A\rho^{-1}}),$$

$$x \equiv x_1 = (u - \sqrt{-A\rho^{-1}})F'(u - \sqrt{-A\rho^{-1}}) - F(u - \sqrt{-A\rho^{-1}}) - (4.5.12) - (u + \sqrt{-A\rho^{-1}})G'(u + \sqrt{-A\rho^{-1}}) + G(u + \sqrt{-A\rho^{-1}}) + k$$

Group properties of gas-dynamics Equations (4.5.1) had been studied by Ovsyannikov and summarized in [162]. Although this investigation was made for arbitrary number of independent variables, the two-dimensional case needs individual consideration. In this case the symmetry group of gas dynamics equations is infinite-dimensional. This result established in [90] does not follow from [162].

Theorem 4.5.1. [90]. One-dimensional equations of gas dynamics (4.5.2)
under
$$p = A\rho^{\gamma}, \frac{2N+1}{2N-1}, N = 0, 1, 2, \dots$$
, that is
 $\gamma = -1, 3, \frac{5}{3}, \frac{7}{5}, \dots$ (4.5.13)

are invariant with respect to the infinite-dimensional Lie group, coordinates $\xi^0, \, \xi^1, \, \eta$ of corresponding IFO

$$X = \xi^0 \frac{\partial}{\partial x_0} + \xi^1 \frac{\partial}{\partial x_1} + \eta \frac{\partial}{\partial u}$$
(4.5.14)

are solutions of Equations (4.5.5), (4.5.6).

Proof. Using notations (4.5.3) Equations (4.5.2) can be rewritten as follows

$$u_0^0 + u^1 u_1^0 + u^0 u_1^1 = 0,$$

$$u_0^1 + u^1 u_1^1 + h(u^0) u_1^0 = 0.$$
(4.5.15)

Applying to (4.5.15) the criterion of invariance (3) one obtains as a result the determining equations for functions ξ^0 , ξ^1 , η :

$$\eta_0^0 + u^1 \eta_1^0 + u^0 \eta_1^1 = 0,$$

$$\eta_0^1 + u^1 \eta_1^1 + h(u^0) \eta_1^0 = 0.$$
(4.5.16)

$$\eta^{1} = -u^{1}(\xi_{0}^{0} - \xi_{1}^{1} + u^{1}\xi_{1}^{0}) - h(u^{0})(u^{0}\xi_{1}^{0} + \eta_{u^{1}}^{0}) + \xi_{0}^{1} + u^{0}\eta_{u^{0}}^{1},$$

$$\eta^{1} = -u^{1}(\xi_{0}^{0} - \xi_{1}^{1} + u^{1}\xi_{1}^{0}) - h(u^{0})(u^{0}\xi_{1}^{0} - \eta_{u^{1}}^{0}) + \xi_{0}^{1} - u^{0}\eta_{u^{0}}^{1};$$
(4.5.17)

$$\eta^{0} = -u^{0}(\xi_{0}^{0} - \xi_{1}^{1} + 2u^{1}\xi_{1}^{0} + \eta_{u^{1}}^{1} - \eta_{u^{0}}^{0}),$$

$$\eta^{0} = -\frac{h(u^{0})}{h'(u^{0})}(\xi_{0}^{0} - \xi_{1}^{1} + 2u^{1}\xi_{1}^{0} - \eta_{u^{1}}^{1} + \eta_{u^{0}}^{0});$$
(4.5.18)

$$\begin{aligned} \xi_{u^0}^1 &= u^1 \xi_{u^0}^0 - h(u^0) \xi_{u^1}^0, \\ \xi_{u^1}^1 &= u^1 \xi_{u^1}^0 - u^0 \xi_{u^0}^0. \end{aligned}$$
(4.5.19)

Compatibility condition of Equations (4.5.19) yields

$$\xi_{u^0 u^0}^0 - \frac{h(u^0)}{u^0} \xi_{u^1 u^1}^0 + 2\frac{\xi_{u^0}^0}{u^0} = 0.$$
(4.5.20)

1. Consider at first the case $p = A\rho^3$. Then (4.5.20) results in the Darboux equation for

$$\xi_{u^0 u^0}^0 - \lambda^2 \xi_{u^1 u^1}^0 + \frac{2}{u^0} \xi_{u^0}^0 = 0, \qquad (\lambda^2 = 3A)$$
(4.5.21)

The general solution of (4.5.21) has the form

$$\xi^{0} = \frac{F(x, u^{1} + \lambda u^{0}) + G(x, u^{1} - \lambda u^{0})}{2\lambda u^{0}}, \qquad (4.5.22)$$
where F and G are arbitrary differentiable functions.

It is not difficult to find function ξ^1 from (4.5.22), (4.5.19)

$$\xi^{1} = \frac{(u^{1} - \lambda u^{0})F(x, u^{1} + \lambda u^{0}) + (u^{1} + \lambda u^{0})G(x, u^{1} - \lambda u^{0})}{2\lambda u^{0}} + H(x), \quad (4.5.23)$$

where H(x) is an arbitrary differentiable function.

Since $p = A\rho^3$ results in $h(u^0) = \lambda^2 u^0$, we get from (4.5.18), (4.5.19)

$$\eta_{u^1}^0 = \eta_{u^0}^1, \qquad \eta_{u^0}^0 = \eta_{u^1}^1; \tag{4.5.24}$$

$$\eta^{0} = -\lambda u^{0} (\xi_{0}^{0} - \xi_{1}^{1} + 2u^{1} \xi_{1}^{0}),$$

$$\eta^{1} = -u^{1} (\xi_{0}^{0} - \xi_{1}^{1} + u^{1} \xi_{1}^{0}) - (\lambda u^{0})^{2} \xi_{1}^{0} + \xi_{0}^{1}.$$
(4.5.25)

Following from (4.5.16), (4.5.23), (4.5.25) we find that functions F, G, and H should satisfy the equations

$$F_{00} + 2(u^{1} + \lambda u^{0})F_{01} + (u^{1} + \lambda u^{0})^{2}F_{11} = 0,$$

$$G_{00} + 2(u^{1} - \lambda u^{0})G_{01} + (u^{1} - \lambda u^{0})^{2}G_{11} = 0,$$

$$H_{00} = H_{11} = H_{01} = 0.$$
(4.5.26)

The general solution of system (4.5.26) has the form

$$F = x_0 F^0 (x_0 (u^1 + \lambda u^0)) - x_1) + G^0 (x_0 (u^1 + \lambda u^0)) - x_1)$$

$$G = x_0 F^1 (x_0 (u^1 - \lambda u^0)) - x_1) + G^1 (x_0 (u^1 - \lambda u^0)) - x_1)$$

$$H = c_{\nu} x_{\nu} + d,$$
(4.5.27)

where F^{ν} , G^{ν} are arbitrary differentiable functions, c_{ν} and d are arbitrary constants, $\nu = 0, 1$.

Now, using formulae (4.5.27), (4.5.25), (4.5.22), (4.5.23) we obtain

$$\begin{aligned} \xi^{0} &= (2\lambda u^{0})^{-1} (x_{0}F^{0} + G^{0} + x_{0}F^{1} + G^{1}), \\ \xi^{1} &= (2\lambda u^{0})^{-1} \left[(u^{1} - \lambda u^{0})(x_{0}F^{0} + G^{0}) + (u^{1} + \lambda u^{0})(x_{0}F^{1} + G^{1}) \right] + c_{\nu}x_{\nu} + d, \\ \eta^{0} &= -\frac{1}{2} (F^{0} + F^{1}) + \lambda c_{1}u^{0}, \end{aligned}$$

$$(4.5.28)$$

$$\eta^{1} &= -\frac{1}{2} (F^{0} - F^{1}) + c_{1}u^{1} + c_{0}, \end{aligned}$$

where F^{ν} , G^{ν} are the same functions as in (4.5.27).

Since ξ^{μ} and η^{μ} depend on arbitrary functions F^{ν} , G^{ν} , it means that the symmetry group of Equations (4.5.15) under $\gamma = 3$ is infinite-dimensional, the infinitesimal transformations having the form

$$\begin{aligned} x'_{\mu} &= x_{\mu} + \epsilon \xi^{\mu} + O(\epsilon^2), \\ u'^{\nu} &= u^{\nu} + \epsilon \eta^{\nu} + O(\epsilon^2), \end{aligned}$$

$$(4.5.29)$$

where ξ^{μ} , η^{ν} are given by (4.5.28). So, case 1 of the theorem is proved.

Under $\gamma = \frac{2N+1}{2N-1}$, N = 0, 2, 3, 4, ... the determining Equations (4.5.19), (4.5.20) coincide with (4.5.5), (4.5.6) and by means of hodograph transformation they are reduced to the initial system (4.5.15); the general solution of this latter, as we have shown above, contain arbitrary functions. This fact accomplishes the proof.

Remark 4.5.1. In a particular case when

$$F^{0} = a\omega^{0}, \qquad F^{1} = -a\omega^{1}, \qquad G^{0} = b\omega^{0}, \qquad G^{1} = -b\omega^{1},$$

$$\omega^{0} = x_{0}(u^{1} + \lambda u^{0}) - x_{1}, \qquad \omega^{1} = x_{0}(u^{1} - \lambda u^{0}) - x_{1}$$

the above proved theorem gives rise to the result obtained under $\gamma = 3$ in [162].

2. In this item we consider another way of constructing the general solution of the two-dimensional equations of gas dynamics with $\gamma = 3$ (i.e., $p = A\rho^3$).

Let us note, that arbitrary functions which are contained in the general solution of Equations (4.5.2) depend on the same arguments as arbitrary functions contained by coordinates of IFO do. So, it is naturally to pass on these arguments in system (4.5.15) $(h(u^0) = \lambda(u^0)^2, \gamma = 3)$:

$$V^{0} = (u^{1} + \lambda u^{0})x_{0} - x_{1},$$

$$V^{1} = (u^{1} - \lambda u^{0})x_{0} - x_{1}.$$
(4.5.30)

This change of variables results in decomposed PDEs

$$x_0 V_0^0 + (V^0 + x_1) V_1^0 = 0,$$

$$x_0 V_0^1 + (V^1 + x_1) V_1^1 = 0.$$
(4.5.31)

the general solution of which has the form

$$V^{0} = x_{0}\varphi^{0}(V^{0}) - x_{1},$$

$$V^{1} = x_{0}\varphi^{1}(V^{1}) - x_{1}.$$
(4.5.32)

 $(\varphi^0, \varphi^1$ are arbitrary differentiable functions), or

$$u^{1} + \lambda u^{0} = \varphi^{0} (x_{0}(u^{1} + \lambda u^{0}) - x_{1}),$$

$$u^{1} - \lambda u^{0} = \varphi^{1} (x_{0}(u^{1} - \lambda u^{0}) - x_{1}).$$
(4.5.33)

This latter coincides with (4.5.10).

3. In this item the symmetry of multidimensional equations of gas dynamics is briefly discussed and then some exact solutions of these equations are obtained.

As we have already noted the symmetry properties of multidimensional gas dynamics equations were studied in [162]. In particular, there is established 13-parameter maximal invariance group G_{13} of these equations in the case n = 3 and

$$p = \lambda \rho^{5/3}.$$
 (4.5.34)

The basis elements of corresponding Lie algebra have the form

$$\partial_{\mu} = \frac{\partial}{\partial x_{\mu}}; \qquad \mu = \overline{0,3}$$

$$(4.5.35)$$

$$J_{ab} = x_a \partial_b - x_b \partial_a + u^a \partial_{u^b} - u^b \partial_{u^a}; \qquad a, b = \overline{1, 3};$$

$$(4.5.36)$$

$$G_a = x_0 \partial_a + \partial_{u^a}; \tag{4.5.37}$$

$$\mathcal{D}_0 = x_0 \partial_0 + x_a \partial_a; \tag{4.5.38}$$

$$\mathcal{D}_1 = x_0 \partial_0 - 3\rho \partial_\rho - u^a \partial_{u^a}; \tag{4.5.39}$$

$$\Pi = x_0 (x_0 \partial_0 + x_a \partial_a - 3\rho \partial_\rho - u^a \partial_{u^a}) + x_a \partial_{u^a}.$$
(4.5.40)

It is the widest symmetry group of Equations (4.5.1). In other cases relation (4.5.34) does not hold, the maximal symmetry group of Equations (4.5.1) is smaller than the above G_{13} . If

$$p = \lambda \rho^{\gamma}, \tag{4.5.41}$$

where λ and γ are arbitrary constants, $\gamma \neq \frac{5}{3}$, then the maximal IA of Equations (4.5.1) is 12-dimensional and determined by operators (4.5.35)–(4.5.38) and the operator

$$\mathcal{D}_1 = x_0 \partial_0 - \frac{2}{\gamma - 1} \rho \partial_\rho - u^a \partial_{u^a}. \tag{4.5.42}$$

If

$$p = f(\rho) \tag{4.5.43}$$

where $f(\rho)$ is an arbitrary differentiable function, $f(\rho) \neq \lambda \rho^{\gamma}$, then the maximal IA of Equations (4.5.1) is 11-dimensional and determined by operators (4.5.35)-(4.5.38).

Using the symmetry of Equations (4.5.1) we find their exact solutions. At first consider the polytropic gas (that is relation (4.5.41) holds). In this case Equations (4.5.1) take the form

$$\vec{u}_0 + (\vec{u}\vec{\nabla})\vec{u} + \lambda\rho^{\gamma-2}\vec{\nabla}\rho = 0,$$

$$\rho_0 + (\vec{u}\vec{\nabla})\rho + \rho \operatorname{div} \vec{u} = 0.$$
(4.5.44)

Since Equations (4.5.44) are invariant under AG(1,3) (4.5.35)-(4.5.40) we look for solutions in the form

$$\rho \equiv u^{0} = x_{0}^{k} \varphi^{0}(\omega),
\vec{u} = \vec{M}(x_{0}) \varphi^{1}(\omega) + \vec{N}(x_{0}) \varphi^{2}(\omega) + \vec{L}(x_{0}) \varphi^{3}(\omega) + \vec{B}(x_{0}).$$
(4.5.45)

One can use invariant variables ω obtained in §3.6. Without going into details we present below corresponding functions \vec{M} , \vec{N} , \vec{L} , \vec{B} and indicate the constant k:

1. $\vec{M}(x_0) = \vec{\alpha} x_0^{-1/2}$, $\vec{N}(x_0) = \vec{\beta} x_0^{-1/2}$, $\vec{L}(x_0) = \vec{\gamma} x_0^{-1/2}$ $\vec{B}(x_0) = \vec{V}, \qquad k = (1 - \gamma)^{-1}.$ where $\vec{\alpha}\vec{V} = -b$. $\vec{\beta}\vec{V} = c\vec{\beta}\vec{\delta} - \alpha\vec{\gamma}\vec{\delta}$. $\vec{\gamma}\vec{V} = c\gamma\delta + \alpha\vec{\beta}\vec{\delta}$: 2. $\vec{M}(x_0) = \vec{\alpha} x_0^{-1/2}$, $\vec{N}(x_0) = \vec{\beta} x_0^{-1/2}$, $\vec{L}(x_0) = \vec{\gamma} x_0^{-1/2}$ $\vec{B}(x_0) = \vec{V}, \qquad k = (1 - \gamma)^{-1}$ where $\vec{\alpha}\vec{V} = \vec{\alpha}\vec{\delta}$, $\vec{\beta}\vec{V} = \vec{\beta}\vec{\delta} - a\vec{\alpha}\vec{\delta}$, $\vec{\gamma}\vec{V} = \vec{\gamma}\vec{\delta} - 2a\vec{\beta}\vec{\delta} + 2a^2\vec{\alpha}\vec{\delta}$; 3. $\vec{M}(x_0) = \vec{\alpha} x_0^{-1/2}, \qquad \vec{N}(x_0) = \vec{\beta} x_0^{-1/2}, \qquad \vec{L}(x_0) = \vec{\gamma} x_0^{-1/2}.$ $\vec{B}(x_0) = \vec{V}, \qquad k = (1 - \gamma)^{-1}$ where $\vec{\alpha}\vec{V} = -b$, $\vec{\beta}\vec{V} = -b_{\perp}$, $\vec{\gamma}\vec{V} = -b_{\perp}$; 4. $\vec{M}(x_0) = \vec{\alpha} x_0^{-1/2}, \qquad \vec{N}(x_0) = \vec{\gamma} x_0^{-1}, \qquad \vec{L}(x_0) = \vec{\beta}.$ $\vec{B}(x_0) = -\frac{\vec{\beta}b_4}{\beta^2} \ln x_0 - \frac{\vec{\gamma}b_2}{\beta^2}, \qquad k = (1 - \gamma)^{-1},$ 5. $\vec{M}(x_0) = \vec{\alpha}, \qquad \vec{N}(x_0) = \vec{\beta},$ $\vec{L}(x_0) = \vec{\gamma},$ $\vec{B}(x_0) = -\vec{\alpha} \frac{2a}{\alpha^2} x_0 + \vec{V}, \qquad k = 0,$ where $\vec{\alpha}\vec{V} = 0$, $\vec{\beta}\vec{V} = \vec{\gamma}\vec{\delta}$, $\vec{\gamma}\vec{V} = -\vec{\beta}\vec{\delta}$. (4.5.45')6. $\vec{M}(x_0) = \vec{\alpha}$, $\vec{N}(x_0) = \vec{\gamma} \exp(-bx_0)$, $\vec{L}(x_0) = \vec{\beta} \exp(bx_0)$, $\vec{B}(x_0) = \vec{\alpha} \frac{2a}{2} x_0 + \vec{V}, \qquad k = 0,$ where $\vec{\alpha}\vec{V} = 0$, $\vec{\beta}\vec{V} = \vec{\beta}\vec{\delta}$, $\vec{\gamma}\vec{V} = -\vec{\gamma}\vec{\delta}$; 7. $\vec{M}(x_0) = \vec{e}_1$, $\vec{N}(x_0) = \vec{e}_2$, $\vec{L}(x_0) = \vec{e}_3$, $\vec{B}(x_0) = 0.$ k=0.

where \vec{e}_a are orthonormal constant vectors, a = 1, 2, 3;

8. $\vec{M}(x_0) = \vec{e}_1(x_0)^{-1/2}$, $\vec{N}(x_0) = \vec{e}_2(x_0)^{-1/2}$, $\vec{L}(x_0) = \vec{e}_3(x_0)^{-1/2}$, $\vec{B}(x_0) = 0$, $k = (1 - \gamma)^{-1}$.

Substitution of (4.5.45), (4.5.45') into (4.5.44) gives rise to the following reduced PDEs (the notation is used: $\varphi = (\varphi^0, \vec{\varphi}), \phi_s = (\phi_s^0, \vec{\varphi}_s)$):

1.
$$(2\varphi^1 - \omega_1)\varphi_1 + (2\varphi^2 - \omega_2 + 2\omega_3)\varphi_2 - (2\varphi^3 + \omega_3 + 2\omega_2)\varphi_3 + (4.5.46) + \phi_1 = 0,$$

where
$$\phi_1^0 = 2\left(\operatorname{div} \vec{\varphi} - \frac{1}{\gamma - 1}\right) \varphi^0$$
, $\vec{\phi} = -\vec{\varphi} + 2\vec{\psi}_1 + 2\lambda(\varphi^0)^{\gamma - 2}\vec{\nabla}\varphi^0$, $\vec{\psi}_1 = (0, -\varphi_3, \varphi_2);$
2. $(2\varphi^1 - \omega_1)\varphi_1 + (2\varphi^2 - \omega_2 + 2\omega_1)\varphi_2 + (4.5.47) + (2\varphi^3 - \omega_3 + 2\varphi^1 + 4\omega_2)\varphi_3 + \phi_2 = 0$,
where $\phi_2^0 = \phi_1^0 + 2\left(\varphi_3^1 + \varphi_1^3 - \varphi_1^1\right)\varphi^0$, $\vec{\phi}_2 = -\vec{\varphi} + 2\vec{\psi}_2 + 2\lambda(\varphi^0)^{\gamma - 2}\vec{\nabla}\varphi^0$, and
 $\vec{\psi}_2 = (\varphi^2, 2\varphi^3, 0);$
3. $(2\varphi^1 + \omega_1)\varphi_1 - (2\varphi^3 - 2a + 2\omega_2)\varphi_2 - (2\varphi^2 - 2a\omega_3)\varphi_3 + \phi_3 = 0, (4.5.48) + (4.5.49) + (4.5.49) + (4.5.49) + (4.5.49) + (2\varphi^1 + \omega_1)\varphi_1 - 2(\varphi^3 - b_3)\varphi_2 - 2(\varphi^3 - \omega_3 + b_4)\varphi_3 + \phi_4 = 0, (4.5.49) +$

Knowing a solution of Equations (4.5.46)-(4.5.53) one obtains by means of (4.5.45) a partial solutions of Equations (4.5.44).

Consider Equations (4.5.46) and suppose $\varphi_1 = 0$, that is $\varphi = \varphi(\omega_2, \omega_3)$. It

follows

$$(2\varphi^2 - \omega_2 + 2\omega_3)\varphi_2 - (2\varphi^3 + \omega_3 + 2\omega_2)\varphi_3 + \phi_1 = 0.$$
(4.5.54)

If $\varphi = \varphi(\omega_1)$ then (4.5.46) is directly reduced to the ODE

$$(2\varphi^1 - \omega_1)\varphi_1 + \phi_1 = 0,$$

which has under $\gamma = 2$ a partial solution

$$\varphi^{0} = c_{0}, \qquad \varphi^{1} = \frac{\omega_{1}}{\alpha^{2}},$$

$$\varphi^{2} = c_{1}\omega_{1}\sin\left(\frac{\alpha}{c}\ln c_{2}\omega_{1}\right), \qquad (4.5.55)$$

$$\varphi^{3} = c_{1}\omega_{1}\cos\left(\frac{\alpha}{c}\ln c_{2}\omega_{1}\right).$$

So, combining (4.5.55) and (4.5.45) we get a partial solution of Equations (4.5.44) under $\gamma = 2$:

$$\rho = c_0 x_0^{-1}, \vec{u} = M(x) x_0^{-1/2} \left[\vec{\alpha} + c_1 \vec{\beta} \sin\left(\ln c_2 M(x)\right) + c_1 \vec{\gamma} \cos\left(\ln c_2 M(x)\right) \right]$$
(4.5.56)

where $M(x) = x_0^{-1/2} (\vec{\alpha} \vec{x} + bx_0)$.

Equation (4.5.47) under $\varphi = \varphi(\omega_1)$ takes the form

$$(2\varphi^1 - \omega_1)\varphi_1 + \phi_2 = 0.$$

This latter system has a partial solution

$$\varphi^{0} = c_{0}\omega_{1}^{\frac{\gamma}{\gamma-1}} \exp(-2\omega_{1}),$$

$$\varphi^{1} = a\omega_{1}(\omega_{1} + c_{2})^{2} + \frac{2\lambda}{\gamma-1}c_{0}^{\gamma-1}\omega_{1}\exp[-2(\gamma-2)\omega_{1}] + c_{1}\omega_{1},$$

$$\varphi^{2} = -2a\omega_{1}(\omega_{1} + c_{2}),$$

$$\varphi^{3} = \omega_{1},$$

(4.5.57)

where c_0 , c_1 , c_2 are arbitrary constants. Using this result we find a partial solution of the system (4.5.44)

$$\rho = x_0^{\frac{1}{\gamma - 1}} [B(x)]^{\frac{2}{\gamma - 1}} \exp[-2B(x)],$$

$$\vec{u} = x_0^{-1/2} B(x) \bigg\{ \vec{\alpha} \bigg[a(B(x) + c_2)^2 + \frac{2\lambda}{\gamma - 1} c_0^{\gamma - 1} \cdot (4.5.58) \bigg\}$$

 $\mathbf{238}$

$$\cdot \exp\left\{-2(\gamma-1)B(x)\right\} + c_1 - 2a\vec{\beta}(B(x) + c_2) + \vec{\gamma} + \vec{V},$$

where $B(x) = x_0^{-1/2} (\vec{\alpha} \vec{x} - \vec{\alpha} \vec{A}(x_0)).$

Consider Equations (4.5.49). They are successfully solved when $\varphi = \varphi(\omega_1)$. In such a case Equations (4.5.49) take the form

$$(2\varphi^1 + \omega_1)\varphi_1 + \phi_4 = 0. \tag{4.5.59}$$

Under $\gamma = 2, b_4 = 0$ we found its solution

$$\varphi^0 = -\frac{1}{2\lambda} \left(\omega_1^2 + c_0 \right), \qquad \varphi^1 = -\frac{1}{2}\omega_1, \qquad \varphi^2 = 0, \qquad \varphi^3 = F(\omega_1),$$

where c_0 is a constant, F an arbitrary differentiable function. Using this result we obtain a partial solution of Equations (4.5.44)

$$\rho = \frac{(\vec{\alpha}\vec{x} + b_1 x_0)^2}{8\lambda x_0},
\vec{u} = \frac{\vec{\alpha}}{2x_0}(\vec{\alpha}\vec{x} + b_1 x_0) + \vec{\beta}F\left(\frac{\vec{\alpha}\vec{x} + b_1 x_0}{\sqrt{x_0}}\right),$$
(4.5.60)

where $\vec{\alpha} = 1$, $\vec{\alpha}\vec{\beta} = 0$, $\vec{\beta}^2 = 0$. Note, that this solution contains arbitrary function F.

Next consider Equations (4.5.50). If function $\varphi(\omega)$ does not depend on ω_2 , ω_3 , we get instead of (4.5.50) the system of ODEs

$$(\varphi^1 + bc)\varphi_1 + \phi_5 = 0 \tag{4.5.61}$$

Under $\gamma = -1$ its general solution is given by

$$\varphi^{0} = \sqrt{\frac{c_{0}^{2} - \lambda}{4a(\omega_{1} + c_{1})}}, \qquad \varphi^{1} = \frac{c_{0}}{\varphi^{0}} - bc, \qquad (4.5.62)$$

$$\varphi^{2} = c_{2} \sin \left[\frac{2b}{c_{0}} (\omega_{1} + c_{1}) \varphi^{0} + c_{3} \right], \qquad (4.5.63)$$
$$\varphi^{3} = c_{2} \cos \left[\frac{2b}{c_{0}} (\omega_{1} + c_{1}) \varphi^{0} + c_{3} \right],$$

where c_0, c_1, c_2, c_3 are arbitrary constants. The corresponding solution of the system (4.5.44) under $\gamma = -1$ has the form

 $\rho = A(x)$

$$\vec{u} = \vec{\alpha} \left(\frac{c_0}{A(x)} + 2ax_0 - bc \right) + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_2 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_3 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_3 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_3 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_3 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_3 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_3 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_4 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_4 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_4 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha} + ax_0^2 + bcx_0 + c_1)A(x) + c_3 \right] + c_4 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha} + ax_0 + bcx_0 + c_1)A(x) + c_1 \right] + c_4 \vec{\beta} \sin\left[\frac{2b}{c_0} (\vec{\alpha} + ax_0$$

Chapter 4. Systems of PDEs Invariant Under Galilei Groups

$$+c_2\vec{\gamma}\cos\left[\frac{2b}{c_0}(\vec{\alpha}\vec{x}+ax_0^2+bcx_0+c_1)A(x)+c_3\right]+\vec{V},$$
(4.5.64)

where

$$A(x) = \sqrt{\frac{c_0^2 - \lambda}{4a(\vec{\alpha}\vec{x} + ax_0^2 + bcx_0 + c_1)}}.$$

It is well to note, that because solutions (4.5.56), (4.5.58), (4.5.60), (4.5.64) contain arbitrary functions, they can be used to solve corresponding boundary-value or initial-value problems.

In the same way one can construct solutions of Equations (4.5.1) when $p = f(\rho)$ with arbitrary differentiable function f. In such a case Equations (4.5.1) are invariant with respect to the Galilei group G(1,3) and one can use invariants of this group and formulae (4.5.45). By substituting these ansatze into (4.5.1) one gets the reduced equations which coincide with (4.5.50)-(4.5.53) with the only exception: instead of the term $\lambda(\varphi^0)^{\gamma-2}$ we will have $(\varphi^0)^{-1}f'(\varphi^0)$.

As an example consider reduced equations corresponding (4.5.52):

$$(1 - \vec{\alpha}\vec{\varphi})\varphi_1 + (1 - \vec{\beta}\vec{\varphi})\varphi_2 + (1 - \vec{\gamma}\vec{\varphi})\varphi_3 + \psi_7 = 0, \psi_7^0 = \phi_7^0, \qquad \psi_7 = (\vec{\alpha}\varphi_1^0 + \vec{\beta}\varphi_2^0 + \vec{\gamma}\varphi_3^0)(\varphi^0)^{-1}f'(\varphi^0).$$
(4.5.65)

Supposing $\varphi = \varphi(\omega_1)$ we obtain from (4.5.65) the system of PDEs

This system has a solution

$$\varphi^0 = c_0 \tag{4.5.67}$$
$$\vec{\varphi} = \vec{F}(\omega_1),$$

where \vec{F} is an arbitrary differentiable vector-function, satisfying the condition $\vec{\alpha}\vec{F} = 1, c_0$ is a constant. There follows a partial solution of Equations (4.5.1)

$$\rho = c_0
\vec{u} = \vec{F}(x_0 - \vec{\alpha}\vec{x}),$$
(4.5.68)

It is appropriate to present here formulae of generating solutions of the system (4.5.1) under $p = \lambda \rho^{5/3}$.

The projective transformations have the form

$$\begin{aligned} x'_{0} &= \frac{x_{0}}{1 - \theta x_{0}}, \qquad \vec{x}' = \frac{\vec{x}}{1 - \theta x_{0}}, \\ \rho' &= \rho (1 - \theta x_{0})^{3}, \qquad \vec{u}' = \vec{u} (1 - \theta x_{0}) + \theta \vec{x}, \end{aligned}$$
(4.5.69)

(θ is a parameter) and the corresponding formulae of generating solutions are

$$\rho = (1 - \theta x_0)^{-3} F\left(\frac{x_0}{1 - \theta x_0}, \frac{\vec{x}}{1 - \theta x_0}\right),$$

$$\vec{u} = (1 - \theta x_0)^{-1} \left[\vec{G}\left(\frac{x_0}{1 - \theta x_0}, \frac{\vec{x}}{1 - \theta x_0}\right) - \theta \vec{x}\right].$$

(4.5.70)

For example, applying (4.5.70) to a trivial constant solution of Equations (4.5.1)

$$\rho = c_0, \qquad \vec{u} = \bar{c}$$

we obtain as a result a solution which depends on all variables

$$\rho = c_0 (1 - \theta x_0)^{-3},$$

$$\vec{u} = (\vec{c} - \theta \vec{x}) (1 - \theta x_0)^{-1}.$$
(4.5.71)

Let us present some more solutions of Equations (4.5.44) obtained with the help of formulae (4.5.70)

$$\rho = c_0 (1 - \theta x_0)^{-3},$$

$$\vec{u} = (1 - \theta x_0)^{-1} \left[\vec{F} \left(\frac{x_0 - \vec{\alpha} \vec{x}}{1 - \theta x_0} \right) - \theta \vec{x} \right].$$
(4.5.72)

Here \vec{F} are the same as in (4.5.68).

4. Now we shall study symmetry of gas dynamics equations describing isochoric process (that is, the process taking place without change in volume (density)) with constant pressure. Equations in question are

$$\vec{u}_0 + (\vec{u}\,\vec{\nabla})\vec{u} = 0,$$

div $\vec{u} = 0$ (4.5.73)

$$(\vec{u} = \vec{u}(x) = \{u^1, u^2, u^3\}, \qquad x = \{x_0, x_1, x_2, x_3\})$$

or
 $\vec{u}_0 + (\vec{u} \, \vec{\nabla}) \vec{u} = 0.$ (4.5.74)

Symmetry of Equation (4.5.74) was investigated in [174] and it was established the maximal in the class of linear representations 24-parameter invariance group $G = \{IGL(4, R), C(4)\}$ (IGL(4,R) is a 20-parameter group of general inhomogeneous linear transformations in R(4); C(4) is a 4-parameter Abelian group of pure conformal transformations in R(4)), the infinitesimal transformations having the form

$$x'_{\mu} = x_{\mu} + \epsilon \xi^{\mu}(x) + O(\epsilon^{2}),$$

$$u^{a'}(x') = u^{a}(x) + \epsilon \eta^{a}(x, u) + O(\epsilon^{2}),$$
(4.5.75)

where

$$\xi^{\mu} = (\vec{\alpha} \cdot \vec{x}) x_{\mu} + \beta_{\mu\nu} x_{\nu} + \gamma_{\mu}, \qquad (4.5.76)$$

$$\eta^{a} = \beta_{a0} + \beta_{ab} u^{b} + (\alpha_{0} + \alpha_{b} u^{b}) x_{a} - [(\alpha_{0} + \alpha_{b} u^{b}) x_{0} + \beta_{00} + \beta_{ab} u^{b}] u^{a};$$

where α_a , $\beta_{\mu\nu}$, γ_{μ} are arbitrary constants.

Note, the most general form of IFO admitted by Equation (4.5.74) is

$$X = \xi^{\mu}(x, u)\partial_{\mu} + \eta^{a}(x, u)\partial_{u^{a}}$$

$$(4.5.77)$$

which includes linear representation $(\xi^{\mu}(x, u) = \xi^{\mu}(x), \eta^{a}(x, u))$ are linear functions of u) as a particular case. It turns out that the Lie-maximal invariance group of Equation (4.5.74) is infinite-dimensional [91].

Lemma 4.5.1. The maximal invariance group of Equation (4.5.74) is infinitedimensional, the basis elements of corresponding Lie algebra having the form (4.5.77) provided

$$\xi^{0} = \xi^{0}(x, u),$$

$$\xi^{a} = u^{a}\xi^{0} + F^{a}(\omega, u) + x_{0}G^{a}(\omega, u),$$

$$\eta^{a} = G^{a}(\omega, u),$$

(4.5.78)

where $\omega = \{\omega^a = x_a - x_0 u_a\}; \xi^0, F^a, G^a$ are arbitrary differentiable functions.

Proof. Applying Lie algorithm to the Equation (4.5.74) one obtains the following defining equations for coordinates of IFO (4.5.77):

$$\eta^{a} = (\partial_{0} + \vec{u} \cdot \vec{\nabla})(\xi^{a} - u^{a}\xi^{0}),$$

$$(\partial_{0} + \vec{u} \cdot \vec{\nabla})\eta^{a} = 0,$$
(4.5.78')

the general solution of which is given by (4.5.78). The Lemma is proved.

The symmetry properties of Equations (4.5.73) are summarized in the following statement.

Theorem 4.5.2. The maximal IA of Equations (4.5.73) is AIGL(4,R), basis elements having the form

$$\partial_{\mu} = \frac{\partial}{\partial x_{\mu}}, \qquad L_{ab} = x_{b}\partial_{a} + u^{b}\partial_{u^{a}}, \qquad a \neq b,$$

$$L_{0a} = x_{a}\partial_{0} - u^{a}u^{b}\partial_{u^{b}}, \qquad G_{a0} = x_{0}\partial_{a} + \partial_{u^{a}},$$

$$\mathcal{D}_{0} = x_{0}\partial_{0} - u^{b}\partial_{u^{b}}, \qquad \mathcal{D}_{a} = x_{a}\partial_{a} + u^{a}\partial_{u^{a}} \qquad \text{(no sum over } a\text{)}.$$

$$(4.5.79)$$

Proof. As a result of applying Lie condition of invariance (3), one obtains defining equations for coordinates of IFO (4.5.77) which include (4.5.78') and equations

div
$$\vec{\eta} = 0, \qquad \frac{\partial \xi^{\mu}}{\partial u^a} = 0; \qquad \mu = \overline{0,3}$$
 (4.5.80)

The general solution of these equation is

$$\xi^{\mu} = \beta_{\mu\nu} x_{\nu} + \gamma_{\mu}; \qquad \mu = \overline{0,3}; \eta^{a} = \beta_{a0} + \beta_{ab} u^{b} - (\beta_{00} + \beta_{ab} u^{b}) u^{a}$$
(4.5.81)

which defines AIGL(4,R) (4.5.79). The theorem is proved.

Remark 4.5.2. Algebra AIGL(4,R) (4.5.79) contains as subalgebras AP(1,3) and AG(1,3). This means that Equations (4.5.73) describe relativistic as well as nonrelativistic processes. It is interesting to note that generators of Lorentz boosts J_{0a} are realized in nonlinear representation

$$J_{0a} = x_0 \partial_a + x_a \partial_0 - u^a u^b \partial_{u^b} + \partial_{u^a}.$$

$$(4.5.82)$$

The final transformations generated by operators (4.5.82) have the form

$$\begin{aligned} x'_{0} &= \gamma(x_{0} + \vec{x} \cdot \vec{v}), \\ \vec{x}' &= \vec{x} + \frac{\gamma - 1}{v^{2}} (\vec{x} \cdot \vec{v}) \vec{v} + \gamma \vec{v} x_{0}, \\ \vec{u}' &= \frac{\vec{u} + \vec{v} \left[(\gamma - 1) \frac{\vec{u} \cdot \vec{v}}{v^{2}} - \gamma \right]}{\gamma(1 - \vec{u} \cdot \vec{v})}, \end{aligned}$$
(4.5.84)

where $\vec{v} = \{v_1, v_2, v_3\}$ are arbitrary constants, $v^2 = \vec{v} \cdot \vec{v}$, and $\gamma = (1 - v^2)^{-1/2}$. Though the transformations (4.5.84) are nonlinear in u, they have a clear interpretation. Formula (4.5.84) is the well-known relativistic-velocity summation.

5. Here we obtain exact solution of Equations (4.5.73), (4.5.74) by making use of their symmetry.

Consider Equation (4.5.74). Since it is invariant under G(1,3), we look for its solutions in the form

$$\vec{u}(x) = \vec{M}(x_0)\varphi^1(\omega) + \vec{N}(x_0)\varphi^2(\omega) + \vec{L}(x_0)\varphi^3(\omega) + \vec{B}(x_0)$$
(4.5.85)

which is a particular case of the ansatz (4.5.45). It is obvious that the reduced equations following as a result of substitution (4.5.85) into (4.5.74) coincide with (4.5.46)-(4.5.53) if to put in these latter $\lambda = 0$ and omit the last equation in every one of them.

Consider some such integrable reduced equations. Putting in (4.5.46) $\vec{\varphi} = \vec{\varphi}(\omega_1)$ we get

$$(2\varphi^1 - \omega_1)\vec{\varphi_1} - \vec{\varphi} + 2\vec{\psi_1} = 0 \tag{4.5.86}$$

The general solution of Equation (4.5.86) has the form

$$\varphi^{1} = \frac{\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{1}}}{2},$$

$$\varphi^{2} = c_{2} \Big(\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{1}} \Big) \sin \ln \Big[c_{3} \Big(\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{1}} \Big) \Big],$$

$$\varphi^{3} = c_{2} \Big(\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{1}} \Big) \cos \ln \Big[c_{3} \Big(\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{1}} \Big) \Big]$$

(4.5.87)

where c_1, c_2, c_3 are integration constants. Formula (4.5.87) together with (4.5.85) give a partial solution to the Equation (4.5.74)

$$\vec{u} = \frac{F(x)}{\sqrt{x_0}} \left[\frac{\vec{\alpha}}{2} + c_2 \vec{\beta} \sin \ln \left(c_3 F(x) \right) + c_2 \vec{\gamma} \cos \ln \left(c_3 F(x) \right) \right] + \vec{v}, \qquad (4.5.88)$$

where $F(x) = x_0^{-1/2} \left(\vec{\alpha} \vec{x} + bx_0 \pm \sqrt{(\vec{\alpha} \vec{x} + bx_0)^2 + c_1 x_0} \right)$. Let function $(\vec{a} \text{ of Equation} (4.5.47)$ depends on (a) on

Let function $\vec{\varphi}$ of Equation (4.5.47) depends on ω_1 only. Then we have the ODE

$$(2\varphi^3 - \omega_1)\vec{\varphi}_1 - \vec{\varphi} + 2\vec{\psi}_2 = 0 \tag{4.5.89}$$

After integrating it we get

$$\varphi^{1} = \frac{1}{2} \left(\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{3}} \right) \left[\ln^{2} c_{2} \left(\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{3}} \right) + c_{1} \right],
\varphi^{2} = -a \left(\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{3}} \right) \ln c_{2} \left(\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{3}} \right),$$

$$\varphi^{3} = \frac{\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{3}}}{2},$$
(4.5.90)

where c_1, c_2, c_3 are integration constants. From (4.5.90) and (4.5.85) we find a solution of Equation (4.5.74)

$$\vec{u} = \frac{M(x)}{2\sqrt{x_0}} \Big[\vec{\alpha} a^2 (\ln^2 c_2 M(x) + c_1) - 2a\vec{\beta} \ln c_2 M(x) + \vec{\gamma} \Big] + \vec{v}, \qquad (4.5.91)$$

where $M(x) = \frac{\vec{\alpha}\vec{x} + kx_0}{\sqrt{x_0}} \pm \sqrt{\frac{(\vec{\alpha}\vec{x} + kx_0)^2}{x_0} + c_3}.$

Let function $\vec{\varphi}$ of Equation (4.5.48) not depend on ω_2, ω_3 . Then we have the ODE

$$(2\varphi^1 + \omega_1)\vec{\varphi} + \vec{\psi}_3 = 0. \tag{4.5.92}$$

The general solution of (4.5.92) has the form

$$\varphi^{1} = \frac{-\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{1}}}{2},$$

$$\varphi^{2} = c_{2} \left(-\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{1}}\right)^{-2a_{+}},$$

$$\varphi^{3} = c_{3} \left(-\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{1}}\right)^{-2a_{-}},$$

(4.5.93)

where c_1, c_2, c_3 are integration constants. Formulae (4.5.93), (4.5.85) give a solution to the Equation (4.5.74)

$$\vec{u} = \frac{\vec{\alpha}}{2} F(x) + c_2 \vec{\beta} [F(x)]^{-2a_+} + c_3 \vec{\gamma} [F(x)]^{-2a_-} + \vec{v},$$

$$F(x) = x_0^{-1} \left(-\vec{\alpha} \vec{x} - bx_0 \pm \sqrt{(\vec{\alpha} \vec{x} + bx_0)^2 + c_1 x_0} \right)$$
(4.5.94)

Putting $\vec{\varphi} = \vec{\varphi}(\omega_3)$ we get from (4.5.48)

$$(\varphi^3 - a_+\omega_2)\vec{\varphi}_2 - \frac{1}{2}\vec{\psi}_3 = 0 \tag{4.5.95}$$

which has a solution

$$\begin{aligned}
\varphi^1 &= c_1(\varphi^3)^{1/2a_-}, \\
\varphi^2 &= c_2(\varphi^3)^{1/2a_+},
\end{aligned}$$
(4.5.96)

function φ^3 being a solution of functional equation

$$\omega_2(\varphi^3)^{a-} + c_3(\varphi^3)^{a+} = 1. \tag{4.5.97}$$

The corresponding solution of Equation (4.5.74) can be written as

$$\vec{u} = \frac{c_1 \vec{\alpha}}{\sqrt{x_0}} (\varphi^3)^{1/2a_-} + c_2 \vec{\beta} x_0^{-a_+} (\varphi^3)^{a_+/a_-} + \vec{\gamma} x_0^{-a_-} \varphi^3 + \vec{v}.$$
(4.5.98)

Let function $\vec{\varphi}$ of Equation (4.5.49) depend only on ω_1 . Then we have the ODE

$$(2\varphi^1 + \omega_1)\vec{\varphi_1} + \vec{\psi_4} = 0$$

the general solution of which having the form

$$\varphi^{1} = \frac{1}{2} \left(-\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{1}} \right),
\varphi^{2} = c_{2} \left(-\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{1}} \right)^{2},
\varphi^{3} = 2b_{4} \ln \left(-\omega_{1} \pm \sqrt{\omega_{1}^{2} + c_{1}} \right) + c_{3},$$
(4.5.99)

where c_1 , c_2 , c_3 are integration constants. Formulae (4.5.99), (4.5.85) give a solution to the Equation (4.5.74)

$$\vec{u} = \vec{\alpha} \left(\frac{M(x)}{2\sqrt{x_0}} + b_1 \right) + \vec{\gamma} \left(c_2 \frac{M^2(x)}{x_0} - b_2 \right) + \vec{\beta} \left(2b_4 \ln M(x) - b_4 \ln x_0 \right) (4.5.100)$$

where M(x) is the same as in (4.5.94).

Let function $\vec{\varphi}$ of Equation (4.5.49) depends only on ω_2 . Then we have the ODE

$$(\varphi^3 + b_3)\vec{\varphi}_2 - \frac{1}{2}\vec{\psi}_4 = 0$$

which is satisfied by

$$\varphi^{1} = c_{1}\sqrt{\varphi^{2}},$$

$$\varphi^{3} = b_{4}\ln\varphi^{2} + c_{3},$$
(4.5.101)

provided function φ^2 is a solution of functional equation

$$b_3 \ln \varphi^2 + \varphi^2 = \omega_2 + c_2 \tag{4.5.102}$$

The corresponding solution of Equation (4.5.74) can be written as

$$\vec{u} = \vec{\alpha} \left(\frac{c_1}{x_0}\sqrt{\varphi^2} + b_1\right) + \vec{\beta} \left(b_4 \ln \frac{\varphi^2}{x_0} + c_3\right) + \vec{\gamma} \left(\frac{\varphi^2}{x_0} - b_2\right).$$
(4.5.103)

Let function $\vec{\varphi}$ of Equation (4.5.50) depend only on ω_1 . Then we have the ODE

$$(\varphi^1 + bc)\vec{\varphi}_1 + \vec{\psi}_5 = 0$$

which has solution

$$\begin{split} \varphi^1 &= \pm \frac{1}{2}\sqrt{4a\omega_1 + c_1} - bc, \\ \varphi^2 &= c_2 \sin\left(\pm \frac{b}{2a}\sqrt{4a\omega_1 + c_1} + c_3\right), \\ \varphi^3 &= c_2 \cos\left(\pm \frac{b}{2a}\sqrt{4a\omega_1 + c_1} + c_3\right). \end{split}$$

Thereby we find one more solution to the Equation (4.5.74)

$$\vec{u} = \vec{\alpha} (M(x) - 2ax_0 - bc) + \vec{\beta}c_2 \sin\left(\frac{b}{2a}M(x) + c_3\right) + (4.5.104) + \vec{\gamma}c_2 \cos\left(\frac{b}{2a}M(x) + c_3\right),$$

where $M(x) = \pm \left[4a(ax_0^2 + bcx_0 + \vec{\alpha}\vec{x}) + c_1 \right]^{1/2}$.

Let function $\vec{\varphi}$ of Equation (4.5.51) depends only on ω_1 . Then we have the ODE

$$(-\varphi^1 + bc)\vec{\varphi}_1 + \vec{\psi}_6 = 0$$

which is satisfied by

$$\varphi^{1} = 2\sqrt{a}\sqrt{\omega_{1} + c_{1}} - bc,$$

$$\varphi^{2} = c_{2} \exp\left(\frac{b}{\sqrt{a}}\sqrt{\omega_{1} + c_{1}}\right),$$

$$\varphi^{3} = c_{2} \exp\left(-\frac{b}{\sqrt{a}}\sqrt{\omega_{1} + c_{1}}\right),$$

(4.5.105)

Formulae (4.5.105) and (4.5.85) give a solution to the Equation (4.5.74)

$$\vec{u} = \vec{\alpha} (M(x) + 2ax_0 - bc) + \vec{\beta}c_2 \exp\left[-\frac{b}{\sqrt{a}}M(x) + bx_0\right] + \vec{\gamma}c_2 \exp\left[\frac{b}{\sqrt{a}}M(x) - bx_0\right] + \vec{v},$$
(4.5.106)

where $M(x) = (ax_0^2 + bcx_0 + \vec{\alpha}\vec{x} + c_1)^{1/2}$. Let function $\vec{\varphi}$ of Equation (4.5.52) depends only on ω_1 . Then we have the ODE

$$(1 - \vec{\alpha}\vec{\varphi})\vec{\varphi}_1 = 0$$

the general solution of which has the form

$$\vec{\varphi} = \vec{f}(\omega_1), \tag{4.5.107}$$

where \vec{f} are arbitrary differentiable functions, such as $\vec{\alpha}\vec{f} = 1$. It follows a solution to the Equation (4.5.74)

$$\vec{u} = \vec{f}(x_0 - \vec{\alpha}\vec{x}). \tag{4.5.108}$$

Let function $\vec{\varphi}$ of Equation (4.5.53) depends only on ω_1 . Then we have the ODE

$$(\omega_1 + \vec{\alpha}\vec{\varphi})\vec{\varphi}_1 = -\frac{1}{2}\vec{\varphi}$$

which has a partial solution

$$\vec{\varphi} = \vec{c}F(\omega_1) \tag{4.5.109}$$

provided $F(\omega_1)$ satisfies the condition

$$F^2(\vec{c}\vec{\alpha}F + \frac{3}{2}\omega_1) = 1.$$

It follows a solution to the Equation (4.5.74)

$$\vec{u} = \frac{\vec{c}}{\sqrt{x_0}} F(x)$$
 (4.5.110)

where function F(x) satisfies functional equation

$$F^{-2} - \vec{\alpha}\vec{c}F = \frac{3}{2}\frac{\vec{\alpha}\vec{x}}{\sqrt{x_0}}.$$
(4.5.111)

Remark 4.5.3. Looking for solutions of Equation (4.5.74) by means of the ansatz

$$\vec{u} = \vec{x}\varphi(\omega) \tag{4.5.112}$$

we get the reduced equation as follows

$$\left(\partial_0\omega + \left[(\vec{x}\,\vec{\nabla})\omega\right]\varphi\right)\varphi' + \varphi^2 = 0 \tag{4.5.113}$$

To require Equation (4.5.113) to be an ODE, one has to put

$$\partial_0 \omega = A(\omega),$$

($\vec{x} \, \vec{\nabla}) \omega = B(\omega).$ (4.5.114)

For example, if

$$\omega = \vec{\alpha} \vec{x} e^{\lambda x_0} \quad \text{or} \quad \omega = \sqrt{\vec{x}^2} e^{\lambda x_0},$$
(4.5.115)

Equations (4.5.113) take the form

$$\omega\varphi'(\varphi+\lambda)+\varphi^2=0. \tag{4.5.116}$$

This latter equation has the general solution

$$\frac{1}{\varphi}e^{\lambda/\varphi} = c_1\omega. \tag{4.5.117}$$

Let us write down some formulae of generating solutions of Equation (4.5.74). Let

$$\vec{u} = \vec{F}(x) \tag{4.5.118}$$

be a solution of (4.5.74). Then using transformations (4.5.83), (4.5.84) and formulae (17), we find new solutions of Equation (4.5.74):

$$\vec{u} = \frac{1 - \vec{v}^2}{1 - \vec{v}\vec{F}(x')}\vec{F}(x') - \vec{v}$$
(4.5.119)

where x' is determined in (4.5.83); analogously, using conformal transformations

$$x' = \frac{x}{\sigma}, \qquad \vec{u}' = \frac{\sigma \vec{u} + (\alpha_0 + \vec{\alpha} \vec{u}) \vec{x}}{\sigma + (\alpha_0 + \vec{\alpha} \vec{u}) x_0}, \qquad \sigma = 1 - \alpha_0 x_0 + \vec{\alpha} \vec{x}$$
(4.5.120)

we find new solutions as

$$\vec{u} = \frac{\vec{F}\left(\frac{x}{\sigma}\right) - \left[\alpha_0 + \vec{\alpha}\vec{F}\left(\frac{x}{\sigma}\right)\right]\vec{x}}{1 - \left[\alpha_0 + \vec{\alpha}\vec{F}\left(\frac{x}{\sigma}\right)\right]x_0}.$$
(4.5.121)

Now, without going into details, we list some solutions of the system (4.5.73) obtained in the same manner as solutions of Equation (4.5.74):

$$\vec{u} = x_0^{-1/2} \exp\left[-M(x)\right] \left[\vec{\beta} \left(c_2 \cos M(x) - c_1 \sin M(x)\right) + \vec{\gamma} \left(c_1 \cos M(x) + c_2 \sin M(x)\right)\right] - \vec{v},$$

$$\begin{split} M(x) &= \ln \frac{\vec{\alpha}\vec{x} + bx_0}{\sqrt{x_0}}; \\ \vec{u} &= \frac{c_2}{B(x)} \left[\vec{\alpha} \left(a \ln \frac{B(x)}{\sqrt{x_0}} + c_1 \right) + \vec{\beta} \right] + \vec{v}, \\ B(x) &= \vec{\alpha}\vec{x} - x_0\vec{\alpha}\vec{A}(x_0); \\ \vec{u} &= c_2\vec{\beta}(\vec{\alpha}\vec{x} + bx_0)^{-2a} + c_3\vec{\gamma}(\vec{\alpha}\vec{x} + bx_0)^{-2a-} + \vec{v}; \\ \vec{u} &= \vec{\alpha}b_1 - \vec{\beta}(2b_4\ln(\vec{\alpha}\vec{x} + b_1x_0) + c_3) + \vec{\gamma}\left(\frac{c_2}{\vec{\alpha}\vec{x} + b_1x_0} - b_2x_0\right); \\ \vec{u} &= \vec{\alpha}(c_1\exp\frac{\vec{\beta}\vec{x} + b_2x_0}{2b_3} - b_1x_0^{-1/2}) + \vec{\beta}\frac{b_4}{b_3}(\vec{\beta}\vec{x} + b_2x_0 + c_3) - \vec{\gamma}b_2; \\ \vec{u} &= \vec{\alpha}c_1 + \vec{\beta}c_2\left(c_3 - \frac{2c_2}{c_1}\vec{\alpha}\vec{x}\right)^{-1/2} \pm \vec{\gamma}\left(c_3 - \frac{2c_2}{c_1}\vec{\alpha}\vec{x}\right)^{1/2} + \vec{v}; \\ \vec{u} &= \vec{f}(x_0 - \vec{\alpha}\vec{x}). \end{split}$$

Note that in this case of generating solutions formula (4.5.119) also holds.

Remark 4.5.4. Let

$$\vec{u} = \vec{\nabla}v \tag{4.5.122}$$

where v is scalar real function. Substitution of (4.5.122) into Equation (4.5.74) gives rise to the Hamilton-Jacobi equation (3.6.1) with m = 1:

$$v_0 + \frac{1}{2} \left(\vec{\nabla} v \right)^2 = 0.$$

Hence, one can use solutions of the Hamilton-Jacobi equation obtained in Paragraph 3.6 to construct via (4.5.122) solutions of Equation (4.5.74). Below we list some of such solutions

$$\vec{u} = \frac{\vec{\alpha}}{2x_0} \left(\vec{\alpha} \vec{x} - bx_0 \pm \sqrt{(\vec{\alpha} \vec{x} + bx_0)^2 - 8ax_0} \right), \qquad \vec{\alpha}^2 = 1;$$

$$\vec{u} = -\vec{\alpha} \left(2ax_0 + b \pm \sqrt{(\vec{\alpha} \vec{x} + ax_0^2 + bx_0 + c)} \right), \qquad \vec{\alpha}^2 = 1;$$

$$\vec{u} = (1 - \theta x_0)^{-1} \left[\vec{\alpha} F \left(\frac{\vec{\alpha} \vec{x}}{1 - \theta x_0} \right) - \theta \vec{x} \right], \qquad \vec{\alpha}^2 = 0,$$

where θ , $\vec{\alpha}$, a, b are arbitrary constants, F is an arbitrary differentiable function.

In conclusion, it will be noted that since equations considered above are translationally invariant all solutions obtained here can be multiplied by making change $x_{\mu} \rightarrow x_{\mu} + \delta_{\mu}$ (δ_{μ} are constants).

4.6. The Galilean invariant generalization of the Lame equations. The superalgebra of symmetry of the Lame equations

The Lame equations

$$L\vec{u} \equiv \frac{\partial^2 \vec{u}}{\partial x_0^2} - \alpha \operatorname{grad}(\operatorname{div} \vec{u}) - \Delta \vec{u} = 0$$
(4.6.1)

where $\vec{u} = \vec{u}(x) = \{u^1, u^2, u^3\}$ is a displacement vector, $x = (x_0, \vec{x}) \in \mathbb{R}^4$, α is a constant, are basic equations of elastodynamics and describe propagation of waves in elastic and isotropic medium. In spite of the fact that Equations (4.6.1) are quite satisfactory from the physical point of view (at least for small-amplitude oscillations) [141] the rightfulness of their use from the group-theoretic point of view gives rise to grave doubts [63]. The point is that the Lame equations do not satisfy any principle of invariance, neither Galilean nor Lorentz. As it was shown in [39] the Lie-maximal invariance algebra of Equation (4.6.1) is $A\widetilde{E}(1,3)$, basis elements having the form

$$P_{0} = \frac{\partial}{\partial x_{0}}, \qquad P_{a} = \frac{\partial}{\partial x_{a}}, \qquad D = x_{0}P_{0} + x_{a}P_{a}$$

$$J_{ab} = x_{a}P_{b} - x_{b}P_{a} + u^{a}\frac{\partial}{\partial u^{b}} - u^{b}\frac{\partial}{\partial u^{a}}$$

$$(4.6.2)$$

In book [10] this symmetry has been used for obtaining exact solutions of Equation (4.6.1). It will be noted that in [73] it is shown (with the help of non-Lie method [57]) that Lame equations (4.6.1) are invariant both under

 $AC(1,3) \supset A\widetilde{P}(1,3)$ and AG(1,3); however, the basis elements of these algebras are pseudodifferential operators.

Following [104] we generalize the Lame equations (4.6.1) so that they will be invariant under the point Galilean transformations

$$\vec{x}' = \vec{x} + \vec{v}_0, \qquad x'_0 = x_0;$$

 $\vec{u}'(x') = \vec{u}(x) + \vec{v}, \qquad \vec{v} = \text{const.}$ (4.6.3)

Theorem 4.6.1. [104]. Equation

$$L\vec{u} + F(\vec{u}, \vec{u}, \vec{u}) = 0, \qquad \vec{u} = \left\{\frac{\partial u^a}{\partial x_b}\right\}, \qquad \vec{u} = \left\{\frac{\partial^2 u^a}{\partial x_b \partial x_c}\right\}$$
(4.6.4)

is invariant under the Galilean transformations (4.6.3) iff

$$F(\vec{u}, \vec{u}, \vec{u}) = 2(\vec{u} \cdot \vec{\nabla})\vec{u}_0 - \left[\vec{u}(\vec{u} \cdot \vec{\nabla})\vec{\nabla}\right]\vec{u}$$
(4.6.5)

Proof. The transformations (4.6.3) are generated by the operator

$$G = v_a G_a = x_0 \vec{v} \cdot \vec{\nabla} + v_a \partial_{u^a}. \tag{4.6.6}$$

According to (5.3.8) we find

$$\vec{\nabla}' = \exp\{x_0 \vec{v} \cdot \vec{\nabla}\} \vec{\nabla} \exp\{-x_0 \vec{v} \cdot \vec{\nabla}\} = \vec{\nabla},$$

$$\partial' = \exp\{x_0 \vec{v} \cdot \vec{\nabla}\} \partial_0 \exp\{-x_0 \vec{v} \cdot \vec{\nabla}\} = \partial_0 - \vec{v} \cdot \vec{\nabla},$$

$$(\partial'_0)^2 = \partial_0^2 - 2(\vec{v} \cdot \vec{\nabla}) \partial_0 + (\vec{v} \cdot \vec{\nabla})^2$$
(4.6.7)

Hence

$$L(\partial')\vec{u}'(x') = L\vec{u} - 2(\vec{v}\cdot\vec{\nabla})\vec{u}_0 + (\vec{v}\cdot\vec{\nabla})^2\vec{u}$$
(4.6.8)

So, to ensure the invariance of Equation (4.6.4) it is necessary and sufficient to require

$$F' = F + 2(\vec{v} \cdot \vec{\nabla})\vec{u}_0 - (\vec{v} \cdot \vec{\nabla})^2 \vec{u}$$
(4.6.9)

whence we obtain (4.6.5). The theorem is proved.

Remark 4.6.1. One can make sure applying Lie's method that the maximal IA of Equation (4.6.4), (4.6.5) is AG(1,3) with basis elements (4.6.2), (4.6.6).

Remark 4.6.2. There is one more realization of AG(1,3), namely when

$$G_a = x_0 \partial_a + \omega^a \omega^b \partial_{\omega^b} \tag{4.6.10}$$

These operators generate the following transformations

$$ec{x}' = ec{x} + ec{v}x_0, \qquad x_0' = x_0$$

 $ec{\omega}' = rac{ec{\omega}}{1 - ec{v}ec{\omega}}.$

However, by virtue of the substitution

$$\vec{\omega} = \frac{\vec{c}}{1 - \vec{c}\vec{u}}, \qquad \vec{c} = \text{const}$$

these transformations and operators (4.6.10) are reduced to (4.6.3) and (4.6.6).

It will be noted that Lame equations are closely connected with the wave equations

$$\frac{\partial^2 \varphi}{\partial x_0^2} = (\alpha + 1) \Delta \varphi, \qquad \frac{\partial^2 \vec{\psi}}{\partial x_0^2} = \Delta \vec{\psi}. \tag{4.6.11}$$

It is easy to verify that

$$\vec{u} = \vec{\nabla}\varphi + \operatorname{rot}\vec{\psi} \tag{4.6.12}$$

is a solution of Lame equations (4.6.1) for any φ and $\vec{\psi}$ satisfying (4.6.11). An infinite sequence of solutions of the wave equation is given in (2.3.60), (2.3.61).

2. If we seek for symmetry operators of the Lame equation (4.6.1) among the second-order differential operators with matrix coefficients

$$Q = A^{\mu\nu}(x)\partial_{\mu}\partial_{\nu} + B^{\mu}(x)\partial_{\mu} + c(x), \qquad (4.6.13)$$

then we will have:

Theorem 4.6.2. [70*] The Lame equations (4.6.1) admit 62 operators of the type (4.6.13) from which nine operators

$$N = (\vec{S}\vec{J})^2 - (\vec{S}\vec{J}),$$

$$N_a = [P_a, N], \qquad N_{ab} = [P_b, N_a]$$
(4.6.14)

where $\vec{J} = \vec{x} \times \vec{P} + \vec{S}$; matrices S_a are written in (4.2.21), $\vec{P} = -i\vec{\nabla}$, do not belong to the enveloping algebra of $A\tilde{E}(1,3)$ (4.6.2).

Theorem 4.6.3. The Lame equations are invariant with respect to the superalgebra SQM(3,3), basis elements having the form

$$Q_{1} = (\vec{S} \cdot \vec{P})^{2}, \qquad Q_{2} = N^{2}, \qquad Q_{3} = Q_{1}Q_{2}$$

$$R_{1} = (\vec{S} \cdot \vec{P})^{2}, \qquad R_{2} = N, \qquad R_{3} = R_{1}R_{2}$$
(4.6.15)

and satisfying relations

$$[Q_a, Q_b] = [Q_a, R_b] = 0,$$

$$[R_a, R_b]_+ \equiv R_a R_b + R_b R_a = \delta_{ab} R_b \text{ (no sum on b)}.$$
(4.6.16)

The proof of these theorems is based on the invariance condition in the form

$$[L,Q] \equiv [L,A^{\mu\nu}\partial_{\mu}\partial_{\nu} + B^{\mu}\partial_{\mu} + C] = \left(\tilde{A}^{\mu\nu}\partial_{\mu}\partial_{\nu} + \tilde{B}\partial_{\mu} + \tilde{C}\right)L$$
(4.6.17)

where $\widetilde{A}^{\mu\nu}$, \widetilde{B}^{μ} , \widetilde{C} are 3×3 matrices depending on x. The truth of formulae (4.6.16) one can verify straightforwardly.

Now we present final transformations generated by operators (4.6.15) calculated according to §5.3 and Appendix 3.5. Using the identities

$$(\vec{S} \cdot \vec{P})\vec{u} = \operatorname{rot} \vec{u},$$

$$(\vec{S} \cdot \vec{P})^{2}\vec{u} = \operatorname{rot} \operatorname{rot} \vec{u} = \vec{\nabla} \operatorname{div} \vec{u} - \Delta \vec{u} \stackrel{\text{def}}{=} \vec{v},$$

$$(\vec{S} \cdot \vec{P})^{2n}\vec{u} = (-1)^{n+1}\Delta^{n-1}\vec{v}, \qquad n = 1, 2, \dots$$

$$(\vec{S} \cdot \vec{P})^{2n+1}\vec{u} = (-1)^{n}\Delta^{n} \operatorname{rot} \vec{u}, \qquad n = 0, 1, 2, \dots$$
(4.6.18)

one can easy obtain

$$Q_{1}: \quad \vec{u}'(x') = \exp\{\theta(\vec{S} \cdot \vec{p})^{2}\}\vec{u} = \sum_{n=0}^{\infty} \frac{\theta^{n}}{n!}(\vec{S} \cdot \vec{p})^{2n}\vec{u} =$$

$$= \vec{u} - \sum_{n=1}^{\infty} (-1)^{n} \frac{\theta^{n}}{n!} \Delta^{n-1}\vec{v} = \vec{u} - \Delta\left(\sum_{n=0}^{\infty} \frac{(-\theta)^{n}}{n!} \Delta^{n} - 1\right)\vec{v} =$$

$$= \vec{u} - \Delta^{-1} \left(e^{-\theta\Delta} - 1\right)\vec{v} =$$

$$= e^{-\theta\Delta}\vec{u} + \left(1 - e^{-\theta\Delta}\right) \Delta^{-1} \operatorname{grad} \operatorname{div} \vec{u} \qquad (4.6.19)$$

$$x' = x$$

 $R_1: \quad \vec{u}'(x') = \vec{u}(x) + \beta \operatorname{rot} \vec{u}$

 θ is a usual real parameter; β is a Grassmann parameter.

Using the identities

$$(\vec{S} \cdot \vec{J})\vec{u} = \vec{u} + \vec{\nabla}(\vec{x} \cdot \vec{u}) - \vec{x}\operatorname{div}\vec{u} \stackrel{\text{def}}{=} \vec{w}$$

$$(\vec{S} \cdot \vec{J})^{2}\vec{u} = (\vec{S} \cdot \vec{J})\vec{w} = \vec{w} + \vec{\nabla}(\vec{x} \cdot \vec{w}) - \vec{x}\operatorname{div}\vec{w}$$

(4.6.20)

we find transformations generated by the odd operator R_2 :

$$u^{a'}(x') = u^a(x) + \beta_1 \left[2u^a + (\vec{x} \cdot \vec{\nabla})u^a + 3\vec{x} \cdot \vec{u}_a + \vec{x}(\vec{x} \cdot \vec{\nabla})\vec{u}_a - \vec{x}^2 \operatorname{div} \vec{u}_a + x_a \left((\vec{x} \cdot \vec{\nabla})\operatorname{div} \vec{u} - \vec{x} \cdot \Delta \vec{u} - 2\operatorname{div} \vec{u} \right) \right]$$

 β_1 is a Grassmann parameter.

4.7. Reduction and exact solutions of nonlinear Galilei-invariant spinor equations

Following $[73^*]$ we consider the nonlinear spinor equation of the form

$$[-i(\gamma_0 + \gamma_4)\partial_t + i\gamma_a\partial_a + m(\gamma_0 - \gamma_4)]\psi = F(\psi^*, \psi), \qquad (4.7.1)$$

where $\psi = \psi(t, x_1, \ldots, x_3)$ is a four-component complex-valued function (column), $\partial_t = \partial/\partial t$, $\partial_a = \partial/\partial x_a$; $\gamma_0, \ldots, \gamma_3$ are Dirac matrices (2.1.2), $\gamma_4 = \gamma_0 \gamma_1 \gamma_2 \gamma_3$; F is a four-component function (column) depending on ψ^*, ψ .

It is known that Equation (4.7.1) under F = 0 is invariant with respect to the 11-dimensional Galilei algebra AG(1,3) [144,78] with basis elements

$$P_{0} = \partial_{t}, \quad P_{a} = \partial_{a}, \quad M = 2im,$$

$$G_{a} = t\partial_{a} + 2imx_{a} + \frac{1}{2}(\gamma_{0} + \gamma_{4})\gamma_{a},$$

$$J_{ab} = x_{a}\partial_{b} - x_{b}\partial_{a} - \frac{1}{2}\gamma_{a}\gamma_{b}.$$
(4.7.2)

Operators G_a generate the three-dimensional group of Galilei transformations

$$t = t', \quad x'_{a} = x_{a} + v_{a}t,$$

$$\psi'(x') = \exp\{-2im(v_{a}x_{a} + \frac{1}{2}v_{a}v_{a}t) - - \frac{1}{2}(\gamma_{0} + \gamma_{4})\gamma_{a}v_{a}\}\psi(x), \quad v_{a} = \text{const}, \quad x = \{t, x_{a}\}.$$
(4.7.3)

Let us require the nonlinear equation (4.7.1) to be invariant under Galilei transformations (4.7.3). (Non-Lie symmetry of Equation (4.7.1) under F = 0 is studied in [138*].)

Theorem 4.7.1. Equation (4.7.1) is invariant under AG(1,3) (4.7.2) iff

$$F = [f_1 + (\gamma_0 + \gamma_4)f_2]\psi$$
 (4.7.4)

where $f_i = f_i (\overline{\psi}\psi, \overline{\psi}(\gamma_0 + \gamma_4)\psi), \ \overline{\psi} = (\psi^*)^T \gamma_0.$

The proof can be obtained by application of standard Lie algorithm.

Remark 4.7.1. When m = 0, Equation (4.7.1) with nonlinearity (4.7.4) is invariant under the infinite-dimensional group of transformations

$$t' = t \qquad x'_{a} = x_{a} + h_{a}(t), \psi'(x') = \exp\{-\frac{1}{2}(\gamma_{0} + \gamma_{4})\gamma_{a}\dot{h}_{a}(t)\},$$
(4.7.5)

where $h_a(t)$ are arbitrary differentiable functions of t, $\dot{h} = dh/dt$.

To construct solutions of Equations (4.7.1), (4.7.4) we will act by analogy with that done in Paragraph 2.1, but in this case we use the three-dimensional subalgebras of AG(1,3) found in [71*]. Below in Table 4.7.1 we list G(1,3)inequivalent ansatze of codimension 1 obtained as solutions of equations

$$Q_a A(x) \equiv \left[\xi_a^0(x)\partial_t + \xi_a^b(x)\partial_b + \eta(x)\right]A = 0,$$

$$\left[\xi_a^0(x)\partial_t + \xi_a^b(x)\partial_b\right]\omega = 0,$$

$$\psi(x) = A(x)\varphi(\omega),$$

(4.7.6)

where Q_a , a = 1, 2, 3 are elements of a three-dimensional subalgebra of AG(1,3).

N	Algebra	Invar. var. ω	Ansatze $\psi(x) =$
1.	P_0, P_1, P_2	x_3	$= \varphi(\omega)$
2.	$P_1, P_2,$	x_3	$= \exp\{\frac{t}{2\alpha}\gamma_1\gamma_2\}\varphi(\omega)$
	$J_{12} + \alpha P_0$		
3.	$P_1, P_2,$ $P_2 + i \alpha m$	x_3	$=\exp\{-i\alpha mt\}\varphi(\omega)$
4.	$P_0, P_3,$	$x_1^2 + x_2^2$	$= \exp\left\{-\frac{1}{2}\gamma_1\gamma_2 \arctan \frac{x_1}{2}\right\} \varphi(\omega)$
	J_{12}		$= \left(\begin{array}{c} 2 \\ 2 \\ 1 \\ 2 \\ 2$
5.	$P_{1}, P_{2},$	$\beta t^2 - 2\alpha x_3$	$=\expiggl\{rac{2}{3}imlpha^{-2}eta t(eta t^2-3lpha x_3)-$
	$J_{12} + \alpha P_0 +$		$-lpha^{-1}eta t\eta_3+rac{t}{2lpha}\gamma_1\gamma_2\bigg\}arphi(\omega)$
	$+\beta G_3$		
6.	$P_1, P_2,$	t	$= \exp\left\{\frac{x_3}{2\alpha t}(\gamma_1\gamma_2 - 2\alpha\eta_3 - \right.$
	$J_{12} + \alpha G_3$		$-2i\alpha m x_3 \Big\} \varphi(\omega)$
7.	$G_1, G_2,$	t	$=\exp\left\{-rac{im}{t}(x_1^2+x_2^2)- ight\}$
	$J_{12} + \alpha G_3$		$\left(-\frac{1}{t}(\eta_1 x_1 + \eta_2 x_2)\right) \exp\left\{-\frac{x_3}{2\alpha t}\right)$
			$\left(\left(2i\alpha m x_3 + \alpha \eta_3 - \gamma_1 \gamma_2 \right) \right) \varphi(\omega)$
8.	$P_{1}, P_{2},$	t	$= \exp\left\{-\frac{x_3}{2\alpha}\gamma_1\gamma_2\right\}\varphi(\omega)$
	$J_{12} + \alpha P_3$		$\zeta_{2\alpha}$
9.	$G_1, G_2,$	t	$= \exp\left\{-\frac{im}{t}(x_1^2 + x_2^2) - \frac{1}{t}(\eta_1 x_1 + y_2^2)\right\}$
	$J_{12} + \alpha P_3$		$+\eta_2 x_2) \bigg\} \exp \left\{ -\frac{x_3}{2\alpha} \gamma_1 \gamma_2 \right\} \varphi(\omega)$

Table 4.7.1. G(1,3)-inequivalent ansatze of codimension 1 for spinor field

Ν	Algebra	Invar. var. ω	Ansatze $\psi(x) =$
10.	P_1, P_2, P_3	t	$= \varphi(\omega)$
11.	G_1, P_2, P_3	t	$\Big = \exp\Big\{-\frac{im}{t}x_1^2 - \frac{1}{t}x_1\eta_1\Big\}\varphi(\omega)$
12.	$G_1 + \alpha P_1,$	t	$=\exp\left\{-\frac{x_2}{t}(imx_2+\eta_2)+\right.$
	G_2, P_3		$+\frac{x_1}{\alpha-t}(imx_1+\eta_1)\Big\}\varphi(\omega)$
13.	$G_1 + \alpha P_1,$	t .	$=\exp\left\{-\frac{x_3}{t}(imx_3+\eta_3)+\frac{x_1}{\alpha-t}\right.$
	$G_2 + \beta P_2,$		$\left[\cdot (imx_1 + \eta_1) + \frac{x_2}{\beta - t}(imx_2 + \eta_2) \right] \varphi(\omega)$
	G_3		
14.	$G_1 + \alpha P_0,$	$t^2 - 2\alpha x_1$	$=\exp\Bigl\{\tfrac{2}{3}i\alpha^{-2}mt(t^2-3\alpha x_1)-$
	P_2, P_3		$\left(-\frac{t}{\alpha}\eta_{1}\right)\varphi(\omega)$
15.	$J_{12} + i\alpha m,$	$x_1^2 + x_2^2$	$= \exp\left\{ (i\alpha m - \frac{1}{2}\gamma_1\gamma_2) \arctan \frac{x_1}{x_2} \right\} \varphi(\omega)$
	P_0, P_3		~2
16.	$G_1 + \alpha P_2,$	t	$= \exp\left\{-\frac{x_1}{t}(imx_1 + \eta_1) - \right.$
	$G_2 + \alpha P_1 +$		$-im(\alpha x_1 + tx_2)^2 [t(t^2 - t\beta - \alpha^2)]^{-1} +$
	$+\beta P_2+$		$+rac{x_3}{ au t}(lpha\eta_1+t\eta_2)\Big\}\cdot$
	$+(\alpha\rho-\delta\beta)P_3,$		$\cdot \exp \left\{ imz^2 \left[f(t)(t^2 - t\beta - lpha^2) \right]^{-1} + \right.$
	$-G_3 + ho G_1 +$		$+\frac{z}{f(t)}\left[\alpha\left(\frac{1}{\tau}-\frac{\delta}{t^2}\right)\eta_1+\right]$
	$+\delta G_2 + \alpha \delta P_1$		$+\left(rac{t}{ au}-\delta ight)\eta_2+\eta_3 ight] ight\}arphi(\omega)$

Table 4.7.1. G(1,3)-inequivalent ansatze of codimension 1 for spinor field

In this table:

$$\begin{split} \eta_a &= \frac{1}{2} (\gamma_0 + \gamma_4) \gamma_a, \qquad \tau = \alpha \rho + \delta \beta \\ f(t) &= \tau [\alpha(\rho t - \alpha \delta) + \delta t^2] - t [t^2 - t\beta - \alpha^2], \\ z &= \tau (\alpha x_1 + t x_2) + [t(t - \beta) - \alpha^2] x_3, \end{split}$$

 $\alpha, \beta, \rho, \delta$ are arbitrary real constants.

Let us substitute ansatze from Table 4.7.1 into Equation (4.7.1), (4.7.4). After some cumbersome calculations we obtain the following system of ODEs:

$$1^{\circ} \quad i\gamma_{3}\dot{\varphi} + m(\gamma_{0} - \gamma_{4})\varphi = R.$$

$$2^{\circ} \quad i\gamma_{3}\dot{\varphi} + \left[\frac{i}{2\alpha}(\gamma_{0} + \gamma_{4})\gamma_{3} + m(\gamma_{0} - \gamma_{4})\right]\varphi = R.$$
(4.7.7)

$$\begin{split} 3^{\circ} & i\gamma_{3}\dot{\varphi} + [\alpha m(\gamma_{0} + \gamma_{4}) + m(\gamma_{0} - \gamma_{4})]\varphi = R. \\ 4^{\circ} & 2i\sqrt{\omega}\gamma_{2}\dot{\varphi} + \left[\frac{i}{2\sqrt{\omega}}\gamma_{2} + m(\gamma_{0} - \gamma_{4})\right]\varphi = R. \\ 5^{\circ} & -2i\gamma_{3}\dot{\varphi} + \left[\frac{i}{2\alpha}(\gamma_{0} + \gamma_{4})\gamma_{3} + \frac{\beta\omega}{\alpha^{2}}(\gamma_{0} + \gamma_{4}) + m(\gamma_{0} - \gamma_{4})\right]\varphi = R. \\ 6^{\circ} & -i(\gamma_{0} + \gamma_{4})\dot{\varphi} + \left[\frac{i}{2\alpha\omega}(\gamma_{0}\gamma_{4} - \alpha(\gamma_{0} + \gamma_{4})) + m(\gamma_{0} - \gamma_{4})\right]\varphi = R. \\ 7^{\circ} & -i(\gamma_{0} + \gamma_{4})\dot{\varphi} + \left[m(\gamma_{0} - \gamma_{4}) - \frac{i}{2\alpha}\gamma_{0}\gamma_{4}\right]\varphi = R. \\ 8^{\circ} & -i(\gamma_{0} + \gamma_{4})\dot{\varphi} + \left[m(\gamma_{0} - \gamma_{4}) - \frac{i}{2\alpha}\gamma_{0}\gamma_{4}\right]\varphi = R. \\ 9^{\circ} & -i(\gamma_{0} + \gamma_{4})\dot{\varphi} + \left[m(\gamma_{0} - \gamma_{4}) - \frac{i}{2\omega}(\gamma_{0} + \gamma_{4}) - \frac{i}{2\alpha}\gamma_{0}\gamma_{4}\right]\varphi = R. \\ 10^{\circ} & -i(\gamma_{0} + \gamma_{4})\dot{\varphi} + \left[m(\gamma_{0} - \gamma_{4}) - \frac{i}{2\omega}(\gamma_{0} + \gamma_{4})\right]\varphi = R. \\ 11^{\circ} & -i(\gamma_{0} + \gamma_{4})\dot{\varphi} + \left[m(\gamma_{0} - \gamma_{4}) - \frac{i}{2}\left(\frac{1}{\omega} + \frac{1}{\omega - \alpha}\right)(\gamma_{0} + \gamma_{4})\right]\varphi = R. \\ 13^{\circ} & -i(\gamma_{0} + \gamma_{4})\dot{\varphi} + \left[m(\gamma_{0} - \gamma_{4}) - \frac{i}{\alpha^{2}}(\gamma_{0} + \gamma_{4})\right]\varphi = R. \\ 14^{\circ} & -2i\alpha\gamma_{1}\dot{\varphi} + \left[m(\gamma_{0} - \gamma_{4}) + \frac{m\omega}{\alpha^{2}}(\gamma_{0} + \gamma_{4})\right]\varphi = R. \\ 15^{\circ} & 2i\sqrt{\omega}\gamma_{2}\dot{\varphi} + \left[\frac{i}{\sqrt{\omega}}(i\alpha m\gamma_{1} + \frac{1}{2}\gamma_{2}) + m(\gamma_{0} - \gamma_{4}) + \beta m(\gamma_{0} + \gamma_{4})\right]\varphi = R. \\ 16^{\circ} & -i(\gamma_{0} + \gamma_{4})\dot{\varphi} + \left\{i(\gamma_{0} + \gamma_{4})\left[(2\omega f(\omega)\right)^{-1}(\omega^{3} + \alpha(\alpha + \rho \tau)\omega - - 2\delta\alpha^{2}\tau) - \frac{1}{\omega}\right] + m(\gamma_{0} - \gamma_{4})\right\}\varphi = R. \end{split}$$

Equations $1^{\circ}-16^{\circ}$ correspond to that of ansatze in Table 4.7.1; dot means differentiation with respect to corresponding ω ;

$$R = \left[f_1(\overline{\varphi}\varphi,\overline{\varphi}(\gamma_0+\gamma_4)\varphi) + (\gamma_0+\gamma_4)f_2(\overline{\varphi}\varphi,\overline{\varphi}(\gamma_0+\gamma_4)\varphi)\right]\varphi.$$

Further we consider the nonlinear equation

$$\begin{bmatrix} -i(\gamma_0 + \gamma_4)\partial_t + i\gamma_a\partial_a + m(\gamma_0 - \gamma_4) - \lambda(\overline{\psi}(\gamma_0 + \gamma_4)\psi)^{1/2k} \end{bmatrix} \psi = 0, \quad (4.7.8)$$

 $\lambda, \ k = \text{const},$

which is a particular case of Equation (4.7.1), (4.7.4). Note that in this case corresponding reduced equations have form (4.7.7), provided $R = \lambda (\overline{\varphi}(\gamma_0 + \gamma_4)\varphi)^{1/2k}\varphi$.

We succeeded in solving Equations $6^{\circ}-13^{\circ}$ (4.7.7). In this connection the following lemma was essentially used.

Lemma 4.7.1. The quantities

$$I_{6} = \overline{\varphi}(\gamma_{0} + \gamma_{4})\varphi\omega, \quad I_{7} = \overline{\varphi}(\gamma_{0} + \gamma_{4})\varphi\omega^{2}, \quad I_{8} = \overline{\varphi}(\gamma_{0} + \gamma_{4})\varphi, \quad (4.7.9)$$

$$I_{9} = \overline{\varphi}(\gamma_{0} + \gamma_{4})\varphi\omega^{2}, \quad I_{10} = \overline{\varphi}(\gamma_{0} + \gamma_{4})\varphi, \quad I_{11} = \overline{\varphi}(\gamma_{0} + \gamma_{4})\varphi\omega,$$

$$I_{12} = \overline{\varphi}(\gamma_{0} + \gamma_{4})\varphi(\omega^{2} - \alpha\omega), \quad I_{13} = \overline{\varphi}(\gamma_{0} + \gamma_{4})\varphi(\omega^{3} - 2(\alpha + \beta)\omega^{2} + \alpha\beta\omega)$$

are the first integrals of the systems of ODEs 6°-13° (4.7.7), respectively.

Proof. Consider Equation 6° of (4.7.7) (the rest of the cases are analogous). Multiplying on γ_0 the complex conjugation of Equation 6° we obtain

$$i\overline{\varphi}(\gamma_0+\gamma_4)+\overline{\varphi}\bigg[m(\gamma_0-\gamma_4)+\frac{i}{2\omega}(\gamma_0+\gamma_4)+\frac{i}{2\alpha\omega}\gamma_0\gamma_4\bigg]=\lambda\Big(\overline{\varphi}(\gamma_0+\gamma_4)\varphi\Big)^{1/2k}\varphi,$$

from which

$$\dot{\overline{\varphi}}(\gamma_0 + \gamma_4)\varphi + \overline{\varphi}(\gamma_0 + \gamma_4)\dot{\varphi} = -\omega^{-1}\overline{\varphi}(\gamma_0 + \gamma_4)\varphi \quad \text{or}$$
$$\frac{d}{d\omega}\left[\overline{\varphi}(\gamma_0 + \gamma_4)\varphi\right] = -\omega^{-1}\overline{\varphi}(\gamma_0 + \gamma_4)\varphi.$$

After integrating this equation we find

$$\overline{\varphi}(\gamma_0 + \gamma_4)\varphi = c\omega^{-1}, \qquad c = \text{const.}$$

So, the lemma is proved.

Resolve relations (4.7.9) with respect to $\overline{\varphi}(\gamma_0 + \gamma_4)\varphi$ and inserting the result obtained in Equations 6°-13° (4.7.7) we get the linear systems of ODEs, having general solutions of the form [73*]:

$$\varphi_N(\omega) = \frac{1}{2} \left[f_N(\omega)(\gamma_0 + \gamma_4) + g_N(\omega)(1 + \gamma_0\gamma_4) \right] \chi, \qquad (4.7.10)$$

where $N = \overline{6-13}$, $\chi = (\chi^0 \chi^1 \chi^2 \chi^3)^T$,

$$g_{\theta}(\omega) = \frac{1}{\sqrt{\omega}} \exp\left\{ (16i\alpha^{2}m\omega)^{-1} + i\phi_{1}(k,\omega) \right\}, \qquad (4.7.11)$$

$$f_{\theta}(\omega) = \frac{1}{2m} (\tilde{\lambda}\omega^{-1/2k} + (2i\alpha\omega)^{-1})g_{\theta}(\omega);$$

$$f_{7}(\omega) = \frac{1}{2m} (\lambda\omega^{-1/k} + (2i\alpha\omega)^{-1})g_{7}(\omega),$$

$$g_{7}(\omega) = \frac{1}{\omega} \exp\left\{ (16i\alpha^{2}m\omega)^{-1} + i\phi_{2}(k,\omega) \right\};$$

$$f_{8}(\omega) = \frac{1}{2m} (\tilde{\lambda} + \frac{i}{2\alpha})g_{8}(\omega),$$

$$g_{8}(\omega) = \exp\left\{ -(1 + 4\alpha^{2}m\tilde{\lambda}^{2})(16i\alpha^{2}m)^{-1}\omega \right\};$$

$$f_{9}(\omega) = \frac{1}{2m} (\tilde{\lambda}\omega^{-1/k} + \frac{i}{2\alpha})g_{9}(\omega),$$

$$g_{9}(\omega) = \omega^{-1} \exp\left\{ -(16i\alpha^{2}m)^{-1}\omega + i\phi_{2}(k,\omega) \right\};$$

$$f_{10}(\omega) = \frac{\tilde{\lambda}}{2m} \exp\left\{ \frac{i\tilde{\lambda}^{2}}{4m} \omega \right\} g_{10}(\omega);$$

$$g_{10}(\omega) = \exp\left\{ \frac{i\tilde{\lambda}^{2}}{4m} \omega \right\};$$

$$f_{11}(\omega) = \frac{\tilde{\lambda}}{2m} \omega^{-1/2k}g_{11}(\omega),$$

$$g_{11}(\omega) = \frac{\tilde{\lambda}}{2m} (\omega^{2} - \alpha\omega)^{-1/2k}g_{12}(\omega),$$

$$g_{12}(\omega) = (\omega^{2} - \alpha\omega)^{-1/2} \exp\left\{ \frac{i\tilde{\lambda}^{2}}{4m} \int (z^{2} - \alpha z)^{-1/k} dz \right\};$$

$$f_{13}(\omega) = \frac{\tilde{\lambda}}{2m} [\omega(\omega - \alpha)(\omega - \beta)]^{-1/2k} \exp\left\{ \frac{i\tilde{\lambda}^{2}}{4m} \int [z(z - \alpha)(z - \beta)]^{-1/k} dz \right\}.$$

Chapter 4. Systems of PDEs Invariant Under Galilei Groups

In formulae (4.7.11) $\tilde{\lambda} = \lambda (\bar{\chi}(\gamma_0 + \gamma_4)\chi)^{1/2k}$,

$$\phi_n(k,\omega) = rac{\widetilde{\lambda}^2}{4m} \left\{ egin{array}{cc} rac{k}{k-n} \omega^{(k-n)/k}, & k
eq n \ \ln \omega, & k=n. \end{array}
ight.$$

Returning to ansatze $6^{\circ}-13^{\circ}$ from Table 4.7.1, we can easily write with the help of (4.7.10), (4.7.11) the corresponding solutions of Equation (4.7.8).

In conclusion, it will be noted that Equation (4.7.8) under $k = \frac{3}{2}$ is additionally invariant with respect to scale and projective transformations [94,73*]:

$$\begin{split} t' &= e^{2\theta}t, \quad x'_a = e^{\theta}x_a, \quad \psi'(x') = \exp\left\{\theta(-2 + \frac{1}{2}\gamma_0\gamma_4)\right\}\psi(x);\\ t' &= \frac{t}{1 - \theta t}, \quad x'_a = \frac{x_a}{1 - \theta t},\\ \psi'(x') &= (1 - \theta t)^2 \exp\left\{im\theta x_a x_a(\theta t - 1)^{-1} - \frac{1}{2t}\ln\left(1 - \theta t\right)[t\gamma_0\gamma_4 + (\gamma_0 + \gamma_4)\gamma_a x_a]\right\}\psi(x). \end{split}$$

4.8. Reduction and exact solutions of the Navier-Stokes equation

Following [63*] below we construct the complete set of $\tilde{G}(1,3)$ -inequivalent ansatze of codimension 1 for the Navier-Stokes(NS) field which reduce the NS equations to systems of ODEs. Having solved these ODEs we obtain thereby solutions of the NS equations.

The NS equations

- -

$$\frac{\partial \vec{u}}{\partial t} + (\vec{u} \cdot \vec{\nabla})\vec{u} - \Delta \vec{u} + \vec{\nabla}p = 0,$$
(4.8.1)
div $\vec{u} = 0,$

where $\vec{u} = \vec{u}(x) = \{u^1, u^2, u^3\}$ is the velocity field of a fluid, p = p(x) is the pressure, $x = \{t, \vec{x}\} \in \mathbb{R}(4)$, $\vec{\nabla} = \{\partial/\partial x_a\}$, a = 1, 2, 3; Δ the Laplacian, are basic equations of hydrodynamics which describe motion of an incompressible viscous fluid. The problem of finding the exact solutions of nonlinear Equations (4.8.1) is rather comlicated, and the symmetry approach turns out very effective. It is well known (see, e.g. [29]) that NS equations (4.8.1) are invariant under the extended Galilei group $\tilde{G}(1,3)$ generated by operators

$$\partial_{t} = \frac{\partial}{\partial t}, \qquad \partial_{a} = \frac{\partial}{\partial x_{a}},$$

$$J_{ab} = x_{a}\partial_{b} - x_{b}\partial_{a} + u^{a}\partial_{u^{b}} - u^{b}\partial_{u^{a}},$$

$$G_{a} = t\partial_{a} + \partial_{u^{a}},$$

$$D = 2t\partial_{t} + x_{a}\partial_{a} - u^{a}\partial_{u^{a}} - 2p\partial_{p},$$
(4.8.2)

N	Algebra	Invar. var. ω	Ansatze
1.	$\partial_1,\partial_2,\partial_3$	t	$u^1 = f(\omega), u^2 = g(\omega),$
			$u^3 = h(\omega), p = \varphi(\omega)$
2.	$\partial_t,\partial_1,\partial_2$	x_3	$u^1 = f(\omega), u^2 = g(\omega),$
	0.0		$u^3 = h(\omega), p = \varphi(\omega)$
3.	∂_t, ∂_1	x_3	$u^1 = x_2 + f(\omega), u^2 = g(\omega),$
4	$G_1 + G_2$	+ ² 2m	$u^{0} = h(\omega), p = \varphi(\omega)$ $u^{1} = f(\omega), u^{2} = g(\omega)$
т.	∂_1, ∂_2 $\partial_1 + G_2$	$\iota - 2x_3$	$u = f(\omega), u = g(\omega),$ $u^{3} = t + h(\omega), n = \omega(\omega)$
5.	∂_1, ∂_2	To	$u^{1} = t + f(\omega), p = \varphi(\omega)$ $u^{1} = t + f(\omega)$
	$\partial_t + G_1$	23	$u^{2} = a(\omega),$ $u^{2} = a(\omega),$
			$u^3 = h(\omega), p = \varphi(\omega)$
6.	$\partial_1, \partial_2 + G_1$	$t^2 - 2x_3$	$u^1 = x_2 + f(\omega), \ u^2 = g(\omega),$
	$\partial_t + G_3$		$u^3 = t + h(\omega), p = \varphi(\omega)$
7.	$\partial_1 + \alpha \partial_3,$	$t^2 + 2\alpha x_1 - 2x_3$	$u^1 = f(\omega), u^2 = g(\omega),$
	$\partial_2, \partial_t + G_3$	1.0	$u^3 = t + h(\omega), p = \varphi(\omega)$
8.	$\partial_1, \partial_t + G_3,$	$\alpha x_2 - x_3 + \frac{1}{2}t^2$	$u^1 = x_2 + f(\omega), u^2 = g(\omega),$
0	$G_1 + O_2 + \alpha O_3$	(-2 + -2)1/2	$u^{3} = t + h(\omega), p = \varphi(\omega)$
9.	O_t, O_3, J_{12}	$(x_1^2 + x_2^2)^{-7}$	$u^{2} = x_{1} f(\omega) - x_{2} g(\omega),$ $u^{2} = x_{1} g(\omega) + x_{2} f(\omega)$
			$u^{3} = h(\omega) + x_{2}f(\omega);$ $u^{3} = h(\omega) + x_{2}f(\omega);$
10.	$\partial_t + G_3$.	$(x_1^2 + x_2^2)^{1/2}$	$u^{1} = x_{1} f(\omega), p = \varphi(\omega)$ $u^{1} = x_{1} f(\omega) - x_{2} q(\omega).$
	∂_3, J_{12}	(~1 + ~2)	$u^{2} = x_{1} g(\omega) + x_{2} f(\omega),$
	.,		$u^3 = t + h(\omega), p = \varphi(\omega)$
11.	$\partial_{t}, \partial_{2}, D$	r_1/r_2	$u^{1} = \frac{1}{2}f(\omega)$ $u^{2} = \frac{1}{2}a(\omega)$
	() ()	~1/~2	x_2 x_2 x_2 x_2
			$u^3 = \frac{1}{r_1}h(\omega), p = \frac{1}{r^2}\varphi(\omega),$
12.	$\partial_t, \partial_3,$	$\ln{(x_1^2 + x_2^2)} +$	$u^1 = (x_1^2 + x_2^2)^{-1}$.
	$J_{12} + \alpha D$	$+2\alpha \arctan \frac{x_1}{2}$	$(x_1 f(\omega) - x_2 g(\omega)),$
		<i>x</i> ₂	$u^2 = (x_1^2 + x_2^2)^{-1}$
			$\cdot (x_1 g(\omega) + x_2 f(\omega)),$
			$u^{3} = (x_{1}^{2} + x_{2}^{2})^{-1/2}h(\omega),$
			$p = (x_1^2 + x_2^2)^{-1}\varphi(\omega)$

Table 4.8.1. $\widetilde{\mathrm{G}}(1,3)\text{-inequivalent ansatze of codimension 1 for the NS field$

Table 4.0.1.	Tabl	e	4.	8	•	1	•
--------------	------	---	----	---	---	---	---

N	Algebra	Invar. var.	Ansatze
13.	$\partial_t, J_{12},$	$\frac{(x_1^2+x_2^2)^{1/2}}{x_2}$	$u^1 = (x_1^2 + x_2^2)^{-1} \cdot$
	D	23	$\cdot (x_1 f(\omega) - x_2 g(\omega)),$
			$u^{2} = (x_{1}^{2} + x_{2}^{2})^{-1} \cdot $
			$ \begin{array}{c} \cdot (x_1 g(\omega) + x_2 f(\omega), \\ u^3 = (x_1^2 + x_2^2)^{-1/2} h(\omega) \end{array} $
			$p = (x_1^2 + x_2^2)^{-1}\varphi(\omega)$
14	de la	$(x_1^2 + x_2^2)^{1/2}$	$u^{1} - \frac{1}{2}(r_{1}f(u) - r_{2}g(u))$
11.	03, 512,	t	$\begin{array}{c} u = \frac{1}{t} \left(u_{1} f(u) - u_{2} g(u) \right) \\ 2 \end{array}$
	D		$u^2 = \frac{1}{t_1} (x_1 g(\omega) + x_2 f(\omega)),$
			$u^3 = \frac{1}{\sqrt{t}}h(\omega),$
			$p = \frac{1}{4}\varphi(\omega)$
15	Ga ha	$(x_1^2 + x_2^2)^{1/2}$	$u^{1} = \frac{1}{2}(x_{1}f(\omega) - x_{2}g(\omega))$
10.	G3, 512,	t	$\frac{u}{t} = \frac{1}{t} \left(\frac{u}{t} + \frac{1}{t} \right)$
	D		$u^{2} = \frac{1}{t_{1}} \frac{(x_{1}g(\omega) + x_{2}f(\omega))}{x_{2}},$
			$u^3 = \frac{1}{\sqrt{t}}h(\omega) + \frac{x_3}{t},$
			$p = \frac{1}{4}\varphi(\omega)$
16.	∂_1, ∂_2 .	$\frac{x_3}{}$	$u^{1} = \frac{1}{-\varepsilon}f(\omega), u^{2} = \frac{1}{-\varepsilon}q(\omega),$
	-1,-2,	\sqrt{t}	$ \begin{vmatrix} 1 & \sqrt{t} & \sqrt{t} \\ 2 & 1 & \sqrt{t} \\ 1 & \sqrt{t} \\$
	D		$u^{3} = \frac{1}{\sqrt{t}}h(\omega), p = \frac{1}{t}\varphi(\omega)$
17.	$\partial_1, D,$	$\frac{x_3}{\sqrt{t}}$	$u^1 = \frac{1}{\sqrt{t}}f(\omega) + \frac{\alpha x_2}{t},$
	$G_2 + \alpha G_1$		$u^2 = \frac{1}{1} a(\omega) + \frac{x_2}{2}.$
			$\begin{bmatrix} 1 & \sqrt{t}^{3} \langle t \rangle & t \\ 2 & 1 \\ 1 & 1 \\ 1 & 1 \end{bmatrix}$
			$u^{3} = \frac{1}{\sqrt{t}}h(\omega), p = \frac{1}{t}\varphi(\omega)$
18.	$G_1, G_2,$	$\frac{x_3}{\sqrt{4}}$	$u^1 = \frac{1}{\sqrt{t}}f(\omega) + \frac{x_1}{t},$
			$u^{2} = \frac{V^{L}}{1} a(w) + \frac{x_{2}}{2}$
			$\begin{bmatrix} u & -\sqrt{t}g(u) + t \\ 1 & 1 \end{bmatrix}$
			$u^{\circ} = \frac{1}{\sqrt{t}}h(\omega), p = \frac{1}{t}\varphi(\omega)$
19.	$\partial_1, G_2,$	$\frac{x_3}{\sqrt{4}}$	$u^1 = \frac{1}{\sqrt{4}}f(\omega),$
	מ		$u^2 = \frac{V^i}{1} q(\omega) + \frac{x_2}{2}$
			$\int_{-\infty}^{\infty} \sqrt{t} \frac{g(\omega) + \overline{t}}{t},$
			$u^{3} = \frac{1}{\sqrt{t}}h(\omega), p = \frac{1}{t}\varphi(\omega)$

where $\partial_{u^a} \equiv \frac{\partial}{\partial u^a}$, $\partial_p = \frac{\partial}{\partial p}$. Recently it was shown [161,74*] that the Lie-maximal invariance algebra of the NS equations (4.8.1) consists of the 11-dimensional $A\tilde{G}(1,3)$ (4.8.2) and the infinite-dimensional algebra A^{∞} with basis elements

$$Q = f^a \partial_a + \dot{f}^a \partial_{u^a} - x_a \ddot{f}^a \partial_p$$

$$R = g \partial_p$$
(4.8.3)

where $f^a = f^a(t)$ and g = g(t) are arbitrary differentiable functions of t; dot means differentiation with respect to t.

To reduce Equations (4.8.1) to a system of ODEs we use ansatze of codimension 1 obtained as invariants of inequivalent 3-dimensional subalgebras of $A\tilde{G}(1,3)$ described in [68,71*]. In Table 4.8.1. we list these 3-dimensional subalgebras and give corresponding invariant variable ω ; $\alpha \neq 0$ is an arbitrary constant; f, g, h, φ are differentiable functions of corresponding invariant variables ω .

Let us substitute ansatze from Table 4.8.1 into the Equations (4.8.1). As a result we obtain the following system of ODEs

$$\begin{split} 1^{\circ} & \dot{f} = 0, \qquad \dot{g} = 0, \qquad \dot{h} = 0. \\ 2^{\circ} & h\dot{f} - \ddot{f} = 0, \qquad h\dot{g} - \ddot{g} = 0, \\ & h\dot{h} - \ddot{h} + \dot{\varphi} = 0, \qquad \dot{h} = 0. \\ 3^{\circ} & g + h\dot{f} - \ddot{f} = 0, \qquad h\dot{g} - \ddot{g} = 0, \\ & h\dot{h} - \ddot{h} + \dot{\varphi} = 0, \qquad \dot{h} = 0. \\ 4^{\circ} & \dot{f}h + 2\ddot{f} = 0, \qquad \dot{g}h + 2\ddot{g} = 0, \\ & 1 - 2h\dot{h} - 4\ddot{h} - 2\dot{\varphi} = 0, \qquad \dot{h} = 0. \\ 5^{\circ} & 1 + h\dot{f} - \ddot{f} = 0, \qquad g\dot{h} - \ddot{g} = 0, \\ & h\dot{h} - \ddot{h} + \dot{\varphi} = 0, \qquad \dot{h} = 0. \\ 6^{\circ} & g - 2h\dot{f} - 4\ddot{f} = 0, \qquad h\dot{g} + 2\ddot{g} = 0, \\ & 1 - 2h\dot{h} - 4\ddot{h} - 2\dot{\varphi} = 0, \qquad \dot{h} = 0. \\ 7^{\circ} & (\alpha f - h)\dot{f} - 2(\alpha^{2} + 1)\ddot{f} + \alpha\dot{\varphi} = 0, \\ & (\alpha f - h)\dot{g} - 2(\alpha^{2} + 1)\ddot{g} = 0, \\ & (\alpha f - h)\dot{h} - 2(\alpha^{2} + 1)\ddot{h} - \dot{\varphi} + \frac{1}{2} = 0, \\ & \alpha\dot{f} - \dot{h} = 0. \\ \end{split}$$

$$\begin{split} 8^{\circ} & -\dot{f}(h-\alpha g) + g - (\alpha^{2}+1)\ddot{f} = 0, \\ & -\dot{g}(h-\alpha g) + \alpha\dot{\varphi} - (\alpha^{2}+1)\ddot{g} = 0, \\ & 1 - \dot{h}(h-\alpha g) - \dot{\varphi} - (\alpha^{2}+1)\ddot{h} = 0, \\ & \dot{h} - \alpha \dot{g} = 0. \end{split}$$

$$\begin{split} 9^{\circ} & f^{2} - g^{2} + \omega f\dot{f} + \frac{1}{\omega}\dot{\varphi} = \frac{3}{\omega}\dot{f} + \ddot{f} \\ & 2fg + \omega f\dot{g} = \frac{3}{\omega}\dot{g} + \ddot{g}, \\ & \omega f\dot{h} = \ddot{h} + \frac{1}{\omega}\dot{h}, \qquad 2f + \omega \dot{f} = 0. \end{split}$$

$$\begin{split} 10^{\circ} & f^{2} - g^{2} + \omega f\dot{f} + \frac{1}{\omega}\dot{\psi} = \frac{3}{\omega}\dot{g} + \ddot{f} \\ & 2fg + \omega f\dot{g} = \frac{3}{\omega}\dot{g} + \ddot{g}, \\ & 1 + \omega f\dot{h} = \ddot{h} + \frac{1}{\omega}\dot{h}, \qquad 2f + \omega \dot{f} = 0. \end{split}$$

$$\begin{split} 11^{\circ} & f\dot{f} - g(f + \omega\dot{f}) + \dot{\psi} = 2f + 4\omega\dot{f} + (\omega^{2} + 1)\ddot{f}, \\ & f\dot{g} - g(g + \omega\dot{g}) - (2\varphi + \omega\dot{\varphi}) = 2g + 4\omega\dot{g} + (\omega^{2} + 1)\ddot{g}, \\ & f\dot{h} - g(h + \omega\dot{h}) = 2h + 4\omega\dot{h} + (\omega^{2} + 1)\ddot{h}, \\ & \dot{f} - (g + \omega\dot{g}) = 0. \end{split}$$

$$\begin{split} 12^{\circ} & -\frac{1}{2}(f^{2} + g^{2}) + (f - \alpha g)\dot{f} - \varphi + \dot{\psi} = 2(-f - \dot{f} + \alpha\dot{g} + (\alpha^{2} + 1)\ddot{f}), \\ & -(f - \alpha g)\dot{g} + \alpha\dot{\varphi} = 2[g + \dot{g} + \alpha\dot{f} - (\alpha^{2} + 1)\ddot{g}], \\ & -fh + 2(f - \alpha g)\dot{h} = h - 4\dot{h} + 4(\alpha^{2} + 1)\ddot{h}, \\ & \dot{f} - \alpha\dot{g} = 0. \end{aligned}$$

$$\begin{split} 13^{\circ} & -f^{2} - g^{2} + \omega f\dot{f} - \omega^{2}\dot{h}\dot{f} - 2\varphi + \omega\dot{\varphi} = \omega(-f - \omega\ddot{f}) + \omega^{3}(2\dot{f} + \omega\ddot{f}), \\ & f\dot{g} - \omega^{2}\dot{h}\dot{g} = \omega(-g + \omega\ddot{g}) + \omega^{3}(2\dot{g} + \omega\ddot{g}), \\ & f(-h + \omega\dot{h}) - \omega^{2}\dot{h} - \omega^{2}\dot{\varphi} = h - \omega\dot{h} + \omega^{2}\ddot{h} + \omega^{3}(2\dot{h} + \omega\ddot{h}) \\ & \dot{f} - \omega\dot{h} = 0. \end{split}$$

$$14^{\circ} \quad f^2 - g^2 + 2\omega f \dot{f} + 2\dot{\varphi} = 4(2\dot{f} + \omega\ddot{f}),$$
$$g + \omega \dot{g} - 2f(g + \omega \dot{g}) = -4(2\dot{g} + \omega \ddot{g}),$$
$$-(\frac{1}{2}h + \omega\dot{h}) + 2\omega f \dot{h} + h = 4(\dot{h} + \omega\ddot{h})$$

$$\begin{aligned} f + \omega \dot{f} &= 0. \\ 15^{\circ} \quad f^{2} - g^{2} + 2\omega f \dot{f} + 2\dot{\varphi} &= 4(2\dot{f} + \omega\ddot{f}), \\ g + \omega \dot{g} - 2f(g + \omega \dot{g}) &= -4(2\dot{g} + \omega \ddot{g}), \\ -(\frac{1}{2}h + \omega\dot{h}) + 2\omega f \dot{h} + h &= 4(\dot{h} + \omega\ddot{h}) \\ f + \omega \dot{f} + \frac{1}{2} &= 0. \end{aligned}$$

$$16^{\circ} \quad -\frac{1}{2}(f + \omega \dot{f}) + h \dot{f} &= \ddot{f}, \\ -\frac{1}{2}(g + \omega \dot{g}) + h \dot{g} &= \ddot{g}, \\ -\frac{1}{2}(h + \omega\dot{h}) + h \dot{h} + \dot{\varphi} &= \ddot{h}, \qquad \dot{h} = 0. \end{aligned}$$

$$17^{\circ} \quad -\frac{1}{2}(f + \omega \dot{f}) + h \dot{f} + \alpha g &= \ddot{f}, \\ -\frac{1}{2}(g + \omega \dot{g}) + h \dot{g} + g &= \ddot{g}, \\ -\frac{1}{2}(g + \omega \dot{g}) + h \dot{g} + g &= \ddot{g}, \\ -\frac{1}{2}(h + \omega \dot{h}) + h \dot{h} + \dot{\varphi} &= \ddot{h}, \qquad \dot{h} + 1 = 0. \end{aligned}$$

$$18^{\circ} \quad \frac{1}{2}(f - \omega \dot{f}) + h \dot{f} &= \ddot{f}, \\ \frac{1}{2}(g - \omega \dot{g}) + h \dot{g} &= \ddot{g}, \\ -\frac{1}{2}(h + \omega \dot{h}) + h \dot{h} &= \ddot{h}, \qquad \dot{h} + 2 = 0. \end{aligned}$$

$$19^{\circ} \quad -\frac{1}{2}(f + \omega \dot{f}) + h \dot{f} &= \ddot{f}, \\ \frac{1}{2}(g - \omega \dot{g}) + h \dot{g} &= \ddot{g}, \\ -\frac{1}{2}(h + \omega \dot{h}) + h \dot{h} &= \ddot{h}, \qquad \dot{h} + 1 = 0. \end{aligned}$$

Enumeration $1^{\circ} - 19^{\circ}$ in (4.8.4) corresponds to that of ansatze in Table 4.8.1; dot means differentiation with respect to corresponding ω .

Equations $1^{\circ} - 10^{\circ}$ (4.8.4) can easily be solved and their general solutions are as follows:

1°. $f = c_1$, $g = c_2$, $\varphi = \varphi(\omega)$.

(Here and in what follows c with a subscript denotes an arbitrary constant; $\varphi = \varphi(\omega)$ means that φ is an arbitrary differentiable function of ω .)

2°.
$$f = \begin{cases} \frac{c_1}{c_3} e^{c_3 \omega} + c_2, & c_3 \neq 0\\ c_1 \omega + c_2, & c_3 = 0, \end{cases}$$

$$\begin{split} f &= \begin{cases} \frac{c_4}{c_3} e^{c_3 \omega} + c_5, & c_3 \neq 0\\ c_4 \omega + c_5, & c_3 = 0, \end{cases} \\ h &= c_3, \qquad \varphi = c_6. \\ 3^\circ. \quad f &= \begin{cases} c_1 + c_2 e^{c_3 \omega} + \frac{c_4}{c_3^2} \left(\omega - \frac{1}{c_3}\right) e^{c_3 \omega} - \frac{c_5}{c_3} \omega, & c_3 \neq 0\\ c_1 + c_2 \omega + \frac{1}{6} c_4 \omega^3 + \frac{1}{2} c_5 \omega^2, & c_3 = 0, \end{cases} \\ g &= \begin{cases} \frac{c_4}{c_3} e^{c_3 \omega} + c_5, & c_3 \neq 0\\ c_4 \omega + c_5, & c_3 = 0, \end{cases} \\ h &= c_3, \qquad \varphi = c_6. \end{cases} \\ 4^\circ. \quad f &= \begin{cases} \frac{c_1}{c_3^2} e^{-\frac{1}{2} c_3 \omega} + c_2, & c_3 \neq 0\\ c_1 \omega + c_2, & c_3 = 0, \end{cases} \\ g &= \begin{cases} \frac{c_4}{c_3} e^{-\frac{1}{2} c_3 \omega} + c_5, & c_3 \neq 0\\ c_4 \omega + c_5, & c_3 = 0, \end{cases} \\ h &= c_3, \qquad \varphi = \frac{1}{2} \omega + c_6. \end{cases} \\ 5^\circ. \quad f &= \begin{cases} -\frac{1}{c_3} \omega + \frac{c_1}{c_3^2} e^{c_3 \omega} + c_2 & c_3 \neq 0\\ \frac{1}{2} \omega^2 + c_1 \omega + c_2, & c_3 = 0, \end{cases} \\ g &= \begin{cases} \frac{c_4}{c_3} e^{c_3 \omega} + c_5, & c_3 \neq 0\\ \frac{1}{2} \omega^2 + c_1 \omega + c_2, & c_3 = 0, \end{cases} \\ g &= \begin{cases} \frac{c_4}{c_3} e^{c_3 \omega} + c_5, & c_3 \neq 0\\ \frac{1}{2} \omega^2 + c_1 \omega + c_2, & c_3 = 0, \end{cases} \\ g &= \begin{cases} \frac{c_4}{c_3} e^{c_3 \omega} + c_5, & c_3 \neq 0\\ \frac{1}{2} \omega^2 + c_1 \omega + c_2, & c_3 = 0, \end{cases} \\ h &= c_3, \qquad \varphi = c_6. \end{cases} \\ 6^\circ. \quad f &= \begin{cases} c_1 + c_2 e^{-\frac{1}{2} c_3 \omega} + \frac{c_5}{2c_3} \omega - \frac{c_4}{c_3^2} \left(\frac{\omega}{2} - \frac{1}{c_3}\right) e^{-\frac{1}{2} c_3 \omega}, & c_3 \neq 0\\ \frac{1}{4} \left(c_1 + c_2 \omega + \frac{1}{2} c_5 \omega^2 + \frac{1}{6} c_4 \omega^3\right), \qquad c_3 = 0, \end{cases} \end{cases}$$

$$g = \begin{cases} \frac{c_4}{c_3} e^{-\frac{1}{2}c_3\omega} + c_5, & c_3 \neq 0\\ c_4\omega + c_5, & c_3 = 0, \end{cases}$$

$$h = c_3, \qquad \varphi = \frac{1}{2}\omega + c_6.$$

$$7^{\circ} \quad f = \begin{cases} c_1 \exp\left\{\frac{c\omega}{2(\alpha^2 + 1)}\right\} + c_2 - \frac{\alpha\omega}{2(\alpha^2 + 1)c}, & c \neq 0, \\\\ \frac{\alpha\omega^2}{2[2(\alpha^2 + 1)]^2} + c_1\omega + c_2, & c = 0 \end{cases}$$

$$g = \begin{cases} c_3 \exp\left\{\frac{c\omega}{2(\alpha^2 + 1)}\right\} + c_4, & c \neq 0, \\ c_3\omega + c_4, & c = 0 \end{cases}$$

$$h = \alpha f - c,$$
 $\varphi = \frac{\omega}{2(\alpha^2 + 1)} + c_5.$

$$8^{\circ} \quad f = \begin{cases} \frac{\alpha\omega^2}{2c^2(\alpha^2+1)} + \frac{\omega}{c}\left(\frac{\alpha}{c^2} - c_4\right) + & c \neq 0, \\ + \left[\frac{c_3}{c}\left(\omega - \frac{\alpha^2+1}{c}\right) + c_1\right] \exp\frac{c\omega}{\alpha^2+1} + c_2, \\ \frac{1}{\alpha^2+1}\left(\frac{\alpha\omega^4}{24(\alpha^2+1)^2} + \frac{c_3}{6}\omega^3 + \frac{c_4}{2}\omega^2 + c_1\omega + c_2\right), \quad c = 0 \end{cases}$$

$$g = \begin{cases} \frac{-\alpha\omega}{c(\alpha^2+1)} + c_3 \exp\left\{\frac{c\omega}{\alpha^2+1}\right\} + c_4, & c \neq 0, \\ \frac{\alpha}{2(\alpha^2+1)}\omega^2 + c_3\omega + c_4, & c = 0, \end{cases}$$

$$h = \alpha g - c, \qquad \varphi = \frac{\omega}{\alpha^2 + 1} + c_6.$$

$$\begin{split} 9^{\circ} \quad f &= \frac{c}{\omega^2}, \qquad g = c_1 \omega^c + \frac{c_2}{\omega^2}, \qquad h = c_3 \omega^c + c_4, \\ \varphi &= \begin{cases} \frac{c_1^2}{2(c+1)} \omega^{2(c+1)} + \frac{2c_1c_2}{c} \omega^c - \frac{c_1^2 + c_2^2}{2\omega^2} + c_5, \quad c \neq -1, 0 \\ c_1^2 \ln \omega - \frac{2c_1c_2}{\omega} - \frac{c_2^2 + 1}{2\omega^2} + c_5, \qquad c = -1 \\ \frac{1}{2} c_1^2 \omega^2 + 2c_1c_2 \ln \omega - \frac{c_2^2}{2\omega^2} + c_5, \qquad c = 0. \end{cases} \end{split}$$

10° f, g, and φ are the same as in the previous case 9°,

$$h = \begin{cases} \frac{\omega^2}{2(2-c)} + c_3 \omega^c + c_4, & c \neq 2, 0\\ \frac{\omega^2}{4} - c_3 \ln \omega + c_4, & c = 0\\ \frac{1}{2} \omega^2 \ln \omega - \frac{1}{4} \omega^2 + c_3 \omega^2 + c_4, & c = 2 \end{cases}$$

For Equation 11° (4.8.4) we did not find solutions. A particular solution of Equations 12° (4.8.4) is

$$\begin{split} 12^{\circ} \quad f = c, \qquad g = 0, \qquad \varphi = 2c - \frac{1}{2}c^{2}, \\ h = \begin{cases} c_{1}e^{\lambda_{1}\omega} + c_{2}e^{\lambda_{2}\omega}, & \frac{c^{2}}{4} > \alpha^{2}(1+c), \\ e^{\lambda\omega}(c_{1}+c_{2}\omega), & \frac{c^{2}}{4} = \alpha^{2}(1+c), \\ e^{\lambda\omega}(c_{1}\cos\beta\omega+c_{2}\sin\beta\omega), & \frac{c^{2}}{4} < \alpha^{2}(1+c), \end{cases} \\ \lambda_{1,2} = \frac{1 + \frac{c}{2} \pm \sqrt{\frac{c^{2}}{4} - \alpha^{2}(1+c)}}{2(1+\alpha^{2})}, \\ \lambda = \frac{1 + \frac{c}{2}}{2(1+\alpha^{2})}, \qquad \beta = \frac{\sqrt{\alpha^{2}(1+c) - \frac{c^{2}}{4}}}{2(1+\alpha^{2})}. \\ \text{A particular solution of Equation 13° (4.8.4) is} \end{split}$$

13° $f = c_1$, $g = c_2$, h = 0, $\varphi = -\frac{1}{2}(c_1^2 + c_2^2)$. (4.8.5) Consider system 14° (4.8.4). Its last equation immediately gives

$$f = \frac{c}{\omega} \tag{4.8.6}$$

(as before, c is an arbitrary constant). Substituting (4.8.6) into the rest of Equations 14° (4.8.4) we get

$$4\frac{d^2}{d\omega^2}(\omega g) + \left(1 - \frac{2c}{\omega}\right)\frac{d}{d\omega}(\omega g) = 0$$
(4.8.7)

and

$$4\omega\ddot{h} + (\omega + 4 - 2c)\dot{h} + \frac{1}{2}h = 0.$$
(4.8.8)

Equation (4.8.7) can be easily integrated and the result is

$$g(\omega) = \frac{c_1}{\omega} \int^{\omega} x^{c/2} e^{-x/4} dx + \frac{c_2}{\omega}.$$
 (4.8.9)
In particular, when c = 0, the general solution of Equation (4.8.7) takes the form

$$g(\omega) = \frac{c_1}{\omega} e^{\omega/4} + \frac{c_2}{\omega}$$
 (4.8.10)

Equation (4.8.8) is in itself an equation for a degenerate hypergeometric function and it can be rewritten in standard Whittaker form

$$4x^2\ddot{W} - (x^2 - 4kx + 4m^2 - 1)W = 0, \qquad (4.8.11)$$

(where W = W(k, m, x); k, m are parameters) by the substitution

$$h(\omega) = \omega^{\frac{c-2}{4}} e^{-\omega/8} W\left(\frac{c}{4}, -\frac{c}{4}, \frac{\omega}{4}\right).$$
(4.8.12)

When c = 0, the substitution

$$h(\omega) = e^{-\tau} \widetilde{Z}_0(\tau), \qquad \tau = \frac{\omega}{8}$$
(4.8.13)

reduces Equation (4.8.8) to the modified Bessel equation of null order, that is

$$\tau \ddot{\widetilde{Z}}_0 + \dot{\widetilde{Z}}_0 - \tau \widetilde{\widetilde{Z}}_0 = 0. \tag{4.8.14}$$

Summarizing results (4.8.6)-(4.8.14) we can write down the general solutions of Equations 14° (4.8.4) as

$$14^{\circ} \quad f = \frac{c}{\omega}$$

$$g = \frac{c_1}{\omega} \int x^{c/2} e^{-x/4} dx + \frac{c_2}{\omega},$$

$$h = \omega^{\frac{c-2}{4}} e^{-\omega/8} W\left(\frac{c}{4}, -\frac{c}{4}, \frac{\omega}{4}\right),$$

$$\varphi = -\frac{c^2}{2\omega} + \frac{1}{2} \int g^2(y) dy + c_3.$$

$$(4.8.5)$$

(We continue to numerate solutions of reduced NS Equations $1^{\circ} - 19^{\circ}$ (4.8.5). Number n° indicates corresponding ansatz of Table 1.) When c = 0 we get from 14° (4.8.5) the following particular solution of Equations 14° (4.8.4)

14°°
$$f = 0, \qquad g = \frac{c_1}{\omega} e^{-\omega/4} + \frac{c_2}{\omega}$$
 (4.8.5)
 $h = e^{-\omega/8} \widetilde{Z}_0\left(\frac{\omega}{8}\right),$
 $\varphi = -\frac{c_2^2}{2\omega} + \frac{c_1^2}{2} \int^{\omega} \frac{e^{-y/2}}{y^2} dy + c_1 c_2 \int^{\omega} \frac{e^{-y/4}}{y^2} dy + c_3.$

where \widetilde{Z}_0 is a modified Bessel function satisfying Equation (4.8.14). Consider system 15° (4.8.4). Its last equation results in

$$f = \frac{c}{\omega} - \frac{1}{2}.$$
 (4.8.15)

The rest of Equation 15° take the form

$$2\frac{d^2}{d\omega^2}(\omega g) + \left(1 - \frac{c}{\omega}\right)\frac{d}{d\omega}(\omega g) = 0, \qquad (4.8.16)$$

$$2\dot{\varphi} = \left(\frac{c}{\omega}\right)^2 + g^2 - \frac{1}{4} \tag{4.8.17}$$

$$\omega \ddot{h} + \left(\frac{1}{2}\omega + 1 - \frac{c}{2}\right)\dot{h} - \frac{1}{8}h = 0.$$
(4.8.18)

Equations (4.8.16), (4.8.17) can be easily integrated and the result is as follows

$$g = \frac{c_1}{\omega} \int x^{c/2} e^{-x/2} dx + \frac{c_2}{\omega}, \qquad (4.8.19)$$

$$\varphi = \frac{1}{2} \int_{-\infty}^{\infty} g^2(y) dy - \frac{c^2}{2\omega} - \frac{1}{8}\omega.$$
 (4.8.20)

Equation (4.8.18) is reduced to the Whittaker equation (4.8.11) by the substitution

$$h(\omega) = \omega^{\frac{c-2}{4}} E^{-\omega/4} W\left(\frac{c-3}{4}, -\frac{c}{4}, \frac{\omega}{4}\right)$$
(4.8.21)

Note, when c = 3, function $W(0, -\frac{3}{4}, \frac{\omega}{2})$ is reduced to the modified Bessel function $\widetilde{Z}_{-\frac{3}{4}}(\frac{\omega}{4})$. The general relation is [130,75*]

$$W(0,m,x) = \sqrt{x} \,\widetilde{Z}_m\left(\frac{x}{2}\right). \tag{4.8.22}$$

So, we can write down the general solution of reduced NS Equations 15° (4.8.4) in the form

$$15^{\circ} \quad f = \frac{c}{\omega} - \frac{1}{2}$$

$$g = \frac{c_1}{\omega} \int x^{c/2} e^{-x/2} dx + \frac{c_2}{\omega},$$

$$h = \omega^{\frac{c-2}{4}} e^{-\omega/4} W\left(\frac{c-3}{4}, -\frac{c}{4}, \frac{\omega}{2}\right),$$

$$\varphi = \frac{1}{2} \int g^2(y) dy - \frac{c^2}{2\omega} - \frac{1}{8}\omega,$$
(4.8.5)

where W satisfies the Whittaker equation (4.8.11).

Consider system 16° (4.8.4). The last two equations of 16° (4.8.4) give rise to

$$h = c, \qquad \varphi = \frac{c\omega}{2} + c_1 \tag{4.8.23}$$

Taking into account (4.8.23) we can rewrite the remaining equations of system 16° as follows:

$$\dot{f} + (\frac{1}{2}\omega - c)\dot{f} + \frac{1}{2}f = 0,$$
 (4.8.24)

$$\ddot{g} + (\frac{1}{2}\omega - c)\dot{g} + \frac{1}{2}g = 0,$$
 (4.8.25)

By substituting

$$f(\omega) = F(\tau), \qquad \tau = \frac{1}{2}\omega - c \qquad (4.8.26)$$

into (4.8.24) we obtain equation

$$\frac{d^2F}{d\tau^2} + 2\tau \frac{dF}{d\tau} + 2F = 0.$$
 (4.8.27)

The general solution of Equation (4.8.27) is

$$F(\tau) = e^{-\tau^2} \left(c_2 + c_3 \int^{\tau} e^{y^2} dy \right).$$
 (4.8.28)

Summarizing results (4.8.23)-(4.8.28) we write the general solutions of Equations 16° (4.8.4):

16°
$$f = \exp\left\{-\left(\frac{\omega}{2}-c\right)^{2}\right\} \left(c_{2}+c_{3}\int^{\frac{\omega}{2}-c}e^{y^{2}}dy\right),$$

$$g = \exp\left\{-\left(\frac{\omega}{2}-c\right)^{2}\right\} \left(c_{4}+c_{5}\int^{\frac{\omega}{2}-c}e^{y^{2}}dy\right),$$

$$h = c, \qquad \varphi = \frac{c\omega}{2}+c_{1}.$$
(4.8.5)

In the same way we find solutions of reduced equations $17^{\circ} - 19^{\circ}$ (4.8.4). The solutions are as follows.

17°
$$\alpha = 1:$$

$$f = g = \left(\frac{3}{2}\omega - c\right)^{-1/2} \exp\left\{-\frac{1}{6}\left(\frac{3}{2}\omega - c\right)^2\right\} \times W\left(-\frac{5}{12}, \frac{1}{4}, \frac{1}{3}\left(\frac{3}{2}\omega - c\right)^2\right),$$

$$h = -\omega + c, \qquad \varphi = \frac{3c}{2}\omega - \omega^2 + c_1,$$

where W(.,.,.) satisfies the Whittaker equation (4.8.11). The above solution 17° (4.8.5) is a particular solution of Equation 17° (4.8.4) with $\alpha = 1$. When α is an arbitrary constant, the general solution of Equation 17 (4.8.4) has the form:

17°°
$$g = \left(\frac{3}{2}\omega - c\right)^{-1/2} \exp\left\{-\frac{1}{6}\left(\frac{3}{2}\omega - c\right)^2\right\} W\left(-\frac{5}{12}, \frac{1}{4}, \frac{1}{3}\left(\frac{3}{2}\omega - c\right)^2\right),$$
$$h = -\omega + c, \qquad \varphi = \frac{3c}{2}\omega - \omega^2 + c_1,$$

and f is determined from the ODE

$$\ddot{f} + \left(\frac{3}{2}\omega - c\right)\dot{f} + \frac{1}{2}f - \alpha g = 0.$$

The general solution of Equation 18° (4.8.4) is

$$18^{\circ} \qquad f = \left(\frac{5}{2}\omega - c\right)^{-1/2} \exp\left\{-\frac{1}{10}\left(\frac{5}{2}\omega - c\right)^{2}\right\} W\left(-\frac{27}{20}, \frac{1}{4}, \frac{1}{5}\left(\frac{5}{2}\omega - c\right)^{2}\right),$$
$$g = \left(\frac{5}{2}\omega - c\right)^{-1/2} \exp\left\{-\frac{1}{10}\left(\frac{5}{2}\omega - c\right)^{2}\right\} W\left(-\frac{27}{20}, \frac{1}{4}, \frac{1}{5}\left(\frac{5}{2}\omega - c\right)^{2}\right),$$
$$h = -2\omega + c, \qquad \varphi = \frac{5}{2}c\omega - 3\omega^{2} + c_{1}.$$
$$19^{\circ} \qquad f = \left(\frac{3}{2}\omega - c\right)^{-1/2} \exp\left\{-\frac{1}{6}\left(\frac{3}{2}\omega - c\right)^{2}\right\} W\left(-\frac{1}{12}, \frac{1}{4}, \frac{1}{3}\left(\frac{3}{2}\omega - c\right)^{2}\right),$$
$$g = \left(\frac{3}{2}\omega - c\right)^{-1/2} \exp\left\{-\frac{1}{6}\left(\frac{3}{2}\omega - c\right)^{2}\right\} W\left(-\frac{5}{12}, \frac{1}{4}, \frac{1}{3}\left(\frac{3}{2}\omega - c\right)^{2}\right),$$
$$h = -\omega + c, \qquad \varphi = \frac{3}{2}c\omega - \omega^{2} + c_{1}.$$

In $17^{\circ}-19^{\circ}$ (4.8.5) W(.,.,.) is an arbitrary solution of the Whittaker equation (4.8.11).

Remark 4.8.1. Solutions of reduced equations $1^{\circ}-19^{\circ}$ (4.8.5) should be considered together with corresponding ansatze of the Table 4.8.1, and then one gets solutions of the NS equations (4.8.1).

The above obtained solutions of the NS equations can be used as basic ones to construct multiparameter families of solutions. For this aim one has to use formulae of generating solutions (group multiplication of solutions) (see Paragraphs 1.4, 2.3). Below we list the final symmetry transformations generated by operators (4.8.2), (4.8.3) and corresponding formulae of group multiplication of solutions for the NS equations (4.8.1).

Table 4.8.2 Final symmetry transformations and corresponding formulae of group multiplication of solutions (GMS) for the NS equation (4.8.1)

Note: In 1-11 p'(x') = p(x) and therefore $p_{II}(x) = p_I(x')$. In this table $\delta_0, \delta_a, \alpha_a, \theta_a, \theta_a, \beta, \epsilon, \mathfrak{x}$ are arbitrary constants, $\alpha = (\alpha_1^2 + \alpha_2^2 + \alpha_3^2)^{1/2}$; \vec{f} and g are arbitrary differentiable functions of t. Formulae of GMS stated above allow us to contruct new solutions $\vec{u}_{II}(x)$ of the NS equations (4.8.1) starting from one $\vec{u}_I(x)$.

Remark 4.8.2. It will be noted that operator Q given in (4.8.3) generates transformations (N 12 in Table 4.8.2) which can be considered as an invariant transition to a frame of reference which is moved arbitrarily: $\vec{x}_{ref} = \epsilon \vec{f}(t)$.

Let us give some examples of application of formulae of GMS. Having applied formulae 5-7 of Table 4.8.2 to solution 16° (4.8.5) we get the following multiparameter families of solutions of the NS equations (4.8.1)

$$\vec{u}(x) = \frac{1}{\sqrt{t}} \left\{ e^{-\tau^2} \left[\vec{a} \left(\alpha_1 + \alpha_2 \int^{\tau} e^{s^2} ds \right) + \vec{b} \left(\alpha_3 + \alpha_4 \int^{\tau} e^{s^2} ds \right) \right] + \vec{c} \right\},$$

$$p(x) = \frac{1}{t} \left(\frac{\vec{c} \cdot \vec{x}}{2\sqrt{t}} + \alpha_5 \right), \qquad \tau = \frac{\vec{c} \cdot \vec{x}}{2\sqrt{t}} - 1, \qquad (4.8.29)$$

where $\alpha_1, \ldots, \alpha_5$ are arbitrary constants, $\vec{a}, \vec{b}, \vec{c}$, are arbitrary orthonormal constant vectors

$$\vec{a}^2 = \vec{b}^2 = \vec{c}^2 = 1, \qquad \vec{a} \cdot \vec{b} = \vec{b} \cdot \vec{c} = \vec{c} \cdot \vec{a} = 0.$$
 (4.8.30)

Further application of formulae of GMS N 8-10 to (4.8.29) gives rise to the following solution of Equations (4.8.1)

$$\vec{u}(x) = \frac{1}{\sqrt{t}} \left\{ e^{-y^2} \left[\vec{a} \left(\alpha_1 + \alpha_2 \int^y e^{s^2} ds \right) + \vec{b} \left(\alpha_3 + \alpha_4 \int^y e^{s^2} ds \right) \right] + \vec{c} \right\} - \vec{\theta},$$

$$p(x) = \frac{1}{t} (y + \alpha_5), \qquad y = \frac{\vec{c} \cdot (\vec{x} + \vec{\theta}t)}{2\sqrt{t}} - 1,$$

(4.8.31)

where $\vec{\theta}$ are arbitrary constants, the remaining designations are the same as in (4.8.29).

The procedure of generating solutions by means of symmetry transformations can be continued until one gets ungenerative families of solutions (see Paragraphs 1.4). Without doubt, the reader can make it, by analogy with the above examples, for any solution $1^{\circ} - 19^{\circ}$ (4.8.5) of the NS equations.

Ansatze collected in Table 4.8.1, of course, do not exhaust all possible ansatze which reduce the NS equations. Let us consider several examples of ansatze which cannot be obtained in the framework of local Lie approach.

The ansatz

$$\vec{u} = \vec{\nabla}\varphi \tag{4.8.32}$$

where $\varphi = \varphi(x)$ is a scalar function, reduces NS equations (4.8.1) to the system of Hamilton-Jacobi and Laplace equations

$$\varphi_t + \left(\vec{\nabla}\varphi\right)^2 + p = 0,$$

$$\Delta\varphi = 0.$$
(4.8.33)

This is an example of nonlocal component reduction.

Ansatz

$$\vec{u} = \vec{a}\varphi(t, \vec{b} \cdot \vec{x}, \vec{c} \cdot \vec{x}), \qquad (4.8.34)$$

where $\vec{a}, \vec{b}, \vec{c}$ are constant vectors satisfying (4.8.30), reduces (4.8.1) to the two-dimensional heat equation

$$\varphi_t - \Delta_2 \varphi = 0, \qquad \Delta_2 \equiv \frac{\partial^2}{\partial \omega_1^2} + \frac{\partial^2}{\partial \omega_2^2}, \qquad (4.8.35)$$
$$\omega_1 = \vec{b} \cdot \vec{x}, \qquad \omega_2 = \vec{c} \cdot \vec{x}.$$

Ansatz

$$\vec{u} = \vec{x}\varphi(x), \qquad p = p(x) \tag{4.8.36}$$

reduces Equations (4.8.1) to the system of PDEs

$$\vec{x}(\varphi_t + \Delta \varphi) + \vec{\nabla}(\varphi + p) = 0,$$

$$\varphi + (\vec{x} \cdot \vec{\nabla})\varphi = 0.$$
(4.8.37)

New ansatze and solutions of the NS equations can be constructed within the framework of conditional symmetry considered in Paragraph 5.7.

Let us make some concluding remarks. It will be noted that the question "What spin is carried by the NS field?" has a rather strange answer [66]: The NS field carries not only spin 1 but all possible integer spins $s = 0, 1, 2, \ldots$. This is due to the fact that the space of solutions of the NS equations can be decomposed into infinite direct sum of subspaces invariant under operators $S_{ab} = u^a \partial_{u^b} - u^b \partial_{u^a}$ from algebra AO(3), and these subspaces are not invariant under operators G_a from (4.8.2) because of the unboundedness of operators ∂_{u^a} . The following linearized NS equations are often used in hydrodynamics:

$$\vec{u}_t - \Delta \vec{u} = 0$$

$$\operatorname{div} \vec{u} = 0.$$
(4.8.38)

The maximal invariance algebra of Equations (4.8.38) is the 9-dimensional Lie algebra with basis elements

$$\partial_t, \quad \partial_a, \qquad J_{ab} = x_a \partial_b - x_b \partial_a + u^a \partial_{u^b} - u^b \partial_{u^a}, D = 2t \partial_t + x_a \partial_a, \qquad I = u^a \partial_{u^a}.$$

$$(4.8.39)$$

It is to be pointed out that Equations (4.8.38) are not Galilei invariant and therefore, generally speaking, they fail to adequately describe real hydrody-namic processes.

The ansatze which reduce NS equations (4.8.1) to linear systems of heat equations are considered in [72*]. The following is an example of such:

$$\vec{u} = \vec{a}f(t,\omega) + bg(t,\omega) + \vec{c}h(t,\omega)$$

$$p = p(t,\omega) \qquad \omega = \vec{c} \cdot \vec{x},$$
(4.8.40)

where $\vec{a}, \vec{b}, \vec{c}$ are constant vectors satisfying (4.8.30)

Ansatz (4.8.40) reduces Equations (4.8.1) to the system of PDEs

$$f_t - f_{\omega\omega} + h f_{\omega} = 0,$$

$$g_t - g_{\omega\omega} + h g_{\omega} = 0,$$

$$h_t + p_{\omega} = 0, \qquad h_{\omega} = 0.$$

(4.8.41)

The last two equations from (4.8.41) are easily integrated and after that the system (4.8.41) takes the form

$$f_t - f_{\omega\omega} + h(t)f_{\omega} = 0,$$

$$g_t - g_{\omega\omega} + h(t)g_{\omega} = 0, \quad p = -\dot{h}(t)\omega + \tilde{p}(t).$$
(4.8.42)

Further, the change of variables

$$t \to \tau = t, \quad \omega \to y = \omega - \int^t h(z) dz$$
 (4.8.43)

reduces Equations (4.8.42) to the two disconnected heat equations

$$f_{\tau} - f_{yy} = 0, g_{\tau} - g_{yy} = 0.$$
(4.8.44)

Chapter 5

Some Special Questions

In the present chapter we consider some problems tightly connected with group-algebraic investigations such as: finding nonlocal transformations to linearize a given nonlinear PDE, symmetry analysis of the three-body problem, calculating final transformations generated by non-Lie symmetry operators, and studying symmetry of integrodifferential equations. Here we introduce the concept of conditional invariance, and study non-Lie symmetry of quasirelativistic generalization of the Schrödinger equation, Galilean invariance of Maxwell's equations, solutions of the Schrödinger equation invariant under the non-Lie Lorentz algebra. Finally, in the concluding topic we introduce the concept of approximate symmetry.

5.1. On nonlocal linearization of nonlinear equations

It is well known that information about Lie (local or point) symmetry of a given nonlinear PDE sometimes allows us to find a substitution by means of which the equation is linearized. For example, if equation

$$\Box u = F\left(x_0, \dots, x_3, \frac{\partial u}{\partial x_0}, \dots, \frac{\partial u}{\partial x_3}\right)$$
(5.1.1)

is invariant under the conformal group C(1,3) then there exists a reversible point change of variables $u \to w = \varphi(u)$ which transforms Equation (5.1.1) to the free wave equation $\Box w = 0$ (see Paragraph 1.2). Class of nonlinear equations which can be linearized essentially extends if to make use of nonlocal transformations, that is, the transformations containing not only dependent and independent variables but also derivatives u, u, \dots Examples of such transformations were known long ago. In mechanics and hydrodynamics nonpoint transformations of Legendre, Euler, Laplace, Backlund, and hodographs have been widely used for a long time. A classical example of nonlocal transformations is the so-called Hopf-Cole transformation [40, 121]

$$u(x,t) = -2\mu \frac{v_x}{v}, \qquad v = v(t,x),$$
 (5.1.2)

(μ is a constant) which transforms Burgers' equation

$$u_t + uu_x - \mu u_{xx} = 0 \tag{5.1.3}$$

into heat equation

$$v_t = \mu v_{xx}.\tag{5.1.4}$$

Below we show how to obtain the Hopf-Cole transformation from the group-theoretical point of view $[81^*]$. Another interesting approach is developed in $[145^*]$.

It is not difficult to calculate the maximal IA of the Burgers' equation (5.1.3). It is five-parameter algebra, basis elements having the form

X:
$$X_1 = \partial_x, \qquad X_2 = \partial_t$$

 $X_3 = x\partial_x + 2t\partial_t - u\partial_u,$
 $X_4 = t\partial_x + \partial_u,$
 $X_5 = tx\partial_x + t^2\partial_t + (x - tu)\partial_u.$
(5.1.5)

It means that Burgers' equation is invariant with respect to space X_1 and time X_2 translations; scale changes X_3 ; Galilean transformations X_4 ; and projective transformations generated by X_5 . There is a remarkable similarity between the symmetry algebra of Burgers' equation and that for the heat equation. The latter has the form

Y:
$$Y_1 = \partial_x, \quad Y_2 = \partial_t$$

 $Y_3 = x\partial_x + 2t\partial_t,$
 $Y_4 = t\partial_x - \frac{1}{2\mu}xv\partial_v,$
 $Y_5 = tx\partial_x + t^2\partial_t - \frac{1}{4\mu}(x^2 + 2\mu t)v\partial_v.$
 $Y_6 = v\partial_v$

$$(5.1.6)$$

and in addition, the heat equation, being linear, admits an infinite-dimensional group generated by operators

$$Y_{\infty} = h(t, x)\partial_v \tag{5.1.7}$$

with arbitrary solution h(t, x) of the equation.

Although the IAs (5.1.5) and (5.1.6) are such similar, nevertheless, the difference between them means that there is not a point change of variables which reduces Burgers' equation to the heat equation.

Suppose

$$u = \varphi(v, v_x) \tag{5.1.8}$$

and require the system of PDEs

$$u_t + uu_x - \mu u_{xx} = 0,$$

$$v_t - \mu v_{xx} = 0,$$

$$u = \varphi(v, v_x)$$

(5.1.9)

to be invariant with respect to the algebra $Z = \{X, Y\}$, i.e.,

$$Z: \qquad Z_1 = \partial_x, \qquad Z_2 = \partial_t$$

$$Z_3 = x\partial_x + 2t\partial_t - u\partial_u,$$

$$Z_4 = t\partial_x + \partial_u - \frac{1}{2\mu}xv\partial_v,$$

$$Z_5 = tx\partial_x + t^2\partial_t + (x - tu)\partial_u - \frac{1}{4\mu}(x^2 + 2\mu t)v\partial_v.$$

$$Z_6 = v\partial_v, \qquad Z_{\infty} = h(t, x)\partial_v.$$
(5.1.10)

Applying the second prolongation of operators (5.1.10) to the system (5.1.9), we obtain from the condition of invariance the defining equations for function $\varphi = \varphi(v, v_x)$. So, using formula (1.1.7) we find the first prolongation of the operator Z_4

$$Z_4 = Z_4 - \frac{1}{2\mu}(v + xv_x)\partial_{v_x}.$$

Then we get

$$Z_{14}(u-\varphi(v,v_x)) = \left\{1 + \frac{1}{2\mu}\left[xv\varphi_v + (v+xv_x)\varphi_{v_x}\right]\right\} = 0$$

Equating coefficients with variable x and then those without x we obtain defining equations

$$v\varphi_v + v_x\varphi_{v_x} = 0,$$

$$1 + \frac{1}{2\mu}v\varphi_{v_x} = 0.$$
(5.1.11)

The general solution of the system (5.1.11) is

$$\varphi(v, v_x) = -2\mu \frac{v_x}{v}.\tag{5.1.12}$$

It can be easily checked that operators Z_5 , Z_6 , and Z_∞ leave the system (5.1.9) with function φ (5.1.12) invariant. From (5.1.8), (5.1.12) follows the Hopf-Cole transformation (5.1.2).

Formula (5.1.2) can be used to obtain a nonlinear superposition principle for the Burgers' equation. Let $u_k(x)$ be a solution of Equation (5.1.3). Inserting it into (5.1.2) and integrating, we get

$$v_k(x,t) = \exp\left\{-\frac{1}{2\mu}\int^x u_k(\tau,t)d\tau\right\}$$
 (5.1.13)

Clearly v_k (5.1.13) satisfies the heat equation (5.1.4). Since the heat equation is linear, it possesses a linear superposition principle. It means that the function

$$v(x,t) = \sum_{k>1} v_k(x,t)$$

will be a solution of the equation as soon as every $v_k(x,t)$ is its solution. So, using (5.1.13) and (5.1.2), we find thereby the superposition principle for Burgers' equation

$$u(x,t) = -2\mu \frac{\partial}{\partial x} \ln \left[\sum_{k>1} \exp\left\{ -\frac{1}{2\mu} \int^x u_k(\tau,t) d\tau \right\} \right].$$
 (5.1.14)

Now consider the Liouville equation

$$u_{xt} + \lambda e^u = 0. \tag{5.1.15}$$

It admits operators

$$X = f(t)\partial_t + g(x)\partial_x - (f_t + g_x)\partial_u \qquad (5.1.16)$$

with arbitrary differentiable functions f = f(t) and g = g(x). Analogous symmetry properties have the free wave equation

$$v_{xt} = 0.$$
 (5.1.17)

It admits operators $Y = \{Y_1, Y_2\}$:

$$Y_1 = f(t)\partial_t + g(x)\partial_x, \qquad Y_2 = v\partial_v. \tag{5.1.18}$$

The similarity between symmetry operators (5.1.16) and (5.1.18) says that a transformation may have to exist which connects Equations (5.1.15) and (5.1.17). To find the constraint equation

$$u = \varphi(v, v_x, v_t)$$

explicitly we require it to be invariant under the algebra $X \oplus Y = \{X, Y_2\}$. Therefore from the condition of invariance

$$\begin{array}{l} X_1(u-\varphi(v,v_x,v_t)) \left|_{\substack{u=\varphi\\v_{xt}=0}} = \left[-\left(f_t+g_x\right) - v_t\varphi_{v_t} - v_x\varphi_{v_x} \right] \right|_{\substack{u=\varphi\\v_{xt}=0}} = 0, \\ Y_2(u-\varphi) \left|_{u=\varphi} = v\varphi_v + v_t\varphi_{v_t} + v_x\varphi_{v_x} = 0 \end{array}$$

we find, setting v = f + g,

$$u = \varphi(v, v_x, v_t) = \ln c + \ln v_x + \ln v_t - 2\ln v = \ln \frac{cv_x v_t}{v^2}$$
(5.1.19)

where c is an arbitrary constant.

Substitution of (5.1.19) into (5.1.15) reduces the latter to (5.1.6) when $c = -2/\lambda$.

Using this fact, it is not difficult to write down the general solution of the Liouville equation (5.1.15)

$$u(x,t) = \ln \left(-\frac{2}{\lambda} \frac{f_t g_x}{(f+g)^2}\right)$$

(Compare with (1.6.2)).

Following [105], consider more general form of nonlocal transformations, namely the transformations

T:

$$\begin{aligned} x_{\nu} \to x_{\nu}' &= f_{\nu} \left(x, u, \underbrace{u}_{1}, \dots, \underbrace{u}_{\ell} \right), \\ u^{s} \to u'^{s} &= \varphi^{s} \left(x, u, \underbrace{u}_{1}, \dots, \underbrace{u}_{k} \right), \end{aligned}$$
(5.1.20)

where

$$u = \{u^s\}, \quad s = \overline{1, m} \qquad u = \frac{\partial u^s}{\partial x_{\nu}}, \qquad u = \frac{\partial^2 u^s}{\partial x_{\nu} \partial x_{\nu}}.$$

From (5.1.20) we get

$$\frac{\partial u'^s}{\partial x'_{\nu}} = \frac{\partial \varphi^s}{\partial x_{\mu}} \frac{\partial x_{\mu}}{\partial x'_{\nu}} + \frac{\partial \varphi^s}{\partial u^k} \frac{\partial u^k}{\partial x_{\mu}} \frac{\partial x_{\mu}}{\partial x'_{\nu}} + \frac{\partial \varphi^s}{\partial u^k_{\mu}} \frac{\partial u^k_{\mu}}{\partial x_{\sigma}} \frac{\partial x_{\sigma}}{\partial x'_{\nu}} + \cdots$$
(5.1.21)

With the help of the operator of total differentiation

$$D_{\nu} = \frac{\partial}{\partial x_{\nu}} + u_{\nu\mu}^{k} \frac{\partial}{\partial u_{\mu}^{k}} + \cdots$$
 (5.1.22)

The equality (5.1.21) can be rewritten as follows:

$$D_{\nu}\varphi^{s} = \frac{\partial u^{\prime s}}{\partial x^{\prime}_{\mu}} D_{\nu} x^{\prime}_{\mu}.$$
 (5.1.23)

Analogous expression hold for the second derivatives

$$D_{\sigma}D_{\nu}\varphi^{s} = \frac{\partial^{2}u^{\prime s}}{\partial x_{\mu}^{\prime}\partial x_{\rho}^{\prime}}(D_{\sigma}x_{\mu}^{\prime})(D_{\nu}x_{\rho}^{\prime}) + \left(\frac{\partial u^{\prime s}}{\partial x_{\mu}}\right)D_{\sigma}D_{\nu}x_{\mu}^{\prime}.$$
 (5.1.24)

Formulae (5.1.23), (5.1.24) are relations for determining the primed derivatives.

A PDE $(L_1 = 0)$ is said to be reducible to a PDE $(L_2 = 0)$ by means of transformation T if on sets of solutions of these equations and their differential consequences denoted as $[L_1] = 0$ and $[L_2] = 0$, the following relations hold

$$TL_1 \Big|_{\substack{[L_1]=0\\[L_2]=0}} = 0.$$
 (5.1.25)

To illustrate what has been said, consider several examples.

Theorem 5.1.1. [105]. The Monge-Ampere equation

$$u_{xx}u_{tt} - u_{xt}^2 = 0 (5.1.26)$$

is reduced to the equation

$$v_{\eta} = 0, \qquad v = \phi(\xi) \tag{5.1.27}$$

by means of nonlocal change of variables

$$\xi = u_t, \qquad \eta = x, \qquad v = u_x.$$

Proof. Using formulae (5.1.23), (5.1.24) we find

$$D_{x}\xi = u_{xt}, \quad D_{x}\eta = 1, \quad D_{x}v = u_{xx},$$
$$D_{t}\xi = u_{tt}, \quad D_{t}\eta = 0, \quad D_{t}v = u_{xt},$$
$$v_{\eta} = -(u_{tt})^{-1}(u_{tx}^{2} - u_{xx}u_{tt}), \quad u_{tt} \neq 0,$$

whence follows (5.1.27).

It will be noted that by force of Equation (5.1.27) any solution of the equation $u_x = \phi(u_t)$ also satisfies Equation (5.1.26).

One can make such in much the same way that the two-dimensional ELBI equation in the Euclidean space

$$(1+u_t^2) u_{xx} = 2u_x u_t u_{xt} - (1+u_x^2) u_{tt}$$
(5.1.28)

is reduced by means of the Monge transformation

$$u = i \int (1 + v_{\xi}^2)^{1/2} d\xi + i \int (1 + v_{\eta}^2)^{1/2} d\eta + c_1$$
$$x = \xi + \eta + c_2, \qquad t = v + c_3$$

 $(c_1, c_2, c_3 \text{ are arbitrary constants})$ to the equation

$$v_{\xi\eta} = 0, \qquad v = \phi(\xi) + \psi(\eta)$$
 (5.1.29)

In [105, 106] is contained many examples of the linearization of nonlinear PDEs by means of nonlocal transformations.

It will be noted that nonlocal transformations can be used to construct formulae of generating solutions of a given nonlinear PDE. Let us give an example of such formulae. New solution $u^{II}(x)$ of the Korteweg-de Vries equation (KdV)

$$u_t + 6uu_x + u_{xxx} = 0 \tag{5.1.30}$$

can be derived from a known one $u^{I}(x)$ according to the formula [118*]

$$u^{II}(x) = -u^{I}(x) - 2v^{2}(x) = u^{I}(x) - 2v_{x}, \qquad (5.1.31)$$

if there is a function v(x) which satisfies the equations

$$v_x = v^2 + u^I(x),$$

$$v_t - 6v^2 v_x + v_{xxx} = 0.$$
(5.1.32)

This statement can be easily proved. Note that Equations (5.1.32) are compatible iff $u^{I}(x)$ satisfies the KdV equation (5.1.30).

Let us demonstrate the usefulness of formula (5.1.31) by considering several simple examples. Using the simplest solution of the KdV equation

$$u^{(1)} = \lambda, \qquad \lambda = \text{const}$$

we get, according to (5.1.31), the new solution

$$u^{(2)} = \lambda - 2v_x^{(1)}.$$

To find $u^{(2)}$ explicitly one has to solve the Riccati equation

$$v_x^{(1)} = \left(v_x^{(1)}\right)^2 + \lambda. \tag{5.1.33}$$

There are three cases: $\lambda = 0$, $\lambda = -1$, $\lambda = 1$. Consider the first one. The general solution of Equation (5.1.33) under $\lambda = 0$ has the form

$$v^{(1)} = \frac{-1}{x + c(t)} \tag{5.1.34}$$

where c(t) is an arbitrary function of t, which should be defined from the second equation of (5.1.32). From this one we find that $\dot{c} = 0$ and due to the invariance of the equation with respect to time translations we can put, without loss of generality, c = 0. So, starting from a trivial solution $u^{(1)} = 0$ of the KdV equation (5.1.30), we find by means of (5.1.31) the new solution $u^{(2)} = -2/x$. Let us repeat the above procedure once more, that is, make up

$$u^{(3)} = u^{(2)} - 2v_x^{(2)}, (5.1.35)$$

where function $v^{(2)}$ satisfies, according to (5.1.32), the system

$$v_x^{(2)} = (v^{(2)})^2 + u^{(2)},$$
 (5.1.36)

$$v_t^{(2)} - 6(v^{(2)})^2 v_x^{(2)} + v_{xxx}^{(2)} = 0.$$
 (5.1.37)

The solution to the Riccati equation (5.1.36) is

$$v^{(2)} = \frac{c(t) - 2x^3}{x(c(t) + x^3)}.$$
(5.1.38)

After substitution of (5.1.38) into (5.1.37) we find $\dot{c} = 12$, and therefore

$$u^{(3)} = \frac{6(24tx - x^4)}{(12t + x^3)^2}.$$
(5.1.39)

Continuing this process we will have on the nth step the formulae

$$u^{(n+1)} = u^{(n)} - 2v_x^{(n)}, (5.1.40)$$

$$v_x^{(n)} = \left(v^{(n)}\right)^2 + u^{(n)},\tag{5.1.41}$$

$$v_t^{(n)} - 6(v^{(n)})^2 v_x^{(n)} + v_{xxx}^{(n)} = 0.$$
(5.1.42)

The main difficulty in application of formulae (5.1.40)–(5.1.42) for generating new solutions of the KdV equation (5.1.30) consists in solving the Riccati equation (5.1.41). However, starting from $u^{(1)} = \lambda = \text{const}$ we succeeded in

obtaining the general solution of Equation (5.1.41) for any $u^{(n)}$ (found from (5.1.40)) as

$$v^{(n)} = -v^{(n-1)} - w^{(n+1)}, (5.1.43)$$

where

$$w^{(n+1)} = \left(\ln \int \exp\left\{2\sum_{m=0}^{n} (-1)^{n-m} \int w^{(m)} dx\right\} dx\right)_{x}$$
(5.1.44)

 $n = 0, 1, 2, \ldots, \qquad w^{(0)} = 0.$

From (5.1.42) one can define more exactly the dependence of $w^{(n)}$ on t and then, from (5.1.40), it follows

$$\lambda \to \lambda + 2 \left(\ln \varphi^{(1)} \right)_{xx} \to 1 + 2 \left(\ln \varphi^{(2)} \right)_{xx} \to \dots$$

where $(\ln \varphi)_x = w$, or

$$u^{(2n)} = \lambda + 2\sum_{m=0}^{n} w_x^{(2m)}, \quad u^{(2m+1)} = \lambda + 2\sum_{m=0}^{n} w_x^{(2m+1)}.$$
 (5.1.45)

So, using formulae (5.1.44), (5.1.45), we have the following chain of solutions

$$0 \rightarrow -\frac{2}{x^2} \rightarrow \frac{6(24tx - x^4)}{(12t + x^3)^2} \rightarrow \dots$$

Analogously, one obtains two chains of solutions of the KdV equation (5.1.30):

$$1 \to 1 - \frac{2}{\cos^2(x - 2t)} \to 1 - \frac{16[(x + 6t)\sin(2x - 4t) + \cos(2x - 4t) + 1]}{[2(x + 6t) + \sin(2x - 4t)]^2} \to \dots$$
$$-1 \to -1 + \frac{2}{\operatorname{ch}^2(x + 2t)} \to -1 + \frac{16[(x - 6t)\sin(2x + 4t) - \operatorname{ch}(2x + 4t) - 1]}{[2(x - 6t) + \sin(2x + 4t)]^2} \to \dots$$

It will be noted that Lie's symmetry of Equation (5.1.30) results in the following formulae of generating solutions:

translations
$$-u^{(2)}(t,x) = u^{(1)}(t+a_0,x+a_1);$$
 (5.1.46)

Galilei transformations — $u^{(2)}(t,x) = u^{(1)}(t,x+\theta t) - \frac{1}{6}\theta;$ (5.1.47)

Chapter 5. Some Special Questions

scale transformations $-u^{(2)}(t,x) = \beta^2 u^{(1)}(\beta^3 t,\beta x),$ (5.1.48)

where a_0 , a_1 , θ , β are arbitrary constants.

Formula (5.1.31) together with those of (5.1.46)-(5.1.48) allow us to construct wide classes of exact solutions (soliton-like and non-soliton) of the KdV equation (5.1.30). Particularly, the solution

$$u = -1 + \frac{2}{\operatorname{ch}^2(2t+x)}$$

after applying formulae (5.1.46)–(5.1.48), provided $\theta = -6$, $a_0 = 0$, takes the form of the well-known soliton solution

$$u = \frac{2\beta^2}{\operatorname{ch}^2\beta(x-4\beta^2t+a_1)}.$$

It is obvious that formula (5.1.31) is not unique for the KdV equation. There is, for example, another one

$$\begin{split} u^{(2)} &= u^{(1)} - 2 \big(v_x^{(1)} + k^{(1)} \big), \\ v_x^{(1)} &= \big(v^{(1)} \big)^2 + k^{(1)} + u^{(1)}, \\ v_t^{(1)} &= -6 \big[\big(v^{(1)} \big)^2 + k^{(1)} \big] u_x^{(1)} + v_{xxx}^{(1)}, \end{split}$$

where $k^{(1)}$ is an arbitrary constant.

In conclusion we note that multidimensional generalization of the KdV equation (5.1.30) is suggested in [63]:

$$u_t + F(u) \left(\frac{\partial u}{\partial x_k} \frac{\partial u}{\partial x_k}\right)^{1/2} + G(u) \Delta \left(\frac{\partial u}{\partial x_k} \frac{\partial u}{\partial x_k}\right)^{1/2} = 0,$$

 $u = u(t, x_k), k = 1, 2, 3; F, G$ are arbitrary smooth functions of u.

5.2. Symmetry, integrals of motion, and some partial solutions of the three-body problem

Here we consider from group-theoretic point of view the classical three-body problem in four-dimensional spacetime.

All of the ten first integrals of the three-body Newtonian problem (that is, when particles are interacting according to Newtonian laws) were known to Euler. Jacobi was the first to show that if particles are interacting with force inversely proportional to the third power of interbody distance, then equations of motions admit one more new the first integral, eleventh, the so-called generalized Jacobi integral. It is, of course, independent of those ten. Further, Poincare, Bruns, and Painleve proved a set of fundamental theorems on integrability of the classical three-body problem. They showed that any integral of motion which represented itself as an arbitrary function of coordinates and an algebraic function of velocities of the particles is just a combination of classical integrals. Yu. D. Sokolov [193] for the first time after Jacobi found a new first integral of the one-dimensional three-body problem in the case when particles have equal mass and they interact with force proportional to the third power of interbody distance. This new integral was generalized on flat and three-dimensional space in [110, 48]. Recently Golubev and Grebennikov [113] developed new fruitful methods of qualitative analysis of equations of motion of the three-body problem.

1. To study the symmetry properties of ODEs

$$\ddot{x}_k \equiv \frac{d^2 x_k}{dt^2} = F_k(t, x, \dot{x}), \qquad k = \overline{1, N}.$$
(5.2.1)

we following [64] rewrite system (5.2.1) in equivalent form

$$\frac{dt}{1} = \frac{dx_1}{\dot{x}_1} = \dots = \frac{dx_N}{\dot{x}_N} = \frac{d\dot{x}_1}{F_1} = \dots = \frac{d\dot{x}_N}{F_N}.$$
(5.2.2)

In turn, the system (5.2.2) is equivalent to one linear PDE

$$L\psi \equiv \left(\frac{\partial}{\partial t} + \dot{x}_k \frac{\partial}{\partial x_k} + F_k \frac{\partial}{\partial \dot{x}_k}\right)\psi = 0$$
 (5.2.3)

for the scalar function ψ depending on t, x_k, \dot{x}_k . Now one can apply to Equation (5.2.3) the standard algorithm of Lie according to which the symmetry operators are looked for in the form

$$X = \xi^0 \partial_t + \xi^k \partial_{x_k} + \tilde{\xi}^k \partial_{\dot{x}_k} + \eta \partial_{\psi}, \qquad (5.2.4)$$

where ξ^0 , ξ^k , $\tilde{\xi}^k$, η are functions on t, x_k, \dot{x}_k, ψ .

The invariance condition can be taken in commutator form

$$\begin{bmatrix} L, X \end{bmatrix} = BL \tag{5.2.5}$$

with an arbitrary function B depending on t, x_k, \dot{x}_k, ψ .

Substitution of (5.2.4) into (5.2.5) results in

$$(L\xi^{0})\frac{\partial}{\partial t} + (L\xi^{k})\frac{\partial}{\partial x_{k}} + (L\tilde{\xi}^{k})\frac{\partial}{\partial \dot{x}_{k}} + (L\eta)\frac{\partial}{\partial \psi} - \tilde{\xi}^{k}\frac{\partial}{\partial x_{k}} -$$
(5.2.6)

Chapter 5. Some Special Questions

$$-\left[\left(\xi^{0}\frac{\partial}{\partial t}+\xi^{s}\frac{\partial}{\partial x_{s}}+\widetilde{\xi}^{s}\frac{\partial}{\partial \dot{x}_{s}}\right)F_{k}\right]\frac{\partial}{\partial \dot{x}_{k}}=B\left(\frac{\partial}{\partial t}+\dot{x}_{k}\frac{\partial}{\partial x_{k}}+F_{k}\frac{\partial}{\partial \dot{x}_{k}}\right)$$

After equating in (5.2.6) coefficients of the various derivatives, we get

$$L\xi^{0} = B, \qquad L\xi^{k} - \tilde{\xi}^{k} = B\dot{x}^{k},$$

$$L\tilde{\xi}^{k} - \left(\xi^{0}\frac{\partial}{\partial t} + \xi^{s}\frac{\partial}{\partial x_{s}} + \tilde{\xi}^{s}\frac{\partial}{\partial \dot{x}_{s}}\right)F_{k} = BF_{k},$$

$$L\eta = 0.$$
(5.2.7)

Excluding from (5.2.7) the function B, we obtain the defining equations

$$L\left(L\xi^{k} - \dot{x}_{k}L\xi^{0}\right) - F_{k}L\xi^{0} = \xi^{0}\frac{\partial F_{k}}{\partial t} + \xi^{s}\frac{\partial F_{k}}{\partial x_{s}} + \left(L\xi^{s} - \dot{x}_{s}(L\xi^{0})\right)$$
(5.2.8)

From (5.2.7), it also follows that coordinates $\tilde{\xi}^k$ are completely determined by means of ξ^0 and ξ^k :

$$\widetilde{\xi}^k = L\xi^k - \dot{x}_k \xi^0. \tag{5.2.9}$$

Now we apply the algorithm stated above to study the symmetry properties of equations of motion of the three-body problem.

$$\ddot{x}_a = \frac{\partial U}{\partial x_a}, \qquad \ddot{y}_a = \frac{\partial U}{\partial y_a}, \qquad \ddot{z}_a = \frac{\partial U}{\partial z_a}, \qquad a = \overline{1,3}$$
 (5.2.10)

with potential

$$U = F(r_{xy}^2) + F(r_{yz}^2) + F(r_{zx}^2)$$
(5.2.11)

where x_a , y_a , z_a are coordinated of the first, second, and third particle; $F(r^2)$ is an arbitrary smooth function;

$$r_{xy}^2 = (\vec{x} - \vec{y})^2, \qquad r_{yz}^2 = (\vec{y} - \vec{z})^2, \qquad r_{zx}^2 = (\vec{z} - \vec{x})^2,$$

Theorem 5.2.1. [155] The system of ODEs (5.2.10) with arbitrary smooth function $F(r^2)$ is invariant under the Galilei group G(1,3), basis elements of corresponding AG(1,3) having the form

$$P_0 = \partial_t, \qquad P_a = \frac{\partial}{\partial x_a} + \frac{\partial}{\partial y_a} + \frac{\partial}{\partial z_a},$$
 (5.2.12)

$$G_a = tP_a, \qquad J_{ab} = x_a \frac{\partial}{\partial x_b} - x_b \frac{\partial}{\partial x_a} + y_a \frac{\partial}{\partial y_b} - y_b \frac{\partial}{\partial y_a} + z_a \frac{\partial}{\partial z_b} - z_b \frac{\partial}{\partial z_a}$$

The proof is not difficult to obtain by means of algorithm described above.

Theorem 5.2.1. determines minimal IA of Equations (5.2.10). It is more or less obvious that depending on special forms of the potential U, the symmetry

of the equations may extend. The following theorem describes the all possible cases of the symmetry extension.

Theorem 5.2.2. [155] The invariance algebra (IA) of Equations (5.2.10), AG(1,3) extends if and only if the potential (5.2.11) has one of the following five forms determined by function $F(r^2)$:

$$F(r^2) = \lambda_1 r^4 + \lambda_2 r^2, \qquad (5.2.13)$$

then $IA = \{AG(1,3), S\},\$

$$S = (y_i - z_i)\frac{\partial}{\partial x_i} + (z_i - x_i)\frac{\partial}{\partial y_i} + (x_i - y_i)\frac{\partial}{\partial z_i}; \qquad (5.2.14)$$

$$F(r^2) = \lambda_3 r^2 \tag{5.2.15}$$

then $IA = \{P_0, P_a, G_a, Y_k^{ab}\}, a, b = \overline{1,3}; k = \overline{1,6},$

$$\begin{split} Y_{1}^{ab} &= v_{a1} \frac{\partial}{\partial v_{b1}} + v_{a2} \frac{\partial}{\partial v_{b2}} + v_{a3} \frac{\partial}{\partial v_{b3}}, \\ Y_{2}^{ab} &= v_{a1} \frac{\partial}{\partial v_{b1}} + v_{a3} \frac{\partial}{\partial v_{b2}} + v_{a2} \frac{\partial}{\partial v_{b3}}, \\ Y_{3}^{ab} &= v_{a2} \frac{\partial}{\partial v_{b1}} + v_{a3} \frac{\partial}{\partial v_{b2}} + v_{a1} \frac{\partial}{\partial v_{b3}}, \\ Y_{4}^{ab} &= v_{a2} \frac{\partial}{\partial v_{b1}} + v_{a1} \frac{\partial}{\partial v_{b2}} + v_{a3} \frac{\partial}{\partial v_{b3}}, \\ Y_{5}^{ab} &= v_{a3} \frac{\partial}{\partial v_{b1}} + v_{a1} \frac{\partial}{\partial v_{b2}} + v_{a2} \frac{\partial}{\partial v_{b3}}, \\ Y_{6}^{ab} &= v_{a3} \frac{\partial}{\partial v_{b1}} + v_{a2} \frac{\partial}{\partial v_{b2}} + v_{a1} \frac{\partial}{\partial v_{b3}}, \\ (v_{a1} \equiv x_{a}, \qquad v_{a2} \equiv y_{a}, \qquad v_{a3} \equiv z_{a},); \end{split}$$

$$F(r^2) = \lambda_4 r^{-2}, \tag{5.2.17}$$

then IA = {AG(1,3), D_1, Π },

$$D_{1} = 2t\partial_{t} + x_{a}\frac{\partial}{\partial x_{a}} + y_{a}\frac{\partial}{\partial y_{a}} + z_{a}\frac{\partial}{\partial z_{a}}$$

$$\Pi = tD_{1} - t^{2}\partial_{t};$$
(5.2.18)

$$F(r^2) = \lambda_5(r^2)^{\beta}, \qquad \beta \neq 0$$
 (5.2.19)

Chapter 5. Some Special Questions

The IA = $\{AG(1,3), D_2\},\$

$$D_2 = (1 - \beta)t\partial_t + x_a\frac{\partial}{\partial x_a} + y_a\frac{\partial}{\partial y_a} + z_a\frac{\partial}{\partial z_a}$$
(5.2.20)

and finally, if

$$F(r^2) = \lambda_6 \ln{(r^2)}$$
 (5.2.21)

then IA = $\{AG(1,3), D_3\},\$

$$D_3 = t\partial_t + x_a \frac{\partial}{\partial x_a} + y_a \frac{\partial}{\partial y_a} + z_a \frac{\partial}{\partial z_a}$$
(5.2.22)

Proof. It is convenient to write IFO (5.2.4) in the form

$$X = \xi \partial t + \eta^{1a} \partial x_a + \eta^{2a} \partial y_a + \eta^{3a} \partial z_a, \qquad (5.2.23)$$

where ξ , η^{1a} , η^{2a} , η^{3a} depend on t, x_1, \ldots, x_3 , provided (x_a, y_a, z_a) denotes coordinates of the *a*th body, a = 1, 2, 3.

In this case, the defining equations (5.2.8) take the form

$$U_{x_{c}}\eta_{x_{c}}^{1a} + U_{y_{c}}\eta_{y_{c}}^{1a} + U_{z_{c}}\eta_{z_{c}}^{1a} - 2U_{x_{a}}\xi_{t} = (\eta^{1a} - \eta^{1b})\dot{F}_{ab} + + (\eta^{1a} - \eta^{1c})\dot{F}_{ac} + (x_{a} - x_{b})\ddot{F}_{ab}X(r_{ab}^{2}) + (x_{a} - x_{b})\ddot{F}_{ac}X(r_{ac}^{2}), U_{x_{c}}\eta_{x_{c}}^{2a} + U_{y_{c}}\eta_{y_{c}}^{2a} + U_{z_{c}}\eta_{z_{c}}^{2a} - 2U_{y_{a}}\xi_{t} = (\eta^{2a} - \eta^{2b})\dot{F}_{ab} + + (\eta^{2a} - \eta^{2c})\dot{F}_{ac} + (y_{a} - y_{b})\ddot{F}_{ab}X(r_{ab}^{2}) + (y_{a} - y_{c})\ddot{F}_{ac}X(r_{ac}^{2}), (5.2.24) U_{x_{c}}\eta_{x_{c}}^{3a} + U_{y_{c}}\eta_{y_{c}}^{3a} + U_{z_{c}}\eta_{z_{c}}^{3a} - 2U_{z_{a}}\xi_{t} = (\eta^{13} - \eta^{3b})\dot{F}_{ab} +$$

$$= + (\eta^{3a} - \eta^{3c})\dot{F}_{ac} + (z_a - z_b)\ddot{F}_{ab}X(r_{ab}^2) + (z_a - z_c)\ddot{F}_{ac}X(r_{ac}^2),$$

$$\begin{aligned} \xi &= b_0 t^2 + \gamma t + d, \\ \eta^{1a} &= b_0 x_a t + a^{1ab} x_b + a^{2ab} y_b + a^{3ab} z_b + a^{a0} t + d^{1a}, \\ \eta^{2a} &= b_0 y_a t + c^{a1b} x_b + c^{a2b} y_b + c^{a3b} z_b + c^{a0} t + d^{2a}, \\ \eta^{3a} &= b_0 z_a t + e^{a1b} x_b + e^{a2b} y_b + e^{a3b} z_b + e^{a0} t + d^{3a}, \end{aligned}$$
(5.2.25)

where

$$\dot{F}_{ab} = \frac{df(r_{ab}^2)}{d(r_{ab}^2)}, \qquad a \neq b \neq c, \qquad a, b, c = 1, 2, 3;$$

$$b_0, \gamma, d, a^{1ab}, a^{2ab}, a^{3ab}, a^{a0}, d^{1a}, d^{2a}, d^{3a},$$

$$c^{a1b}, c^{a2b}, c^{a3b}, c^{a0}, e^{1a}, e^{a2b}, e^{a3b}, e^{a0}$$

are arbitrary real constants (group parameters);

$$X(r_{ab}^2) = 2(x_a - x_b) (\eta^{1a} - \eta^{1b}) + (5.2.26) + 2(y_a - y_b) (\eta^{2a} - \eta^{2b}) + 2(z_a - z_b) (\eta^{3a} - \eta^{3b}).$$

Further analysis of the defining equations (5.2.24), (5.2.25) results in five different cases, namely those listed in the theorem. By this remark, we end the proof.

As a consequence of Theorem 5.2.2, we note that Equations (5.2.10) with the potential

$$U = \lambda_4 \left(\frac{1}{r_{xy}^2} + \frac{1}{r_{yz}^2} + \frac{1}{r_{zx}^2} \right)$$
(5.2.27)

are invariant with respect to the Schrödinger group Sch(1,3). One can make sure that operators of invariance algebra stated in (5.2.12) and (5.2.18) satisfy commutation rules of ASch(1,3) (compare with (4.2.2)):

$$[P_a, P_b] = [P_0, P_a] = [G_a, G_b] = [P_0, J_{ab}] =$$
$$= [P_a, G_b] = [J_{ab}, D] = [J_{ab}, \Pi] = 0$$
$$[P_a, I_b] = \delta \cdot P_b = \delta \cdot P_b$$

$$\begin{split} & [G_a, J_{bc}] = \delta_{ab}G_c - \delta_{ac}G_b, \\ & [P_0, G_a] = P_a, \qquad [P_0, D] = 2P_0, \qquad [P_0, \Pi] = D, \\ & [P_a, D] = P_a, \qquad [P_a, \Pi] = 2G_a, \qquad [D, \Pi] = 2\Pi. \end{split}$$

Let us write down the final transformations generated by (5.2.14). In the new notations they have the form:

$$\begin{aligned} x'_{a} &= \left(x_{a} - \frac{x_{1} + x_{2} + x_{3}}{3}\right) \cos\sqrt{3}\theta + \frac{1}{2\sqrt{3}}\epsilon_{abc}(x_{b} - x_{c}) \sin\sqrt{3}\theta + \frac{x_{1} + x_{2} + x_{3}}{3}, \\ y'_{a} &= \left(y_{a} - \frac{y_{1} + y_{2} + y_{3}}{3}\right) \cos\sqrt{3}\theta + \frac{1}{2\sqrt{3}}\epsilon_{abc}(y_{b} - y_{c}) \sin\sqrt{3}\theta + \frac{y_{1} + y_{2} + y_{3}}{3}, \\ z'_{a} &= \left(z_{a} - \frac{z_{1} + z_{2} + z_{3}}{3}\right) \cos\sqrt{3}\theta + \frac{1}{2\sqrt{3}}\epsilon_{abc}(z_{b} - z_{c}) \sin\sqrt{3}\theta + \frac{z_{1} + z_{2} + z_{3}}{3}, \end{aligned}$$

where θ is an arbitrary real parameter. In this notation operator S takes the form

$$S = \frac{1}{2}\epsilon_{abc} \left[(x_a - x_b) \frac{\partial}{\partial x_c} + (y_a - y_b) \frac{\partial}{\partial y_c} + (z_a - z_b) \frac{\partial}{\partial z_c} \right].$$

2. Now consider integrals of motion of the three-body problem. It is well known that invariance of the Lagrangian

$$\mathcal{L} = \frac{1}{2} \left(\dot{\vec{x}}^2 + \dot{\vec{y}}^2 + \dot{\vec{z}}^2 \right) - U$$
 (5.2.30)

of the three-body problem (5.2.10) with respect to 10-parameter Galilei group G(1,3) means that there are 10 quantities which are constants of motion of the system (also known as first integrals):

$$H = \frac{1}{2} \left(\dot{\vec{x}}^{2} + \dot{\vec{y}}^{2} + \dot{\vec{z}}^{2} \right) + U,$$

$$\mathcal{P}_{a} = \dot{x}_{a} + \dot{y}_{a} + \dot{z}_{a},$$

$$\mathcal{Y}_{a} = x_{a} + y_{a} + z_{a} - t \left(\dot{x}_{a} + \dot{y}_{a} + \dot{z}_{a} \right),$$

$$j_{ab} = \dot{x}_{a} x_{b} - \dot{x}_{b} x_{a} + \dot{y}_{a} y_{b} - \dot{y}_{b} y_{a} + \dot{z}_{a} z_{b} - \dot{z}_{b} z_{a}$$
(5.2.31)

Additional first integrals come into being when function $F(r^2)$ defining the potential U has the form (5.2.13) and (5.2.17). In the first case (that is when $F(r^2)$ is given by (5.2.13)), the Lagrangian (5.2.30) takes the form

$$\mathcal{L}_{4} = \frac{1}{2} \left(\dot{\vec{x}}^{2} + \dot{\vec{y}}^{2} + \dot{\vec{z}}^{2} \right) -$$

$$- \lambda_{1} \left[(r_{xy}^{2})^{2} + (r_{yz}^{2})^{2} + (r_{zx}^{2})^{2} \right] - \lambda_{2} \left(r_{xy}^{2} + r_{yz}^{2} + r_{zx}^{2} \right).$$
(5.2.32)

The equations of motion following from the Lagrangian (5.2.32) possess an additional symmetry operator (5.2.14). This means that there exists an additional conserved quantity. To construct it, one can use the Noether theorem (the condition of the theorem is fulfilled: $S\mathcal{L}_4 = 0$). It yields the 11th first integral

$$\overline{S} = (y_a - z_a)\dot{x}_a + (z_a - x_a)\dot{y}_a + (x_a - y_a)\dot{z}_a,$$
ions
(5.2.33)

or in new notations

$$\overline{S} = \frac{1}{2} \epsilon_{abc} \left[(x_a - x_b) \dot{x}_c + (y_a - y_b) \dot{y}_c + (z_a - z_b) \dot{z}_c \right].$$

This integral was obtained for the first time by Yu. D. Sokolov in [193] for one-dimensional space and then it was extended on flat and three-dimensional space in [48, 110].

In the second case (that is when the function $F(r^2)$ is given by (5.2.17)), the Lagrangian has the form

$$\mathcal{L}_{2} = \frac{1}{2} \left(\dot{\vec{x}}^{2} + \dot{\vec{y}}^{2} + \dot{\vec{z}}^{2} \right) - \lambda_{4} \left(\frac{1}{r_{xy}^{2}} + \frac{1}{r_{yz}^{2}} + \frac{1}{r_{zx}^{2}} \right).$$
(5.2.34)

According to the Theorem (5.2.2) the equations of motion following from the Lagrangian (5.2.34) have two additional symmetry operators D_1 and Π written

in (5.2.18). But since $\Pi \mathcal{L}_2 \neq 0$, we cannot make use of the Noether theorem to construct the corresponding first integral. Nevertheless, the first integral can be constructed in the following way, which holds for any equation of the form (5.2.3). For any symmetry operator Q of a given equation (5.2.3) and for the arbitrary solution ψ of this equation the function

$$\psi_1 = Q\psi \tag{5.2.35}$$

will also have a solution. An analogous statement holds for the conserved quantities. It means, in particular, that the expressions

$$\pi_1 = \Pi H_2 = \left(x_a \dot{x}_a + y_a \dot{y}_a + z_a \dot{z}_a \right) - 2H_2 t, \tag{5.2.36}$$

$$\pi_2 = \Pi \pi_1 = \vec{x}^2 + \vec{y}^2 + \vec{z}^2 - 2H_2t^2 - 2\pi_1t, \qquad (5.2.37)$$

where

$$H_2 = \frac{1}{2} \left(\dot{\vec{x}}^2 + \dot{\vec{y}}^2 + \dot{\vec{z}}^2 \right) + \lambda_4 \left(\frac{1}{r_{xy}^2} + \frac{1}{r_{yz}^2} + \frac{1}{r_{zx}^2} \right),$$

give new constants of motion. Integral π_2 is known as the Jacobi integral. Since

$$\Pi \mathcal{P}_a = \mathcal{G}_a, \qquad \Pi \mathcal{G}_a = \Pi j_{ab} = \Pi \pi_2 = 0$$

we do not obtain any new integrals by this procedure.

When the potential U is determined by (5.2.19) and (5.2.21), the equations of motion (5.2.10) are additionally invariant under operators D_2 (5.2.20) and D_3 (5.2.22) respectively. One finds that

$$D_2 \mathcal{L}_\beta \neq 0, \qquad D_3 \mathcal{L}_0 \neq 0,$$

where \mathcal{L}_{β} and \mathcal{L}_{0} are corresponding Lagrangians. It means that the Noether theorem is impossible to apply here. On the other hand we find

$$\begin{split} D_2 \mathcal{P}_a &= \beta \mathcal{P}_a, \quad D_2 \mathcal{G}_a = \mathcal{G}_a, \quad D_2 H_\beta = 2\beta H_\beta, \\ D_2 j_{ab} &= (1+\beta) j_{ab}, \quad D_2 \pi_1 = 0, \quad D_2 \pi_2 = 2\pi_2 \quad (\beta = -1); \\ D_3 \mathcal{P}_a &= 0, \quad D_3 \mathcal{G}_a = \mathcal{G}_a, \\ D_3 j_{ab} &= j_{ab}, \quad D_3 H_0 = 2\lambda_6; \end{split}$$

where

$$\begin{aligned} H_{\beta} &= \frac{1}{2} \left(\dot{\vec{x}}^2 + \dot{\vec{y}}^2 + \dot{\vec{z}}^2 \right) + \lambda_5 \left[(r_{xy}^2)^{\beta} + (r_{yz}^2)^{\beta} + (r_{zx}^2)^{\beta} \right], \\ H_0 &= \frac{1}{2} \left(\dot{\vec{x}}^2 + \dot{\vec{y}}^2 + \dot{\vec{z}}^2 \right) + \lambda_6 \left[\ln \left(r_{xy}^2 \right) + \ln \left(r_{yz}^2 \right) + \ln \left(r_{zx}^2 \right) \right] \end{aligned}$$

It means that operators (5.2.20) and (5.2.22) do not give new integrals of motion.

Equations (5.2.10) with the potential U defined by (5.2.15), as follows from the theorem (5.2.2), are invariant with respect to the 61-parameter Lie algebra (5.2.16). This case is studied in full detail in [125, 126] and therefore we do not consider it here.

3. In this point we describe some exact solutions of the three-body problem.

The problem of integration of equations of motion of the three-body system was studied by many workers since Newtonian times. Until now only three cases of full integrability of this problem are known, namely the potential of interaction is defined by (5.2.13) and (5.2.17). Some partial solutions of Equations (5.2.10) are obtained for other potentials.

The equations of motion (5.2.10) represent themselves the system of PDEs for 18 unknown functions. The fact that this system possesses 10 first integrals (5.2.31) allows us to reduce the number of unknown functions to 6. When equations of motion possess additional first integrals (they are written in (5.2.33), (5.2.36), (5.2.37)), the number of unknowns is reduced to 4. The reduction is achieved by means of appropriate change of variables.

Consider the three-body system with the potential of interaction given by (5.2.11), (5.2.13). As it was already said in this case, equations of motion have an additional first integral (5.2.33). The one-dimensional case of this problem was fully integrated by Yu. S. Sokolov [194]. The change of variables found in [194] was generalized in the case of flat (two-dimensional) space which resulted in some partial solutions of the corresponding equations of motion. Some exact solutions in three-dimensional space were obtained in [204].

Below we construct some partial solutions of equations of motion (5.2.10) with the potential (5.2.11) defined by (5.2.13).

For the sake of convenience we combine the center-of-mass system with the reference system, that is, we put

$$x_a + y_a + z_a = 0. (5.2.38)$$

Further, we make the following change of variables

$$3x_{k} = S_{1} \cos\left(\alpha_{1} + \frac{2\pi(k-1)}{3}\right) + S_{2} \cos\left(\alpha_{2} - \frac{2\pi(k-1)}{3}\right) + S_{3} \cos\left(\frac{\alpha_{1} - \alpha_{2}}{2} + \frac{2\pi(k-1)}{3}\right) + S_{4} \sin\left(\frac{\alpha_{1} - \alpha_{2}}{2} + \frac{2\pi(k-1)}{3}\right),$$

$$3y_k = S_1 \cos\left(\alpha_1 + \frac{2\pi k}{3}\right) + S_2 \cos\left(\alpha_2 - \frac{2\pi (k+1)}{3}\right) +$$
(5.2.39)

$$+S_3\cos\left(\frac{\alpha_1-\alpha_2}{2}+\frac{2\pi(k-1)}{3}\right)+S_4\sin\left(\frac{\alpha_1-\alpha_2}{2}+\frac{2\pi(k-1)}{3}\right),$$

$$3z_{k} = S_{1} \cos\left(\alpha_{1} + \frac{2\pi(k+1)}{3}\right) + S_{2} \cos\left(\alpha_{2} - \frac{2\pi k}{3}\right) + S_{3} \cos\left(\frac{\alpha_{1} - \alpha_{2}}{2} + \frac{2\pi(k-1)}{3}\right) + S_{4} \sin\left(\frac{\alpha_{1} - \alpha_{2}}{2} + \frac{2\pi(k-1)}{3}\right),$$

$$k = \overline{1, 3}$$

In the two-dimensional case an analogous change of variables was used in [48].

In new variables expressions of kinetic and potential energy take the following form

$$T = \frac{1}{2}(\dot{\vec{x}}^2 + \dot{\vec{y}}^2 + \dot{\vec{z}}^2) = \frac{1}{4}[\dot{S}_1^2 + \dot{S}_2^2 + \dot{S}_3^2 + \dot{S}_4^2 + (5.2.40)$$

$$+ (S_{1}\dot{\alpha}_{1})^{2} + (S_{2}\dot{\alpha}_{2})^{2} + (S_{3}^{2} + S_{4}^{2})\left(\frac{\dot{\alpha}_{1} - \dot{\alpha}_{2}}{2}\right)^{2} + (\dot{S}_{3}S_{4} - S_{3}\dot{S}_{4})(\dot{\alpha}_{1} - \dot{\alpha}_{2})\Big],$$

$$U = \lambda_{1}\left(r_{12}^{4} + r_{23}^{4} + r_{31}^{4}\right) + \lambda_{2}\left(r_{12}^{2} + r_{23}^{2} + r_{31}^{2}\right) =$$

$$= \frac{3}{2}\lambda_{2}\left(S_{1}^{2} + S_{2}^{2} + S_{3}^{2} + S_{4}^{2}\right) + \frac{3}{4}\lambda_{1}\left[\left(S_{1}^{2} + S_{2}^{2} + S_{3}^{2} + S_{4}^{2}\right)^{2} + (5.2.41) + \frac{1}{2}\left(S_{4}^{2} - S_{3}^{2} - 2S_{1}S_{2}\right)^{2} + 2\left(S_{3}S_{4}\right)^{2}\right].$$

Equations of motion in new variables are written as follows

$$\begin{split} \frac{1}{2}\ddot{S}_{1} &- \frac{1}{2}S_{1}\dot{\alpha}_{1}^{2} = -\frac{\partial U}{\partial S_{1}}, \\ \frac{1}{2}\ddot{S}_{2} &- \frac{1}{2}S_{2}\dot{\alpha}_{2}^{2} = -\frac{\partial U}{\partial S_{2}}, \end{split} (5.2.42) \\ \frac{d}{dt} \left[\frac{1}{2}\dot{S}_{3} + \frac{1}{8}S_{4}(\dot{\alpha}_{1} - \dot{\alpha}_{2}) \right] &- \frac{1}{2}S_{3} \left(\frac{\dot{\alpha}_{1} - \dot{\alpha}_{2}}{2} \right)^{2} + \frac{1}{8}\dot{S}_{4}(\dot{\alpha}_{1} - \dot{\alpha}_{2}) = -\frac{\partial U}{\partial S_{3}}, \\ \frac{d}{dt} \left[\frac{1}{2}\dot{S}_{4} + \frac{1}{8}S_{3}(\dot{\alpha}_{1} - \dot{\alpha}_{2}) \right] &- \frac{1}{2}S_{4} \left(\frac{\dot{\alpha}_{1} - \dot{\alpha}_{2}}{2} \right)^{2} + \frac{1}{8}\dot{S}_{3}(\dot{\alpha}_{2} - \dot{\alpha}_{1}) = -\frac{\partial U}{\partial S_{4}}, \\ \frac{d}{dt} \left[\dot{S}_{1}^{2}\dot{\alpha}_{1} - \frac{1}{8}(S_{3}^{2} + S_{4}^{2})(\dot{\alpha}_{1} - \dot{\alpha}_{2}) + \frac{1}{2}(\dot{S}_{3}S_{4} - S_{3}\dot{S}_{4}) \right] = 0, \\ \frac{d}{dt} \left[\dot{S}_{2}^{2}\dot{\alpha}_{2} + \frac{1}{8}(S_{3}^{2} + S_{4}^{2})(\dot{\alpha}_{2} - \dot{\alpha}_{1}) + \frac{1}{2}(S_{3}\dot{S}_{4} - \dot{S}_{3}S_{4}) \right] = 0. \end{split}$$

Integrals of motion \mathcal{P}_a and \mathcal{G}_a in new variables become zero.

Let S_1 , S_2 , S_3 , S_4 , $\dot{\alpha}_1$, $\dot{\alpha}_2$ be constant, then the system (5.2.42) is reduced to the system of algebraic equations

Chapter 5. Some Special Questions

$$S_1^2 \dot{\alpha}_1 + \frac{1}{4} \left(S_3^2 + S_4^2 \right) (\dot{\alpha}_2 - \dot{\alpha}_2) = c_1,$$

$$S_1^2 \dot{\alpha}_1 + \frac{1}{4} \left(S_3^2 + S_4^2 \right) (\dot{\alpha}_2 - \dot{\alpha}_2) = c_1,$$
(5.2.43)

where c_1 and c_2 are constants of integration.

Further, we put $S_1 = S_2$, $S_3 = S_4$ (and therefore $\dot{\alpha}_1 = \dot{\alpha}_2$, $c_1 = c_2$). After this suggestion, the system (5.2.43) takes the form

$$\dot{\alpha}_1^2 = \lambda_1 \left[6(S_1^2 + S_2^2) + \frac{3}{4}S_1^2 \right] + 3\lambda_2 \lambda_1 \left(\frac{5}{2} + 3S_3^2 \right) + \lambda_2, \qquad \dot{\alpha}S_1^2 = c_1$$
(5.2.44)

The system (5.2.44) has a solution when $\lambda_1 \lambda_2 < 0$. Having expressed S_3^2 and $\dot{\alpha}_1$ from the second and third equation of the system (5.2.44) via S_1^2 , we get cubic equation with respect to S_1^2

$$\dot{\alpha}_1 = \frac{c_1}{S_1^2}, \qquad S_3^2 = \frac{\lambda_2}{3\lambda_1} - \frac{5}{6}S_1^2,$$

$$\frac{7}{4}\lambda_1(S_1^2)^3 + \lambda_2S_1^2 - c_1^2 = 0.$$
(5.2.45)

Using Cardano formula, it is not difficult to obtain the solution of Equation (5.2.45)

$$S_{1} = \left\{ \frac{4}{7\lambda_{1}} \left[\frac{c_{1}^{2}}{2} - \left(\frac{\lambda_{2}}{3} \right)^{3} \left(\frac{4}{7\lambda_{1}} \right)^{2} \right] + \frac{4}{7} \left| \frac{c_{1}}{\lambda_{1}} \right| \left[\frac{c_{1}}{4} - \left(\frac{\lambda_{2}}{3} \right)^{3} \left(\frac{4}{7\lambda_{1}} \right)^{2} \right]^{1/2} \right\}^{1/3} + \left\{ \frac{4}{7\lambda_{1}} \left[\frac{c_{1}}{2} - \left(\frac{\lambda_{2}}{3} \right)^{3} \left(\frac{4}{7\lambda_{1}} \right)^{2} \right] - \frac{4}{7} \left| \frac{c_{1}}{\lambda_{1}} \right| \left[\frac{c_{1}^{2}}{4} - \left(\frac{\lambda_{2}}{3} \right)^{3} \left(\frac{4}{7\lambda_{1}} \right)^{2} \right]^{1/2} \right\}^{1/3} - \frac{4}{21} \frac{\lambda_{2}}{\lambda_{1}},$$

$$\frac{c_1^2}{4} \ge \left(\frac{\lambda_2}{3}\right)^3 \left(\frac{4}{7\lambda_1}\right)^2 \tag{5.2.46}$$

From (5.2.45), we find

$$\alpha_1 = \frac{c_1}{S_1^2}t + c_3, \qquad \alpha_2 = \frac{c_1}{S_1^2}t + c_4,
S_3^2 = S_4^2 = \frac{\lambda_2}{3\lambda_1} - \frac{5}{6}S_1^2,$$
(5.2.47)

where c_1, c_3, c_4 are constants of integration.

Formulae (5.2.46), (5.2.47), and (5.2.39) give a solution of equations of motion (5.2.10) of the three-body system with the potential (5.2.11), (5.2.13).

5.3 Non-Lie symmetry and nonlocal transformations

The infinitesimal Lie method is far from providing the possibility of finding all symmetries which a system of differential equations possesses. A familiar example of a "non-Lie" symmetry is the invariance of the Schrödinger equation for the hydrogen atom under the group 0(4) first discovered by Fock [53].

Let us consider an arbitrary linear system of PDEs

$$L(x,\partial)\psi(x) = 0, \qquad (5.3.1)$$

where $L(x,\partial)$ is a linear operator, ψ is a multi-component function with components $\{\psi^1, \psi^2, \dots, \psi^m\}, x \in \mathbb{R}^n, \partial = \left\{\frac{\partial}{\partial x_{\mu}}\right\} \mu = \overline{0, n-1}.$

In Lie's approach, the infinitesimal operators of the invariance algebra are sought in the form of first-order differential operators (2). Taking into account that the manifold of solutions of linear system (5.3.1) is also linear, we can simplify the general form of the symmetry operators (2) as follows

$$X = \xi^{\mu}(x)\frac{\partial}{\partial x^{\mu}} - (\eta(x)\psi)^{k}\frac{\partial}{\partial \psi^{k}} \Leftrightarrow Q = \xi^{\mu}(x)\frac{\partial}{\partial x^{\mu}} + \eta(x), \qquad (5.3.2)$$

where $\eta(x)$ are matrices of dimension $m \times m$. Here we used different notations for the same operator to emphasize that operator Q acts in the linear m-dimensional space $\{\psi(x)\}$ while the operator X acts in linear (m + n)-dimensional space $\{x, \psi\}$.

By means of operator Q (5.3.2), the invariance condition of the system (5.3.1) can be written in the form (9), (10):

$$[L,Q]\psi(x)\Big|_{L\psi=0} = 0.$$
 (5.3.3)

It is obvious that this condition does not impose any restrictions on the order of generators Q. So the formulation of the problem of investigating symmetry properties of the system (5.3.1) can be considerable generalized by extending the class of the desired operators Q. For example, it is possible to seek IA in a class of second-order differential operators or even of integrodifferential operators. It will be emphasized that the condition of invariance (5.3.3) does not guarantee for the set of invariance operators Q of second and higher orders to form a Lie algebra as it was for the first-order differential operators in Lie approach, and what is more operators Q can form, for example, superalgebra or possess other specific algebraic properties. If we desire operators Q to form a Lie algebra, we must also require the following relations to hold:

$$[Q_A, Q_B] = C_{ABC}Q_C, \qquad (5.3.4)$$

where C_{ABC} are structure constants. A set of operators $\{Q_A\}$ satisfying condition (5.3.) and (5.3.4) is called an invariance algebra (IA) of the system (5.3.1). This non-Lie algebraic approach was suggested in 1971 in [57] and since then new IA of the Dirac equations. Maxwell's equations and many equations of quantum mechanics have been found in just this manner. A largely complete review of these results the reader will find in [82,87*].

Below we consider how to find final transformations generated by non-Lie symmetry operators. It is clear that these transformations cannot be local; naturally they are nonlocal.

An arbitrary differential operator of any order can be written as follows:

$$Q \equiv Q(x,\partial) = \xi^{\mu}(x)\partial_{\mu} + \eta(x,\partial), \qquad (5.3.5)$$

where $\xi^{\mu}(x)$ are scalar functions, $\eta(x,\partial)$ is the matrix-differential part of Q does not contain terms like $\xi^{\mu}(x)\partial_{\mu}$.

Theorem 5.3.1. [99] If operator Q (5.3.5) satisfies condition of invariance (5.3.3) on the manifold of solutions of the system (5.3.1), then the system (5.3.1) is invariant under one-parameter group of transformations.

$$\begin{aligned} x_{\mu} \to x'_{\mu} &= \exp\{\theta\xi\partial\}x_{\mu}\exp\{-\theta\xi\partial\},\\ \psi(x) \to \psi(x') &= \exp\{\theta\xi\partial\}\exp\{-\thetaQ\}\psi(x), \end{aligned} \tag{5.3.6}$$

where θ is an arbitrary real constant (group parameter), generated by this operator Q.

Proof. As a result of transformations $x_{\mu} \to x'_{\mu}$ from (5.3.6), any differential operator constructed from x_{μ} and ∂_{μ} , and of course, the operator of the system (5.3.1), $L(x, \partial)$ transforms according to the law

$$L(x,\partial) \to L(x',\partial') = \exp\{\theta\xi\partial\}L(x,\partial)\exp\{-\theta\xi\partial\}.$$

then using the expression for $\psi(x)$ from (5.3.6) and taking into account (5.3.3) we find

$$\begin{split} L(x',\partial')\psi'(x') &= \exp\{\theta\xi\partial\}L(x,\partial)\exp\{-\theta\xi\partial\}\cdot\\ &\cdot \exp\{\theta\xi\partial\}\exp\{-\thetaQ\}\psi(x) = \exp\{\theta\xi\partial\}L(x,\partial)\exp\{-\thetaQ\}\psi(x) =\\ &= \exp\{\theta\xi\partial\}L(x,\partial)\left(1-\thetaQ+\frac{(\thetaQ)^2}{2!}\mp\cdots\right)\psi(x) = 0. \end{split}$$

It proves the first part of the theorem, that is, the system (5.3.1) is indeed invariant with respect to transformations (5.3.6). And now we convince ourselves that transformations (5.3.6) form a one-parameter group. We find

$$\begin{aligned} x''_{\mu} &= \left(\exp\{\beta\xi\partial\}\right)' x'_{\mu} \left(\exp\{-\beta\xi\partial\}\right)' = \\ &= \exp\{\theta\xi\partial\} \exp\{\beta\xi\partial\} \exp\{-\theta\xi\partial\} \exp\{\theta\xi\partial\} \exp\{\theta\xi\partial\} x_{\mu} \cdot \\ &\cdot \exp\{-\theta\xi\partial\} \exp\{\theta\xi\partial\} \exp\{-\beta\xi\partial\} \exp\{-\theta\xi\partial\} = \\ &= \exp\{(\theta+\beta)\xi\partial\} x_{\mu} \exp\{-(\theta+\beta)\xi\partial\}, \end{aligned}$$

$$\begin{split} \psi''(x'') &= \left(\exp\{\beta\xi\partial\}\exp\{-\beta Q\}\right)'\psi'(x') = \\ &= \exp\{\theta\xi\partial\}\exp\{\beta\xi\partial\}\exp\{-\beta Q\}\exp\{-\theta\xi\partial\}\exp\{\theta\xi\partial\}\exp\{-\theta Q\}\psi(x) = \\ &= \exp\{(\theta+\beta)\xi\partial\}\exp\{-(\theta+\beta)\xi\partial\}\psi(x), \end{split}$$

which completes the proof.

Remark 5.3.1. If operator Q satisfying invariance condition (5.3.2) belongs to the class of Lie-type operators (5.3.2), then expressions (5.3.6) give a formal solution to the Lie equations.

$$\frac{dx'_{\mu}}{d\theta} = \xi^{\mu}(x'), \qquad x'_{\mu} \Big|_{\theta=0} = x_{\mu},$$

$$\frac{d\psi'(x')}{d\theta} = -\eta(x')\psi'(x'), \qquad \psi'(x') \Big|_{\theta=0} \equiv \psi(x).$$
(5.3.7)

Indeed, by substituting (5.3.6) into (5.3.7), we obtain identities

$$x'_{\mu}\Big|_{\theta=0}\equiv x_{\mu}, \qquad \psi'(x')\Big|_{\theta=0}\equiv \psi(x),$$

$$\begin{aligned} \frac{dx'_{\mu}}{d\theta} &= \exp\{\theta\xi\partial\}[\xi\partial, x_{\mu}]\exp\{-\theta\xi\partial\} = \exp\{\theta\xi\partial\}\xi^{\mu}(x)\exp\{-\theta\xi\partial\} \equiv \xi^{\mu}(x'),\\ \frac{d\psi'(x')}{d\theta} &= \exp\{\theta\xi\partial\}(\xi\partial - Q)\exp\{-\thetaQ\}\psi(x) = -\exp\{\theta\xi\partial\}\eta(x)\exp\{-\theta\xi\partial\}.\end{aligned}$$

 $\cdot \exp\{\theta\xi\partial\}\exp\{-\theta Q\}\psi(x)\equiv -\eta(x')\psi(x').$

If operator Q does not belong to the class of Lie's operators (5.3.2), it generates nonlocal transformations. As a simple example of non-Lie operator, let us consider the operator

$$Q=\frac{\partial^2}{\partial x^2}.$$

According to (5.3.6), we have

1

$$x = x,$$

$$\psi'(x') = \exp\left\{\theta \frac{\partial^2}{\partial x^2}\right\}\psi(x) = \frac{1}{\sqrt{4\pi\theta}} \int_{-\infty}^{\infty} \exp\left\{-\frac{(x-y)^2}{4\theta}\right\}\psi(y)dy.$$

This result follows by noting that the $F(\theta, x) \equiv \psi'(x')$ satisfies the Cauchy problem

$$\frac{\partial F}{\partial \theta} = \frac{\partial^2 F}{\partial x^2}, \qquad F(\theta = 0, x) = \psi(x),$$

and, therefore, we may use the well-known expression for the fundamental solution of the Cauchy problem for the heat equation

$$F(heta, x) \equiv \psi'(x') = rac{1}{\sqrt{4\pi heta}} \int\limits_{-\infty}^{\infty} \exp\left\{-rac{(x-y)^2}{4 heta}
ight\}\psi(y)dy$$

To obtain functional form of final transformations (5.3.6), it is necessary to do additional calculations connected with expansion of operational exponents. So, we further consider this question.

Let \mathcal{A} be an algebra with the following properties: for any elements \hat{x} , \hat{y} from \mathcal{A} their commutator $[\hat{x}, \hat{y}] = \hat{x}\hat{y} - \hat{y}\hat{x}$ belongs to \mathcal{A} , and the sum of a infinite convergent series with elements belonging to \mathcal{A} also belongs to \mathcal{A} . In what follows we will use hat as the notation of elements of algebra \mathcal{A} .

Theorem 5.3.2. For any \hat{x} , $\hat{y} \in \mathcal{A}$ the following expansion holds (the Campbell-Baker-Hausdorff (CBH) formula)

$$\exp\{-\theta \widehat{x}\}\widehat{w}\exp\{\theta \widehat{x}\} = \sum_{n=0}^{\infty} \frac{\theta^n}{n!} \{\widehat{w}, \widehat{x}^n\},$$
(5.3.8)

where

$$\{\widehat{w}, \widehat{x}^{0}\} \equiv \widehat{w}, \{\widehat{w}, \widehat{x}\} \equiv [\widehat{w}, \widehat{x}] \equiv \widehat{w}\widehat{x} - \widehat{x}\widehat{w}, \{\widehat{w}, \widehat{x}^{k}\} \equiv [\{\widehat{w}, \widehat{x}^{k-1}\}, \widehat{x}], \qquad k = 1, 2, \dots;$$

$$(5.3.9)$$

 θ is an arbitrary real constant.

Proof. Consider the function of θ

$$F(\theta) = \exp\{-\theta \widehat{x}\}\widehat{w}\exp\{\theta \widehat{x}\}$$
(5.3.10)

Let us expand it in Maclaurin's series. But at first we find

$$\begin{split} \frac{\partial F}{\partial \theta} &= \exp\{-\theta \widehat{x}\} [\widehat{w}, \widehat{x}] \exp\{\theta \widehat{x}\} \\ \frac{\partial^2 F}{\partial \theta^2} &= \exp\{-\theta \widehat{x}\} \left(-\widehat{x} [\widehat{w}, \widehat{x}] + [\widehat{w}, \widehat{x}] \widehat{x}\right) \exp\{\theta \widehat{x}\} \\ &= \exp\{-\theta \widehat{x}\} [\widehat{w}, \widehat{x}^2] \exp\{\theta \widehat{x}\} \end{split}$$

whence follows

$$F(\theta) = \sum_{n=0}^{\infty} \frac{\theta^n}{n!} \left(\frac{\partial^n F}{\partial \theta^n} \right) \bigg|_{\theta=0} = \widehat{w} + \frac{\theta}{1!} \{ \widehat{w}, \widehat{x} \} + \frac{\theta^2}{2!} \{ \widehat{w}, \widehat{x}^2 \} + \ldots = \sum_{n=0}^{\infty} \frac{\theta^n}{n!} \{ \widehat{w}, \widehat{x}^n \}.$$

The theorem is proved.

Theorem 5.3.3. For any $\hat{x}, \hat{y} \in A$, the following identity holds (the formula of Hausdorff)

$$e^{\widehat{x}}e^{\widehat{y}} = e^{\widehat{z}}, \qquad \widehat{z} \in \mathcal{A},$$
 (5.3.11)

where

$$\widehat{z} = \sum_{n=0}^{\infty} \widehat{z}_{n},
\widehat{z}_{0} = \widehat{y}
\widehat{z}_{1} = \sum_{n=0}^{\infty} \frac{(-1)^{n}}{n!} B_{n}\{\widehat{x}, \widehat{y}^{n}\} = \widehat{x} + \frac{1}{2}[\widehat{x}, \widehat{y}] + \sum_{k=1}^{\infty} \frac{B_{2k}}{(2k)!} \{\widehat{x}, \widehat{y}^{2k}\},
\widehat{z}_{2} = \frac{1}{2} (\widehat{z}_{1} \partial_{\widehat{y}}) \widehat{z}_{1}, \dots, \quad \widehat{z}_{k} = \frac{1}{k} (\widehat{z}_{1} \partial_{\widehat{y}}) \widehat{z}_{k-1}, \quad k = 1, 2, \dots,$$
(5.3.12)

where B_n are the Bernouilli numbers:

$$B_0 = 1, \qquad B_1 = -\frac{1}{2}, \qquad B_2 = \frac{1}{6},$$

 $B_3 = B_5 = \dots = B_{2k+1} = \dots = 0,$ (5.3.13)

Chapter 5. Some Special Questions

$$B_4 = -\frac{1}{30}, \qquad B_6 = \frac{1}{42}, \dots;$$

 $\hat{z}_1 \partial_{\hat{y}}$ is the so-called Hausdorff operator which operates in the following manner: for example,

$$(\hat{z}_1\partial_{\hat{y}})F\hat{y}^2G\hat{y} = F\hat{z}_1\hat{y}G\hat{y} + F\hat{y}\hat{z}_1G\hat{y} + F\hat{y}^2G\hat{z}_1, \qquad (5.3.14)$$

where F and G are operator-valued functions independent of \hat{y} .

Before proceeding to the proof of this theorem, we will first prove three lemmas.

Lemma 5.3.1. For smooth operator-valued function $F(\hat{x})$, the analog of Taylor expansion holds true

$$F(\widehat{x} + \widehat{u}) = F(\widehat{x}) + (\widehat{u}\partial_{\widehat{x}})F(x) + \frac{1}{2}(\widehat{u}\partial_{\widehat{x}})^2F(\widehat{x}) + \cdots$$
(5.3.15)

Proof. Since in both parts of equality (5.3.15), function $F(\hat{x})$ comes linearly, it is sufficient to prove the lemma for the function \hat{x}^n . For n = 2 we have

$$(\widehat{x} + \widehat{u})^2 \equiv (\widehat{x} + \widehat{u})(\widehat{x} + \widehat{u}) = \widehat{x}^2 + \widehat{x}\widehat{u} + \widehat{u}\widehat{x} + \widehat{u}^2$$

From the other hand, if $F(\hat{x}) = \hat{x}^2$, we find the same result from (5.3.15)

$$(\widehat{x}+\widehat{u})^2 = \widehat{x}^2 + (\widehat{u}\partial_{\widehat{x}})\widehat{x}^2 + \frac{1}{2!}(\widehat{u}\partial_{\widehat{x}})^2\widehat{x}^2 = \widehat{x}^2 + \widehat{u}\widehat{x} + \widehat{x}\widehat{u} + \widehat{u}^2.$$

To complete the proof, one has to apply further the method of mathematical induction.

Remark 5.3.2. Analogous expansion can be written in two-dimensional case

$$F(\hat{x} + \hat{u}, \hat{y} + \hat{v}) = F(\hat{x}, \hat{y}) + \left(\hat{u}\partial_{\hat{x}} + \hat{v}\partial_{\hat{y}}\right)F(\hat{x}, \hat{y}) + \frac{1}{2!}\left(\hat{u}\partial_{\hat{x}} + \hat{v}\partial_{\hat{y}}\right)^{2}F(\hat{x}, \hat{y}) + \cdots$$
(5.3.16)

Lemma 5.3.2. In algebra A, the following identities are valid

$$\left(\widehat{u}\partial_{\widehat{x}}\right)e^{\widehat{x}} = e^{\widehat{x}}\varphi\left(\widehat{u},\widehat{x}\right), \qquad (5.3.17)$$

$$\left(\widehat{u}\partial_{\widehat{x}}\right)e^{\widehat{x}} = \psi(\widehat{u},\widehat{x})e^{\widehat{x}}, \qquad (5.3.18)$$

where

$$\varphi(\hat{u},\hat{x}) = \hat{u} + \frac{1}{2!}\{\hat{u},\hat{x}\} + \frac{1}{3!}\{\hat{u},\hat{x}^2\} + \ldots = \sum_{n=0}^{\infty} \frac{\{\hat{u},\hat{x}^n\}}{(n+1)!} \stackrel{\text{def}}{=} \hat{p}, \quad (5.3.19)$$

$$\psi(\widehat{u},\widehat{x}) = \varphi(\widehat{u},-\widehat{x}) = \sum_{n=0}^{\infty} \frac{(-1)^n}{(n+1)!} \{\widehat{u},\widehat{x}^n\} \stackrel{\text{def}}{=} \widehat{q}.$$
 (5.3.20)

Proof. It follows as a trivial consequence of (5.3.15) that

$$\left(\widehat{u}\partial_{\widehat{x}}\right)e^{\widehat{x}} = \widehat{u} + \frac{1}{2!}(\widehat{u}\widehat{x} + \widehat{x}\widehat{u}) + \frac{1}{3!}(\widehat{u}\widehat{x}^2 + \widehat{x}\widehat{u}\widehat{x} + \widehat{x}^2\widehat{u}) + \cdots$$

Setting in this expansion

$$\widehat{u} = [\widehat{w}, \widehat{x}] \tag{5.3.21}$$

we find

$$([\widehat{w},\widehat{x}]\partial_{\widehat{x}}) e^{\widehat{x}} = [\widehat{w},\widehat{x}] + \frac{1}{2!} [\widehat{w},\widehat{x}^2] + \frac{1}{3!} [\widehat{w},\widehat{x}^3] + \dots = [\widehat{w},e^{\widehat{x}}]$$
(5.3.22)

Further, we use CBH formula (5.3.8) under $\theta = 1$

$$[\widehat{w}, e^{\widehat{x}}] \equiv \widehat{w}e^{\widehat{x}} - e^{\widehat{x}}\widehat{w} \equiv e^{\widehat{x}}\left(e^{-\widehat{x}}\widehat{w}e^{\widehat{x}} - \widehat{w}\right) = e^{\widehat{x}}\left(\{\widehat{w}, \widehat{x}\} + \frac{1}{2!}\{\widehat{w}, \widehat{x}^2\} + \cdots\right)$$

It yields together with (5.3.21), (5.3.22) formulae (5.3.17), (5.3.19):

$$\left(\widehat{u}\partial_{\widehat{x}}\right)e^{\widehat{x}} = e^{\widehat{x}}\left(\widehat{u} + \frac{1}{2!}\{\widehat{u},\widehat{x}\} + \frac{1}{3!}\{\widehat{u},\widehat{x}^2\} + \cdots\right) = e^{\widehat{x}}\varphi(\widehat{u},\widehat{x}).$$

By noting that

$$\begin{split} [\widehat{w}, e^{\widehat{x}}] &\equiv \widehat{w}e^{\widehat{x}} - e^{\widehat{x}}\widehat{w} \equiv \left(\widehat{w} - e^{-\widehat{x}}\widehat{w}e^{\widehat{x}}\right)e^{\widehat{x}} = \\ &= \left(\{\widehat{w}, \widehat{x}\} - \frac{1}{2!}\{\widehat{w}, \widehat{x}^2\} + \frac{1}{3!}\{\widehat{u}, \widehat{x}^2\} \mp \cdots\right)e^{\widehat{x}} \end{split}$$

one can easily show the validity of (5.3.18), (5.3.20).

Lemma 5.3.3. The series (5.3.19) is reversible, the reversion having the form

$$\widehat{u} = \sum_{n=0}^{\infty} \frac{B_n}{n!} \{ \widehat{p}, \widehat{x}^n \},$$
(5.3.23)

where B_n are Bernoulli numbers (5.3.13).

Proof. Starting from (5.3.19), we can successively write

$$\begin{aligned} \widehat{p} &= \widehat{u} + \frac{1}{2!} \{ \widehat{u}, \widehat{x} \} + \frac{1}{3!} \{ \widehat{u}, \widehat{x}^2 \} + \dots + \frac{1}{(n+1)!} \{ \widehat{u}, \widehat{x}^n \} + \dots \\ \{ \widehat{p}, \widehat{x} \} &= \{ \widehat{u}, \widehat{x} \} + \frac{1}{2!} \{ \widehat{u}, \widehat{x}^2 \} + \dots + \frac{1}{n!} \{ \widehat{u}, \widehat{x}^n \} + \dots \end{aligned}$$

$$\begin{aligned} \{ \widehat{p}, \widehat{x}^2 \} &= \{ \widehat{u}, \widehat{x}^2 \} + \frac{1}{2!} \{ \widehat{u}, \widehat{x}^3 \} + \dots + \frac{1}{(n-1)!} \{ \widehat{u}, \widehat{x}^n \} + \dots \\ \{ \widehat{p}, \widehat{x}^{n-1} \} &= \{ \widehat{u}, \widehat{x}^{n-1} \} + \frac{1}{2!} \{ \widehat{u}, \widehat{x}^n \} + \dots \\ \{ \widehat{p}, \widehat{x}^n \} &= \{ \widehat{u}, \widehat{x}^n \} + \dots \end{aligned}$$

$$(5.3.24)$$

Now we make the sum $\sum_{n=0}^{\infty} b_n\{\hat{p}, \hat{x}^n\}$ and require the coefficients $b_n (n = 0, 1, ...; b_0 = 1)$ to satisfy the relations

$$\frac{1}{(n+1)!} + \frac{b_1}{n!} + \dots + \frac{b_{n-1}}{2!} + b_n = 0.$$
 (5.3.25)

All terms in the right-hand side of our sum vanish except the addend u. Thus we get

$$\widehat{u} = \sum_{n=0}^{\infty} b_n \{ \widehat{p}, \widehat{x}^n \}.$$
(5.3.26)

Equations (5.3.25) should be considered as recurrent relations for determination of coefficients b_1, b_2, \ldots These relations can be written as a single identity

$$\left(\sum_{n=0}^{\infty} \frac{t^n}{(n+1)!}\right) \left(\sum_{n=0}^{\infty} b_n t^n\right) = 1.$$

But as it is well known

$$\sum_{n=0}^{\infty} \frac{t^n}{(n+1)!} = \frac{e^t - 1}{t}.$$

Therefore, numbers b_n are just the coefficients of the expansion of the function $t/(e^t - 1)$ in a power series. So we have

$$\sum_{n=0}^{\infty} b_n t^n = \frac{t}{e^t - 1} = \sum_{n=0}^{\infty} \frac{B_n}{n!} t^n,$$

 $(B_n \text{ are Bernouilli numbers } (5.3.13))$ whence follows

$$b_n = \frac{B_n}{n!}.\tag{5.3.27}$$
From (5.3.26) and (5.3.27) one obtains the formula (5.3.23).

Remark 5.3.3. Similarly one can reverse the series (5.3.20). But since $\hat{q} = \varphi(\hat{u}, -\hat{x})$ we get from (5.3.23)

$$\widehat{u} = \sum_{n=0}^{\infty} \frac{(-1)^n B_n}{n!} \{ \widehat{q}, \widehat{x}^n \}.$$
(5.3.28)

Now we are ready to prove Theorem 5.3.3. Let \hat{x} is changed to $\hat{x} + \theta \hat{u}$ and \hat{y} is changed to $\hat{y} - \theta \hat{v}$, where \hat{u} and \hat{v} are arbitrary, in such a way that

$$\widehat{z}(\widehat{x} + \theta \widehat{u}, \widehat{y} - \theta \widehat{v}) = \widehat{z}(\widehat{x}, \widehat{y})$$
(5.3.29)

Then according to Lemma 5.3.1 using (5.3.16) we get identity

$$\left(\widehat{u}\partial_{\widehat{x}}\right)\widehat{z} = \left(\widehat{v}\partial_{\widehat{y}}\right)\widehat{z}.$$
(5.3.30)

The condition (5.3.29) means that

$$e^{\widehat{x}+\theta\widehat{u}}e^{\widehat{y}-\theta\widehat{v}} = \left(e^{\widehat{x}}+\theta\left(\widehat{u}\partial_{\widehat{x}}\right)e^{\widehat{x}}+\cdots\right)\left(e^{\widehat{y}}-\theta\left(\widehat{v}\partial_{\widehat{y}}\right)e^{\widehat{y}}+\cdots\right) = e^{\widehat{x}}e^{\widehat{y}}$$

And together with Lemmas 5.3.1 and 5.3.2, it yields

$$\varphi(\widehat{u}, \widehat{x}) = \psi(\widehat{v}, \widehat{y}) \tag{5.3.31}$$

Since \hat{u} is arbitrary, we may choose it equal to \hat{x} , from which we may find

$$(\widehat{x}\partial_{\widehat{x}})\,\widehat{z} = \left(\widehat{v}\partial_{\widehat{y}}\right)\widehat{z}.$$
 (5.3.32)

and

$$\varphi(\hat{x}, \hat{x}) \equiv \hat{x} = \psi(\hat{v}, \hat{y}). \tag{5.3.33}$$

Having used formulae (5.3.20), (5.3.28) we obtain

$$\widehat{v} = \sum_{n=0}^{\infty} \frac{(-1)^n B_n}{n!} \{\widehat{x}, \widehat{y}^n\}$$
(5.3.34)

Let us expand \hat{z} as a series

$$\hat{z} = \hat{z}_0 + \hat{z}_1 + \hat{z}_2 + \cdots,$$
 (5.3.35)

where \hat{z}_k is a polynomial containing k factors of \hat{x} . Since \hat{v} contains \hat{x} only to the first power (see (5.3.34)), we can rewrite equality (5.3.32) as

$$\widehat{z}_1 + 2\widehat{z}_2 + 3\widehat{z}_3 + \cdots = \left(\widehat{v}\partial_{\widehat{y}}\right)(\widehat{z}_0 + \widehat{z}_1 + \widehat{z}_2 + \cdots)$$

and after equating coefficients of the terms with equal power of \hat{x} , we get identities

$$\widehat{z}_1 = \left(\widehat{v}\partial_{\widehat{y}}\right)\widehat{z}_0, \qquad 2\widehat{z}_2 = \left(\widehat{v}\partial_{\widehat{y}}\right)\widehat{z}_1, \qquad 3\widehat{z}_3 = \left(\widehat{v}\partial_{\widehat{y}}\right)\widehat{z}_2, \cdots.$$

To find \hat{z}_0 we set in (5.3.35) and in (5.3.11) $\hat{x} = O$. It immediately yields

$$\widehat{z}\Big|_{\widehat{x}=0}=\widehat{y}=\widehat{z}_0.$$

Thus Theorem 5.3.3 is proved.

Remark 5.3.4. In the same way, using the substitution $\hat{v} = \hat{y}$, we find

$$\left(\widehat{y}\partial_{\widehat{y}}\right)\widehat{z} = \left(\widehat{u}\partial_{\widehat{x}}\right)\widehat{z}, \qquad \psi(\widehat{y},\widehat{y}) \equiv \widehat{y} = \varphi(\widehat{u},\widehat{x}). \tag{5.3.36}$$

Using (5.3.19), (5.3.23) we obtain another equivalent expression for \hat{z} :

$$\hat{z} = \hat{x} + \hat{r}_1 + \hat{r}_2 + \cdots,$$

$$\hat{r}_1 = \sum_{n=0}^{\infty} \frac{B_n}{n!} \{ \hat{y}, \hat{x}^n \} = \hat{y} - \frac{1}{2} [\hat{y}, \hat{x}] + \sum_{k=1}^{\infty} \frac{B_{2k}}{(2k)!} \{ \hat{y}, \hat{x}^{2k} \},$$

$$\hat{r}_2 = \frac{1}{2} (\hat{r}_1 \partial_{\hat{x}}) \hat{r}_1, \qquad \hat{r}_3 = \frac{1}{3} (\hat{r}_1 \partial_{\hat{x}}) \hat{r}_2, \dots.$$
(5.3.37)

Consequence 5.3.1. When operators \hat{x} and \hat{y} satisfy commutation relations

$$\left[\widehat{x}, \left[\widehat{x}, \widehat{y}\right]\right] = \left[\widehat{y}, \left[\widehat{x}, \widehat{y}\right]\right] = 0, \qquad (5.3.38)$$

formulae (5.3.11)-(5.3.13), (5.3.37) result in the well-known identity

$$e^{\widehat{x}}e^{\widehat{y}} = e^{\widehat{x}+\widehat{y}+\frac{1}{2}[\widehat{x},\widehat{y}]},\tag{5.3.39}$$

which is widely used in quantum mechanics and in the theory of coherent states [151,164].

Remark 5.3.5. The problem of finding an explicit expression for $\hat{z} = \ln(e^{\hat{x}}e^{\hat{y}})$ was first attacked by Campbell in 1898 and was soon thereafter followed by investigations of Baker and of Hausdorff. The author last named found an expression for \hat{z} in terms of repeated commutators of \hat{x} and \hat{y} [117]. Hausdorff results were expounded in [38,146,189,206,208] and here we used these works. More recently Dynkin [45] found explicit form of coefficients of the commutator series for \hat{z} but this result was quite unwieldy and therefore is not used in applications, while the Hausdorff formulae (5.3.11), (5.3.12), (5.3.37) give, as we see it, an effective way of finding \hat{z} .

Below we consider an example of using the Hausdorff formula to find nonlocal transformations of invariance of the Dirac equation.

$$(i\gamma_{\nu}\partial^{\nu} - m)\psi(x) = 0 \tag{5.3.40}$$

where m is a constant (mass of particle), γ_{ν} are Dirac matrices written in (2.1.2).

As it is well known the Dirac equation (5.3.40) is invariant with respect to the Poincare group P(1,3) and this group is the maximal one in the sense of Lie (here we do not consider transformations which mix up ψ and $\overline{\psi}$). But the Poincare group does not exhaust all symmetry properties of the Dirac equation. It was shown in [64] by means of non-Lie method the duality of the spacetime symmetry of the Dirac equation. This duality is stipulated by the existence of two IA: the known one AP(1,3) with basis elements (2.1.11)), and the algebra AF(1,3) with basis elements

$$P_{0} = i\partial_{0}, \qquad \mathcal{P}_{0} = H = \gamma_{0}\gamma_{a}P_{a} + \gamma_{0}m, \qquad P_{a} = -i\partial_{a},$$

$$J_{ab} = x_{a}P_{b} - x_{a}P_{a} + S_{ab} \equiv \epsilon_{abc}J_{c},$$

$$R_{a} = x_{0}P_{a} - \frac{1}{2}(\mathcal{P}_{0}x_{a} + x_{a}\mathcal{P}_{0}),$$
(5.3.41)

which satisfy commutation rules

$$\begin{aligned} [\mathcal{P}_{0}, P_{0}] &= [\mathcal{P}_{0}, P_{a}] = [P_{a}, P_{b}] = [P_{0}, P_{a}] = [\mathcal{P}_{0}, R_{a}] = 0, \\ [P_{0}, R_{a}] &= iP_{a}, \qquad [P_{a}, R_{b}] = i\delta_{ab}\mathcal{P}_{0}, \\ [J_{a}, J_{b}] &= i\epsilon_{abc}J_{c}, \qquad [J_{a}, R_{b}] = -i\epsilon_{abc}R_{c}, \\ [R_{a}, R_{b}] &= -i\epsilon_{abc}J_{c}. \end{aligned}$$

$$(5.3.42)$$

As operators \mathcal{P}_0 and R_a are of non-Lie type, to find their group action, we shall use the formulae (5.3.6).

Theorem 5.3.4 [99, 189]. The Dirac equation (5.3.40) is invariant under Galilean transformations

$$x'_{0} = x_{0}, \qquad x'_{1} = x_{1} + \theta x_{0}, \qquad x'_{2} = x_{2}, \qquad x'_{3} = x_{3},$$

$$\psi'(x') = \exp\left\{i\theta\left[\left(1 - \frac{1}{2}\theta \operatorname{cth} \frac{1}{2}\theta\right)x_{0}P_{1} + \frac{1}{2}(Hx_{1} + x_{1}H + \theta x_{0}H)\right]\right\}\psi(x).$$

(5.3.43)

generated by the operator R_1 from AF(1, 3) (5.3.41).

Proof. At first we convince ourselves that operator

$$R_1 = x_0 P_1 - \frac{1}{2} (Hx_1 + x_1 H) = -i \left(x_0 \partial_1 - \frac{1}{2} i (Hx_1 + x_1 H) \right)$$
(5.3.44)

satisfies the invariance condition (5.3.3), and to do it we rewrite Equation (5.3.40) into the equivalent form

$$L\psi \equiv (P_0 - H)\psi = 0, \qquad H = \gamma_0 \gamma_a P_a + \gamma_0 m. \tag{5.3.45}$$

It is easy to find

$$[L, R_1] = iP_1 + \frac{1}{2} [H^2, x_1]$$

and using the identities

$$H^2 \equiv (\gamma_0 \gamma_a P_a + \gamma_0 m)^2 = P_a^2 + m^2, \qquad [H^2, x_1] = -2iP_1$$

we get

$$[L, R_1] = 0$$

So, the invariance condition (5.3.3) is satisfied and one can use formulae (5.3.6) for finding the group action of the operator (5.3.44). According to (5.3.6) we have

$$x'_{\mu} = e^{\theta x_0 \partial_1} x_{\mu} e^{-\theta x_0 \partial_1}, \qquad (5.3.46)$$

$$\psi'(x') = \exp\{\theta x_0 \partial_1\} \exp\{-\theta[x_0 \partial_1 - \frac{1}{2}i(Hx_1 + x_1 H)]\} \psi(x).$$
 (5.3.47)

To obtain the functional form of transformations (5.3.46), (5.3.47) we apply formulae (5.3.8), (5.3.11)-(5.3.13).

From (5.3.46) and (5.3.8) we find

$$x'_{\mu} = x_{\mu} - \theta[x_{\mu}, x_0 \partial_1] = x_{\mu} + \theta x_0 \delta_{\mu 1}.$$
 (5.3.48)

To use expansion (5.3.11) let us introduce notations

$$\widehat{x} = \theta x_0 \partial_1, \qquad \widehat{y} = -\widehat{x} + \frac{1}{2}i\theta(Hx_1 + x_1H) \tag{5.3.49}$$

Then we successively calculate

 $\{\widehat{x},\widehat{y}^{2n}\}=\theta^{2n}\widehat{x}, \qquad n=1,2,\ldots,$

whence follows that the series for \hat{z}_1 from (5.3.12) takes the form

$$\widehat{z}_{1} = \widehat{x} + \frac{1}{2} [\widehat{x}, \widehat{y}] + \sum_{k=1}^{\infty} \frac{B_{2k}}{(2k)!} \{\widehat{x}, \widehat{y}^{2k}\} =$$

$$= \frac{1}{2} i \theta^{2} x_{0} H + \left(\frac{\theta}{2} \operatorname{cth} \frac{\theta}{2}\right) \widehat{x}, \qquad |\theta| < 2\pi.$$
(5.3.50)

Here we used the known expansion

$$\frac{\theta}{2}\operatorname{cth}\frac{\theta}{2} = \sum_{k=0}^{\infty} \frac{B_{2k}}{(2k)!} \theta^{2k}, \qquad |\theta| < 2\pi.$$
(5.3.51)

To find the second term in the expansion of (5.3.12) we note that

$$\begin{split} &(\hat{z}_{1} \ \partial_{\hat{y}})\{\hat{x}, \hat{y}\} = \{\hat{x}, \hat{z}_{1}\} = 0, \\ &(\hat{z}_{1} \ \partial_{\hat{y}})\{\hat{x}, \hat{y}^{2}\} = \left[[\hat{x}, \hat{z}_{1}] \hat{y} \right] + \left[[\hat{x}, \hat{y}] \hat{z}_{1} \right] = 0, \\ &(\hat{z}_{1} \ \partial_{\hat{y}})\{\hat{x}, \hat{y}^{3}\} = \theta(\hat{z}_{1}, \partial_{\hat{y}})\{\hat{x}, \hat{y}\} = 0, \end{split}$$

and therefore

$$\widehat{z}_2 = \frac{1}{2} \left(\widehat{z}_1 \partial_{\widehat{y}} \right) \widehat{z}_1 = \frac{1}{2} \widehat{z}_1 \partial_{\widehat{y}} \left(\widehat{x} + \frac{1}{2} [\widehat{x}, \widehat{y}] + \sum_{k=0}^{\infty} \frac{B_{2k}}{(2k)!} \{ \widehat{x}, \widehat{y}^{2k} \} \right) = 0$$

and hence $\hat{z}_3 = \hat{z}_4 = \cdots = 0$. Finally we obtain the following expression for \hat{z} :

$$\widehat{z} = \widehat{y} + \widehat{z}_1 = \widehat{y} + rac{i}{2} heta^2 x_0 H + \left(rac{ heta}{2}\operatorname{cth}rac{ heta}{2}
ight)\widehat{x}, \qquad | heta| < 2\pi,$$

or in the terms of the original notations (5.3.49)

$$\widehat{z} = \left(\frac{\theta}{2}\operatorname{cth}\frac{\theta}{2} - 1\right)\theta x_0\partial_1 + \frac{1}{2}i\theta(Hx_1 + x_1H) + \frac{1}{2}i\theta^2 x_0H.$$

So the theorem is proved.

Remark 5.3.6. One can make sure in the validity of Theorem 5.3.4 straightforwardly. Indeed, under the Galilei transformations (5.3.48) we have

$$P'_{0} = i\frac{\partial}{\partial x'_{0}} = P_{0} + \theta P_{1}, \qquad P'_{a} = -i\frac{\partial}{\partial x'_{a}} = P_{a}.$$
(5.3.52)

For $\psi'(x')$ (5.3.43) we find

$$\psi'(x') = \psi(x) + \frac{1}{2}i\theta(Hx_1 + x_1H)\psi(x) + \cdots$$
(5.3.53)

After substitution of (5.3.52), (5.3.53) into the primed Equation (5.3.40)

$$(P_0'-H')\psi'(x')=0$$

and retaining terms linear in θ , we successively obtain

Chapter 5. Some Special Questions

$$(P_0 + \theta P_1) \left(\psi + \frac{1}{2} i \theta (Hx_1 + x_1 H) \psi \right) - H\psi - \frac{1}{2} i \theta H (Hx_1 + x_1 H) \psi = 0,$$

$$(P_0 - H)\psi + \theta \left[\frac{1}{2} i (Hx_1 + x_1 H) P_0 \psi + P_1 \psi - \frac{1}{2} i (H^2 x_1 + Hx_1 H) \psi \right] = 0,$$

$$(P_0 - H)\psi + \theta \left[\frac{1}{2} i (Hx_1 P_0 + x_1 H P_0 - H^2 x_1 - Hx_1 H) + P_1 \psi \right] = 0.$$

Since on the manifold of solutions of the Dirac equation (5.3.40), we have $P_0\psi = H\psi$ then the latter equality we can write as

$$\left[\frac{1}{2}i(x_1H^2 - H^2x_1) + P_1\right]\psi \equiv 0,$$

which proves our statement.

Remark 5.3.7. As follows from the proof of Theorem 5.3.4 the formula (5.3.43) for $\psi'(x')$ is valid when $|\theta| < 2\pi$. This restriction is stipulated by the range of convergence of the series (5.3.51). But for real values of the parameter, the series (5.3.51) is convergent everywhere in R. This means that there is no restriction on the value of the parameter which can be treated as the velocity of an inertial reference frame, while in the case of the Lorentz transformation a restriction exists: the speed of light.

Remark 5.3.8. Theorem 5.3.4 can be easily generalized on equations of arbitrary spin. These equations should have the form

$$P_0\psi = H\psi \tag{5.3.54}$$

and the Hamiltonian H must satisfy the condition

$$[H^2, x_a] = -2iP_a. \tag{5.3.55}$$

(A wide class of such equations for particles of arbitrary spin is described in $[76,82,87^*]$.)

Formulae (5.3.43) are written in the form suitable for any equation of the type (5.3.54), (5.3.55).

In conclusion, we note that some other applications of CBH and Hausdorff formulae are given in Sections 5.4, 5.5, 5.8-5.10 and in Appendix 3.

5.4 Lie-Backlund symmetry of the Dirac equation

Following [75, 82, 211] we shall study the symmetry properties of the Dirac equation (5.3.40) in the class of the first-order matrix-differential operators. The investigation will be done in two different approaches: non-Lie and Lie-Backlund.

Theorem 5.4.1 [74,82]. The Dirac equation (5.3.40) is invariant under an eight-dimensional Lie algebra defined over the field of real numbers. The basis elements of this algebra have the form

$$\widehat{\Sigma}_{\mu\nu} = \frac{1}{2} im[\gamma_{\mu}, \gamma_{\nu}] + i(1 - i\gamma_{4})(\gamma_{\mu}P_{\nu} - \gamma_{\nu}P_{\mu}),
\widehat{\Sigma}_{0} = I, \qquad \widehat{\Sigma}_{1} = \gamma_{4}m - i(1 - i\gamma_{4})\gamma_{\nu}P^{\nu},$$
(5.4.1)

where $\gamma_4 = \gamma_0 \gamma_1 \gamma_2 \gamma_3$ and I is a unit matrix. In the case $m \neq 0$ the algebra (5.4.1) is isomorphic to the Lie algebra of the group U(2) \otimes U(2), while for m = O the operators (5.4.1) form an abelian algebra.

Proof. The validity of the theorem can most simply be seen by direct verification. One has

$$\begin{split} \left[\widehat{\Sigma}_{\mu\nu}, L\right] &= i(\gamma_4 P_\nu - \gamma_\nu P_\mu)L, \qquad (L \equiv \gamma_\nu P^\nu - m), \\ \left[\widehat{\Sigma}_1, L\right] &= -i2\gamma_4 \gamma_\nu P^\nu L, \qquad \left[\widehat{\Sigma}_0, L\right] = 0, \\ \left[\widehat{\Sigma}_{\mu\nu}, \widehat{\Sigma}_{\lambda\sigma}\right] &= 2im^2 \left(g_{\mu\sigma}\widehat{\Sigma}_{\nu\lambda} + g_{\nu\lambda}\widehat{\Sigma}_{\mu\sigma} - g_{\mu\lambda}\widehat{\Sigma}_{\nu\sigma} - g_{\nu\sigma}\widehat{\Sigma}_{\mu\lambda}\right), \\ \left[\widehat{\Sigma}_1, \widehat{\Sigma}_0\right] &= \left[\widehat{\Sigma}_1, \widehat{\Sigma}_{\mu\nu}\right] = \left[\widehat{\Sigma}_0, \widehat{\Sigma}_{\mu\nu}\right] = 0, \end{split}$$
(5.4.2)

whence it is evident that operators (5.4.1) satisfy the invariance condition (10).

According to (5.4.2), in the case m = 0 the operators (5.4.1) commute. If $m \neq 0$ then, setting

$$\Sigma_{\mu\nu} = \frac{1}{m} \widehat{\Sigma}_{\mu\nu}, \qquad \Sigma_0 = \widehat{\Sigma}_0, \qquad \Sigma_1 = \frac{1}{m} \widehat{\Sigma}_1 \tag{5.4.3}$$

from (5.4.2) we find that operators $\Sigma_{\mu\nu}, \Sigma_0, \Sigma_1$ form the algebra

$$[\Sigma_{\mu\nu}, \Sigma_{\lambda\sigma}] = 2i \left(g_{\mu\sigma} \Sigma_{\nu\lambda} + g_{\nu\lambda} \Sigma_{\mu\sigma} - g_{\mu\lambda} \Sigma_{\nu\sigma} - g_{\nu\sigma} \Sigma_{\mu\lambda} \right),$$
(5.4.4)
$$[\Sigma_1, \Sigma_0] = [\Sigma_1, \Sigma_{\mu\nu}] = [\Sigma_0, \Sigma_{\mu\nu}] = 0.$$

The algebra (5.4.4) is isomorphic to the Lie algebra of the group $U(2) \otimes U(2)$. This isomorphism can be established by the following relations:

$$\begin{split} \Sigma_{ab} &\leftrightarrow i\epsilon_{abc}Q_c, \qquad \Sigma_{0a} \leftrightarrow iQ_{3+a}, \qquad \Sigma_0 \leftrightarrow iQ_7, \quad \Sigma_1 \leftrightarrow iQ_8; \\ [Q_a, Q_b] &= -[Q_{3+a}, Q_{3+b}] = -\epsilon_{abc}Q_c \\ [Q_{3+a}, Q_b] &= \epsilon_{abc}Q_{3+c} \qquad [Q_7, Q_8] = [Q_8, Q_A] = 0 \qquad A = \overline{1,8}. \end{split}$$

Over the field of real numbers all elements of the algebra (5.4.3) are linearly independent. In order to see this, it suffices to subject the operators (5.4.3) to the transformation

$$\begin{split} \Sigma_{\mu\nu} &\to \Sigma'_{\mu\nu} = V \Sigma_{\mu\nu} V^{-1} = \frac{1}{2} i [\gamma_{\mu}, \gamma_{\nu}], \\ \Sigma_0 &\to \Sigma'_0 = V \Sigma_0 V^{-1} = 1 \qquad \Sigma_1 \to \Sigma'_1 = V \Sigma_1 V^{-1} = \gamma_4, \end{split}$$

where

$$V = \exp\left\{-\frac{1}{2m}(1-i\gamma_4)\gamma_{\nu}P^{\nu}\right\} \equiv 1 - \frac{1}{2m}(1-i\gamma_4)\gamma_{\nu}P^{\nu}.$$

The theorem is proved.

Since operators (5.4.1) do not belong to the class of first-order differential operators of the form (7), they correspond to nonlocal symmetry of the Dirac equation (5.3.40). In order to find the group action of the operators (5.4.1) we shall use formulae (5.3.6). Since $\xi^{\mu}(x) = 0$ for all operators (5.4.1) it means that $x'_{\mu} = x_{\mu}$. For operator $\theta_a \Sigma_{0a}$ (θ_a are arbitrary constants) one successively finds

$$\begin{split} \psi'(x') &= \exp\left\{-i\Sigma_{0a}\theta_{a}\right\}\psi(x) = \\ &= \left[\sum_{k=0}^{\infty}\frac{1}{(2k)!}\left(i\Sigma_{0a}\theta_{a}\right)^{2k} + \sum_{k=0}^{\infty}\frac{1}{(2k+1)!}\left(-i\Sigma_{0a}\theta_{a}\right)^{2k+1}\right]\psi(x) = \quad (5.4.5) \\ &= \left[\sum_{k=0}^{\infty}\frac{\theta^{2k}}{(2k)!} + \frac{1}{\theta}\sum_{k=0}^{\infty}\frac{\theta^{2k+1}}{(2k+1)!}\left(\gamma_{0}\gamma_{a} + \frac{1}{m}(1-i\gamma_{4})(\gamma_{0}P_{a} - \gamma_{a}P_{0})\right)\theta_{a}\right]\psi(x) = \\ &= \left(\operatorname{ch}\theta + \frac{\operatorname{sh}\theta}{\theta}\gamma_{0}\gamma_{a}\theta_{a}\right)\psi(x) - \frac{i}{\theta m}\operatorname{sh}\theta(1-i\gamma_{4})\left(\gamma_{0}\frac{\partial\psi}{\partial x_{a}} + \gamma_{a}\frac{\partial\psi}{\partial x_{0}}\right)\theta_{a}\psi(x), \end{split}$$

where $\theta \equiv \left(\theta_1^2 + \theta_2^2 + \theta_3^2\right)^{1/2}$. Here we have used the identity

$$\left(i\Sigma_{0a}\theta_{a}\right)^{2}=\theta^{2}.$$

Similarly one obtains

$$\psi'(x') = \exp\left\{-\frac{1}{2}i\epsilon_{abc}\Sigma_{ab}\beta_{c}\right\}\psi(x) =$$

$$= \left[\cos\beta - \frac{1}{2\beta}\sin\beta\epsilon_{abc}\gamma_{a}\gamma_{b}\beta_{c} + \frac{1}{m\beta}\sin\beta(1 - i\gamma_{4})\epsilon_{abc}\gamma_{a}P_{b}\beta_{c}\right]\psi(x),$$

$$\psi'(x') = \exp\left\{-\Sigma_{1}\alpha\right\}\psi(x) = \left[\cosh\alpha + i\gamma_{4}\sin\alpha + \frac{1}{m}\sin\alpha(1 - i\gamma_{4})\gamma_{\nu}P^{\nu}\right]\psi(x),$$
(5.4.6)

where β_a , α are arbitrary constants, $\beta = (\beta_1^2 + \beta_2^2 + \beta_3^2)^{1/2}$.

The principal difference of the transformations (5.4.5), (5.4.6) from Lorentz transformations for the Dirac spinor $\psi(x)$ is that the function $\psi'(x')$ depends not only on $\psi(x)$ (and parameters of the transformation) but also on the derivatives $\partial \psi / \partial x_{\mu}$.

2. We will show that operators (5.4.1) can be obtained within the framework of Lie-Backlund approach.

As is known [161], the Lie-Backlund symmetry operators are looked for in the form

$$X = \eta^{\sigma} \left(x, \psi^{\alpha}, \psi^{\alpha}_{\mu_{1}}, \psi^{\alpha}_{\mu_{1}\mu_{2}}, \ldots \right) \frac{\partial}{\partial \psi^{\sigma}} + \sum_{\substack{0 \le \mu_{k} \le 3, \\ k \ge 1}} \left(D_{\mu_{1}} D_{\mu_{2}} \dots D_{\mu_{k}} \right) \eta^{\sigma} \frac{\partial}{\partial \psi^{\sigma}_{\mu_{1}\mu_{2}\dots\mu_{k}}},$$
(5.4.7)

where

$$D_{\mu} = \frac{\partial}{\partial x^{\mu}} + \psi^{\sigma}_{\mu} \frac{\partial}{\partial \psi^{\sigma}} + \dots + \psi^{\sigma}_{\mu\mu_{1}\dots\mu_{k}} \frac{\partial}{\partial \psi^{\sigma}_{\mu_{1}\dots\mu_{k}}} + \dots,$$
$$\psi^{\sigma}_{\mu} \equiv \frac{\partial \psi^{\sigma}}{\partial x^{\mu}}, \qquad \psi^{\sigma}_{\mu\nu} \equiv \frac{\partial^{2}\psi^{\sigma}}{\partial x^{\mu}\partial x^{\nu}}, \dots.$$

The problem of finding Lie-Backlund symmetry is the problem of finding functions $\eta(x, \psi^{\alpha}, \psi^{\alpha}_{\mu}, ...)$. The functions η^{σ} are found from the condition of invariance.

$$X[L] \Big|_{[L]=0} = 0, \tag{5.4.8}$$

where X is given in (5.4.7); [L] = 0 denotes the manifold of solutions of the equation and all its differential consequences. It is rather difficult (if not impossible) to solve the problem in such general formulation without any restrictions on the coordinates η^{σ} of IFO (5.4.7). The needed calculations are enormous even in the simplest cases. Besides that one has to study additionally the algebraic properties of the found operators, which is rather difficult by itself.

Below we apply the Lie-Backlund method to the study of symmetry properties of the Dirac equation (5.3.40). Following [211] we rewrite it as eightcomponent system of PDEs

$$i\gamma_{\mu}\frac{\partial\psi}{\partial x_{\mu}} - m\psi = 0,$$

$$i\frac{\partial\overline{\psi}}{\partial x_{\mu}}\gamma_{\mu} + m\overline{\psi} = 0,$$

(5.4.9)

or in equivalent form

$$i\Gamma_{\mu}\partial^{\mu}\phi + m\phi = 0,$$

$$\Gamma_{4}\Gamma_{5}\phi + \Gamma_{2}\phi^{*} = 0,$$
(5.4.10)

where $\phi = \phi(x)$ is an eight-component spinor, $\Gamma_{\mu} = \begin{pmatrix} \gamma_{\mu} & 0 \\ 0 & \gamma_{\mu} \end{pmatrix}$,

$$\Gamma_4 = \begin{pmatrix} 0 & \gamma_4 \\ \gamma_\mu & 0 \end{pmatrix}, \quad \Gamma_5 = i \begin{pmatrix} 0 & \gamma_4 \\ -\gamma_\mu & 0 \end{pmatrix}, \quad \Gamma_6 = \begin{pmatrix} \gamma_4 & 0 \\ 0 & -\gamma_4 \end{pmatrix}$$
(5.4.11)

We look for the Lie-Backlund symmetry operators (5.4.7) setting $\eta = A_{\mu}\phi_{\mu} + F(\phi)$, A_{μ} are 8-column constant matrices.

Theorem 5.4.2 [211]. The basis of Lie-Backlund symmetry operators (5.4.7) under $\eta = A_{\mu}\phi_{\mu} + F(\phi)$ of the Dirac equation (5.4.10) is given by the following operators

$$I = \phi, \qquad \hat{Q}_{a} = Q_{a}\phi$$

$$P_{\mu}^{(0)} = \phi_{\mu}, \qquad P_{\mu}^{(a)} = Q_{a}P_{\mu}^{(0)};$$

$$\Sigma_{\mu\nu}^{(0)} = i(I - i\tilde{\Gamma}_{4})(\Gamma_{\mu}\phi_{\nu} - \Gamma_{\nu}\phi_{\mu}) - 2imG_{\mu\nu}\phi,$$

$$\Sigma_{\mu\nu}^{(a)} = Q_{a}\Sigma_{\mu\nu}^{(0)};$$

$$\Sigma_{\mu}^{(0)} = \tilde{\Gamma}_{4}\phi_{\mu} + im\Gamma_{\mu}\Gamma_{4}\phi, \qquad \Sigma_{\mu}^{(a)} = Q_{a}\Sigma_{\mu}^{(0)},$$
(5.4.12)

where

 $\Gamma_4\Gamma_5\eta+\Gamma_2\eta^*=0,$

$$\begin{aligned} Q_1 &= \Gamma_4 \Gamma_5, \qquad Q_2 = i \Gamma_5 \Gamma_6, \qquad Q_3 = i \Gamma_4 \Gamma_6, \qquad \widetilde{\Gamma}_4 = \Gamma_0 \Gamma_1 \Gamma_2 \Gamma_3, \\ G_{\mu\nu} &= \frac{1}{4} (\Gamma_{\mu} \Gamma_{\nu} - \Gamma_{\nu} \Gamma_{\mu}). \end{aligned}$$

Proof. Since the calculations are involved in the present case, we shall state them in a sketchy manner, for the sake of brevity. From the condition of invariance (5.4.8) with $\eta^{\sigma} = (A_{\mu}\phi_{\mu})^{\sigma} + F^{\sigma}(\phi)$ we get the defining equations

$$\left(\Gamma_{\mu}B\phi_{\mu}+\Gamma_{\mu}A_{\nu}\phi_{\mu\nu}-im\eta\right)\Big|_{[i\Gamma_{\mu}\phi_{\mu}+m\phi=0]}=0,$$
(5.4.13)

where $B = \left(B^{\alpha}_{\beta} \equiv \partial \eta^{\alpha} / \partial \phi^{\beta}; \alpha, \beta = \overline{0,7}\right); [i\Gamma_{\mu}\phi_{\mu} + m\phi = 0]$ is the totality of all differential consequences of Equation (5.4.10). Taking into account differential consequences we obtain from (5.4.13)

$$\Gamma_0 B \left(-\Gamma_0 \Gamma_a \phi_a + im \Gamma_0 \phi \right) + \Gamma_a B \phi_a + \Gamma_0 A_0 (\phi_{aa} - m^2 \phi) + + (\Gamma_0 A_a + \Gamma_a A_0) (-\Gamma_0 \Gamma_b \phi_{ba} + im \Gamma_0 \phi_a) + \Gamma_0 A_b \phi_{ab} - im \eta = 0, \qquad (5.4.14)$$

$$\Gamma_4 \Gamma_5 \eta + \Gamma_2 \eta^* = 0,$$

Decomposing the first equation of (5.4.14) into independent variables ϕ_{ab} , one finds the defining equations for matrices A_{μ} :

$$\begin{split} &-\Gamma_0 A_a \Gamma_0 \Gamma_b - \Gamma_a A_0 \Gamma_0 \Gamma_b + \Gamma_a A_b + \Gamma_b \Gamma_a - (\Gamma_0 A_b + \Gamma_b A_0) \Gamma_0 \Gamma_a = 0, \qquad a \neq b \\ &\Gamma_0 A_0 - (\Gamma_0 A_a + \Gamma_a A_0) + \Gamma_a A_a = 0, \, (\text{no sum over } a), \end{split}$$

whence follows

$$A_{\mu} = \begin{pmatrix} A_{\mu}^{(1)} & 0\\ 0 & 0 \end{pmatrix} + \begin{pmatrix} 0 & A_{\mu}^{(2)}\\ 0 & 0 \end{pmatrix} + \begin{pmatrix} 0 & 0\\ A_{\mu}^{(3)} & 0 \end{pmatrix} + \begin{pmatrix} 0 & 0\\ 0 & A_{\mu}^{(4)} \end{pmatrix},$$

where

$$\begin{split} A^{(k)}_{\mu} &= a^{\mu(k)}I + b^{\mu(k)}_{\alpha}\gamma^{\alpha} + C^{\mu(k)}_{\alpha\beta}\sigma^{\alpha\beta} + d^{\mu(k)}_{\alpha}\gamma_{4}\gamma^{\alpha} + e^{\mu(k)}\gamma_{4}, \\ b^{(k)}_{0} &= -b^{a(k)}_{a}, \qquad b^{\mu(k)}_{\nu} = -b^{\nu(k)}_{\mu}, \qquad \mu \neq \nu \\ d^{0(k)}_{0} &= -d^{a(k)}_{a}, \qquad b^{\mu(k)}_{\nu} = -d^{\nu(k)}_{\mu}, \qquad \mu \neq \nu \\ c^{a(k)}_{ab} &= -c^{0(k)}_{0b}, \qquad c^{a(k)}_{0b} = -c^{0(k)}_{ab}, \qquad c^{a(k)}_{bc} = -c^{b(k)}_{ac}, \\ \sigma^{\alpha\beta} &= \frac{1}{4}(\gamma^{\alpha}\gamma^{\beta} - \gamma^{\beta}\gamma^{\alpha}). \end{split}$$

Here there is no sum over a. It leads to (5.4.12).

By substituting this result into (5.4.13) it is cumbersome but not difficult to convince ourselves that the operators (5.4.12) exhaust the set of linearly independent solutions of the defining Equations (5.4.13). So, the theorem is proved.

In much the same way one can prove another statement.

Theorem 5.4.3 [211]. The basis of Lie-Backlund symmetry operators (5.4.7) under $\eta = A_{\mu}\phi_{\mu} + F(\phi)$ of the Dirac equation (5.4.10) with m = 0 is given by the following operators

$$I = \phi, \qquad \widehat{Q}_{a} = Q_{a}\phi$$
(5.4.12')

$$P_{\mu}^{(0)} = \phi_{\mu}, \qquad P_{\mu}^{(a)} = Q_{a}\phi_{\mu};$$

$$\Sigma_{\mu\nu}^{(0)} = i(I - \widetilde{\Gamma}_{4})(\Gamma_{\mu}\phi_{\nu} - \Gamma_{\nu}\phi_{\mu}), \qquad \Sigma_{\mu\nu}^{(a)} = Q_{a}\Sigma_{\mu\nu}^{(0)}$$

$$\Sigma_{\mu}^{(0)} = \widetilde{\Gamma}_{4}\phi_{\mu}, \qquad \Sigma_{\mu}^{(a)} = Q_{a}\Sigma_{\mu}^{(0)},$$

$$\Sigma^{(0)} = \widetilde{\Gamma}_{4}\phi, \qquad \Sigma^{(a)} = Q_{a}\Sigma^{(0)}.$$

Remark 5.4.1. It is easy to show that the sets of operators (5.4.12), (5.4.12') contain operators (5.4.1) and (5.4.1) with m = 0, respectively.

3. Following [81] we will show that any relativistic wave equation for a particle with non-zero mass and arbitrary spin is additionally invariant under the Lie algebra of the group GL(2s + 1, C).

Let us write an arbitrary linear (differential or integrodifferential) equation

$$L\psi = 0 \tag{5.4.15}$$

where L is a linear operator defined on a vector space \mathcal{H} ,

$$\psi = \psi(x) \in \mathcal{H}.$$

Definition 5.4.1. Equation (5.4.15) is Poincare invariant and describes a particle of mass m and spin s if it has the 10 symmetry operators P_{μ} , $J_{\mu\nu}$, $\mu, \nu = \overline{0,3}$, which form the basis of the Lie algebra of the Poincare group, and any solution ψ satisfies the conditions

$$P_{\mu}P^{\mu}\psi = m^{2}\psi, \qquad W_{\mu}W^{\mu}\psi = -m^{2}s(s+1)\psi, \qquad (5.4.16)$$

where W_{μ} is the Lubansky-Pauli vector

$$W_{\mu} = \frac{1}{2} \epsilon_{\mu\nu\rho\sigma} J^{\nu\rho} P^{\sigma}. \tag{5.4.17}$$

We consider only such equations as (5.4.15) which satisfy the given definition and so may be interpreted as equations for relativistic particle of spin s and mass m. The symmetry operators P_{μ} , $J_{\mu\nu}$ of such an equation satisfy the commutation relations of AP(1,3).

$$[P_{\mu}, P_{\nu}] = 0, \qquad [P_{\mu}, J_{\nu\sigma}] = i(g_{\mu\nu}P_{\sigma} - g_{\mu\sigma}P_{\nu}), \tag{5.4.18}$$
$$[J_{\mu\nu}, J_{\lambda\sigma}] = i(g_{\mu\sigma}J_{\nu\lambda} + g_{\nu\lambda}J_{\mu\sigma} - g_{\mu\lambda}J_{\nu\sigma} - g_{\nu\sigma}J_{\mu\lambda}).$$

The eigenvalues of the corresponding Casimir operators $P_{\mu}P^{\mu}$ and $W_{\mu}W^{\mu}$ are fixed and given by the relations (5.4.16). It is to be pointed out that we do not make any supposition on the explicit form of the operators P_{μ} and $J_{\mu\nu}$, they can be as differential operators of the first order as nonlocal (integrodifferential) ones.

Theorem 5.4.4 [81]. Any Poincare invariant equation for a particle of mass m and spin s is invariant under the algebra AGL(2s + l, C).

Proof. Let P_{μ} , $J_{\mu\nu}$ be the symmetry operators of Equation (5.4.15). Then according to the definition and conditions (5.4.16) the following combinations

$$Q_{\mu\nu}^{\pm} = \frac{1}{m^2} \left[\epsilon_{\mu\nu\rho\sigma} W^{\rho} P^{\sigma} \pm \left(P_{\mu} W_{\nu} - P_{\nu} W_{\mu} \right) \right]$$
(5.4.19)

are also the symmetry operators of this equation.

Using (5.4.18) and the relations

$$[W_{\mu}, P_{\nu}] = 0, \qquad [W_{\mu}, W_{\nu}] = i\epsilon_{\mu\nu\rho\sigma}P^{\rho}W^{\sigma}$$
(5.4.20)

one can make sure that the operators (5.4.19) satisfy the conditions

$$[Q_{\mu\nu}^{\pm}, Q_{\lambda\sigma}^{\pm}] = i (g_{\mu\sigma} Q_{\nu\lambda}^{\pm} + g_{\nu\lambda} Q_{\mu\sigma}^{\pm} - g_{\mu\lambda} Q_{\nu\sigma}^{\pm} - g_{\nu\sigma} Q_{\mu\lambda}^{\pm}) m^{-4} (P_{\rho} P^{\rho})^{2}$$
(5.4.21)

and

$$c_{1} = \frac{1}{4}Q_{\mu\nu}^{\pm}Q^{\pm\mu\nu} = -m^{-4}W_{\lambda}W^{\lambda}P_{\sigma}P^{\sigma},$$

$$c_{2} = \frac{1}{4}\epsilon_{\mu\nu\rho\sigma}Q^{\pm\mu\nu}Q^{\pm\rho\sigma} = \mp im^{-4}W_{\lambda}W^{\lambda}P_{\sigma}P^{\sigma},$$
(5.4.22)

It follows from (5.4.16) and (5.4.21) that on the set of solutions of Equation (5.4.15) operators (5.4.19) satisfy the commutation relations

$$[Q_{\mu\nu}^{\pm}, Q_{\lambda\sigma}^{\pm}] = i \left(g_{\mu\sigma} Q_{\nu\lambda}^{\pm} + g_{\nu\lambda} Q_{\mu\sigma}^{\pm} - g_{\mu\lambda} Q_{\nu\sigma}^{\pm} - g_{\nu\sigma} Q_{\mu\lambda}^{\pm} \right) \psi$$
(5.4.23)

which characterize the Lie algebra of the group SL(2,C). From (5.4.16) and 5.4.22) one obtains the eigenvalues of corresponding Casimir operators

$$c_1\psi = \frac{1}{2}(\ell_0^2 + \ell_1^2 - 1)\psi, \qquad c_2\psi = -i\ell_0\ell_1\psi, \tag{5.4.24}$$

where $\ell_0 = s$, $\ell_1 = \pm (s+1)$.

So we demonstrated that any Poincare-invariant equation of non-zero mass and spin s is additionally invariant under the algebra ASL(2,C), the basis elements of which belong to the enveloping algebra of AP(1,3) and are given exactly by the relations (5.4.19). According to (5.4.24) operators (5.4.19) realize the representation $D(\ell_0, \ell_1) = D(s, \pm(s+l))$ of AGL(2,C). Now we see that this invariance algebra may be extended to 2(2s + l)-dimensional Lie algebra isomorphic to the algebra AGL(2s + l, C). Exactly the basis elements of AGL(2s + l, C) have the following form on the set of solutions of Equation (5.4.15):

$$\lambda_{n+k\,n} = a_{kn} (Q_{23}^{+} - Q_{02}^{+}) P_{n}^{s},$$

$$\lambda_{n\,n+k} = a_{kn} P_{n}^{s} (Q_{23}^{+} + Q_{02}^{+}),$$

$$\widetilde{\lambda}_{m\,n} = Q_{1} \lambda_{m\,n},$$

(5.4.25)

where

$$P_n^s = \prod_{n' \neq n} \frac{Q_{12} - s - 1 + n'}{n' - n}, \qquad Q_1 = \frac{\epsilon_{abc}}{2s(s+1)} Q_{0a}^+ Q_{bc}^+,$$
$$m, n = 1, 2, \dots, 2s + 1; \qquad k = 0, 1, \dots, 2s - n$$

and a_{kn} are coefficients determined by the recurrent relations

$$a_{0n} = 1,$$
 $a_{1n} = [n(2s+1-n)]^{-1/2},$
 $a_{\lambda n} = a_{\lambda-1} a_{\lambda-1} a_{\lambda-1} a_{\lambda-1},$ $\lambda = 2, 3, \dots, 2s - n.$

Actually, the polynomials of the symmetry operators $Q^{\pm}_{\mu\nu}$ given by the relations (5.4.25) are the symmetry operators of the equation (5.4.15). Operators (5.4.25) form the basis of the algebra AG(2s + l, C) inasmuch as they satisfy the commutation relations of AGL(2s + l, C)

$$\begin{aligned} [\lambda_{ab}, \lambda_{cd}] &= -[\widetilde{\lambda}_{ab}, \lambda_{cd}] = \delta_{bc} \lambda_{ad} - \delta_{ad} \lambda_{bc}, \\ [\lambda_{ab}, \widetilde{\lambda}_{cd}] &= \delta_{bc} \widetilde{\lambda}_{ad} - \delta_{ad} \widetilde{\lambda}_{bc}, \quad a, b, c, d = 1, 2, \dots, 2s + 1. \end{aligned}$$
(5.4.26)

The relations (5.4.26) are correct on the manifold of solutions of Equation (5.4.15). The validity of the above formulae can be verified by direct calculation using the equivalent matrix representation for the basis elements of ASL(2,C)

$$Q_{ab}^+ = \epsilon_{abc} S_c, \qquad Q_{0a}^+ = -i S_a$$

where S_a are matrices realizing the representation D(s) of ASO(3) in the Gelfand-Zetlin basis. Thus the theorem is proved.

So, if Equation (5.4.15) is Poincare invariant and describes a particle of spin s and mass m > 0, it is invariant also under AGL(2s + l, C) basis elements of which belong to the enveloping algebra of AP(1,3). The operators (5.4.25) together with the Poincare generators P_{μ} and $J_{\mu\nu}$ form the basis of the [10 + 2(2s+l)]-dimensional Lie algebra isomorphic to the algebra AP(1,3) \oplus AGL(2s + l, C). The last statement can easily be verified by moving to the new basis

$$P_{\mu} \to P_{\mu}, \qquad J_{\mu\nu} \to J_{\mu\nu} - Q_{\mu\nu}, \qquad \lambda_{mn} \to \lambda_{mn}, \qquad \widetilde{\lambda}_{mn} \to \widetilde{\lambda}_{mn},$$

where

$$Q_{12} = \sum_{n} (s - n + 1)\lambda_{nn}, \qquad Q_{03} = \sum_{n} (s - n + 1)\overline{\lambda}_{nn}$$
$$Q_{23} = \sum_{n} \frac{1}{2a_{1n}} (\lambda_{n n+1} + \lambda_{n+1 n}), \qquad Q_{31} = -i[Q_{12}, Q_{23}],$$
$$Q_{02} = i[Q_{23}, Q_{03}], \qquad Q_{01} = -i[Q_{31}, Q_{03}].$$

The theorem proved has a constructive character insofar as it gives the explicit form of the basis elements of additional invariance algebra via the Poincare generators. Starting, for example, from the Poincare generators for the Dirac equation (2.1.11) we obtain by means of the formula (5.4.19) the additional symmetry operators

$$Q_{\mu\nu}^{\pm} = \frac{i}{4} [\gamma_{\mu}, \gamma_{\nu}] + \frac{i}{2m} (\gamma_{\mu} P_{\nu} - \gamma_{\nu} P_{\mu}) (1 \pm i\gamma_{4}), \qquad (5.4.27)$$

which one easily recognizes as operators from (5.4.1).

In conclusion, let us present the result on complete symmetry of the Dirac equation (5.3.40) in the class of matrix differential operators of the first order

(that is operators of the form $Q = A^{\mu}(x)\partial_{\mu} + B(x)$, where A_{μ} and B are 4-column matrices. Note, in Lie approach A_{μ} are obligatory scalar functions). Let us denote this class as M_1 .

It turns out that operators (2.1.11) and (5.4.1) do not exhaust all symmetry operators of the Dirac equation (5.3.40) in the class M_1 .

Theorem 5.4.5 [81]. The Dirac equation (5.3.40) has only 26 linearly independent symmetry operators $Q \in M_1$, These operators include the Poincare generators (2.1.11), identity operator and 15 operators given below

$$\eta_{\mu} = \frac{i}{4} \gamma_4 (P_{\mu} - m \gamma_{\mu}),$$

$$\omega_{\mu\nu} = m S_{\mu\nu} + \frac{i}{2} (\gamma_{\mu} P_{\nu} - \gamma_{\nu} P_{\mu}),$$

$$A_{\mu} = \omega_{\mu\nu} x^{\nu} + x^{\nu} \omega_{\mu\nu} - i \gamma_{\mu},$$

$$B = i \gamma_4 (D - m \gamma_{\nu} x^{\nu}),$$

(5.4.28)

where

$$D = x^{\nu} P_{\nu} + \frac{3}{2}i, \qquad S_{\mu\nu} = \frac{i}{4} [\gamma_{\mu}, \gamma_{\nu}], \qquad \mu, \nu = \overline{0, 3}.$$
(5.4.29)

The proof consists in constructing the general solution of the defining equations following from the invariance condition

$$[L,Q] = f_Q L, (5.4.30)$$

where $L = \gamma P - m$, Q, and f_Q are unknown operators belonging to M_1 .

It will be noted that operators (5.4.28) do not form the basis of a Lie algebra since commutators $[\omega_{\mu\nu}, \omega_{\lambda\sigma}]$ do not belong to the class M_1 .

5.5 Symmetry of integrodifferential equations

Here we consider (following [88*]) a method of studying local symmetry properties of nonlinear systems of integrodifferential equations (IDEs) of the form

$$L\psi + \lambda_1 F(\psi) + \lambda_2 \int\limits_{\mathbb{R}^n} K(x, y, \psi(x), \psi(y)) dy = 0, \qquad (5.5.1)$$

where L is a linear differential operator: $x \in \mathbb{R}^n$; ψ , F, K are columns with m components; λ_1 , λ_2 are arbitrary constants. In what follows we shall suppose that the integral of (5.5.1) exists.

Before we proceed to formulate the principal result we note that the standard Lie algorithm is inapplicable to the IDE (5.5.1). The symmetry properties of Equation (5.5.1) can be studied by means of method of differential forms [116, 84], but in this case one faces a problem of unwieldy calculations which become really enormous when the order of the differential operator L and the number of components of ψ increase. This circumstance essentially restricts the applicability of the method of differential forms.

We shall investigate symmetry properties of IDE (5.5.1) in the class of firstorder differential operators of the form (7), (8). As a matter of convenience we write here formulae (7), (8) once again

$$Q = \xi^{\mu}(x)\frac{\partial}{\partial x_{\mu}} + \eta(x) \Leftrightarrow$$

$$X = \xi^{\mu}(x)\frac{\partial}{\partial x_{\mu}} - (\eta(x)\psi)^{k}\frac{\partial}{\partial \psi^{k}},$$

$$\mu = \overline{0, n-1}, \qquad k = \overline{1, m}.$$
(5.5.2)

Theorem 5.5.1. The maximal invariance algebra of the system of IDE (5.5.1) in the class of first-order differential operators (5.5.2) is determined by the following defining equations

1° $[L,Q] = \lambda(x)L$, $(\lambda(x)$ is an m-component matrix)

2°
$$\left(\eta(x) + \lambda(x) - (\eta(x)\psi)^k \frac{\partial}{\partial \psi^k}\right) F(\psi) = 0,$$

$$3^{\circ} \quad \left\{ \frac{\partial \xi^{\mu}(y)}{\partial y_{\mu}} + \eta(x) + \lambda(x) + \xi^{\mu}(x) \frac{\partial}{\partial x^{\mu}} + \xi^{\mu}(y) \frac{\partial}{\partial y^{\mu}} - (\eta(y)\psi(y))^{k} \frac{\partial}{\partial \psi^{k}(y)} - (\eta(x)\psi(x))^{k} \frac{\partial}{\partial \psi^{k}(x)} \right\} K(x, y, \psi(x), \psi(y)) = 0.$$

$$(5.5.3)$$

Proof. Using formulae (5.3.6) one finds transformations generated by operator (5.5.2):

$$\begin{aligned} x'_{\mu} &= e^{\theta\xi\partial}x_{\mu}e^{-\theta\xi\partial} = x_{\mu} + \theta\xi^{\mu}(x) + \cdots \\ \psi'(x') &= e^{\theta\xi\partial}e^{-\theta Q}\psi(x) = \psi(x) - \theta\eta(x)\psi(x) + \cdots \\ L(x',\partial') &= e^{\theta\xi\partial}L(x,\partial)e^{-\theta\xi\partial} = L(x,\partial) + \theta[\xi\partial,L] + \cdots \end{aligned}$$
(5.5.4)

Further, according to the fundamental theorem of integral calculus, under arbitrary coordinate transformation $y_{\mu} \rightarrow y'_{\mu}$ an element of volume dy =

 $dy_0 \ldots dy_{n-1}$ transforms as

$$dy \to dy' = \det\left(\frac{\partial y'}{\partial y}\right) dy,$$
 (5.5.5)

or in infinitesimal form, when $y'_{\nu} = y_{\nu} + \theta \xi^{\nu}(y) + \cdots$,

$$dy' = \left(1 + \theta \frac{\partial \xi^{\nu}(y)}{\partial y_{\nu}} + \cdots\right) dy$$
 (5.5.6)

Now we substitute (5.5.4), (5.5.6) into the primed Equation (5.5.1):

$$\begin{split} L(x',\partial')\psi'(x') + \lambda_1 F(\psi') + \lambda_2 & \int_{R^n} K(x',y',\psi'(x'),\psi'(y'))dy' = \\ &= L(x,\partial)\psi(x) + \lambda_1 F(\psi) + \lambda_2 \int_{R^n} K(x,y,\psi(x),\psi(y))dy + \\ &+ \theta \Big\{ ([Q(x,\partial),L(x,\partial)] - \eta(x)L(x,\partial))\psi(x) - \lambda_1 (\eta(x)\psi(x))^k \frac{\partial}{\partial \psi^k} F(\psi) + \\ &+ \lambda_2 \int_{R^n} \Big[\frac{\partial \xi^{\nu}(y)}{\partial y_{\nu}} + \xi^{\mu}(x) \frac{\partial}{\partial x_{\mu}} + \xi^{\mu}(y) \frac{\partial}{\partial y_{\mu}} - (\eta(x)\psi(x))^k \frac{\partial}{\partial \psi^k} - \\ &- (\eta(y)\psi(y))^k \frac{\partial}{\partial \psi^k(y)} \Big] K(x,y,\psi(x),\psi(y)) \Big\} dy + O(\theta^2). \end{split}$$

The requirement of invariance means that the above expression should be equal to zero on the set of solutions of Equation (5.5.1). It immediately gives the condition

$$([Q, L] - \eta(x)L)\psi(x) - \lambda_1 (\eta(x)\psi(x))^k \frac{\partial}{\partial\psi^k}F(\psi) + + \lambda_2 \int_{R^n} \left(\frac{\partial\xi^{\nu}(y)}{\partial y_{\nu}} + \xi^{\nu}(x)\frac{\partial}{\partial x_{\nu}} + \xi^{\nu}(y)\frac{\partial}{\partial y_{\nu}} - - (\eta(x)\psi(x))^k \frac{\partial}{\partial\psi^k(x)} - (\eta(y)\psi(y))^k \frac{\partial}{\partial\psi^k(y)}\right) K(x, y, \psi(x), \psi(y)) = 0.$$

$$(5.5.7)$$

The condition (5.5.7) together with Equation (5.5.1) lead to Equations (5.5.3). The general solutions of Equations (5.5.3) determines the maximal IA of IDE (5.5.1) in the class of operators (5.5.2). The theorem is proved.

Now we consider several examples of IDE of the type (5.5.1).

Theorem 5.5.2. [88*] The equation

Chapter 5. Some Special Questions

$$\Box u + \lambda_1 u^3 + \lambda_2 \int_{R^4} \frac{u^{3-\alpha}(x)u^{4-\alpha}(y)}{|x-y|^{2\alpha}} G\big(|x-y|^2 u(x)u(y)\big) dy = 0, \qquad (5.5.8)$$

where $|x - y| = [(x_{\nu} - y_{\nu})(x^{\nu} - y^{\nu})]^{1/2}$; $\nu = \overline{0,3}$; G is an arbitrary integrable function; u = u(x) is a real scalar function; $\lambda_1, \lambda_2, \alpha$ are arbitrary constants, is invariant under the conformal group C(1,3) and has the most general form in the class of conformally invariant IDEs of the type

$$\Box u + \lambda_1 F(u) + \lambda_2 \int_{R^4} K(x, y, u(x), u(y)) dy = 0.$$
 (5.5.9)

Proof. Let us use Theorem 5.5.1 and apply it to Equation (5.5.9). It is appropriate to recall that the free wave equation $\Box u = 0$ is invariant under AC(1,3) (basis elements are given in (1.1.2), (1.3.2)), provided

$$[\Box, P_{\mu}] = [\Box, J_{\mu\nu}] = 0, [\Box, D] = 2\Box, \qquad [\Box, K_{\mu}] = 4x_{\mu}\Box.$$
 (5.5.10)

Using (5.5.10) we can write the defining equations 2° , 3° from (5.5.3):

$$\begin{pmatrix} \frac{\partial}{\partial x_{\mu}} + \frac{\partial}{\partial y_{\mu}} \end{pmatrix} K(x, y, u(x), u(y)) = 0, \begin{pmatrix} x_{\mu} \frac{\partial}{\partial x_{\nu}} - x_{\nu} \frac{\partial}{\partial x_{\mu}} + y \frac{\partial}{\partial y_{\nu}} - y_{\nu} \frac{\partial}{\partial y_{\mu}} \end{pmatrix} K = 0 \begin{pmatrix} 7 + x_{\mu} \frac{\partial}{\partial x_{\mu}} + y_{\mu} \frac{\partial}{\partial y_{\mu}} - u(x) \frac{\partial}{\partial u(x)} - u(y) \frac{\partial}{\partial u(y)} \end{pmatrix} K = 0,$$
 (5.5.11)
$$3F - u \frac{\partial F}{\partial u} = 0.$$

The general solution of the system (5.5.11) has the form

$$F = F(u) = \lambda_1 u^3,$$

$$K = K(x, y, u(x), u(y)) = \lambda_2 \frac{u^{3-\alpha}(x)u^{4-\alpha}(y)}{|x-y|^{2\alpha}} G(|x-y|^2 u(x)u(y))$$

Thus the theorem is proved.

Remark 5.5.1. One can directly verify the conformal invariance of Equation (5.5.8) by means of final conformal transformations (see formulae (2.3.2), (2.3.33))

$$x'_{\mu} = rac{x_{\mu} - c_{\mu}x^2}{\sigma(x,c)}, \qquad u'(x') = \sigma(x,c)u(x), \qquad \sigma(x,c) = 1 - 2cx + c^2x^2,$$

$$|x' - y'| = \frac{|x - y|}{\sqrt{\sigma(x, c)\sigma(y, c)}}, \qquad dy' = \frac{dy}{\sigma^4(y, c)}.$$
 (5.5.12)

In much the same way one can prove the validity of the following statements.

Theorem 5.5.3. The nonlinear IDE for Dirac spinor field

$$\begin{cases} i\gamma\partial + \lambda_1(\overline{\psi}\psi)^{1/3} + \lambda_2 \frac{(\overline{\psi}\gamma_\nu\psi)\gamma^\nu}{\left[(\overline{\psi}\gamma_\mu\psi)(\overline{\psi}\gamma^\mu\psi)\right]^{1/3}} + \\ + \lambda_3 \int_{R^4} \frac{\left[\overline{\psi}(y)\psi(y)\right]^\alpha}{|x-y|^{2(4-3\alpha)}} \left[\overline{\psi}(x)\psi(x)\right]^{\alpha-1} \cdot \\ \cdot G\Big(|x-y|^6(\overline{\psi}(x)\psi(x))\left(\overline{\psi}(y)\psi(y)\right)\Big) dy \bigg\} \psi(x) = 0, \end{cases}$$

where λ_1 , λ_2 , λ_3 , α are arbitrary constants; G is an arbitrary integrable function (other notations see in Paragraphs 2.1, 2.3), is invariant the conformal group C(1,3).

Theorem 5.5.4. The nonlinear IDE for complex scalar field

$$\left(i\frac{\partial}{\partial x_0} + \frac{\Delta}{2m}\right)\psi + \lambda_1(\psi^*\psi)^{2/3}\psi +$$

$$+ \lambda_2 \int_{R^4} \exp\left\{i\frac{|\vec{x} - \vec{y}|^2}{2(x_0 - y_0)}\right\}\psi(y)(x_0 - y_0)^{-7/2} \cdot \\ \cdot \phi\left(\frac{\psi^*(x)\psi(x)\psi^*(y)\psi(y)}{(x_0 - y_0)^3}, \exp\left\{-i\frac{|\vec{x} - \vec{y}|}{(x_0 - y_0)}\frac{\psi^*(y)\psi(x)}{\psi^*(x)\psi(y)}\right\}\right)d^4y = 0.$$

$$(5.5.14)$$

where ψ is an arbitrary integrable function, is invariant under the Schrödinger group Sch(1,3) and generalizes the Schrödinger equation.

5.6. On exact and approximate solutions of the multidimensional Van der Pol equation

In [63] the multidimensional generalization of Van der Pol equation was suggested. Here we following [191] construct some exact and approximate solutions of this multidimensional Van der Pol equation.

So, the equation in question is

$$\Box u + \lambda_1 u + \lambda_2 (1 - \lambda_3 u^2) \frac{\partial u}{\partial x_0} = 0$$
(5.6.1)

where u = u(x) is a real scalar function, $x \in \mathbb{R}^4$; $(\lambda_1, \lambda_2, \lambda_3) = \text{const.}$ By means of Lie's method one can show that the Lie-maximal IA of Equation (5.6.1) is the 7-dimensional Lie algebra $AE(3)\oplus P_0$ with basis elements

$$P_0 = \partial_0, \qquad P_a = \partial_a, \qquad J_{ab} = x_a P_b - x_b P_a. \tag{5.6.2}$$

Using the algorithm stated in Paragraph 1.4 and symmetry operators (5.6.2) one can construct an ansatze to solve Equation (5.6.1). Below we consider some of such ansatze.

Let

$$u(x) = \varphi(\omega), \qquad \omega = \beta_0 x_0 + \vec{\beta} \cdot \vec{x}, \qquad \beta_0^2 - \vec{\beta}^2 \equiv \beta^2 \neq 0. \tag{5.6.3}$$

The substitution of (5.6.3) into (5.6.1) results in the original Van der Pol equation for function $\varphi(\omega)$:

$$\beta^2 \frac{d^2 \varphi}{d\omega^2} + \lambda_1 \varphi + \lambda_2 \beta_0 (1 - \lambda_3 \varphi^2) \frac{d\varphi}{d\omega} = 0.$$
 (5.6.4)

Unfortunately, exact solutions of this equation are not known yet. Therefore, we apply the Krilov-Bogolubov-Mitropolski method [32] to obtain an approximate solution of Equation (5.6.4). So, in a first approximation we find

$$\varphi(\omega) = a_0 e^{\epsilon \omega/2} \left[1 + a_0^2 (e^{\epsilon \omega} - 1)/4 \right]^{-1/2} \cos\left(\sqrt{\frac{\lambda_1}{\beta^2}} \omega + \theta\right)$$

where a_0 , θ are arbitrary constants; $\epsilon = -\frac{\lambda_2\beta_0}{\beta^2}$. Inserting this result into (5.6.3) we obtain an approximate solution of Equation (5.6.1):

$$u(x) = \frac{a_0 \exp\left\{\frac{1}{2}\epsilon\beta x\right\}}{\left[1 + \frac{1}{4}a_0^2\left(\exp\{\epsilon\beta x\} - 1\right)\right]^{-1/2}}\cos\left(\sqrt{\frac{\lambda_1}{\beta^2}}\,\beta x + \theta\right)$$
(5.6.5)

Setting in (5.6.4) $\lambda_1 = 0$ we get the ODE

$$\beta^2 \frac{d^2 \varphi}{d\omega^2} + \lambda_2 \beta_0 (1 - \lambda_3 \varphi^2) \frac{d\varphi}{d\omega} = 0$$

which can be fully integrated. Its general solution has the form

$$\varphi(\omega) = \pm \sqrt{\frac{3}{\lambda_3}} \left[1 - \left(\frac{\lambda_2 \lambda_3 \beta_0}{3\beta^2} \right)^2 \exp\left\{ \frac{2\lambda_2 \beta_0}{\beta^2} \omega + c_2 \right\} \right]^{-1/2}, \quad (c_2 = \text{const}).$$

Correspondingly, we can write a partial solution to Equation (5.6.1) under $\lambda_1 = 0$:

$$u(x) = \pm \sqrt{\frac{3}{\lambda_3}} \left[1 - \left(\frac{\lambda_2 \lambda_3 \beta_0}{3\beta^2}\right)^2 \exp\left\{\frac{2\lambda_2 \beta_0}{\beta^2} \beta^{\nu} x_{\nu} + c_2\right\} \right]^{-1/2}, \quad (5.6.6)$$

Consider the ansatz

$$u(x) = \varphi(\omega), \qquad \omega = \frac{1}{2} (\vec{\alpha} \cdot \vec{x})^2 - \theta \vec{\beta} \cdot \vec{x},$$

$$\vec{\alpha}^2 = \theta^2, \qquad \vec{\beta}^2 = -\ell^2 \neq 0,$$

(5.6.7)

which reduces Equation (5.6.1) to the ODE

$$\ell^2 \theta^2 \frac{d^2 \varphi}{d\omega^2} + \lambda_1 \varphi = 0 \tag{5.6.8}$$

The general solution of Equation (5.6.8) has the form

$$\varphi(\omega) = c_1 \cos \sqrt{\frac{\lambda_1}{\ell^2 \theta^2}} \omega + c_2 \sin \sqrt{\frac{\lambda_1}{\ell^2 \theta^2}} \omega, \qquad \lambda_1 > 0;$$

$$\varphi(\omega) = c_1 \exp \left\{ \sqrt{\frac{-\lambda_1}{\ell^2 \theta^2}} \omega \right\} + c_2 \exp \left\{ -\sqrt{\frac{-\lambda_1}{\ell^2 \theta^2}} \omega \right\}, \qquad \lambda_1 < 0.$$

Inserting this result into (5.6.7) we obtain solutions of Equation (5.6.1):

$$u(x) = c_1 \cos\left(\sqrt{\frac{\lambda_1}{\ell^2 \theta^2}} \left[\frac{1}{2} (\vec{\alpha} \cdot \vec{x})^2 - \theta \vec{\beta} \cdot \vec{x}\right]\right) + c_2 \sin\left(\sqrt{\frac{\lambda_1}{\ell^2 \theta^2}} \left[\frac{1}{2} (\vec{\alpha} \cdot \vec{x})^2 - \theta \vec{\beta} \cdot \vec{x}\right]\right), \quad \lambda_1 > 0;$$

$$u(x) = c_1 \exp\left\{\sqrt{\frac{-\lambda_1}{\ell^2 \theta^2}} \left[\frac{1}{2} (\vec{\alpha} \cdot \vec{x})^2 - \theta \vec{\beta} \cdot \vec{x}\right]\right\} + c_2 \exp\left\{-\sqrt{\frac{-\lambda_1}{\ell^2 \theta^2}} \left[\frac{1}{2} (\vec{\alpha} \cdot \vec{x})^2 - \theta \vec{\beta} \cdot \vec{x}\right]\right\}, \quad \lambda_1 < 0.$$
(5.6.9)

Consider another ansatz

$$u(x) = \varphi(\omega), \qquad \omega = x_0 + \theta \ln \vec{\alpha} \cdot \vec{x}, \qquad \vec{\alpha}^2 = 0.$$
 (5.6.10)

Which reduces Equation (5.6.1) to the original Van der Pol equation

$$\frac{d^2\varphi}{d\omega^2} + \lambda_1\varphi + \lambda_2(1-\lambda_3\varphi^2)\frac{d\varphi}{d\omega} = 0.$$
 (5.6.11)

By analogy with Equation (5.6.4) we construct an approximate solution of Equation (5.6.11):

$$\varphi(\omega) = \frac{a_0 e^{\epsilon \omega/2}}{\left[1 + \frac{1}{4} a_0^2 \left(e^{\epsilon \omega} - 1\right)\right]^{1/2}} \cos\left(\sqrt{\lambda_1} \,\omega + \theta\right) \tag{5.6.12}$$

Integrating (5.6.11) under $\lambda_1 = 0$ we find a family of solutions of Equations (5.6.1) under $\lambda_1 = 0$:

$$u(x) = \pm \sqrt{\frac{3}{\lambda_3}} \left[1 - \left(\frac{\lambda_2 \lambda_3}{3}\right)^2 \exp\left\{ 2\lambda_2 (x_0 + \theta \ln \vec{\alpha} \cdot \vec{x}) + c_2 \right\} \right]^{-1/2}.$$
 (5.6.13)

In conclusion, let us note that exact solutions of Equations (5.6.1) can be used for obtaining new approximate solutions of the equation. Indeed, let $u(x) = \varphi(x)$ be an exact solution of Equation (5.6.1), then we shall look for an approximate solution in the form

$$u(x) = \varphi(x) + \epsilon g(x) \tag{5.6.14}$$

where ϵ is a small parameter. The substitution of (5.6.14) into (5.6.1) and taking only first order in ϵ gives a linear PDE for function g = g(x):

$$\Box g + \lambda_1 g + \lambda_2 \left[(1 - \lambda_3 \varphi^2) \frac{\partial g}{\partial t} - 2\lambda_3 \varphi \frac{\partial \varphi}{\partial t} g \right] = 0$$

which should be considered as defining an equation for function g(x).

5.7 Conditional symmetry of PDEs

Generalizing results of works [94, 67, 108, 7^{*}, Appendix 4 in [82]] here we introduce the concept of conditional invariance (briefly: c-invariance) of PDEs or, in other words, invariance of PDEs on submanifolds of their solutions. The usefulness and efficiency of this conception is illustrated on a set of nonlinear equations from heat-conduction, Born-Infeld theory, gas dynamics, nonlinear acoustics, and some others. With the help of c-invariance, we construct new classes of exact solutions of these equations, which cannot be obtained based on the Lie approach.

1. The conditional invariance. Consider a system of PDEs of sth order

$$L(x, u, u, \dots, u) = 0, (5.7.1)$$

where $u = u(x), x \in \mathbb{R}^n$, u_s is a totality of s-order derivatives. Equation (5.7.1) admits (in sense of Lie) an algebra $A = \{X\}$ (the general form of operators X

is written in (2)). It means that condition of invariance (3) holds true for any operator $X \in A$. Here we rewrite (3) as follows

$$XL = RL, (5.7.2)$$

where $R = R(x, u, \theta, \partial)$ are some differential operators, depending on x and u.

Definition 5.7.1. A system of PDEs (5.7.1) we shall call *c*-invariant if there is an additional condition (equation), compatible with (5.7.1), which enlarges the symmetry of the system (5.7.1).

A clear example of c-symmetry gives us Maxwell's equations. The maximal IA of the system

$$\frac{\partial \vec{E}}{\partial t} = \operatorname{rot} \vec{H}, \qquad \frac{\partial \vec{H}}{\partial t} = -\operatorname{rot} \vec{E}$$

is 10-dimensional [82, p.121]:

$$\begin{split} A_{10} &= \left\langle P_0 = \partial_t, \quad P_a = \partial_a, \quad J_{ab} = x_a \partial_b - x_b \partial_a + E_a \partial_{E_b} - E_b \partial_{E_a} + \right. \\ &+ H_a \partial_{H_b} - H_b \partial_{H_a}, \quad D = t \partial_t + x_a \partial_a, \quad F = E_a \partial_{H_a} - H_a \partial_{E_a}, \\ &I = E_a \partial_{E_a} + H_a \partial_{H_a} \right\rangle \end{split}$$

and we see that it is not Lorentz invariant. But, by supplementing the above system to the Maxwell's equations by the well-known additional conditions

$$\operatorname{div} \vec{E} = 0, \qquad \operatorname{div} \vec{H} = 0$$

we thereby essentially enlarge its symmetry: the IA of Maxwell's equation is the 16-dimensional Lie algebra $AC(1,3)\otimes F$.

If A is the maximal IA of Equation (5.7.1), then for any operator Y not belonging to A, the condition of invariance (5.7.2) does not hold. Instead of (5.7.2), we will have

$$YL = R_0 L + R_1 L_1, \qquad R_1 \neq 0, \tag{5.7.3}$$

where R_0 , R_1 , L_1 are some differential expressions.

Definition 5.7.2. System of PDEs (5.7.1) we shall call *c*-invariant under an operator Y which does not belong to IA of (5.5.1) if it is invariant under this operator only together with an additional condition

$$L_1(x, u, u, \dots, u) = 0, (5.7.4)$$

or, in other words, instead of (5.7.2), now we have (5.7.3) and

$$YL_1 = R_2 L + R_3 L_1. (5.7.5)$$

The additional condition (5.7.4) select from the whole set of solutions of Equation (5.7.1) a subset and this subject may well be true to have the symmetry wider than that of the whole set. It is most desirable to know how to select such subsets of solutions.

It is obvious that the concept of c-invariance makes sense only if the couple of equations (5.7.1) and (5.7.4) is consistent. Relation (5.7.5) represents the necessary condition of compatibility for Equations (5.7.1), (5.7.4). So, to find c-invariant solutions one has to solve compiling systems (5.7.1), (5.7.4) and the equation

$$Yu = 0 \tag{5.7.6}$$

and for which purpose to find the very operator Y. In general case, it is rather complicated to solve this problem. However, if additional condition (5.7.4) coincides with (5.7.6), then one succeeds in obtaining a constructive algorithm for finding *c*-invariant operators.

Definition 5.7.3. System of PDEs (5.7.1) we shall call c(Y)-invariant (or Y-conditional invariant) if

$$YL = R_0 L + R_1 (Yu). (5.7.7)$$

As in the previous case (see Definition 5.7.2) relation (5.7.7) represents a necessary condition of compatibility of system (5.7.1), (5.7.6). It will be noted that even in Lie's case, when $R_1 = 0$, fulfillment of relation (5.7.7) does not guarantee compatibility of system (5.7.1), (5.7.6). For example, any linear inhomogeneous PDE of first order $Qu = \text{const} \neq 0$ is, clearly, invariant under the operator Q, though it has no Q-invariant solutions because it is inconsistent with the condition Qu = 0 which defines such solutions.

Definition 5.7.3 can be generalized on differential operators Y of second or higher order. To do it one has to use the condition of invariance (5.7.7) in the Lie-Backlund form.

It is clear that ansatze constructed within the framework of conditional symmetry include Lie's in particular, as well as others which we shall call non-Lie ansatze.

2. Conditional invariance of nonlinear heat equation under the Galilei algebra. As is well known (see $\S3.3$) the linear heat equation

$$u_0 + \frac{1}{2m}\Delta u = 0$$

admits AG(1, n), provided

$$G_a = x_0 \partial_a + m x_a u \partial_u, \tag{5.7.8}$$

and there is no one nonlinear heat equation of the form

$$u_0 + \vec{\nabla}[f(u)\vec{\nabla}u] = 0, \qquad f(u) \neq \text{const},$$
 (5.7.9)

which would be invariant under Galilean transformations. But it only means that the whole set of solutions of Equations (5.7.9) is not invariant under Galilean transformations [66], though there may be a subset of solutions which will be Galilean invariant. An affirmative answer on this question gives the following theorem.

Theorem 5.7.1. [39*] The nonlinear heat equation (5.7.9) is invariant under Galilean operators

$$G_a = x_0 \partial_a + M(u) x_a \partial_u \tag{5.7.10}$$

if

$$u_0 + \frac{(\vec{\nabla}u)^2}{2M(u)} = 0, \qquad (5.7.11)$$

where

$$M(u) = \frac{u}{2f(u)}.$$
 (5.7.12)

Proof. Having acted on (5.7.9) by G_2a (G_2a is constructed by virtue of formulae (1.1.7)), we find

$$\begin{aligned} G_{2a}\{u_{0} + \vec{\nabla}[f(u)\vec{\nabla}u]\} &= \left[M'u_{0} + (Mf)''(\vec{\nabla}u)^{2} + (fM)'\Delta u\right]x_{a} + \left[2(Mf)' - 1\right]u_{a} \end{aligned}$$

whence follows

$$2(Mf)'-1=0,$$

which gives (5.7.12). Substituting (5.7.12) into (5.7.13), we get

$$\begin{array}{l}
G_{2a}\left\{u_{0}+\vec{\nabla}\left[(f(u)\vec{\nabla}u\right]\right\} = \frac{x_{a}}{2f(u)}\left\{u_{0}+\vec{\nabla}\left[f(u)\vec{\nabla}u\right]\right\} - \\
-\frac{x_{a}f'(u)}{2f(u)}\left[u_{0}+\frac{(\vec{\nabla}u)^{2}}{2M(u)}\right]
\end{array}$$
(5.7.14)

whence follows (5.7.11).

To complete the proof, one can make sure that

$$G_{1a}\left[u_0 + \frac{(\vec{\nabla}u)^2}{2M(u)}\right] = M'(u)x_a\left[u_0 + \frac{(\vec{\nabla}u)^2}{2M(u)}\right].$$
 (5.7.15)

Theorem 5.7.2. The nonlinear heat equation (5.7.9) is $c(G_a)$ -invariant under Galilean operators (5.7.10) if

$$f(u) = \frac{1}{2m}u^{k}, \qquad M(u) = \frac{2m}{kn+2}u^{1-k};$$

$$f(u) = e^{u}, \qquad M(u) = 1,$$
(5.7.16)

where m, k are arbitrary constants, $kn + 2 \neq 0, m \neq 0$).

Proof. Conditions (5.7.6) in this case have the form

$$G_a u = x_0 u_a - M(u) x_a = 0 (5.7.17)$$

whence follows

$$u_{a} = x_{0}^{-1} [G_{a}u + M(u) \cdot x_{a}]$$

$$\Delta u = x_{0}^{-1} \partial_{a} G_{a}u + x_{0}^{-2} M(u) (nx_{0} + M(u)\vec{x}^{2}).$$
(5.7.18)

Substituting (5.7.18) into (5.7.13), we have

$$\begin{aligned} G_{2a}\left[u_{0}+\vec{\nabla}[f(u)\vec{\nabla}u]\right] &= M'x_{a}\left\{u_{0}+\vec{\nabla}[f(u)\vec{\nabla}u]\right\}+x_{0}[2(Mf)'-1]\delta_{ab}\right\}G_{b}u+\\ &+x_{0}^{-2}x_{a}\left\{\left[(Mf)''-M'f'\right](G_{b}u+2Mx_{b})+Mf'x_{0}\partial_{b}+\right.\\ &+x_{0}^{-2}[2(Mf)'+nMf'-1]x_{0}+M(Mf)''\vec{x}^{2}\right\}, \end{aligned}$$
(5.7.19)

whence follows (5.7.16). Since $[G_a, G_b] = 0$, the theorem is proved. It will be noted, that if $f(u) = \frac{1}{m}u^k$ $(m \neq 0, k \neq -\frac{2}{n})$, then Equation (5.7.9) is c-invariant under operators G_a (5.7.10), provided M(u) has two different forms, (5.7.12) and (5.7.16).

So, to describe nonlinear heat condition, one may use the completed Equations (5.7.9), (5.7.11), that is the system

$$u_{0} + \vec{\nabla}[f(u)\vec{\nabla}u] = 0,$$

$$u_{0} + \frac{(\vec{\nabla}u)^{2}}{2M(u)} = 0,$$

(5.7.20)

where $M(u) = \frac{u}{2f(u)}$.

Next we study the maximal IA of the system (5.7.20).

Theorem 5.7.3. The maximal IA of coupled Equations (5.7.20) is $A\widetilde{G}(2,n)$, basis element having the form

$$\partial_0, \, \partial_w, \, \partial_a, \qquad J_{ab} = x_a \partial_b - x_b \partial_a,$$

$$G_{1a} = x_0 \partial_a + m x_a \partial_w, \qquad G_{2a} = w \partial_a + m x_a \partial_0, \tag{5.7.21}$$

$$\mathcal{D}_1 = 2x_0\partial_0 + x_a\partial_a, \qquad \mathcal{D}_2 = 2w\partial_w + x_a\partial_a,$$

where $a, b = \overline{1, n}, m = \text{const} \neq 0$,

$$w = 2m \int \frac{f(u)}{u} du. \tag{5.7.22}$$

Proof. After changing variables according to (5.7.22), Equations (5.7.20) take the form

$$\Delta w = 0,$$

$$w_0 + \frac{(\vec{\nabla}w)^2}{2m} = 0.$$
(5.7.23)

Then applying to (5.7.23) the Lie algorithm, we find that maximal IA of these coupled equations is AG(2, n) with basis elements (5.7.21).

Now we find some exact solutions of Equations (5.7.20). From equation

$$\theta_a G_a u = 0 \tag{5.7.24}$$

where θ_a are arbitrary constants, G_a are given in (5.7.10), (5.7.12) we find the ansatz:

$$\frac{1}{2m}w = \int \frac{f(u)}{u}du = \varphi(\vec{\omega}) + \frac{\vec{x}^2}{4x_0},$$
(5.7.25)

where $\varphi(\vec{\omega})$ is an unknown function of variables $\vec{\omega} \in \mathbb{R}^n$. In three-dimensional space $\vec{\omega}$ are as follows

$$\omega_1 = x_0, \qquad \omega_2 = \alpha \vec{x}, \qquad \omega_3 = \beta \vec{x}, \qquad (5.7.26)$$

where $\vec{\alpha}, \vec{\beta}$ are constant vectors ($\vec{\alpha}^2 = \vec{\beta}^2 = 1, \vec{\alpha} \cdot \vec{\beta} = 0$). Substituting ansatz (5.7.25) into (5.7.23) we get

$$\begin{cases} \varphi_{22} + \varphi_{33} + \frac{n}{2\omega_1} = 0, \\ \varphi_2^2 + \varphi_3^2 + \frac{1}{\omega_1} \omega_a \varphi_a = 0. \end{cases}$$
(5.7.27)

A particular solution of Equations (5.7.27) has the form

$$\varphi = -\frac{n}{4}\frac{\omega_2^2}{\omega_1}, \qquad (n=1).$$
 (5.7.28)

Then the formula

$$\int \frac{f(u)}{u} du = \frac{\vec{x}^2 - n(\vec{\alpha}\vec{x})^2}{4x_0}, \qquad (n=1)$$
(5.7.29)

gives a Galilean-invariant solution of Equation (5.7.9)

Consider the nonlinear equation

$$u_0 + \frac{1}{m} \vec{\nabla} (u^k \vec{\nabla} u) = 0.$$
 (5.7.30)

From (5.7.17), (5.7.10), (5.7.16), we find the ansatz

$$u^{k} = \varphi(x_{0}) + \frac{km}{kn+2} \frac{\vec{x}^{2}}{2x_{0}}$$
(5.7.31)

which reduces (5.7.30) to the following ODE

$$\varphi' + \frac{kn}{kn+2} \frac{1}{x_0} \varphi = 0$$
 (5.7.32)

The general solution of (5.7.32) can easily be obtained and therefore via (5.7.31) we find a solution of (5.7.30)

$$u^{k} = \lambda x_{0}^{-kn/(kn+2)} + \frac{km}{kn+2} \frac{\vec{x}^{2}}{2x_{0}}, \qquad k \neq -\frac{2}{n}, \quad \lambda = \text{const}$$
(5.7.33)

3. Conditional invariance of the Born-Infeld equation under conformal algebra. In $\S1.9$ we established that the maximal IA of the *n*-dimensional Born-Infeld equation

$$(1 - u_{\nu}u^{\nu})\Box u + u^{\mu}u^{\nu}u_{\mu\nu} = 0$$
 (5.7.34)

is $A\widetilde{P}(1, n+1)$.

Theorem 5.7.4. Born-Infeld equation (5.7.34) is invariant under AC(1, n+1) if

 $1-u_{\nu}u^{\nu}=0$

(Basis elements of AC(1, n + 1) are written in (1.2.2)).

Remark 5.7.1. The Born-Infeld equation (5.7.34) is a differential consequence of the eikonal equation.

$$(1 - u_{\nu}u^{\nu}) \Box u + u^{\mu}u^{\nu}u_{\mu\nu} = (\Box u - \frac{1}{2}u^{\mu}\partial_{\mu})(1 - u_{\nu}u^{\nu}).$$

Except for conformally invariant subset of solutions, Equation (5.7.34) possesses a subset of solutions which is invariant under an infinite-dimensional algebra.

Theorem 5.7.5. Equation (5.7.34) is invariant under the infinite-dimensional algebra AP(1, n) with basis elements written in (1.2.19) if

$$u_{\nu}u^{\nu}=0$$

Theorem 5.7.4, 5.7.5 can be proved analogously to Theorem 5.7.1.

Using invariants of AC(1, n + 1)

$$\omega_1 = \frac{x_A x^A}{\alpha_A x^A}, \qquad \omega_2 = \frac{\beta_A x^A}{\alpha_A x^A}, \qquad (A = \overline{0, n+1}, \ x_{n+1} \equiv u)$$

and invariants of $\widetilde{AP}_{\infty}(1,n)$

$$\omega_1 = d_\mu(u) x^\mu, \qquad \omega_2 = u, \quad d_\mu d^\mu = 0$$

where α_A , β_A are arbitrary constants, $d_{\mu}(u)$ are arbitrary differential functions, we obtain the following ansatze

$$\frac{x_A x^A}{\alpha_A x^A} = \varphi\left(\frac{\beta_A x^A}{\alpha_A x^A}\right),\tag{5.7.35}$$

 $d_{\mu}(u)x^{\mu} = \varphi(\omega) \tag{5.7.36}$

However, these ansatze are not invariant under IA of the Equation (5.7.34), nevertheless they reduce it. Substitution ansatz (5.7.35) into (5.7.34) we get the ODE

$$(\alpha_A \alpha^A \omega_2^2 - 2\alpha_A \beta^A \omega_2 + \beta_A \beta^A) \varphi'^2 - 2(\alpha_A \alpha^A \omega_2 - \alpha_A \beta^A) \varphi \varphi' + \alpha_A \alpha^A \varphi^2 = 0.$$

Ansatz (5.7.36) satisfies (5.7.34) under arbitrary φ .

When

$$\alpha_A \alpha^A = \alpha_A \beta^A = \beta_A \beta^A = 0,$$

formula (5.7.35) gives a solution of the Born-Infeld equation (5.7.34) with arbitrary differentiable function φ ; otherwise it is a linear function of w_2 :

$$\varphi = \lambda_1 \omega_2 + \lambda_2, \qquad \lambda_1, \lambda_2 = \text{const.}$$

4. Extension of symmetry of acoustic nonlinear equations.

In many cases equations of nonlinear acoustics have the form

$$u_{00} = C(x, u, u) \Delta u$$
 (5.7.37)

where u = u(x), $x(x_0, \vec{x}) \in \mathbb{R}^{1+n}$, C(x, u, u) is a smooth function depending on x, u, u. One can make sure that Equation (5.7.37) is not Galilean invariant, but, as we shall show it, it has Galilean invariant subset of solutions.

Theorem 5.7.6. Equation (5.7.37) is conditionally G_a -invariant under G_a from (5.7.10) if

$$M(u) = m = \text{const}, \quad C(x, u, u_1) = F(\vec{w}, \vec{\nabla}w_2) + \frac{\vec{x}^2}{nx_0^2},$$
 (5.7.38)

where F is a arbitrary differentiable function, $w \in \mathbb{R}^3$,

$$w_1 = x_0, \qquad w_2 = u - \frac{m\vec{x}^2}{2x_0}, \qquad w_3 = u_0 + \frac{(\vec{\nabla}u)^2}{2m}.$$
 (5.7.39)

Ansatz following as a solution of equations $G_a u = 0$

$$u = \varphi(x_0) + \frac{m}{2} \frac{\vec{x}^2}{x_0} \tag{5.7.40}$$

reduces (5.7.37), (5.7.38) to the ODE

$$x_0\varphi'' = nmF(x_0,\varphi,\varphi',\vec{O}) \tag{5.7.41}$$

When n = 1, C(x, u, u) = u Equation (5.7.37) takes the form

$$u_{00} = u u_{11}. \tag{5.7.42}$$

As is shown in [160] Equation (5.7.42) is reduced to the ODE by the ansatz obtained with the help of operator

$$Q = \partial_0 + 2x_0\partial_1 + 8x_0\partial_u \tag{5.7.43}$$

which does not belong to IA of this equation.

Using condition (5.7.7) one can obtain more operators of *c*-invariance of Equation (5.7.42).

Theorem 5.7.7. Equation (5.7.42) is c(Y)-invariant, provided

$$Y = A(x)\partial_0 + B(x)\partial_1 + [\alpha(x)u + \beta(x)]\partial_u \qquad (5.7.44)$$

and functions A(x), B(x), $\alpha(x)$, $\beta(x)$ satisfy the following equations:

a).
$$A \neq 0$$
. $B \neq 0$
 $\alpha = 2(B_1 - A_0 - \frac{B}{A}A_1), \quad \beta = 2\frac{B}{A}B_0, \quad \alpha_{11} = \frac{\alpha}{A}A_{11} + 2\left(\frac{\alpha}{A}\right)_1 A_1,$
 $\alpha_{00} + 2\frac{\alpha}{A}\alpha_0 - \left[\frac{\alpha}{A}A_{00} + 2\left(\frac{\alpha}{A}\right)_1 B_0\right] = \beta_{11} - \left[\frac{\beta}{A}A_{11} + 2\left(\frac{\beta}{A}\right)_1 A_1\right],$

$$\beta_{00} + 2\frac{\beta}{A}\alpha_0 - \left[\frac{\beta}{A}A_{00} + 2\left(\frac{\alpha}{A}\right)_1 B_0\right] = 0, \qquad (5.7.45)$$

$$B_{11} - 2\alpha_1 - \left[\frac{B}{A}A_{11} + 2\left(\frac{B}{A}\right)_1 A_1\right] + 2\frac{\alpha}{A}A_1 = 0,$$

$$B_{00} + 2\frac{B}{A}\alpha_0 - \left[\frac{B}{A}A_{00} + 2\left(\frac{B}{A}\right)_1 B_0\right] + 2\frac{\alpha}{A}B_0 = 0.$$
b). $A = 0, B \neq 0$ (without loss of generality one can put $B = 1$)

$$\alpha_0 = 0, \qquad \alpha_{11} + 3\alpha\alpha_1 + \alpha^3 = 0, \qquad (5.7.46)$$

$$\beta_{11} + \alpha \beta_1 + (3\alpha_1 + 2\alpha^2)\beta = 0, \qquad \beta_{00} - \beta \beta_1 - \alpha \beta^2 = 0.$$
 (5.7.47)

c). A = 1, B = 0

$$\alpha_1 = 0, \qquad \alpha_{00} + \alpha \alpha_0 - \alpha^3 = \beta_{11}, \beta(\beta_0 + \alpha\beta) = 0, \qquad \beta_{00} + \alpha_0\beta - \alpha^2\beta = 0.$$
(5.7.48)

The proof of this theorem is not complicated but is rather cumbersome and for which reason we shall omit them.

Partial solutions of Equations (5.7.45)-(5.7.48) give explicit forms of operators (5.7.44) and we list them in Table 5.7.1., where we used notations: a_i, b_j $(i, j = \overline{1, 10})$ are arbitrary constants, $\omega = \frac{1}{2}a_4x_0^2 + a_5x_0x_1$, \wp is a Weierstrass elliptic function, which satisfies the equation

$$\wp'' = \wp^2 \tag{5.7.49}$$

 $f(x_0)$ is a solution of the Lame equation

$$f'' = \wp f \tag{5.7.50}$$

and $F(x_0)$ satisfies the equation

$$F''(x_0) = F^2(x_0) \left(\int F^{-2}(x_0) dx_0 + b_9 \right).$$
 (5.7.51)

Having solved some reduced ODEs from Table 5.7.1, we succeed in finding solutions of Equation (5.7.4):

$$u = P_{1}(x_{0})Q_{1}(x_{1}), \qquad u = x_{0}^{-2}[3x_{1}^{2} + Q_{1}(x_{1})] + x_{0}^{3}R_{1}(x_{1}),$$

$$u = \frac{1}{2}\wp(x_{0})x_{1}^{2} + f(x_{0})x_{1} + f^{2}(x_{0}), \qquad (5.7.52)$$

$$u = \left[P_{2}(x_{0}) + Q_{2}(x_{1})\int F^{-2}(x_{0})dx_{0}\right]F(x_{0}),$$

Ν	Operator Y	Ansatz $u =$	Reduced ODE
1.	$\partial_1 + a_1 \partial_u$	$=\varphi(x_0)+a_1x_1$	$\varphi'' = 0$
2.	$\partial_0 + (a_2 x_1 + a_3) \partial_u$	$= \varphi(x_1) + x_0(a_2x_1 + a_3)$	$\varphi'' = 0$
3.	$\partial_0 + (a_4x_0 + a_5)$	$= \varphi(\omega) + 2a_4x_1$	$(\varphi - 2a_4\omega -$
	$(\partial_1 + 2a_4\partial_u)$		$ -a_5)\varphi''=a_4\varphi'$
4.	$\partial_1 + [\wp(x_0)x_1 +$	$= \wp(x_0)\frac{x_1^2}{2} + f(x_0)x_1 + \varphi(x_0)$	$\varphi'' = \wp \varphi$
-	$+f(x_0)]\partial_u$		
5.	$x_0 \partial_0 + (u + a_7 x_1 + a_7) \partial_1$	$= x_0 \varphi(x_1) - (a_7 x_1 + a_8)$	$\varphi^{\mu} = 0$
	$+u_8)o_u$	r_{1}^{3}	
6.	$x_0\partial_0 + [x_0^3(a_9x_1 +$	$=x_0^{-2}\varphi(x_1)+\frac{x_0}{5}(a_9x_1+a_{10})$	$\varphi'' = 6$
_	$(+a_{10})-2u]\partial_u$		
7.	$x_1\partial_1 + (u+b_1x_0 + \dots + b_1x_0)$	$=x_1\varphi(x_0)-(b_1x_0+b_2)$	$\varphi'' = 0$
	$+b_2)O_u$	2	
8.	$x_1\partial_1 + [u + \wp(x_0)\frac{x_1}{2} -$	$= \wp(x_0)\frac{x_1}{2} + \varphi(x_0)x_1 + f(x_0)$	$\varphi'' = \wp \varphi$
	$-f(x_0)$] ∂_u	2	
9.	$x_0^3 \partial_0 + (3x_0^2 u - 15x_1^2 +$	$= x_0^3 \varphi(x_1) + 3x_0^{-2} x_1^2 -$	$\varphi'' = 0$
	$+b_3x_1+b_4)\partial_u$	$-\frac{1}{5}(b_3x_1-b_4)x_0^{-2}$	0 // -
10.	$x_0^2 x_1 \partial_1 + (x_0^2 u + 3x_1^2 + 3x_1^2 + 3x_1^2)$	$= x_1 \varphi(x_0) + 3x_0^{-2} x_1^2 - \frac{1}{2} x_1^{-2} + \frac{1}{2} x_1^{-2}$	$x_0^2\varphi^{\prime\prime}=6\varphi$
	$+b_5x_0^3+b_6)\partial_u$	$-b_5 x_0^3 - b_6 x_0^{-2}$	// 1
11.	$\wp(x_0)\partial_0 + \wp'(x_0)u\partial_u$	$= \varphi(x_0)\varphi(x_1)$	$\varphi'' = 1$
12.	$F(x_0)O_0 + [F'(x_0)u +$	$= F(x_0)\varphi(x_1) + (\frac{1}{2}x_1^2 + b_7x_1 + b_7x$	$\varphi^{\cdot\cdot} = b_9$
	$+\frac{1}{2}x_1^2 + b_7x_1 + b_8]\partial_u$	$(+ b_8)F(x_0)\int F^{-2}(x_0)dx_0$	

Table 5.7.1. Operators Y of c(Y)-invariance of Equation (5.7.42), ansatz and reduced ODEs.

$$u = \varphi(\omega) + a_4 x_0 (a_4 x_0 + 2a_5), \tag{5.7.53}$$

where $P_k(x)$, $Q_k(x)$, $R_k(x)$ are arbitrary polynomials of order k = 1, 2, and the other notations are the same as in Table 5.7.1.

Note, when $a_4 = 2$, $a_5 = 0$, operator Y_3 coincides with (5.7.43). The corresponding solution (5.7.53) is also obtained in [160].

The above results on two-dimensional Equation (5.7.42) can be generalized on the multi-dimensional case, that is, for the equation

$$u_{00} = u\Delta u. \tag{5.7.54}$$

We present these results in Table 5.7.2.

Table 5.7.2. (In this table $\vec{\alpha}$ is a constant vector, $\vec{\alpha}^2 = 1$, $\omega = \frac{1}{2}c_2x_0^2 + c_3x_0\vec{\alpha}\vec{x}$.)

Ν	Operator Y	Ansatz $u =$	Reduced ODE
1.	$\vec{\nabla} + \vec{\alpha}\partial_u$	$= \varphi(x_0) + \vec{lpha} \vec{x}$	$\varphi'' = 0$
2.	$\partial_0 + (\vec{lpha} \vec{x} + c_1) \partial_u$	$=\varphi(\vec{x})+x_0(\vec{\alpha}\vec{x}+c_1)$	$\Delta \varphi = 0$
3.	$ec{lpha}\partial_0+(c_2x_0+c_3)ec{ abla}+$	$= \varphi(\omega) + 2c_2 \vec{lpha} \vec{x}$	$(\varphi - 2c_2\omega -$
	$+2c_2(c_2x_0+c_3)ec lpha \partial_u$	-	$-c_3^2)arphi''=c_2arphi'$
4.	$ec{ abla} + [\wp(x_0)ec{x} + f(x_0)ec{lpha}]\partial_u$	$=\frac{1}{2n}\wp(x_0)\vec{x}^2+$	$\varphi'' = \wp \varphi$
		$+\frac{1}{\pi}f(x_0)\vec{lpha}\vec{x}+\frac{1}{\pi}\varphi(x_0)$	
5.	$x_0\partial_0 + (u + \vec{lpha}\vec{x} + c_1)\partial_u$	$=x_0arphi(ec{x})-ec{lpha}ec{x}-c_1^n$	$\Delta \varphi = 0$
6.	$x_0\partial_0 + [x_0^3(\vec{\alpha}\vec{x} + c_1) - 2u]\partial_u$	$= x_0^{-2}\varphi(\vec{x}) + \frac{x_0^2}{5}(\vec{\alpha}\vec{x} + c_1)$	$\Delta arphi = 6$
7.	$(\vec{lpha}\vec{x})\vec{ abla}+\vec{lpha}(u+c_2x_0+c_3)\partial_u$	$=\vec{\alpha}\vec{x}\varphi(x_0)-(\vec{c_2}x_0+c_3)$	$\varphi'' = 0$
8.	$(\vec{lpha}\vec{x})\vec{ abla}+\vec{lpha}[u+$	$= \frac{1}{2} \wp(x_0) (\vec{\alpha} \vec{x})^2 +$	$\varphi'' = \wp \varphi$
	$+\wp(x_0)rac{(ec lpha ec x)^2}{2}-f(x_0)]\partial_u$	$+\varphi(x_0)\vec{\alpha}\vec{x}+f(x_0)$	
9.	$x_0^3 \partial_0 + (3x_0^3 u - 15 \vec{x}^2 +$	$=x_0^3\varphi(\vec{x})+\frac{3}{n}x_0^{-2}\vec{x}^2-$	$\Delta arphi = 0$
	$+\vec{lpha}\vec{x}+c_1)\partial_u$	$-\frac{4}{5}x_0^{-2}(\vec{\alpha}\vec{x}+c_1)$	
10.	$x_0^2(\vec{lpha}\vec{x})\vec{ abla} + \vec{lpha}[x_0^2u + 3(\vec{lpha}\vec{x})^2 +$	$=\vec{\alpha}\vec{x}\varphi(x_0)+3x_0^{-2}(\vec{\alpha}\vec{x})^2-$	$x_0^2 arphi'' = 6 arphi$
	$+c_4x_0^5+c_3]\partial_u$	$-c_4 x_0^3 - c_3 x_0^{-2}$	
11.	$\wp(x_0)\partial_0+\wp'(x_0)u\partial_u$	$=\wp(x_0)\varphi(\vec{x})$	$\Delta \varphi = 1$
12.	$F(x_0)\partial_0 + [F'(x_0)u +$	$= F(x_0)\varphi(\vec{x}) + \left(\frac{1}{2}\vec{x}^2 + \vec{\alpha}\vec{x} + \frac{1}{2}\vec{x}^2 + \vec{\alpha}\vec{x}^2 + \vec{\alpha}\vec{x} + \frac{1}{2}\vec{x}^2 + \vec{\alpha}\vec{x}^2 + \frac{1}{2}\vec{x}^2 + \frac{1}{2}$	$\Delta \varphi = b_9$
	$+\frac{1}{2}\vec{x}^2 + \vec{\alpha}\vec{x} + c_1]\partial_u$	$+c_1)F(x_0)\int F^{-2}(x_0)dx_0$	

5. Extension of symmetry of equations $\Box u + \lambda u u_0 = 0$, Hamilton-Jacobi, gas dynamics, and Navier-Stokes

Theorem 5.7.8. Equation

$$\Box u + \lambda u u_0 = 0 \tag{5.7.55}$$

is c(Y)-invariant with

$$Y = x_0^2 \partial_0 + \left(x_0 u - \frac{4}{\lambda}\right) \partial_u \tag{5.7.56}$$

Theorem 5.7.9. The Hamilton-Jacobi equation

$$u_0 + \frac{(\vec{\nabla}u)^2}{2m} = 0 \tag{5.7.57}$$

Chapter 5. Some Special Questions

is c(Y)-invariant with

$$Q_a = r\partial_a + x_a \left(x_0 + \sqrt{x_0^2 - 2mr} \right) \partial_u, \qquad \left(r = \sqrt{\vec{x}^2} \right) \tag{5.7.58}$$

Theorem 5.7.10. Equations of gas dynamics

$$\begin{cases} \vec{u}_0 + (\vec{u}\,\vec{\nabla})\vec{u} = -\frac{1}{\rho}\vec{\nabla}\left(\frac{\lambda}{2}\rho^2\right),\\ \rho_0 + \operatorname{div}\left(\rho\vec{u}\right) = 0 \end{cases}$$
(5.7.59)

are c(Y)-invariant with

$$Y_a = x_0^{n+1} \partial_a + x_0^n \partial_{u^a} + \alpha_a \partial_\rho, \qquad (\alpha_a = \text{const}). \tag{5.7.60}$$

Theorem 5.7.11. The Navier-Stokes equations

$$\vec{u}_0 + (\vec{u}\,\vec{\nabla})\vec{u} + \lambda\Delta\vec{u} = 0, \qquad (5.7.61)$$

when n = 1, is c(Y)-invariant with

$$Y = x_1^2 \partial_1 - (x_1 u^1 - 4\lambda) \partial_{u^1}.$$
 (5.7.62)

Ansatz obtained by virtue of operator (5.7.56) has the form

$$u = x_0 \varphi(\vec{x}) + 2/\lambda x_0$$

and it reduces Equation (5.7.55) to the nonlinear Laplace equation

$$\Delta \varphi = \lambda \varphi^2. \tag{5.7.63}$$

Operators (5.7.58) give the ansatz

$$u = \varphi(x_0) + x_0 r - \frac{(x_0^2 - 2mr)^{2/3}}{3m}$$
(5.7.64)

which reduces the Hamilton-Jacobi equation to the ODE

$$m\varphi' + x_0^2 = 0.$$

This latter can be easy solved, and as a result we find a solution of Equation (5.7.57)

$$u = x_0 r - \frac{(x_0^2 - 2mr)^{3/2} + x_0^3}{3m}, \qquad \left(r \le \frac{x_0^2}{2m}\right). \tag{5.7.65}$$

Operators (5.7.60) give the ansatz

$$\vec{u} = \vec{\varphi}(x_0) + \frac{\vec{x}}{x_0}, \qquad \rho = \varphi^0(x_0) + x_0^{-n-1} \vec{\alpha} \vec{x}$$
 (5.7.66)

which reduces the gas-dynamics equations to the system of ODEs

$$\begin{cases} x_0 \vec{\varphi_0} + \vec{\varphi} + \lambda x_0^{-n} \vec{\alpha} = 0, \\ x_0 \varphi_0^0 + n \varphi^0 + x_0^{-n} \vec{\alpha} \vec{\varphi} = 0. \end{cases}$$
(5.7.67)

This latter ODE can be easily integrated, and as a result we find a solution of Equations (5.7.59)

$$\begin{cases} \vec{u} = (\vec{x} + \vec{c})x_0^{-1} + \frac{\lambda}{n-1}x_0^{-n}\vec{\alpha}, \\ \rho = c_0 x_0^{-n} + \vec{\alpha}(\vec{x} + \vec{c})x_0^{-n-1} + \frac{\lambda \vec{\alpha}^2 x_0^{-2n}}{n(n-1)}. \end{cases}$$
(5.7.68)

We do not succeed in generalizing operator (5.7.62) for the multi-dimensional case, but the ansatz

$$u^{1} = x_{1}\varphi(x_{0}) + \frac{2\lambda}{x_{1}}$$
(5.7.69)

is easily generalized as

$$\vec{u} = \vec{x} \left[\varphi(x_0) + \frac{2\lambda(2-n)}{\vec{x}^2} \right].$$
 (5.7.70)

Ansatz (5.7.70) reduces the multi-dimensional Navier-Stokes equation (5.7.61) to the ODE

$$\varphi' + \varphi^2 = 0$$

(it will be note, that in this case reduction is made on independent and dependent variables) whence follows a solution of Equations (5.7.61)

$$\vec{u} = \vec{x} \left[\frac{1}{x_0} + \frac{2\lambda(2-n)}{\vec{x}^2} \right].$$
 (5.7.71)

Remark 5.7.2. Since some reduced equations have wider symmetry than the initial equation, we can use this wider symmetry to construct nontrivial formulae of GS. In particular, the nonlinear Laplace equation (5.7.63), under n = 6, is conformally invariant, and if it has a solution $u_I(x)$, then

$$u_{II}(x) = x_0 \sigma^2 u_I \left(\frac{\vec{x} + \vec{\theta} \vec{x}^2}{\sigma}\right) + \frac{2}{\lambda x_0}, \qquad (\sigma = 1 + 2\vec{\theta} \vec{x} + \vec{\theta}^2 \vec{x}^2) \qquad (5.7.72)$$

will be also its solution.

6. Conditional symmetry and exact solutions of the Boussinesq equation

It is known that the maximal invariance algebra of the Boussinesq equation

$$u_{00} = \frac{1}{2}\Delta u^2 + \Delta^2 u = 0, \quad u = u(x), \quad x = (x_0, \vec{x}) \in \mathbb{R}^{1+n}$$
(5.7.73)

is the extended Euclid algebra $A\widetilde{E}(1,n)$ with basis elements

$$\partial_0 \equiv \frac{\partial}{\partial x_0}, \quad \partial_a \equiv \frac{\partial}{\partial x_a}, \quad J_{ab} = x_a \partial_b - x_b \partial_a, \quad D = 2x_0 \partial_0 + x_a \partial_a - 2u \partial_u.(5.7.74)$$

All inequivalent ansatze which reduce the two-dimensional (n = 1) equation (5.7.73) to the ODE constructed with the help of operators (5.7.74) are as follows

$$u = \varphi(\omega), \quad \omega = \alpha_0 x_0 + \alpha_1 x_1, \quad \alpha_0, \alpha_1 = \text{const};$$

$$u = \frac{1}{x_0} \varphi(\omega), \qquad \omega = \frac{x_1}{\sqrt{x_0}}.$$
 (5.7.75)

Olver and Rosenau [160] considered ansatz

$$u = \varphi(\omega) - 4\mu^2 x_0^2, \quad \omega = x_1 + \mu x_0^2, \quad \mu = \text{const},$$
 (5.7.76)

which reduces the two-dimensional equation (5.7.73) to the ODE

$$\ddot{\varphi} + \varphi \dot{\varphi} + 2\mu \varphi = 8\mu^2 \omega + c_1. \tag{5.7.77}$$

Ansatz (5.7.76) is invariant under the operator

$$Q = \partial_0 - 2\lambda x_0 \partial_1 - 8\lambda^2 x_0 \partial_u, \quad \lambda = -2\mu, \tag{5.7.78}$$

which does not belong to the invariance algebra (5.7.74), and therefore it is a non-Lie ansatz. Below (following [83^{*}]) we systematically describe in terms of conditional symmetry non-Lie ansatze of type (5.7.76) which reduce the two-dimensional Boussinesq equation (5.7.73) to ODEs. Similar results were independently obtained by Levi and Winternitz [85^{*}].

It will be noted that Clarkson and Kruskal [84*] have described ansatze

$$u(x) = f(x)\varphi(\omega) + g(x) \tag{5.7.79}$$

which reduce the two-dimensional Boussinesq equation to ODEs using direct substitution. The main difference of our approach from that of theirs consists crucially in using the conditional invariance of the equation which allows us
to obtain not only the ansatze found in $[84^*]$ but other ones which cannot be obtained within the framework of the direct method $[84^*]$.

So, consider the two-dimensional Boussinesq equation

$$u_{00} + uu_{11} + u_1^2 + u_{1111} = 0. (5.7.80)$$

Theorem 5.7.12. [83*] Equation (5.7.80) is Q-conditionally invariant under the operator

$$Q = A(x)\partial_0 + B(x)\partial_1 + [\alpha(x)u + \beta(x)]\partial_u \qquad (5.7.81)$$

if functions A(x), B(x), $\alpha(x)$, $\beta(x)$ satisfy the following equations:

1)
$$A \neq 0$$
 (without loss of generality one can put $A = 1$)
 $\alpha = -2B_1, \quad \alpha_1 = B_{11} = 0, \quad \beta = -2B(B_0 + 2BB_1),$
 $\beta_1 = \frac{1}{2}B_{00} + (\alpha B)_0 + B_1(B_0 - 2BB_1 + \alpha B),$
 $\beta_{11} = -(\partial_0 + 4B_1)(\alpha_0 + \alpha^2),$ (5.7.82)
 $\beta_{00} - 2B_0\beta_1 + 4B_1(\beta_0 - \beta_1 B + \alpha\beta) + 2\alpha_0\beta = 0;$

A = 0, B = 1,

$$\begin{aligned} \alpha_0 &= 0, \quad \alpha_{11} + 5\alpha\alpha_1 + 2\alpha^3 = 0, \\ \beta_{11} + 3\alpha\beta_1 + 4\alpha^2\beta + 5\alpha_1\beta + 5\alpha_{11}(\alpha^2 - \alpha_1) + 5\alpha\alpha_1(\alpha_1 + 2\alpha^2) = 0, \\ \beta_{1111} + 4\alpha_{111}\beta + 6\alpha_{11}(\beta_1 + \alpha\beta) + 4\alpha_1[(\alpha^2 + \alpha_1)\beta + (5.7.83) \\ &+ (\beta_1 + \alpha\beta)_1 + \beta_{00} + 3\beta\beta_1 + 2\alpha\beta^2] = 0. \end{aligned}$$

The proof is obtained by direct calculation according to formula (5.7.7).

In the first case there exists the general solutions of Equations (5.7.82) and it results in the operator

$$Q = \partial_0 + (ax_1 + b)\partial_1 - 2[au + a(a' + 2a^2)x_1^2 + (a'b + ab' + 4a^2b)x_1 + b(b' + 2ab)]\partial_u,$$
(5.7.84)

where $a = a(x_0)$, $b = b(x_0)$ satisfy the ODEs

$$a'' + 2aa' - 4a^3 = 0, \quad b'' + 2ab' - 4a^2b = 0$$
 (5.7.85)

Depending on explicit form of a, b there are several inequivalent operators

$$Q_{1} = \partial_{0} + x_{0}\partial_{1} - 2x_{0}\partial_{u}, \quad (a = 0, \ b = x_{0});$$

$$Q_{2} = x_{0}\partial_{0} - (x_{1} + 6x_{0}^{5})\partial_{1} + 2[u + 3(x_{1}^{2}x_{0}^{-2} - 24x_{0}^{8} + 2x_{1}x_{0}^{3})]\partial_{u},$$
(5.7.86)

$$\begin{pmatrix} a = -\frac{1}{x_0}, b = 6x_0^5 \end{pmatrix}; Q_3 = 2x_0\partial_0 + (x_1 - 3x_0^2)\partial_1 - 2(u - 3x_1 + 9x_0^2)\partial_u, \quad \left(a = \frac{1}{2x_0}, b = -\frac{3}{2}x_0\right); Q_4 = 2W\partial_0 + W'[x_1\partial_1 - (2u + Wx_1^2)\partial_u], \quad \left(a = \frac{W'}{2W}, b = 0\right); Q_5 = 2W\partial_0 + W'(x_1 + \rho)\partial_1 - [2W'u + WW'(x_1 + \rho)^2 + x_1 + \rho]\partial_u, \left(a = \frac{1}{2}\frac{W'}{W}, b = a\rho, \rho = \int \frac{W}{(W')^2}dx_0\right),$$

where $W = W(x_0)$ is the Weierstrass elliptic function satisfying the ODE $W'' = W^2$ or $(W')^2 = \frac{2}{3}(W^3 + \text{const})$.

In the second case we succeeded in obtaing several partial solutions of Equations (5.7.83) which results in the following operators

$$Q_{6} = x_{0}^{2}\partial_{1} + (x_{0}^{5} - 2x_{1})\partial_{u}, \quad \alpha = 0, \ \beta = x_{0}^{3} - 2x_{1}x_{0}^{-2};$$

$$Q_{7} = \partial_{1} + (\Lambda - \frac{1}{3}Wx_{1})\partial_{u}, \quad \alpha = 0, \ \beta = \Lambda + \frac{1}{3}Wx_{1};$$

$$Q_{8} = x_{1}\partial_{1} + 2u\partial_{u}, \quad \alpha = \frac{2}{x_{1}}, \ \beta = 0;$$

$$Q_{9} = x_{1}^{3}\partial_{1} + 2(x_{1}^{2}u + 24)\partial_{u}, \quad \alpha = \frac{2}{x_{1}}, \ \beta = \frac{48}{x_{1}^{3}},$$
(5.7.87)

where $\Lambda = \Lambda(x_0)$ is the Lame function satisfying equation $\Lambda'' = W\Lambda$. Using operators (5.7.86), (5.7.87) we find ansatze:

$$1^{\circ}. \quad u = \varphi(\omega) - 4x_{0}^{2}, \quad \omega = x_{1} + x_{0}^{2}; \qquad (5.7.88)$$

$$2^{\circ}. \quad u = x_{0}^{2}\varphi(\omega) - \left(\frac{x_{1}}{x_{0}} + 6x_{0}^{4}\right)^{2}, \quad \omega = x_{0}(x_{1} + x_{0}^{5});$$

$$3^{\circ}. \quad u = x_{0}^{-1}\varphi(\omega) + 2(x_{1} - x_{0}^{2}), \quad \omega = \frac{x_{1} + x_{0}^{2}}{\sqrt{x_{0}}};$$

$$4^{\circ}. \quad u = \frac{1}{W}\varphi(\omega) - \frac{1}{6}Wx_{1}^{2}, \quad \omega = \frac{x_{1}}{\sqrt{W}};$$

$$5^{\circ}. \quad u = \frac{1}{W}\varphi(\omega) - \frac{1}{4}\left(\frac{W'}{W}\right)^{2}(x_{1} + \rho)^{2}, \quad \omega = \frac{x_{1}}{\sqrt{W}} - \frac{1}{2}\int \frac{W'}{W^{3/2}}\rho dx_{0};$$

$$6^{\circ}. \quad u = \varphi(\omega) - x_{0}^{-2}x_{1}^{2} + x_{0}^{3}x_{1}, \quad \omega = x_{0};$$

$$7^{\circ}. \quad u = \varphi(\omega) - \frac{1}{6}x_{1}^{2}W + \Lambda x_{1}, \quad \omega = x_{0};$$

8°.
$$u = x_1^2 \varphi(\omega), \quad \omega = x_0;$$

9°. $u = x_1^2 \varphi(\omega) - 12x_1^{-2}, \quad \omega = x_0.$

After substitution of ansatze (5.7.88) into (5.7.80) we get, respectively, the following reduced ODEs:

1°.
$$\varphi''' + \varphi \varphi' + 2\varphi = 8\omega + c_1;$$
 (5.7.89)
2°. $\varphi''' + \varphi \varphi' + 30\varphi = 1800\omega + c_2;$
3°. $\varphi'''' + (\varphi + \frac{1}{4}\omega^2)\varphi'' + (\varphi')^2 + \frac{7}{4}\omega\varphi' + 2\varphi = 0;$
4°. $\varphi'''' + \varphi \varphi'' + (\varphi')^2 + \frac{\lambda}{6}(\omega^2\varphi'' + 7\omega\varphi' + 8\varphi) = 0;$
5°. $\varphi'''' + \varphi \varphi'' + (\varphi')^2 + \frac{\lambda}{2}(\omega\varphi' + 2\varphi - \lambda\omega) = 0;$
6°. $\varphi'' - \frac{2}{\omega^2}\varphi + \omega^6 = 0;$
7°. $\varphi'' - \frac{1}{3}W\varphi + \Lambda^2 = 0;$
8°, 9°. $\varphi'' + 6\varphi^2 = 0,$

where c_1 , c_2 are arbitrary constants.

Having solved the reduced Equations (5.7.89) and using once more formulae (5.7.88) we obtain solutions of Equation (5.7.80):

$$u = -\frac{1}{6}x_1^2 W(x_0), \quad u = -12x_1^{-2}, \tag{5.7.90}$$

$$u = -\frac{1}{6}x_1^2 W(x_0) - 12x_1^{-2}, \quad u = 2(x_1 - x_0^2)$$

$$u = 2(x_1 - x_0^2) - \frac{12}{(x_1 + x_0^2)^2}, \quad u = \frac{x_1^2}{x_0^2} - 6c_3^2 x_0^8 + 18c_3 x_0^3 x_1 + \frac{c_4}{x_0} + c_5 x_0^2.$$

Now let us give some results on studying conditional symmetry of the multidimensional Boussinesq equation.

Theorem 5.7.13. [83*] Equation (5.7.73) under n = 6 is invariant with respect to the conformal algebra AC(6) with basis elements

$$\partial_{a} = \frac{\partial}{\partial x_{a}}, \quad J_{ab} = x_{a}\partial_{b} - x_{b}\partial_{a}, \quad D = x_{a}\partial_{a} - 4u\partial_{u},$$

$$K_{a} = 2x_{a}D - x^{2}\partial_{a}, \quad a, b = 1, 2, \dots, 6, \quad x^{2} \equiv x_{a}x_{a},$$
(5.7.91)

if

$$\Delta u + \frac{1}{2}u^2 = 0. \tag{5.7.92}$$

An ansatz obtained by means of operator K_a has the form

$$u = \frac{1}{x^2}\varphi(\omega_1, \omega_2), \quad \omega_1 = x_0, \quad \omega_2 = \frac{bx - b^2 x^2}{x^4},$$
 (5.7.93)

where b_a are arbitrary constants. The corresponding reduced equations are

$$\varphi_{11} = 0, \quad 2\omega_2\varphi_{22} + 5\varphi_2 = \frac{1}{4b^2}\varphi^2$$
 (5.7.94)

Partial solution of Equation (5.7.94) is $\varphi = -4b^2\omega_2^{-1}$ and it results in the following solution of the Boussinesq equation (5.7.73) under n = 6:

$$u = \frac{4}{x^2 - (\alpha x)^2}, \qquad \alpha_a = \text{const}, \quad \alpha^2 = 1.$$

7. In this point we give a brief review of conditional symmetry carried out by us and our collaborators up to 1991.

1°. Nonlinear wave equation

$$\Box u = \frac{n}{u} \tag{5.7.95}$$

is conformally invariant under the condition (see Appendix 6)

$$1 - u_{\nu}u^{\nu} = 0. \tag{5.7.96}$$

2°. Nonlinear wave equation [100*]

$$\Box u = \lambda_1 u^k \tag{5.7.97}$$

is conformally invariant under the condition

$$u_{\nu}u^{\nu} = \lambda_2 u^{k+1}. \tag{5.7.98}$$

The Liouville equation

$$\Box u = \lambda_1 e^u \tag{5.7.99}$$

is conformally invariant under the condition

$$u_{\nu}u^{\nu} = \lambda_2 e^u. \tag{5.7.100}$$

The Monge-Ampere equation

$$\det(u_{\mu\nu}) = 0 \tag{5.7.101}$$

344

is conformally invariant under the condition (5.7.96).

3°. Equations of the form

$$u_{00} - \vec{\nabla} \left[f(u) \vec{\nabla} u \right] = F(x, u, u)$$
(5.7.102)

were considered in [108*] and it has been established that the following equations -1

$$u_{00} - \partial_1 (u^{-2}u_1) = -u^{-1},$$

$$u_{00} - \partial_1 (\operatorname{ch}^{-2}u u_1) = -(u_0 + \operatorname{th} u),$$

$$u_{00} - \partial_1 (\cos^{-2}u u_1) = -(u_0 + \tan u),$$

$$u_{00} - \vec{\nabla} (e^{2u}\vec{\nabla}u) = 0$$

(5.7.103)

are invariant under the Poincare algebra if the conditions

$$u_0^2 - u^{-2}u_1^2 = 1,$$

$$u_0^2 - \operatorname{ch}^{-2}u \, u_1^2 = 1,$$

$$u_0^2 - \cos^{-2}u \, u_1^2 = 1,$$

$$u_0^2 - e^{2u}(\vec{\nabla}u)^2 = 1$$

(5.7.104)

hold true, respectively.

Equations

$$u_{00} - \partial_1 (u^{-2} u_1) = -u^{-2} u_1 \cot a x_1,$$

$$u_{00} - \partial_1 (ch^{-2} u_1) = 2 sh^{-1} 2u,$$

$$u_{00} - \partial_1 (cos^{-2} u_1) = 2 sin 2u,$$

$$u_{00} - \vec{\nabla} (e^{2u} \vec{\nabla} u) = -n(u_0 th x_0 + 1)$$

(5.7.105)

are conformally invariant under the conditions (5.7.104), respectively.

4°. Equation of nonlinear acoustic

$$u_{01} - \partial_1(f(u)u_1) - u_{22} - u_{33} = 0$$
 (5.7.106)

which coincides, under f(u) = u, with the Khokhlov-Zabolotskaya equation

$$u_{01} - \partial_1(uu_1) - u_{22} - u_{33} = 0 \tag{5.7.107}$$

was considered in $[58^*, 106^*]$. In particular, there it is established that Equation (5.7.107) under the condition

$$u_0 u_1 - u u_1^2 - u_2^2 - u_3^2 = 0 (5.7.108)$$

is invariant under infinite-dimensional algebra of special form.

It will be noted that Equations (5.7.103) under corresponding conditions (5.7.104) are locally equivalent to the system

$$\Box w = 0, (5.7.109) \\ w_{\nu} w^{\nu} = 1, (5.7.109)$$

and Equations (5.7.105) under corresponding conditions (5.7.104) are locally equivalent to the system

$$\Box w = \frac{n}{w},$$

$$w_{\nu}w^{\nu} = 1.$$
(5.7.110)

 5° . Q-conditional symmetry of the nonlinear wave equation

$$u_{00} - \partial_1(f(u)u_1) = g(u) \tag{5.7.111}$$

under

$$\begin{split} f(u) &= \langle \lambda e^u, \, \lambda u^k, \, \lambda u, \, \lambda/\sqrt{u}, \, \lambda u^{-2/3}, \, e^u + \lambda^2, \, \lambda u^{-4/5}, \, \lambda u^4, \, \lambda u^{-4/3}, \, \lambda u^{-4} \rangle, \\ g(u) &= \langle 0, \, \lambda u, \, \lambda u^{-1/3}, \, \lambda u^{-(2k+1)}, \, \lambda e^{2u} \rangle \end{split}$$

was studies in [94*, 105*].

6°. In [192] it is established that the d'Alembert equation

 $\Box u = 0$

is invariant under the infinite-dimensional algebra $\widetilde{P}^{\infty}(1,n)$ if the condition

$$u_{\nu}u^{\nu}=0$$

holds true.

7°. In $[56^*, 90^*, 129^*]$ was treated the Q-conditional invariance of the nonlinear heat equation

~

$$u_0 + u_{11} = F(u), \tag{5.7.112}$$

which especially distinguished the following cases

$$u_{0} + u_{11} = \lambda u^{3},$$

$$u_{0} + u_{11} = \lambda (u^{3} - u),$$

$$u_{0} + u_{11} = \lambda (u^{3} - u),$$

$$u_{0} + u_{11} = \lambda (u^{3} - 3u + 2)$$

(5.7.113)

346

8°. In [91*,104*] we considered the nonlinear heat equation

$$u_0 + \vec{\nabla}(f(u)\vec{\nabla}u) = F(u).$$
 (5.7.114)

Generators of Q-conditional symmetry for the equations

$$u_0 + \partial_1(e^u u_1) = 0,$$

$$u_0 + \partial_1(u^{-1/2} u_1) = 0,$$
(5.7.115)

were found, and it was established that the equation

$$u_0 + \vec{\nabla}(e^u \vec{\nabla} u) + \lambda e^{-u} = 0$$
 (5.7.116)

under the condition

$$u_0 + \frac{n}{2}e^u(\vec{\nabla}u)^2 + \frac{\lambda}{2}e^{-u} = 0$$
 (5.7.117)

is conformally invariant.

9°. Conditional symmetry of the Kadomtsev-Petviashvili equation

$$u_{01} + (u^2)_{11} + u_{22} + u_{1111} = 0 (5.7.118)$$

is studied in [93*].

10°. In [92*] is considered the generalized Korteweg-de Vries equation

$$u_0 + F(u)u_1^k + u_{111} = 0. (5.7.119)$$

It is shown that Equation (5.7.119) is Q-conditionally invariant with respect to the Galilei algebra under the conditions

$$F(u) = \lambda_1 u^{(2-k)/2} + \lambda_2 u^{(1-k)/2},$$

$$F(u) = u^{1-k} (\lambda_1 \ln u + \lambda_2),$$

$$F(u) = (1 - u^2)^{(1-k)/2} (\lambda_1 \arcsin u + \lambda_2),$$

$$F(u) = (1 + u^2)^{(1-k)/2} (\lambda_1 \operatorname{arcsh} u + \lambda_2).$$
(5.7.120)

11°. In [99*,101*] it is established that the Boussinesq equation

$$u_0 + \Delta u^2 = 0 \tag{5.7.121}$$

is Q-conditionally invariant under the Galilei algebra.

12°. Q-conditional symmetry of the Born-Infeld equation (5.7.35) is studied in [102*].

13°. In [107*] it is established that the nonlinear polywave equation

$$\Box^m u = \lambda u^{1-2m} \tag{5.7.122}$$

is invariant under the conformal algebra provided

$$\Box u = \frac{n}{u}, \quad u_{\nu}u^{\nu} = 1.$$

14°. In $[95^*, 96^*]$ Q-conditional symmetry of the gas dynamics equations is studied:

$$u_0^1 - u_1^2 = 0,$$

$$u_0^2 - u_1^3 = 0,$$

$$u_0^3 - F(u^1, u^3)u_1^2 = 0$$

(5.7.123)

under

$$F(u^1, u^3) = \langle \lambda u^1, \lambda(u^1)^k, \lambda(u^1)^{-2}, \lambda(u^1)^{-1}, \lambda(u^1)^{-3/2}, \lambda(u^1)^2, \lambda(u^3)^{-1} \rangle.$$

15°. In [31*, 137*] it is shown that the nonlinear Dirac equation

$$(i\gamma\partial - \lambda\overline{\psi}\psi)\psi = 0 \tag{5.7.124}$$

under the condition $\overline{\psi}\psi = 1$ is invariant with respect to the operators

$$Q_1 = i\partial_0 - \lambda\gamma_0, \quad Q_2 = i\partial_3 + \lambda\gamma_3. \tag{5.7.125}$$

16°. In $[57^*,103^*,109^*]$ it is studied the conditional symmetry of the nonlinear Schrödinger equation

$$\left(P_0 - \frac{\vec{P}^2}{2m}\right)\psi = F(|\psi|)\psi.$$
(5.7.126)

It is shown that Equation (5.7.126) with the nonlinearity

$$F(|\psi|) = \lambda_1 |\psi|^{-k} + \lambda_2 |\psi|^k$$
(5.7.127)

is invariant under the operator

$$Q = x_a P_a - \frac{2}{k} I + \ln \frac{\psi}{\psi^*} \left(\psi \partial_{\psi} - \psi^* \partial_{\psi} \right), \qquad (5.7.128)$$

provided

$$\lambda_1 \Delta |\psi| + \lambda_2 |\psi|^{k+1} = 0.$$
 (5.7.129)

5.8 Nonlocal symmetry of quasirelativistic wave equation

Consider a PDE of fourth order (on spatial variables) which, following [60,75], we shall call a *quasirelativistic* wave equation

$$P_0\psi = H(P)\psi, \qquad H(P) \equiv a_0 \frac{P^2}{2m_0} + a_4 \frac{P^4}{8},$$
 (5.8.1)

where $\psi = \psi(x)$ is a complex scalar function: $x = \{x_0 \equiv t, x_a\} \in \mathbb{R}^4$; a_0, a_4, m_0 are arbitrary real constants;

$$P_0 = i \frac{\partial}{\partial t}, \qquad P_a = -i \frac{\partial}{\partial x_a}, \qquad P^2 = P_a P_a, \qquad a = 1, 2, 3.$$
 (5.8.2)

Under $a_0 = a_4 = 0$ Equation (5.8.1) coincides with the well-known Schrödinger equation, and under $a_0 = m_0$, $a_4 = -m_0^3$ the Hamiltonian of (5.8.1) represents by itself the first three terms of Taylor expansion of the relativistic Hamiltonian

$$\mathcal{H}(P) = \left(m_0^2 + P^2\right)^{1/2} \tag{5.8.3}$$

By means of Lie's method one can make sure that the maximal IA of Equation (5.8.1) is the 8-parameter Lie algebra $A_8 = AE(1,3)$, with basis elements

$$P_0, P_a, J_{ab} = x_a P_b - x_b P_a, \quad I$$

This result means that Equation (5.8.1) is invariant neither under Lorentz transformations nor under Galilean ones. But, of course, it does not mean that symmetry properties of Equation (5.8.1) are exhausted by the first-order differential operators of AE(1,3). Equation (5.8.1) possesses a wide nonlocal symmetry.

Theorem 5.8.1 [60,75]. Equation (5.8.1) is invariant under the 20-dimensional Lie algebra $A_{20} \supset A_8$ with basis elements

$$P_{0}, P_{a}, J_{ab}, I,$$

$$V_{a} = i[H, x_{a}] = \left(\frac{1}{m_{0}} + \frac{a_{4}}{2}P^{2}\right)P_{a},$$

$$G_{a} = tV_{a} - x_{a},$$

$$R_{ab} = -a_{4}(P_{a}P_{b} + \frac{1}{2}\delta_{ab}P^{2}).$$
(5.8.4)

Proof. One can directly verify that operators (5.8.4) satisfy invariance condition (10). Operators P_0 , P_a , V_a , R_{ab} , and I commute and together with operators J_{ab} , G_a satisfy the following commutation relations

$$\begin{split} [J_{ab}, J_{cd}] &= i(\delta_{ac}J_{bd} + \delta_{bd}J_{ac} - \delta_{ad}J_{bc} - \delta_{bc}J_{ad}), \\ [J_{ab}, G_c] &= i(\delta_{ac}G_b - \delta_{bc}G_a), \qquad [J_{ad}, P_0] = 0, \\ [P_a, G_b] &= i\delta_{ab}I, \qquad [G_a, G_b] = 0, \\ [P_0, G_a] &= iV_a, \qquad [J_{ab}, P_c] = i(\delta_{ac}P_b - \delta_{bc}P_a), \\ [V_a, G_b] &= i(R_{ab} - \delta_{ab}a_2I), \\ [J_{ab}, R_{cd}] &= i(\delta_{ac}R_{bd} + \delta_{bd}R_{ac} - \delta_{bc}R_{ad} - \delta_{ad}R_{bc}), \\ [J_{ab}, V_c] &= i(\delta_{ac}V_b - \delta_{bc}V_a), \\ [G_a, R_{bc}] &= ia_4(\delta_{ab}P_c + \delta_{bc}P_a + \delta_{ac}P_b), \end{split}$$

and thereby form a Lie algebra. It will be noted that operators V_a , G_a , R_{ab} belong to the class of third-order differential operators and hence they generate nonlocal transformations.

Using formulae (5.3.6) we calculate final transformations generated by the operator

$$Q = imv_a G_a = t\left(1 + \frac{m_0}{2}a_4 P^2\right)v_a \partial_a - im\vec{x}\cdot\vec{v}, \qquad (5.8.6)$$

where v_a are arbitrary real parameters. So, we have

$$t' = e^{tv_a \partial_a} t e^{-tv_a \partial_a} = t,$$

$$x'_b = e^{tv_a \partial_a} x_b e^{-tv_a \partial_a} = x_b + v_b t,$$
(5.8.7)

 \mathbf{and}

$$\psi'(x') = e^{tv_a \partial_a} e^{-Q} \psi(x) \tag{5.8.8}$$

Since

$$-[tv_a\partial_a, Q] = im_0v^2t,$$

$$[tv_a\partial_a, im_0v^2t] = [Q, im_0v^2t] = 0,$$

one can use formula (5.3.39). Applying it to (5.8.8) one gets

$$\psi'(x') = \exp\left\{tv_a\partial_a - Q + \frac{i}{2}m_0v^2t\right\}\psi(x) = = \exp\left\{im\left[x_av_a + \frac{v^2}{2}t - \frac{1}{2}a_4tP^2v_aP_a\right]\right\}\psi(x).$$
(5.8.9)

~

So, we can conclude that Equation (5.8.1) is invariant under Galilean transformations (5.8.7), provided ψ -function is transformed according to (5.8.9). Unlike Galilean transformations given in the Table 4.1.2, the transformation (5.8.9) is nonlocal. It becomes local when $a_4 = 0$, and in this case it coincides with the standard Galilean transformation of the Schrödinger wave function.

Now we will show that Equation (5.8.1) describes uniform motion of a particle with rest mass m_0 and takes into account dependence of the full mass of the particle on its velocity. But before doing this we shall demonstrate that the relativistic integrodifferential wave equation $P_0\psi = \mathcal{H}(P)\psi$ with the Hamiltonian given in (5.8.3) describes the well-known effect of the growing mass of a particle by increasing its velocity according to the law

$$m = \frac{m_0}{\sqrt{1 - v^2}}.$$
 (5.8.10)

With the help of formulae (A.3.5), (A.3.6) one easily finds

$$\dot{x}_{a} \equiv V_{a} = i[\mathcal{H}(P), x_{a}] = i\left[\sqrt{P^{2} + m_{0}^{2}}, x_{a}\right] = \frac{P_{a}}{\sqrt{P^{2} + m_{0}^{2}}};$$
(5.8.11)
$$\ddot{x}_{a} = \dot{V}_{a} = i[\mathcal{H}(P), V_{a}] = 0,$$

where dot means differentiation with respect to t. Let us recall that in (5.8.11) quantities x_a , $\dot{x}_a \equiv V_a$, \dot{V}_a , P_a are operators of coordinates, velocities, accelerations, and of impulse, respectively. The transition to corresponding classical quantities is made by means of the change

$$x_a \to x_a, \qquad P_a = -i \frac{\partial}{\partial x_a} \to p_a$$
 (5.8.12)

So, one finds from (5.8.11), (5.8.12) the expression for particle velocity

$$v_a = \frac{p_a}{\sqrt{p^2 + m_0^2}} \tag{5.8.13}$$

On the other hand, according to the definition of velocity, we have

$$v_a = \frac{p_a}{m} \tag{5.8.14}$$

where m is the full mass of the particle. Equating these two expressions (5.8.13) and (5.8.14) and then solving the obtained equation with respect to m, we get the formula (5.8.10).

In the same spirit we shall act in the case of Equation (5.8.4), we can write according to (5.8.12) the corresponding classical expression

$$v_a = \frac{p_a}{m_0} + \frac{a_4}{2} p^2 p_a. \tag{5.8.15}$$

Substituting (5.8.14) into (5.8.15) we get a cubic equation with respect to m:

$$\frac{m}{m_0} + \frac{a_4}{2}m^3v^2 - 1 = 0. (5.8.16)$$

Let $a_4 < 0$. Solution of Equation (5.8.16) can be looked for in the form

$$m = \frac{m_0}{c} \sin \alpha \tag{5.8.17}$$

where α , c are arbitrary real constants. Substitution of (5.8.17) into (5.8.16) followed by multiplication on 3c, yields

$$3\sin\alpha - 4\sin^3\alpha \left(\frac{3}{8}m_0^3|a_4|\frac{v^2}{c^2}\right) = 3c.$$
 (5.8.18)

Let us define c so that

$$\frac{3}{8}m_0^3|a_4|\frac{v^2}{c^2}=1,$$

that is

$$c = \left(\frac{3}{8}m_0^3|a_4|\right)^{1/2} v \stackrel{\text{def}}{=} \frac{1}{3}w, \qquad w = \left(\frac{3}{2}\right)^{3/2} v \sqrt{m_0^3|a_4|}. \tag{5.8.19}$$

Now we can rewrite Equation (5.8.18) as

$$3c = 3\sin\alpha - 4\sin^3\alpha \equiv \sin 3\alpha$$

whence follows

$$\alpha = \frac{1}{3} \arcsin 3c \equiv \frac{1}{3} \arcsin w \equiv \frac{1}{3} \arctan \frac{w}{\sqrt{1 - w^2}}.$$

Finally, we get the formula

$$m = m_0 \frac{3}{w} \sin\left(\frac{1}{3}\arctan\frac{w}{\sqrt{1-w^2}}\right).$$
 (5.8.20)

Starting from (5.8.20) we can conclude that in mechanics based on Equation (5.8.1) with $a_4 < 0$, as well as in relativistic mechanics, the limiting speed exists. But unlike relativistic mechanics, where the mass of a particle infinitely increases when velocity v tends to its limit $v \to 1$ (see formula (5.8.10)), in the considered case we have according to (5.8.20)

$$m \underset{w \to 1}{\longrightarrow} \frac{3}{2} m_0. \tag{5.8.21}$$

Now consider the case of Equation (5.8.1) with $a_4 > 0$. Solution of Equation (5.8.16) we look for in the form

$$m=\frac{m_0}{c}\operatorname{sh}\alpha.$$

As a result we obtain

$$m = m_0 \frac{3}{w} \operatorname{sh}\left(\frac{1}{3} \ln\left(w + \sqrt{1 + w^2}\right)\right),$$
 (5.8.22)

where w is defined in (5.8.19). So, in mechanics based on the Equation (5.8.1) with $a_4 > 0$ the limiting speed does not exist, and when velocity v aims at infinity the mass of particle tends to zero.

In conclusion, it will be noted that the same analysis can be made for an equation of the type (5.8.1) with the Hamiltonian

$$H(P) = \sum_{n=0}^{N} a_{2n} P^{2n}, \qquad N < \infty.$$

5.9 Are Maxwell's equations invariant under Galilean transformations?

More than eighty years ago Lorentz, Poincare, and Einstein gave negative answer on this question we are about to discuss. At present it is common knowledge that Maxwell equations (ME)

$$\vec{E} \equiv \frac{\partial \vec{E}}{\partial t} = \operatorname{rot} \vec{H}, \qquad \operatorname{div} \vec{E} = 0,$$

$$\vec{H} \equiv \frac{\partial \vec{H}}{\partial t} = -\operatorname{rot} \vec{E}, \qquad \operatorname{div} \vec{H} = 0$$
(5.9.1)

are invariant under the Lorentz transformations and are not invariant under the Galilei ones

$$x'_a = x_a + v_a t, \qquad t' = t; \qquad a = 1, 2, 3,$$
 (5.9.2)

where v_a are arbitrary constants (the speed of inertial frame of reference). The negative answer on the discussed question is connected with the implicit suggestion that the relationship between \vec{E} , \vec{H} and $\vec{E'}$, $\vec{H'}$ is local, that is, the field transformation $\vec{E} \rightarrow \vec{E'}$, $\vec{H} \rightarrow \vec{H'}$ depends only on unprimed fields (and, of course, on parameters v_a) and do not depend on their derivatives. But if we assume that field transformations may be nonlocal, we will have new possibilities, and our question will have a positive answer [111*].

Theorem 5.9.1. Maxwell equations (5.9.1) are invariant under Galilei transformations (5.9.2), provided the electric and magnetic fields are transformed as

$$\vec{E}' = \vec{E} - \vec{v} \times \vec{H} - (\vec{v} \cdot \vec{x}) \operatorname{rot} \vec{H} + O(v^2), \vec{H}' = \vec{H} + \vec{v} \times \vec{E} + (\vec{v} \cdot \vec{x}) \operatorname{rot} \vec{E} + O(v^2).$$
(5.9.3)

Proof. We offer a straightforward proof of this theorem. As follows from (5.9.2)

$$\vec{\nabla}' = \vec{\nabla}, \qquad \partial_{t'} = \partial_t - \vec{v} \cdot \vec{\nabla}$$
 (5.9.4)

Substituting (5.9.3), (5.9.4) into primed ME we get, omitting terms quadratic in v_a :

$$(\operatorname{div} \vec{E})' = \operatorname{div} \vec{E}' = \operatorname{div} \vec{E} - \operatorname{div} (\vec{v} \times \vec{H}) - \operatorname{div} [(\vec{v} \cdot \vec{x}) \operatorname{rot} \vec{H}]$$

= $\operatorname{div} \vec{E} + \vec{v} \cdot \operatorname{rot} \vec{H} - \vec{v} \cdot \operatorname{rot} \vec{H} = \operatorname{div} \vec{E} = 0,$

where we used the identities

$$\operatorname{div}\left(\vec{v}\times\vec{H}\right) = -\vec{v}\cdot\operatorname{rot}\vec{H},$$
$$\operatorname{div}\left[\left(\vec{v}\cdot\vec{x}\right)\operatorname{rot}\vec{H}\right] = \vec{v}\cdot\operatorname{rot}\vec{H}.$$

Further

$$\partial_{t'}\vec{E}' - (\operatorname{rot}\vec{H})' = (\partial_t - \vec{v}\cdot\vec{
abla})\vec{E}' - \operatorname{rot}\vec{H}' = \dot{\vec{E}} - \vec{v}\times\dot{\vec{H}} - (\vec{v}\cdot\vec{x})\operatorname{rot}\dot{\vec{H}} - (\vec{v}\cdot\vec{
abla})\vec{E} - (\vec{v}\cdot\vec{
abla})\vec{E} - \operatorname{rot}(\vec{v}\times\vec{E}) - \operatorname{rot}[(\vec{v}\cdot\vec{x})\operatorname{rot}\vec{E}],$$

(dot means differentiation with respect to t). Taking into account the identities

$$\operatorname{rot} \left(ec{v} imes ec{E}
ight) = -ec{v} \cdot ec{E} + ec{v} \operatorname{div} ec{E},$$

 $\operatorname{rot} \left[\left(ec{v} \cdot ec{x}
ight) \operatorname{rot} ec{E}
ight] = ec{v} \cdot ec{x} \operatorname{rot} \operatorname{rot} ec{E} + ec{v} imes \operatorname{rot} ec{E}$

and using (5.9.1) we get

$$\partial_{t'}\vec{E}' - (\operatorname{rot}\vec{H})' = \vec{E} - \vec{v} \times \vec{H} - (\vec{v} \cdot \vec{x})\operatorname{rot}\vec{H} - (\vec{v} \cdot \vec{\nabla})\vec{E} - \operatorname{rot}\vec{H} + (\vec{v} \cdot \vec{\nabla})\vec{E} - \vec{v}\operatorname{div}\vec{E} - (\vec{v} \cdot \vec{x})\operatorname{rot}\operatorname{rot}\vec{E} - \vec{v} \times \operatorname{rot}\vec{E} = \vec{E} - \operatorname{rot}\vec{H} - \vec{v} \times (\vec{H} + \operatorname{rot}\vec{E}) - (\vec{v} \cdot \vec{x})\operatorname{rot}(\vec{H} + \operatorname{rot}\vec{E}) - \vec{v}\operatorname{div}\vec{E} = 0.$$

In the same way one can test the invariance of the rest of the equations of the system (5.9.1) with respect to Galilei transformations (5.9.2), (5.9.3). Thus the theorem is proved.

Comparing transformations (5.9.2), (5.9.3) with the infinitesimal Lorentz transformations

$$\vec{x}' = \vec{x} + \vec{v}t + O(v^2), \qquad t' = t + \vec{v} \cdot \vec{x} + O(v^2);$$
 (5.9.5)

$$\vec{E}' = \vec{E} - \vec{v} \times \vec{H} + O(v^2), \vec{H}' = \vec{H} + \vec{v} \times \vec{E} + O(v^2)$$
(5.9.6)

one immediately sees that the simplicity of geometrical transformations (5.9.2) results in complexity (nonlocality) of transformations (5.9.3). To clarify the meaning of the result obtained, let us write down the system (5.9.1) in another equivalent form

$$i\frac{\partial\psi}{\partial t} = \mathcal{H}\psi, \qquad \mathcal{H} = i\widehat{\sigma}_2(\vec{S}\cdot\vec{\nabla}),$$

div $\vec{E} = 0, \qquad \text{div } \vec{H} = 0,$ (5.9.7)

where

$$\psi = \psi(x) = \text{column}(E_1 E_2 E_3 H_1 H_2 H_3), \qquad \widehat{\sigma}_2 = i \begin{pmatrix} \widehat{0} & -I_3 \\ I_3 & \widehat{0} \end{pmatrix}$$
 (5.9.8)

 I_3 , $\hat{0}$ are 3×3 unit and zero matrices, respectively;

$$S_1 = \begin{pmatrix} 0 & 0 & 0 \\ 0 & 0 & -i \\ 0 & i & 0 \end{pmatrix}, \quad S_2 = \begin{pmatrix} 0 & 0 & i \\ 0 & 0 & 0 \\ -i & 0 & 0 \end{pmatrix}, \quad S_3 = \begin{pmatrix} 0 & -i & 0 \\ i & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}$$
(5.9.9)

It is not difficult to show that system (5.9.7) is invariant under the following two Poincare algebras

$$P^{I}_{\mu} = P_{\mu} = \frac{\partial}{\partial x^{\mu}}, \qquad \mu = \overline{0,3}, \qquad x^{0} \equiv t,$$

$$J^{I}_{ab} = x_{a}P_{b} - x_{b}P_{a} + i\epsilon_{abc} \begin{pmatrix} S_{c} & 0\\ 0 & S_{c} \end{pmatrix},$$

$$J^{I}_{0a} = x_{0}P_{a} - x_{a}P_{0} + \widehat{\sigma}_{2}S_{a};$$
(5.9.10)

and

$$P_{0}^{II} = -i\mathcal{H}, \qquad P_{a}^{II} = P_{a}^{I}, \qquad J_{ab}^{II} = J_{ab}^{I}, J_{0a}^{II} = tP_{a} - \frac{i}{2}(\mathcal{H}x_{a} + x_{a}\mathcal{H}) + \frac{1}{2}\widehat{\sigma}_{2}S_{a}.$$
(5.9.11)

Operators J_{0a}^{I} generate the well-known Lorentz transformations, while operators J_{0a}^{II} generate another (nonlocal) transformation. Since J_{0a}^{II} are not differential operators of the first order (in the Lie sense), we cannot use Lie's equations to find the corresponding group action. In this case we shall use formulae (5.3.6), according to which we find

$$t' = e^{t(\vec{v}\cdot\vec{\nabla})}te^{-t(\vec{v}\cdot\vec{\nabla})} = t$$

$$\vec{x}' = e^{t(\vec{v}\cdot\vec{\nabla})}\vec{x}e^{-t(\vec{v}\cdot\vec{\nabla})} = \vec{x} + \vec{v}t;$$

(5.9.12)

 $\psi'(x') = \exp\{t(\vec{v}\cdot\vec{\nabla})\} \exp\{-t(\vec{v}\cdot\vec{\nabla}) + \hat{\sigma}_2(\vec{S}\cdot\vec{v}+\vec{x}\cdot\vec{v}\vec{S}\cdot\vec{\nabla})\}\psi(x), \quad (5.9.13)$ where function $\psi(x)$ is defined in (5.9.8). Infinitesimal transformations (5.9.3) result from (5.9.13) in first order of v_a . Final geometrical transformations (5.9.12) coincide with the Galilei ones (5.9.2). As we see, representations of AP(1,3) given by (5.9.10), (5.9.11) are inequivalent because they result in different group actions. But on the manifold of solutions of ME (5.9.7) we have

$$J_{0a}^{I}\psi = J_{0a}^{II}\psi, \qquad (J_{0a}^{I})^{2}\psi = (J_{0a}^{II})^{2}\psi, \dots$$
(5.9.14)

where ψ is an arbitrary solution of ME, and therefore they are equivalent in a sense. The idea of duality of spacetime symmetry of relativistic equations was first put forward in [59] (see also §5.3).

An important application of transformations (5.9.12), (5.9.13) consists in the possibility of correctly introducing the concept of approximate Galilei invariance with absolute time for relativistic equations [111*]. It is obvious that we can interpret transformations (5.9.2), (5.9.3) as an approximate Galilei invariance of ME. Formula (5.9.13) allows us to calculate Galilei transformations for the electromagnetic field in any approximation in v_a . So, for example, the second approximation has the form

$$\vec{E}' = \vec{E} - \vec{v} \times \vec{H} - (\vec{v} \cdot \vec{x} + \frac{1}{2}\vec{v}^2 t) \text{ rot } \vec{H} + \frac{1}{2} \left[\vec{v}^2 \vec{E} - \vec{v} (\vec{v} \cdot \vec{E}) + (\vec{x} \cdot \vec{v}) ((\vec{v} \cdot \vec{\nabla}) \vec{E} - 2\vec{v} \times \text{ rot } \vec{E}) + (\vec{x} \cdot \vec{v})^2 \Delta \vec{E} \right] + O(v^3),$$

$$\vec{H}' = \vec{H} + \vec{v} \times \vec{E} + (\vec{v} \cdot \vec{x} + \frac{1}{2}\vec{v}^2 t) \text{ rot } \vec{E} + (5.9.15) + \frac{1}{2} \left[\vec{v}^2 \vec{H} - \vec{v} (\vec{v} \cdot \vec{H}) + (\vec{x} \cdot \vec{v}) ((\vec{v} \cdot \vec{\nabla}) \vec{H} - 2\vec{v} \times \text{ rot } \vec{H}) + (\vec{x} \cdot \vec{v})^2 \Delta \vec{H} \right] + O(v^3).$$

The Lorentz transformations in second approximation look like

$$\begin{aligned} \vec{x}' &= \vec{x} + \vec{v}t + \frac{1}{2}\vec{v}(\vec{x}\cdot\vec{v}) + O(v^3), \\ t' &= t + \vec{x}\cdot\vec{v} + \frac{1}{2}\vec{v}^2t + O(v^3), \\ \vec{E}' &= \vec{E} - \vec{v}\times\vec{H} + \frac{1}{2}\left[\vec{v}^2\vec{E} - \vec{v}(\vec{v}\cdot\vec{E})\right] + O(v^3), \\ \vec{H}' &= \vec{H} + \vec{v}\times\vec{E} + \frac{1}{2}\left[\vec{v}^2\vec{H} - \vec{v}(\vec{v}\cdot\vec{H})\right] + O(v^3). \end{aligned}$$
(5.9.16)

The main difference between transformations (5.9.2), (5.9.15) and (5.9.16) (as well as between (5.9.2), (5.9.3) and (5.9.5), (5.9.6)) is that the time t in Galilei transformations (5.9.2) is unchanged in contrast with Lorentz transformations (5.9.5), (5.9.16). This advantage of Galilei transformations is due to the non-local law of transformations of \vec{E} and \vec{H} (compare (5.9.3), (5.9.6) and (5.9.15), (5.9.16)).

5.10 Nonlocal spacetime symmetry of the Klein-Gordon equation

Consider the Klein-Gordon wave equation

$$(\Box + m^2)\varphi = 0 \tag{5.10.1}$$

and write it down in equivalent form [59]

$$i\frac{\partial\phi}{\partial t} = H\phi,$$

$$H = \frac{1}{2\omega} [(E^2 + \omega^2)\sigma_1 + (E^2 - \omega^2)i\sigma_2],$$

$$E^2 = -\Delta + m^2,$$

(5.10.2)

where ϕ is the two-component function

$$\phi = \begin{pmatrix} \phi_1 \\ \phi_2 \end{pmatrix}, \qquad \phi_1 = \frac{i}{x} \frac{\partial \varphi}{\partial t}, \qquad \phi_2 = \varphi$$
 (5.10.3)

 $x \neq 0$ is an arbitrary constant, σ_1 and σ_2 are 2 × 2 Pauli matrices (2.1.3). According to Remark 5.3.8 and Theorem 5.3.4, Equation (5.10.2) is invariant under Galilei transformation (5.3.43) (with H given in (5.10.2)).

It will be noted that, in contrast with the Dirac equation, operators

$$J_{0a}^{II} = x_0 P_a - \frac{1}{2} (H x_a + x_a H)$$
(5.10.4)

are nonlocal even on the set of solutions of Equation (5.10.2), where they take the form $[111^*]$

$$J_{0a}^{I} = t\partial_a + x_a\partial_t + \frac{i}{2\omega}(\sigma_1 + i\sigma_2)\partial_a$$
(5.10.5)

Operators (5.10.5) generate transformation (A.3.3) and

$$\phi'(x') = \frac{1}{2} \left[(\sigma_0 + \sigma_3) \operatorname{ch} \theta + \sigma_0 - \sigma_3 - \frac{i}{\mathfrak{a}} (\sigma_1 + i\sigma_2) \frac{\vec{\theta} \cdot \vec{\nabla}}{\theta} \operatorname{sh} \theta \right] \phi(x), \quad (5.10.6)$$

where $\vec{\theta} = \{\theta_1, \theta_2, \theta_3\}, \ \theta = (\theta_1^2 + \theta_2^2 + \theta_3^2)^{1/2}, \ \sigma_0$ is a unit 2×2 matrix. It will be stressed that Equation (5.10.2) is not Lorentz invariant in the Lie sense. As we see, Lorentz transformations of function $\phi(x)$ (5.10.6) are nonlocal and therefore non-Lie.

From (5.10.6) one can derive, according to the general relation (17), the formula of generating new solutions. This formula has the form $[111^*]$

$$\phi_{II}(x) = \frac{1}{2} \left[(1+\sigma_3) \operatorname{sech} \theta + 1 - \sigma_3 + \frac{i}{2} (\sigma_1 + i\sigma_2) \frac{\vec{\theta} \cdot \vec{\nabla}}{\theta} \operatorname{th} \theta \right] \phi_I(x'), \quad (5.10.7)$$

where x' are given in (A.3.3).

5.11 On solutions of the Schrödinger equation, invariant under the Lorentz algebra

The Lie-maximal invariance algebra of the Schrödinger Equation (3.3.24) is given in (3.3.25). The algebra ASch(1,3) does not contain the Lorentz algebra ASO(1,3). But if to look for symmetry operators among non-Lie operators we will have new possibilities. So, in $[115^*]$ it is established that the Schrödinger equation (3.3.24) is invariant under the Lorentz algebra with basis elements belonging to pseudodifferential operators. Indeed, one can make sure that the operators

$$J_{0a} = \frac{1}{2m} (PG_a + G_a P), \qquad J_{ab} = x_a P_b - x_b P_a, \tag{5.11.1}$$

where $P = \sqrt{P_a P_a} (P_a = -i\partial_a, G_a = tP_a - mx_a)$, satisfy invariance condition

$$[\overset{\vee}{S}, J_{\mu\nu}]u = 0; \qquad \overset{\vee}{S} \equiv P_0 - \frac{P^2}{2m}, \qquad \mu, \nu = \overline{0,3}$$
 (5.11.2)

and form the Lorentz algebra ASO(1,3) with commutation relations written in (A.2.2). The action of pseudodifferential operators J_{0a} from (5.11.1) is defined by means of integral Fourier transformation in (1.3.17).

Following $[89^*]$, we look for solutions of the Schrödinger equation (3.3.24), invariant under the operators (5.11.1). The corresponding ansatz is defined from the condition

$$J_{0a}u = 0, \qquad J_{ab}u = 0. \tag{5.11.3}$$

If we go over in (5.11.3) to Fourier transform, we get

$$\begin{bmatrix} \left(\frac{t}{m} - \frac{i}{2k^2}\right)k_a - i\frac{\partial}{\partial k_a} \end{bmatrix} \widetilde{u}(t, \vec{k}) = 0,$$

$$\begin{pmatrix} k_a\frac{\partial}{\partial k_b} - k_b\frac{\partial}{\partial k_a} \end{pmatrix} \widetilde{u}(t, \vec{k}) = 0.$$
(5.11.4)

The general solution of Equations (5.11.4) has the form

$$\widetilde{u}(t,\vec{k}) = (\vec{k}^2)^{-1/4} \exp\left\{-it\frac{\vec{k}^2}{2m}\right\}\varphi(t).$$
(5.11.5)

Substitution of (5.11.5) into the Fourier transform of the Schrödinger equation results in

$$\frac{\partial\varphi}{\partial t} = 0. \tag{5.11.6}$$

358

So, we obtain that the solution of the Schrödinger equation which is invariant under ASO(1,3) (5.11.1)

$$u(t,\vec{x}) = c \int_{R^3} e^{i\vec{k}\cdot\vec{x}} (\vec{k}^2)^{-1/4} \exp\left\{-it\frac{\vec{k}^2}{2m}\right\} d^3k, \qquad (5.11.7)$$

where c is an arbitrary constant. To simplify the expression (5.11.7) let us go over to spherical coordinates, choosing the polar axis along the vector \vec{x} . In this case expression (5.11.7) can be rewritten as

$$u(t,\vec{x}) = c \int_{0}^{\infty} k^{3/2} dk \int_{0}^{\pi} \sin\theta d\theta \int_{0}^{2\pi} d\varphi \exp\left\{ikx\cos\theta - it\frac{\vec{k}^2}{2m}\right\} =$$
$$= \frac{4\pi c}{x} \int_{0}^{\infty} k^{1/2} \exp\left\{-it\frac{\vec{k}^2}{2m}\right\} \sin kx \, dk,$$
(5.11.8)

where $x = \sqrt{\vec{x}^2}$, $k = \sqrt{\vec{k}^2}$. After calculation of the last integral, we finally find [89*]

$$u(t,\vec{x}) = \frac{c\sqrt{x}}{t^{3/2}} e^{i\omega} \left(J_{-1/4}(\omega) + iJ_{3/4}(\omega) \right),$$
(5.11.9)

where $\omega = mx^2/4t$; $J_{-1/4}$, $J_{3/4}$ are Bessel functions. One can make sure that (5.11.9) satisfies Schrödinger equation (3.3.24) and it is solution invariant under the Lorentz algebra (5.11.1).

In conclusion, it will be noted that the substitution $t \rightarrow -it$ transforms the Schrödinger equation (3.3.24) into the heat equation

$$\frac{\partial u}{\partial t} - \frac{1}{2m} \Delta u = 0. \tag{5.11.10}$$

If we do the same transformation with expression (5.11.9), we obtain the following (real) solution of the heat equation (5.11.10):

$$u = \frac{c\sqrt{x}}{t^{3/2}} e^{-\omega} \left(I_{-1/4}(\omega) - I_{3/4}(\omega) \right), \qquad \omega = \frac{mx^2}{4t}, \tag{5.11.11}$$

where c is an arbitrary real constant, $I_{-1/4}$, $I_{3/4}$ are modified Bessel functions.

It will be appropriate to give here the solution of the Schrödinger equation (3.3.24) invariant under 6-dimensional Galilei algebra

$$G_a = tP_a - mx_a, \qquad J_{ab} = x_a P_b - x_b P_a.$$
 (5.11.12)

The corresponding ansatz is (compare with (5.11.5)):

$$u(t,\vec{x}) = \exp\left\{i\frac{m\vec{x}^2}{2t}\right\}\varphi(t).$$
(5.11.13)

Substitution of (5.11.13) into (3.3.24) results in the ODE

$$t\dot{\varphi} + \frac{3}{2}\varphi = 0,$$

the general solution of which is $\varphi = \text{const} \cdot t^{-3/2}$. So, the solution of the Schrödinger equation invariant under the Galilei algebra (5.11.12) is

$$u(t, \vec{x}) = ct^{-3/2} \exp\left\{i\frac{m\vec{x}^2}{2t}\right\},$$
(5.11.14)

and one can compare it with the solution (5.11.9). It will be stressed that solution (5.11.9) is not Lorentz invariant in the standard meaning, because basis elements of the Lorentz algebra (5.11.1), under which (5.11.9) is invariant, are non-Lie (nonlocal).

5.12 On approximate symmetry and approximate solutions of the nonlinear wave equation with a small parameter

Following [62^{*}], we introduce the concept of approximate symmetry and describe all nonlinearities F(u) with which the nonlinear wave equation

$$\Box u + \lambda u^3 + \epsilon F(u) = 0 \tag{5.12.1}$$

 $(\Box = \partial_{\mu}\partial^{\mu}, \mu = \overline{0,3}; \lambda \text{ is an arbitrary constant}, \epsilon \ll 1 \text{ is a small parameter}, u = u(x), x \in R(1,3))$ is approximately scale and conformally invariant.

By means of Lie's method one can make sure that when $F(u) \neq 0$ and $F(u) \neq u^3$, Equation (5.12.1) is invariant under the Poincare group P(1,3) only, because the term $\epsilon F(u)$ breaks down the scale and conformal symmetry of the equation $\Box u + \lambda u^3 = 0$.

Let us represent an arbitrary solution, analytic in ϵ , of Equation (5.12.1) in the form

$$u = w + \epsilon v, \tag{5.12.2}$$

where w and v are some smooth functions of x. After sustitution of (5.12.2) into (5.12.1) and equating to zero the coefficients of the zero and first power of ϵ , we get the following system of PDEs

$$\Box w + \lambda w^3 = 0$$

$$\Box v + 3\lambda w^2 v + F(w) = 0.$$
 (5.12.3)

Definition 5.12.1. We shall call the approximate symmetry of Equation (5.12.1) the exact symmetry of the system (5.12.3).

Theorem 5.12.1. $[62^*]$ Equation (5.12.1) is approximately scale invariant (in the sense of the definition above) if and only if

$$F(u) = \begin{cases} \frac{2\lambda b}{k+1}u^3 + \frac{3\lambda c}{k}u^2 + au^{2-k}, & k \neq 0, -1; \\ 2\lambda bu^3 + 3\lambda cu^2 \ln u + au^2, & k = 0; \\ 2\lambda bu^3 \ln u - 3\lambda cu^2 + au^3, & k = -1 \end{cases}$$
(5.12.4)

(k, a, b, c are arbitrary constants), with the generator of scale transformations having the form

$$D = x\partial - w\partial_w + (kv + bw + c)\partial_v \tag{5.12.5}$$

Proof. Using Lie's algorithm we find from the condition of invariance that the generator of scale transformations should have the form

$$D = x\partial - w\partial_w + \eta^2(v, w)\partial_v,$$

provided that from the invariance of the second equation of system (5.12.3),

$$\eta_{vv}^{2} = \eta_{ww}^{2} = \eta_{wv}^{2} = 0 \Rightarrow \eta^{2} = kv + bw + c,$$

$$2\lambda bw^{3} + 3\lambda cw^{2} + (2 - k)F - w\frac{dF}{dw} = 0.$$
(5.12.6)

The general solution of Equation (5.12.6) is given in (5.12.4). Thus the theorem is proved.

In particular, as follows from the above equation, the equation

$$\Box u + \lambda u^3 + \epsilon u = 0 \tag{5.12.7}$$

is approximately scale invariant and the corresponding generator has the form $D = x\partial - w\partial_w + v\partial_v$. This statement holds true even if $\lambda = 0$.

Theorem 5.12.2. [62*] Equation (5.12.1) is approximately conformally invariant if and only if

$$F(u) = -3\lambda\beta u^2 + au^3, (5.12.8)$$

with the generator of conformal transformations having the form

$$\mathcal{K} = 2cx \left[x\partial - w\partial_w - (v - \beta)\partial_v \right] - x^2 c\partial \qquad (5.12.9)$$

where β , a, c_{μ} are arbitrary constants. The proof of Theorem 5.12.2 is performed in the same spirit as that of the Theorem 5.12.1.

It will be noted that Equation (5.12.1) with F(u) given in (5.12.8) is reduced within to ϵ to the equation $\Box u + \lambda' u^3 = 0$, where $\lambda' = \lambda + \epsilon a$ be means of the transformation $u \to u + \epsilon \beta$. It is clear that the exact solutions of the associated system (5.12.3) produce the approximate solutions of the Equation (5.12.1). To find exact solutions of the system (5.12.3) one may use its symmetry under $\widetilde{P}(1,3)$ and C(1,3) described above, applying the algorithm we have used throughout the book. Below, we consider another possibility. Suppose that in (5.12.2)

$$v = f(w),$$
 (5.12.10)

where f is an arbitrary differentiable function. In this case the system (5.12.3) takes the form

$$\Box w + \lambda w^3 = 0 \tag{5.12.11}$$

$$(w_{\mu}w^{\mu})\ddot{f} + (\Box w)\dot{f} + 3\lambda w^{2}f + F(w) = 0,$$

$$w_{\mu} = \frac{\partial w}{\partial x^{\mu}}, \qquad \dot{f} = \frac{df}{dw}$$
(5.12.12)

From the condition of splitting of Equation (5.12.12) one has to put

$$w_{\mu}w^{\mu} = A(w), \qquad (5.12.13)$$

where A is some function of w. Equation (5.12.13) is compatible with (5.12.11) if $A(w) = \lambda w^4$, that is

$$w_{\mu}w^{\mu} = \lambda w^4 \tag{5.12.14}$$

(see Appendix 6).

Taking account of (5.12.11) and (5.12.14) we rewrite (5.12.12) as

$$\lambda(w^2\ddot{f} - w\dot{f} + 3f) + w^{-2}F(w) = 0.$$
 (5.12.15)

So, if we find function f(w) as solution of Equation (5.12.15), we thereby obtain by means of expressions (5.12.2), (5.12.10) approximate solutions of Equation (5.12.1). It will be noted that a subset of such solutions of Equation (5.12.11) and (5.12.14) is conformally invariant since the corresponding approximate system (5.12.11) and (5.12.15) is conformally invariant. Solutions of Equation (5.12.15) for functions F(w) given in (5.12.4) have the form

$$f(w) = \begin{cases} -\frac{a}{\lambda[k(k+2)+3]w^k} - \frac{b}{k+1}w - \frac{c}{k}, & k \neq -1, 0\\ -c\ln w - bw - \frac{1}{3}\left(2c + \frac{a}{\lambda}\right), & k = 0\\ -w\left(\frac{a}{2\lambda} + b\ln w\right) + c, & k = -1 \end{cases}$$
(5.12.16)

The solution of the system (5.12.11) and (5.12.14) is the function

$$w = \pm \left[\lambda(x_{\nu} + a_{\nu})(x^{\nu} + a^{\nu})\right]^{-1/2}$$
(5.12.17)

362

where a_{ν} are arbitrary constants.

When $\lambda = 0$, the nontrivial condition of splitting of Equation (5.12.12) compatible with the Equation $\Box w = 0$ is

$$w_{\mu}w^{\mu} = 1 \tag{5.12.18}$$

So, in this case we find approximate solutions of equation (5.12.1) be means of expressions (5.12.2) and (5.12.10), where function is determined from the equation

$$\hat{f} + F(w) = 0$$
 (5.12.19)

and w, in turn, is determined from the system

$$\Box w = 0, \qquad w_{\mu}w^{\mu} = 1. \tag{5.12.20}$$

The system (5.12.20) is invariant under the extended Poincare group $\widetilde{P}(1,4)$ and has solution

$$w = \alpha_{\nu} x^{\nu} + a, \qquad \alpha_{\nu} a^{\nu} = 1$$
 (5.12.21)

where α_{ν} , a are arbitrary constants.

In particular, equation

$$\Box u + \epsilon u = 0 \tag{5.12.22}$$

is approximately invariant under the group $\widetilde{P}(1,4)$ on the subset of solutions

$$u = w - \epsilon \left(\frac{1}{6}w^3 + a_1w + a_2\right), \qquad (5.12.23)$$

where w is given in (5.12.21) and a_1 , a_2 are arbitrary constants.

In conclusion, let us note some generalizations of the concept of approximate symmetry studied above. First of all, one can consider higher orders of approximation of u in ϵ , i.e., $u = w + \epsilon v^{(1)} + \epsilon^2 v^{(2)} + \cdots$, and can study the symmetry of the corresponding approximate system of PDEs for functions w, $v^{(1)}$, $v^{(2)}$, and so on. Secondly, one can expand in ϵ -series not only dependent variables, but also independent ones, e.g., $x_0 = t + \epsilon z^{(1)} + \epsilon^2 z^{(2)} + \cdots$, and can construct in this way the corresponding approximate system and then study its symmetry. Another approach to the study of approximate symmetry is to use some special approximates, say the two-point Padé approximants

$$u = \sum_{k=0}^{m} \epsilon^k f_k \left(\sum_{j=0}^{n} \epsilon^j g_j \right)^{-1}, \qquad (5.12.24)$$

where functions f_k , g_j are determined from the condition: when $\epsilon \to 0$ expression (5.12.24) coincides with the expansion

$$u = v^{(0)} + \epsilon v^{(1)} + \epsilon^2 v^{(2)} + \cdots \qquad \epsilon \ll 1,$$

and when $\epsilon \to \infty$, (5.12.24) coincides with the expansion

$$u = w^{(0)} + \epsilon^{-1} w^{(1)} + \epsilon^{-2} w^{(2)} + \cdots \qquad \epsilon \gg 1.$$

Jacobi elliptic functions [3,4,24]

Jacobi elliptic functions E(z, k) (z is the variable argument, k is the parameter) are solutions of the nonlinear ODE

$$\ddot{E} + aE + bE^3 = 0,$$
 $(\dot{E})^2 + aE^2 + \frac{1}{2}bE^4 = c$ (A.1.1)

where dot means differentiation with respect to z; a = a(k). b = b(k), c = c(k) are constants which depend on the parameter k. For the first time elliptic functions were introduced as inverse functions of an elliptic integral. This problem of inversion was solved in 1827 by Jacobi and Abel independently.

There are twelve elliptic functions. The three basic elliptic functions are determined as follows

$$sn(z,k) = sin \varphi$$

$$cn(z,k) = cos \varphi$$

$$dn(z,k) = \frac{d\varphi}{dz} = (1 - k^2 sin^2 \varphi)^{1/2},$$
(A.1.2)

where φ is implicitly defined by the elliptic integral of the first kind

$$z = \int_{0}^{\varphi} \frac{d\tau}{\sqrt{1 - k^2 \sin^2 \tau}}.$$
 (A.1.3)

The rest of the nine elliptic functions are reciprocals of these three functions, and the quotients of any two of them.

Below in Table A.1.1 we present, following [4], the constants a, b, c for the twelve Jacobi elliptic functions.

Table A.1.1. Jacobi elliptic functions and parameters of the Equation (A.1.1).

N	E = E(z, k)	a	b	С
1	sn	$1 + k^2$	$-2k^{2}$	1
2	\mathbf{cn}	$1 - 2k^2$	$2k^2$	$1 - k^2$
3	dn	$-(2-k^2)$	2	$-(1-k^2)$
4	$ns \equiv 1/sn$	$1 + k^2$	-2	k^2
5	$nc \equiv 1/cn$	$1 - 2k^2$	$-2(1-k^2)$	$-k^{2}$
6	$nd \equiv 1/dn$	$-(2-k^2)$	$2(1-k^2)$	-1
7	$sc \equiv sn / cn$	$-(2-k^2)$	$-2(1-k^2)$	1
8	$sd \equiv sn/dn$	$1 - 2k^2$	$2k^2(1-k^2)$	1
9	$cs \equiv cn/sn$	$-(2-k^2)$	-2^{-2}	$1 - k^2$
10	$cd \equiv cn / dn$	$1 + k^2$	$-2k^{2}$	1
11	$ds \equiv dn / sn$	$1 - 2k^2$	-2	$-k^2(1-k^2)$
12	$dc \equiv dn / cn$	$1 + k^2$	-2	k^2

From the above definition of the basic elliptic functions it follows immediately that

$$\operatorname{sn}(0,k) = 0, \qquad \operatorname{cn}(0,k) = \operatorname{dn}(0,k) = 1, \qquad (A.1.4)$$

and

$$\begin{split} \dot{sn} &= cn \, dn \,, \qquad \dot{cn} = -sn \, dn \,, \\ \dot{dn} &= -k^2 \, sn \, cn \,; \\ sn^2 &= 1 - cn^2 = \frac{1 - dn^2}{k^2} \,, \\ cn^2 &= 1 + \frac{dn^2 - 1}{k^2} \,, \\ dn^2 &= 1 - k^2 \, sn^2 = 1 - k^2 + k^2 \, cn^2 \end{split}$$
(A.1.5)

Jacobi elliptic functions are doubly periodic functions of the complex argument z. Specifically,

$$\operatorname{sn}(z + 4N_1\mathcal{K} + i2N_2\mathcal{K}') = \operatorname{sn} z,$$

$$\operatorname{cn}(z + 4N_1\mathcal{K} + 2N_2(\mathcal{K} + i\mathcal{K}')) = \operatorname{cn} z,$$

$$\operatorname{dn}(z + 2N_1\mathcal{K} + i4N_2\mathcal{K}') = \operatorname{dn} z,$$

(A.1.6)

where N_1 and N_2 are any integers and

$$\mathcal{K} = \mathcal{K}(k) = \int_{0}^{\pi/2} \frac{d\tau}{\sqrt{1 - k^2 \sin^2 \tau}},$$

$$\mathcal{K}' = \mathcal{K}'(k) = \mathcal{K}(k'), \qquad k' = \sqrt{1 - k^2}$$
(A.1.7)

are complete elliptic integrals. Under k = 0 and k = 1 one of the periods becomes infinite and the elliptic functions degenerate:

k = 0, then $\mathcal{K}' = \infty$, $\mathcal{K} = \pi/2$,

and $\operatorname{sn} z = \sin z$, $\operatorname{cn} z = \cos z$, $\operatorname{dn} z = 1$;

$$k = 1$$
, then $\mathcal{K} = \infty$, $\mathcal{K}' = \pi/2$, (A.1.8)

and $\operatorname{sn} z = \operatorname{th} z$, $\operatorname{cn} z = \operatorname{dn} z = 1/\operatorname{ch} z$.

Restricting z to real values we see that $\operatorname{sn} z$, $\operatorname{cn} z$, and $\operatorname{dn} z$ have periods $4\mathcal{K}, 4\mathcal{K}, 2\mathcal{K}$, respectively. The shortest period corresponds to k = 0, when $\mathcal{K}(0) = \pi/2$. For k > 0, $\mathcal{K}(k) > \pi/2$. All Jacobi elliptic functions are real for real z and for $0 \le k^2 \le 1$.

The basic functions $\operatorname{sn} z$, $\operatorname{cn} z$, and $\operatorname{dn} z$ are finite everywhere on the real-z axis and have the following zeros on this axis

$$\begin{split} & \mathrm{sn}\,(z,k)=0 \mbox{ at } z=2N\mathcal{K},\\ & \mathrm{cn}\,(z,k)=0 \mbox{ at } z=(2N+1)\mathcal{K},\\ & \mathrm{dn}\,(z,k)\neq 0 \mbox{ for } k<1, \end{split}$$

where N is any integer. Only for parameter k = 1 does dn z have zeros on the real axis, namely, at $z = \pm \infty$.

The Jacobi identities

sn
$$(iz, k) = isc(z, k'),$$
 $(k' = \sqrt{1 - k^2})$
cn $(iz, k) = nc(z, k'),$ (A.1.9)
dn $(iz, k) = dc(z, k')$

enable one to change from real to imaginary arguments or conversely.

As an example of the use of elliptic functions we describe, following [4], elliptic solutions of the nonlinear wave equation

$$\Box u + \lambda u^3 = 0, \tag{A.1.10}$$

where u = u(x) is a scalar function, $x \in \mathbb{R}(1,3)$.

366

Let f(x) be a given explicit solution of Equation (A.1.10) with λ equal to $-a\lambda/b$, that is

$$\Box f - \frac{a\lambda}{b} f^3 = 0. \tag{A.1.11}$$

Then the ansatz

$$u(x) = f(x)E(\omega, k). \tag{A.1.12}$$

where $E(\omega, k)$ is an elliptic function which satisfies ODE (A.1.1), will be a solution of Equation (A.1.10) provided

$$2\frac{\partial f}{\partial x^{\nu}}\frac{\partial \omega}{\partial x_{\nu}} + f\Box\omega = 0,$$

$$\frac{\partial \omega}{\partial x_{\nu}}\frac{\partial \omega}{\partial x^{\nu}} - \frac{\lambda}{b}f^{2} = 0.$$
(A.1.13)

The system (A.1.13) can be considered as defining equations for the argument $\omega = \omega(x)$ of an elliptic function $E(\omega, k)$. If a solution of this system exists then an elliptic generalization of f(x) exists. Solving the conditions (A.1.13) for $\omega(x)$ is the essential step in the construction.

As far as there are 12 Jacobi elliptic functions, one might expect that 12 different elliptic generalizations of f(x) exist. Actually, five of these are redundant. The remaining ones are [4]:

$$u = f \operatorname{sn} \left(\frac{\omega'}{\sqrt{-1 - k^2}}, k \right) = i f \operatorname{sc} \left(\frac{\omega'}{\sqrt{2 - k'^2}}, k' \right),$$

$$u = f \operatorname{cn} \left(\frac{\omega'}{\sqrt{-1 + 2k^2}}, k \right) = f \operatorname{nc} \left(\frac{\omega}{\sqrt{-1 + 2k'^2}}, k' \right),$$

$$u = f \operatorname{dn} \left(\frac{\omega'}{\sqrt{2 - k^2}}, k \right) = f \operatorname{dc} \left(\frac{\omega'}{\sqrt{-1 - k'^2}}, k' \right),$$

$$u = f \operatorname{nd} \left(\frac{\omega'}{\sqrt{-1 - k^2}}, k \right) = i f \operatorname{cs} \left(\frac{\omega'}{\sqrt{-1 - k'^2}}, k' \right),$$

$$u = f \operatorname{nd} \left(\frac{\omega'}{\sqrt{2 - k^2}}, k \right) = f \operatorname{cd} \left(\frac{\omega'}{\sqrt{-1 - k'^2}}, k' \right),$$

$$u = f \operatorname{sd} \left(\frac{\omega'}{\sqrt{-1 + 2k^2}}, k \right) = i f \operatorname{sd} \left(\frac{\omega'}{\sqrt{-1 + 2k'^2}}, k' \right),$$

$$u = f \operatorname{ds} \left(\frac{\omega'}{\sqrt{-1 + 2k^2}}, k \right) = i f \operatorname{ds} \left(\frac{\omega'}{\sqrt{-1 + 2k'^2}}, k' \right),$$

where $\omega' = \sqrt{-a(k)}\omega$, and Jacobi imaginary identities (A.1.9) have been used to obtain the second form of the solution with parameter $k' = \sqrt{1-k^2}$.

$A\widetilde{P}(1,3)$ -nonequivalent one-dimensional subalgebras of the extended Poincare algebra $A\widetilde{P}(1,3)$

Here we would like to demonstrate as example of finding one-dimensional subalgebras of a given algebra. As a given algebra we take the extended Poincare algebra $A\widetilde{P}(1,3)$ whose one-dimensional subalgebras have been used in Paragraph 2.2 in constructing $\widetilde{P}(1,3)$ -nonequivalent ansatze for spinor field.

In what follows, an essential role will be played by the CBH formula, which we write once again for the sake of convenience

$$\exp\{-\theta Q_2\}Q_1 \exp\{\theta Q_2\} = \sum_{n=0}^{\infty} \frac{\theta^n}{n!} \{Q_1, Q_2^n\},$$
 (A.2.1)

where $\{Q_1, Q_2^0\} = Q_1, \{Q_1, Q_2^k\} = [\{Q_1, Q_2^{k-1}\}, Q_2],$

with k = 1, 2, ..., and Q_1, Q_2 are some operators.

Lemma A.2.1. [109,11*] By the transformation

$$Q \to Q' = VQV^{-1}$$

where $V = \exp\{\theta^{\mu\nu} J_{\mu\nu}\}$, the operator

$$Q = C_{\mu\nu} J_{\mu\nu} = A_k M_k + B_k N_k$$

where $M_k = -\frac{1}{2} \epsilon_{kij} J_{ij}$ and $N_k = J_{0k}$ (A_k , B_k are arbitrary constants) can be reduced to one of the following forms

1°
$$Q' = \alpha J_{01} + \beta J_{23}$$
, if $(\vec{A} \cdot \vec{B})^2 + (\vec{A}^2 - \vec{B}^2)^2 \neq 0$,
2° $Q' = \alpha (J_{01} + J_{12})$, if $\vec{A} \cdot \vec{B} = \vec{A}^2 - \vec{B}^2 = 0$.

Proof. Let us introduce a new operator

$$J_a = \frac{i}{2}(M_a + iN_a), \qquad K_a = \frac{i}{2}(M_a - iN_a)$$

Using commutation relations of AP(1,3)

$$[P_{\mu}, P_{\nu}] = 0, \qquad [P_{\mu}, J_{\nu\sigma}] = i(g_{\mu\nu}P_{\sigma} - g_{\mu\sigma}P_{\nu}), [J_{\mu\nu}, J_{\lambda\sigma}] = i(g_{\mu\sigma}J_{\nu\lambda} + g_{\nu\sigma}J_{\mu\sigma} - g_{\mu\lambda}J_{\nu\sigma} - g_{\nu\sigma}J_{\mu\lambda}),$$
(A.2.2)

one can easily check the validity of relations

$$[J_a, J_b] = i\epsilon_{abc}J_c,$$

$$[K_a, K_b] = i\epsilon_{abc}K_c, \qquad [J_a, K_b] = 0.$$
(A.2.3)

Operator Q can be rewritten as $Q = a_k J_k + b_a K_a$, provided

$$a_k = -B_k - iA_k, \qquad b_k = B_k - iA_k.$$

By means of (A.2.1) and (A.2.3) one obtains

$$Q' = V_1 Q V_1' = \sqrt{\vec{a}^2} J_1 + (\sqrt{\vec{a}^2})^* K_1 \equiv \alpha J_{01} + \beta J_{23},$$

where

$$V_{1} = \exp\left\{-i\arctan\frac{a_{2}}{a_{3}}J_{1}\right\}\exp\left\{i\left(\arctan\frac{a_{1}}{\sqrt{a_{2}^{2}+a_{3}^{2}}}+\frac{\pi}{2}\right)J_{2}\right\}$$

$$\cdot\exp\left\{-i\arctan\frac{b_{2}}{b_{3}}K_{1}\right\}\exp\left\{i\left(\arctan\frac{b_{1}}{\sqrt{b_{2}^{2}+b_{3}^{2}}}+\frac{\pi}{2}\right)K_{2}\right\}$$
(A.2.4)

It is evident that these formulae lose their validity in the case

$$\vec{a}^2 \equiv a_1^2 + a_2^2 + a_3^2 = 0 \Leftrightarrow \vec{A}^2 = \vec{B}^2, \qquad \vec{A} \cdot \vec{B} = 0$$

Therefore, one can apply operator (A.2.4) only in the case 1° . Let us now consider the second case 2° . It follows from (A.2.1) that

$$\exp\{\theta M_a\}A_k M_k \exp\{-\theta M_a\} = A_k M_k \cos\theta +$$
(A.2.5)

$$+A_a M_a (1 - \cos \theta) + \epsilon_{ak\ell} A_k M_\ell \sin \theta \qquad \text{(no sum over } a\text{)},$$

$$\exp\{\theta M_a\}B_k N_k \exp\{-\theta M_a\} = B_k N_k \cos\theta +$$
(A.2.6)

$$+B_a N_a (1-\cos\theta) + \epsilon_{ak\ell} B_k N_\ell \sin\theta \qquad \text{(no sum over } a\text{)},$$

Using identities (A.2.5), (A.2.6) one can make sure that the following equality holds

$$Q' = V_2 Q V_2^{-1} = V_2 (\vec{A} \cdot \vec{M} + \vec{B} \cdot \vec{N}) V_2^{-1} = -\sqrt{\vec{A}^2} \operatorname{sign} A_3 (J_{01} + J_{12}),$$

where

$$V_{2} = \exp\left\{\arctan\frac{A_{1}}{A_{2}}M_{3}\right\} \exp\left\{\arctan\frac{\sqrt{A_{1}^{2} + A_{2}^{2}}}{A_{3}}M_{1}\right\} \cdot \left(A.2.7\right)$$
$$\cdot \exp\left\{\left[\arctan\frac{B_{3}\sqrt{\vec{A}^{2}}}{B_{2}A_{1} - A_{2}B_{1}} + \pi\theta(B_{1}A_{2} - B_{2}A_{1})\right]M_{3}\right\}$$
$$\operatorname{sign} x = \left\{\begin{array}{cc}1, & x \ge 0\\ -1, & x < 0\end{array}, \qquad \theta(x) = \left\{\begin{array}{cc}1, & x > 0\\ 0, & x < 0\end{array}\right.\right\}$$

This completes the proof.

Theorem A.2.1. [109,212,11*] *The operator*

$$Q = A_k M_k + B_k N_k + CD + C^{\mu} P_{\mu}$$
(A.2.8)
$$(A_k, B_k, C, C^{\mu} \text{ are constants})$$

can be reduced by means of transformation

$$Q \to Q' = VQV^{-1}, \qquad V = \exp\{\theta^{\mu\nu}J_{\mu\nu} + \theta D + \theta^{\mu}P_{\mu}\}$$
(A.2.9)

to one of the following forms: If $\vec{A} \cdot \vec{B} = 0$, $\vec{A}^2 = \vec{B}^2$, then

1)
$$Q' = J_{01} + J_{12} + aD$$
,
2) $Q' = J_{01} + J_{12} + \beta P_3 - P_0$,
3) $Q' = J_{01} + J_{12} + \beta P_3$,
 $If (\vec{A} \cdot \vec{B})^2 + (\vec{A}^2 - \vec{B}^2)^2 \neq 0$, then

4)
$$Q' = J_{23} + aD,$$

5) $Q' = J_{01} + bJ_{23} + aD,$ (A.2.10)

370

- 6) $Q' = J_{01} + bJ_{23} + D + \beta P_0$,
- 7) $Q' = J_{01} + P_2$,
- 8) $Q' = J_{23} + \alpha_1 P_0 + \alpha_2 P_1,$

If
$$\vec{A} = \vec{B} = 0$$
, then

- 9) Q' = D,
- 10) $Q' = P_0 + P_1$,
- 11) $Q' = P_0$,
- 12) $Q' = P_1$.

Proof. Let us consider at first the case when $\vec{A}^2 + \vec{B}^2 \neq 0$. Then, as follows from Lemma A.2.1, there exists an operator $V_1(V_2)$ of the form (A.2.9) such that

- (a) under $\vec{A} \cdot \vec{B} = \vec{A}^2 \vec{B}^2 = 0$, $V_1 Q V_1^{-1} = \alpha (J_{01} + J_{12}) + \theta D + \theta^{\mu} P_{\mu}$,
- (b) under $(\vec{A} \cdot \vec{B})^2 + (\vec{A}^2 \vec{B}^2)^2 \neq 0$,

$$V_2 Q V_2^{-1} = \alpha J_{01} + \beta J_{23} + \theta D + \theta^{\mu} P_{\mu},$$

Below we shall use the identities

$$\exp\{i\lambda^{\mu}P_{\mu}\}J_{\alpha\beta}\exp\{-i\lambda^{\mu}P_{\mu}\} = J_{\alpha\beta} + \lambda_{\beta}P_{\alpha} - \lambda_{\alpha}P_{\beta}$$
$$\exp\{i\lambda^{\mu}P_{\mu}\}D\exp\{-i\lambda^{\mu}P_{\mu}\} = D - \lambda^{\mu}P_{\mu},$$
(A.2.11)
$$\exp\{i\lambda^{\mu}P_{\mu}\}P_{\alpha}\exp\{-i\lambda^{\mu}P_{\mu}\} = P_{\alpha},$$

which can be easily proved be means of the CBH formula (A.2.1). In the case (a) we have

$$\begin{aligned} Q' \to Q'' &= \exp\{i\lambda^{\mu}P_{\mu}\}(J_{01} + J_{12} + \theta D + \theta^{\alpha}P_{\alpha})\exp\{-i\lambda^{\mu}P_{\mu}\} = \\ &= J_{01} + J_{12} + \theta D + \theta^{\mu}P_{\mu} + \lambda_1P_0 - \lambda_0P_1 - \lambda_2P_1 - \lambda_1P_2 - \theta\lambda^aP_{\alpha}. \end{aligned}$$

Under $\theta \neq 0$ one can always choose λ_{α} so that

$$Q'' = J_{01} + J_{12} + \theta D$$

and under $\theta = 0$ so that

$$Q'' = J_{01} + J_{12} + \alpha P_0 + \beta P_3, \qquad \alpha \le 0$$

If in this latter operator $\alpha \neq 0$, then

$$Q''' = \exp\{-i\ln|\alpha|D\}(J_{01} + J_{12} + \alpha P_0 + \beta P_3)\exp\{i\ln|\alpha|D\} =$$
$$= J_{01} + J_{12} - \operatorname{sign} \alpha P_0 + \frac{\beta}{|\alpha|}P_3$$

If $\alpha = 0$, then

$$Q'' = J_{01} + J_{12} + \beta P_3.$$

Let us now consider the case (b). If $\alpha \neq 0$, then on dividing into α and on transforming the operator Q according to (A.2.11) we obtain

$$Q' = \exp\{i\lambda^{\mu}P_{\mu}\}(J_{01} + bJ_{23} + \theta D + \theta^{\mu}P_{\mu})\exp\{-i\lambda^{\mu}P_{\mu}\} = = J_{01} + \lambda_{1}P_{0} - \lambda_{0}P_{1} + bJ_{23} + b(\lambda_{3}P_{2} - \lambda_{2}P_{3}) + \theta D - \theta\lambda^{\mu}P_{\mu} + \theta^{\mu}P_{\mu}.$$

Under $\theta \neq \pm 1$, $\theta^2 + b^2 \neq 0$ it is always possible to choose λ_{μ} so that

$$Q' = J_{01} + bJ_{23} + \theta D.$$

Under $\theta = \pm 1$, one can choose λ_{μ} so that

$$Q' = J_{01} + bJ_{23} \pm D + \beta P_0.$$

Under $\theta = b = 0$ there exist such λ_{μ} that

$$Q' = J_{01} + P_2.$$

Under $\alpha = 0$, using identities (A.2.11), one can reduce the operator Q' to one of the following forms:

$$\begin{aligned} Q'' &= J_{23} + \theta D, \qquad \theta \neq 0 \\ Q'' &= J_{23} + \alpha_1 P_0 + \alpha_2 P_1, \qquad \theta = 0. \end{aligned}$$

Remaining to be considered is the case $\vec{A} = \vec{B} = 0$, i.e., $Q = \theta D + \theta^{\mu} P_{\mu}$. Using (A.2.11) one easily obtains under $\theta \neq 0$:

$$\exp\left\{\frac{i}{\theta}\theta^{\mu}P_{\mu}\right\}Q\exp\left\{\frac{-i}{\theta}\theta^{\mu}P_{\mu}\right\}=\theta D.$$

If $\theta = 0$, then analyzing three possibilities $\theta_{\mu}\theta^{\mu} = 0$, $\theta_{\mu}\theta^{\mu} > 0$, $\theta_{\mu}\theta^{\mu} < 0$ we obtain operators 10)-12) from (A.2.10). Thus the theorem is proved.

Note A.2.1. When proving the theorem we used only commutation relations of $\widetilde{AP}(1,3)$ given by (A.2.2) and

$$[P_{\mu}, D] = iP_{\mu}, \qquad [J_{\mu\nu}, D] = 0 \tag{A.2.12}$$

and we did not use any concrete representation.

Some applications of Campbell-Baker-Hausdorff operator calculus

In the present Appendix we consider some applications of operator calculus expounded in Paragraph 5.3.

1. Let us calculate finite transformations of spacetime variables for the Lorentz group. We take generators of the Lorentz algebra AO(1,3) in the form

$$J_a = (\vec{x} \times \vec{\nabla})_a, \qquad J_{0a} = x_0 \partial_a + x_a \partial_0, \qquad a = 1, 2, 3$$
 (A.3.1)

and consider two operators

$$M = \alpha_a J_a = \vec{\alpha} \cdot (\vec{x} \times \nabla),$$

$$N = \beta_a J_{0a} = x_0 \vec{\beta} \cdot \vec{\nabla} + \vec{\beta} \cdot \vec{x} \partial_0,$$

where α_a , β_a are arbitrary real constants (group parameters). Using formulae

(5.3.6), (5.3.8), (5.3.11) we find $\begin{aligned} x'_0 &= \exp\{M\}x_0 \exp\{-M\} = x_0, \\ x'_a &= \exp\{M\}x_a \exp\{-M\} = x_a + \\ &+ \left(\sum_{i=1}^{\infty} \frac{(-1)^n}{(2\pi + 1)!} \alpha^{2n+1}\right) \frac{(\vec{x} \times \vec{\alpha})_a}{(\vec{x} - \vec{\alpha})_a} + \left(\sum_{i=1}^{\infty} \frac{(-1)^n}{(2\pi + 1)!} \alpha^{2n}\right) \frac{x_a \alpha^2 - \alpha_a \vec{\alpha} \cdot \vec{x}}{\alpha^2} = \end{aligned}$

$$\left(\frac{1}{n=0}\left(2n+1\right)!\right) \alpha \left(\frac{1}{n=1}\left(2n\right)!\right) \alpha^{\alpha}$$
$$= x_{a}\cos\alpha + \frac{(\vec{x}\times\vec{\alpha})_{a}}{\alpha}\sin\alpha + \frac{\alpha_{a}(\vec{\alpha}\cdot\vec{x})}{\alpha^{2}}(1-\cos\alpha),$$
(A.3.2)

$$\begin{aligned} x'_{0} &= \exp\{N\}x_{0}\exp\{-N\} = x_{0}\sum_{n=0}^{\infty}\frac{\beta^{2n}}{(2n)!} + \frac{\vec{\beta}\cdot\vec{x}}{\beta}\sum_{n=0}^{\infty}\frac{\beta^{2n+1}}{(2n+1)!} = \\ &= x_{0}\operatorname{ch}\beta + \frac{\vec{\beta}\cdot\vec{x}}{\beta}\operatorname{sh}\beta, \\ x'_{a} &= \exp\{N\}x_{a}\exp\{-N\} = x_{a} + \frac{\beta_{a}\vec{\beta}\cdot\vec{x}}{\beta^{2}}(\operatorname{ch}\beta - 1) + \beta_{a}x_{0}\frac{\operatorname{sh}\beta}{\beta}, \end{aligned}$$
(A.3.3)

where $\alpha = (\alpha_1^2 + \alpha_2^2 + \alpha_3^2)^{1/2}, \ \beta = (\beta_1^2 + \beta_2^2 + \beta_3^2)^{1/2}.$

2. Here we derive useful formulae for evaluating commutator of operatorvalued function with an operator.

Theorem A.3.1. Let \hat{x} , \hat{w} be some non-commuting operators belonging to the algebra \mathcal{A} (definition of \mathcal{A} see in Paragraph 5.3). Let $f(\hat{x})$ be an operator-valued differentiable function which can be represented as a power series

$$f(\hat{z}) = \sum_{n=0}^{\infty} a_n \hat{x}^n \tag{A.3.4}$$

 $(a_n \text{ are numerical coefficients})$. Then in \mathcal{A} the following identities hold

$$[\widehat{w}, f(\widehat{x})] = \sum_{n=1}^{\infty} \frac{1}{n!} \left(\partial_{\widehat{x}}^n f(\widehat{x}) \right) \{ \widehat{w}, \widehat{x}^n \}, \tag{A.3.5}$$

$$[f(\widehat{x}),\widehat{w}] = \sum_{n=1}^{\infty} \frac{1}{n!} \{\widehat{x}^n, \widehat{w}\} \left(\partial_{\widehat{x}}^n f(\widehat{x})\right), \tag{A.3.6}$$

Proof. At first we evaluate the commutator $[\widehat{w}, \widehat{x}^n]$. Using CBH formula (5.3.8) we find

$$e^{-\theta \widehat{x}} \widehat{w} e^{\theta \widehat{x}} - \widehat{w} \equiv e^{-\theta \widehat{x}} [\widehat{w}, e^{\theta \widehat{x}}] = \sum_{n=1}^{\infty} \frac{\theta^n}{n!} \{\widehat{w}, \widehat{x}^n\}.$$

Therefore

$$\begin{split} [\widehat{w}, e^{\widehat{\theta x}}] &= e^{\widehat{\theta x}} \sum_{n=1}^{\infty} \frac{\theta^n}{n!} \{\widehat{w}, \widehat{x}^n\}.\\ \sum_{n=0}^{\infty} \frac{\theta^n}{n!} [\widehat{w}, \widehat{x}^n] &= \left(\sum_{k=0}^{\infty} \frac{\theta^k}{k!} \widehat{x}^k\right) \left(\sum_{l=1}^{\infty} \frac{\theta^l}{l!} \{\widehat{w}, \widehat{x}^l\}\right) =\\ &= \sum_{k=0}^{\infty} \sum_{l=1}^{\infty} \frac{\theta^{k+l}}{k! \, l!} \widehat{x}^k \{\widehat{w}, \widehat{x}^l\}. \end{split}$$

After equating in this identity terms with equal degree of θ we get

$$\frac{1}{n!} [\widehat{w}, \widehat{x}^n] = \sum_{l=1}^n \frac{1}{l!(n-l)!} \widehat{x}^{n-l} \{\widehat{w}, \widehat{x}^l\} \equiv \sum_{l=1}^n \frac{1}{n!} C_n^l \widehat{x}^{n-l} \{\widehat{w}, \widehat{x}^l\},$$
$$C_n^l \equiv \frac{n!}{l!(n-l)!}.$$

Hence

$$[\widehat{w}, \widehat{x}^{n}] = \sum_{l=1}^{n} C_{n}^{l} \widehat{x}^{n-l} \{ \widehat{w}, \widehat{x}^{l} \}$$
(A.3.7)

From (A.3.4) we find

$$\partial_{\widehat{x}}^{l}f(\widehat{x}) = \sum_{n=0}^{\infty} a_n \frac{n!}{(n-l)!} \widehat{x}^{n-l}.$$
(A.3.8)

Using formulae (A.3.4), (A.3.7), (A.3.8) we obtain (A.3.5):

$$\begin{split} [\widehat{w}, f(\widehat{x})] &= \sum_{n=0}^{\infty} a_n [\widehat{w}, \widehat{x}^n] = \\ &= \sum_{n=0}^{\infty} a_n \sum_{l=1}^n C_n^l \widehat{x}^{n-l} \{\widehat{w}, \widehat{x}^l\} = \\ &= \sum_{l=1}^{\infty} \sum_{n=l}^{\infty} a_n C_n^l \widehat{x}^{n-l} \{\widehat{w}, \widehat{x}^l\} = \\ &= \sum_{l=1}^{\infty} \frac{1}{l!} \left(\partial_{\widehat{x}}^l f(\widehat{x}) \right) \{\widehat{w}, \widehat{x}^l\}. \end{split}$$

In the same way one can derive formula (A.3.6). To do it one should start from the indentity

$$[\hat{x}^{n}, \hat{w}] = \sum_{k=1}^{n} C_{n}^{k} \{\hat{x}^{k}, \hat{w}\} \hat{x}^{n-k}, \qquad (A.3.7')$$

which follows from (5.3.8) analogously to (A.3.7). For examples of applications of these formulae see Sec. 1.3.

3. In this point we prove the basic Lie's theorems for local groups by means of the CBH operator calculus. For the first time such a proof had been done in 1906 by Hausdorff [117]. Below we follow [38,189].

As it was shown in Paragraph 5.3, final transformations generated by infinitesimal operator of Lie type

$$X = \xi^{\mu}(x,\psi)\frac{\partial}{\partial x_{\mu}} + \eta^{k}(x,\psi)\frac{\partial}{\partial \psi^{k}}, \qquad \mu = \overline{0,n-1}, \ k = \overline{1,m}$$
(A.3.9)

can be obtained be means of the formula

$$x_A \to x'_A = \exp\{\theta X\} x_A \exp\{-\theta X\}, \qquad x_A = \{x_\mu, \psi^k\},$$

(A.3.10)
 $A = 0, 1, \dots, n + m - 1.$

Here it is convenient not to distinguish independent and dependent variables, but further we will write instead of index A the index μ bearing in mind that μ can always be extended to A = n + m - 1.

Theorem A.3.2. Let operators (A.3.9) form an r-dimensional Lie algebra AL(r), in which the following relations hold

$$[X_i, X_j] = c_{ij}^k X_k, \qquad i, j, k = \overline{1, r}$$
(A.3.11)

$$c_{ij}^{k} = -c_{ji}^{k}, \qquad c_{ij}^{s}c_{ks}^{t} + c_{jk}^{s}c_{is}^{t} + c_{ki}^{s}c_{js}^{t} = 0 \qquad (Jacobi \ identity)$$
(A.3.12)

 $(c_{ij}^k$ are certain constants, called the structure constants). Then there exists an r-dimensional group of point transformations G(r) which corresponding to this AL(r).

Proof. Consider two infinitesimal operators

$$\widehat{x} = a^i X_i, \qquad \widehat{y} = b^j X_j \tag{A.3.13}$$

where a^i , b^j are arbitrary constants, $i, j = \overline{1, r}$. According to (A.3.10) we have

$$x'_{\mu}=e^{\widehat{y}}x_{\mu}e^{-\widehat{y}},\qquad x''_{\mu}=e^{\widehat{x}}e^{\widehat{y}}x_{\mu}e^{-\widehat{y}}e^{-\widehat{x}},$$

which can be rewritten be means of (5.3.11) as

$$x''_{\mu} = e^{\hat{x}} e^{\hat{y}} x_{\mu} e^{-\hat{y}} e^{-\hat{x}} = e^{\hat{z}} x_{\mu} e^{-\hat{z}}.$$
 (A.3.14)
The theorem will be proved if we convince ourselves that operator \hat{z} from (A.3.14) belong to the AL(r), that is it can be represented as

$$\widehat{z} = c^i X_i, \qquad c^i = \varphi^i(a, b), \tag{A.3.15}$$

where c^i are parameters depending on a and b. Note: formula (A.3.10) as well as (A.3.14) with \hat{z} given by (A.3.15) lead to local (point) transformations insofar as the expression

$$f(x') = e^{\theta \xi \partial} f(x) e^{-\theta \xi \partial} = \sum_{n=0}^{\infty} \frac{\theta^n}{n!} (\xi \partial)^n f(x)$$

is just a Maclaurin's series of an expansion of function f(x').

To find \hat{z} from (A.3.14) let us use formulae (5.3.12). Taking into account (A.3.11) we successively obtain

where $\beta = (\beta_i^k \equiv b^j c_{ij}^k)$. So we can write the first term \hat{z}_1 of the series (5.3.12):

$$\widehat{z}_1 = \sum_{n=0}^{\infty} \frac{(-1)^n}{n!} B_n\{\widehat{x}, \widehat{y}^n\} = \left(\sum_{n=0}^{\infty} \frac{(-1)^n}{n!} B_n \beta^n\right)_i^s a^i X_s \stackrel{\text{def}}{=} f^s X_s.$$
(A.3.17)
where

wnere

$$f^{s} = a^{i} \mathcal{B}_{i}^{s}, \qquad \mathcal{B} = \sum_{n=0}^{\infty} \frac{(-1)^{n}}{n!} B_{n} \beta^{n} = \frac{\beta e^{\beta}}{e^{\beta} - 1}.$$
 (A.3.18)

As it is known from linear algebra a power series $f(\beta)$ converges if and only if all characteristic roots of the matrix β lie inside the ring of convergence of the series f(z). As was already said (see formula (5.3.51)) the seriess

$$f(z) = \sum_{n=0}^{\infty} \frac{(-1)^n}{n!} B_n z^n$$

converges if |z| is less than 2π . Speaking more precisely, this series converges everywhere except points where $z = 2\pi i k$. Therefore element \hat{z}_1 is well-defined everywhere except the points where $det|\beta - 2\pi ik|=0, k=1,2,...$

To find the rest terms of the series (5.3.12) we shall use the following statement.

Lemma A.3.1. Let

 $\widehat{f} = f(\widehat{y}) = f^s(b)X_s, \qquad \widehat{u} = d^j X_j, \qquad d_j \text{ are constants}$

Then the equality holds

$$\left(\widehat{u}\,\partial_{\widehat{y}}\right)\widehat{f} = d^j \frac{\partial f^s(b)}{\partial b^j} X_s. \tag{A.3.19}$$

Proof. According to the Taylor series expansion of operator-valued function $f(\hat{y} + \theta \hat{u})$ (see Lemma 5.3.1) we have

$$f(\widehat{y} + \theta \widehat{u}) = f(\widehat{y}) + \theta \left(\widehat{u}\partial_{\widehat{y}}\right) f(\widehat{y}) + \cdots$$

On the other hand we can write

$$f(\hat{y} + \theta \hat{u}) = f^s(b + \theta d) X_s = \left[f^s(b) + \theta d^j \frac{\partial f^s}{\partial b^j} + \cdots \right] X_s$$

After equating these two expressions we obtain the identity (A.3.19). Thus the lemma is proved.

By means of (A.3.19) we find $\hat{z}_2, \hat{z}_3, \ldots$ from the expansion (5.3.12):

$$\widehat{z}_{2} = \frac{1}{2} \left(\widehat{z}_{1} \partial_{\widehat{y}} \right) \widehat{z}_{1} = \frac{1}{2} f^{j} \frac{\partial f^{s}}{\partial b^{j}} X_{s} = -\frac{1}{2} [f^{s}, \widehat{q}],$$

$$\widehat{z}_{3} = \frac{1}{3} \left(\widehat{z}_{1} \partial_{\widehat{y}} \right) \widehat{z}_{2} = \frac{1}{3!} \{ f^{s}, \widehat{q}^{2} \}, \dots,$$
(A.3.20)

where $\hat{q} = f^j \frac{\partial}{\partial b^j}$. Hence

$$\widehat{z} = \widehat{y} + \left(e^{\widehat{q}} f^s e^{-\widehat{q}}\right) X_s \tag{A.3.21}$$

and thereby we get

$$c^{s} = b^{s} + e^{\widehat{q}} f^{s} e^{-\widehat{q}} = b^{s} + \sum_{n=0}^{\infty} \frac{(-1)^{n}}{n!} \{ f^{s}, (f^{k} \partial_{b^{k}})^{n} \},$$
(A.3.22)

where f^s and \hat{q} are defined in (A.3.18) and (A.3.20), respectively. In particular, in a first approximation we have

$$\mathcal{B} = 1 + \frac{1}{2}\beta + \cdots, \qquad \mathcal{B}_i^s = \delta_i^s + \frac{1}{2}b^i c_{ij}^k + \cdots$$
$$f^s = a^s + \frac{1}{2}c_{ij}^s a^i b^j$$

and therefore

$$c^{s} = a^{s} + b^{s} + \frac{1}{2}c^{s}_{ij}a^{i}b^{j} + \cdots$$
 (A.3.23)

Analogously, in a second approximation (A.3.22) results in [189]:

$$c^{s} = a^{s} + b^{s} + \frac{1}{2}c^{s}_{ij}a^{i}b^{j} + \frac{1}{12}c^{k}_{ij}c^{s}_{kl}a^{i}b^{j}(b^{l} - a^{l}) + \cdots$$
(A.3.24)

This completes the proof of Theorem A.3.2.

Now we shall obtain the so-called Maurer-Cartan equations. Consider an operator-valued function

$$\exp\left\{\widehat{y} + \epsilon\widehat{u} + \delta\widehat{v}\right\} \tag{A.3.25}$$

where $\hat{y} = b^j X_j$, $\hat{u} = d^j X_j$, $\hat{v} = h^j X_j$, $j = \overline{1, r}$; and b^j , d^j , h^j , ϵ , δ are arbitrary constants. Expanding it in the Taylor series according to (5.3.15) and using the formula (5.3.18) one successively obtains

$$\exp\{\widehat{y} + \epsilon \widehat{u} + \delta \widehat{v}\} = \exp\{\widehat{y} + \epsilon \widehat{u}\} + \delta\left(\widehat{v}\partial_{\widehat{y}}\right)\exp\{\widehat{y} + \epsilon \widehat{u}\} + \dots =$$

$$= \exp\{\widehat{y} + \epsilon \widehat{u}\} + \delta\psi\left(\widehat{v}, \widehat{y} + \epsilon \widehat{u}\right)\exp\{\widehat{y} + \epsilon \widehat{u}\} + \dots =$$

$$= \left[1 + \delta\psi(\widehat{v}, \widehat{y}) + \epsilon\delta\left(\widehat{u}\partial_{\widehat{y}}\right)\psi(\widehat{v}, \widehat{y}) + \dots\right]\left(\exp\{\widehat{y}\} + \epsilon\psi(\widehat{u}, \widehat{y})\exp\{\widehat{y}\} + \dots\right) =$$

$$= \left[1 + \epsilon\psi(\widehat{u}, \widehat{y}) + \delta\psi\left(\widehat{v}, \widehat{y}\right) + \epsilon\delta(\widehat{u}, \partial_{\widehat{y}})\psi(\widehat{v}, \widehat{y}) + \epsilon\delta\psi(\widehat{v}, \widehat{y})\psi(\widehat{u}, \widehat{y}) + \dots\right]\exp\{\widehat{y}\}.$$

Let us write down a coefficient under $\epsilon\delta$. It is

$$\left[(\widehat{u} \,\partial_{\widehat{y}}) \psi(\widehat{v}, \widehat{y}) + \psi(\widehat{v}, \widehat{y}) \psi(\widehat{u}, \widehat{y}) \right] \exp\{\widehat{y}\}.$$

Interchanging \hat{u} and \hat{v} and equating the two expressions, we get the indentity

$$\begin{split} \left(\widehat{u}\,\partial_{\widehat{y}}\right)\psi(\widehat{v},\widehat{y}) - \left(\widehat{v}\,\partial_{\widehat{y}}\right)\psi(\widehat{u},\widehat{y})\psi\left(\widehat{u},\widehat{y}\right) = \\ &= \psi(\widehat{u},\widehat{y})\psi(\widehat{v},\widehat{y}) - \psi(\widehat{v},\widehat{y})\psi(\widehat{u},\widehat{y}). \end{split}$$

But according to (5.3.20) we have

$$\psi(\hat{u},\hat{y}) = \sum_{n=0}^{\infty} \frac{(-1)^n}{(n+1)!} \{\hat{u},\hat{y}^n\} = \left(\sum_{n=0}^{\infty} \frac{(-1)^n}{(n+1)!} \beta^n\right)_i^s d^i X_s =$$
(A.3.27)
$$= \psi_i^s(\beta) d^i X_s,$$

where matrix $\psi(\beta)$ (see also (A.3.18)) is

$$\psi(\beta) = \sum_{n=0}^{\infty} \frac{(-1)^n}{(n+1)!} \beta^n = \mathcal{B}^{-1}$$
(A.3.28)

Analogous expression one obtains for $\psi(\hat{v}, \hat{y})$:

$$\psi(\hat{v},\hat{y}) = h^i \psi_i^s(\beta) X_s. \tag{A.3.29}$$

Note that the series (A.3.28) is convergent for any matrix β . Using (A.3.19) we get

$$\begin{pmatrix} \widehat{u} \,\partial_{\widehat{y}} \end{pmatrix} \psi(\widehat{v}, \widehat{y}) = d^i h^j \frac{\partial \psi_j^{\gamma}(b)}{\partial b^i} X_k,$$

$$\begin{pmatrix} \widehat{v} \,\partial_{\widehat{y}} \end{pmatrix} \psi(\widehat{u}, \widehat{y}) = d^i h^j \frac{\partial \psi_i^k(b)}{\partial b^j} X_k.$$
(A.3.30)

Substituting (A.3.27)-(A.3.30) into (A.3.26) and equating coefficients under equal products of $d^i h^j$, we obtain equations

$$\frac{\partial \psi_i^k}{\partial b^j} - \frac{\partial \psi_j^k}{\partial b^i} = c_{sl}^k \psi_i^s \psi_j^l, \qquad (A.3.31)$$

which are known as the Maurer-Cartan equations.

Consider operators

$$\widehat{q}_i = \mathcal{B}_i^s \frac{\partial}{\partial b^s} \tag{A.3.32}$$

acting in dual space of parameters. The third Lie's theorem will be proved if we show that operators (A.3.32) are linearly independent and form a Lie algebra with structure constants c_{ij}^k (A.3.12).

Linear independence of operators (A.3.32) follows from the fact that $det|\mathcal{B}| \neq 0$ insofar as the inverse matrix $\mathcal{B}^{-1} = \psi(\beta)$ exists (series (A.3.28) is convergent for any β). Let us compute the commutator

$$[\widehat{q}_i, \widehat{q}_j] = \left(\mathcal{B}_i^s \frac{\partial \mathcal{B}_j^t}{\partial b^s} - \mathcal{B}_j^s \frac{\partial \mathcal{B}_i^t}{\partial b^s}\right) \frac{\partial}{\partial b^t} \stackrel{\text{def}}{=} R_{ij}^t \frac{\partial}{\partial b^t}.$$
 (A.3.33)

Differentiating the identity

$$\psi_t^k \mathcal{B}_i^t = \delta_i^k \tag{A.3.34}$$

with respect to b^s we get

$$\psi_t^k \frac{\partial \mathcal{B}_i^t}{\partial b^s} + \mathcal{B}_i^t \frac{\partial \psi_t^k}{\partial b^s} = 0.$$
(A.3.35)

Multiplying (A.3.33) by ψ_i^k and using (A.3.35) we find

$$\mathcal{B}_{i}^{s}\mathcal{B}_{j}^{t}\left(\frac{\partial\psi_{s}^{k}}{\partial b^{t}}-\frac{\partial\psi_{t}^{k}}{\partial b^{s}}\right)=\psi_{t}^{k}R_{ij}^{t}.$$
(A.3.36)

By means of the Maurer-Cartan equations (A.3.31) we can rewrite (A.3.36) as follows

$$\mathcal{B}_i^s \mathcal{B}_J^t c_{lm}^k \psi_s^l \psi_t^m = c_{lm}^k \mathcal{B}_i^s \psi_s^l \mathcal{B}_j^t \psi_t^m = c_{ij}^k = \psi_t^k R_{ij}^t$$

As a result we have

$$R_{ij}^s = \mathcal{B}_k^s c_{ij}^k \tag{A.3.37}$$

Substituting (A.3.37) into (A.3.33) we get

$$[\hat{q}_i, \hat{q}_j] = c_{ij}^k \mathcal{B}_k^s \frac{\partial}{\partial b^s} \equiv c_{ij}^k \hat{q}_k.$$
(A.3.38)

Thus we have proved the third Lie's theorem.

4. An important extension of the Hausdorff formula results when one considers the solutions of the ODE for operator-valued function $\hat{y}(t)$ depending on real variable t,

$$\dot{\hat{y}} \equiv \frac{d\hat{y}}{dt} = \hat{q}(t)\hat{y}, \qquad \hat{y}(0) = I,$$
(A.3.39)

where $\hat{q}(t)$ is a given operator. Following [146] we look for a solution of (A.3.39) in the form

$$\widehat{y}(t) = \exp\{\widehat{x}(t)\} \tag{A.3.40}$$

and the problem then becomes one of finding an expression for $\hat{x}(t)$. It follows from (5.3.18), (5.3.20) that

$$\begin{split} \dot{\hat{y}} &= \left(\dot{\hat{x}}\,\partial_{\hat{x}}\right)e^{\hat{x}} = \psi(\dot{\hat{x}},\hat{x})e^{\hat{x}} = \hat{q}e^{\hat{x}},\\ \hat{q} &= \psi\left(\dot{\hat{x}},\hat{x}\right) = \sum_{n=0}^{\infty} \frac{(-1)^n}{(n+1)!} \{\dot{\hat{x}},\hat{x}\}, \end{split}$$
(A.3.41)

Using Lemma 5.3.3 we can invert the series for \hat{q} (A.3.41) according to formula (5.3.28):

$$\dot{\widehat{x}} = \sum_{n=0}^{\infty} \frac{(-1)^n}{n!} B_n\{\widehat{q}, \widehat{x}^n\},$$

where B_n are the Bernouilli numbers (5.3.13). Finally, this equation is solved by iteration, by setting

$$\begin{aligned} \widehat{x}_{0} &= 0, \\ \widehat{x}_{1}(t) &= \int_{0}^{t} \widehat{q}(\tau) d\tau, \\ \widehat{x}_{2}(t) &= \int_{0}^{t} \left(\sum_{n=0}^{\infty} \frac{(-1)^{n}}{n!} B_{n} \left\{ \widehat{q}(\tau), \left(\int_{0}^{\tau} \widehat{q}(\tau_{1}) d\tau_{1} \right)^{n} \right\} \right) d\tau, \\ \dots \\ \widehat{x}_{k}(t) &= \int_{0}^{t} \left(\sum_{n=0}^{\infty} \frac{(-1)^{n}}{n!} B_{n} \left\{ \widehat{q}(\tau), x_{k-1}^{n}(\tau) \right\} \right) d\tau \end{aligned}$$
(A.3.42)

and putting

$$\widehat{x}(t) = \lim_{k \to \infty} \widehat{x}_k(t). \tag{A.3.43}$$

This result was first obtained by Magnus [146], and formulae (A.3.42), (A.3.43) are the continuous analog to the Hausdorff fromula.

In particular, if

$$\left[\widehat{q}(\tau), \int_{0}^{\tau} \widehat{q}(\tau_{1}) d\tau_{1}\right] = 0, \qquad (A.3.44)$$

then solution of Equation (A.3.39) has the form

$$\widehat{y}(t) = \exp\left\{\int_{0}^{t} \widehat{q}(\tau)d\tau\right\}.$$
(A.3.45)

Consider another example of Equation (A.3.39), namely when operator $\hat{q}(t)$ has the form

$$\widehat{q}(t) = \sum_{j=1}^{n} a_j(t) Q_j, \qquad n < \infty$$
(A.3.46)

where $a_j(t)$ are scalar differentiable functions of t, and the Q_j are timeindependent operators forming an *n*-dimensional Lie algebra AL(n), *n* is finite. The following statement holds.

Theorem A.3.3 [26*,27*]. The solution of Equation (A.3.46) with the operator $\hat{q}(t)$ given by (A.3.46) may be expressed in the form (at least in a neighborhood of t = 0)

$$\hat{y}(t) = \prod_{i=1}^{n} \exp\{g_i(t)Q_i\}$$
(A.3.47)

where $g_i(t)$ are scalar functions which are solutions of the following system of nonlinear ODEs:

$$\begin{pmatrix} a_1 \\ a_2 \\ \vdots \\ a_n \end{pmatrix} = \begin{pmatrix} \xi_{11} & \xi_{12} & \cdots & \xi_{1n} \\ \xi_{21} & \ddots & \vdots \\ \vdots & & \ddots & \vdots \\ \xi_{n1} & \cdots & \cdots & \xi_{nn} \end{pmatrix} \begin{pmatrix} \dot{g}_1 \\ \dot{g}_2 \\ \vdots \\ \dot{g}_n \end{pmatrix}, \quad g_i(0) = 0,$$
 (A.3.48)

 ξ_{ij} are certain analytic functions of g. If the operators Q_i form a solvable Lie algebra, then Equations (A.3.48) take the form

$$\begin{pmatrix} a_1 \\ a_2 \\ \vdots \\ a_n \end{pmatrix} = \begin{pmatrix} \xi_{11} & 0 & 0 & \dots & 0 \\ \xi_{21} & \xi_{22} & 0 & \dots & 0 \\ \vdots & \vdots & & \ddots & \vdots \\ \xi_{n1} & \dots & \dots & \xi_{nn} \end{pmatrix} \begin{pmatrix} \dot{g}_1 \\ \dot{g}_2 \\ \vdots \\ \dot{g}_n \end{pmatrix}, \quad g_i(0) = 0, \qquad (A.3.49)$$

and in this case solution (A.3.47) exists globally.

The proof of Theorem A.3.3 based on the CBH formula (5.3.8) and on the properties of Lie algebras, in particular of solvable Lie algebras. Let us remind some definitions.

A Lie algebra is said to be solvable if its derived series (also called a commutant series)

$$L \supset L^{(1)} \supset L^{(2)} \supset \ldots \supset L^{(k)} \supset \ldots$$
(A.3.50)

ends by zero ideal $L^{(k)}$, provided k is finite. A subalgebra S is called an *ideal* if the commutator $[\hat{x}, \hat{y}]$, of $\hat{x} \in S$ and $\hat{y} \in L$, is in S.

The set of those elements of algebra AL which are the result of commututation of elements \hat{u} and \hat{v} , where $\hat{u} \in N$, $\hat{v} \in M$ (N, M are some subspaces of AL) forms a Lie product and is denoted by [N, M].

The ideal $L^{(1)} = [L, L]$ is called a derived algebra or commutant. By analogy with $L^{(1)}$ we define $L^{(2)} = [L^{(1)}, L^{(1)}], L^{(3)} = [L^{(2)}, L^{(2)}]$, and so on.

As an example to what has been said above we consider, following [28*], the time-dependent Fokker-Planck equation

$$\frac{\partial u}{\partial t} = \left\{ -\frac{\partial}{\partial x} \left(a(t)x + b(t) \right) + c(t) \frac{\partial^2}{\partial x^2} \right\} u, \tag{A.3.51}$$

where u = u(x, t) is a scalar function. We look for solution of this equation in the form

$$u(x,t) = \hat{y}(t)u(x,0), \qquad (A.3.52)$$

where operator $\hat{y}(t)$ satisfies Equation (A.3.39), provided

$$\widehat{q}(t) = -a(t)\frac{\partial}{\partial x}x - b(t)\frac{\partial}{\partial x} + c(t)\frac{\partial^2}{\partial x^2}.$$
(A.3.53)

Comparing (A.3.53) with (A.3.46) we have

$$Q_1 = \frac{\partial}{\partial x} x \equiv x \frac{\partial}{\partial x} + 1, \qquad Q_2 = \frac{\partial}{\partial x}, \qquad Q_3 = \frac{\partial^2}{\partial x^2}.$$
 (A.3.54)

Operators (A.43.54) form a Lie algebra with commutation relations

$$[Q_1, Q_2] = -Q_2,$$
 $[Q_2, Q_3] = 0,$ $[Q_1, Q_3] = -2Q_3,$

whence follows that it is a solvable algebra insofar as

$$L^{(1)} = \langle Q_2, Q_3 \rangle, \qquad L^{(2)} = \langle 0 \rangle$$

and commutant series (A.3.50) ends under k = 2.

Searching for operator-valued function $\hat{y}(t)$ in the form (A.3.47)

$$\widehat{y}(t) = \exp\left\{\alpha(t)\frac{\partial}{\partial x}x\right\} \exp\left\{\beta(t)\frac{\partial}{\partial x}\right\} \exp\left\{\gamma(t)\frac{\partial^2}{\partial x^2}\right\}$$
(A.3.55)

we find for the unknown functions $\alpha(t), \beta(t), \gamma(t)$ the linear system of ODEs:

$$\begin{pmatrix} -a(t) \\ -b(t) \\ c(t) \end{pmatrix} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & e^{-\alpha(t)} & 0 \\ 0 & 0 & e^{-2\alpha(t)} \end{pmatrix} \begin{pmatrix} \dot{\alpha}(t) \\ \dot{\beta}(t) \\ \dot{\gamma}(t) \end{pmatrix},$$
(A.3.56)

with $\alpha(0) = \beta(0) = \gamma(0) = 0$; whence follows

$$\begin{aligned} \alpha(t) &= -\int_{0}^{t} a(\tau) d\tau, \\ \beta(t) &= -\int_{0}^{t} b(\tau) e^{\alpha(\tau)} d\tau, \\ \gamma(t) &= -\int_{0}^{t} c(\tau) e^{2\alpha(\tau)} d\tau. \end{aligned}$$
(A.3.57)

Using the easily derived identities

$$\exp\left\{f(t)\frac{\partial}{\partial x}\right\}u(x) = u\left(x + f(t)\right),$$

$$\exp\left\{f(t)\frac{\partial}{\partial x}x\right\}u(x) = e^{f(t)}u\left(e^{f(t)}x\right),$$

$$\exp\left\{f(t)\frac{\partial^2}{\partial x^2}\right\}u(x) = \frac{1}{\sqrt{4\pi f(t)}}\int_{-\infty}^{\infty}\exp\left\{-\frac{(x-y)^2}{4f(t)}\right\}u(y)dy$$
(A.3.58)

we successively obtain be means of (A.3.52), (A.3.55), (A.3.57) the solution of the Fokker-Planck equation (A.3.51):

$$\begin{split} u(x,t) &= \exp\left\{\alpha(t)\frac{\partial}{\partial x}x\right\} \exp\left\{\beta(t)\frac{\partial}{\partial x}\right\} \exp\left\{\gamma(t)\frac{\partial^2}{\partial x^2}\right\} u(x,0) = \\ &= \exp\left\{\alpha(t)\frac{\partial}{\partial x}x\right\} \exp\left\{\beta(t)\frac{\partial}{\partial x}\right\} \frac{1}{\sqrt{4\pi\gamma(t)}} \int_{-\infty}^{\infty} dy \, u(y,0) \exp\left\{-\frac{(y-x)^2}{4\gamma(t)}\right\} = \\ &= \exp\left\{\alpha(t)\frac{\partial}{\partial x}x\right\} \frac{1}{\sqrt{4\pi\gamma(t)}} \int_{-\infty}^{\infty} dy \, u(y,0) \exp\left\{-\frac{[y-(x+\beta(t)]^2}{4\gamma(t))}\right\} = \end{split}$$

384

$$=\frac{e^{\alpha(t)}}{\sqrt{4\pi\gamma(t)}}\int_{-\infty}^{\infty}\exp\left\{-\frac{[y-(xe^{\alpha(t)}+\beta(t))]^2}{4\gamma(t)}\right\}u(y,0)dy,\tag{A.3.59}$$

where functions $\alpha(t)$, $\beta(t)$, $\gamma(t)$ are given in (A.3.57).

It will be noted that the Fokker-Planck equation (A.3.51) is reduced by means of the change of variables (see §3.8)

$$u(x,t) = e^{\alpha(t)}w(y,\tau),$$

$$y = e^{\alpha(t)}x + \beta(t),$$

$$\tau = -\gamma(t)$$

(A.3.60)

to the standard heat equation

$$w_{\tau} = w_{yy}.\tag{A.3.61}$$

5. The CBH formula (5.3.8) can be used for finding final transformations generated by operators forming a superalgebra. Without going into details let us present an example of such calculations.

Consider, following [199], an operator

$$X = a_j X_j, \qquad j = 1, 2, 3, \tag{A.3.62}$$

where X_j are basis elements of the superalgebra SQM(1,2), satisfying relations

$$[X_1, X_2] = [X_1, X_3] = 0,$$

$$[X_2, X_3]_+ \equiv X_2 X_3 + X_3 X_2 = X_1,$$

$$[X_2, X_2]_+ = [X_3, X_3]_+ = 0.$$

(A.3.63)

Operator X_1 is called even and two others are called *odd*. In the above definition of operator X (A.3.62) parameters a_j are as follows: a_1 is a real number, a_2 , a_3 are Grassmann numbers. It should be kept in mind that an odd generator anticommutes with a Grassmann variable. For example,

$$X_2 a_3 = -a_3 X_2, \tag{A.3.64}$$

The superalgebra SQM(1,2) (A.3.63) has a faithful 2×2 matrix representation

$$X_1 = \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}, \quad X_2 = \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix}, \quad X_3 = \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix}.$$
 (A.3.65)

We adopt the convention that matrices with Grassmann variable entries are linear combinations of the generators and their products, multiplied from the

 $\mathbf{385}$

left by Grassmann parameters. The matrices multiplying the product a_2a_3 are treated as even.

So, for example,

$$\begin{bmatrix} a_1 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} \end{bmatrix}^2 = \\ = a_1 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} a_1 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + a_1 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + \\ + a_1 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} + a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} a_1 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + \\ + a_2 \begin{pmatrix} 0 & 1 \\ 0 & 1 \end{pmatrix} a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} + a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} a_1 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + \\ + a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} =$$
(A.3.66)
$$= a_1^2 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + 2a_1a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} = \\ = a_1^2 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + 2a_1a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} - \\ = a_1^2 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + 2a_1a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} - \\ = a_1^2 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + 2a_1a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} - \\ = a_1^2 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + 2a_1a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} - \\ = a_1^2 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + 2a_1a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} - \\ = a_1^2 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + 2a_1a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} - \\ = a_1^2 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + 2a_1a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} - \\ = a_1^2 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + 2a_1a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix} - \\ = a_1^2 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + 2a_1a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 0 & 1 \end{pmatrix} - \\ = a_1^2 \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix} + 2a_1a_2 \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + 2a_1a_3 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\ = a_1^2 \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix} + \\$$

$$-a_2a_3\left(egin{smallmatrix} 1&0\0&0 \end{smallmatrix}
ight)+a_2a_3\left(egin{smallmatrix} 0&0\0&1 \end{smallmatrix}
ight).$$

Let us determine a one-parameter subgroup of the supergroup as

$$\exp\{\theta X\}, \quad \theta = \text{const.}$$
 (A.3.67)

The expression (A.3.67) in the case of X given by (A.3.62) takes the form

$$\exp\{\theta X\} \equiv \exp\{\theta a_{j} X_{j}\} = 1 + \theta a_{j} X_{j} + \frac{\theta^{2}}{2!} (a_{j} X_{j})^{2} + \dots =$$

$$= e^{\theta a_{1}} \begin{pmatrix} 1 - \frac{\theta^{2}}{2} a_{2} a_{3} & \theta a_{2} \\ \theta a_{3} & 1 + \frac{\theta^{2}}{2} a_{2} a_{3} \end{pmatrix} =$$

$$= e^{\theta a_{1}} \begin{pmatrix} e^{-\frac{\theta^{2}}{2} a_{2} a_{3}} & \theta a_{2} \\ \theta a_{3} & e^{\frac{\theta^{2}}{2}} a_{2} a_{3} \end{pmatrix}.$$
(A.3.68)

For more details on superalgebras see $[112, 134, 199, 144^*]$ and the references cited therein.

Differential invariants (DI) of Poincare algebras AP(1,n), $A\tilde{P}(1,n)$ and conformal algebra AC(1,n)

Following [43*] we give functional bases of second-order differential invariants (DI) of algebras AP(1,n), A $\tilde{P}(1,n)$, and AC(1,n) which are realized on (n + 1 + m)-dimensional space $\mathbb{R}^{n+1}(x) \times \mathbb{R}^m(u)$, $x = (x_0, x_1, \ldots, x_n)$, $u = (u^1, \ldots, u^m) \equiv u(x)$.

1°. DI of AP(1,n) in scalar case m = 1:

$$S_k = S_k(u_{\mu\nu}) = u_{\mu_0\mu_1}u_{\mu_1\mu_2}\cdots u_{\mu_k\mu_0},$$

$$R_k = R_k(u_{\mu}, u_{\mu\nu}) = u_{\mu_0}u_{\mu_k}u_{\mu_0\mu_1}\cdots u_{\mu_{k-1}\mu_k}.$$

 $1^{\circ\circ}$. DI of AP(1,n) when m > 1:

$$\begin{aligned} R_{k} &= R_{k}(u_{\mu}^{r}, u_{\mu\nu}^{1}), \\ S_{jk}(u_{\mu\nu}^{r}, u_{\mu\nu}^{1}) &= u_{\mu_{1}\mu_{2}}^{1} \cdots u_{\mu_{j-1}\mu_{j}}^{1} u_{\mu_{j}\mu_{j+1}}^{r} \cdots u_{\mu_{k}\mu}^{r}, \\ j &= \overline{0, k}, \qquad k = \overline{1, n+1}, \qquad r = \overline{1, m}, \text{ no sum over } r. \end{aligned}$$

$$2^{\circ}. \text{ DI of } A\widetilde{P}(1, n) &= \{AP(1, n), \ D = x\partial + \lambda u^{r}\partial_{u^{r}}\}: \quad \lambda \neq 0, \\ \frac{u^{r}}{u^{1}}, \qquad S_{jk}(u_{\mu\nu}^{r}, u_{\mu\nu}^{1})(u^{1})^{k(2/\lambda - 1)} \end{aligned}$$

$$R_{k}(u_{\mu}^{r}, u_{\mu\nu}^{1})(u^{1})^{(2k/\lambda-k-1)};$$

$$\lambda = 0:$$

$$u^{r}, \qquad S_{jk}(u_{\mu\nu}^{r}, u_{\mu\nu}^{1})(u_{\alpha\alpha}^{1})^{-k},$$

$$R_{r}(u_{\mu}^{r}, u_{\mu\nu}^{1})(u_{\alpha\alpha}^{1})^{-k}.$$

3°. DI of AC(1,n):

$$\begin{aligned} \lambda \neq 0, \qquad S_{jk} \big(\theta^r_{\mu\nu}, \theta^1_{\mu\nu}\big) (u^1)^{k(2/\lambda - 1)} \\ \frac{u^r}{u^1}, \qquad R_k \big(\theta^r_{\mu}, \theta^1_{\mu\nu}\big) (u^1)^{[k(2/\lambda - 1) - 1]}, \end{aligned}$$

where $j = \overline{0, k}$, $k = \overline{1, n+1}$, $r = \overline{2, m}$, with no sum over r;

$$\begin{aligned} \theta^r_{\mu} &= \frac{u^r_{\mu}}{u^r} - \frac{u^1_{\mu}}{u^1}, \\ \theta^r_{\mu\nu} &= \lambda u^r_{\mu\nu} + (1-\lambda) \frac{u^r_{\mu} u^r_{\nu}}{u^r} + \frac{\lambda}{1-n} g_{\mu\nu} \left(u^r_{\beta\beta} - \frac{u^r_{\beta} u^r_{\beta}}{u^r} \right), \\ \lambda &= 0, \quad \Rightarrow \\ u^r, \quad \left(u^1_{\alpha} u^1_{\alpha} \right)^{-2k} S_{jk} \left(w^1_{\mu\nu}, w^r_{\mu\mu} \right), \\ R_k \left(u^r_{\mu}, w^1_{\mu\nu} \right) \left(u^1_{\alpha} u^1_{\alpha} \right)^{-2k+1} \end{aligned}$$

where

here

$$\begin{split} w^r_{\mu\nu} &= u^r_{\alpha} u^r_{\alpha} \left(u^r_{\mu\nu} + \frac{g_{\mu\nu}}{1-n} u^r_{\beta\beta} \right) - u^r_{\beta} (u^r_{\mu} u^r_{\beta\nu} + u^r_{\nu} u^r_{\beta\mu}), \\ j &= \overline{0, k}, \qquad k = \overline{1, n+1}, \qquad r = \overline{2, m}, \end{split}$$

no sum over r.

Differential invariants (DI) of Galilei algebras AG(1,n), $A\widetilde{G}(1,n)$ and Schrödinger algebra ASch(1,n)

Following [42*] we give functional bases of second-order DI of algebras AG(1,n), $A\widetilde{G}(1,n)$, and ASch(1,n).

1°. DI of $AG(1,n) = \{\partial_t, \partial_a, u\partial_u, J_{ab} = x_a\partial_b - x_b\partial_a, G_a = t\partial_a + \mu x_a u\partial_u\}:$ $S_j(\varphi_{ab}), \qquad \varphi = \ln u$ $M_1 = 2\mu\varphi_t + \varphi_a\varphi_a,$ $M_2 = \mu^2\varphi_{tt} + 2\mu\varphi_a\varphi_{at} + \varphi_a\varphi_b\varphi_{ab},$ $R_j(\theta_a, \varphi_{ab}),$

where

$$\theta_a = \mu \varphi_{at} + \varphi_a \varphi_{ab}, \qquad j = \overline{1, n}.$$

1°°. DI of A $\widetilde{\mathbf{G}}(1,n) = \{ \mathbf{A}\mathbf{G}(1,n), D = 2t\partial_t + x_a\partial_a + \lambda u\partial_u \} :$

$$\frac{M_2}{M_1^2}, \qquad \frac{R_j}{M_1^{3+j}}, \qquad \frac{S_j}{M_1^{1+j}}.$$

1°°°. DI of ASch(1,n) = $\left\{ A\widetilde{G}(1,n), \Pi = tD - t^2 \partial_t + \frac{\mu x^2}{2} u \partial_u, \quad \lambda = -\frac{n}{2} \right\}$:

$$\frac{N_2}{N_1^2}, \qquad \frac{\widehat{R}_j}{N_1^{3+j}}, \qquad \frac{\widehat{S}_j}{N_1^{1+j}},$$

$$N_{1} = 2\mu\varphi_{t} + \varphi_{aa} + \varphi_{a}\varphi_{a},$$

$$N_{2} = \mu^{2}\varphi_{tt} + 2\mu\left(\frac{1}{n}\varphi_{t}\varphi_{aa} + \varphi_{a}\varphi_{at}\right) + \varphi_{a}\varphi_{b}\varphi_{ab} + \frac{1}{n}\varphi_{a}\varphi_{a}\varphi_{bb} + \frac{1}{2n}(\varphi_{bb})^{2},$$

where

$$R_{j} = \sum_{\ell=0}^{j} R_{\ell}(\varphi_{aa})^{j-\ell} \frac{(-n)^{\ell} j!}{\ell! (j-\ell)!},$$
$$\widehat{S}_{j} = \sum_{\ell=0}^{j} \frac{(-n)^{\ell} (j-1)! (j+1)!}{(\ell+1)! (j-\ell)!} S_{\ell}(\varphi_{aa})^{j-\ell},$$

 R_j, S_ℓ are given in (A.5.1).

2° DI of AG(1,n), $\mu = 0$:

$$\begin{split} M_1 &= \varphi_t - \varphi_a \sigma_a, \qquad M_2 = \varphi_{tt} - \varphi_{at} \sigma_a, \\ R_j &= R_j(\varphi_a, \varphi_{ab}), \qquad S_j = S_j(\varphi_{ab}) \qquad j = \overline{1, n}. \end{split}$$

 $2^{\circ\circ}$ DI of A $\tilde{G}(1,n)$, $\mu = 0$:

$$\frac{M_1^2}{M_2}, \qquad \frac{R_j}{M_1^{j+1}}, \qquad \frac{S_j}{M_1^{j+1}}.$$

 $2^{\circ\circ\circ}$ DI of ASch(1,n), $\mu = 0$:

$$R_j M^{-\frac{1}{2}(j+1)}, \qquad S_j M^{-\frac{1}{2}(j+1)},$$

where

$$M = (\varphi_t - \sigma_a \varphi_a)^2 + (\varphi_{tt} - \varphi_{at} \sigma_a)(\lambda + \varphi_a \varphi_b r_{ab}),$$
$$\{r_{ab}\} = \{\varphi_{ab}\}^{-1}, \qquad \sigma_a = r_{ab} \varphi_{bt}.$$

Below we list DI in the case of complex scalar fields ψ .

3°. DI of AG(1,n) = { ∂_t , ∂_a , $J_{ab} = x_a \partial_b - x_b \partial_a$, $G_a = t \partial_a + im x_a Q$, $I = \psi \partial_{\psi} + \psi^* \partial_{\psi^*} (Q \equiv \psi \partial_{\psi} - \psi^* \partial_{\psi^*})$ }:

 $m \neq 0 \Rightarrow$

$$\begin{split} \phi + \phi^*, \qquad & \left(\phi = \ln \psi, \qquad \operatorname{Im} \phi = \arctan \frac{\operatorname{Re} \psi}{\operatorname{Im} \psi}\right) \\ M_1 &= 2im\phi_t + \phi_a \phi_a, \qquad M_1^*, \\ M_2 &= -im^2 \phi_{tt} + 2im\varphi_a \varphi_{at} + \phi_a \phi_b \phi_{ab}, \qquad M_2^*, \\ S_{jk} &= S_{jk}(\varphi_{ab}, \varphi_{ab}^*), \\ R_j^1 &= R_j(\theta_a, \phi_{ab}), \qquad R_j^2 = R_j(\theta_a^*, \phi_{ab}), \\ R_j^3 &= R_j(\phi_a + \phi_a^*, \phi_{ab}), \qquad \theta_a = im\phi_{at} + \phi_b \phi_{ab}. \\ 3^{\circ \circ}. \text{ DI of } A\widetilde{G}(1, n) = \{A\widetilde{G}(1, n), D = 2t\partial_t + x_a\partial_a - \frac{1}{2}nI\}, \qquad m \neq 0: \end{split}$$

$$-rac{M_1^*}{M_1}, \qquad rac{M_2}{M_1^2}, \qquad rac{M_2^*}{M_1^2},$$

 $R_j^{\ell} M_1^{-(3+j)}, \quad \ell = 1, 2; \qquad R_j^3 M_1^{-(j+1)}, \qquad S_{jk} M_1^{-(j+1)},$

and $\phi + \phi^*$ under $\lambda = 0$,

$$M_1 \exp \{\lambda(\phi + \phi^*)\} ext{ under } \lambda
eq 0.$$

3°°°. DI of

$$\begin{split} \operatorname{ASch}(1,n) &= \{ \operatorname{A}\widetilde{\mathrm{G}}(1,n), \, \Pi = tD - t^2 \partial_t + \frac{1}{2} m x_a x_a I \}, \quad \lambda = -\frac{1}{2}n, \quad m \neq 0 : \\ N_1 \exp\left\{\frac{4}{n}(\phi + \phi^*)\right\}, \quad \frac{N_1}{N_1^*}, \quad \frac{N_2}{N_1^2}, \quad \frac{N_2^*}{N_1^2} \\ R_j^\ell N_1^{-(3+j)}, \quad (\ell = 1,2); \quad \widehat{R}_j^3 N_1^{-(j+1)}, \quad \widehat{S}_{jk} N_1^{-(1+j)}, \end{split}$$

where

$$\begin{split} N_{1} &= 2im\phi_{t} + \phi_{a}\phi_{a} + \phi_{aa}, \\ N_{2} &= -m^{2}\phi_{tt} + 2im\left(\phi_{a}\phi_{at} + \frac{1}{n}\phi_{t}\phi_{aa}\right)\phi_{a}\phi_{b}\phi_{ab} + \frac{1}{n}\phi_{a}\phi_{a}\phi_{bb} + \frac{1}{2n}\varphi_{aa}^{2}, \\ \widehat{S}_{jk} &= \sum_{\ell=0}^{j}\sum_{r=0}^{k} \left[S_{r\ell}(-n)^{\ell}C_{k}^{r}C_{k}^{\ell+1-r}(\phi_{aa})^{k-r}(\phi_{aa}^{*})^{j-\ell-k+r} + \\ &+ j(\phi_{aa})^{k}(\phi_{0a}^{*})^{j-k-1}\right], \end{split}$$

$$\widehat{R}_{j}^{\ell} = \sum_{k=0}^{j} R_{k}^{j} (\varphi_{aa})^{j-k} \frac{(-n)^{k} j!}{k! (j-k)!}, \qquad \ell = 1, 2, 3.$$

4°. DI of AG(1,n), m = 0:

$$\begin{split} \phi + \phi^*, & M_1 = \phi_t - \phi_a \sigma_a, & M_1^* \\ M_2 = \phi_{tt} - \phi_{at} \sigma_a, & M_2^*, & (\phi_{at} = \sigma_a, \phi_{ab}), \\ R_j^1 = R_j(\phi_a, \phi_{ab}), & R_j^2(\phi_a^*, \phi_{ab}), \\ R_j^3 = R_j(\sigma_a - \sigma_a^*, \psi_{ab}), & S_{jk}. \end{split}$$

4°°. DI of A $\widetilde{\mathrm{G}}(1,n), m = 0$:

$$rac{M_1}{M_1^*}, \qquad rac{N_2}{M_1^2}, \qquad rac{M_2^*}{M_1^2}, \quad rac{R_j^{t}}{M_1^{j+1}}, \quad rac{S_{jk}}{M_1^{j+1}}$$

and $M_1 \exp \left\{ \frac{2}{\lambda} (\phi + \phi^*) \right\}$ under $\lambda \neq 0$; $\phi + \phi^*$ under $\lambda = 0$. 4°°°°. DI if ASch(1,n), m = 0:

$$\begin{split} \lambda &= 0 \; \Rightarrow \; \phi + \phi^*, \qquad \frac{N_1^2}{N_2}, \qquad \frac{N_1^{*2}}{N_2}, \qquad \frac{(S_{jk})^2}{N_1^{j+1}}, \\ & \left(R_j^\ell\right)^2 N_1^{-(j+1)}, \qquad \ell = 1, 2, 4; \\ \lambda &\neq 0 \; \Rightarrow \; N_1 \exp\left\{\frac{4}{\lambda}(\phi + \phi^*)\right\}, \qquad \frac{N_1}{N_1^*}, \\ & N_3 \exp\left\{\frac{3}{\lambda}(\phi + \phi^*)\right\}, \qquad \frac{(R_j^\ell)^2}{N_1^{j+1}}, \quad (\ell = 1, 2, 3), \qquad \frac{(S_{jk})^2}{N_1^{j+1}}, \end{split}$$

where

$$\begin{split} N_{1} &= (\phi_{t} - \sigma_{a}\phi_{a})^{2} + (\phi_{tt} - \sigma_{a}\phi_{at}) \left(\lambda + \phi_{a}\phi_{b}r_{ab}\right), \\ &\{r_{ab}\} = \{\phi_{ab}\}^{-1}, \\ N_{2} &= (\phi_{t} - \phi_{c}\sigma_{c}) \phi_{a}^{*}\phi_{b}^{*}r_{ab}^{*} - (\phi_{t}^{*} - \phi_{c}^{*}\sigma_{c}^{*}) \phi_{a}\phi_{b}r_{ab}, \\ N_{3} &= \tau_{a} \left(\phi_{a}^{*} - \phi_{a}\right) + \phi_{t} - \phi_{t}^{*}, \\ &\tau_{a} &= (\phi_{b}\phi_{t} + \lambda\phi_{bt}) r_{ab}, \qquad \{\widehat{r}_{ab}\} = \{\lambda\phi_{ab} + \phi_{a}\phi_{b}\}^{-1}, \\ R_{j}^{1} &= R_{j}(\phi_{a}, \phi_{ab}), \qquad R_{j}^{2} &= R_{j}(\phi_{a}^{*}, \phi_{ab}), \\ R_{j}^{3} &= R_{j}(\sigma_{a} - \sigma_{a}^{*}, \phi_{ab}), \qquad R_{j}^{4} &= R_{j}(\rho_{a}, \phi_{ab}), \\ &(\rho_{a} &= (\phi_{t} - \sigma_{b}\phi_{b})(\phi_{c}^{*}r_{ac} - \phi_{c}r_{ac}^{*}) - \phi_{b}\phi_{d}r_{bd}(\sigma_{a} - \sigma_{a}^{*})); \\ S_{jk} &= S_{jk}(\phi_{ab}, \phi_{ab}^{*}). \end{split}$$

Compatibility and solutions of the overdetermined d'Alembert-Hamilton system

The system

$$\Box \omega = F_1(\omega),$$

$$\omega_{\mu} \omega^{\mu} = F_2(\omega),$$
(A.6.1)

where $\omega_{\mu} = \partial \omega / \partial x^{\mu}$, $\mu = \overline{0,3}$, F_1 , F_2 are arbitrary smooth functions, often arises from reduction of a Poincare invariant PDE to an ODE (see Paragraphs 1.4, 2.1, 5.12). Here we consider, following [86^{*}, 122^{*}, 130^{*}], the compatibility of the overdetermined system (A.6.1) and describe its solutions.

By means of an appropriate change of dependent variable, system (A.6.1) can be transformed to the form

$$\Box \omega = F(\omega),$$

$$\omega_{\mu}\omega^{\mu} = \lambda, \qquad \lambda = \text{const.}$$
(A.6.2)

Lemma A.6.1. Solutions of the system (A.6.2) satisfy the indentities

$$\omega_{\mu\nu_{1}}\omega_{\nu_{1}\mu} = -\lambda\dot{F},$$

$$\omega_{\mu\nu_{1}}\omega_{\nu_{1}\nu_{2}}\omega_{\nu_{2}\mu} = \frac{1}{2!}\lambda^{2}\ddot{F},...$$

$$(A.6.3)$$

$$\omega_{\mu\nu_{1}}\omega_{\nu_{1}\nu_{2}}\cdots\omega_{\nu_{n}\mu} = \frac{(-1)^{n}}{n!}\lambda^{n}\frac{d^{n}F}{d\omega^{n}},$$

where
$$\omega_{\alpha\beta} = \frac{\partial^2 \omega}{\partial x^{\alpha} \partial x^{\beta}}, \, \alpha, \, \beta = \overline{0,3}, \, n \ge 1, \, \dot{F} \equiv \frac{dF}{d\omega}.$$

Lemma A.6.2. Solutions of the system (A.6.2) satisfy the equality

$$\det\left(\omega_{\mu\nu}\right) = 0. \tag{A.6.4}$$

The proofs of these lemmas one can make straightforwardly.

Theorem A.6.1. The necessary condition of compatibility of the overdetermined system (A.6.2) is

$$F(\omega) = \begin{cases} 0, \\ \lambda(\omega + c_1)^{-1}, \\ 2\lambda(\omega + c_1)[(\omega + c_1)^2 + c_2]^{-1}, \\ 3\lambda[(\omega + c_1)^2 + c_2][(\omega + c_1)^3 + 3c_2(\omega + c_1) + c_3]^{-1}, \end{cases}$$
(A.6.5)

where c_1, c_2, c_3 are arbitrary constants.

~

Proof. By direct (and rather tiresome) verification one can be convinced of the following identity

$$\delta(\omega_{\mu\nu_1}\omega_{\nu_1\nu_2}\omega_{\nu_2\nu_3}\omega_{\nu_3\mu}) - 8(\Box\omega)(\omega_{\mu\nu_1}\omega_{\nu_1\nu_2}\omega_{\nu_2\mu}) -$$

$$-3(\omega_{\mu\nu_1}\omega_{\nu_1\mu})^2 + 6(\Box\omega)^2(\omega_{\mu\nu_1}\omega_{\nu_1\mu}) - (\Box\omega)^4 = 24\det(\omega_{\alpha\beta}).$$
(A.6.6)

Substituting (A.6.3), (A.6.4) into (A.6.6) we obtain the nonlinear ODE for

$$\lambda^{3}\bar{F} + 4\lambda^{2}F\bar{F} + 3\lambda^{2}\dot{F}^{2} + 6\lambda\dot{F}F^{2} + F^{4} = 0, \qquad (A.6.7)$$

The general solution of Equation (A.6.7) is given in (A.6.5). Thus, the theorem is proved.

Remark A.6.1. Compatibility of the three-dimensional d'Alembert-Hamilton system has been investigated in detail by Collins [41], who essentially used geometrical methods which could not be generalized to higher dimensions.

Using Lie's method one can prove the following statement.

Theorem A.6.2. [86*, 130*] System (A.6.2) is invariant under the 15-parameter conformal group C(1,3) iff

$$F(\omega) = 3\lambda(\omega+c)^{-1}, \qquad \lambda > 0, \qquad c = \text{const.}$$
 (A.6.8)

Remark A.6.2. Formula (A.6.8) can be obtained from (A.6.5) by putting $c_2 = c_3 = 0$. So Theorem A.6.2 demonstrates the close connection between compatibility of system (A.6.2) and its symmetry.

394

In conclusion, let us list the explicit form of exact solution of system (A.6.2), taking into account Theorem A.6.1.

 Table A.6.1. Cases of compatibility of system (A.6.2) and corresponding solution.

N	λ	$F(\omega)$	$\omega = \omega(x)$
1	1	0	dx
2	1	ω^{-1}	$[(dx)^2 - (ax)^2]^{1/2}$
3	1	$2\omega^{-1}$	$[(dx)^2 - (ax)^2 - (bx)^2]^{1/2}$
4	1	$3\omega^{-1}$	$(x_ u x^ u)^{1/2}$
5	-1	0	$ax\cos h_1 + bx\sin h_1 + g_1,$
			$dx - ax\cos h_2 - bx\sin h_2 - g_2 = 0$
6	-1	$-\omega^{-1}$	$[(ax+h_1)^2+(bx+h_2)^2]^{1/2}$
7	-1	$-2\omega^{-1}$	$[(ax)^2 + (bx)^2 + (cx)^2]^{1/2}$
8	0	0	h_1

In this table, h_1 , g_1 are arbitrary smooth functions of dx + cx, and h_2 , g_2 are arbitrary smooth functions of $\omega + cx$; a_{μ} , b_{μ} , c_{μ} , d_{μ} are arbitrary real constants satisfying conditions (2.1.27).

Q-Conditional Symmetry of the Heat Equation

Here we consider in full detail, as a simple but non-trivial example, how to find and use Q-conditional symmetry of the one-dimensional heat equation*

$$u_0 = u_{11}$$
 (A.7.1)

 $(u = u(x_0, x_1), u_0 = \partial u / \partial x_0, u_1 = \partial u / \partial x_1, \text{ and so on}).$

The definition of Q=symmetry is given in Sec. 5.7 (see Definition 5.7.3, p.328). The Lie-maximal invariance algebra of Equation (A.7.1) is written in (5.1.6). The problem of finding non-classical symmetry (in our terminology: Q-conditional symmetry) was first put forward by Bluman and Cole [131*]. However, in this important paper the authors did not give explicitly any operators which would differ from those of (5.1.6). Below we will present a complete investigation of this problem.

The general form of a first-order operator is

$$Q = A(x_0, x_1, u)\partial_0 + B(x_0, x_1, u)\partial_1 + C(x_0, x_1, u)\partial_u,$$
(A.7.2)

where A, B, C are some differentiable functions of x_0, x_1, u to be determined from the invariance condition (5.7.7). It will be noted that because of the imposed condition (5.7.6)

$$Qu = 0 \iff Au_0 + Bu_1 = C, \tag{A.7.3}$$

there are really only two independent cases of operator (A.7.2).

Theorem A.7.1. The heat equation (A.7.1) is Q-conditionally invariant under operator (A.7.2) if and only if its coordinates are as follows:

^{*} Some results herein were obtained in collaboration with R.E.Popovich.

Case 1:

$$A = 1, \quad B = W^1(x_0, x_1), \quad C = W^2(x_0, x_1)u + W^3(x_0, x_1)$$
 (A.7.4)

and functions $ec{W} = ec{W}(x_0, x_1) = \{W^1, W^2, W^3\}$ satisfy

$$(\partial_0 + 2W_1^1 - \partial_{11}) \vec{W} = \vec{F}, \qquad \vec{F} \equiv \{2W_1^2, 0, 0\};$$
 (A.7.5)

Case 2:

$$A = 0, \quad B = 1, \quad C = v(x_0, x_1, u)$$
 (A.7.6)

and functions $v = v(x_0, x_1, u)$ satisfies the PDE

$$v_0 = v_{11} + 2vv_{1u} + v^2 v_{uu}. \tag{A.7.7}$$

Proof. From the criterion of invariance

$$Q(u_0 - u_{11}) \Big|_{\substack{u_0 = u_{11} \\ Qu = 0}} = 0,$$
 (A.7.8)

absolutely analogously to the standard Lie's algorithm, one finds the defining equation for the coordinates of operator (A.7.2) which can be reduced to (A.7.4)–(A.7.7). It is to be pointed out that unlike Lie's algorithm, in the cases considered above, the defining equations (A.7.5), (A.7.7) are nonlinear ones, which is a typical feature of Q-conditional invariance.

It goes without saying that Q-conditional invariance includes Lie's invariance in particular. So, in our case of the heat equation, we obtain infinitesimals (5.1.6) as simplest solutions of (A.7.5), (A.7.7):

$$\begin{split} A &= 1, \quad \vec{W} = 0 \Rightarrow Q = \partial_0 \\ A &= v = 0, \quad B = 1 \Rightarrow Q = \partial_1, \\ A &= 0, \quad B = 1, \quad v = -(x_1/2x_0)u \Rightarrow Q = G = x_0\partial_1 - \frac{1}{2}x_1u\partial_u, \\ A &= 1, \quad W^1 = x_1/2x_0, \quad W^2 = W^3 = 0 \Rightarrow Q = D, \\ A &= 1, \quad W^1 = x_1/x_0, \quad W^2 = -(2x_0 + x_1^2)/4x_0^2, \quad W^3 = 0 \Rightarrow Q = \Pi. \end{split}$$
(A.7.9)

Remark A.7.1. The system of defining equations (A.7.5) was first obtained by Bluman and Cole [131*]. Further investigation of system (A.7.5) was continued in [132*], where the question of linearization of the first two equations of (A.7.5) had been studied. The general solution of the problem of linearization of Equations (A.7.5), (A.7.7) will be given after a while.

Now let us list some concrete operators (A.7.2) of Q-conditional invariance of Equation (A.7.1) obtained as partial solutions of the defining equations (A.7.5), (A.7.7). In the following table we also give corresponding invariant ansatze and the reduced equations.

Table A.7.1.

N	Operator Q	Ansatz $u =$	Reduced ODE
1.	$-x_1\partial_0+\partial_1$	$=\varphi(x_0+x_1^2/2)$	$\varphi^{\prime\prime} = 0$
2.	$-x_1\partial_0+\partial_1+x_1^3\partial_u$	$= \varphi(x_0 + x_1^2/2) + \frac{1}{4}x_1^4$	$arphi^{\prime\prime}=-3$
3.	$x_1^2\partial_0 - 3x_1\partial_1 - 3u\partial_u$	$= x_1\varphi(x_0 + \tfrac{1}{6}x_1^2)$	$\varphi'' = 0$
4.	$x_1^2\partial_0 - 3x_1\partial_1 -$	$= x_1 \varphi(x_0 + \frac{1}{6}x_1^2) + \frac{1}{15}x_1^5$	$\varphi = -15$
	$-(3u+x_1^5)\partial_u$		
5.	$x_1\partial_1 + u\partial_u$	$=x_1\varphi(x_0)$	arphi'=0
6.	$\operatorname{cth} x_1 \partial_1 + u \partial_u$	$= \varphi(x_0) \operatorname{ch} x_1$	$\varphi' - \varphi = 0$
7.	$\cot x_1 \partial_1 - u \partial_u$	$= \varphi(x_0) \cos x_1$	$\varphi' + \varphi = 0$
8.	$\partial_1 - u \partial_u - rac{u}{2x_0 - x_1} \partial_u$	$=(2x_0-x_1)e^{-x_1}\varphi(x_0)$	$\varphi' - \varphi = 0$
9.	$\partial_1 - \sqrt{-2(x_0+u)}\partial_u$	$= -x_0 - rac{1}{2}[x_1 + \varphi(x_0)]^2$	arphi'=0
10.	$(x_0+rac{1}{2}x_1^2)\partial_0-x_1\partial_1$	$=\varphi(x_0x_1+\tfrac{1}{3!}x_1^3)$	$\varphi^{\prime\prime}=0$

Theorem A.7.2 The Lie-maximal invariance algebra of system (A.7.5) is given by the operators

$$\begin{aligned} \partial_{0}, \quad \partial_{1}, \quad G^{(1)} &= x_{0}\partial_{1} + \partial_{W^{1}} - \frac{1}{2}W^{1}\partial_{W^{2}} - \frac{1}{2}x_{1}W^{3}\partial_{W^{3}}, \\ D^{(1)} &= 2x_{0}\partial_{0} + x_{1}\partial_{1} - W^{1}\partial_{W^{1}} - 2W^{2}\partial_{W^{2}}, \\ \Pi^{(1)} &= x_{0}\left(x_{0}\partial_{0} + x_{1}\partial_{1} - W^{1}\partial_{W^{1}} - 2W^{2}\partial_{W^{2}} - \frac{5}{2}W^{3}\partial_{W^{3}}\right) + \\ &+ x_{1}\left(\partial_{W^{1}} - \frac{1}{2}W^{1}\partial_{W^{2}}\right) - \frac{1}{2}\partial_{W^{2}} - \frac{1}{4}x_{1}^{2}W^{3}\partial_{W^{3}}, \\ X^{(1)} &= (f_{0} + f_{1}W^{1} - fW^{2})\partial_{W^{3}}, \qquad I^{(1)} = W^{3}\partial_{W^{3}}, \end{aligned}$$

where $f = f(x_0, x_1)$ is an arbitrary solution of (A.7.1), that is, $f_0 = f_{11}$.

Theorem A.7.3 The Lie-maximal invariance algebra of system (A.7.7) is given by the operators

$$\partial_{0}, \quad \partial_{1}, \quad D^{(2)} = 2x_{0}\partial_{0} + x_{1}\partial_{1} + u\partial_{u}, \quad D^{(3)} = u\partial_{u} + v\partial_{v},$$

$$G^{(2)} = x_{0}\partial_{1} - \frac{1}{2}x_{1}(u\partial_{u} + v\partial_{v}) - \frac{1}{2}u\partial_{v}, \qquad (A.7.11)$$

$$\Pi^{(2)} = x_{0}\left(x_{0}\partial_{0} + x_{1}\partial_{1} - \frac{1}{2}u\partial_{u} - \frac{3}{2}v\partial_{v}\right) - \frac{1}{4}x_{1}^{2}\left(u\partial_{u} + v\partial_{v}\right) - \frac{1}{2}x_{1}^{2}u\partial_{v},$$

$$X^{(2)} = f\partial_{u} + f_{1}\partial_{v}, \qquad (f_{0} = f_{11}).$$

One can get the proofs of these two theorems by means of the standard Lie's algorithm.

Operators (A.7.10), (A.7.11) can be used to find exact solutions of Equations (A.7.5), (A.7.7). In particular, using the formula of generating solutions at the expense on invariance under $\Pi^{(2)}$

$$v^{II}(x_0, x_1, u) = (1 - \theta x_0)^{3/2} \exp\left\{\frac{\theta x_1^2}{4(1 - \theta x_0)}\right\} v^{I}(x'_0, x'_1, u') + + \frac{\theta}{1 - \theta x_0} \cdot \frac{x_1 u}{2}; \qquad x'_0 = \frac{x_0}{1 - \theta x_0}, \qquad x'_1 = \frac{x_1}{1 - \theta x_0}, \qquad (A.7.12)$$
$$u' = (1 - \theta x_0)^{1/2} \exp\left\{-\frac{1}{4}\frac{\theta x_1^2}{1 - \theta x_0}\right\}, \qquad (\theta = \text{const})$$

one can construct new solutions of Equations (A.7.7) starting from known ones.

Solutions of Equations (A.7.5), (A.7.11) can be obtained by the use of reduction on subalgebras of the invariance algebras (A.7.10), (A.7.11). For example, using the subalgebra $\langle \partial_0 + aI^{(1)} \rangle$ of the algebra (A.7.10) we find the following solution of the system (A.7.5)

$$W^{1} = \frac{c_{1}^{2} - c_{3}^{2}}{-c_{1} \tan(c_{1}x + c_{2}) + c_{3} \tan(c_{3}x + c_{4})},$$

$$W^{2} = -c_{1}c_{3}\frac{c_{1} \tan(c_{3}x + c_{4}) - c_{3} \tan(c_{1}x + c_{2})}{-c_{1} \tan(c_{1}x + c_{2}) + c_{3} \tan(c_{3}x + c_{4})},$$

$$W^{3} = (\varphi_{xx} - W^{1}\varphi_{x} - W^{2}\varphi)e^{at},$$

(A.7.13)

where c_1, \ldots, c_4 are arbitrary constants, $\varphi = \varphi(x), \ \varphi_{xx} = a\varphi$.

Theorem A.7.4. The system (A.7.5) is reduced to the system of disconnected heat equations

$$\vec{z}_0 = \vec{z}_{11}, \qquad \left(\vec{z} = \vec{z}(x_0, x_1) = \{z^1, z^2, z^3\}\right)$$
 (A.7.14)

with the help of the nonlocal transformations

$$W^{1} = -\frac{z_{11}^{1} z^{2} - z^{1} z_{11}^{2}}{z_{1}^{1} z^{2} - z^{1} z_{1}^{2}},$$
(A.7.15)

$$W^{2} = -\frac{z_{11}^{1}z_{1}^{2} - z_{1}^{1}z_{11}^{2}}{z_{1}^{1}z^{2} - z^{1}z_{1}^{2}}, \qquad W^{3} = z_{11}^{3} + W^{1}z_{1}^{3} - W^{2}z^{3}.$$

Expressions (A.7.15) result in (after using the corresponding operator (A.7.2), (A.7.4)) the ansatz

$$u = z^1 \varphi(\omega) + z^3, \qquad \omega = \frac{z^2}{z^1}$$
 (A.7.16)

 $(z^1,z^2,z^3$ are solutions of (A.7.14)), and the reduced equation is $\varphi''=0.$ This means that

$$u = c_1 z^1 + c_2 z^2 + c_3 z^3. (A.7.17)$$

So, we get just the well-known supersosition principle for the heat equation.

Letting $W^2 = W^3 = 0$ we get from (A.7.5) the Burger's equation

$$W_0^1 + 2W^1 W_1^1 = W_{11}^1. (A.7.18)$$

Using Hopf-Cole transformation (see Sec. 5.1) one obtains solutions of Equation (A.7.18) in the form

$$W^1 = -\partial_1 \ln f = -\frac{f_1}{f}, \qquad (f_0 = f_{11}).$$
 (A.7.19)

This results in the operator

$$Q = f\partial_0 - f_1\partial_1. \tag{A.7.20}$$

Q-conditional symmetry of Equation (A.7.1) under the operator Q (A.7.20) leads to the following statement.

Theorem A.7.5. If function f is an arbitrary solution of the heat equation (A.7.1) and u is the general integral of the ODE

$$f_1 dx_0 + f dx_1 = 0, \tag{A.7.21}$$

then u satisfies Equation (A.7.1).

Proof. We note that Equation (A.7.21) is a perfect differential equation and therefore its general solution, $u(x_0, x_1) = c$, possesses the following property

$$u_0 = f_1, \qquad u_1 = f.$$
 (A.7.22)

Having used (A.7.22) we obtain

$$u_0 - u_{11} = f_1 - f_2 \equiv 0,$$

and the theorem is proved.

Theorem A.7.5 may be considered as another algorithm of generating solutions of Equation (A.7.1). Indeed, even starting from a rather trivial solution of the heat equation, u = 1, we get the chain of quite interesting solutions

$$1 \to x_1 \to x_0 + \frac{x_1^2}{2!} \to x_0 x_1 + \frac{x_1^3}{3!} \to \cdots,$$
 (A.7.23)

400

and among them the solutions

$$\frac{x_1^{2m}}{(2m)!} + \frac{x_0}{1!} \frac{x_1^{2m-2}}{(2m-2)!} + \frac{x_0^2}{2!} \frac{x_1^{2m-4}}{(2m-4)!} + \dots + \frac{x_0^{m-1}}{(m-1)!} \frac{x_1^2}{2!} + \frac{x_0^m}{m!}, \quad (A.7.24)$$

$$\frac{x_1^{2m+1}}{(2m+1)!} + \frac{x_0}{1!} \frac{x_1^{2m-1}}{(2m-1)!} + \frac{x_0^2}{2!} \frac{x_1^{2m-3}}{(2m-3)!} + \dots + \frac{x_0^{m-1}}{(m-1)!} \frac{x_1^2}{3!} + \frac{x_0^m}{m!} \frac{x_1}{1!}, \quad (A.7.25)$$

where m = 1, 2, 3, ...

It will be also noted thay supposing function v in (A.7.7) to be independent of x_1 , and denoting

$$v = \frac{1}{w(x_0, u)}$$
 (A.7.26)

we get instead of (A.7.7) the following remarkable nonlinear heat equation

$$w_0 = \partial_u \left(w^{-2} w_u \right). \tag{A.7.27}$$

One easily see that the operator

$$Q = w(x_0, u)\partial_1 + \partial_u \tag{A.7.28}$$

sets the connection between Equations (A.7.27) and (A.7.1):

$$w_{0} - \partial_{u}(w^{-2}w_{u}) = \frac{1}{u_{1}}\partial_{1}\left(\frac{u_{0} - u_{11}}{u_{1}}\right),$$

$$u_{0} - u_{11} = \frac{1}{w}\int [w_{0} - \partial_{u}(w^{-2}w_{u})]du$$
(A.7.29)

by means of the change of variables

$$w(x_0, u) = \frac{\partial x_1(x_0, u)}{\partial u}, \qquad \frac{\partial u(x_0, x_1)}{\partial x_1} = \frac{1}{w(x_0, u)}.$$
 (A.7.30)

This result has been obtained differently in $[133^*, 134^*]$.

If we suppose v from (A.7.7) to have the form

$$v = \varphi(x_0, x_1)u, \tag{A.7.31}$$

then (A.7.7) is reduced to the Burger's equation for φ

$$\varphi_0 = 2\varphi\varphi_1 + \varphi_{11}, \tag{A.7.32}$$

and one may say that operator

$$Q = \partial_1 + \varphi u \partial_u \tag{A.7.33}$$

sets the connection between Equations (A.7.32) and (A.7.1) via the substitution

$$\varphi = f_1 / f \tag{A.7.34}$$

Letting

$$v = \varphi(x_0, x_1)u + h(x_0, x_1) \tag{A.7.35}$$

and substituting it into (A.7.7) one finds the Burger's equation (A.7.32) for function φ and the following equation for h

$$h_0 = 2h\varphi_1 + h_{11}. \tag{A.7.36}$$

Having made the change of variables

$$h = -(f_1/f)g + g_1 \tag{A.7.37}$$

we reduce (A.7.32), (A.7.36) to two disconnected heat equations

$$f_0 = f_{11} \qquad g_0 = g_{11}. \tag{A.7.38}$$

Now we see how to linearize Equation (A.7.7) in the general case. Let us introduce the notations

$$S_1(x_0, x_1, u, v) = v_0 - (v_{11} + 2vv_{1u} + v^2 v_{uu}).$$
 (A.7.39)

After changing the variables to

$$v = -\frac{z_1}{z_u}, \qquad z = z(x_0, x_1, u)$$
 (A.7.40)

we get

$$S_1(x_0, x_1, u, v) = -\frac{1}{z_u} (\partial_1 + v \partial_u) S_2(x_0, x_1, u, z),$$
(A.7.41)

where

$$S_2(x_0, x_1, u, z) = z_0 - z_{11} + 2\frac{z_1}{z_u}z_{1u} - \frac{z_1^2}{z_u^2}z_{uu}.$$
 (A.7.42)

Having applied the hodograph transformation

$$y_0 = x_0, \qquad y_1 = x_1, \qquad y_2 = z, \qquad R = u$$
 (A.7.43)

we get

$$S_2(x_0, x_1, u, z) = -\frac{1}{R_2}(R_0 - R_{11}), \qquad (A.7.44)$$

where

$$R=R(y_0,y_1,y_2).$$

So, we see that Equations (A.7.5), (A.7.7) are reduced to the heat equations (A.7.14), (A.7.44), respectively.

402

On Nonlocal Symmetries of Nonlinear Heat Equation*

We will show how to find and use nonlocal symmetries of the nonlinear heat equation to construct its exact solutions.

The complete group classification of nonlinear one-dimensional heat equation

$$u_0 = \partial_1(F(u)u_1), \tag{A.8.1}$$

 $(u = u(x_0, x_1), F(u)$ is an arbitrary differentiable function, $u_0 = \partial u / \partial x_0$, and so on) was made by Ovsyannikov (see [161]). His results may be summarized as follows.

Theorem A.8.1. Depending on the function F(u), the maximal invariance algebra (MIA) of Equation (A.8.1) is determined by the following basis elements:

1) F(u) is an arbitrary function, then MIA is three-dimensional,

$$\partial_0 = \frac{\partial}{\partial x_0}, \qquad \partial_1 = \frac{\partial}{\partial x_1}, \qquad D_1 = 2x_0\partial_0 + x_1\partial_1$$
(A.8.2)

2) $F(u) = \lambda u^k$, then MIA is four-dimensional,

$$\partial_0, \qquad \partial_1, \qquad D_1, \qquad D_2 = 2x_1\partial_1 + \frac{2}{k}u\partial_u$$
 (A.8.3)

3) $F(u) = \lambda e^{u}$, then MIA is four-dimensional,

$$\partial_0, \quad \partial_1, \quad D_1, \quad D_3 = x_1 \partial_1 + 2 \partial_u$$
 (A.8.4)

^{*} Results in this appendix were obtained with the collaboration of V.A.Tychinin and T.K.Amerov.

4) $F(u) = \lambda u^{-4/3}$, then MIA is five-dimensional,

$$\partial_0, \quad \partial_1, \quad D_1, \quad D_4 = x_1 \partial_1 - \frac{3}{2} u \partial_u, \quad \Pi = x_1^2 \partial_1 - 3x_1 u \partial_u.$$
 (A.8.5)

It is also known (see [135*], for example) that the chain of transformations

$$u(x_0, x_1) = \frac{\partial v(x_0, x_1)}{\partial x_1}, \qquad (A.8.6)$$

$$x_0 = t,$$
 $x_1 = w(t, x),$ $v = x;$ (A.8.7)

$$\frac{\partial w(t,x)}{\partial x} = z(t,x) \tag{A.8.8}$$

does not go out from the class of equations (A.8.1). It means that having made transformations (A.8.6)-(A.8.8) one gets from (A.8.1) another equation

$$z_t = \partial_x (F^*(z) z_x), \tag{A.8.9}$$

where

$$F^*(z) = z^{-2}F(z^{-1}).$$
 (A.8.10)

Below we use transformations (A.8.6)-(A.8.8) to construct nonlocal ansatze which reduce Equation (A.8.1) to ODEs. We also give nonlocal formulae of generating solutions and superposition of solutions of Equation (A.8.1) for some nonlinearities F(u).

First of all we consider the equation

$$u_0 = \partial_1(u^{-2}u_1). \tag{A.8.11}$$

As it follows from (A.8.10) this equation is reduced by means of (A.8.6)-(A.8.8) to the linear heat equation

$$z_t = z_{xx} \tag{A.8.12}$$

whose maximal invariance algebra (MIA) is 6-dimensional, and it is given in (5.1.6). The difference between MIAs of these two (equivalent in a sense) Equations (A.8.11) and (A.8.12) says that nonlinear Equation (A.8.11) possessess a non-Lie symmetry which corresponds to Lie symmetry of the heat equation (A.8.12) generated by Galilei and projective operators (compare MIAs (A.8.3) and (5.1.6)). It is obvious that this non-Lie symmetry of equation (A.8.11) cannot be obtained with the help of Lie's method.

Let us make use of this fact and construct ansatz for Equation (A.8.11) which correspond to those of Equation (A.8.12) invariant under operators

$$G = t\partial_x - \frac{1}{2}xz\partial_z,$$

$$\Pi = t(t\partial_t + x\partial_x - \frac{1}{2}z\partial_z) - \frac{1}{4}x^2z\partial_z.$$
(A.8.13)

404

Using transformations (A.8.6)-(A.8.8) we find

$$u(x_0, x_1) = [x_0 x_1 + x_1 h(\omega)]^{-1}, \qquad \omega = \tau + x_0^2,$$

$$\exp\{x_0 \tau + \frac{2}{3} x_0^3\}\varphi(\omega) = x_1;$$
(A.8.14)

$$u(x_0, x_1) = \frac{2(x_0^2 + 1)}{x_1 [2(x_0^2 + 1)^{1/2} h(\omega) - \tau x_0]}, \qquad \omega = \frac{\tau}{\sqrt{1 + x_0^2}},$$

$$\exp\left\{\lambda \arctan x_0 - \frac{x_0 \tau}{4(1 + x_0^2)}\right\} \varphi(\omega) = x_1 (1 + x_0^2)^{1/4}.$$
(A.8.15)

In (A.8.14), (A.8.15) $\tau = \tau(x_0, x_1)$ is a function parameter, $h = \dot{\varphi}/\varphi$. Ansatze (A.8.14), (A.8.15) reduced (A.8.11) to the Riccati equation

$$\dot{h} + h^2 = \omega, \tag{A.8.16}$$

$$\dot{h} + h^2 = -\frac{1}{4}\omega^2 + \lambda,$$
 (A.8.17)

respectively, or in terms of φ we have

$$\begin{split} \ddot{\varphi} - \omega \varphi &= 0, \\ \ddot{\varphi} + (\frac{1}{4}\omega^2 - \lambda)\varphi &= 0. \end{split}$$

Solutions of these equations are expressed in terms of Bessel functions.

The connection between solutions of Equations (A.8.11) and (A.8.12), as follows from (A.8.6)-(A.8.8), is

$$u(x_0, x_1) = \left[\frac{\partial z(x_0, \tau)}{\partial \tau}\right]^{-1}, \qquad (A.8.18)$$

where $\tau = \tau(x_0, x_1)$ is a functional parameter to be determined from the condition

$$z(x_0, \tau) = x_1. \tag{A.8.19}$$

Now consider the following question. Since Equation (A.8.12) is linear, then function

$$z^{2}(t,x) = Qz^{1}(t,x)$$
 (A.8.20)

will be its solution provided $\frac{1}{2}$ is a given solution and Q is any symmetry operator from (5.1.6). How to obtain analogous results for Equation (A.8.11)? To this end one has to use the connection between the two Equations (A.8.11) and (A.8.12) given by (A.8.6)–(A.8.8). So, choosing in (A.8.20) as Q operator ∂_x we find corresponding formula of generating solutions for Equation (A.8.11)

$$u^{2}(x_{0}, x_{1}) = -[u^{1}(x_{0}, \tau)]^{3} \left[\frac{\partial u(x_{0}, \tau)}{\partial \tau}\right]^{-1},$$
 (A.8.21)

where $\overset{2}{u}$ is a new solution and $\overset{1}{u}$ is a given one; τ is to be determined from the condition

$$u^{1}(x_{0},\tau) = x_{1}^{-1}.$$
 (A.8.22)

For example, starting from solution

$$u'(x_0,\tau) = \frac{\sqrt{x_0}}{x_1} \left(-\ln\sqrt{x_0} \, x_1\right)^{-1/2},\tag{A.8.23}$$

we obtain with the help of (A.8.21), (A.8.22) the new parametrical solution

$$u^{1}(x_{0},\tau) = x_{0}^{3/2}\tau \left(\ln\tau - \frac{1}{2}\right)^{-1}, \quad \ln\tau = (x_{0}x_{1}\tau)^{2}.$$
 (A.8.24)

When $Q = \partial_{\tau}$, then formulae (A.8.18)-(A.8.20) result in

where $\tau = \tau(x_0, x_1)$ is to be determined from the condition

$${}^{1}_{u_{\tau}}(x_{0},\tau) + x_{1}[{}^{1}_{u}(x_{0},\tau)]^{3} = 0.$$
(A.8.26)

In the same way one can construct the explicit form of formulae (A.8.20) for Equation (A.8.11) for any symmetry operator from MIA of the heat equation (5.1.6)

Formulae (A.8.3) mean that Equation (A.8.11) is not invariant under the Galailei transformations while the heat equation (A.8.12) is, the transformation having the form

$$t' = t, \qquad x' = x + 2at, \qquad z' = \exp\{-a(x+at)\}z$$
 (A.8.27)

(a is an arbitrary constant). However, having used the nonlocal connection (A.8.18), (A.8.19) between the two equations one can write down corresponding Galilean transformations for Equation (A.8.11) and afterwards the formula of generating solutions, the latter having the form

$$\hat{u}(x_0, x_1) = \frac{\dot{u}(x_0, \tau)}{a x_1 \dot{u}(x_0, \tau) + x_1 / \tau},$$
(A.8.28)

where $\tau = \tau(x_0, x_1)$ is a function parameter to be determined from the equations

$$\tau_{1} = \left[a x_{1} \overset{1}{u}(x_{0},\tau) + x_{1}/\tau \right],$$

$$[\overset{1}{u}(x_{0},\tau)]^{2} \tau_{0} = \tau_{11}/\tau_{1}^{2} + 2a \overset{1}{u}(x_{0},\tau).$$
(A.8.29)

It will be noted that the first equation of (A.8.29) contains x_0 as a parameter (so this equation may be considered as an ODE with separate variables) and the second equation serves for sharpening dependence τ on x_0 . The following example shows the efficiency of formulae (A.8.28), (A.8.29). Choosing as u the simplest solution of Equation (A.8.11), $u(x_0, x_1) = 1$, we get, with the help of (A.8.28), (A.8.29), the new highly non-trivial solution

$$-\ln\left[(x_1u)^{-1} - a\right] + a\left[(x_1u)^{-1} - a\right]^{-1} = \ln x_1 + a^2 x_0.$$
 (A.8.30)

This is a parametric solution.

Next we will consider how to construct formula of superposition of solutions for Equation (A.8.11) analogous to that of linear superposition $\overset{3}{z}(t,x) = \overset{1}{z}(t,x) + \overset{2}{z}(t,x)$ of the heat equation (A.8.12) $(\overset{1}{z},\overset{2}{z}$ are given solutions, $\overset{3}{z}$ is the new one). Having used once more formulae (A.8.18), (A.8.19) one finds

$$\frac{1}{\overset{3}{u}(x_0, x_1)} = \frac{1}{\overset{1}{u}(x_0, \overset{1}{\tau})} + \frac{1}{\overset{2}{u}(x_0, \overset{2}{\tau})},$$
(A.8.31)

where $\dot{\tau} = \dot{\tau}(x_0, x_1), k = 1, 2$ are functional parameters to be determined from the conditions

Note, letting $u(x_0, x_1) = 1/w(x_0, x_1)$ we rewrite Equation (A.8.11) and formulae (A.8.31), (A.8.32) as

$$w_0 = w^2 w_{11};$$
 (A.8.33)

$$\overset{3}{w}(x_0, x_1) = \overset{1}{w}(x_0, \dot{\tau}) + \overset{2}{w}(x_0, \dot{\tau});$$
 (A.8.34)

$$\frac{d\tau}{\dot{u}(x_0, x_1)} = \frac{d\tau}{\dot{w}(x_0, \tau)}, \qquad \dot{\tau} + \dot{\tau} = x_1,$$

$$\dot{\tau}_0 = \dot{\tau}_{11} \left[\dot{w}(x_0, \dot{\tau}) / \dot{\tau}_1 \right]^2, \qquad k = 1, 2.$$
(A.8.35)

For example, starting from two simple solutions of Equation (A.8.33)

$${}^{1}w(x_0, x_1) = x_1, \qquad {}^{2}w(x_0, x_1) = 2x_1$$
 (A.8.36)

we find by means of (A.8.34), (A.8.35) another solution

$$\overset{3}{w}(x_0, x_1) = \pm e^{-2x_0} \Big[1 - 2x_1 e^{2x_0} \pm \sqrt{1 - 2x_1 e^{2x_0}} \Big],$$
 (A.8.37)

and one may compare (A.8.36) with (A.8.37).

Let us consider the equation

$$u_0 = \partial_1 \left(u^{-2/3} u_1 \right). \tag{A.8.38}$$

This equation is transformed through the use of (A.8.6)-(A.8.8) into the equation

$$z_t = \partial_x \left(z^{-4/3} z_x \right) \tag{A.8.39}$$

which is remarable by its local symmetry propertites (see Theorem A.8.1, case 4). Without going into details we list below Lie ansatze for Equation (A.8.39) and then transform some of them with the help of (A.8.6)–(A.8.8) into ansatze for Equation (A.8.38), acting in much the same way as when considering Equations (A.8.11), (A.8.12).

Ansatze for Equation (A.8.30):

1)
$$z = x^{-3}\varphi(\omega), \qquad \omega = t;$$

2) $z = x^{-3}\varphi(\omega), \qquad \omega = at + \frac{1}{x};$
3) $z = t^{3/4}x^{-3}\varphi(\omega), \qquad \omega = a\ln t + \frac{1}{x};$
4) $z = (x^2 + 1)^{-3/2}\varphi(\omega), \qquad \omega = t + \lambda \arctan x;$ (A.8.40)
5) $z = (x^2 - 1)^{-3/2}\varphi(\omega), \qquad \omega = t + \lambda \operatorname{arcth} x;$
6) $z = t^{3/4}(x^2 + 1)^{-3/2}\varphi(\omega), \qquad \omega = \ln t + \lambda \arctan x;$
7) $z = t^{3/4}(x^2 - 1)^{-3/2}\varphi(\omega), \qquad \omega = \ln t + \lambda \operatorname{arcth} x.$
Ansatze for Equation (A.8.38):
1) $u = [\varphi^1(x_0)x_1^2 + \varphi^2(x_0)]^{-3/2};$

2)
$$[x_1 + \varphi^1(x_0)] [\dot{\varphi}^2(x_0)]^{3/4} = -\tau \dot{\varphi}^3(\omega) + \varphi^3(\omega),$$

 $\omega = \varphi^2(x_0) + \tau, \qquad -\frac{\tau_1}{\tau} = u;$ (A.8.41)

3)
$$[x_1 + \varphi^1(x_0)] [\dot{\varphi}^2(x_0)]^{3/4} = \int [\dot{\varphi}^3(\tau)]^{3/2} \varphi^4(\omega) d\tau,$$

 $\omega = \varphi^2(x_0) + \varphi^3(\tau), \qquad \tau_1 = u.$

After substitution of ansatze 1)–3) (A.8.41) into Equation (A.8.38) we get, respectively

1) $\dot{\varphi}^{1} + 4(\varphi^{1})^{2} = 0,$ (A.8.42) $\dot{\varphi}^{2} - 2\varphi^{1}\varphi^{2} = 0;$

408

2) $\dot{\varphi}^{1} = \lambda_{1}(\dot{\varphi}^{2})^{1/4},$ $\ddot{\varphi}^{2} = \lambda_{2}(\dot{\varphi}^{2})^{2},$ $3(\ddot{\varphi}^{3})^{-1/3} + \lambda_{3}\dot{\varphi}^{3} + \frac{3}{4}\lambda_{2}\varphi^{3} - \lambda_{1} = 0;$ 3) $\dot{\varphi}^{1} = 0,$ $\ddot{\varphi}^{2} = \lambda_{2}(\dot{\varphi}^{2})^{2},$ $2\ddot{\varphi}^{3}\dot{\varphi}^{3} - 3(\ddot{\varphi}^{3})^{2} = 2\lambda_{1}(\dot{\varphi}^{3})^{4},$ $(\varphi^{4})^{-4/3}\ddot{\varphi}^{4} - \frac{4}{3}(\varphi^{4})^{-7/3}(\dot{\varphi}^{4})^{2} + 3\lambda_{1}(\varphi^{4})^{-1/3} + \frac{3}{4}\lambda_{2}\varphi^{4} - \dot{\varphi}^{4} = 0,$

where $\lambda_1, \lambda_2, \lambda_3$ are arbitrary constants.

Having integrated system 2) from (A.8.42) under $\lambda_2 = 0$, we obtain the following parametric solution of Equation (A.8.38)

$$u^{1/3} = \frac{-c\left(\frac{5}{4}c_1^3 x_1 + c_2 x_0\right)}{\tau(\tau - 4c_3 x_0)},$$

(\tau + c_3 x_0)(\tau - 4c_3 x_0)^4 = \left(\frac{5}{4}c_1^3 x_1 + c_2 x_0\right)^4, \text{(A.8.43)}

where c_1, c_2, c_3 are constants of integration.

In conclusion we consider such case of transformations (A.8.6)-(A.8.8) when they are symmetry transformations of Equation (A.8.1). One sees from (A.8.10) that it meets the case when

$$z^{-2}F(z^{-1}) = F(z).$$
 (A.8.44)

This functional equation has solution

$$F(z) = z^{-1} f(\ln z), \qquad (A.8.45)$$

where f is an arbitrary differentiable even function. So, formulae (A.8.6)–(A.8.8) determine nonlocal symmetry transformations of the equation

$$u_0 = \partial_1 \big[f(\ln u) u_1/u \big], \qquad (f(\alpha) = f(-\alpha)). \tag{A.8.46}$$

Using this fact we construct formula of generating solutions of Equation (A.8.46). It looks like

$$\hat{u}(x_0, x_1) = \frac{1}{\hat{u}(x_0, \tau)},$$
(A.8.47)

where the functional parameter $\tau = \tau(x_0, x_1)$ is to be determined from the equations

$$\tau_1 = \frac{1}{u(x_0, \tau)}, \qquad \tau_0 = \frac{f(\ln \tau_1)\tau_{11}}{\tau_1}.$$
 (A.8.48)

For example, starting from the solution

$$u'(x_0, x_1) = \frac{x_0}{1 + \cos x_1},$$
 (A.8.49)

of equation

$$u_0 = \partial_1 \left(\frac{u_1}{u}\right) \tag{A.8.50}$$

we find, with the help of (A.8.47), (A.8.48), the new solution of Equation (A.8.50)

$$\hat{u}(x_0, x_1) = \frac{2x_0}{x_0^2 + x_1^2},$$
(A.8.51)

It will be noted that solutions (A.8.49) and (A.8.51) essentially differ from each other on their properties, such as boundedness, periodicity, and analyticity behavior at infinity, at zero, and so on. It is a distinguishing feature of nonlocal transformations that they may essentially change properties of solutions unlike those of Lie's solutions.

References

- 1. M.J.Ablowitz and H.Segur, "Solitons and the Inverse Scattering Transform," SIAM, Philadelphia (1981).
- A.I.Akhiezer and V.B.Berestezkii, Quantum Electrodynamics, Nauka, Moscow (1969).
- 3. N.I.Achiezer, Elements of Theory of Elliptic Functions, Nauka, Moscow (1970).
- A.Actor, "Classical solutions of SU(2) Yang-Mills theories," Rev. Mod. Phys., 51, No.3, 461-525 (1979).
- K.G.Akdeniz, "On classical solutions of Gürsey's conformal invariant spinor model," Lett. Nouvo Cim., 33, No.2, 40-44 (1982).
- K.G.Akdeniz and A.Smailagic, "Classical solutions for fermionic models," Nuovo Cim. 51A, No. 3, 345-357 (1979).
- W.F.Ames, Nonlinear Partial Differential Equations in Engineering, N.Y.: Academic Press, vol.1 (1965), vol.2 (1972).
- 8. W.F.Ames, R.J.Lohner, and E.Adams, "Group properties of $u_{tt} = (f(u)u_x)_x$," in: Nonlinear Phenomena in Mathematical Sciences, Ed. V.Lakshmikanthan, N.Y.: Academic Press, 1982, p.1–6.
- R.L.Anderson and N.H.Ibrgimov, "Lie-Bäcklund Transformations in Applications," SIAM, Philadelphia (1979).
- 10. B.D.Annin, N.O.Bytev and S.I.Senashov, Group Properties of Elasticity and Plasticity Equations, Nauka, Novosibirsk (1985).
- 11. V.G.Bagrov, D.M.Gitman, I.M.Ternov, V.R.Chalilov, and VN.Shapovalov, Exact Solutions of Relativistic Wave Equations, Nauka, Novosibirsk (1982).
- L.F.Barannik and A.F.Barannik, "Subalgebras of the generalized Galilean algebra," in: Group-Theoretic Investigations of Equations of Mathematical Physics, Inst. Matematiki AN Ukr SSR, Kiev, 1985, p. 39-43.
- 13. L.F.Barannik, V.I.Lagno, and W.I.Fushchich, "Subalgebras of the generalized Poincare algebra $\widehat{AP}(1,n)$," Preprint Inst. Matematiki AN Ukr SSR,

No. 85.89, Kiev (1985).

- 14. L.F.Barannik and W.I.Fushchich, "Subalgebras of the extended Poincare algebra $A\tilde{P}(1, n)$," Preprint Inst. Matematiki AN Ukr SSR, No. 85.90, Kiev (1985).
- 15. L.F.Barannik and W.I.Fushchich, "Invariants of the generalized Poincare group P(1,n)," Preprint Inst. Matematiki AN Ukr SSR, No. 86.86, Kiev (1986).
- L.F.Barannik and W.I.Fushchich, "On subalgebras of the Lie algebra of the extended Poincare group P(1,n)," J. Math. Phys., 28 No. 7, 1445-1458 (1987).
- B.M.Barbashov and N.A.Chernikov, "Solution and quantization of two-dimensional model of the Born-Infeld type," Zhurn. Eksperim. i Teoret. Fiziki, 60, No.5, 1296-1308 (1966).
- 18. B.M.Barbashov and V.V.Nesterenko, Relativistic String Model in Hadron Physics, Energoatomizdat, Moscow (1987).
- A.O.Barut and X.U.Bo-Wei, "Derivation and uniqueness of vacuum solutions of conformally invariant coupled nonlinear field equations," Phys. Lett. B, 102 No.1, 37-39 (1981).
- 20. A.O.Barut and X.U.Bo-Wei, "New exact solutions of coupled nonlinear field equations," Physica D, 6 No.2, 137-139 (1982).
- 21. A.O.Barut and R.Raczka, Theory of Group Representations and Applications, Vols. 1,2, Moscow (1980).
- 22. F.Bayen and M.Flato, "Remarks on conformal space," J. Math. Phys., 17, No.7, 1112-1114.
- 23. O.I.Bazula and V.P.Gusinin, "Plane-wave solutions of SU(2) Yang-Mills theory," Preprint Inst. Teoret. Fiziki AN Ukr SSR, No. 80.89P, Kiev (1980).
- 24. H.Bateman and A.Erdèlyi, Higher Transendental Functions, Vol. 3, McGraw-Hill, New York (1955).
- 25. H.Bateman, "The transformations of electrodynamical equations," Proc. London Math. Soc., 8, 223-264 (1909).
- 26. P.L.Bhatnagar, Nonlinear Waves in One-dimensional Dispersive Systems, Clarendon Press, Oxford (1979).
- 27. G.W.Bluman and I.D.Cole, Similarity Methods for Differential Equations, Springer, Berlin (1974).
- G.Bluman and S.Kumei, "On invariance properties of the wave equation," J. Math. Phys., 28, No. 2, 307-318 (1987).
- 29. H.Birkhoff, Hydrodynamics, Inostr. Literatura, Moscow (1963).
- 30. A.W.Bizadze, Some Classes of Partial Differential Equations, Nauka, Moscow (1981).
- 31. N.N.Bogolubov, A.A.Logunov, A.I.Oksak, and I.T.Todorov, General Principles of Quantum Field Theory, Nauka, Moscow (1987).
- 32. N.N.Bogolubov and Yu.A.Mitropolsky, Asymptotic Methods In Theory of Nonlinear Oscillations, Nauka, Moscow (1974).
- P.A.Borroni and W.M.Shtelen, "The maximally extensive local group of invariance of Dirac-Lorentz's equation," Lett. Nuovo Cim., 28, No. 5, 169-170 (1980).
- AJ.Bracken, "A comment on the conformal invariance of zero-mass Klein-Gordon equation," Ibid., 2, No. 11, 574-576 (1971).
- 35. L.de Broglie, Thèorie Generale des Particules à Spin Méthode de Fusion, Paris (1954).
- 36. C.P.Boyer and M.N.Penafiel, "Conformal symmetry of the Hamilton-Jacobi equation and quantization," Nuovo Cim., **31B**, No. 2, 195-210 (1976).
- J.Challis, "On the velocity of sound," Philos. Magaz., 32, No. 3, 494-499 (1848).
- 38. N.G.Chebotarov, The Theory of Lie Groups, Gostekhizdat, Moscow (1949).
- 39. Yu.A.Chirkunov, "The group property of Lame equations," in: Dynamics of Continuous Medium, No. 14, Novosibirsk, 1983, p. 138-140.
- 40. J.D.Cole, "On a quasi-linear parabolic equation occurring in aerodynamics," Quart. Appl. Math., No. 9, 225-236 (1951).
- 41. C.B.Coliins, "Complex potential equations. 1. A technique for solution," Math. Proc. Cambr. Phil. Soc., 80, No. 1, 165-187 (1976).
- 42. J.P.Crawford, "On the algebra of Dirac bispinor densities: factorization and inversion theorem," J. Math. Phys., 26, No. 7, 1439-1441 (1985).
- 43. E.Cunningham, "The principle of relativity in electrodynamics and an extension thereof," Proc. Lond. Math. Soc., 8, No. 1, 77-97 (1909).
- 44. Yu.A.Danilov, "On nonlinear Dirac equations admitting conformal group," Teoret. i Mat. Fizika, 2, No. 3, 40-45 (1970).
- 45. E.B.Dynkin, "On representation of series $\ln (e^x e^y)$ for noncommutative x and y via commutators," Mat. Sbornik, 25, No. 2, 155-162 (1949).
- 46. B.A.Dubrovin, S.P.Novikov, and A.T.Fomenko, Modern Geometry: Methods and Applications, Nauka, Moscow (1986).
- 47. S.Earnshaw, "On the mathematical theory of sound," Transact. of the Soc. of London, 150, No. 1, 133-148 (1860).
- 48. E.Egervary, "On a generalization of Lagrange solution of the three-body problem," Dokl. AN SSR, 55, No. 9, 805-807 (1947).
- I.A.Egorchenko, "Poincare invariant quasilinear wave equations for complexvalued functions," Preprint Inst. Matematiki AN Ukr SSR, No. 87.3, Kiev (1987).
- 50. L.P.Eisenhart, Groups of Continuous Transformatins, Inostr. Literatura, Moscow (1947).
- 51. V.M.Fedorchuk, "Continuous subgroups of the inhomogeneous de Sitter group P(1,4)," Preprint Inst. Matematiki AN Ukr SSR, No. 78.18, Kiev (1978).
- 52. M.Flato, J.Simon, and D.Sternheimer, "Conformal covariance of field equations," Ann. Phys. N.Y., **61**, No. 1, 78-97 (1970).
- V.A.Fock, "Zur theorie des Wassrstoffatoms," Fortschr. Phys., 98, 145-147 (1935).

- 54. A.R.Forsyth, Theory of Differential Equations, Vols. 5,6, Dover Publications, New York (1959).
- 55. W.I.Fushchich, "On additional invariance of the relativistic equations of motion," Preprint Inst. Theoret. Phys. AN Ukr SSR, No. 70.32E, Kiev (1970).
- W.I.Fushchich, "Representations of the complete inhomogeneous de Sitter group and equations in a five-dimensional approach," Teoret. i Mat Fizika, 4, No. 3, 360-382 (1970).
- 57. W.I.Fushchich, "On additional invariance of relativistic equations of motion," Ibid., 7 No. 11, 3-12 (1971).
- 58. W.I.Fushchich, "On additional invariance of the Dirac and Maxwell equations," Lett. Nuovo Cim. 11, No. 10, 508-512 (1974).
- 59. W.I.Fushchich, "On additional invariance of Klein-Gordon-Fock equation," Dokl. AN SSR, 230, No. 3, 570-572 (1976).
- W.I.Fushchich, "Group properties of quantum mechanics equations," in: Problems Of Asymptotic Theory of Nonlinear Oscillation, Nauk. Dumka, Kiev, 1977, p. 75-87.
- W.I.Fushchich, "On the new method of studying group properties of equations of mathematical physics," Dokl. AN SSR, 246. No. 4, 946-850 (1979).
- 62. W.I.Fushchich, "On a method of studying group properties of integro-differential equations," Ukr. Mat. Zhurn., **33**, No. 6, 834-838 (1981).
- W.I.Fushchich, "Symmetry in problems of mathematical physics," on: Algebraic-Theoretic Studies in Mathematical Physics, Inst. Matematiki AN UKr SSR, Kiev, 1981, p. 6–44.
- 64. W.I.Fushchich, "On the new method of investigation of group properties of systems of partial differential equations," in: Group-Theoretic Methods in Mathematical Physics, Inst. Matematiki AN Ukr SSR, Kiev 1978, p.5-44.
- 65. W.I.Fushchich, "On Poincare- and Galilei-invariant nonlinear equations and methods of their solution," in: Group-Theoretic Investigations of Equations of Mathematical Physics, Inst. Matematiki AN Ukr SSR, Kiev 1985, p.4–19.
- 66. W.I.Fushchich, "On symmetry and partial solutions of some many-dimensional equations of mathematical physics," in: Algebraic-Theoretical Studies in Mathematical Physics Problems, Inst. Matematiki AN Ukr SSR, Kiev, 1983, p.4-23.
- 67. W.I.Fushchich, "On symmetry and exact solutions of the many-dimensional nonlinear wave equations," Ukr. Mat. Zhurn., **39**, No. 1, 116-123 (1987).
- W.I.Fushchich, A.F.Barannik, and L.F.Barannik "Continuous subgroups of the generalized Galilei group. 1," Preprint Inst. Matematiki AN Ukr SSR, No. 85.19, Kiev (1985).
- 69. W.I.Fushchich, A.F.Barannik, and L.F.Barannik, "Continuous subgroups of the generalized Euclid group," Ukr. Mat. Zhurn., 38, No. 1, 67-72 (1986).
- W.I.Fushchich and R.M.Cherniha, "On exact solutions of two Schrödingertype nonlinear equations," Preprint Inst. Matematiki AN Ukr SSR, No. 86.85 (1986).

- W.I.Fushchich and R.M.Cherniha, "The Galilean relativistic principle and nonlinear partial differential equations," J. Phys. A: Math. and Gen., 18, 3491-3503 (1985).
- W.I.Fushchich and S.S.Moskaliuk, "On some exact solutions of the nonlinear Schrödinger equation in three spatial dimensions," Lett. Nuovo Cim., 31, No. 16, 371-376 (1981).
- 73. W.I.Fushchich and V.V.Nakonechny, "Algebraic-theoretic analysis of the Lame equation," Ukr, Mat. Zhurn., No. 2, 267-273 (1980).
- W.I.Fushchich and A.G.Nikitin, "On the new invariance groups of the Dirac and Kemmer-Duffin-Petiau equations," Lett. Nuovo Cim., 19, No. 9, 347-352 (1977).
- 75. W.I.Fushchich and A.G.Nikitin, "On invariance group of quasirelativistic equation," Dokl. AN SSR, 238, No. 1, 46-49 (1978).
- W.I.Fushchich and A.G.Nikitin, "Poincare-invariant differential equations of motion of arbitrary spin particles," Fizika Elementar. Chastit i Atom Yadra, 9, No. 3, 501-553 (1978).
- W.I.Fushchich and A.G.Nikitin, "On the new invariance algebras of relativistic equations for massless particles," J. Phys. A: Math, and Gen., 12 No. 6, 747-757 (1979).
- W.I.Fushchich and A.G.Nikitin, "Nonrelativistic equations of motion for arbitrary spin particles," Fizika Elemntar. Chastits i Atom Yardra, 12, No. 5, 1157-1219 (1981).
- 79. W.I.Fushchich and A.G.Nikitin, "On new and old symmetries of Maxwell and Dirac equations," Ibid., 14, No. 1, 5-57 (1983).
- 80. W.I.Fushchich and A.G.Nikitin, Symmetries of Maxwell's Equations, Nauk. Dumka, Kiev (1983).
- W.I.Fushchich and A.G.Nikitin, "On new invariance algebras and superalgebras of relativistic wave equations," J. Phys. A: Math. and Gen., 20, No. 2, 537-549 (1987).
- 82. W.I.Fushchich and A.G.Nikitin, Symmetries of Maxwell's Equations, D.Reidel (1987).
- W.I.Fushchich and Yu.N.Seheda, "Some exact solutions of the multi-dimensional sine-Gordon equation," Lett. Nuovo Cim., 41, No. 14, 462-464 (1984).
- W.I.Fushchich and N.A.Selehman, "Integro-differential equations invariant under group of Galilei, Poincare, Schrödinger, and conformal group," Dokl. AN Ukr SSR, A, No. 5, 21-24 (1983).
- W.I.Fushchich and N.I.Serov, "On exact solutions of Born-Infeld equation," Dokl. AN SSSR, 263, No. 3, 582-586 (1981).
- 86. W.I.Fushchich and N.I.Serov, "Symmetry and some exact solutions of multidimensional Monge-Ampere equation," Ibid., 273, No. 3, 543-546 (1983).
- 87. W.I.Fushchich and N.I.Serov, "On some exact solutions of multi-dimensional Euler-Lagrange equation," Ibid., **278**, No. 4, 847–851 (1984).

- W.I.Fushchich and N.I.Serov, "The symmetry and exact solutions of the nonlinear multi-dimensional Liouville, d'Alembert and eikonal equations," J. Phys. A: MAth. and Gen., 16, 3645-3658 (1983).
- W.I.Fushchich, N.I.Serov, and W.M.Shtelen, "Some exact solutions of multidimensional nonlinear d'Alembert, Liouville, eikonal, and Dirac equations," in: Group-theoretical Methods in Physics, Harwood Academic Publ. London, 1984, p. 489-496.
- W.I.Fushchich and M.M.Serova, "On the maximal invariance group and general solution of one-dimensional gas dynamics equations," Dokl. AN SSSR, 268, No. 5, 1102-1104 (1983).
- W.I.Fushchich and M.M.Serova, "On exact solutions of some nonlinear differential equations invariant under Galilei and Euclid groups," in: Algebraic-Theoretic Studies in Mathematical Physics Problem, Inst. Matematiki AN Ukr SSR, Kiev, 1983, p. 24-54.
- W.I.Fushchich and W.M.Shtelen, "The symmetry and some exact solutions of the relativistic eikonal equation," Lett. Nuovo Cim., 34, No. 16, 498-502 (1982).
- 93. W.I.Fushchich and W.M.Shtelen, "On invariant solutions of the nonlinear Dirac equation," Dokl. AN SSSR, 269, No. 1, 88-92 (1983).
- W.I.Fushchich and W.M.Shtelen, "On linear and nonlinear systems of differential equations invariant under the Schrödinger group," Teoret. i Mat. Fizika, 56, No. 3, 387-394 (1983).
- 95. W.I.Fushchich and W.M.Shtelen, "On some exact solutions of the nonlinear Dirac equation," J. Phys. A: Math. and Gen., 16, No. 2, 271-277 (1983).
- W.I.Fushchich and W.M.Shtelen, "Conformal symmetry and new exact solutions of SU(2) Yang-Mills theory," Lett. Nuovo Cim., 38 No.2, 37-40 (1983).
- W.I.Fushchich and W.M.Shtelen, "On some exact solutions of the nonlinear equations of quantum electrodynamics," Phys. Lett. B, 128, No. 3/4, 215-217 (1985).
- W.I.Fushchich and W.M.Shtelen, "Conformal symmetry and exact solutions of nonlinear field equations," Ukr. Fiz. Zhurn., 30, No. 5, 787-790 (1985).
- 99. W.I.Fushchich and W.M.Shtelen, "On nonlocal transformations," Lett. Nuovo Cim., 44, No. 1, 40-42 (1985).
- 100. W.I.Fushchich and W.M.Shtelen, "On reduction and exact solutions of the nonlinear Dirac equation," Teoret. i Mat. Fizika, 72, No. 1, 35-44 (1987).
- 101. W.I.Fushchich and W.M.Shtelen, and R.Z.Zhdanov, "Conformally invariant generalization of the Dirac-Heiseberg equation and its exact solution," in: Group-Theoretic Methods in Physics (3-d International Seminar, Yurmala), Vol. 1, Nauka, Moscow, 1986, p. 497-501.
- 102. W.I.Fushchich and W.M.Shtelen, and R.Z.Zhdanov, "On the new conformally invariant equations for spinor fields and their exact solutions," Phys. lett. B, 159, No. 2-3, 189-191 (198).

- 103. W.I.Fushchich and S.L.Slavutsky, "On symmetry of some equations of motion of ideal fluid," in: Investigations on Theory of Complex Variable Functions with Applications to Mechanics of Continuous Medium, Nauk. Dumka, Kiev, 1986, p. 161–165.
- 104. W.I.Fushchich and S.L.Slavutsky, "On nonlinear Galilei-invariant generalization of Lame equation," Dokl. AN SSSR, 287, No. 2, 320-323 (1986).
- 105. W.I.Fushchich and V.A.Tychinin, "On linearization of some nonlinear equations by virtue of nonlocal transformations," Preprint. Inst. Matematiki AN Ukr SSR, No. 82.33, Kiev (1982).
- 106. W.I.Fushchich and V.A.Tychinin, and R.Z.Zhdanov, "Nonlocal linearization and exact solutions of some equations of Monge-Ampere and Dirac equation," Preprint. Inst. Matematiki AN Ukr SSR, No. 85.88, Kiev (1985).
- 107. W.I.Fushchich and I.M.Tsifra, "On symmetry of nonlinear equations of electrodynamics," Teoret. i Mat. Fizika, 64, No. 1, 41-50 (1985).
- 108. W.I.Fushchich and I.M.Tsifra, "On reduction and solutions of nonlinear wave equations with broken symmetry," J. Phys. A: Math. and Gen., 20, No. 2, 45-47 (1987).
- 109. W.I.Fushchich and R.Z.Zhdanov, "Exact solutions of nonlinear system of partial differential equations for vector and spinor fields," in: Group-Theoretic Investigations of Equations of Mathematical Physics, Inst. Matematiki AN Ukr SSR, Kiev, 1985 p. 20-30.
- 110. L.A.Gazarkhi, "On new (with respect to velocity) integral of the generalization three-body problem," Ukr. Mat. Zhurn., 8, No. 1, 5-11 (1956).
- 111. I.M.Gel'fand, R.A.Minlos, and Z.Ya.Shapiro, Representations of the Rotation and Lorentz Groups, Fizmatigiz, Moscow (1958).
- 112. L.E.Gendenshtein and I.V.Krive, "Supersymmetry in quantum mechanics," Uspekhi Fiz. Nauk, 146, No. 4, 553-590 (1985).
- 113. V.G.Golubev and E.A.Grebennikov, The Three-Body Problem in Celestial Mechanics, Izd. MGU, Moscow (1985).
- 114. A.M.Grundland, J.Harnad, and P.Winternitz, "Symmetry reduction for nonlinear relativistically invariant equations," J.Math.Phys., 25, No. 4, 791–806 (1984).
- 115. F.Gürsey, "On a conform-invariant spinor wave equation," Nuovo Cim., 3, No. 5, 988-1006 (1956).
- B.K.Harrison and F.B.Estrabrook, "Geometric approach to invariance groups and solutions of partial differential system," J. Math. Phys., 12, No. 4, 653-666 (1971).
- 117. F.Hausdorff, "Die simbolische exponentialformel in der gruppentheorie," Ber. Verh.Sächs.Wiss. Leipzig, Math.-Phys., 58, K1, 19-48 (1906).
- 118. W.Heisenberg, Introduction to the Unified Field Theory of Elementary Particles, Intersciences Publishers, London (1966).
- 119. W.Heisenberg, "On quantum theory of non-renormalized wave equations," Zs. Nuturforsch, 9a, 292-303 (1954).

- 120. W.J.Herley, "Nonrelativistic quantum mechanics for particles with arbitrary spin," Phys, Rev., 3, No.10, 2339–2347 (1971).
- 121. E.Hopf, "The partial differential equation $u_t + uu_x = \mu u_{xx}$," Comm. Pure Appl. Math., 3, 201–230 (1950).
- 122. H.Hugoniot, "Sur la propagation du mouvement dens lez gaz parfaits," J. de l'ecole politechnique, 58, 1-125 (1889).
- 123. N.Ch.Ibragimov, Lie Groups in Some Problems of Mathematical Physics, Izd. NGU, Novosibirsk (1972).
- 124. N.Ch.Ibragimov, Group Transformations in Mathematical Physics, Nauka, Moscow (1983).
- 125. V.G.Imshenenezky, Integration of First- and Second-Order Partial Differential Equations, Izd. Mosk. Mat. Ob., Moscow (1916).
- 126. A.R.Its, "Reversion of hyperelliptic integrals and integration of nonlinear differential equations," Vestnik Leningrad. Universiteta, No. 7, 39-46 (1976).
- 127. G.P.Jorjadze, A.K.Pogrebkov, and M.C.Polivanov, "Singular solutions of the equation $\Box u + \frac{1}{2}m^2 \exp u = 0$ and dynamics of singularities," Teoret. i Mat. Fizika, 40, No. 3, 221-234 (1979).
- 128. G.P.Jorjadze, A.K.Pogrebkov, and M.C.Polivanov, "Liouville field theory: IST and Poisson bracket structure," J. Phys. A: Math. and Gen., 19, No. 1, 121-139 (1986).
- 129. V.G.Kadyshevsky, "A new approach to the theory of electromagnetic interactins," Fizika Elemenar. Chastits i Atom. Yadra, 11, No. 1, 5-36 (1980).
- 130. E.Kamke. Hand-Book on Ordinary Differential Equations, Nauka, Moscow (1976).
- 131. E.Kartan, Integral Invariants, Gostenkhizdat, Moscow (1940).
- 132. F.Kortel, "On some solutions of Gürsey's conformal-invariant spinor wave equation," Nuovo Cim., 4, No. 12, 211-215 (1956).
- 133. J.Kosmann-Schwartzbach, "Generalized symmetries of nonlinear differential equations," Lett. Math. Phys., No. 3, 395-404 (1979).
- 134. V.A.Kostelecky and M.M.Nieto, "Baker-Campbell-Hausdorff relations for supergroups," J. Phys., 27., No. 5, 1419-1429 (1986).
- 135. V.G.Kostenko, Integration of Some Partial Differential Equations with the Help of Group Method, Izdat. Lvovskoho Universiteta, Lvov (1959).
- 136. S.Kumei and G.W.Bluman, "When nonlinear differential equations are equivalent to linear differential equations," SIAM J. Appl. Math., 42, No. 5, 1157-1173 (1982).
- 137. R.Courant, Partial Differential Equations, Mir, Moscow (1964).
- D.F.Kurdhelaidze. "Periodic solutions of nonlinear generalization of the Dirac equation," Zhurn. Eksperim. i Teoret. Fiziki, 34, No. 6, 1587-1592 (1958).
- M.Lakshmanan and P.Kaliappan, "Lie transformations, nonlinear evolution equations, and Painleve forms," J. Math. Phys., 24, No. 4, 795-806 (1983).
- 140. L.D.Landau and E.M.Lifshits, Hydrodynamics, Nauka, Moscow (1986).
- 141. L.D.Landau and E.M.Lifshits, Theory of Elasticity, Nauka, Moscow (1987).

References

- 142. L.D.Landau and E.M.Lifshits, *Electrodynamics of Continuous Medium*, Nauka, Moscow (1982).
- 143. A.N.Leznov and M.V.Savel'ev, Group Methods in Integrating Nonlinear Dynamical Systems, Nauka, Moscow (1985).
- 144. J.M.Levi-Leblond, "Nonrelativistic particles and wave equations," Comm. Math. Phys., 16, No. 4, 884-893 (1975).
- 145. G.Mack and A.Salam, "Finite-component field representation of conformal group," Ann. Phys., 53, No. 1, 174-202 (1969).
- 146. W.Magnus, "On the exponential solution of differential equations for a linear operator," Comm. Pure Appl. Math., 7, 649-673 (1954).
- 147. I.A.Malkin and V.I.Man'ko, Dynamical Symmetries and Coherent States of Quantum Systems, Nauka, Moscow (1979).
- D.H.Mayer, "Vector and tensor fields on conformal space," J. Math. Phys. 16, No. 4, 884-893 (1975).
- 149. P.du T.Merwe, "Space-time symmetries and nonlinear field theory," Nuovo Cim., 60A, No. 4, 247-261 (1980).
- 150. P.du T.Merwe, "Classical solutions of the conformally invariant Gürsey equation," Phys. Lett. B, **106**, No. 6, 485-486 (1981).
- 151. A.Messiah, Quantum Mechanics, Vols 1,2, Mir, Moscow (1979).
- 152. A.V.Michailov, A.B.Shabat, and R.I.Yamilov, "Symmetry approach to classification of nonlinear equations, complete lists of integrable systems," Uspechi Mat. Nauk, 42, No. 4, 3-53 (1987).
- 153. W.Miller, Jr., Symmetry and Separation of Variables, Addison-Wesley Publ. Company (1977).
- 154. Yu.A.Mitropolsky, N.N.Bogolubov, Jr., A.K.Prikarpatsky, and V.G.Samoilenko, Integrable Dynamical Systems: Spectral and Differential-Geometry Aspects, Nauk. Dumka, Kiev (1987).
- 155. Yu.A.Mitropolsky, I.V.Revenko, and W.I.Fushchich, "On symmetry, integral of motion, and some exact solutions of the spatial three-body problem," Dokl. AN SSSR, **280**, No. 4, 799-904 (1985).
- 156. Nonlinear Field Theory, Collection of Papers, ed. D.D.Ivanenko, Izdat. Inostr. Literat., Moscow (1959).
- 157. U.Niederer, "The maximal kinematical invariance group of Schrödinger equation," Helv. Phys. Acta., 45, No. 5, 802-810 (1972).
- 158. T.Nishitani and M.Tajiri, "On similarity solutions of the Boussinesq equation," Phys. Lett. A, 89, No. 4, 379-380 (1982).
- 159. P.Olver, Applications of Lie Groups to Differential Equations, Springer (1986).
- 160. P.Olver and Ph.Rosenau, "The construction of special solutions to partial differential equations," Phys. Lett. A, 114, No. 3, 107-112 (1986).
- L.V.Ovsyannikov, Group Analysis of Differential Equations, Nauka, Moscow (1978).
- 162. L.V.Ovsyannikov, Lectures on Gas Dynamics, Nauka, Moscow (1981).
- 163. J.Patera, P.Winternitz, and H.Zassenhaus, "Continuous subgroups of the fundamental groups of physics. 1. General method and the Poincare group,"

J. Math. Phys., 16, No. 8, 1597-1624 (1975).

- A.M.Perelomov, Generalized Coherent States and Applications, Nauka, Moscow (1987).
- A.V.Pogorelov, The Multi-Dimensional Minkowsky Problem, Nauka, Moscow (1971).
- 166. L.S.Polak, Variational Principles of Mechanics, Fizmatgiz, Moscow (1960).
- 167. E.M.Polishchuk, Vito Volterra, Nauka, Leningrad (1977).
- 168. E.M.Polishchuk, Sophus Lie, Nauka, Leningrad (1983).
- 169. P.Raczka, Jr., "On the class of simple solutions of the SU₂ Yang-Mills equations," Nuovo Cim., **72A**, No. 3, 289-302 (1982).
- 170. R.Rajaraman, An Introduction to Soliton and Instantons in quantum Field Theory, North-Holland Publ. Comp., Amsterdam (1982).
- 171. B.Riman, "On spreading of air wave final amplitude," in: B.Riman, Collection of Works, Gostenkhizdat, Moscow (1948).
- 172. G.Rosen, "Solutions of certain nonlinear wave equations," J. Math. Phys., 45, No. 3-4.
- 173. G.Rosen, "Conformal transformation matrix for fields," Ann. Phys. (New York), 77, No. 2, 452-453 (1973).
- 174. G.Rosen and G.Ullrich, "Invariance group of the equation $\vec{u} + (\vec{u} \cdot \vec{\nabla})\vec{u} = 0$," SIAM J. Appl. Math., 24, 286–288 (1973).
- 175. B.L.Rozdestvensky and N.N.Yanenko, Systems of quasilinear Equations and their Applications to Gas dynamics, Nauka, Moscow (1968).
- 176. N.I.Serov, "Conformal symmetry of nonlinear wave equations," in: Algebraic-Theoretic Studies in Mathematical Physics, Inst. Matematiki AN Ukr SSR, Kiev, 1981, p. 59-63.
- 177. N.I.Serov, "Nonlinear Lame and Weyl wave equations," Ibid., p. 49-44.
- 178. M.M.Serova, "Exact solutions of a second-order nonlinear partial differential equation," Ibid., p.29-34.
- 179. M.M.Serova, "On exact solutions of the Darboux equation," Ibid., 42-44.
- 180. M.M.Serova, "On exact solutions of the Boussinesq equation," in: Algebraic-Theoretic Studies in Mathematical Physics Problems, Inst. Matematiki AN Ukr SSR, Kiev, 1983, p.55-58.
- 181. M.M.Serova, "On nonlinear heat equations invariant under the Galilei group," in: Group-Theoretic Investigations of Equations of Mathematical Physics, Inst. Matematiki AN Ukr SSR, Kiev, 1985, p. 119-123.
- 182. F.Schwarz, "Symmetry of SU(2) invariant Yang-Mills Theories," Lett. Math. Phys., 6, No. 5, 355-359 (1982).
- 183. W.M.Shtelen, "Group analysis of nonlinear systems of differential equations connected with the Schrödinger equation," Ukr Fiz. Zhurn., 26, No. 2, 323-326 (1981).
- 184. W.M.Shtelen, "Group analysis of a parabolic system of differential equations," in: Algebraic-Theoretic Studies in Mathematical Physics, Inst Matematiki AN Ukr SSR, Kiev, 1981 p. 64-67.

References

- M.Shtelen, "On a system of differential equations invariant under the Schrödinger group," Ibid., p. 104–107.
- 186. W.M.Shtelen, "Some exact solutions of coupling nonlinear equations of wave mechanics," in: Mathematical Problems of Continuous Mechanics and Thermodynamics, Inst. Matematiki AN Ukr SSR, Kiev, 1982, p.100-107.
- 187. W.M.Shtelen, "Contact transformations of the relativistic Hamilton-Jacobi equation," in: Algebraic-Theoretic Studies in Mathematical Physics Problems, Inst. Matematiki AN Ukr SSR, Kiev, 1983, p.62-65.
- 188. W.M.Shtelen, "The formula of generating conformally-invariant solutions for field equations of arbitrary spin," in: Group-Theoretic Investigations of Equations of Mathematical Physics, Inst. Matematiki AN Ukr SSR, Kiev, 1985, p. 60-66.
- 189. W.M.Shtelen, "Non-Lie symmetry and nonlocal transformations," Preprint. Inst. Matematiki AN Ukr SSR, No. 87.6, Kiev (1987).
- 190. M.A.Shubin, Pseudodifferential Operators and Spectral Theory, Nauka, Moscow (1978).
- 191. M.W.Shulga, "On exact and approximate solutions of a nonlinear wave equation," in: Methods of Nonlinear Mechanics and their Applications, Inst Matematiki AN Ukr SSR, Kiev, 1982, p. 149-155.
- 192. M.W.Shulga, "Symmetry and some partial solutions of d'Alembert equations with nonlinear condition," in: Group-Theoretic Investigations of Equations of Mathematical Physics, Inst. Matematiki AN Ukr SSR, Kiev, 1985, p. 36-38.
- 193. Yu.D.Sokolov, "On trajectories of general collision of rectilinear three material points interacting via forces depending on mutual distances," Dokl. AN SSSR, 33, No. 2, 112-115 (1941).
- 194. Yu.D.Sokolov, "On the case of integrability in the rectilinear three-body problem," Ibid., 46, No. 3, 99-109 (1945).
- 195. Yu.D.Sokolov, "On some partial solutions of the Boussinesq equation," Ukr Mat. Zhurn., 8, No. 1, 55-58 (1956).
- 196. K.P.Stanyukovitch, Unsettled Motion of Continuous Medium, Gostekhizdat, Moscow (1955).
- 197. E.E.Stokes, "On a difficulty in the theory of sound," Phil. Mag., 33, No. 3, 349-356 (1848).
- 198. L.A.Tachtadjan and L.D.Faddeev, Hamilton Approach in Theory of Solitons, Nauka, Moscow (1986).
- 199. D.R.Truax, "On the theory of one-parameter subgroups of supergroups," J. Math. Phys., 27, No. 1, 354-364 (1986).
- 200. E.Vessiot, "Sur l'integration des systéms differentiel, qui admettent des groups continus de transformations," Acta Math. Zeitschr., 28, 307–349 (1904).
- V.P.Vizgin, "From the history of conformal symmetry," in: Historic-Mathematic Investigations, Nauka, Moscow, 1974, vol. 19, p. 189-219.
- 202. N.Ya.Vilenkin, Special Functions and Theory of Group Representations, Nauka, Moscow (1965).

- 203. V.S.Vladimirov, Equations of Mathematical Physics, Nauka, Moscow (1981).
- 204. P.Voronetz in: Universit. Izvestiya, 28, No. 2 (1907).
- 205. J.Wermer, Potential Theory, Springer-Verlag (1974).
- 206. R.M.Wilcox, "Exponential operators and parameter differentiation in quantum physics," J. Math. Phys., 8, No. 4, 962-892 (1967).
- 207. G.B.Whitham, Linear and Nonlinear Waves, Mir, Moscow (1977).
- 208. G.H.Weiss and A.A.Maradudin, "The Baker-Hausdorff formula and a problem in crystal physics," J. Math. Phys., 3, No. 4, 771-777 (1962).
- 209. V.E.Zacharov and A.B.Shabat, "Exact theory of two-dimensional self-focusing and one-dimensional auto-modulation of waves in nonlinear mediums," Zhurn. Teoret. i Mat. Fiziki, 61, No. 1, 118-134 (1971).
- V.E.Zacharov, S.V.Manakov, S.P.Novikov, and L.P.Pitoevsky, Theory of Solitons. Method of Inverse Problem (ed. S.P.Novikov), Nauka, Moscow (1980).
- 211. R.Z.Zhdanov, "On application of Lie-Bäcklund method to study of symmetry properties of Dirac equation," in: Group-Theoretic Investigations of Equations of Mathematical Physics, Inst. Matematiki AN Ukr SSR, Kiev, 1985, p. 70-73.
- 212. R.Z.Zhdanov, "Theoretic-algebraic analysis and exact solutions of nonlinear spinor equations," Author's Abstract of Thesis, Inst. Matematiki AN Ukr SSR, Kiev (1987).

Additional References

- A.A.Belavin, A.M.Polyakov, A.S.Schawartz, and Yu.S.Tyupkin, "Pseudoparticle solutions of the Yang-Mills equations," Phys. Lett. B, 59, No. 1, 85-87 (1975).
- 2*. H.Boerner, Representations of Groups, North-Holland, Amsterdam (1970).
- 3*. A.V.Dorodnizin, I.V.Knyazeva, and S.R.Svirshchevski, "Group properties of the heat equation with a source in two and three dimensions," Differential Uravneniya, 19, No. 7, 1215-1224 (1983).
- 4^{*}. V.de Alfaro, S.Fubini, and G.Furlan, "A new classical solution of the Yang-Mills field equations," Phys. Lett. B, 65, No. 2, 163-166 (1976).
- 5*. V.M.Fedorchuk, "On reduction and some exact solutions of Nonlinear Equations of Mathematical Physics," Inst. Matematiki AN Ukr SSR, Kiev, 1987, p.73-76.
- 6^{*}. I.M.Fedorchuk, "On some exact solutions of the relativistic Hamilton equations," Ibid., p. 76-79.
- 7^{*}. W.I.Fushchich, "How to extend the symmetry of differential equations," Ibid., p. 4-16.
- 8*. W.I.Fushchich and I.A.Egorchenko, "On symmetry properties of nonlinear complex wave equations," Dokl. AN SSSR, 298, No. 2, 347-351 (1988).

- 9*. W.I.Fushchich, V.M.Fedorchuk, and I.M.Fedorchuk, "The subgroup structure of the generalized Poincare group and exact solutions of some nonlinear wave equations," Preprint. Inst Matematiki AN Ukr SSR, No. 86.27, Kiev (1986).
- 10*. W.I.Fushchich and N.I.Serov, "On some exact solutions of the three-dimensional nonlinear Schrödinger equation," J. Phys. A, 20, No. 16, L929-L933 (1987).
- 11*. W.I.Fushchich and R.Z.Zhdanov, "On some exact solutions of the threedimensional nonlinear differential equations for spinor and vector fields," J. Phys. A: Math. and Gen., 20, No. 13, 4173-4190 (1987).
- 12*. W.I.Fushchich and R.Z.Zhdanov, "On generalization of the method of separation of variables for systems of linear differential equations," in: Symmetry and Exact Solutions of Nonlinear Equations of Mathematical Physics, Inst. Matematiki AN Ukr SSR, Kiev, 1987, p. 17-22.
- 13^{*}. W.I.Fushchich and R.Z.Zhdanov, "On the reduction and some exact solution of the nonlinear Dirac and Dirac-Klein-Gordan equations," J. Phys. A: Math. and Gen., 21, No. 1, L5-L9 (1988).
- 14*. W.I.Fushchich and R.Z.Zhdanov, "Symmetry and some exact solutions of the nonlinear Dirac equation," Fizika Elementar. Chastits i Atom. Yadra, 19, No. 5, (1988).
- 15*. A.M.Grundland and J.A.Tuszynski, "Symmetry breaking and bifurcation solutions in the classical complex ϕ^6 field theory," J. Phys. A: Math, and Gen., No 19, 6243-6258 (1987).
- 16*. E.G.Kalnins, Separation of Variables for Riemannian Spaces of Constant Curvature, John Wiley and Sons, Inc., New York (1986).
- 17*. B.G.Konopelchenko, Nonlinear Integrable Equations, in: Lecture Notes in Physics, 270, (1987), Springer-Verlag, Berlin.
- 18*. Logunov A.A., Lecture notes on Relativity and Gravitation, The Modern Consideration of the Problem, Nauka, Moscow (1987).
- 19*. Yu.A.Mitropolsky and M.W.Schulga, "Asymptotic and exact solutions of multidimensional equation of the Schrödinger type," Ukr. Mat. Zhurn., 39, No. 6, 744-751 (1987).
- 20*. Yu.A.Mitropolsky and M.W.Schulga, "Asymptotic solutions of multidimensional nonlinear wave equation," Dokl. AN SSR, 295, No. 1, 30-33 (1987).
- 21*. C.Rogers, W.F.Shadwick, Bäcklund transformations and their Applications, Academic Press, New York (1982).
- 22*. L.H.Ryder, Quantum Field Theory, Cambridge University Press, Cambridge (1985)
- 23*. N.I.Serov, "On solution of Riccati equations" in: Algebraic-Theoretic Studies in Mathematical Physics Problems, Inst. Matematiki AN Ukr SSR, Kiev 1983, p. 59-62.
- 24*. W.M.Shtelen, "On a method of constructing exact solutions of multidimensional linear differential equations," in: Symmetry and Exact Solutions of

Nonlinear Equations of Mathematical Physics, Inst. Matematiki AN Ukr SSR, Kiev 1987, p. 31-36.

- 25*. P.Winternitz, A.M.Grundland, and J.A.Tuszynski, "Exact solutions of the multidimensional classical ϕ^6 field equations obtained by symmetry reduction," J. Math. Phys., 28, No. 9, 2194-2212 (1987).
- 26^{*}. J.Wei and E.Norman, "Lie algebraic solutions of linear differential equations," J. Math. Phys., 4, No. 4, 575-581 (1963).
- 27*. J.Wei and E.Norman, "On global representation of the solutions of linear differential equations as a product of exponentials," Proc. Amer. Math. Soc., 15, No. 2, 327-334 (1964).
- 28*. F.Wolf, "Lie algebraic solutions of linear Fokker-Planck equations," J. Math. Phys., No. 2, 305-307 (1988).
- 29*. R.Z.Zhdanov, "On nonlinear spinor equation admitting infinite-dimensional symmetry group," in: Symmetry and Exact Solutions of Nonlinear Equations of Mathematical Physics, Inst. Matematiki AN Ukr SSR, Kiev, 1987, p. 44-47.
- 30^{*}. W.I.Fushchich and R.Z.Zhdanov, "On some exact solutions of nonlinear d'Alembert and Hamilton equations," Preprint. Institute for Mathematics and Applications, University of Minnesota, 5p. (1988).
- 31^{*}. W.I.Fushchich and R.Z.Zhdanov, "Symmetry and exact solutions of nonlinear spinor equations," Phys. Reports, **172** No. 4, 123–174 (1989).
- 32*. W.I.Fushchich and Barannik L.F., "On continuous subgroups of the generalized Schrödinger group," J. Math. Phycs., 30, N1, (1989).
- 33*. Jackiw R., "Introduction in scale symmetry," Physics Today, 25, N1, 23-33 (1972).
- 34*. Kastrup H.A., "Position operators, gauge transformations and the conformal group," Phys. Rev., 143, No. 4, 1021-1028 (1966).
- 35*. Tajiri M., "Similarity reductions of the one and two dimensional nonlinear Schrödinger equation," J. of the Physical Society of Japan, 52, No. 7, 1908-1917 (1983).
- 36*. Tajiri M., "On N-soliton solutions of coupled Higgs field equation," Ibid., 52, No. 7, 2277-2280 (1983).
- 37^{*}. Tajiri M., "N-soliton solutions of the two- and three-dimensional nonlinear Klein-Gordon equations," Ibid., **53**, No. 4, 1221-1228 (1984).
- 38*. Gagnon L and P.Winternitz, "Lie symmetries of a generalized nonlinear Schrödinger equation. 1. The Symmetry group and its subgroups," J. Phys. A: Math. and Gen., 21, No. 7, 1493-1511 (1988).
- 39*. D.David, D.Levi, and P.Winternitz, "Bäcklund transformations and infinitedimensional symmetry group of the Kadomtsev-Petviashvili equation," Phys. Lett., B, 118, No. 8, 380-384 (1986).
- 40^{*}. T.Cazenave and F.Weissler, "The Cauchy problem for the nonlinear Schrödinger equation in H," Mannuscr. Math., **61**, No. 3, 477-494 (1988).
- 41^{*}. T.Cazenave, "Uniform estimates for solutions of nonlinear Klein-Gordon equations," J. Funct. Anal., **60**, No. 1, 36-55 (1985).

References

- 42*. W.I.Fushchich and I.A.Egorchenko, "Differential invariants of the Galilei group," Dokl. AN Ukr SSR, No. 4 (1989). Ser. A. 29–32.
- 43*. W.I.Fushchich and I.A.Egorchenko, "Differential invariants of the Poincare group," Dokl. AN Ukr SSR, No. 5 (1989), Ser. A. 46-53.
- 44*. W.I.Fushchich, N.I.Serov, and V.I.Chopik, "Conditional invariance and nonlinear heat equations," Ibid., Ser. A., No. 9, 17-20 (1988).
- 45*. W.I.Fushchich and N.I.Serov, "Conditional invariance and exact solutions of nonlinear acoustic equation," Ibid., Ser. A., No. 10, 27-31 (1988).
- 46*. Symmetry analysis and solutions of equations of mathematical physics, Kiev: Inst. of Mathematics (1988).
- 47*. A.A.Samarsky, V.A.Galaktionov, S.P.Kurdumov, and A.P.Mikhailov, Regime with Sharpening in Quasilinear Parabolic Problems, Nauka, Moscow (1987).
- 48*. R.K.Dodd, J.C.Eilbeck, J.D.Gibbon, and H.C.Morris, Solitons and Nonlinear Wave Equations, Academic Press Inc., London (1984).
- 49*. V.A.Marchenko, Nonlinear Equations and Operator Algebras, Naukova Dumka, Kiev (1986).
- 50*. V.P.Maslov, V.H.Danilov, and K.A.Volosov, Mathematical Analogue of Heat and Mass Conduction. Evolution of Dissipative Structures, Nauka, Moscow (1987).
- 51*. V.A.Yatsun, "O(4)-invariant solutions of Yang-Mills theory with scalar field," Teoret. i Mat. Fizika, 68, No. 3, 392-400 (1986).
- 52*. V.A.Yatsun, "On quasi-self-dual fields in N = 4 supersymmetric Yang-Mills," Lett. Math. Phys., 15, No. 1, 7–11 (1988).
- 53*. H.D.Doebner, J.D.Henning (Eds.) Differential Geometric Methods in Theoretical Physics. Proceedings of the 15-th International Conference, 1987. Singapore, World Scientific.
- 54*. E.A.Tagirov, I.T.Todorov, "A geometric approach to the solutions of conformal invariant nonlinear field equations," Acta Physica Austriaca, 51, No. 2, 135-148 (1979).
- 55*. J.Sniatycki, "Geometric quantization and quantum mechanics," Berlin, Springer-Verlag (1980).
- 56^{*}. N.I.Serov, "Conditional invariance and exact solutions of nonlinear heat equation," Ukr. Math. J., **42** No. 10, 1370–1376 (1990).
- 57*. W.I.Fushchich and V.I.Chopik, "Conditional invariance of the nonlinear Schrödinger equation," Dokl. AN USSR (Ukraine), No. 4, 30–33 (1990).
- 58*. W.I.Fushchich and V.I.Chopik, and P.I.Mironyuk, "Conditional invariance and exact solutions of three-dimensional nonlinear acoustic equations," Ibid., No. 9, 25–28 (1990).
- 59*. W.I.Fushchich, "On a generalization of Lie's method," in: Theoretic-Algebraic analysis of equations of mathematical physics, Kiev: Institute of Mathematics, 1990, p.4–11.
- 60*. W.I.Fushchich, R.Z.Zhdanov, and I.V.Revenko, "Compatibility and solutions of nonlinear d'Alembert and Hamilton equations," Preprint 90.30: Institute of Mathematics, Kiev, 1990, p. 65.

- 61^{*}. W.I.Fushchich and W.M.Shtelen, "Merons and Instantons as products of selfinteraction of the Dirac-Gürsey spinor field," J. Phys. A: Math. Gen. 23, No. 10, L.517-520 (1990).
- 62*. W.I.Fushchich and W.M.Shtelen, "On approximate symmetry and approximate solutions of nonlinear wave equation with a small parameter," Ibid., 22, No. 16, L887-L890 (1989).
- 63*. W.I.Fushchich, W.M.Shtelen, and S.L.Slavutsky, "Reduction and exact solutions of the Navier-Stokes equations," Ibid., 24, No. 5, 971–984 (1991).
- 64^{*}. W.I.Fushchich, W.M.Shtelen, and S.L.Spichak, "On the connection between solutions of Dirac and Maxwell equations, dual Poincare invariance and superalgebra of invariance solutions of nonlinear Dirac equations," Ibid., 24, 1683–1698 (1991).
- 65*. W.M.Shtelen, "On connection between solutions of Dirac and Maxwell equations," In: Symmetry and solutions of equations of mathematical physics, Kiev, Institute of Mathematics, 110–113 (1989).
- 66*. N.H.Ibragimov, "On invariance of Dirac equation," Dokl. AN USSR, 185, No. 6, 1225–1228 (1969).
- 67*. K.Ljoje, "Some remarks on variational formulations of physical fields," Fortschr. Phys., **36**, No. 1, 9-32 (1988).
- 68*. W.I.Fushchich and R.M.Cherniha, "Galilei-invariant nonlinear equations of Schrödinger type and their exact solutions I," Ukrain. Math. J. 41, No. 10, 1349–1357 (1989).
- 69*. W.I.Fushchich and R.M.Cherniha, Id. II, Ibid., 41, No. 12, 1687-1694 (1989).
- 70*. W.I.Fushchich and A.G.Nikitin, "On new symmetries and conservations laws of elastic wave equation," Dokl. AN USSR, 304, No. 2, 333-335 (1989).
- 71^{*}. P.Sorba, "The Galilei group and its connected subgroups," J. Math. Phys., 17, No. 6, 941–953 (1976).
- 72*. W.I.Fushchich, W.M.Shtelen, and R.E.Popovich, "On reduction of the Navier-Stokes equations to linear heat equations," Dokl. AN Ukraine, No. 2, 23-30 (1992).
- 73^{*}. R.Z.Zhdanov, "Symmetry and exact solutions of nonlinear Galilei-invariant equations for spinor filed," Ukrain. Math. J., 43, No. 4, 496-503 (1991).
- 74^{*}. S.P.Lloyd, "The infinitesimal group of the Navier-Stokes equations," Acta Mechanica, **38**, No. 1–2, 85–98 (1981).
- 75*. H.Bateman and A.Erdelyi, "Higher transcendental functions," vol. 3, Mc-Graw-Hill: New York p. 299 (1955).
- 76*. W.M.Shtelen and V.I.Stogny, "Symmetry properties of one- and two-dimensional Fokker-Planck equations," J. Phys. A: Math. Gen., 22, L539-543 (1989).
- 77*. C.W.Gardiner, Handbook of Stochastic Methods, Berlin: Springer (1985).
- 78*. M.Kimura, "Diffusion Models in population genetics," J. Appl. Probab., 1, No. 2, 178-230 (1964).
- 79*. M.Suzuki, "Decomposition formulas of exponential operators and Lie exponentials with some applications to quantum mechanics and statistical

physics," J. Math. Phys., 26, No. 4, 601-612 (1985).

- 80*. G.Cicogna and D.Vitali, "Generalized symmetries of Fokker-Plank-type equations," J. Phys. A: Math.Gen., 22, No. 11, L453-456 (1989).
- 81*. W.M.Shtelen, "On group method of linearization of Burger's equation," Math. Phys. and Nonlinear Mechanics (Kiev), 11(45), 89-91 (1989).
- 82*. M.M.Serova and N.I.Serov, "Nonlinear Schrödinger equation for particle with variable mass invariant under the Galilei group," Depon. VINITI 03.11. 1986, No. 7517-B86.
- 83*. W.I.Fushchich and N.I.Serov, "Conditional symmetry and exact solutions of the Boussinesq equation," in: "Symmetry and exact solutions of equations of mathematical physics," Kiev: Institute of Mathematics, 95–102 (1989).
- 84*. P.A.Clarkson and M.D.Kruskal, "New similarity reductions of the Boussinesq equation," J. Math. Phys., 30, No. 10, 2201-2213(1989).
- 85*. D.Levi and P.Winternitz., "Non-classical symmetry reduction: example of the Boussinesq equation," J. Phys. A: Math.Gen., 22, No. 15, 2915–2924 (1989).
- 86*. W.I.Fushchich and R.Z.Zhdanov, "On some new exact solutions of the nonlinear d'Alembert-Hamilton system," Phys. Lett. A, 141, No. 3-4, 113-115 (1989)
- 87*. W.I.Fushchich and A.G.Nikitin, "Symmetry of Equations of Quantum Mechanics," Moscow: Nauka, p. 400 (1990).
- 88*. W.M.Shtelen, "On integrodifferential nonlinear equations for scalar field," In: Symmetry analysis and solutions of equations of mathematical physics," Kiev: Institute of Mathematics, 22–26 (1988).
- 89*. W.M.Shtelen, "On solutions of the Schrödinger equation invariant under the Lorentz algebra," In: Theoretic-algebraic analysis of equations of mathematical physics, Kiev: Institute of Mathematics, 109-112 (1990).
- 90*. W.I.Fushchich and N.I.Serov, "Conditional symmetry and reduction of nonlinear heat equation," Dokl. AN Ukrain.SSR, No. 7, 24-28 (1990).
- 91*. W.I.Fushchich, N.I.Serov, and T.K.Amerov, "Conditional invariance of the heat equation," Ibid., No. 11, 16-21 (1990).
- 92*. W.I.Fushchich, N.I.Serov, and T.K.Amerov, "Conditional symmetry of generalized Korteweg-de Vries equation," Ibid., No. 12, 20-23 (1991).
- 93*. W.I.Fushchich and P.I.Mironyuk, "Conditional symmetry and exact solutions of nonlinear acoustic equations," Ibid., No. 6, 12–16 (1991).
- 94*. W.I.Fushchich, N.I.Serov, and V.K.Repeta, "Conditional symmetry and exact solutions of nonlinear wave equation," No. 5, 29-34 (1991).
- 95*. W.I.Fushchich, N.I.Serov, and V.K.Repeta, "Exact solutions of one-dimensional equations of gas dynamics and nonlinear acoustics," No. 8, 15–20 (1991).
- 96^{*}. W.I.Fushchich, N.I.Serov, and V.K.Repeta, "Non-Lie symmetry and exact solutions of one-dimensional equations of gas dynamics," No. 11, (1991).
- 97*. R.Z.Zhdanov and A.Yu.Andreyzev, "On non-Lie reduction of Galilei-invariant spinor equations," Ibid., No. 7, 8-11 (1990).

- 98*. W.I.Fushchich and R.Z.Zhdanov, "Non-Lie ansatze and exact solutions of nonlinear spinor equation," Ukrain. Math. J., 42, No. 7, 958-962 (1990).
- 99*. M.M.Serova, "Conditional invariance of the Boussinesq equation under the Galilei algebra," in: Symmetry analysis and solutions of equations of mathematical physics, Kiev: Institute of Mathematics, 92–94 (1988).
- 100*. W.I.Fushchich and N.I.Serov, "On conditional invariance of d'Alembert, Liouville, Born-Infeld, and Monge-Ampere equations under the conformal algebra," Ibid., 98–102.
- 101*. M.M.Serova, "On some conditionally invariant solutions of equations of mathematical physics," Kiev: Institute of Mathematics, 71-73 (1989).
- 102*. N.I.Serov, "On some conditionally invariant solutions of the Born-Infeld equations," Ibid., 74-76.
- 103*. V.I.Chopik, "Conditional invariance of the Schrödinger equation with a real nonlinearity," Ibid., 108-109.
- 104*. T.K.Amerov, "On conditional invariance of nonlinear heat equation," in: Theoretic-algebraic analysis of equations of mathematical physics, Kiev: Institute of Mathematics, 12-14 (1990).
- 105*. V.K.Repeta, "On some conditionally invariant solutions of nonlinear wave equation," Ibid., 61-64.
- 106*. V.I.Chopik and P.I.Mironyuk, "Conditional invariance and reduction of the two-dimensional Khohlov-Zabolotskaya equation," Ibid., 107–109.
- 107*. N.I.Serov, "Conditional invariance of the nonlinear polywave equation under the conformal algebra," Ibid., 65–66.
- 108*. W.I.Fushchich and N.I.Serov, "Conditional symmetry of nonlinear wave equation," Ukrain. Math. J., (1992).
- 109*. W.I.Fushchich and I.A.Yegorchenko, "Non-Lie ansatze and conditional symmetry of nonlinear Schrödinger equation," Ibid., N12, 1620–1628 (1991).
- 110*. W.I.Fushchich and I.A.Yegorchenko, "The symmetry and exact solutions of the nonlinear d'Alembert equation for complex fields," J. Phys. A: Math. Gen., 22, 2643-2652 (1989).
- 111*. W.I.Fushchich and W.M.Shtelen, "Are Maxwell's equations invariant under the Galilei transformations?" Dokl. AN Ukrain.SSR, No. 3, 23-27 (1991).
- 112*. W.I.Fushchich, A.S.Galizin, and A.S.Polubinsky, "New mathematical model of diffusion processes with final rate," Ibid., No. 8, 22–27 (1988).
- 113*. I.O.Parasyuk and W.I.Fushchich, "Qualitative analysis of bounded solutions of nonlinear three-dimensional Schrödinger equation," Ukrain. Math. J., 42, No. 10, 1344–1349 (1990).
- 114*. W.I.Fushchich and V.V.Kornyak, "Computer algebra applications for determining Lie and Lie-Backlund symmetries of differential equations," J. Symbolic Computation, 7, 611-619 (1989).
- 115*. W.I.Fushchich and Yu.N.Segeda, "On new invariance algebra of the free Schödinger equation," Dokl. AN USSR, 232, No. 4, 800-801 (1977).
- 116*. W.I.Fushchich and I.O.Parasyuk, "Qualitative analysis of families of constrained solutions of the multidimensional nonlinear Schrödinger equation,"

Ukrain. Math. J., 43, No. 6, 821-829 (1991).

- 117*. Yu.A.Mitropolsky and M.W.Shulga, "Approximate symmetry of the nonlinear heat equation," Ibid., 43, No. 6, 833-837 (1991).
- 118*. W.I.Fushchich, V.A.Tychinin, and N.I.Serov, "Formula of generation of solutions of the Korteweg-de Vries equation," Ukrain. Math. J., No. 5, (1992).
- 119*. W.I.Fushchich and V.A.Tychinin, "Exact solutions and principle of superposition for nonlinear wave equation," Dokl. AN Ukrain.SSR, A, No.5, 32–36 (1990).
- 120*. C.Rogers and W.F.Ames, Nonlinear Boundary Value Problems in Science and Engineering, Acad. Press Boston (1989).
- 121*. P.L.Sachdev, Nonlinear Diffusive Waves, Cambridge University Press (1987).
- 122*. W.I.Fushchich, R.Z.Zhdanov, and I.A.Yegorchenko, "On the reduction of the nonlinear multi-dimensional wave equations and compatibility of the d'Alembert-Hamilton system," J.Math.Anal.Appl., 161, No.2, 352-360 (1991).
- 123*. W.H.Steeb and J.A.Louw, "Parametrically driven Sine-Gordon equation and Painleve test," Phys.Lett., A113, 61-62 (1985).
- 124*. R.Sechadri and T.Y.Na, Group Invariance in Engineering Boundary Value Problems, Springer-Verlag, New York, 1985.
- 125*. D.David, N.Kamran, D.Levi, and P.Winternitz, "Symmetry reduction for the Kadomtsev-Petviashvili equation using a loop algebra," J.Math.Phys., 27 1225-1237 (1986).
- 126^{*}. W.H.Steeband and N.Euler, Nonlinear Evolution Equations and Painleve Test, World Scientific Publ. Co., Singapore (1988).
- 127*. W.I.Fushchich, N.I.Serov, and T.K.Amerov, "On nonlocal ansatze for a nonlinear one-dimensional heat equation," Dokl. AN Ukraine, No. 1, 26–30 (1992).
- 128*. I.A.Yegorchenko and A.I.Vorobieva, "Conditional symmetry and exact solutions of the Klein-Gordon-Fock equation," Ibid., No. 1, 19–22 (1992).
- 129*. W.I.Fushchich, "Conditional symmetry of equations of nonlinear mathematical physics," Ukrain. Math. J., No. 11, 1456-1470 (1991).
- 130*. W.I.Fushchich, R.Z.Zhdanov, and I.V.Revenko, "The general solution of coupled nonlinear wave and eikonal equations," Ibid. No. 11, 1471–1487 (1991).
- 131^{*}. G.W.Bluman and J.D.Cole, "The general similarity solution of the heat equation," J. Math. Mechan., 18, No. 11, 1025-1042 (1969).
- 132*. G.M.Webb, "Lie symmetries of coupled nonlinear Burgers-heat equation system," J. Phys. Math.: Math. Gen., 23, No. 17, 3885–3894 (1990).
- 133*. G.Rosen, "Nonlinear heat equation in solid H_2 ," Phys. Rev. B, 19, No. 4, 2398–2399, (1979).
- 134^{*}. G.W.Bluman and S.Kumei, "On the remarkable nonlinear diffusion equation $\frac{\partial}{\partial x} \left[a(u+b)^{-2} \frac{\partial u}{\partial x} \right] - \frac{\partial u}{\partial t} = 0,$ " J. Math. Phys., **21**, No. 5, 1019–1023 (1980).
- 135*. J.R. King, "Some non-local transformations between nonlinear diffusion equation," J. Phys. A: Math. Gen. 23, 5441-5464 (1990).

- 136^{*}. W.I.Fushchich and R.Z.Zhdanov, Nonlinear Spinor Equation: Symmetry and Exact Solutions, Naukova Dumka, Kiev (1992).
- 137*. W.I.Fushchich and R.Z.Zhdanov, "On the non-Lie reduction of the nonlinear Dirac equation," J. Math. Phys., **32**, No. 11, 3315–3318 (1991).
- 138^{*}. R.Z.Zhdanov and W.I.Fushchich, "On non-Lie symmetry of a Galilei invariant equation for particle with spin s = 1/2," Teoret. i Mat. Fizika, 89, No. 3, 413-419 (1991).
- 139*. W.I.Fushchich, L.F.Barannik, and A.F.Barannik, Subgroup Analysis of the Galilei and Poincare Groups and Reduction of Nonlinear Equation, Naukova Dumka, Kiev (1991).
- 140*. W.I.Fushchich, "The symmetry and exact solutions of multi-dimensional nonlinear spinor equations," Proceedings of the annual seminar of the Canadian Math. Soc.: Lie theory, differential equations and representation theory, 161-170 (1989).
- 141*. A.F.Barannik, V.A.Marchenko, and W.I.Fushchich, "On reduction and exact solutions of the nonlinear Schrödinger equation," Teoret. i Mat. Fizika, 87, No. 2, 220-233 (1991).
- 142*. M.Torrisi and M.V.Lalicata, "Some similarity solutions for hyperelastic half space with finite deformations," Inter. Jour. Non-linear Mechanics, 26, No. 1, 15-23 (1991).
- 143^{*}. W.I.Fushchich, "New nonlinear equations for electromagnetic field having the velocity different from c," Dokl. AN Ukraine, No. 4, 16–19 (1992).
- 144*. J.Beckers, L.Gagnon, V.Hussin, and P.Winternitz, "Nonlinear differential equations and Lie superalgebras," Lett. Math. Phys., 13, No. 2, 113-120 (1987).
- 145*. F.Magri, "Equivalence transformations for nonlinear evolution equations," J. Math. Phys., 18, No. 7, 1405-1411 (1977).

Index

Acoustic nonlinear equations 333, 345

- anstatz xiv, 16, 18, 56
 - conformally invariant -87
 - E(1, 3)-inequivalent—for scalar field 150, 151
 - G(1,3)-inequivalent—205
 - G(1,3)-inequivalent—codimension 1 for spinor field 255
 - G(1,3)-inequivalent—codimension 1 for the Navier-Stokes field 261
 - P(1, 2)-inequivalent—for scalar field 20, 21
 - P(1,3)-inequivalent—for spinor field 58; in covariant form 66
 - P(1,3)-inequivalent—for vector field 69
 - P(1, 3)-inequivalent—for spinor field 71
 - Sch(1,3)-inequivalent-201
 - Sch(1,3)-inequivalent—for a complex scalar field 177, 182
- approximate Galilei invariance 355 approximate symmetry 360

Backlund transformations 278 basis elements of AE(1,3) for vector field 250 AC(1,3) for field of arbitrary spin 79 A $\widetilde{E}(1,n)$ for scalar field 147, 148, 349 AG(1, n) for scalar field 164, 165

AG(1,3) for field of arbitrary 200, 208-212, 216, 217 spin AG(1,3) for spinor field 254AG(1,3) for Navier-Stokes field 260 $A\widetilde{P}(1, n)$ for scalar field 1, 4 AP(1, n) for vector field 52AP(1,3) for spinor field 57ASch(1,3) for field of arbitrary spin 200ASch(1,3) for scalar field 163, 167 ASch(1,3) for spinor field 228Bernoulli equation 45, 153, 161 Bernoulli numbers 301 68, 138, 269 Bessel equation ---functions 68, 77, 138, 269, 358 bispinor densities 92 -(Fierz-Pauli) identities 93 Born-Infeld equation 121, 332 Boussinguesq equation 188, 340, 347 Burger's equation 278 - 280Campbell-Baker-Hausdorf (CBH) formula 300, 368, 373 characteristic function 8 classical electrodynamics 117

commutant series 382 conditional symmetry xvi, xvii,

326 Y-conditional invariance 328 conformal degree 79, 80—operator 10, 89–scalar field; 89–spinor field; 52–vector field; 79–arbitrary spin field; 130–Yang-Mills field

conformal symmetry (invariance) xi, 6, 10, 13, 14, 27, 78 conformal transformations xvi, 27, 78, 83(field of arbitrary spin); 84 (spinor f.); 85 (scalar f.); 85 (vector f.); 85, 86 (tensor f.); 130 (Yang-Mills f.) contact symmetry 7–10 -of eikonal equations 9, 10 -of Hamilton-Jacobi equations 186 -transformations 7-9, 186, 187 **Darboux** equation 231 nonlinear-25, 158 Dirac equation xi, 55, 68, 146, 289, 306, 310, 357 massless-89, 141 nonlinear-55, 70, 348 Dirac-Gürsey equation 56, 80, 93, 96, 135 Dirac-Heisenberg equation 56 generalized-93, 95 Dirac-Heisenberg-Thirring equation 112 Dirac matrices 55 differential invariants of AP(1, n)387; AP(1, n)387; AC(1,n)388; AG(1, n)389: AG(1,n)389; ASch(1, n)389 dual Poincare invariance 141, 145, 146 dual spacetime symmetry 307, 356 Eikonal equations xi, 3, 6, 10, 35, 80, 332, 346 elliptic functions 28, 34, 126, 135, 138, 364 Emden-Fowler equation 22, 23, 178 Euler eq. 46 Euler-Lagrange-Born-Infeld (ELBI) eq. 5, 40, 282, 332, 347 nonrelativistic counterpart of-188 Equation(s)

Abel of second kind 140 d'Alembert-Hamilton xvii, 17, 67, 344, 347, 361, 362, 393 accoustic 333, 345 Bernoulli 45, 153, 161 Bessel 68, 138, 269 Born-Infeld 121, 332 Boussinesq 188, 340, 347 Burger's 278 - 280classical electrodynamics 117 continuity 80 231, 25, 158 (nonlinear) Darboux xi, 55, 68, 146, 298, 306, Dirac 310, 357 massless-89, 141 nonlinear-55, 70, 348 Dirac-Gürsey 56, 80, 93, 96, 135 generalized-80, 96 Dirac-Heisenberg 56 generalized 93.95 Dirac-Heisenberg-Thirring 112eikonal xi, 3, 5, 10, 35, 80, 346 Emden-Fowler 22, 23, 178 Euler 46 Euler-Langrage-Born-Infeld (ELBI) 5, 40, 187, 282, 332, 347 nonrelativistic-188 Fokker-Planck 193, 383 gas dynamics 229, 338, 348 ---for isochoric process 241-for polytropic process 235Hagen-Herley 211nonlinear generalization 215Hamilton xi, 3, 5, 35, 80 Hamilton-Jacobi xi, 164, 165, 182, 249, 274, 330, 337, 338 heat x, 163, 172, 181, 194, 195, 197, 275, 278, 328, 358, 385 163, 328-331, 347 nonlinear Helmholtz 110 integrodifferential 319 Kadomtsev-Petviashvili 347 Khohlov-Zabolotskaya 345

Index

Killing 6, 54 Klein-Gordan 356 Korteweg-de Vries 283, 286 generalized 347 Kramers 195 Lagrange-Euler 18, 59, 150 Lame 128, 250, 335 Laplace 78, 274, 330, 338 Legendre $\mathbf{26}$ Levi-Leblond 210, 215, 225, 258 Lie 8, 186, 299 linear xiii, xiv, 88, 297, 328 Liouville 3, 28, 280, 344 Maurer-Cartan 379, 380 Maxwell x, xi, xvi, 80, 90, 91, 117, 141, 298, 353 -for vector potential 80, 90, 126, 131 Monge-Ampere xi, 5, 47, 174, 187, 282, 344 Navier-Stokes 260, 339 Ornstein-Uhlenbeck 193 Penleve 34 Poisson 141 polywave 10, 80 Proca (Galilei invariant) 210poseudodifferential 14 quantum electrodynamics 102, 111 two-dimensional—112, 115 quasirelativistic 348 Riccati 152, 154, 155, 159, 283, 284Schrödinger 167, 199, 358 nonlinear 168-170, 176, 348 Van der Pol 323 wave x, 10, 89, 142, 181, 252, 362 nonlinear—xv, xvii, 1, 3, 16, 21, 32, 80, 96, 133, 134, 277, 345, 346, 348, 360, 366 with interaction 51Weyl 80, 91 Whittaker 269 - 272

Yang-Mills 96, 129 extended group: Euclidean 25, 147; Galilei 200; Poincare 16, 70 Fierz-Pauli identities 93 Fokker-Planck equation 193, 383 "fusion" method 56 Galilei algebra 164, 165 (scalar representation); 200, 208-212, 216, 217 (arbitrary spin representation); 254 (spinor representation) Galilei invariant PDEs 163, 215; 254 (for spinor field) Galilei generalization of Dirac equation 223Galilei relativistic principle 48, 120, 169Galilei transformations 202, 213 Galilei nonlocal transformations 307 (for Dirac equation); 355 (for Maxwell eq.); 351 (for quasirelativistic eq.) gas dynamics equations 229, 337, 347 ---for isochoric process 241-for polytropic process 235gauge group SU(2)130 gauge transformations for SU(2)Yang-Mills field 131 generating solution xv, 18, 19, 64 (by Poincare transformations); 27, 83-86, 131 (by conformal transform.); 202 (by Schrödinger group transform.); 273 (for the Navier-Stokes field); 283 (nonlocal for KdV equation) Hagen-Herley equation 211; 215(nonlinear generalization) Hamilton eq., xi, 3, 5, 35, 80 Hamilton-Jacobi eq. xi, 164, 165, 182, 249, 274, 330, 337, 338 Hausdorff formula 301, 306, 308,

310, 350 -continuous analog 382 Hausdorff operator 301 x, 163, 172, 181, heat equation 194, 195, 197, 275, 278, 328, 358, 385: nonlinear 163, 328 - 331,347 Heisenberg ansatz 56, 67, 98 Helmoltz equation 110 hodograph transformation 278Hopf-Cole transformation 278t'Hooft-Corrigan-Fairlie-Wilczek ansatz 96, 133, 134 Ideal 383 infinitesimal operator (IFO) xii. xiii, 2 infinite-dimensional invariance group of one-dimensional gas dynamics eq. 231 integral Fourier transformation 14, 358integrodifferential equation 319 invariance condition xii-xiv. 326 - 328invariant variables of $\tilde{P}(1,2)$ 20; P(1,3) 58, 66; $\tilde{P}(1,3)$ 71; Sch(1,3) 182; G(1,3) 261, 262; G(1,3) 255; E(1,3) 151 inversion 78 inverse scattering method x, xvii Jacobi elliptic functions 28, 34, 126, 135, 138, 364 Jacobi identity 376 Kadomtsev-Petviashvili eq. 347 Kelvin transformation 78 Khohlov-Zabolotskaya eq. 345 Killing eq. 6, 54 kink 128 Klein-Gordon eq. 356 Korteweg - de Vries eq. 283 - 286;

generalized 347 Kramers eq. 195 Krilov-Bogolubov-Mitropolski method 324 Lagrange-Euler eq. 18, 59, 150 Lame eq. 128, 250, 335 Lame functions 128, 342 Laplace eq. 78, 274, 330, 338 Legendre eq. 26Levi-Leblond eq. 210, 215, 225, 258 Lie algebra vii, 376 Lie algorithm (method) x-xiv Lie equations 8, 186, 299 Lie's theorems 376-381 Lie-Backlund symmetry 312, 328 linear PDEs xiii, xiv, 88, 297, 328 3, 28, 280, 344 Liouville ea. Markovian processes 193 379, 380 Maurer-Cartan equations maximal invariance group (algebra) in Lie's sense xi, xiii; 189-191 of Boussinesq eq.; 278 of Burger's eq.; 144 of Dirac massless eq.; of Dirac eq. 57, 306; of eikonal eq. 6, 10; of ELBI eq. 41; of Fokker-Planck eq. 193; of gas dynamics eq. 235, 241; of Hamilton-Jacobi eq. 183; of heat eq. 163, 278; of Kramers eq. 195; of Lame eq. 250; of Liouville eq. 29, 30; of Maxwell eq. xvi, 327; of Monge-Ampere eq. 47; of Navier-Stokes eq. 263; of Ornstein-Uhlenbeck eq. 193; of quasirelativistic eq. 349; of Schrödinger eq. 167; of wave eq. 10 Maxwell equations x, xi, xvi, 80, 90, 91, 117, 141, 298, 353 -for vector potential 80, 90, 126,131 26, 134 meron 47 Minkowsky problem

Monge-Ampere eq. xi, 5, 47, 174, 187, 282, 344 Monge transformations 282, 283 **Navier-Stokes** equations 260, 339 non-Abelian solutions 136 274, 328, 340 non-Lie ansatz non-Lie method xi, 250, 298, 310 nonlocal linearization 277nonlocal transformations 278, 297, 307, 350, 353-357 Operator of full differentiation (total derivative operator) xii Ornshtein-Uhlenbeck equation 193 Padé approximants 363 Pauli matrices 55Penleve eq. 34 Poincare algebra 2, 369 Poincare group xi, 64 -transformations 64 Poisson eq. 141 polywave eq. 10.80 Proca Galilei invariant eq. 210projective Galilean transformations 169 projective transformations 199, 202, 213 prolongation of operator xii, 2 pseudodifferential equation 14 pseudodifferential operator 14, 358 quantum electrodynamics equation 102, 111 two-dimensional-112, 115 quasirelativistic equation 348 Rayleigh process 193 **Riccati** equation 152, 154, 155, 159, 283, 284 Riemannian invarints 230 - 231

Separation of variables xv, xvi, 110 15, 91, 199, scale transformations 202200Schrödinger algebra Schrödinger equation 167, 199, 357 nonlinear-168-170, 176, 348 Schrödinger group 199, 200, 202 solitary waves 128 383 solvable Lie algebra splitting conditions 17, 57 376 structure constants subalgebras of AP(1,3) 58; of AP(1,3) 71, 368; of AG(1,3) 205, 255; of AG(1,3) 261, 262; of ASch(1,3) 181, 201 superalgebra 145, 252, 385 supergroup 253, 386 symmetry analysis vii three-body problem 286 topological index 135 total derivative operator xii ungenerative solutions xv, 19, 88, 94, 95, 104, 204, 226 unitary field theory 56 323 Van der Pol equation

wave equation x, 10, 89, 142, 181, 252, 362
nonlinear—xv, xvii, 1, 3, 16, 21, 32, 80, 96, 133, 134, 277, 345, 346, 348, 360, 366
—with interaction 51
Weyl eq. 80
Weyl field 91
Whittaker eq. 269–272

Yang-Mills equations 80, 96, 129